



# On the Boltzmann equation, Quantitative studies and hydrodynamical limits



Marc Briant

Homerton College Department of Pure Mathematics and Mathematical Statistics University of Cambridge

> A thesis submitted for the degree of Doctor of Philosophy

> > September, 2014

## Statement of Originality

I hereby declare that my dissertation entitled "On the Boltzmann equation, quantitative studies and hydrodynamical limits" is not substantially the same as any that I have submitted for a degree or diploma or other qualification at any other University. I further state that no part of my dissertation has already been or is concurrently submitted for any such degree of diploma or other qualification.

This dissertation is the result of my own work and includes nothing which is the outcome of work done in collaboration except where specifically indicated in the text.

Chapter 1 introduces the work done in the present dissertation. It contains some elementary properties about the Boltzmann equation and mostly summarises the results presented later on.

Chapter 2 studies quantitatively the strict positivity of solutions to the Boltzmann equation. It is original work and has been submitted. Its arXiv link is: http://arxiv.org/abs/1302.1755.

Chapter 3 contains a detailed presentation of Hilbert's  $6^{th}$  problem and a brief state of the art. It is mainly to motivate the studies made in the following two chapters.

Chapter 4 rigorously derives the incompressible Navier-Stokes limit of the Boltzmann equation with constructive arguments, in Sobolev spaces with exponential weight. It is original work and has been submitted. Its arXiv link is: http://arxiv.org/abs/1207.4379.

Chapter 5 extends the results obtained in the previous chapter to Sobolev spaces with polynomial weight. It is original work in collaboration with S. Merino-Aceituno and C. Mouhot.

Chapter 6 establishes a Cauchy theory on a quantic version of the Boltzmann equation and discusses the phenomenon of Bose-Einstein condensate. It is original work and its arXiv link is: http://arxiv.org/abs/1310.7220.

### Acknowledgements

I would like to first thank my supervisor Clément Mouhot for all the interesting and fruitful projects he presented to me. I am really grateful for all the pieces of advice he gave me that helped improve my research and all the ones he gave me but that I did not understand. It was an honour to start my career as his student.

I thank Mihalis Dafermos and François Golse for accepting to be the examiners of my thesis. I also would like to express my gratitude to my former professor Guillaume Roussel who played such an important role in my love of mathematics and Denis Matignon and Grégoire Casalis without whom I would have never taken this path. I am also grateful towards Deborah Wilde who helped me walk it.

I thank all my CCA and CMS friends who made my everyday's life so great and with whom I have now so many memories: Damon, Ludovic, Kostas, Bati, Julio, Nayia and Ed as well as Meline and Marion. I want to express special thanks to La Sarita, Kolyan and Amit for all the discussions and the fun we had and most of all for their frienship and all that we have built together. Good luck to all of you and see you during future conferences.

My life in Cambridge would have not be the same without my crazy handball friends who were always amazing (during and after trainings) and also offered me the European dream on my last year: Agi, Tiburón, Lionel, Cédric Petardito, Big Daddy, Peter, La Moritzalidad and all the first and second team I do not have the space to name here. I will miss you all.

Of course, I thank all my friends who were not in Cambridge but the distance between us was only physical: Khala, mon Rémi, mon Adri, mon Druggite, mon Jérèm and mes Alex. I am not forgetting my friends from Monta, Bordeaux and Toulouse; unfortunately there is not enough space to name you all but I want to let you know that you make my life an awesome place to be.

I now want to express very special thanks to my mom, my dad and mon P'tit Du Pli. You have always surrounded me with your support and your love and I always felt you so close to me. You have helped me all my life to become the person I am now. I thank you for all.

I also thank my family, especially my grandparents who have always been so fond of what I was doing and provided such a great environment to grow up in, as well as my aunts, uncles and cousins. Also to all who are not legally part of it but are in my heart: Marraine, Nessica, Fabien, Bastien, Audrey, Damien and their little wonders. You are all so important to me.

Finally, I would like to thank Stéphanie who walked by my side all along this work, boosting up my joys and being an unfailing warm and bright light when I was in darkness. Your understanding, your support and your love helped me through this PhD and the present achievement owes a lot to you.

On the Boltzmann equation, Quantitative studies and hydrodynamical limits

Marc Briant

# Contents

Intr	oduction	13
1.1	General presentation of the Boltzmann equation	15
1.2	Quantitative study of the positivity of solutions (Part $I$ )	27
1.3	The incompressible Navier-Stokes limit of the Boltzmann equation (Part $II$ )	31
1.4	A quantic version of Boltzmann equation (Part <i>III</i> )	43
App	endices	46
1.A	Notations	46
	<b>Intr</b> 1.1 1.2 1.3 1.4 App 1.A	Introduction1.1General presentation of the Boltzmann equation1.2Quantitative study of the positivity of solutions (Part I)1.3The incompressible Navier-Stokes limit of the Boltzmann equation (Part II)1.4A quantic version of Boltzmann equation (Part III)1.4Notations

# IQUALITATIVE AND QUANTITATIVE STUDY OF SOLUTIONSTO THE BOLTZMANN EQUATION49

2	Inst	tantaneous filling of the vacuum		
	2.1	Introduction	54	
	2.2	Main results	60	
	2.3	The free transport equation: proof of Theorem 2.2.1 $\ldots$	68	
	2.4	The cutoff case: localized "upheaval points"	82	
	2.5	The cutoff case: characteristics passing by a point far from the boundary $\ .$	90	
	2.6	The cutoff case: geometry and grazing trajectories	94	
	2.7	Maxwellian lower bound in the cutoff case: proof of Theorem 2.2.3 $\ldots$	107	
	2.8	Exponential lower bound in the non cutoff case: proof of Theorem 2.2.6 $\ .$ .	110	

## II THE HYDRODYNAMICAL LIMITS OF THE BOLTZMANN EQUATION 117

3	Fro	m many-body problems to physics of continua	119
	3.1	Particles, gases and fluids in physics	. 121
	3.2	Hilbert's sixth problem: the mathematical coherence of models	. 124
	3.3	A brief description of the following chapters	. 138

4	From Boltzmann to incompressible Navier-Stokes on the torus			
	4.1	Introduction	143	
	4.2	Main results	149	
	4.3	Toolbox: fluid projection and a priori energy estimates	157	
	4.4	Linear case: proof of Theorem 4.2.1	162	
	4.5	Estimate for the full equation: proof of Proposition 4.2.2	166	
	4.6	Existence and exponential decay: proof of Theorem 4.2.3	170	
	4.7	Exponential decay of $v$ -derivatives: proof of Theorem 4.2.4	175	
	4.8	Incompressible Navier-Stokes Limit: proof of Theorem 4.2.5	181	
	App	endices	189	
	4.A	Validation of the assumptions	189	
	4.B	Proofs of the results given in the toolbox	194	
	4.C	Proof of the hydrodynamical limit lemmas	208	
5	The	Incompressible Navier-Stokes limit in polynomial weighted spaces 2	219	
	5.1	Introduction	221	
	5.2	Main results	226	
	5.3	The linear part: a $C^0$ -semigroup in spaces with polynomial weight, proof of		
		Theorem 5.2.1	230	
	5.4	An <i>a priori</i> estimate for the full perturbed equation: proof of Theorem 5.2.3	243	

# IIIA QUANTIC VERSION OF BOLTZMANN EQUATION FORGASES OF BOSONS AND FERMIONS255

6	The	homogeneous Boltzmann-Nordheim equation for bosons	257
	6.1	Introduction	259
	6.2	Main results	265
	6.3	An extended version of a Povzner-type inequality	266
	6.4	A priori control on the $L_v^{\infty}\left((1+ v ^{\gamma})L_t^1\right)$	271
	6.5	Creation of moments of all order	276
	6.6	Uniqueness of solution for the Boltzmann-Nordheim equation	280
	6.7	Local existence of solutions	290

## Bibliography

305

# Chapter 1

# Introduction

The present thesis deals with the mathematical treatment of kinetic theory and focuses more precisely on the Boltzmann equation. The latter equation describes the evolution in position and velocity of rarefied gas particles with a statistical point of view. It plays a central role in mathematical physics as it builds a bridge between Newtonian systems of particles and fluid dynamics. In this chapter, we start with a brief overview of the Boltzmann equation and its main features (Section 1.1). We then present some mathematical problems such as the quantification of positivity of solutions (Section 1.2) and the Cauchy theory and the trend to equilibrium in a perturbative setting (Section 1.3); which is a short introduction to the hydrodynamical limits of the Boltzmann equation which will be studied more thoroughly in Part II. We conclude by a quantic version of the Boltzmann equation that is used to describe gases of bosons and fermions and also contains the mechanisms of the Bose-Einstein condensate (Section 1.4). We give, in each section, a brief description of our main contributions in those domains.

The reader will also find an index of the notations we use for functional spaces in Appendix 1.A.

### Contents

1.1	Gen	eral presentation of the Boltzmann equation
-	1.1.1	Models in kinetic theory
		1.1.1.1 Kinetic theory and physics of particles and fluids
		1.1.1.2 Evolution equations in kinetic theory
	1.1.2	The collisional model of the Boltzmann equation $\ldots \ldots \ldots$
		1.1.2.1 The Boltzmann collision operator
		1.1.2.2 The different collision kernels of the Boltzmann operator
		1.1.2.3 Initial data and boundary conditions $\ldots$
		1.1.2.4 Conservation laws and entropy dissipation
1.2	Qua	ntitative study of the positivity of solutions (Part $I$ )
1	1.2.1	Motivations and state of the art
-	1.2.2	The free transport equation in convex bounded domain (Chapter 2)
-	1.2.3	An exponential lower bound (Chapter 2)
1.3	The	incompressible Navier-Stokes limit of the Boltzmann equa-
	tion	(Part <i>II</i> )
1	1.3.1	Going from Boltzmann equation to incompressible Navier-Stokes equations (Chapter 3)
1	1.3.2	Hydrodynamical limit in $H^s_{x,v}\left(\mu^{-1/2}\right)$ (Chapter 4)
		1.3.2.1 The linear Boltzmann operator
		1.3.2.2 The perturbed Cauchy problem and trend to equilibrium
		1.3.2.3 The limit towards the incompressible Navier-Stokes equa-
		$tions \ldots \ldots$
-	1.3.3	Hydrodynamical limit in polynomial weighted spaces (Chapter 5)
		1.3.3.1 $$ Semigroup properties in Lebesgue and Sobolev spaces $$ .
		1.3.3.2 Cauchy problem and exponential decay in Sobolev spaces with polynomial weight
1.4	A qu	antic version of Boltzmann equation (Part III)
-	1.4.1	The Boltzmann-Nordheim equation
1	1.4.2	Cauchy problem and the Bose-Einstein condensate (Chapter 6) .
App	endic	es
1.A	Nota	ations
-	1.A.1	Function spaces for one variable
1	1.A.2	Function spaces for several varaiables

### 1.1 General presentation of the Boltzmann equation

In this section we gather some general definitions, descriptions and properties of kinetic theory and the Boltzmann equation in rather informal statements. We refer to the standard books [28][32][30] or the review article [112] for deeper considerations and rigorous statements.

### 1.1.1 Models in kinetic theory

### 1.1.1.1 Kinetic theory and physics of particles and fluids

If one considers a system of N bodies, which can be particles, stars or galaxies for instance, moving in a domain  $\Omega \subset \mathbb{R}^d$  with velocities in  $\mathbb{R}^d$ , the Newton's laws of motion describes the dynamics of the latter system. These laws are the foundations of classical mechanics and generate, in that case, an Hamiltonian system in the phase space  $(\Omega \times \mathbb{R}^d)^N$ . This system of evolution equations encodes the particular dynamics of each body, this is called the microscopic scale.

Unfortunately, the N-body problem is renowned to be of a tremendous mathematical difficulty (even when N = 3) and is therefore hardly useful to study or predict the behaviour of a large number of entities. In the case of a large system, it is often more interesting to look into the general, the average, behaviour rather than following each particle individually. Applying Newton's laws to infinitesimal volume of particles at equilibrium inside this volume leads to the equations of fluids mechanics, such as Euler or Navier-Stokes equations. This point of view is called the macroscopic scale.

The macroscopic scale proved itself useful to describe the average dynamics of systems (sea or car traffic for example). It however comes with the drawback that is the loss of the microscopic dynamics inside the system. Kinetic theory stands right between the microscopic and the macroscopic scale, it is called the mesoscopic scale. It adopts a probabilistic approach to the problem in order to decrease the degrees of freedom of the Hamiltonian system but still keeps track of the microscopic dynamics.

A deeper presentation of these physical points of view as well as their different connections is given in Chapter 3.

The aim of kinetic theory is to model a system constituted of a large number of particles by a distribution function, in the one particle phase space of position and velocity, that evolves with time. More precisely, the dynamics of the system is encoded in a density function

$$\begin{array}{rccc} f: & [0,T] \times \Omega \times \mathbb{R}^d & \longrightarrow & \mathbb{R}^+ \\ & (t,x,v) & \longmapsto & f(t,x,v), \end{array}$$

where T > 0 can be infinite and  $\Omega \times \mathbb{R}^d$  is the particle phase space introduced earlier.

Physically speaking, for a given position x and velocity v, the quantity f(t, x, v)dxdv is the probability of having a particle in the ball B(x, dx) with a velocity in the ball B(v, dv)at time t. One can understand f as an approximation, in the limit when N tends to infinity, of the first marginal of the empirical measure of the system

$$\frac{1}{N}\sum_{i=1}^N \delta_{x_i(t)}(x)\delta_{v_i(t)}(v),$$

where  $(x_i(t), v_i(t))$  is the position and velocity of the  $i^{th}$  particle at time t. A more precise description is given in Chapter 3.

Moreover, for kinetic theory to have a physical meaning, one expects that the total mass of the system remains finite in bounded domains and therefore, the minimal assumption required for f is that

$$\forall t \in [0,T], \quad f(t,\cdot,\cdot) \in L^1_{loc}\left(\Omega, L^1_v\left(\mathbb{R}^d\right)\right).$$

In this point of view, physical observables can be expressed as averages in velocities. We therefore obtain the following local macroscopic quantities of the system of N particles.

• the local density:

$$\rho(t,x) = \int_{\mathbb{R}^d} f(t,x,v) \, dv,$$

• the local velocity:

$$u(t,x) = \frac{1}{\rho(t,x)} \int_{\mathbb{R}^d} v f(t,x,v) \, dv,$$

• the local temperature:

$$\theta(t,x) = \frac{1}{d\rho(t,x)} \int_{\mathbb{R}^d} |v-u|^2 f(t,x,v) \, dv,$$

or, equivalently, the local energy:

$$E(t,x) = \int_{\mathbb{R}^d} \frac{|v|^2}{2} f(t,x,v) \, dv = \rho(t,x) \frac{|u|^2}{2} + d\frac{\rho(t,x)\theta(t,x)}{2}.$$

The mass, mean velocity and temperature of the system being the integral against the space variable x over the spatial domain  $\Omega$  of the local observables. In order to be physically relevant, kinetic theory focuses on density functions f that have finite mass, mean velocity and energy at each time.

### 1.1.1.2 Evolution equations in kinetic theory

As described before, the kinetic theory point of view is to model the dynamics of a large number of particles thanks to an evolution equation satisfied by a density function f = f(t, x, v). This equation has to take into account the free motion of a particle and the possible distortion it undergoes due to an external force or interactions with other particles. As we will see, the latter interactions play a major role in physical studies and contain most of the mathematical difficulties.

In the case of non-interacting particles and in the absence of external force, the motion remains straight lines travelled along with constant velocity. The corresponding equation is the free transport equation

$$\partial_t f + v \cdot \nabla_x f = 0. \tag{1.1.1}$$

When the system is influenced by an external force  $F_{ext} = F_{ext}(x)$  acting on the particles, corrections have to be made to (1.1.1). The new equation is called the linear Vlasov equation and reads

$$\partial_t f + v \cdot \nabla_x f + F_{ext}(x) \cdot \nabla_v f = 0. \tag{1.1.2}$$

Even though these equations are deeper than they look, especially in bounded domains, they neglect the interactions which may exist between particles. These interactions could be attractive or repulsive, thinking of electromagnetism for instance, but also should model what happens when two, or more, particles collide with each other.

The modelling of one-to-one interaction between particles can be done in two different ways, and one can, of course, combine them. The idea of how to derive them from microscopic behaviours is given in Chapter 3, Section 3.2.

If the range of the interaction is macroscopic then the evolution equation is called a mean-field equation. This type of kinetic equation is non-linear and has the following form.

$$\partial_t f + v \cdot \nabla_x f + \nabla_x \Psi(t, x) \cdot \nabla_v f = 0, \qquad (1.1.3)$$

with

$$\Psi(t,x) = -\psi *_x \int_{\mathbb{R}^d} f(t,x,v) dv$$

A typical example of a mean-field equation is the Vlasov-Poisson equation used to described plasmas and for which  $\psi$  is the Coulomb interaction for electromagnetism:

$$\psi(z) = \frac{q^2}{4\pi\varepsilon_0 \left|z\right|}$$

where q is the electric charge of a particle and  $\varepsilon_0$  is the vacuum permittivity.

Our work will, however, be about another way of modelling interactions between par-

ticles. In that case, the range of the interaction is assumed to be so small that it can be considered as a localised interaction. This happens when the trajectory of a particle is distorted when passing very close to another one or, in the simplest physical case, when the particles bounce again each other when colliding. The kinetic equations describing this type of interactions are called collision equations and read

$$\partial_t f + v \cdot \nabla_x f = Q(f), \tag{1.1.4}$$

where Q can be non-linear and encodes the physical properties of the collision process.

One of the most fundamental collision equation in kinetic theory gives the dynamics of rarefied gases. This equation is called the Boltzmann equation and will be the subject of this entire thesis.

### 1.1.2 The collisional model of the Boltzmann equation

As mentionned above, we will only be interested in the case of the collisional model (1.1.4) described by the Boltzmann equation (even if some of our results will apply to more models, see Chapter 4. We give below some elementary properties of the Boltzmann collisional operator Q.

#### 1.1.2.1 The Boltzmann collision operator

The kinetic theory point of view begins with the microscopic modelling of the collisional interactions between particles. The Boltzmann equation rules a particular sort of many particles system. We restrain ourselves to the case of monoatomistic system with elastic collisions. The formal derivation of the Boltzmann equation relies on the following assumptions on the physical process. We refer to [28][30], first chapter, for a complete description.

- 1. We suppose that the interaction is a binary collision, which means that when two particles are close enough to each other their trajectories are deviated. The consequence of such a postulate is that one can neglect collisions involving more than two particles, which implies that the system is comparable to a dilute (rarefied) gas. Mathematically, if the system contains N particles of radius r, we suppose that we are in the Boltzmann-Grad limit:  $Nr^3 << 1$  and  $Nr^2 = O(1)$ .
- 2. The collisions are considered to be localised both in space and time. This conveys the idea of the fact that the trajectories are deviated very quickly and it translates mathematically under the hypothesis that a collision takes place at a position x and a time t.

3. We also suppose that the collisions are elastic. In other terms, the momentum and the kinetic energy are preserved throughout the collision process. If two particles of respective velocities v' and  $v'_*$  collide, then their outcoming velocities v and  $v_*$  satisfies

$$\begin{cases} v' + v'_{*} = v + v_{*} \\ |v'|^{2} + |v'_{*}|^{2} = |v|^{2} + |v_{*}|^{2} \end{cases}$$

We remark here that the mass is the same for all the particles in a monoatomistic gas and considering several species requires a different version of the preservation of kinetic energy and therefore different outcoming velocities.

- 4. The physics of the process is assumed to be microreversible, which means that the microscopic dynamics are reversible in time. In other terms, the probability that velocities  $(v', v'_*)$  are changed into  $(v, v_*)$  during a collision is equal to the probability of changing velocities from  $(v, v_*)$  into  $(v', v'_*)$ .
- 5. We further suppose Boltzmann molecular choas inside the system. This states that before they collide, two particles evolve independently one from the other. This hypothesis implies an asymmetry in the arrow of time since after collision the velocities of the two particles are correlated (*via* the preservation of momentum and kinetic energy).

The formal derivation, from Newton's laws, of the kinetic model associated to the assumptions above (see Chapter 3.2 for a brief explaination or [28], chapter 3) yields the Boltzmann equation. Note that the rigorous mathematical derivation is still a very hard problem even if in 1974 Lanford [65], and recently ameliorated in [44][96], proved it for very short time (typically, shorter than the mean time of first collision).

The Boltzmann collision operator is therefore a bilinear operator encoding the probability for two particles with velocities v' and  $v'_*$  to undergo a collision resulting in velocities v and  $v_*$ . The laws of elasticity link  $(v', v'_*)$  to  $(v, v_*)$  in a bijective correspondence (easily deduced from Figure 1.1) we call the " $\sigma$ -representation". If we denote

$$\sigma = \frac{v' - v'_*}{|v' - v'_*|}$$

then  $\sigma$  varies on  $\mathbb{S}^{d-1}$  when  $(v', v'_*)$  varies in  $\mathbb{R}^{2d}$  and we have the following relation

$$v' = \frac{v + v_*}{2} + \frac{|v - v_*|}{2}\sigma$$
$$v'_* = \frac{v + v_*}{2} - \frac{|v - v_*|}{2}\sigma$$



Figure 1.1: Correspondance between pre and post collision velocities

Under this representation we obtain an explicit form for the Boltzmann equation,

$$\partial_t f + v \cdot \nabla_x f = Q(f, f), \tag{1.1.5}$$

with Q being the Boltzmann collision operator given by

$$Q(f,f) = \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} B(|v - v_*|, \cos \theta) \left[ f' f'_* - f f_* \right] dv_* d\sigma.$$
(1.1.6)

In the expression above, B is called the Boltzmann collision kernel and encodes the physics of the collision process,  $\theta$  is the angle between  $v - v_*$  and  $\sigma$ , and we use the standard notations  $f = f(t, x, v), f_* = f(t, x, v_*), f' = f(t, x, v')$  and  $f'_* = f(t, x, v'_*)$ .

### 1.1.2.2 The different collision kernels of the Boltzmann operator

Alternative representations. The first thing one can notice about the Boltzmann collision kernel is that its form (1.1.6) depends on the choice we made to express  $(v', v'_*)$  in terms of  $(v, v_*)$ . There exists other parametrisations and we refer to [112] Chapter 1 for advantages and inconveniences of each of them. We nonetheless present two alternative representations of the Boltzmann collision operator that will be use later in this work.

The most common alternative way of writing the Boltzmann operator Q is the so-called " $\omega$ -representation" (which can also be easily deduced from Figure 1.1). In this case, we consider the unit vector

$$\omega = \frac{v - v'}{|v - v'|}$$

to obtain a new bijective correspondance betteen  $(v', v'_*)$  and  $(v, v_*)$ , namely

$$\begin{cases} v' = v - \langle v - v_*, \omega \rangle \, \omega \\ v'_* = v_* - \langle v_* - v, \omega \rangle \, \omega. \end{cases}$$

In the " $\omega$ -representation", the collision operator reads

$$Q(f,f) = \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} \tilde{B}\left(v - v_*, \omega\right) \left[f'f'_* - ff_*\right] dv_* d\omega,$$

where the correspondance with the " $\sigma$ -representation" is given by

$$\tilde{B}(z,\omega) = 2^{d-1} \left| \left\langle \frac{z}{|z|}, \omega \right\rangle \right|^{d-2} B(|z|, \sigma).$$

The second representation we want to introduce has been proposed by Carleman [27] and takes v' and  $v'_*$  as integration parameters and defines  $v_* = v' + v'_* - v$ . With these choices, the Boltzmann collision operator reads

$$Q(f,f) = \int_{\mathbb{R}^d} \left( \int_{E_{v,v'}} \frac{1}{|v-v'|^{d-1}} \tilde{B}\left( 2v - v' - v'_*, \frac{v' - v'_*}{|v' - v'_*|} \right) \left[ f'f'_* - ff_* \right] \, dv'_* \right) \, dv',$$

where  $E_{v,v'}$  is the hyperplane going through v and orthogonal to v - v'.

The Carleman representation will play an important role in Chapter 6.

**Different collisional interactions.** The physics of the collision process is encoded in the Boltzmann collision kernel B. For simplicity reasons, we will always assume that this kernel is of the form

$$B\left(\left|v-v_{*}\right|,\cos\theta\right) = \Phi\left(\left|v-v_{*}\right|\right)b(\cos\theta),$$

where  $\Phi$  and b are positive and locally integrable functions except, eventually, at respectively 0 and 1. This assumption is made without loss of generality on the kernel but reduce the complexity of later computations. Moreover, it is satisfied in all the physically relevant cases that we describe below. We refer to [28] Chapter 2 or [112] Chapter 1 for a derivation of B from the interaction laws.

A very important case is the one of hard spheres which correspond to the case where particles are considered as billiard balls bouncing on each other. For this specific interaction one has

$$\exists C_{\Phi} > 0, \quad B(|v - v_*|, \cos \theta) = C_{\Phi} |v - v_*|.$$

In a more general setting, we will always assume that the kinetic collision kernel  $\Phi$ 

satisfies either

$$\forall z \in \mathbb{R}, \quad c_{\Phi} |z|^{\gamma} \leqslant \Phi(z) \leqslant C_{\Phi} |z|^{\gamma}$$

or a mollified version

$$\begin{cases} \forall |z| \ge 1 \in \mathbb{R}, \quad c_{\Phi} |z|^{\gamma} \le \Phi(z) \le C_{\Phi} |z|^{\gamma} \\ \forall |z| \le 1 \in \mathbb{R}, \quad c_{\Phi} \le \Phi(z) \le C_{\Phi}, \end{cases}$$

 $c_{\Phi}$  and  $C_{\Phi}$  being strictly positive constants and  $\gamma$  belonging to (-d, 1]. The collision kernel is said to be "hard potential" in the case of  $\gamma > 0$ , "soft potential" if  $\gamma < 0$  and "Maxwellian" if  $\gamma = 0$ .

The angular collision kernel b is seldom known explicitly. However, we will assume  $(b \circ \cos)$  to be a continuous function on  $\theta$  in  $(0, \pi]$ , strictly positive near  $\theta \sim \pi/2$ , which satisfies

$$b(\cos\theta)\sin^{d-2}\theta \underset{\theta\to 0^+}{\sim} b_0 \theta^{-(1+\nu)},$$

for  $b_0 > 0$  and  $\nu$  in  $(-\infty, 2)$ . The case when b is locally integrable,  $\nu < 0$ , is referred to by the Grad's cutoff assumption (first introduce in [48]) and therefore B will be said to be a cutoff collision kernel. This case is of tremendous importance since it allows to decompose the Boltzmann operator  $Q = Q^+ - Q^-$ . The case  $\nu \ge 0$  will be designated by non-cutoff collision kernel.

We can mention here that in the physically important case of inverse-power laws in dimension d = 3,

$$\Phi(z) = C_{\Phi} |z|^{\gamma}$$

and  $\gamma$  and  $\nu$  are not independent since there exists s > 2 such that

$$\begin{pmatrix} \gamma = \frac{s-5}{s-1} \\ \nu = \frac{2}{s-1}. \end{pmatrix}$$

Moreover, in the case of Coulomb interactions s = 2, we have an explicit formula for the angular kernel in dimension d = 3, which is

$$b\left(\cos\theta\right) = \frac{b_0}{\sin^4\theta}.$$

The mathematical treatment of these different collision kernels reveals different behaviours for solutions to the Boltzmann equation, depending where the singularities of both the kinetic and the angular collision kernels occur. In other terms, the decay at infinity or even the regularity properties, both in the velocity variable, of solutions to the Boltzmann equation (1.1.5) are very sensitive to the way  $\Phi$  behaves for small or large relative velocities and the way ( $b \circ \cos$ ) blows up at  $\theta \sim 0^+$ .

### 1.1.2.3 Initial data and boundary conditions

The Boltzmann equation has to describe, at a mesoscopic scale, the motion of particles evolving in time in the spatial domain  $\Omega$ . We therefore need to prescribe an initial distribution  $f_{in}(x, v)$  as well as a modelling of the interactions between a particle and the boundary of  $\Omega$ , in the case it exists.

The problem of the initial data is quite obvious

$$\forall x \in \Omega, \ \forall v \in \mathbb{R}^d, \quad f(0, x, v) = f_{in}(x, v).$$

However, there exists density functions that are not physically relevant, as discussed in Section 1.1.1.1. In that respect, the physically relevant solutions to the Boltzmann equation must have finite mass and energy at least in bounded sets. The minimum requirements one should ask for  $f_{in}$  are thus

- 1.  $f_{in}(x,v) \ge 0$  almost everywhere in  $\Omega \times \mathbb{R}^d$ ,
- 2. for all K compact in  $\Omega$ ,

$$\int_K \int_{\mathbb{R}^d} \left( 1 + |v|^2 \right) f_{in}(x, v) \, dx \, dv < +\infty.$$

There exist several modellings of the interactions between a particle and the boundary of  $\Omega$ .

In the case  $\Omega = \mathbb{R}^d$ , no boundary condition is needed. However, the relevant solutions need to satisfy an integrability condition at infinity.

In the case  $\partial \Omega \neq \emptyset$ , particles will interact with the frontier of the domain. The most common behaviours are the following.

• The bounce-back condition

$$\forall (t, x, v) \in \mathbb{R}^+ \times \partial\Omega \times \mathbb{R}^d, \quad f(t, x, v) = f(t, x, -v).$$

• If  $\Omega$  is regular enough, then one can consider the specular reflection boundary condition

 $\forall (t, x, v) \in \mathbb{R}^+ \times \partial\Omega \times \mathbb{R}^d, \quad f(t, x, v) = f(t, x, \mathcal{R}_x(v)),$ 

 $\mathcal{R}_x$ , for x on  $\partial\Omega$ , stands for the specular reflection at that point against the boundary. One can compute, denoting by  $n_x$  the outward normal at a point x on  $\partial\Omega$ ,

$$\forall v \in \mathbb{R}^d, \quad \mathcal{R}_x(v) = v - 2\langle v, n_x \rangle n_x.$$

• If  $\Omega$  is regular enough and  $\partial \Omega$  has a temperature  $T_{\partial}$  then one can impose the Maxwellian diffusion boundary condition

$$\forall (t,x,v) \in \mathbb{R}^+ \times \partial\Omega \times \mathbb{R}^d, \quad f(t,x,v) = \left[ \int_{v \cdot n_x > 0} f(t,x,v) \left( v \cdot n_x \right) \, dv \right] \frac{1}{(2\pi)^{\frac{d-1}{2}} T_\partial^{\frac{d+1}{2}}} e^{-\frac{|v|^2}{2T_\partial}}.$$

Note that the first two boundary conditions convey the idea of particles bouncing against the wall, in two different manners, whereas the third one expresses the fact that particles are absorbed by the wall and then emitted back into  $\Omega$  according to the thermodynamical equilibrium distribution  $M_{\partial}$  between the wall and the gas,

$$M_{\partial}(v) = \frac{1}{(2\pi)^{\frac{d-1}{2}} T_{\partial}^{\frac{d+1}{2}}} e^{-\frac{|v|^2}{2T_{\partial}}}.$$

The last case we will consider is the periodic case when  $\Omega$  is the *d*-dimensional torus  $\mathbb{T}^d$ . This will be of particular interest since it is a bounded domain without boundary conditions except for the periodicity condition. This case is also physically and mathematically interesting because it has been proven (see [30] Chapter 7) that it is equivalent to the case when  $\Omega$  is a box with specular reflection boundary conditions.

### 1.1.2.4 Conservation laws and entropy dissipation

There are some few interesting facts that one can rapidly discover about the solutions to the Boltzmann equation, at least formally. Its collision operator

$$Q(f,f) = \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} B\left(|v - v_*|, \cos \theta\right) \left[f'f'_* - ff_*\right] dv_* d\sigma$$

encodes the microscopic behaviour of the gas in the case of elastic collisions. This particular case of interaction preserves the mass, the momentum and the energy and this reflects on the macroscopic observables. Indeed, the Boltzmann collision kernel is invariant, for instance, under the changes of variable

$$(v, v_*, \sigma) \to (v', v'_*, k)$$
 with  $k = \frac{v - v_*}{|v - v_*|}$ 

and

$$(v, v_*, \sigma) \rightarrow (v_*, v, -\sigma).$$

These invariances formally give (see [46] Chapter 1 for a rigorous statement) that for a given test function  $\phi(v)$ ,

$$\int_{\mathbb{R}^d} Q(f,f)(v)\phi(v) \, dv = -\frac{1}{4} \int_{\mathbb{R}^d \times \mathbb{R}^d \times \mathbb{S}^{d-1}} B\left[f'f'_* - ff_*\right] \left(\phi'_* + \phi' - \phi_* - \phi\right) \, dv dv_* d\sigma.$$
(1.1.7)

The latter property has two major consequences that are related to macroscopic laws. In full generality (1.1.7) implies first that

$$\int_{\mathbb{R}^d} Q(f, f) \begin{pmatrix} 1 \\ v \\ |v|^2 \end{pmatrix} dv = 0$$
(1.1.8)

and second

$$D(f) := -\int_{\mathbb{R}^d} Q(f, f) \log f \, dv \ge 0.$$
(1.1.9)

In the case when f is solution to the Boltzmann equation

$$\partial_t f + v \cdot \nabla_x f = Q(f, f),$$

we have that, by integrating in v the differential equation, (1.1.8) comes out as

• the preservation of the total mass

$$\frac{d}{dt}\int_{\Omega}\rho(t,x)\,dx=0,$$

• the preservation of total energy if Ω has no boundary or if boundary conditions are bounce-back or specular reflections

$$\frac{d}{dt}\int_{\Omega}E(t,x)\,dx=0,$$

• the preservation of total momentum if  $\Omega$  has no boundary

$$\frac{d}{dt} \int_{\Omega} \rho(t, x) u(t, x) \, dx = 0,$$

The consequence of (1.1.9) applied to solutions to the Boltzmann equation, at least at a formal level, is known as the Boltzmann H-theorem. The latter theorem states, if  $\Omega$  has no boundary or if boundary conditions are bounce-back or specular reflections, that the entropy of a solution f, defined as

$$S(f) = \int_{\Omega \times \mathbb{R}^d} f \log f \, dx dv$$

which is the opposite of the thermodynamical entropy, can only decrease in time

$$\frac{d}{dt}S(f) = -\int_{\Omega} D(f) \, dx \leqslant 0.$$

Such a result is much deeper than it looks and was subject to a lot of discussions and rejections from the scientific community when Boltzmann discovered it. The entropy dissipation indeed implies time irreversibility of the Boltzmann equation which seems unnatural since the Newton dynamics of the microscopic processes it describes are timereversible.

We conclude this brief introduction to the Boltzmann equation by describing its steady states. The entropy dissipation states that we are at a local thermodynamic equilibrium if

$$D(f)(t,x) = 0$$

which is possible if and only if

$$\forall v \in \mathbb{R}^d, \quad f(t, x, v) = M_{(\rho(t, x), u(t, x), \theta(t, x))}(v),$$

where  $M_{(\rho,u,\theta)}$  is called a Maxwellian distribution and is defined by

$$M_{(\rho,u,\theta)}(v) = \frac{\rho}{(2\pi\theta)^{d/2}} e^{-\frac{|v-u|^2}{2\theta}}.$$

Moreover, for all  $\rho = \rho(t, x)$ , u = u(t, x) and  $\theta = \theta(t, x)$  the following holds

$$Q(M_{(\rho,u,\theta)}, M_{(\rho,u,\theta)}) = 0,$$

and therefore a local thermodynamic equilibrium is global if and only if

$$\forall (x,v) \in \Omega \times \mathbb{R}^d, \quad v \cdot \nabla_x M_{(\rho,u,\theta)} = 0.$$

In the case of the torus, this condition yields a unique global equilibrium for the Boltzmann equation that is the Maxwellian independent of t and x that has the same total mass, momentum and energy as the initial configuration  $f_{in}$ . This is also the case if  $\Omega$  is a non axis-symmetric bounded domain with bounce-back or specular reflection boundary conditions. We refer to [46] Chapter 1 and [112] Section 2.5 for more details and references.

In these cases, we can always consider, without loss of generality, that the equilibrium

is a centred gaussian with mass and variance 1. We denote

$$\mu(v) = \frac{1}{(2\pi)^{d/2}} e^{-\frac{|v|^2}{2}}.$$

# 1.2 Quantitative study of the positivity of solutions (Part I)

Part I deals with some quantitative and qualitative aspects of solutions to the Boltzmann equation (1.1.5).

The aim of Chapter 2 is to prove that non-negative solutions to the Boltzmann equation are in fact strictly positive and bounded from below by an exponential lower bound. This is an *a priori* result where we will not tackle the issue of existence. We solely make the hypothesis that a solution exists and it has some uniform regularity properties we present below.

The framework of this study is quite wide since we consider all types of collision operator (hard and soft potentials with or without angular cutoff) and  $\Omega$  is a  $C^2$  convex bounded domain with specular reflection boundary conditions.

### **1.2.1** Motivations and state of the art

The issue of quantifying the positivity of solutions to the Boltzmann equation has been investigated for a long time. It is of great interest for physical purposes but, more recently, it has also proven itself of significant importance for the mathematical study of the Boltzmann equation. The development of entropy-entropy production methods (see [112] Chapter 3 and [113]) to study the convergence to equilibrium [35][36][37] requires this type of exponential lower bounds.

The first quantitative statement on positivity of the solutions to the Boltzmann equation goes back to Carleman [26] where he dealt with the spatially homogeneous equation. The radially symmetric solutions f(t, v) = f(t, |v|) he constructed in dimension d = 3 for hard sphere collision operator, satisfy an almost Maxwellian lower bound

$$\forall t \ge t_0, \ \forall v \in \mathbb{R}^3, \quad f(t,v) \ge C_1 e^{-C_2 |v|^{2+\varepsilon}},$$

 $C_1, C_2 > 0$  for all  $t_0 > 0$  and  $\varepsilon > 0$ . The constants  $C_1$  and  $C_2$  depends only on  $t_0, \varepsilon$  and a priori estimates on the solution f.

Pulvirenti and Wennberg [95] then extended the latter inequality to solutions to the spatially homogeneous Boltzmann equation with hard potential and cutoff in dimension d = 3 with more general initial data. They proved that if the solution has finite mass,

energy and entropy then it is bounded from below by a Maxwellian lower bound of the type

$$\forall t \ge t_0, \, \forall v \in \mathbb{R}^3, \quad f(t,v) \ge C_1 e^{-C_2 |v|^2},$$

for all  $t_0 > 0$ .

Finally, Mouhot [78] dealt with the full Boltzmann equation on the torus. He not only proved the same result as Pulvirenti and Wennberg in the case of hard potential with angular cutoff but he also obtained the Maxwellian lower bound for soft potential with cutoff collision kernels. He also derived the same kind of results in the non-cutoff case in the torus, the immediate appearance of an exponential lower bound of the form

$$\forall t \ge t_0, \, \forall (x,v) \in \mathbb{T}^d \times \mathbb{R}^d, \quad f(t,v) \ge C_1(\varepsilon) e^{-C_2(\varepsilon)|v|^{K+\varepsilon}}$$

for all  $t_0 > 0$ , all  $\varepsilon > 0$  and  $K = K(\nu)$  with K(0) = 2 (thus recovering the cutoff case in the limit).

All these results deeply rely on a spreading property of the gain part  $Q^+$  of the Boltzmann collision operator that arises as soon as the solution has a non-concentration property which means, roughly speaking, an initial lower bound. This "upheaval point" results from non-concentration properties of the gain operator ([95]) or continuity-compactness arguments (Chapter 2). The case of spatially inhomogeneous solutions [78] is based on these arguments and a method to make them uniform under the flow of characteristics.

The case where  $\Omega$  is bounded implies a different behaviour for the characteristics and our main contribution is the derivation of a spreading method that remains invariant under the characteristics flow. For instance, boundaries imply the existence of grazing collisions where the strategy develop on the torus fails and we had to create a geometrical approach of those problematic trajectories.

# 1.2.2 The free transport equation in convex bounded domain (Chapter 2)

The first task is to establish a rigorous description of characteristic trajectories for the free transport equation

$$\partial_t f + v \nabla_x f = 0$$

with specular reflection boundary conditions, which can be seen has billiard balls trajectories. Although it has been studied in numerous works [93][33][104][105] and has been used in kinetic theory [52][59], a complete study in the case of mere specular reflections and convexity seemed to be missing. The cited works indeed contain assumptions on the boundary (electromagnetism or strict convexity for example) that lead to clear rebounds against the boundary whereas a general study should also consider rolling trajectories for instance.

Chapter 2 starts with an extensive and descriptive study of the characteristics of the free transport equation in a  $C^1$  convex bounded domain. One of our most relevant contribution is the extension of a result of Tabachnikov [104] that states that the set of points (x, v) that comes from infinitely many rebounds in finite time is negligeable. More precisely, we proved the following.

**Proposition 1.2.1** Let  $\Omega$  be a  $C^1$  open, bounded domain in  $\mathbb{R}^d$  and let (x, v) be in  $\overline{\Omega} \times \mathbb{R}^d$ . Then for all  $t \ge 0$  the trajectory finishing at (x, v) after a time t has at most a countable number of rebounds.

Moreover, this number is finite almost surely with respect to the Lebesgue measure on  $\bar{\Omega} \times \mathbb{R}^d$ .

The main idea was to generate a parametrisation of  $\overline{\Omega}$  that links the trajectories to their footprints, where this result is known to hold thanks to the work of Tabachnikov.

### 1.2.3 An exponential lower bound (Chapter 2)

In what follows, we are going to need bounds on some physical observables of solution to the Boltzmann equation and we describe them below.

We consider a function  $f(t, x, v) \ge 0$  defined on  $[0, T) \times \Omega \times \mathbb{R}^d$  and we recall the definitions of its local physical quantities.

• its local energy

$$e_f(t,x) = \int_{\mathbb{R}^d} |v|^2 f(t,x,v) dv,$$

• its local weighted energy

$$e'_f(t,x) = \int_{\mathbb{R}^d} |v|^{\tilde{\gamma}} f(t,x,v) dv,$$

where  $\tilde{\gamma} = (2 + \gamma)^+$ ,

• its local  $L^p$  norm  $(p \in [1, +\infty))$ 

$$l_f^p(t,x) = \|f(t,x,\cdot)\|_{L_v^p},$$

• its local  $W^{2,\infty}$  norm

$$w_f(t,x) = \|f(t,x,\cdot)\|_{W^{2,\infty}_{u}}.$$

The solutions to the Boltzmann equation are assumed to satisfy some properties about their local hydrodynamical quantities. These properties will differ depending on which case of collision kernel we are considering and are given by what follows.

• In the case of hard or Maxwellian potentials with cutoff ( $\gamma \ge 0$  and  $\nu < 0$ ):

$$\sup_{(t,x)\in[0,T)\times\Omega} e_f(t,x) < +\infty.$$
(1.2.1)

• In the case of a singularity of the kinetic collision kernel  $(\gamma \in (-d, 0))$  we shall make the additional assumption

$$\sup_{(t,x)\in[0,T)\times\Omega} l_f^p(t,x) < +\infty, \tag{1.2.2}$$

where  $p_{\gamma} > d/(d+\gamma)$ .

• In the case of a singularity of the angular collision kernel ( $\nu \in [0, 2)$ ) we shall make the additional assumption

$$\sup_{(t,x)\in[0,T)\times\Omega} w_f(t,x) \quad \text{and} \quad \sup_{(t,x)\in[0,T)\times\Omega} e'_f(t,x) < +\infty.$$
(1.2.3)

We now state the result of Chapter 2 in a rather informal way. For a more detailed and more rigorous statement, we refer to Section 2.2.

**Theorem 1.2.2** Let  $\Omega$  be  $\mathbb{T}^d$  or a  $C^2$  open convex bounded domain in  $\mathbb{R}^d$  and let  $f_{in}$  be a non-negative continuous function on  $\overline{\Omega} \times \mathbb{R}^d$  with strictly positive mass and finite energy. Let f(t, x, v) be a continuous non-negative solution of the Boltzmann equation in  $\overline{\Omega} \times \mathbb{R}^d$ on some time interval  $[0, T), T \in (0, +\infty]$ , which satisfies

- if the collision kernel is hard or Maxwellian potential with cutoff, then f satisfies (1.2.1);
- if the collision kernel is soft potential, then f satisfies (1.2.1) and (1.2.2);
- if the collision kernel is non-cutoff, then f satisfies (1.2.3).

Then

(i) for cutoff collision kernels: for all  $\tau \in (0,T)$  there exists  $\rho > 0$  and  $\theta > 0$ , depending on  $\tau$ ,  $E_f$  (and  $L_f^{p\gamma}$  if B is a soft potential kernel), such that for all  $t \in [\tau, T)$  the solution f is bounded from below, almost everywhere, by a global Maxwellian distribution with density  $\rho$  and temperature  $\theta$ , i.e.

$$\forall t \in [\tau, T), \, \forall (x, v) \in \bar{\Omega} \times \mathbb{R}^d, \quad f(t, x, v) \geqslant \frac{\rho}{(2\pi\theta)^{d/2}} e^{-\frac{|v|^2}{2\theta}}.$$

(ii) for non-cutoff collision kernels: for all  $\tau \in (0,T)$  and for any exponent K such that

$$K > 2 \frac{\log\left(2 + \frac{2\nu}{2-\nu}\right)}{\log 2},$$

there exists  $C_1, C_2 > 0$ , depending on  $\tau$ , K,  $E_f, E'_f, W_f$  (and  $L_f^{p\gamma}$  if B is a soft potential kernel), such that

$$\forall t \in [\tau, T), \, \forall (x, v) \in \bar{\Omega} \times \mathbb{R}^d, \quad f(t, x, v) \ge C_1 e^{-C_2 |v|^K}.$$

As an important remark, let us emphasize that in the case of a  $C^3$  bounded strictly convex domain with f having uniformly bounded local mass and entropy, our proofs are entirely constructive.

## 1.3 The incompressible Navier-Stokes limit of the Boltzmann equation (Part *II*)

# 1.3.1 Going from Boltzmann equation to incompressible Navier-Stokes equations (Chapter 3)

The Boltzmann equation rules the mesocopic evolution of a rarefied gas and is established on the microscopic dynamics of the particles. It therefore stands in between the microscopic scale and the macroscopic scale described by the acoustics and fluids evolution equations. A natural question thus arises: does there exist a link between the physical observables of solutions to the Boltzmann equation and solutions to fluid dynamics ?

It is physically relevant to derive a non-dimensional form of the Boltzmann equation [46][98] which reads

$$\partial_t f_{\varepsilon} + v \cdot \nabla_x f_{\varepsilon} = \frac{1}{\varepsilon} Q(f_{\varepsilon}, f_{\varepsilon}), \qquad (1.3.1)$$

where  $\varepsilon$  is called the Knudsen number of the gas. Physically,  $\varepsilon^{-1}$  represents the average number of collisions for each particle per unit of time. Therefore, as reviewed in [111], one can expect a convergence from the Boltzmann model towards the acoustics and the fluids dynamics as the Knudsen number  $\varepsilon$  tends to 0. The study of the latter convergence is called the hydrodynamical limits of the Boltzmann equation and is of tremendous importance to mathematically prove the coherence of the different scales of description in physics. Chapter 3 is dedicated to the issue of hydrodynamical limits and gives a state of the art in the domain. We therefore just describe briefly here the incompressible Navier-Stokes framework that we thoroughly study in Chapters 4 and 5. The incompressible Navier-Stokes equations read

$$\partial_t u - \nu \Delta u + u \cdot \nabla u + \nabla p = 0,$$
  

$$\nabla \cdot u = 0,$$
  

$$\partial_t \theta - \kappa \Delta \theta + u \cdot \nabla \theta = 0,$$
  
(1.3.2)

to which we can add the Boussineq relation

$$\nabla(\rho + \theta) = 0, \tag{1.3.3}$$

where p is the pressure,  $\nu$  and  $\kappa$  are respectively the dynamic viscosity and the thermal conductivity of the fluid.

The Boltzmann equation and the incompressible Navier-Stokes equations describe physical phenomenon that do not evolve at the same timescale. As suggested in previous studies [46][111][98] we need to rescale (1.3.1) in time by a factor  $\varepsilon$ , to get rid of these time scale differences. Moreover, they also suggested that a perturbation of order  $\varepsilon$ around the global equilibrium

$$\mu(v) = \frac{1}{(2\pi)^{d/2}} e^{-\frac{|v|^2}{2}}$$

should approximate, as the Knudsen number tends to 0, the incompressible Navier-Stokes equations.

We hence study the following equation

$$\partial_t f_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x f_{\varepsilon} = \frac{1}{\varepsilon^2} Q(f_{\varepsilon}, f_{\varepsilon}) , \text{ on } \mathbb{R}^+ \times \mathbb{T}^d \times \mathbb{R}^d,$$
(1.3.4)

under the linearization  $f_{\varepsilon}(t, x, v) = \mu(v) + \varepsilon h_{\varepsilon}(t, x, v)$ . This leads to the perturbed Boltzmann equation

$$\partial_t h_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x h_{\varepsilon} = \frac{1}{\varepsilon^2} L(h_{\varepsilon}) + \frac{1}{\varepsilon} \Gamma(h_{\varepsilon}, h_{\varepsilon}).$$
(1.3.5)

that we will study thoroughly, and where we defined

$$\left\{ \begin{array}{ll} L(h) &= \left[Q(\mu,h) + Q(h,\mu)\right] \\ \Gamma(g,h) &= \frac{1}{2}\left[Q(g,h) + Q(h,g)\right]. \end{array} \right.$$

Roughly speaking, the dissipation of entropy discussed in Section 1.1.2.4 is expected, in the case of small initial perturbation  $h_{\varepsilon}(0, x, v)$ , to make  $f_{\varepsilon}(t, x, v) = \mu(v) + \varepsilon h_{\varepsilon}(t, x, v)$ converge towards its global equilibrium  $\mu(v)$  as time goes to infinity. This trend to equilibrium would give bounds on  $f_{\varepsilon}$  and if the latter bounds are uniform in  $\varepsilon$  one can study the hydrodynamical limit

$$\lim_{\varepsilon \to 0} \|f_{\varepsilon}\|,$$

where the norm of the convergence will be rigorously defined later. The main goal to study the limit for Boltzmann equation towards incompressible Navier-Stokes equation is therefore to develop a Cauchy theory and prove a trend to equilibrium for (1.3.5) that will be uniform in  $\varepsilon$ .

## 1.3.2 Hydrodynamical limit in $H_{x,v}^{s}(\mu^{-1/2})$ (Chapter 4)

Chapter 4 rigorously justifies the discussion of previous subsection in the Sobolev space  $H_{x,v}^s(\mu^{-1/2})$  for *s* large. More precisely, it constructs a Cauchy theory for small initial data of the perturbed Boltzmann equation (1.3.5). This theory is uniform in the Knudsen number, that is to say the smallness assumption is independent of  $\varepsilon$ . Moreover, we show an exponential decay for  $h_{\varepsilon}$ , uniformly in  $\varepsilon$ . The latter decay allows us to rigorously prove the convergence of the observables of  $h_{\varepsilon}$  towards solutions to the incompressible Navier-Stokes equation (1.3.2), satisfying the Boussineq equation (1.3.3).

We emphasize here that all the results in Chapter 4 are obtained constructively, which is of great importance for physical purposes and seldom the case in Boltzmann perturbative theory. Our main contribution is the derivation of hypocoercive estimates independent on  $\varepsilon$  in new distorted norms catching the dependencies in the Knudsen number.

We refer to Sections 4.1.4 and 5.1.2 for a state of the art of the study of the semigroup and the Cauchy problem.

In this section we consider the Boltzmann equation with hard potential or Maxwellian potential ( $\gamma = 0$ ), that is to say there is a constant  $C_{\Phi} > 0$  such that

$$B(|v - v_*|, \theta) = \Phi(|v - v_*|) b(\cos \theta)$$

with

$$\Phi(z) = C_{\Phi} z^{\gamma} , \ \gamma \in [0,1],$$

and a strong form of Grad's angular cutoff, expressed here by the fact that we assume b to be  $C^1$  with the controls from above

$$\forall z \in [-1, 1], \quad b(z), b(z') \leqslant C_b$$

For the sake of clearness, we study (1.3.4) with the linearization  $f_{\varepsilon}(t, x, v) = \mu(v) + \varepsilon \mu^{1/2} h_{\varepsilon}(t, x, v)$  which amounts to working on  $h_{\varepsilon}$  in the space  $H_{x,v}^s$  without any weight. The sole changes are in the linear and bilinear operators:

$$\begin{cases} L(h) = \left[Q(\mu, \mu^{\frac{1}{2}}h) + Q(\mu^{\frac{1}{2}}h, \mu)\right] \mu^{-\frac{1}{2}} \\ \Gamma(g, h) = \frac{1}{2} \left[Q(\mu^{\frac{1}{2}}g, \mu^{\frac{1}{2}}h) + Q(\mu^{\frac{1}{2}}h, \mu^{\frac{1}{2}}g)\right] \mu^{-\frac{1}{2}}. \end{cases}$$
(1.3.6)

### 1.3.2.1 The linear Boltzmann operator

A common strategy in perturbative framework is to study the properties of the linear operator part and then consider the bilinear as a remainder term. We therefore focus first on the linear part of the perturbed Boltzmann equation

$$G_{\varepsilon} = \frac{1}{\varepsilon^2} L - \frac{1}{\varepsilon} v \cdot \nabla_x.$$

In the case of hard potential with angular cutoff, it is known that L is a negative self-adjoint operator in  $L_v^2$ . More importantly, L in hypocoercive. This translates into the following properties.

**Properties in**  $H^1_{x,v}$  :

### (H1): Coercivity and general controls

 $L: L_v^2 \longrightarrow L_v^2$  is a closed and self-adjoint operator with  $L = K - \Lambda$  such that:

•  $\Lambda$  is coercive:

– it exists  $\|.\|_{\Lambda_v}$  norm on  $L^2_v$  such that

$$orall h\in L^2_v \ , \ 
u_0^\Lambda \left\|h
ight\|^2_{L^2_v}\leqslant 
u_1^\Lambda \left\|h
ight\|^2_{\Lambda_v}\leqslant \langle\Lambda(h),h
angle_{L^2_v}\leqslant 
u_2^\Lambda \left\|h
ight\|^2_{\Lambda_v} ,$$

 $-\Lambda$  has a defect of coercivity regarding its v derivatives:

$$\forall h \in H_v^1, \ \langle \nabla_v \Lambda(h), \nabla_v h \rangle_{L_v^2} \ge \nu_3^{\Lambda} \| \nabla_v h \|_{\Lambda_v}^2 - \nu_4^{\Lambda} \| h \|_{\Lambda_v}^2.$$

• There exists  $C^L > 0$  such that

$$\forall h \in L_v^2, \ \forall g \in L_v^2, \ \langle L(h), g \rangle_{L_v^2} \leqslant C^L \, \|h\|_{\Lambda_v} \, \|g\|_{\Lambda_v},$$

where  $(\nu_k^{\Lambda})_{1 \leq k \leq 4}$  are strictly positive constants depending on the operator and the dimension d.

We define a new norm on  $L^2_{x,v}$ :

$$\|.\|_{\Lambda} = \|\|.\|_{\Lambda_v}\|_{L^2_{w}}.$$

### (H2): Mixing property in velocity

$$\forall \delta > 0 , \exists C(\delta) > 0 , \forall h \in H_v^1 , \quad \langle \nabla_v K(h), \nabla_v h \rangle_{L_v^2} \leqslant C(\delta) \|h\|_{L_v^2}^2 + \delta \|\nabla_v h\|_{L_v^2}^2.$$

### (H3): Relaxation to equilibrium

The kernel of L is generated by d+2 functions which form an orthonormal basis for Ker(L):

$$\operatorname{Ker}(L) = \operatorname{Span}\{\phi_1(v), \dots, \phi_{d+2}(v)\}.$$

Moreover, the  $\phi_i$  are of the form  $P_i(v)e^{-|v|^2/4}$ , where  $P_i$  is a polynomial.

Furthermore, denoting by  $\pi_L$  the orthogonal projector in  $L_v^2$  on Ker(L) we have the following local coercivity property:

$$\exists \lambda > 0 , \, \forall h \in L^2_v , \quad \langle L(h), h \rangle_{L^2_v} \leqslant -\lambda \left\| h^{\perp} \right\|_{\Lambda_v}^2$$

where  $h^{\perp} = h - \pi_L(h)$  denotes the microscopic part of h (the orthogonal to Ker(L) in  $L_v^2$ ).

### Assumptions in $H_{x,v}^s$ , s > 1 :

### (H1'): Defect of coercivity for higher derivatives

L satisfies the following property: for all  $s \ge 1$ , for all |j| + |l| = s such that  $|j| \ge 1$ ,

$$\forall h \in H^s_{x,v} , \quad \langle \partial^j_l \Lambda(h), \partial^j_l h \rangle_{L^2_{x,v}} \geqslant \nu_5^{\Lambda} \left\| \partial^j_l h \right\|_{\Lambda}^2 - \nu_6^{\Lambda} \left\| h \right\|_{H^{s-1}_{x,v}},$$

where  $\nu_5^{\Lambda}$  and  $\nu_6^{\Lambda}$  are strictly positive constants depending on L and d. We also define a new norm on  $H_{x,v}^s$ :

$$\|.\|_{H^s_\Lambda} = \left(\sum_{|j|+|l|\leqslant s} \left\|\partial_l^j.\right\|_\Lambda^2\right)^{1/2}$$

### (H2'): Mixing properties

(H2) extends to higher Sobolev's spaces: for all  $s \ge 1$ , for all |j| + |l| = s such that  $|j| \ge 1$ ,

$$\forall \delta > 0 \ , \ \exists C(\delta) > 0 \ , \ \forall h \in H^s_{x,v} \ , \quad \langle \partial_l^j K(h), \partial_l^j h \rangle_{L^2_{x,v}} \leqslant C(\delta) \left\| h \right\|_{H^{s-1}_{x,v}}^2 + \delta \left\| \partial_l^j h \right\|_{L^2_{x,v}}^2.$$

All the constants are explicit thanks to the works of C. Baranger and C. Mouhot [4] and C. Mouhot [79].

The first important result derive in Chapter 4, is the fact that the linear part of the Boltzmann operator generates a contraction semigroup in  $H^s_{x,v}$ .

**Theorem 1.3.1** If L is a linear operator satisfying the conditions (H1'), (H2') and (H3) then there exists  $0 < \varepsilon_d \leq 1$  such that for all s in  $\mathbb{N}^*$ ,

- 1. for all  $0 < \varepsilon \leq \varepsilon_d$ ,  $G_{\varepsilon} = \varepsilon^{-2}L \varepsilon^{-1}v \cdot \nabla_x$  generates a  $C^0$ -semigroup on  $H^s_{x,v}$ .
- 2. there exist  $C_G^{(s)} > 0$  and a norm  $\|\cdot\|_{\mathcal{H}^s_{\varepsilon}}$  such that for all  $0 < \varepsilon \leq \varepsilon_d$ :

$$\left\|\cdot\right\|_{\mathcal{H}^s_{\varepsilon}}^2 \sim \left(\left\|\cdot\right\|_{L^2_{x,v}}^2 + \sum_{|l|\leqslant s} \left\|\partial_l^0\cdot\right\|_{L^2_{x,v}}^2 + \varepsilon^2 \sum_{\substack{|l|+|j|\leqslant s\\|j|\geqslant 1}} \left\|\partial_l^j\cdot\right\|_{L^2_{x,v}}^2\right),$$

and for all h in  $H^s_{x,v}$ ,

$$\langle G_{\varepsilon}(h), h \rangle_{\mathcal{H}^{s}_{\varepsilon}} \leqslant -C_{G}^{(s)} \|h - \pi_{G_{\varepsilon}}(h))\|_{H^{s}_{\Lambda}}^{2}.$$

The modified norm  $\|\cdot\|_{\mathcal{H}^s_{\varepsilon}}$  is dependent on  $\varepsilon$ . We can however make two remarks.

- 1. The dependence on  $\varepsilon$  only appears in front of v-derivatives which disappear in the process of the hydrodynamical limit since only integral against the v variable are of interest.
- 2. In the next subsection, another norm is constructed and do not involve any  $\varepsilon$  dependencies. With this norm, a same result than Theorem 1.3.1 can be obtained with similar arguments.

#### 1.3.2.2 The perturbed Cauchy problem and trend to equilibrium

The hypocoercivity features of the linear Boltzmann operator and the generation of a strongly continuous semigroup in  $H_{x,v}^s$  discussed in the previous subsection were used by C. Mouhot and L. Neumann [82] to obtain existence, uniqueness and exponential decay to equilibrium in the case  $\varepsilon = 1$ , with constructive methods. Such results were known to exist since the first rigorous studies by S. Ukai [107][108] but the methods of the proof were not constructive and thus did not give explicit statements.

The controls we have on the bilinear remainder term  $\Gamma$  are the following.

(H4): Control on the second order operator

 $\Gamma: L_v^2 \times L_v^2 \longrightarrow L_v^2$  is a bilinear symmetric operator such that for all multi-indexes j and l such that  $|j| + |l| \leq s, s \geq 0$ ,

$$\left| \langle \partial_l^j \Gamma(g,h), f \rangle_{L^2_{x,v}} \right| \leqslant \begin{cases} \left| \mathcal{G}^s_{x,v}(g,h) \| f \|_{\Lambda} &, \text{ if } j \neq 0 \\ \left| \mathcal{G}^s_x(g,h) \| f \|_{\Lambda} &, \text{ if } j = 0 \end{cases}$$

 $\mathcal{G}_{x,v}^s$  and  $\mathcal{G}_x^s$  being such that  $\mathcal{G}_{x,v}^s \leqslant \mathcal{G}_{x,v}^{s+1}$ ,  $\mathcal{G}_x^s \leqslant \mathcal{G}_x^{s+1}$  and satisfying the following property:

$$\exists s_0 \in \mathbb{N} , \forall s \ge s_0 , \exists C_{\Gamma} > 0 , \quad \begin{cases} \mathcal{G}_{x,v}^s(g,h) & \leq C_{\Gamma} \left( \|g\|_{H_{x,v}^s} \|h\|_{H_{\Lambda}^s} + \|h\|_{H_{x,v}^s} \|g\|_{H_{\Lambda}^s} \right) \\ \mathcal{G}_{x}^s(g,h) & \leq C_{\Gamma} \left( \|h\|_{H_{x}^s L_{v}^2} \|g\|_{H_{\Lambda}^s} + \|g\|_{H_{x}^s L_{v}^2} \|h\|_{H_{\Lambda}^s} \right). \end{cases}$$

The uniform Cauchy theory we present in Chapter 4 is an extension of the results derived in [82] to obtain estimates that are uniform in the Knudsen number. However, in the case  $\varepsilon = 1$ , the linear part  $G_1$  and the bilinear remainder term  $\Gamma$  are of the same order and can be compared. The main difficulty for general Knudsen number is the fact that the linear part  $G_{\varepsilon}$  generates a contraction semigroup with a spectral gap of order O(1)whereas the bilinear part is of order  $O(\varepsilon^{-1})$ . This makes impossible to consider  $\varepsilon^{-1}\Gamma$  as a mere error term since it explodes as  $\varepsilon$  goes to 0. Our main contributions are

• A method mixing the hypocoercivity properties of the linear operator L with the *a* priori estimates on the bilinear operator, in particular thanks to an orthogonality property of the symmetrised operator  $\Gamma$ .

### (H5): Orthogonality to the Kernel of the linear operator

$$\forall h, g \in \text{Dom}(\Gamma) \cap L^2_v, \quad \Gamma(g,h) \in \text{Ker}(L)^{\perp}.$$

• The construction of a new norm in  $H^s_{x,v}$  combining the idea of [82] and [56] to study both the microscopic and the fluid part of the solution.

The main result is the following theorem.

**Theorem 1.3.2** Let Q be a bilinear operator such that:

- the equation (1.3.4) admits an equilibrium  $0 \leq \mu \in L^1(\mathbb{T}^d \times \mathbb{R}^d)$ ,
- the linearized operator L = L(h) around  $\mu$  with the scaling  $f = \mu + \varepsilon \mu^{1/2} h$  satisfies (H1'), (H2') and (H3),
- the bilinear remaining term  $\Gamma = \Gamma(h, h)$  in the linearization satisfies (H4) and (H5).

Then

• there exists  $0 < \varepsilon_d \leq 1$  and a norm  $\|\cdot\|_{\mathcal{H}^s_{\varepsilon\perp}}$  such that for any  $s \geq s_0$  (defined in (H4)) and any  $0 < \varepsilon \leq \varepsilon_d$ ,  $\|\cdot\|_{\mathcal{H}^s_{\varepsilon\perp}} \sim \|\cdot\|_{H^s_{x,v}}$ , independently of  $\varepsilon$ , • there exist  $\delta_s > 0$ ,  $C_s > 0$  and  $\tau_s > 0$  such that for all  $0 < \varepsilon \leq \varepsilon_d$ :

For any distribution  $0 \leq f_{in} \in L^1(\mathbb{T}^d \times \mathbb{R}^d)$  with  $f_{in} = \mu + \varepsilon \mu^{1/2} h_{in} \geq 0$ ,  $h_{in}$  in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$ and

$$\|h_{in}\|_{\mathcal{H}^{s}_{c+}} \leq \delta_s,$$

there exists a unique global smooth (in  $H^s_{x,v}$ , continuous in time) solution  $f_{\varepsilon} = f_{\varepsilon}(t, x, v)$ to (1.3.4) which, moreover, satisfies  $f_{\varepsilon} = \mu + \varepsilon \mu^{1/2} h_{\varepsilon} \ge 0$  with:

$$\|h_{\varepsilon}\|_{\mathcal{H}^{s}_{\varepsilon^{\perp}}} \leqslant \delta_{s} e^{-\tau_{s} t},$$

We emphasize here that this Theorem is more general than just the case of the Boltzmann equation. It is indeed applicable to several other kinetic models such as the linear relaxation, the semi-classical relaxation, the linear Fokker-Planck equation and the Landau equation with hard and moderately soft potential.

### 1.3.2.3 The limit towards the incompressible Navier-Stokes equations

Theorem 1.3.2 implies that the sequence  $(h_{\varepsilon})_{0 < \varepsilon \leq \varepsilon_d}$  is bounded in  $L_t^{\infty} H_{x,v}^s$ . Such a boundedness property is enough (see [8]) to obtain a weak convergence result  $h_{\varepsilon} \rightharpoonup h$  in distributions as  $\varepsilon$  goes to 0 with

1. h is in Ker(L), so of the form

$$h(t, x, v) = \left[\rho(t, x) + v \cdot u(t, x) + \frac{1}{2}(|v|^2 - d)\theta(t, x)\right]\mu(v)^{1/2},$$

- 2.  $(\rho_{\varepsilon}, u_{\varepsilon}, \theta_{\varepsilon})$  converges weakly-\* in  $L_t^{\infty}(H_x^s)$  towards  $(\rho, u, \theta)$ ,
- 3.  $(\rho, u, \theta)$  satisfies the incompressible Navier-Stokes equations (1.3.2) in the Leray sense [66] as well as the Boussineq equation (1.3.3).

In fact this convergence is strong and Chapter 4 gives explicit rates of convergence.

**Theorem 1.3.3** Consider  $s \ge s_0$  and  $h_{in}$  in  $H^s_{x,v}$  such that  $||h_{in}||_{\mathcal{H}^s_x} \le \delta_s$ .

Then,  $(h_{\varepsilon})_{\varepsilon>0}$  exists for all  $0 < \varepsilon \leq \varepsilon_d$  and converges weakly\* in  $L^{\infty}_t(H^s_x L^2_v)$  towards h such that  $h \in Ker(L)$ , with  $\nabla_x \cdot u = 0$  and  $\rho + \theta = 0$ .

Furthermore,  $\int_0^T h dt$  belongs to  $H_x^s L_v^2$  and it exists C > 0 such that,

$$\left\|\int_{0}^{+\infty} h dt - \int_{0}^{+\infty} h_{\varepsilon} dt\right\|_{H^{s}_{x}L^{2}_{v}} \leq C\sqrt{\varepsilon |ln(\varepsilon)|}$$
One can have a strong convergence in  $L^2_{[0,T]}H^s_x L^2_v$  only if  $h_{in}$  is in Ker(L) with  $\nabla_x \cdot u_{in} = 0$ and  $\rho_{in} + \theta_{in} = 0$  (initial layer conditions).

Moreover, in that case we have

$$\|h - h_{\varepsilon}\|_{L^{2}_{[0,+\infty)}H^{s}_{x}L^{2}_{v}} \leq C\sqrt{\varepsilon |ln(\varepsilon)|},$$

and for all  $\delta$  in [0,1], if  $h_{in}$  belongs to  $H_x^{s+\delta}L_v^2$ ,

$$\sup_{t\in[0,+\infty)}\|h-h_{\varepsilon}\|_{H^{s}_{x}L^{2}_{v}}(t)\leqslant C\varepsilon^{\min(\delta,1/2)}.$$

This theorem gives a strong convergence for  $(\rho_{\varepsilon}, u_{\varepsilon}, \theta_{\varepsilon})$  towards  $(\rho, u, \theta)$  but above all it gives us that  $(\rho, u, \theta)$  is the solution to the incompressible Navier-Stokes equations together with the Boussineq equation satisfying the initial conditions:

- $u(0,x) = Pu_{in}(x)$ , where  $Pu_{in}(x)$  is the divergence-free part of  $u_{in}(x)$ ,
- $\rho(0,x) = -\theta(0,x) = \frac{1}{2}(\rho_{in}(x) \theta_{in}(x)).$

A similar convergence was known to exist (see [10]) in the case where the spatial domain was  $\mathbb{R}^d$ , but did not require any integration in time. Our main contribution was to adapt the arguments to the case of the torus where the integration in time is compulsory to control the Fourier transform of the semigroup generated by  $G_{\varepsilon}$  that was derived in [39].

#### **1.3.3** Hydrodynamical limit in polynomial weighted spaces (Chapter 5)

This work has been done jointly with Sara Merino-Aceituno and Clément Mouhot, both from the University of Cambridge.

The aim of Chapter 5 is to extend the previous semigroup properties of the linear part  $G_{\varepsilon}$  and the Cauchy theory for the full perturbed equation to more general space. The ultimate goal is to derive those results, uniform in the Knudsen number, in  $L_v^1 L_x^{\infty} (1 + |v|^2)$ . This space is indeed optimal in the velocity variable, since it incorporates bounded mass and energy densities, in the Boltwmann framework whereas  $L_x^{\infty}$  is problematic for the Navier-Stokes equations. We would therefore be able to construct solutions to the incompressible Navier-Stokes equations in  $L_x^{\infty}$  via the Boltzmann equation and its hydrodynamical limit. Here again we hope to use constructive arguments and obtain explicit rates of convergence.

This aim has not been achieved yet but it is still a work in progress with Sara Merino-Aceituno and Clément Mouhot. We so far managed to drop the strong exponential weight for a polynomial one, almost optimal, and we can deal with spaces without any derivatives in v for the Cauchy problem. The semigroup properties are extended in all Lebesgue and Sobolev spaces with a polynomial weight  $(1 + |v|)^k$ , for k large enough and k > 2 in the  $L_v^1$  case. Chapter 5 presents our joint work.

We recall that we are working on the dimensionless Boltzmann equation

$$\partial_t f_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x f_{\varepsilon} = \frac{1}{\varepsilon^2} Q(f_{\varepsilon}, f_{\varepsilon}) , \text{ on } \mathbb{T}^N \times \mathbb{R}^N, \qquad (1.3.7)$$

under the linearization  $f_{\varepsilon}(t, x, v) = \mu(v) + \varepsilon h_{\varepsilon}(t, x, v)$ , which leads to the perturbed Boltzmann equation

$$\partial_t h_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x h_{\varepsilon} = \frac{1}{\varepsilon^2} \mathcal{L}(h_{\varepsilon}) + \frac{1}{\varepsilon} Q(h_{\varepsilon}, h_{\varepsilon}), \qquad (1.3.8)$$

where we defined

$$\mathcal{L}(h) = 2Q(\mu, h).$$

Note that we will use curly letters for operators in that Section and standard ones to talk about the restrictions of these operators to  $H_{x,v}^s(\mu^{-1/2})$ . For instance, we recover the operator of previous section

$$(\mathcal{G}_{\varepsilon})|_{H^s_{x,v}(\mu^{-1/2})} = G_{\varepsilon}$$

We still consider the Boltzmann equation with hard potential or Maxwellian potential  $(\gamma = 0)$ , that is to say

$$B\left(|v - v_*|, \cos \theta\right) = \Phi\left(|v - v_*|\right) b\left(\cos \theta\right), \qquad (1.3.9)$$

with  $\Phi$  and b be positive functions. This hypothesis is satisfied for all physical model and is more convenient to work with but do not impede the generality of our results.

We also restrict ourselves to the case of hard potential or Maxwellian potential ( $\gamma = 0$ ), that is to say there is a constant  $C_{\Phi} > 0$  such that

$$\Phi(z) = C_{\Phi} z^{\gamma}, \quad \gamma \in [0, 1], \tag{1.3.10}$$

with a strong form of Grad's *angular cutoff* (see [48]), expressed here by the fact that we assume b to be  $C^1$  with the controls from above

$$\forall z \in [-1, 1], \quad b(z), \ b(z') \leqslant C_b.$$
 (1.3.11)

#### 1.3.3.1 Semigroup properties in Lebesgue and Sobolev spaces

In a recent article [51], an abstract extension theorem allows, under certain assumptions, to extend semigroup properties from a space E into a larger space  $\mathcal{E}$ . The latter theorem allowed to prove that  $\mathcal{G}_1$  generates a strongly continuous semigroup in Lebesgue and Sobolev spaces with polynomial weight [51].

In the same spirit, we show that  $\mathcal{G}_{\varepsilon}$  generates a strong continuous semigroup in Lebesgue and Sobolev spaces of the form  $W_v^{\alpha,1}W_x^{\beta,p}\left(1+|v|\right)^k\right)$  for  $\alpha \leq \beta$  and k large enough with explicit thresholds independent of  $\varepsilon$ . It is done by starting from existing results in  $H_{x,v}^s\left(\mu^{-1/2}\right)$  and then decomposing the linear operator  $\mathcal{G}_{\varepsilon}$  into a dissipative part and a regularising part that are then treated in larger and larger spaces up to the space where the semigroup properties have been derived in previous articles. We thus improve the existing result [23]. Our main contribution is an adapted version of the abstract extension theorem developed in [51] that takes into account the dependencies on the Knudsen number as well as a careful study of the dissipative and the regularising parts of the operator  $\mathcal{G}_{\varepsilon}$ .

**Theorem 1.3.4** Let B be a Boltzmann collision kernel satisfying (1.3.9)-(1.3.10)-(1.3.11). There exists  $0 < \varepsilon_d \leq 1$  such that for all p, q in  $[1, +\infty]$ , all  $\alpha$ ,  $\beta$  in  $\mathbb{N}$  with  $\alpha \leq \beta$  and all  $k > k_q^*$ , where

$$k_q^* = \frac{3 + \sqrt{49 - 48/q}}{2} + \gamma \left(1 - \frac{1}{q}\right), \qquad (1.3.12)$$

- 1. for all  $0 < \varepsilon \leq \varepsilon_d$ ,  $\mathcal{G}_{\varepsilon} = \varepsilon^{-2} \mathcal{L} \varepsilon^{-1} v \cdot \nabla_x$  generates a  $C^0$ -semigroup,  $S_{\mathcal{G}_{\varepsilon}}(t)$ , on  $W_v^{\alpha,q} W_x^{\beta,p}(\langle v \rangle^k)$ ,
- 2. for all  $\tau > 0$ , there exist  $C_{\mathcal{G}}(\tau)$ ,  $\lambda_0 > 0$ , such that for all  $0 < \varepsilon \leq \varepsilon_d$  and for all  $h_{in}$ in  $W_v^{\alpha,q} W_x^{\beta,p}(\langle v \rangle^k)$ , for all  $t \ge \tau$

$$\|S_{\mathcal{G}_{\varepsilon}}(t)(h_{in}) - \Pi_{\mathcal{G}}(h_{in})\|_{W_{v}^{\alpha,q}W_{x}^{\beta,p}(\langle v \rangle^{k})} \leq C_{\mathcal{G}}(\tau)e^{-\lambda_{0}t} \|h_{in} - \Pi_{\mathcal{G}}(h_{in})\|_{W_{v}^{\alpha,q}W_{x}^{\beta,p}(\langle v \rangle^{k})},$$

where  $\Pi_{\mathcal{G}}$  is the spectral projector onto Ker  $(\mathcal{G}_{\varepsilon})$  which is given, for all  $\varepsilon$ , by

$$\Pi_{\mathcal{G}}(g) = \sum_{i=0}^{d+1} \left( \int_{\mathbb{T}^d \times \mathbb{R}^d} g\phi_i \, dx dv \right) \phi_i \mu, \qquad (1.3.13)$$

where  $\phi_0(v) = 1$ , for i = 1, ..., d we defined  $\phi_i(v) = v_i$  and  $\phi_{d+1} = \left(|v|^2 - d\right) / \sqrt{2d}$ . The constants  $\varepsilon_d$ ,  $C_{\mathcal{G}}(\tau)$  and  $\lambda_0$  are constructive and only depends on d, p, q, k,  $\alpha$ ,  $\beta$  and the kernel of the Boltzmann operator.

The rate of decay  $\lambda_0$  can be taken equal to the spectral gap of  $\mathcal{L}|_{H^s_{x,v}(\mu^{-1/2})}$  (see [23]), for s as large as wanted, when k is large enough (and we obtained a constructive threshold).

Finally, we emphasize that in the case q = 1, the result holds for all k > 2. This is almost the minimal regularity  $L_v^2 \left(1 + |v|^2\right)$  for the Boltzmann equation, that is solutions with bounded mass and energy.

# 1.3.3.2 Cauchy problem and exponential decay in Sobolev spaces with polynomial weight

The second part of Chapter 5 deals with the uniform, in the Knudsen number, Cauchy problem and the exponential decay towards equilibrium in larger spaces than the exponential weight framework we dealt with in Chapter 4.

The spaces where we developed our theory are of the following forms

$$W_v^{\alpha,1} W_x^{\beta,1} \left( 1 + |v|^{2+0} \right)$$
 and  $W_v^{\alpha,1} H_x^{\beta} \left( 1 + |v|^{2+0} \right)$ ,

for s large enough and all  $\alpha \leq \beta$ . This improves the Cauchy theory developed in Chapter 4 by dropping the exponential weight and the v-derivatives. Again, the polynomial weight is almost the optimal one for the Boltzmann equation (conservation of mass and energy).

Such results have recently been obtained [51] for fixed  $\varepsilon$ , in which case the rate of decay of the semigroup generated by  $\mathcal{G}_{\varepsilon}$  is of the same order than the remainder term Q(h, h). However, in order to obtain uniform results we have to handle the remainder term  $\varepsilon^{-1}Q$ and it cannot be treated as a mere perturbation that evolves under the flow of  $S_{\mathcal{G}_{\varepsilon}}$ , the semigroup generated by  $\mathcal{G}_{\varepsilon}$ , since the latter has an exponential decay of order O(1).

Our main contribution is a new analytic point of view about the extension theorem in [51] and includes the bilinear term. More precisely, we decompose the perturbed equation (1.3.8) into a hierarchy of equations taking place in spaces that have more and more regularity up to  $H_{x,v}^s(\mu^{-1/2})$  where estimates had been derived in Chapter 4. At each step we use the dissipative part of the linear operator to control the remainder term  $\varepsilon^{-1}Q$  whereas the regularising part is controlled in spaces with higher regularity.

We hence state the following theorem tackling the Cauchy problem and the exponential convergence towards the equilibrium  $\mu$ .

**Theorem 1.3.5** Let B be a Boltzmann collision kernel satisfying (1.3.9)-(1.3.10)-(1.3.11)and let p = 1 or p = 2.

There exists  $0 < \varepsilon_d \leq 1$  and  $\beta_0$  in  $\mathbb{N}$  such that

• for all  $\alpha$ ,  $\beta$  in  $\mathbb{N}$  such that  $\beta \ge \beta_0$  and  $\alpha \le \beta$  and for all k > 2 define

$$\mathcal{E}^p = W_v^{\alpha,1} W_x^{\beta,p} \left( \langle v \rangle^k \right)$$

• for any  $\lambda'_0$  in  $(0, \lambda_0)$  ( $\lambda_0$  defined in Theorem 1.3.4) there exist  $C_{\alpha,\beta}$ ,  $\eta_{\alpha,\beta} > 0$  such that for any  $0 < \varepsilon \leq \varepsilon_d$ , for any distribution  $f_{in} = \mu + \varepsilon h_{in} \geq 0$ :

If

(i)  $h_{in}$  is in  $\operatorname{Ker}(\mathcal{G}_{\varepsilon})^{\perp}$  in  $\mathcal{E}^p$ ,

(*ii*)  $||h_{in}||_{\mathcal{E}^p} \leq \eta_{\alpha,\beta}$ ,

Then there exists a unique global solution  $f_{\varepsilon} = f_{\varepsilon}(t, x, v)$  to (1.3.7) in  $\mathcal{E}^p$  which, moreover, satisfies  $f_{\varepsilon} = \mu + \varepsilon h_{\varepsilon} \ge 0$  with:

- $h_{\varepsilon}$  belongs to  $\operatorname{Ker}(\mathcal{G}_{\varepsilon})^{\perp}$  for all times,
- •

$$\|h_{\varepsilon}\|_{\mathcal{E}^p} \leqslant C_{\alpha,\beta} \|h_{in}\|_{\mathcal{E}^p} e^{-\lambda_0' t}.$$

The constants  $C_{\alpha,\beta}$  and  $\eta_{\alpha,\beta}$  are constructive and depends only on  $\alpha$ ,  $\beta$ , k, d,  $\lambda'_0$  and the kernel of the Boltzmann operator.

# 1.4 A quantic version of Boltzmann equation (Part III)

#### 1.4.1 The Boltzmann-Nordheim equation

As we mentioned before, the Boltzmann equation describes, at a mesoscopic level, the dynamics of a monoatomistic rarefied gas with elastic collisions. There exists different modifications of this kinetic model, for polyatomistic gases for instance (see [29]).

For all these models, the Boltzmann equation arises from microscopic behaviours ruled by classical physics, where the probability of two particles colliding depends only on the number of particles moving at the incoming velocities. The case of quantum mechanics is rather different since the probability of two particles colliding not only depends on the velocity of the particles undergoing the collision but also the outcoming velocity the collision yields. We refer to [32] Chapter 17 for more details.

Using quatum statistical physics instead of classical statistical physics, Nordheim [89] derived a quantic version of the Boltzmann equation for bosons and fermions.

The latter evolution equation is called the Boltzmann-Nordheim equation and reads as follow, with the usual shorthand notations.

$$\partial_t f + v \cdot \nabla_x f = Q_\alpha(f),$$

with

$$Q_{\alpha}(f) = \int_{\mathbb{R}^N \times \mathbb{S}^{d-1}} B(v, v_*, \theta) \left[ f'(1 + \alpha f) f'_*(1 + \alpha f_*) - f(1 + \alpha f') f_*(1 + \alpha f'_*) \right] dv_* d\sigma.$$

The Boltzmann-Nordheim equation thus rules the dynamics of the distribution of particles for a dilute quantum gas of bosons ( $\alpha = 1$ ) or fermions ( $\alpha = -1$ ). Note that in the classical case  $\alpha = 0$  one recovers the Boltzmann equation. In Chapter 6, our study applies to the case where the collision kernel B is hard potential with angular cutoff. More precisely, the collision operator is supposed to satisfy the following properties.

- 1.  $B(v, v_*, \theta) = \Phi(|v v_*|) b(\cos \theta)$ ,
- 2. there exist  $C_{\Phi} > 0$  and  $\gamma$  in [0,1] such that  $\Phi(z) = C_{\Phi} z^{\gamma}$ ,
- 3.  $(b \circ \cos)$  is continuous on  $(0, \pi)$  and integrable on the sphere:

$$l_b = \int_{\mathbb{S}^{d-1}} b\left(\cos\theta\right) d\sigma = \left|\mathbb{S}^{d-2}\right| \int_0^\pi b\left(\cos\theta\right) \sin^{d-2}\theta \, d\theta < +\infty.$$

Moreover, we restrain ourselves to the spatially homogeneous case for a gas of bosons

$$\partial_t f + v \cdot \nabla_x f = Q(f), \tag{1.4.1}$$

with

$$Q(f) = \int_{\mathbb{R}^N \times \mathbb{S}^{d-1}} B(v, v_*, \theta) \left[ f'(1+f) f'_*(1+f_*) - f(1+f') f_*(1+f'_*) \right] dv_* d\sigma. \quad (1.4.2)$$

#### 1.4.2 Cauchy problem and the Bose-Einstein condensate (Chapter 6)

The Boltzmann-Nordheim collision operator (1.4.2) is in fact the addition of the classical Boltzmann collision operator with a trilinear operator. If some properties of the classical Boltzmann equation still hold true for the Boltzmann-Nordheim equation, such as the *a priori* preservation of mass, momentum and energy

$$\int_{\mathbb{R}^d} \begin{pmatrix} 1\\ v\\ |v|^2 \end{pmatrix} f(v) \, dv = \int_{\mathbb{R}^d} \begin{pmatrix} 1\\ v\\ |v|^2 \end{pmatrix} f_0(v) \, dv,$$

the trilinear term implies rather different behaviours.

Indeed, physical observations and numerical simulations (see [40] for an overview of these results) in the isotropic setting f(t, v) = f(t, |v|) showed that there exists a critical temperature  $T_c(M_0)$ , depending on the mass  $M_0$  of the bosonic gas. If the temperature of the initial datum  $f_{in}$  is below  $T_c(M_0)$  then the solution of the Boltzmann-Nordheim equation will develop a dirac mass at |v| = 0 in finite time. This blow-up phenomenon is known as the Bose-Einstein condensate.

From the mathematical point of view, the only rigorously known results focused on the isotropic framework. X. Lu [69][70][71] built solutions in  $L_2^1$  and proved a Cauchy theory for measures. He also proved, with not entirely constructive methods, a concentration

phenomenon for subcritical temperature in the limit t goes to infinity. We emphasize here that this asymptotic result does not imply the appearance of a Bose-Einstein condensate in finite time. Recently, other isotropic solutions have been constructed in  $L^1(1 + |v|^{3+0})$ by M. Escobedo and J. J. L. Velázquez [40]. Moreover, they made a major breakthrough by proving the appearance of a Bose-Einstein condensate in finite time under some assumptions on the solution [40] and for subcritical temperatures [41].

In Chapter 6 we develop a local in time Cauchy theory in the non-isotropic setting in  $L_2^1 \cap L^\infty$ . The latter space is the most general one can hope for a Cauchy theory that catches the Bose-Einstein condensate. Solutions are indeed physically expect to have finite mass and energy and the creation of a dirac mass creates a blow-up in  $L^\infty$  whereas it only leads to a loss of mass in  $L^1$ .

Our main contributions are a new version of Povzner inequality [94], which bounds the evolution of convex functions through a collision, and a new control on the operator  $Q^+$  for high and small relative velocities  $v - v_*$ . We also control the higher moments of solutions to the Boltzmann-Nordheim equation and derive a precise estimate on the blow-up of the  $(2 + \gamma)^{th}$  moment of solutions at time t = 0, in the spirit of [77], to obtain uniqueness.

We denote, for all s and t in  $\mathbb{R}^+$ ,

$$M_s(t) = \int_{\mathbb{R}^d} |v|^s f(t, v) \, dv$$

the  $s^{th}$  moment of a function f(t, v). The main result of Chapter 6 is the following Cauchy theorem for the Boltzmann-Nordheim equation for bosons.

**Theorem 1.4.1** Let  $f_0(v)$  be in  $L^1_{2,v} \cap L^{\infty}_v$ .

Then there exists  $T_0 > 0$ , depending only on  $C_{\Phi}$ ,  $l_b$ ,  $\gamma$ ,  $||f_0||_{L^1_{2,v}}$  and  $||f_0||_{L^{\infty}_v}$ , such that there exists a unique f in  $L^{\infty}_{loc}([0, T_0), L^1_{2,v} \cap L^{\infty}_v)$  solution of (6.1.3) on  $[0, T_0) \times \mathbb{R}^d$  that preserves mass and energy.

Moreover, this solution satisfies

- $\bullet \ T_0=+\infty \quad or \quad \lim_{T\rightarrow T_0^-}\|f\|_{L^\infty_{[0,T]\times \mathbb{R}^d}}=+\infty,$
- f preserves the momentum of  $f_0$ ,
- for all s > 0 and for all  $0 < T < T_0$ ,

$$M_s(t) \in L^{\infty}_{loc}([T, T_0)).$$

• for all  $T < T_0$ ,

$$\sup_{[0,T]\times\mathbb{R}^d} \left( f(t,v) + \int_0^t \left(1 + |v|^\gamma\right) f(s,v) \, ds \right) < \infty.$$

Let us mention here that Theorem 6.2.1 implies a Bose-Einstein concentration phenomenon as time goes to infinity for subcritical initial data if they are globally defined thanks to the work of Lu ([71] Theorem 2).

The latter argument is however non explicit and does not prove any blow-up in finite time whereas [40] gives the apparition of the Bose-Einstein condensate in finite time in the isotropic setting. A work in progress is the proof of the creation of a condensate in finite time in our more general framework.

# Appendices

# 1.A Notations

We will work in different function spaces. We gather in this appendix the different notations we will use throughout the sequel.

We first emphasize the fact that we consider that 0 belongs to the following sets:  $\mathbb{N}$ ,  $\mathbb{Z}^+$ ,  $\mathbb{Q}^+$ ,  $(\mathbb{R} - \mathbb{Q})^+$  and  $\mathbb{R}^+$ .

We then define the following shorthand notation,

$$\langle \cdot \rangle = \sqrt{1 + |\cdot|^2}.$$

### **1.A.1** Function spaces for one variable

Here, the term "variable" has to be understood as being in a particular vector space of dimension N, namely  $\mathbb{R}^N$ . Basically, when there is not any combination of time, space and velocity spaces.

Let p be in  $[1, +\infty)$ , q in  $\mathbb{R}$ , s in  $\mathbb{R}^+$  and  $m : \mathbb{R}^N \longrightarrow \mathbb{R}^+$  a strictly positive measurable function.

Weighted Lebesgue spaces. We define the space  $L^{p}(m)$  by the norm

$$||f||_{L^p(m)} = \left[\int_{\mathbb{R}^d} |f(y)|^p \, m(y)^p \, dy\right]^{\frac{1}{p}},$$

and the space  $L^{\infty}(m)$  by the norm

$$\left\|f\right\|_{L^{\infty}(m)} = \sup_{y \in \mathbb{R}^{N}} \left(\left|f(y)\right| m(y)\right).$$

In the case when  $m(y) = \langle y \rangle$  is a power of  $\langle \cdot \rangle$  we use the shorthand notations

$$L_{q}^{p} = L^{p}(m^{q})$$
 and  $L_{q}^{\infty} = L^{\infty}(m^{q})$ .

Weighted Sobolev spaces In the case where s is a natural number, for any multi-index  $k = (k_1, \ldots, k_N)$  in  $\mathbb{N}^N$  we denote

• the  $k^{th}$  partial derivative by

$$\partial^k = \frac{\partial^{k_1}}{\partial y_1^{k_1}} \cdots \frac{\partial^{k_N}}{\partial y_N^{k_N}},$$

- for i in  $\{1, \ldots, N\}$  we denote by  $c_i(k)$  the  $i^{th}$  coordinate of k,
- the length of k will be written  $|k| = \sum_{i} c_i(k)$ ,
- the multi-index  $\delta_{i_0}$  by :  $c_i(\delta_{i_0}) = 1$  if  $i = i_0$  and 0 elsewhere.

With these conventions, we define the space  $W^{s,p}(m)$  by the norm

$$\|f\|_{W^{s,p}(m)} = \left[\sum_{|k| \leqslant s} \left\|\partial^k f\right\|_{L^p(m)}^p\right]^{\frac{1}{p}},$$

and the space  $W^{\infty,p}(m)$  by the norm

$$\|f\|_{W^{s,\infty}(m)} = \sum_{|k| \leqslant s} \left\|\partial^k f\right\|_{L^{\infty}(m)}.$$

In the case  $m(y) = \langle y \rangle$  we use the obvious shorthand notations  $W_q^{s,p}$  and  $W_q^{s,\infty}$ .

These definitions can be extended by interpolation, or *via* the theory of Fourier transform, to the case s in  $\mathbb{R}^+$ .

In the particular case p = 2, we will write  $H^{s}(m) = W^{s,2}(m)$  and  $H_{q}^{s} = W_{q}^{s,2}$ .

### 1.A.2 Function spaces for several variables

In the case where the functions we consider are functions of time, space and velocity we need distinctive notations. The convention we chose is to index the space by the name of the concerned variable. For instance, for a measurable function

$$f(t, x, v) : [0, T) \times \Omega \times \mathbb{R}^d \longrightarrow \mathbb{R}^+,$$

with  $\Omega \subset \mathbb{R}^d$ , we will denote for p in  $[1, +\infty]$ 

$$L_t^p = L^p([0,T)), \quad L_x^p = L_x^p(\Omega) \text{ and } L_v^p = L^p(\mathbb{R}^d).$$

We extend these notations verbatim to weighted Lebesgue and weighted Sobolev spaces.

In the case of norm involving all the different variables we need new definitions. We consider functions f(x, v) defined in  $\Omega \times \mathbb{R}^d$  with  $\Omega \subset \mathbb{R}^d$ .

Let p and q be in  $[1, +\infty)$ ,  $\alpha$  and s in  $\mathbb{R}^+$  and  $m : \mathbb{R}^d \longrightarrow \mathbb{R}^+$  a strictly positive measurable function.

In the case where s is a natural number, for any multi-indexes  $j = (j_1, \ldots, j_N)$  and  $l = (l_1, \ldots, l_N)$  in  $\mathbb{N}^N$  we denote the  $(j, l)^{th}$  partial derivative by

$$\partial_l^j = \partial_x^l \partial_v^j$$

with multi-index partial derivatives being defined in previous subsection. We define the space  $W_v^{\alpha,q} W_x^{\beta,p}(m)$  by the norm

$$\|f\|_{W_v^{\alpha,q}W_x^{\beta,p}(m)} = \sum_{\substack{|j| \leq \alpha, |l| \leq \beta \\ |l|+|j| \leq \max(\alpha,\beta)}} \left\|\partial_l^j f\right\|_{L_v^q L_x^p(m)}.$$

We emphasize here that in the case  $\alpha = \beta$  and p = q this definition is equivalent to the  $W_{x,v}^{\beta,p}(m)$ -norm on  $\Omega \times \mathbb{R}^d$  we defined in the previous subsection. Again, in the particular case p = 2 or q = 2 we will use the notations, respectively,  $H_x^{\beta}$  and  $H_v^{\beta}$ .

Part I

# QUALITATIVE AND QUANTITATIVE STUDY OF SOLUTIONS TO THE BOLTZMANN EQUATION

# Chapter 2

# Instantaneous filling of the vacuum for the full Boltzmann equation in bounded domains

We prove the immediate appearance of a lower bound for continuous mild solutions to the full Boltzmann equation in the torus or a  $C^2$  convex domain with specular boundary conditions, under the sole assumption of regularity of the solution. We investigate a wide range of collision kernels, some satisfying Grad's cutoff assumption and others not. We show that this lower bound is exponential, independent of time and space with explicit constants depending only on the a priori bounds on the solution. In particular, this lower bound is Maxwellian in the case of cutoff collision kernels. A thorough study of characteristic trajectories, as well as a geometric approach of grazing collisions against the boundary are derived.

These results are entirely constructive if the domain is  $C^3$  and strictly convex.

# Contents

<ul> <li>2.1.1</li> <li>2.1.2</li> <li>2.1.3</li> <li>Main</li> <li>2.2.1</li> <li>2.2.2</li> <li>2.2.3</li> <li>2.2.4</li> </ul>	Motivations and comparison with previous results       !         Our strategy       !         Organisation of the chapter       !         In results       !         Notations       !         Results about the free transport equation       !
<ul> <li>2.1.2</li> <li>2.1.3</li> <li>Main</li> <li>2.2.1</li> <li>2.2.2</li> <li>2.2.3</li> <li>2.2.4</li> </ul>	Our strategy       §         Organisation of the chapter       6         n results       6         Notations       6         Results about the free transport equation       6         Manual Weak and the state of the
2.1.3 Main 2.2.1 2.2.2 2.2.3 2.2.4	Organisation of the chapter       6         n results       6         Notations       6         Results about the free transport equation       6         Notations
Main 2.2.1 2.2.2 2.2.3 2.2.4	n results       6         Notations       6         Results about the free transport equation       6         Note With the free transport equation       6
<ol> <li>2.2.1</li> <li>2.2.2</li> <li>2.2.3</li> <li>2.2.4</li> </ol>	Notations
$2.2.2 \\ 2.2.3 \\ 2.2.4$	Results about the free transport equation
2.2.3 2.2.4	
2.2.4	Maxwellian lower bound for cutoff collision kernels
	Exponential lower bound for non-cutoff collision kernels
The	free transport equation: proof of Theorem 2.2.1
2.3.1	Study of rebounds on the boundary
2.3.2	Description of characteristics
2.3.3	Existence and uniqueness of solution to $(2.2.5) - (2.2.7)$
	2.3.3.1 Proof of uniqueness
	2.3.3.2 Construction of the solution
	2.3.3.3 Boundary and initial conditions
	2.3.3.4 Time differentiability $\ldots \ldots \ldots$
	2.3.3.5 Space differentiability and solvability of the transport equation
The	cutoff case: localized "upheaval points" 8
2.4.1	Partition of the phase space and first localised lower bounds
2.4.2	A constructive approach to the initial lower bound, Corollary 1 .
2.4.3	A lower bound depending only on the norm of the velocity: strat- egy of the proof of Proposition 2.2.4
The	cutoff case: characteristics passing by a point far from
$\mathbf{the}$	boundary
The	cutoff case: geometry and grazing trajectories
2.6.1	Geometric study of grazing trajectories
2.6.2	Spreading effect along grazing trajectories 10
Max	wellian lower bound in the cutoff case: proof of Theorem
2.2.3	
	2.7.0.1 Proof of Proposition $(2.2.4)$
2.7.1	Proof of Theorem 2.2.3
	2.3.1 2.3.2 2.3.3 The 2.4.1 2.4.2 2.4.3 The the 1 2.6.1 2.6.2 Max 2.2.3 2.7.1

2.8.1	A lower bound only depending on the norm of the velocity $\ . \ .$	112
2.8.2	Proof of Theorem 2.2.6	114

# 2.1 Introduction

This chapter deals with the Boltzmann equation, which rules the behaviour of rarefied gas particles moving in a domain  $\Omega$  of  $\mathbb{R}^d$  with velocities in  $\mathbb{R}^d$  ( $d \ge 2$ ) when the only interactions taken into account are binary collisions. More precisely, the Boltzmann equation describes the time evolution of f(t, x, v), the distribution of particles in position and velocity, starting from an initial distribution  $f_0(x, v)$ .

We investigate the case where  $\Omega$  is either the torus or a  $C^2$  convex bounded domain. The Boltzmann equation reads

$$\forall t \ge 0 \quad , \quad \forall (x,v) \in \Omega \times \mathbb{R}^d, \quad \partial_t f + v \cdot \nabla_x f = Q(f,f),$$

$$\forall (x,v) \in \bar{\Omega} \times \mathbb{R}^d, \quad f(0,x,v) = f_0(x,v),$$

$$(2.1.1)$$

with f being periodic in the case of  $\Omega = \mathbb{T}^d$ , the torus, or with f satisfying the specular reflections boundary condition if  $\Omega$  is a  $C^2$  convex bounded domain:

$$\forall (x,v) \in \partial\Omega \times \mathbb{R}^d, \quad f(t,x,v) = f(t,x,\mathcal{R}_x(v)). \tag{2.1.2}$$

 $\mathcal{R}_x$ , for x on the boundary of  $\Omega$ , stands for the specular reflection at that point of the boundary. One can compute, denoting by n(x) the outward normal at a point x on  $\partial\Omega$ ,

$$\forall v \in \mathbb{R}^d, \quad \mathcal{R}_x(v) = v - 2(v \cdot n(x))n(x).$$

The quadratic operator Q(f, f) is local in time and space and is given by

$$Q(f,f) = \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} B\left(|v - v_*|, \cos \theta\right) \left[f'f'_* - ff_*\right] dv_* d\sigma_*$$

where f',  $f_*$ ,  $f'_*$  and f are the values taken by f at v',  $v_*$ ,  $v'_*$  and v respectively. Define:

$$\begin{cases} v' = \frac{v + v_*}{2} + \frac{|v - v_*|}{2}\sigma \\ v'_* = \frac{v + v_*}{2} - \frac{|v - v_*|}{2}\sigma \end{cases}, \text{ and } \cos\theta = \langle \frac{v - v_*}{|v - v_*|}, \sigma \rangle.$$

The collision kernel  $B \ge 0$  contains all the information about the interaction between two particles and is determined by physics (see [28] or [30] for a formal derivation for the hard sphere model of particles). In this chapter we shall only be interested in the case of B satisfying the following product form

$$B\left(|v - v_*|, \cos \theta\right) = \Phi\left(|v - v_*|\right) b\left(\cos \theta\right), \qquad (2.1.3)$$

which is a common assumption as it is more convenient and also covers a wide range of physical applications. Moreover, we shall assume that  $\Phi$  satisfies either

$$\forall z \in \mathbb{R}, \quad c_{\Phi} |z|^{\gamma} \leqslant \Phi(z) \leqslant C_{\Phi} |z|^{\gamma} \tag{2.1.4}$$

or a mollified assumption

$$\begin{cases} \forall |z| \ge 1 \in \mathbb{R}, \quad c_{\Phi} |z|^{\gamma} \le \Phi(z) \le C_{\Phi} |z|^{\gamma} \\ \forall |z| \le 1 \in \mathbb{R}, \quad c_{\Phi} \le \Phi(z) \le C_{\Phi}, \end{cases}$$
(2.1.5)

 $c_{\Phi}$  and  $C_{\Phi}$  being strictly positive constants and  $\gamma$  in (-d, 1]. The collision kernel is said to be "hard potential" in the case of  $\gamma > 0$ , "soft potential" if  $\gamma < 0$  and "Maxwellian" if  $\gamma = 0$ .

Finally, we shall consider b to be a continuous function on  $\theta$  in  $(0, \pi]$ , strictly positive near  $\theta \sim \pi/2$ , which satisfies

$$b(\cos\theta)\sin^{d-2}\theta \underset{\theta\to 0^+}{\sim} b_0\theta^{-(1+\nu)}$$
(2.1.6)

for  $b_0 > 0$  and  $\nu$  in  $(-\infty, 2)$ . The case when b is locally integrable,  $\nu < 0$ , is referred to by the Grad's cutoff assumption (first introduce in [48]) and therefore B will be said to be a cutoff collision kernel. The case  $\nu \ge 0$  will be designated by non-cutoff collision kernel.

#### 2.1.1 Motivations and comparison with previous results

The aim of this chapter is to show and to quantify the strict positivity of the solutions to the Boltzmann equation when the gas particles move in a bounded domain. This issue has been investigated for a long time since it not only presents a great physical interest but also appears to be of significant importance for the mathematical study of the Boltzmann equation.

Moreover, our results only require some regularity on the solution and no further assumption on its local density, which was assumed to be uniformly bounded from below in previous studies (which is equivalent of assuming *a priori* either that there is no vacuum or that the solution is strictly positive).

More precisely, we shall prove that continuous solutions to the Boltzmann equation with angular cutoff in a  $C^2$  convex bounded domain or the torus which have uniformly bounded energy satisfy an immediate Maxwellian lower bound:

$$\forall t_0 > 0, \ \exists \rho, \theta > 0, \ \forall t \ge t_0, \ \forall (x, v) \in \Omega \times \mathbb{R}^d, \quad f(t, x, v) \ge \frac{\rho}{(2\pi\theta)^{d/2}} e^{-\frac{|v|^2}{2\theta}}.$$

The strict positivity of the solutions to the Boltzmann equation standing in the form of an exponential lower bound was already noticed by Carleman in [26] for the spatially homogeneous equation. In his article he proved that such a lower bound is created immediately in time in the case of hard potential kernels with cutoff in dimension 3. More precisely, the radially symmetric solutions he constructed in [26] satisfies an almost Maxwellian lower bound,

$$\forall t \ge t_0, \forall v \in \mathbb{R}^3, \quad f(t,v) \ge C_1 e^{-C_2 |v|^{2+\varepsilon}},$$

 $C_1, C_2 > 0$  for all  $t_0 > 0$  and  $\varepsilon > 0$ . His breakthrough was to notice that a part  $Q^+$  of the Boltzmann operator Q satisfies a spreading property, roughly speaking

$$Q^+(\mathbf{1}_{B(\bar{v},r)},\mathbf{1}_{B(\bar{v},r)}) \geqslant C_+\mathbf{1}_{B(\bar{v},\sqrt{2}r)},$$

with  $C_+ < 1$  (see Lemma 2.4.2 for an exact statement).

The spreading strategy was used by Pulvirenti and Wennberg in [95] to extend the latter inequality to solutions to the spatially homogeneous Boltzmann equation with hard potential and cutoff in dimension 3 with more general initial data. Their contribution was to get rid of the initial boundedness suggested in [26] by Carleman thanks to the use of an iterative regularity property of the  $Q^+$  operator. This property allowed them to immediately create an "upheaval point" that they then spread with the method of Carleman. Moreover, they obtain an exact Maxwellian lower bound of the form by controlling the decay of  $C^n_+$ 

$$\forall t \ge t_0, \forall v \in \mathbb{R}^3, \quad f(t,v) \ge C_1 e^{-C_2 |v|^2},$$

for all  $t_0 > 0$ .

Finally, Mouhot in [78] dealt with the full Boltzmann equation in the torus. He derived a spreading method that is invariant under the flow of the characteristics, obtaining lower bounds uniformly in space as long as the solution has uniformly bounded density, energy and entropy (for the hard potential case) together with uniform bounds on higher moments (for the soft and Maxwellian potentials case). However, he also implicitly assumed that the initial data had to be bounded from below uniformly in space. He also derived ([78]) the same kind of results in the non-cutoff case in the torus, the immediate appearance of an exponential lower bound of the form

$$\forall t \ge t_0, \forall (x,v) \in \mathbb{T}^d \times \mathbb{R}^d, \quad f(t,v) \ge C_1(\varepsilon) e^{-C_2(\varepsilon)|v|^{K+\varepsilon}},$$

for all  $t_0 > 0$ , all  $\varepsilon > 0$  and  $K = K(\nu)$  with K(0) = 2 (thus recovering the cutoff case in the limit). His idea was to split further the Q operator into a cutoff part and a non-cutoff part that is seen as a small perturbation of his original spreading method.

Our results extend those in [78] in the case of  $C^2$  bounded convex domain. Our main contribution is the derivation of a spreading method that remains invariant under the characteristics flow that, unlike the torus case, changes the direction of velocities at the boundary. Moreover, we emphasize here that the existence of boundaries implies the existence of grazing collisions against the latter, where the strategies developed in [95] and [78] fail. We therefore to derive a geometrical approach to those problematic trajectories.

Furthermore, we do not assume any uniform boundedness on the initial data but we require the continuity of the solution to the Boltzmann equation. However, if we keep the assumptions made in [78] and further assume that the domain is  $C^3$  and strictly convex then our proofs are constructive.

The quantification of the strict positivity, and above all the appearance of an exponential lower bound, has been seen to be of great mathematical interest thanks to the development of the entropy-entropy production method. This method (see [112], Chapter 3, and [113]) provides a useful way of investigating the long-time behaviour of solutions to kinetic equations. Indeed, it has been successfully used to prove convergence to the equilibrium in non-pertubative cases for the Fokker-Planck equation, [36], and the full Boltzmann equation in the torus or in  $C^1$  bounded connected domains with specular reflections, [37]. This entropy-entropy production method requires (see Theorem 2 in [37]) uniform boundedness on moments and Sobolev norms for the solutions to the Boltzmann equation but also an *a priori* exponential lower bound of the form

$$f(t, x, v) \geqslant C_1 e^{-C_2 |v|^q},$$

with  $q \ge 2$ .

Therefore, the present chapter allows us to prove that the latter *a priori* assumption is in fact satisfied for a lot of different cases (see [78], Section 5 for an overview). We also emphasize here that the assumption of continuity of the solution we have made does not reduce the range of applications since a lot more regularity is usually asked for the entropy-entropy production method. Moreover, our method, unlike the ones developed in [95] and [78], does not require a uniform bound on the local density of solutions, which is not a requirement for the entropy-entropy production method either (see [37], Theorem 2).

To conclude we note that our investigations require a deep and detailed understanding of the geometry and properties of characteristic trajectories for the free transport equation. In particular, a geometric approach of grazing collisions against the boundary is derived and is the key ingredient to study the strict positivity of solutions to the Boltzmann equation. The existing strategies as well as our improvements are discussed in the next section.

## 2.1.2 Our strategy

Our strategy to tackle this issue will follow the method introduced by Carleman [26] together with the idea of Mouhot [78] to find a spreading method that will be invariant along the characteristic trajectories. Roughly speaking we shall built characteristics in a  $C^2$  bounded convex domain, create an "upheaval point" (as in [95] and [78]) that we spread and expand uniformly along the characteristics. Finally, once the lower bound can be compared to an exponential one we reach the expected result.

However, the existence of rebounds against the boundary leads to difficulties. We describe them below and point out how we shall overcome them.

Creating an "upheaval point" was achieved, in [95] and [78], by using an iterated Duhamel formula and a regularity property of the collision operator relying on a uniform lower bound of the local density of the function. But the use of this property requires a uniform control along the characteristics of the density, the energy and the entropy of the solutions to the Boltzmann equation which is natural in the homogeneous case but made Mouhot consider initial datum bounded from below uniformly in space. Our way of dealing with the appearance of the "upheaval point" is rather different but includes more general initial datum. We make the assumption of continuity of solutions to the Boltzmann equation and by compactness arguments we can construct a partition of our phase space where initial localised lower bounds exist, i.e., localised "upheaval points".

The case on the torus studied by Mouhot tells us that an exponential lower bound should arise immediately and therefore we expect the same to happen as long as the characteristic trajectory is a straight line. Unfortunately, the possibility for a trajectory to remain a line depends on the distance from the boundary of the starting point, which can be as short as one wants. This thought is the basis of our means for spreading the initial lower bound. We divided our trajectories into two categories, the ones which always stay close to the boundary (grazing collisions) and the others. For the latter we can spread our lower bound uniformly as noticed in [78]. The key contribution of our proof is a thorough investigation of the geometry of grazing collisions. We show that their velocity does not evolve a lot along time and mix it with the spreading property of the collision operator. Notice here that the convexity of  $\Omega$  is needed for the study of grazing trajectories.

The last behaviour to notice is the fact that specular reflections completely change velocities but preserve their norm. Therefore, the existence of rebounds against the boundary prevents us from obtaining a uniform spreading method straight from the "upheaval point" unless it is depending only on the norm of the velocity. Our strategy is to spread the lower bound created at the "upheaval points" independently for grazing and non-grazing collisions up to the point when the lower bound we obtain depends only on the norm of the velocity. Roughly, our lower bounds will be balls in velocity that can be centred away from the origin and we shall grow them up finitely many times to balls containing the origin and finally be able to generate a uniform spreading method.

Collision kernels satisfying a cutoff property as well as collision kernels with a noncutoff property will be treated following the strategy described above. The only difference is the decomposition of the Boltzmann bilinear operator Q we consider in each case. In the case of a non-cutoff collision kernel, we shall divide it into a cutoff collision kernel and a remainder. The cutoff part will already be dealt with and a careful control of the  $L^{\infty}$ -norm of the remainder will give us the expected lower bound, smaller than a Maxwellian lower bound.

A preliminary to our study is to be able to construct the characteristic trajectories associated to the Boltzmann equation with specular reflections in a  $C^2$  bounded convex domain. These trajectories are merely those of the free transport and so can be seen as the movement of a billiard ball inside the boundary of our domain.

Such a free transport in a convex domain has been studied in [33] (see also [93], [104] or [105] for geometrical properties) and has been used in kinetic theory by Guo, [52], or Hwang, [59], for instance. Yet, the common feature in [33], [52] and [59] is that their assumptions on the boundary always lead to clear rebounds of the characteristic trajectories. That is to say, the absoption phenomenon of [33], the electromagnetic field in [52] and [59] or the smooth strict convexity assumption used in [57], prevent the characteristics to roll on the boundary which is one of the possible behaviour we have to take into account in our general settings. As briefly mentionned in the introduction of [105], the behaviour at some specific boundary points is mathematically quite unexpected, even if that is of no physical relevance. We thus classify all the possible outcomes of a rebound against the boundary and study them carefully to analytically build the characteristics for the free transport equation in our domain  $\Omega$ .

Finally, we need to control the number of rebounds that can happen in a finite time. In [104], Tabachnikov focuses on the footprints on the boundary of the trajectories of billiard balls and shows that the initial conditions leading to infinitely many rebounds on the boudary is a set of measure 0. We extend this to the whole trajectory (see Appendix 2.3.1, Proposition 2.3.4), not only its footprints on the boundary, allowing us to consider only finitely many rebounds in finite time and to have an analytic formula for the characteristics which we shall use throughout the chapter.

Notice that all this study of the free transport equation will be done in the case of a merely  $C^1$  bounded domain.

#### 2.1.3 Organisation of the chapter

Section 2.2 is dedicated to the statement and the description of the main results proved in this chapter. It contains four different parts

Section 2.2.1 defines all the notations which will be used throughout the chapter.

As mentioned above, we shall investigate in detail the characteristics and the free transport equation in a  $C^1$  bounded domain. Section 2.2.2 mathematically formulates the intuitive ideas of trajectories.

The last subsections, 2.2.3 and 2.2.4, are dedicated to a mathematical formulation of the results related to the lower bound in, respectively, the cutoff case and the non-cutoff case, described in Section 2.1.2. It also defines the concept of mild solutions to the Boltzmann equation in each case.

Sections 2.4 to 2.7 focuse on the Maxwellian lower bound in the cutoff case. It is divided into the four main arguments of the proof.

Following our strategy, Section 2.4 creates the localised "upheaval points" whereas Section 2.5 and Section 2.6 spread them along non-grazing and grazing trajectories respectively.

Section 2.7 concludes by describing the immediate appearance of a lower bound depending only on the norm of the velocity (Proposition 2.2.4) as well as proving the immediate Maxwellian lower bound (proof of Theorem 2.2.3).

Finally, we deal with non-cutoff collision kernels in Section 2.8 where we prove the immediate appearance of an exponential lower bound (Theorem 2.2.6). The proof follows exactly the same steps as in the case of cutoff kernels and is thus divided into Section 2.8.1, where we construct a lower bound only depending on the norm of the velocity, and Section 2.8.2, where we derive the exponential lower bound.

As mentioned before, we need to study the free transport equation and the different important properties of the characteristics. Appendix 2.3 formulates these issues, investigates all the different behaviours of rebounds against the boundary (Section 2.3.1), builds the characteristics and derives their properties (Section 2.3.2) and solves the free transport equation (Section 2.3.3).

# 2.2 Main results

We begin with the notations we shall use all along the chapter.

# 2.2.1 Notations

We denote  $\langle \cdot \rangle = \sqrt{1 + |\cdot|^2}$  and  $y^+ = \max\{0, y\}$ , the positive part of y. This study will hold in specific functional spaces regarding the v variable that we describe here and use throughout the sequel. Most of them are based on natural Lebesgue spaces  $L_v^p = L^p(\mathbb{R}^d)$  with a weight:

• for  $p \in [1, \infty]$  and  $q \in \mathbb{R}$ ,  $L^p_{q,v}$  is the Lebesgue space with the following norm

$$||f||_{L^{p}_{q,v}} = ||\langle v \rangle^{q} f||_{L^{p}_{v}},$$

• for  $p \in [1, \infty]$  and  $k \in \mathbb{N}$  we use the Sobolev spaces  $W_v^{k, p}$  by the norm

$$\|f\|_{W^{k,p}_{v}} = \left[\sum_{|s| \leqslant k} \|\partial^{s} f(v)\|_{L^{p}_{v}}^{p}\right]^{1/p},$$

with the usual convention  $H_v^k = W_v^{k,2}$ .

In what follows, we are going to need bounds on some physical observables of solution to the Boltzmann equation (2.1.1).

We consider here a function  $f(t, x, v) \ge 0$  defined on  $[0, T) \times \Omega \times \mathbb{R}^d$  and we recall the definitions of its local hydrodynamical quantities.

• its local energy

$$e_f(t,x) = \int_{\mathbb{R}^d} |v|^2 f(t,x,v) dv,$$

• its local weighted energy

$$e_f'(t,x) = \int_{\mathbb{R}^d} |v|^{\tilde{\gamma}} f(t,x,v) dv,$$

where  $\tilde{\gamma} = (2 + \gamma)^+$ ,

• its local  $L^p$  norm  $(p \in [1, +\infty))$ 

$$l_f^p(t,x) = \|f(t,x,\cdot)\|_{L_v^p},$$

• its local  $W^{2,\infty}$  norm

$$w_f(t,x) = \|f(t,x,\cdot)\|_{W^{2,\infty}_x}.$$

Our results depend on uniform bounds on those quantities and therefore, to shorten calculations we will use the following

$$\begin{split} E_f &= \sup_{(t,x) \in [0,T) \times \Omega} e_f(t,x) \quad , \quad E'_f = \sup_{(t,x) \in [0,T) \times \Omega} e'_f(t,x), \\ L^p_f &= \sup_{(t,x) \in [0,T) \times \Omega} l^p_f(t,x) \quad , \quad W_f = \sup_{(t,x) \in [0,T) \times \Omega} w_f(t,x). \end{split}$$

In our theorems we are giving a priori lower bound results for solutions to (2.1.1) satisfying some properties about their local hydrodynamical quantities. Those properties will differ depending on which case of collision kernel we are considering. We will take them as assumptions in our proofs and they are the following.

• In the case of hard or Maxwellian potentials with cutoff ( $\gamma \ge 0$  and  $\nu < 0$ ):

$$E_f < +\infty. \tag{2.2.1}$$

• In the case of a singularity of the kinetic collision kernel  $(\gamma \in (-d, 0))$  we shall make the additional assumption

$$L_f^{p_\gamma} < +\infty, \tag{2.2.2}$$

where  $p_{\gamma} > d/(d+\gamma)$ .

• In the case of a singularity of the angular collision kernel ( $\nu \in [0, 2)$ ) we shall make the additional assumption

$$W_f < +\infty, \ E'_f < +\infty. \tag{2.2.3}$$

As noticed in [78], in some cases several assumptions might be redundant.

Furthermore, in the case of the torus with periodic conditions or the case of bounded domain with specular boundary reflections, solutions to (2.1.1) also satisfy the following conservation laws (see [28], [30] or [112] for instance) for the total mass and the total energy:

$$\exists \mathbf{M}, \mathbf{E} \ge 0, \, \forall t \in \mathbb{R}^+, \quad \begin{cases} \int_{\Omega} \int_{\mathbb{R}^d} f(t, x, v) \, dx dv = M, \\ \int_{\Omega} \int_{\mathbb{R}^d} |v|^2 \, f(t, x, v) \, dx dv = E. \end{cases}$$
(2.2.4)

# 2.2.2 Results about the free transport equation

Our investigations start with the study of the characteristics of the free transport equation. We only focus on the case where  $\Omega$  is not the torus (the characteristics in the torus being merely straight lines) but we will use the same notations in both cases. This is achieved by the following theorem.

**Theorem 2.2.1** Let  $\Omega$  be an open, bounded and  $C^1$  domain in  $\mathbb{R}^d$ . Let  $u_0: \overline{\Omega} \times \mathbb{R}^d \longrightarrow \mathbb{R}$  be  $C^1$  in  $x \in \Omega$  and in  $L^2_{x,v}$ .

The free transport equation with specular reflections reads

$$\forall t \ge 0 \quad , \quad \forall (x,v) \in \Omega \times \mathbb{R}^d, \quad \partial_t u(t,x,v) + D_x(v)(u)(t,x,v) = 0, \qquad (2.2.5)$$

$$\forall (x,v) \in \overline{\Omega} \times \mathbb{R}^d, \quad u(0,x,v) = u_0(x,v), \tag{2.2.6}$$

$$\forall (x,v) \in \partial\Omega \times \mathbb{R}^d, \quad u(t,x,v) = u(t,x,\mathcal{R}_x(v)), \tag{2.2.7}$$

where  $\mathcal{R}_x$  stands for the specular reflection at a point x and  $D_x(v)$  is the directional derivative at x in the direction of v.

Then this equation has a unique solution  $u : \mathbb{R}^+ \times \overline{\Omega} \times \mathbb{R}^d \longrightarrow \mathbb{R}$  which is  $C^1$  in time, admits a directional derivative in space in the direction of v and is in  $L^2_{x,v}$ .

Moreover, for all (t, x, v) in  $\mathbb{R}^+ \times \overline{\Omega} \times \mathbb{R}^d$ , there exists  $x_{fin}(t, x, v)$ ,  $v_{fin}(t, x, v)$  and  $t_{fin}(t, x, v)$  (see Definition 2.3.6) such that

$$u(t, x, v) = u_0 \left( x_{fin} - (t - t_{fin}) v_{fin}, v_{fin} \right).$$

#### 2.2.3 Maxwellian lower bound for cutoff collision kernels

The final theorem we prove in the case of cutoff collision kernel is the immediate appearance of a uniform Maxwellian lower bound. We use, in that case, the Grad's splitting for the bilinear operator Q such that the Boltzmann equation reads

$$Q(g,h) = \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \Phi(|v - v_{*}|) b(\cos \theta) [h'g'_{*} - hg_{*}] dv_{*} d\sigma$$
  
=  $Q^{+}(g,h) - Q^{-}(g,h),$ 

where we used the following definitions

$$Q^{+}(g,h) = \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \Phi(|v - v_{*}|) b(\cos \theta) h'g'_{*} dv_{*} d\sigma,$$
  

$$Q^{-}(g,h) = n_{b} (\Phi * g(v)) h = L[g](v)h,$$
(2.2.8)

where

$$n_b = \int_{\mathbb{S}^{d-1}} b\left(\cos\theta\right) d\sigma = \left|\mathbb{S}^{d-2}\right| \int_0^\pi b\left(\cos\theta\right) \sin^{d-2}\theta \, d\theta.$$
(2.2.9)

In Section 2.3 we prove that we are able to construct the characteristics  $(X_t(x, v), V_t(x, v))$ , for all (t, x, v) in  $\mathbb{R}^+ \times \overline{\Omega} \times \mathbb{R}^d$ , of the transport equation (Proposition (2.3.8)). Thanks to this Proposition we can define a mild solution of the Boltzmann equation in the cutoff case. This weaker form of solutions is actually the key point for our result and also gives a more general statement.

**Definition 2.2.2** Let  $f_0$  be a measurable function, non-negative almost everywhere on  $\overline{\Omega} \times \mathbb{R}^d$ .

A measurable function f = f(t, x, v) on  $[0, T) \times \overline{\Omega} \times \mathbb{R}^d$  is a mild solution of the Boltzmann equation associated to the initial datum  $f_0(x, v)$  if

- 1. f is non-negative on  $\overline{\Omega} \times \mathbb{R}^d$ ,
- 2. for every (x, v) in  $\Omega \times \mathbb{R}^d$ :

$$t \longmapsto L[f(t, X_t(x, v), \cdot)](V_t(x, v)), t \longmapsto Q^+[f(t, X_t(x, v), \cdot), f(t, X_t(x, v), \cdot)](V_t(x, v))$$

are in 
$$L^{1}_{loc}([0,T))$$
,

3. and for each  $t \in [0,T)$ , for all  $x \in \Omega$  and  $v \in \mathbb{R}^d$ ,

$$f(t, X_t(x, v), V_t(x, v)) = f_0(x, v) \exp\left[-\int_0^t L[f(s, X_s(x, v), \cdot)](V_s(x, v)) \, ds\right] \\ + \int_0^t \exp\left(-\int_s^t L[f(s', X_{s'}(x, v), \cdot)](V_{s'}(x, v)) \, ds'\right)$$

$$Q^+[f(s, X_s(x, v), \cdot), f(s, X_s(x, v), \cdot)](V_s(x, v)) \, ds.$$
(2.2.10)

Now we state our result.

**Theorem 2.2.3** Let  $\Omega$  be  $\mathbb{T}^d$  or a  $C^2$  open convex bounded domain in  $\mathbb{R}^d$  and let  $f_0$  be a non-negative continuous function on  $\overline{\Omega} \times \mathbb{R}^d$ . Let  $B = \Phi b$  be a collision kernel satisfying (2.1.3), with  $\Phi$  satisfying (2.1.4) or (2.1.5) and b satisfying (2.1.6) with  $\nu < 0$ . Let f(t, x, v) be a mild solution of the Boltzmann equation in  $\overline{\Omega} \times \mathbb{R}^d$  on some time interval [0,T),  $T \in (0, +\infty]$ , which satisfies

- f is continuous on  $[0,T) \times \overline{\Omega} \times \mathbb{R}^d$ ,  $f(0,x,v) = f_0(x,v), M > 0$  and  $E < \infty$  in (2.2.4);
- if  $\Phi$  satisfies (2.1.4) with  $\gamma \ge 0$  or if  $\Phi$  satisfies (2.1.5), then f satisfies (2.2.1);
- if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ , then f satisfies (2.2.1) and (2.2.2).

Then for all  $\tau \in (0,T)$  there exists  $\rho > 0$  and  $\theta > 0$ , depending on  $\tau$ ,  $E_f$  (and  $L_f^{p_{\gamma}}$ if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ ), such that for all  $t \in [\tau,T)$  the solution f is bounded from below, almost everywhere, by a global Maxwellian distribution with density  $\rho$  and temperature  $\theta$ , i.e.

$$\forall t \in [\tau, T), \, \forall (x, v) \in \bar{\Omega} \times \mathbb{R}^d, \quad f(t, x, v) \geqslant \frac{\rho}{(2\pi\theta)^{d/2}} e^{-\frac{|v|^2}{2\theta}}.$$

If we add the assumptions of uniform boundedness of  $f_0$  and of the mass and entropy of the solution f we can use the arguments originated in [95] to construct explicitly the initial "upheaval point", without any compactness argument (see Section 2.4.2). Moreover, if we further suppose that  $\Omega$  is  $C^3$  and strictly convex, the use of tools developed by Guo [57] yields a constructive method to control grazing collisions (see Remark 2.6.3). We thus have the following corollary.

**Corollary 1** Suppose that conditions of Theorem 2.2.3 are satisfied (the continuity assumption on  $f_0$  can be dropped) and further assume that  $\Omega$  is  $C^3$  and strictly convex, i.e. there exists  $\xi : \mathbb{R}^d \longrightarrow \mathbb{R}$  to be  $C^3$  such that

$$\Omega = \{ x \in \mathbb{R}^d, \quad \xi(x) < 0 \}$$

and such that  $\nabla \xi \neq 0$  on  $\partial \Omega$  and there exists  $C_{\xi} > 0$  such that

$$\partial_{ij}\xi(x)v_iv_j \ge C_{\xi} \|v\|^2$$

for all x in  $\overline{\Omega}$  and all v in  $\mathbb{R}^d$ . Further assume that  $f_0$  is uniformly bounded from below

$$\forall (x,v) \in \Omega \times \mathbb{R}^d, \quad f_0(x,v) \ge \varphi(v) > 0,$$

and that f has a bounded local mass and entropy

$$R_f = \inf_{\substack{(t,x)\in[0,T)\times\Omega}} \int_{\mathbb{R}^d} f(t,x,v) \, dv > 0$$
  
$$H_f = \sup_{\substack{(t,x)\in[0,T)\times\Omega}} \left| \int_{\mathbb{R}^d} f(t,x,v) \log f(t,x,v) \, dv \right| < +\infty$$

Then conclusion of Theorem 2.2.3 holds true with the constants  $\rho$  and  $\theta$  being explicitly constructed in terms of  $\tau$ ,  $E_f$ ,  $H_f$ ,  $L_f^{p_{\gamma}}$  and upper and lower bounds on  $|\nabla \xi|$  and  $|\nabla^2 \xi|$  on  $\partial \Omega$ .

As stated in Subsection 2.1.2, the main result to reach Theorem 2.2.3 is the construction of an immediate lower bound only depending on the norm of the velocity:

**Proposition 2.2.4** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3.

For all  $0 < \tau < T$  there exists  $r_V$ ,  $a_0(\tau) > 0$  such that

$$\forall t \in [\tau/2, \tau], \, \forall (x, v) \in \bar{\Omega} \times \mathbb{R}^d, \quad f(t, x, v) \ge a_0(\tau) \mathbf{1}_{B(0, r_V)}(v),$$

 $r_V$  and  $a_0(\tau)$  only depending on  $\tau$ ,  $E_f$  (and  $L_f^{p_{\gamma}}$  if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ ).

## 2.2.4 Exponential lower bound for non-cutoff collision kernels

In the case of non-cutoff collision kernels  $(0 \le \nu < 2 \text{ in } (2.1.6))$ , Grad's splitting does not make sense anymore and so we have to find a new way to define mild solutions to the Boltzmann equation (2.1.1). The splitting we are going to use is a standard one and it reads

$$Q(g,h) = \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \Phi(|v - v_{*}|) b(\cos \theta) [h'g'_{*} - hg_{*}] dv_{*} d\sigma$$
  
=  $Q_{b}^{1}(g,h) - Q_{b}^{2}(g,h),$ 

where we used the following definitions

$$Q_{b}^{1}(g,h) = \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \Phi(|v-v_{*}|) b(\cos\theta) g'_{*}(h'-h) dv_{*} d\sigma,$$
  

$$Q_{b}^{2}(g,h) = -\left(\int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \Phi(|v-v_{*}|) b(\cos\theta) [g'_{*} - g_{*}] dv_{*} d\sigma\right) h \quad (2.2.11)$$
  

$$= S[g](v)h.$$

We would like to use the properties we derived in the study of collision kernels with cutoff. Therefore we will consider additional splitting of Q.

For  $\varepsilon$  in  $(0, \pi/4)$  we define a cutoff angular collision kernel

$$b_{\varepsilon}^{CO}\left(\cos\theta\right) = b\left(\cos\theta\right)\mathbf{1}_{\left|\theta\right| \ge \varepsilon}$$

and a non-cutoff one

$$b_{\varepsilon}^{NCO}\left(\cos\theta\right) = b\left(\cos\theta\right) \mathbf{1}_{|\theta| \leqslant \varepsilon}$$

Considering the two collision kernels  $B_{\varepsilon}^{CO} = \Phi b_{\varepsilon}^{CO}$  and  $B_{\varepsilon}^{NCO} = \Phi b_{\varepsilon}^{NCO}$ , we can combine Grad's splitting (2.2.8) applied to  $B_{\varepsilon}^{CO}$  with the non-cutoff splitting (2.2.11) applied to  $B_{\varepsilon}^{NCO}$ . This yields the splitting we shall use to deal with non-cutoff collision kernels,

$$Q = Q_{\varepsilon}^{+} - Q_{\varepsilon}^{-} + Q_{\varepsilon}^{1} - Q_{\varepsilon}^{2}, \qquad (2.2.12)$$

where we use the shortened notations  $Q_{\varepsilon}^{\pm} = Q_{b_{\varepsilon}^{CO}}^{\pm}$  and  $Q_{\varepsilon}^{i} = Q_{b_{\varepsilon}^{NCO}}^{i}$ , for i = 1, 2.

Thanks to the splitting (2.2.12) and the study of characteristics mentionned in Section 2.2.2, we are able to define mild solutions to the Boltzmann equation with non-cutoff collision kernels. This is obtained by considering the Duhamel formula associated to the splitting (2.2.12) along the characteristics (as in the cutoff case).

**Definition 2.2.5** Let  $f_0$  be a measurable function, non-negative almost everywhere on  $\overline{\Omega} \times \mathbb{R}^d$ .

A measurable function f = f(t, x, v) on  $[0, T) \times \overline{\Omega} \times \mathbb{R}^d$  is a mild solution of the Boltzmann equation with non-cutoff angular collision kernel associated to the initial datum  $f_0(x, v)$  if there exists  $0 < \varepsilon_0 < \pi/4$  such that for all  $0 < \varepsilon < \varepsilon_0$ :

- 1. f is non-negative on  $\overline{\Omega} \times \mathbb{R}^d$ ,
- 2. for every (x, v) in  $\Omega \times \mathbb{R}^d$ :

$$t \longmapsto L_{\varepsilon}[f(t, X_t(x, v), \cdot)](V_t(x, v)), t \longmapsto Q_{\varepsilon}^+[f(t, X_t(x, v), \cdot), f(t, X_t(x, v), \cdot)](V_t(x, v))$$
$$t \longmapsto S_{\varepsilon}[f(t, X_t(x, v), \cdot)](V_t(x, v)), t \longmapsto Q_{\varepsilon}^1[f(t, X_t(x, v), \cdot), f(t, X_t(x, v), \cdot)](V_t(x, v))$$
$$are \ in \ L^1_{loc}([0, T)),$$

3. and for each  $t \in [0,T)$ , for all  $x \in \Omega$  and  $v \in \mathbb{R}^d$ ,

$$f(t, X_t(x, v), V_t(x, v)) = f_0(x, v) \exp\left[-\int_0^t (L_{\varepsilon} + S_{\varepsilon}) \left[f(s, X_s(x, v), \cdot)\right] (V_s(x, v)) \, ds\right] \\ + \int_0^t \exp\left(-\int_s^t (L_{\varepsilon} + S_{\varepsilon}) \left[f(s', X_{s'}(x, v), \cdot)\right] (V_{s'}(x, v)) \, ds'\right) \\ \left(Q_{\varepsilon}^+ + Q_{\varepsilon}^1\right) \left[f(s, X_s(x, v), \cdot), f(s, X_s(x, v), \cdot)\right] (V_s(x, v)) \, ds.$$

$$(2.2.13)$$

Now we state our result.

**Theorem 2.2.6** Let  $\Omega$  be  $\mathbb{T}^d$  or a  $C^2$  open convex bounded domain in  $\mathbb{R}^d$  and  $f_0$  be a continuous function on  $\overline{\Omega} \times \mathbb{R}^d$ . Let  $B = \Phi b$  be a collision kernel satisfying (2.1.3), with  $\Phi$  satisfying (2.1.4) or (2.1.5) and b satisfying (2.1.6) with  $\nu$  in [0,2). Let f(t,x,v) be a mild solution of the Boltzmann equation in  $\overline{\Omega} \times \mathbb{R}^d$  on some time interval  $[0,T), T \in (0,+\infty]$ , which satisfies

- f is continuous on  $[0,T) \times \overline{\Omega} \times \mathbb{R}^d$  and  $f(0,x,v) = f_0(x,v), M > 0$  and  $E < \infty$  in (2.2.4);
- if  $\Phi$  satisfies (2.1.4) with  $\gamma \ge 0$  or if  $\Phi$  satisfies (2.1.5), then f satisfies (2.2.1) and (2.2.3);
- if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ , then f satisfies (2.2.1), (2.2.2) and (2.2.3).

Then for all  $\tau \in (0,T)$  and for any exponent K such that

$$K > 2 \frac{\log\left(2 + \frac{2\nu}{2-\nu}\right)}{\log 2},$$

there exists  $C_1, C_2 > 0$ , depending on  $\tau$ , K,  $E_f, E'_f, W_f$  (and  $L_f^{p\gamma}$  if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ ), such that

$$\forall t \in [\tau, T), \, \forall (x, v) \in \bar{\Omega} \times \mathbb{R}^d, \quad f(t, x, v) \ge C_1 e^{-C_2 |v|^K}.$$

Moreover, in the case  $\nu = 0$ , one can take K = 2 (Maxwellian lower bound).

We emphasize here that, in the same spirit as in the cutoff case, the main part of the proof will rely on the establishment of an equivalent to Proposition 2.2.4 for non-cutoff collision kernels.

**Corollary 2** As for Corollary 1, if if  $f_0$  is bounded uniformly from below as well as the local mass of f, the local entropy of f is uniformly bounded from above and  $\Omega$  is  $C^3$  and strictly convex then the conclusion of Theorem 2.2.6 holds true with constants being explicitly constructed in terms of  $\tau$ , K,  $E_f$ ,  $E'_f$ ,  $W_f$ ,  $H_f$ ,  $L_f^{p\gamma}$  and upper and lower bounds on  $|\nabla \xi|$  and  $|\nabla^2 \xi|$  on  $\partial \Omega$ .

**Remark 2.2.7** Throughout the chapter, we are going to deal with the case where  $\Omega$  is a  $C^2$  convex bounded domain since it is the case where the most important difficulties arise. However, if  $\Omega = \mathbb{T}^d$ , we can follow the same proofs by letting the first time of collision with the boundary to be  $+\infty$  (see Section 2.3) and by making the definition that the distance to the boundary (which does not exist) is  $+\infty$  (which rules out the case of grazing trajectories).

# 2.3 The free transport equation: proof of Theorem 2.2.1

In this section, we study the transport equation with a given initial data and boundary condition in a bounded domain  $\Omega$ . We will only consider the case of purely specular reflections on the boundary  $\partial \Omega$ . Those kind of interaction cannot occur for all velocities at the boundary. Indeed, for a particle to bounce back at the boundary, we need its velocity to come from inside the domain  $\Omega$ . To express this fact mathematically, we define

$$\Lambda^{+} = \left\{ (x, v) \in \partial \Omega \times \mathbb{R}^{d} : v \cdot n(x) \ge 0 \right\},\$$

where we denote by n(x) the exterior normal to  $\partial \Omega$  at x.

Consider  $u_0: \overline{\Omega} \times \mathbb{R}^d \longrightarrow \mathbb{R}$  which is  $C^1$  in  $x \in \Omega$  and  $L^2(\overline{\Omega} \times \mathbb{R}^d) = L^2_{x,v}$ . We are interested in the problem stated in Theorem 2.2.1, (2.2.5) – (2.2.7).

If  $D_x(v)(u)$  denotes the directional derivate of u in x in the direction of v we have, in the case of functions that are  $C^1$  in x,

$$D_x(v)(u) = v \cdot \nabla_x u.$$

Therefore, instead of imposing that the solution to the transport equation should be  $C^1$  in x, we reformulate the problem with directional derivatives.

Physically, the free transport equation means that a particle evolves freely in  $\Omega$  at a velocity v until it reaches the boundary. Then it bounces back and moves straight until it reaches the boundary for the second time and so on so forth up to time t. The method of characteristics is therefore the best way to link u(t, x, v) to  $u_0$  by just following the path used by the particle, backwards from t to 0 (see Figure 2.1). This method has been used



Figure 2.1: Backward trajectory with standard rebounds

in [52] on the half-line and in [33], [59], for instance, in the case of convex media. However, in both articles they only deal with finite, or countably many, numbers of rebounds in finite time. Indeed, the electrical field in [52] and [59] makes the particles always reach the boundary with  $v \cdot n(x) > 0$  and [33] has a specular boundary problem with an absorption coefficient  $\alpha \in [0, 1)$ :  $u(t, x, v) = \alpha u(t, x, \mathcal{R}_x(v))$ . Therefore, in the case the particle arrives tangentially to the boundary, i.e.  $v \cdot n(x) = 0$ , we have  $\mathcal{R}_x(v) = v$  and so u(t, x, v) = 0. This vanishing property allowed the authors to not care about the special cases where the particle starts to roll on the boundary.

Another way of looking at the characteristics method is to study the footprints of the trajectories on the boundary. This problem, as well as the possibility of having infinitely many rebounds in a finite time, has been tackled by Tabachnikov in [104]. Tabachnikov only focused on boundary points since the description of the trajectories by only considering their collisions with the boundary holds a symplectic property and a volume-preserving transformation. Such properties allowed him to show that the set of points on the boundary that lead to infinitely many rebounds in finite time is of measure 0 ([104], Lemma 1.7, 1). Unfortunately, in our case we would like to follow the characteristics and the study of trajectories only via their footprints on the boundary is no longer a volume-preserving transformation.

In our case we need to follow the path of a particle along the characteristics of the equation to know the value of our function at each step. If the particle starts to roll on the boundary (see Figure 2.2) we require to know for how long it will do so. The major issue is the fact that  $v \cdot n(x) = 0$  does not tell us much about the geometry of  $\partial \Omega$  at x and the possibility, or lack of, for the particle to keep moving tangentially to the boundary. Moreover, some cases lead to non physical behaviour since the sole specular collision condition implies that some pairs  $(x, v) \in \partial \Omega \times \mathbb{R}^d$  can only be starting points, they cannot be generated by any trajectories (see Figure 2.3). This case is mentioned quickly in the first chapter of [105] but not dealt with.



Figure 2.2: Backward trajectory rolling on the boundary



Figure 2.3: Backward trajectory that reaches an end

Therefore, in order to prove the well-posedness of the transport equation (2.2.5) - (2.2.7), we follow the ideas developed in [52] and [59], which consist of studying the backward trajectories that can lead to a point (t, x, v), combined with the idea of countably many collisions in finite time used in [33]. However, we have to deal with the issues described above and to do so we introduce a new classification of possible interactions with the boundary (see Definition 2.3.1). We also extend the result of [104], in terms of pair (x, v) leading to infinitely many rebounds in finite time, to the whole domain  $\Omega$  (Proposition 2.3.4). To do so we link up the study on the boundary made in [104] with the Lebesgue measure on  $\Omega$  by artificially creating volume on  $\partial\Omega$  thanks to time and a foliation of the domain by parallel trajectories.

The section is divided as follows. First of all we shall describe and classify the collisions with the boundary in order to describe very accurately the backward trajectories of a point (x, v) in  $\partial \Omega \times \mathbb{R}^d$ . We will name trajectory or characteristic any solution (X(t, x, v), V(t, x, v)) satisfying the initial condition (X(0, x, v), V(0, x, v)) = (x, v), the boundary condition (2.2.7) and satisfying, in  $\Omega$ ,

$$\begin{cases} \frac{dX}{dt} = V\\ \frac{dV}{dt} = 0. \end{cases}$$

This will give us an explicit form for the characteristics and allow us to link u(t, x, v) with  $u_0(x^*, v^*)$ , for some  $x^*$  and  $v^*$ . Finally, we will show that the function we constructed is, indeed, a solution to the transport equation with initial data  $u_0$  and specular boundary condition and that such a solution is unique.

### 2.3.1 Study of rebounds on the boundary

As mentionned in the introduction of this section, when a particle reaches a point at the boundary with a velocity v it can bounce back (Figure 2.1), keep moving straight (Figure 2.2) or stop moving because the specular reflection does not allow it to do anything else (Figure 2.3), which is physically unexpected. The next definition gives a partition of the points at the boundary which takes into account those properties.

**Definition 2.3.1** We define here a partition of  $\partial \Omega \times \mathbb{R}^d$  that focuses on the outcome of a collision in each of the sets.

• The set coming from a rebound without rolling

$$\Omega_{rebounds} = \left\{ (x, v) \in \partial \Omega \times \mathbb{R}^d : v \cdot n(x) < 0 \right\}.$$

• The set coming from rolling on the boundary

$$\Omega_{rolling} = \left\{ (x, v) \in \partial\Omega \times \mathbb{R}^d : v \cdot n(x) = 0 \text{ and } \exists \delta > 0, \forall t \in [0, \delta], \ x - vt \in \overline{\Omega} \right\}.$$

• The set of only starting points

$$\Omega_{stop} = \left\{ (x, v) \in \partial\Omega \times \mathbb{R}^d : v \cdot n(x) = 0 \text{ and } \forall \delta > 0, \exists t \in [0, \delta], \ x - vt \notin \overline{\Omega} \right\}.$$

• The set coming from straight line

$$\Omega_{line} = \left\{ (x, v) \in \partial \Omega \times \mathbb{R}^d : v \cdot n(x) > 0 \right\}.$$

One has to notice that any point of  $\Omega_{line}$  indeed comes from a straight line arriving at x with direction v since  $\Omega$  is open and is  $C^1$  (so there is no cusp).

In order to understand the behaviour expected at  $\Omega_{stop}$  we have the following proposition. The proof of it gives insight into the nature of specular reflections.

**Proposition 2.3.2** If we have (x, v) in  $\Omega_{stop}$  then there is no trajectory with specular boundary reflections that leads to (x, v).

**Proof of Proposition 2.3.2** Let us assume the contrary, that is to say (x, v) is in  $\Omega_{stop}$  comes from a trajectory with specular boundary reflection.

We have that (x, v) belongs to  $\partial \Omega \times \mathbb{R}^d$  and so if (x, v) comes from a straight line it can only be (by definition of trajectories) a line containing x with direction v which means

that (x, v) comes from  $\{(x - vt, v), t \in [0, T]\}$ , for some T > 0. But the trajectory is necessarily in  $\overline{\Omega}$  and this is in contradiction with the definition of  $\Omega_{stop}$ .

Therefore, (x, v) must come from a rebound after a straight line trajectory. But again we obtain a contradiction because the velocity before the rebound is  $\mathcal{R}_x(v) = v$  and the backward trajectory is the one studied above.

Now we have our partition of points on the boundary of  $\Omega$ , we are able to generate the backward trajectory associated to a starting point (x, v) in  $\overline{\Omega} \times \mathbb{R}^d$ . The first step towards its resolution is to find the first point of real collision (if it exists) that generates (x, v) (see Figure 2.1). The next proposition-definition proves mathematically what the figure shows.

**Proposition 2.3.3** Let  $\Omega$  be an open, bounded and  $C^1$  domain in  $\mathbb{R}^d$ . Let (x, v) be in  $\overline{\Omega} \times \mathbb{R}^d$ , then we can define

$$t_{min}(x,v) = \max\left\{t \ge 0 : x - vs \in \overline{\Omega}, \, \forall \, 0 \le s \le t\right\}.$$

Moreover we have the following properties:

- 1. if there exists t in  $(0, t_{min}(x, v))$  such that x vt hits  $\partial \Omega$  then (x vt, v) belongs to  $\Omega_{rolling}$ .
- 2.  $t_{min}(x, v) = 0$  if and only if (x, v) belongs to  $\Omega_{stop} \cup \Omega_{rebounds}$ .
- 3.  $(x vt_{min}(x, v), v)$  belongs to  $\Omega_{stop} \cup \Omega_{rebounds}$ .

Property (1) emphasises the fact that if, on the straight line between x and  $x - vt_{min}(x, v)$ , the particle hits the boundary it will not be reflected and so just rolls on. Then property (2) tells us than  $t_{min(x,v)}$  is always strictly positive except if (x, v) does not come from any trajectory of a particle or if it is the outcome of a rebound without rolling. Finally, property (3) finishes the study since at  $x - vt_{min}(x, v)$  the particles either come from a reflection (case  $\Omega_{rebounds}$ ), and we can keep tracking backwards, or started its trajectory at  $x - vt_{min}(x, v)$  (case  $\Omega_{stop}$ ).

**Proop of Proposition** 2.3.3 First of all we have that  $\Omega$  is bounded and so there exists R such that  $\overline{\Omega} \subset B(0, R)$ , the ball of radius R in  $\mathbb{R}^d$ .

Then we notice that 0 belongs to

$$A(x,v) = \left\{ t \ge 0 : x - vs \in \overline{\Omega}, \, \forall \, 0 \le s \le t \right\}.$$

Therefore A(x, v) is not empty. Moreover, this set is bounded above by 2R/||v|| since for all t in A(x, v)

$$R > ||x - vt|| \ge t ||v|| - ||x||.$$

Therefore we can talk about the supremum  $t_{min}(x, v)$  of A(x, v). Let  $(t_n)_{n \in \mathbb{N}}$  be increasing sequence in A(x, v) that tends to  $t_{min}(x, v)$ . As  $\overline{\Omega}$  is closed we have that  $x - vt_{min}(x, v)$ belongs to  $\overline{\Omega}$ . Then, if  $0 \leq s < t_{min}(x, v)$  there exists n such that  $0 \leq s \leq t_n$  and so, by the property of  $t_n, x - vs$  is in  $\overline{\Omega}$ . This conclude the fact that  $t_{min}(x, v)$  belongs to A(x, v)and so is a maximum.

We now turn to the proof of properties.

Let (x, v) be in  $\Omega$  and  $0 < t < t_{min}(x, v)$  such that x - vt belongs to  $\partial\Omega$ . Then for all  $0 < t_1 < t < t_2 < t_{min}(x, v), x - vt_1$  and  $x - vt_2$  are in  $\overline{\Omega}$  and so, by the definition of an exterior normal to a surface we have

$$[(x-vt)-(x-vt_1)]\cdot n(x-vt) \ge 0 \text{ and } [(x-vt)-(x-vt_2)]\cdot n(x-vt) \ge 0,$$

which gives  $v \cdot n(x - vt) = 0$ .

Moreover, since  $t_2$  belongs to A(x, v), for all s in  $[0, t_2 - t]$ , (x - vt) - vs is in  $\overline{\Omega}$ , which means that (x - vt, v) belongs to  $\Omega_{rolling}$ .

Property (2) is direct since if  $t_{min}(x, v) = 0$  then for all t > 0, there exists  $0 < s \leq t$ such that x - vs does not belong to  $\overline{\Omega}$  and then  $v \cdot n(x) \leq 0$ . So (x, v) belongs to  $\Omega_{rebounds}$ , if  $v \cdot n(x) > 0$ , or to  $\Omega_{stop}$ .

Finally, property (3) is straightforward since  $x - vt_{min}(x, v)$  is in  $\partial\Omega$  (because  $\Omega$  is open) and since for all  $0 \leq t \leq t_{min}(x, v)$ , x - vt is in  $\overline{\Omega}$ . Thus  $[(x - vt_{min}) - (x - vt)] \cdot n(x - vt_{min}(x, v)) \geq 0$ , which yields  $v \cdot n(x - vt_{min}(x, v)) \leq 0$ . Then, by the definition of A(x, v) and the fact that  $t_{min}(x, v)$  is its maximum, we have that either  $(x - vt_{min}(x, v), v)$  belongs to  $\Omega_{rebounds}$  or belongs to  $\Omega_{stop}$ .

Up to now we focused solely on the case of the first possible collision with the boundary. In order to conclude the study of rebounds for any given characteristics we have to, in some sense, count the number of rebounds without rolling that can happen in finite time. This is the purpose of the next proposition.

**Proposition 2.3.4** Let  $\Omega$  be a  $C^1$  open, bounded domain in  $\mathbb{R}^d$  and let (x, v) be in  $\overline{\Omega} \times \mathbb{R}^d$ . Then for all  $t \ge 0$  the trajectory finishing at (x, v) after a time t has at most a countable number of rebounds without rolling.

Moreover, this number is finite almost surely with respect to the Lebesgue measure on  $\bar\Omega\times\mathbb{R}^d$
**Proof of Proposition 2.3.4** The fact that there is countably many rebounds without rolling comes directly from the fact that  $t_{min}(x, v) > 0$  except if (x, v) is a starting/stopping point (and then did not move from 0 to t) or if (x, v) is the outcome of a rebound (and so comes from  $(x, \mathcal{R}_x(v))$  which belongs to  $\Omega_{line}$ , implying that  $t_{min}(x, \mathcal{R}_x(v)) > 0$ ).

Now we shall prove that the set of points in  $\overline{\Omega} \times \mathbb{R}^d$  which lead to an infinite number of rebounds in a finite time is of measure 0. To do so, we first need some definitions. The measure  $\mu$  in  $\overline{\Omega} \times \mathbb{R}^d$  is the one induced by the Lebesgue measure and we denote by  $\lambda$  the measure on  $\partial\Omega \times \mathbb{R}^d$  (see Section 1.7 of [104]). We will also denote

$$\Omega = \left\{ (x,v) \in \Omega \times \left( \mathbb{R}^d - \{0\} \right) \text{ coming from an infinite number of rebounds} \right\},$$
  
$$\Omega_\partial = \left\{ (x,v) \in \partial\Omega \times \left( \mathbb{R}^d - \{0\} \right) \text{ coming from an infinite number of rebounds} \right\}.$$

We know ([104] Lemma 1.7.1) that  $\lambda(\Omega_{\partial}) = 0$  and we are going to establish a link between the measure of  $\Omega$  and the one of  $\Omega_{\partial}$ . Those two sets do not live in the same topology nor same dimension and so we build a function that artificially recreates them via time.

Because  $\Omega$  is bounded we can find time  $T_M > 0$  such that for all x in  $\overline{\Omega}$  and v in  $\mathbb{R}^d - \{0\}$ ,  $(x - T_M v / ||v||)$  does not belong to  $\overline{\Omega}$ . Furthermore, in the same way as for  $t_{min}(x, v)$ , we can define, for (x, v) in  $\overline{\Omega} \times \mathbb{R}^d$ ,

$$T(x,v) = \begin{cases} \min\{t > 0 : x + vt \in \partial\Omega\} \text{ if } (x,v) \in \Omega \cup \Omega_{rebounds} \\ 0 \text{ otherwise} \end{cases}$$

We define the following function which is clearly  $C^1$ .

$$F: [0, T_M] \times \mathbb{R}^d \times \left(\mathbb{R}^d - \{0\}\right) \longrightarrow \mathbb{R}^d \times \left(\mathbb{R}^d - \{0\}\right)$$
$$(t, x, v) \longmapsto (x + \frac{v}{\|v\|}t, v).$$

We also define the set

$$B = \left\{ (t, x, v) : x \in \partial\Omega, v \in (\mathbb{R}^d - \{0\}), t \in [0, T(x, v)) \right\}$$

and claim that F is injective on the set B. Indeed, if (t, x, v) and  $(t^*, x^*, v^*)$  are in B such that  $F(t, x, v) = F(t^*, x^*, v^*)$  then  $v = v^*$  and  $x + tv/||v|| = x^* + t^*v/||v||$ . Let assume that  $t^* > t$ , therefore we have that

$$x = x^* + (t^* - t)\frac{v}{\|v\|} \in \partial\Omega$$

and thus  $t^* - t \ge T(x^*, v)$ . However,  $t^* \le T(x^*, v)$  so we reach a contradiction and  $t^* \le t$ . By symmetry we have  $t = t^*$  and then  $x = x^*$ . We also notice that  $[0, T_M] \times \Omega_{stop}$  and  $[0, T_M] \times \Omega_{rolling}$  do not intersect B.

Finally we have that  $\Omega = F(B \cap ([0, T_M] \times \Omega_\partial))$ . Indeed, if (t, x, v) belongs to  $B \cap ([0, T_M] \times \Omega_\partial)$  then F(t, x, v) = (x + tv/||v||, v) and x + tv/||v|| is in  $\Omega$  and its first rebound backward in time is (x, v) which lead to infinitely many rebounds in finite time. Therefore

$$x + t \frac{v}{\|v\|} \in \Omega.$$

The converse is direct, by considering the first collision with the boundary of the backward trajectory starting at (x, v) in  $\Omega$ .

All those properties allow us to compute  $\mu(\Omega)$  by a change of variable in  $B \cap \Omega_{\partial}$ .

$$\begin{split} \mu(\Omega) &= \mu(F\left(B \cap \left([0, T_M] \times \Omega_{\partial}\right)\right)) \\ &= \int_{\bar{\Omega} \times \mathbb{R}^d} \mathbf{1}_{F(B \cap \left([0, T_M] \times \Omega_{\partial}\right))}(x, v) dx dv dt \\ &= \int_{B \cap \left([0, T_M] \times \Omega_{\partial}\right)} \left|\operatorname{Jac}(F^{-1})\right| d\lambda(x, v) dt \\ &\leqslant T_M \sup_{[0, T_M] \times \bar{\Omega}} \left(\left|\operatorname{Jac}(F^{-1})\right|\right) \lambda(\Omega_{\partial}) = 0. \end{split}$$

### 2.3.2 Description of characteristics

In the previous section we derived all the relevant properties of when, where and how a trajectory can bounce against the boundary of  $\Omega$ . As was shown, the characteristic starting from a point (t, x, v) in  $\mathbb{R}^+ \times \overline{\Omega} \times \mathbb{R}^d$  is the backward trajectory satisfying specular boundary reflections that leads to (x, v) in time t. Basically, it consists in a straight line as long as it stays inside  $\Omega$  or it rolls on the boundary. Then it reaches a boundary point where it does not move any more  $(\Omega_{stop})$  or bounces back  $(\Omega_{rebounds})$ .

Thanks to Proposition 2.3.4 we can generate the countable (and almost surely finite) sequence of collisions with the boundary associated to the future point (x, v). We shall construct it by induction. We consider (x, v) in  $\overline{\Omega} \times \mathbb{R}^d$ .

• Step 1: initialisation: we define

$$\begin{cases} x_0(x,v) &= x, \\ v_0(x,v) &= v, \\ t_0(x,v) &= 0. \end{cases}$$

• Step 2: induction: if  $(x_k(x, v), v_k(x, v)) \in \Omega_{stop}$  then we define

$$\begin{cases} x_{k+1}(x,v) = x_k(x,v), \\ v_{k+1}(x,v) = v_k(x,v), \\ t_{k+1}(x,v) = +\infty, \end{cases}$$

if  $(x_k(x,v), v_k(x,v)) \notin \Omega_{stop}$  then we define

$$\begin{cases} x_{k+1}(x,v) &= x_k(x,v) - v_k(x,v)t_{min}(x_k(x,v),v_k(x,v)), \\ v_{k+1}(x,v) &= \mathcal{R}_{x_{k+1}(x,v)}(v_k(x,v)), \\ t_{k+1}(x,v) &= t_k(x,v) + t_{min}(x_k(x,v),v_k(x,v)). \end{cases}$$

Remark 2.3.5 Let us make a few comments on the accuracy of the sequence we just built.

- 1. Looking at Proposition 2.3.3, we know that at each step (apart from 0) we necessary have that  $(x_k(x, v), v_k(x, v))$  belongs to either  $\Omega_{stop}$  or  $\Omega_{rebounds}$  and so the characteristic stops for ever (case 1 in induction) or bounces without rolling and start another straight line (case 2). Thus the sequence of footprints defined above captures the trajectories as long as there are rebounds and then becomes constant once the trajectory reach a stopping point.
- 2. If  $t_{min}(x_k(x,v), v_k(x,v)) = 0$  for some k > 0 then, by properties 2. and 3. of Proposition 2.3.3, we must have  $(x_k(x,v), v_k(x,v)) \in \Omega_{stop}$  (since  $v_k(x,v)$  is the specular reflection at  $x_k(x,v)$  of  $v_{k-1}(x,v)$  and  $(x_k(x,v), v_{k-1}(x,v))$  is in  $\Omega_{rebounds} \cup \Omega_{stop}$ ). Thus,  $(t_k(x,v))_{k\in\mathbb{N}}$  is strictly increasing as long as it does not reach the value  $+\infty$ , where it remains constant.

Finally, it remains to connect the time variable to those quantities. In fact, the time will determine how many rebounds can lead to (x, v) in a time t. The reader must remember that the backward trajectory can lead to a point in  $\Omega_{stop}$  before time t.

Since the characteristics method helps us to find the value of the solution of the transport equation at a given point using its trajectory, the next definition links a triplet (t, x, v) to the first rebound of the trajectory that leads to (x, v) in a time t.

**Definition 2.3.6** Let  $\Omega$  be an open, bounded and  $C^1$  domain in  $\mathbb{R}^d$ . Let (t, x, v) be in  $\mathbb{R}^+ \times \overline{\Omega} \times \mathbb{R}^d$ . Then we can define

$$n(t, x, v) = \begin{cases} \max\{k \in \mathbb{N} : t_k(x, v) \leq t\}, & \text{if it exists,} \\ +\infty, & \text{if } (t_k(x, v))_k & \text{is bounded by t.} \end{cases}$$

The last rebound is then define by

• if  $n(t, x, v) < +\infty$  and  $t_{n(t, x, v)+1} = +\infty$ , then

$$\begin{cases} x_{fin}(t, x, v) = x_{n(t, x, v)}(x, v), \\ v_{fin}(t, x, v) = v_{n(t, x, v)}(x, v), \\ t_{fin}(t, x, v) = t, \end{cases}$$

• if  $n(t, x, v) < +\infty$  and  $t_{n(t, x, v)+1} < +\infty$ , then

$$\begin{cases} x_{fin}(t, x, v) = x_{n(t,x,v)}(x, v), \\ v_{fin}(t, x, v) = v_{n(t,x,v)}(x, v), \\ t_{fin}(t, x, v) = t_{n(t,x,v)}(x, v), \end{cases}$$

• if  $n(t, x, v) = +\infty$ , then

$$\begin{cases} x_{fin}(t, x, v) = \lim_{k \to +\infty} x_k(x, v), \\ v_{fin}(t, x, v) = \lim_{k \to +\infty} v_k(x, v), \\ t_{fin}(t, x, v) = \lim_{k \to +\infty} t_k(x, v). \end{cases}$$

**Remark 2.3.7** Let us make a few comments on the definition above and the existence of limits.

1. After the last rebound, occuring at  $t_{n(t,x,v)}$ , the backward trajectory can only be a straight line during the time period  $t - t_{n(t,x,v)}$  (see Figure 2.1). That is why we defined  $t_{fin}(t,x,v) = t_{n(t,x,v)}$  if we reached a point on  $\Omega_{rebounds}$  and  $t_{fin}(t,x,v) = t$  if the last rebound reaches  $\Omega_{stop}$  (the trajectory can only start from there).

- 2. In the last case of the definition, we remind the reader that  $(t_k(x,v))_{k\in\mathbb{N}}$  is strictly increasing and so converges if bounded by t. But then, because  $(||v_k(x,v)||)_{k\in\mathbb{N}}$  is constant and  $x_k(x,v) = x_{k-1}(x,v) - t_{min}(x_k(x,v),v_k(x,v))v_k(x,v)$ , we have that  $(x_k(x,v))_{k\in\mathbb{N}}$  is a Cauchy sequence.
- 3. The last case in Definition 2.3.6 almost surely never happens, as proved in Proposition 2.3.4.

To conclude this study of the characteristics we just have to make one more comment. We studied the characteristics that go backward in time because it simplifies the construction of a solution to the free transport equation. However, it is easy to prove (just requires the inductive construction of  $v_k$  and  $x_k$ ) that the forward trajectory of (x, v)during a period t is the backward trajectory over a period t of (x, -v). This gives the final proposition.

**Proposition 2.3.8** Let  $\Omega$  be an open, bounded and  $C^1$  domain in  $\mathbb{R}^d$ . Then for all (x, v)in  $\overline{\Omega} \times \mathbb{R}^d$  we have existence and uniqueness of the characteristic  $(X_t(x, v), V_t(x, v))$  given by, for all  $t \ge 0$ ,

$$\begin{aligned} X_t(x,v) &= x_{fin}(t,x,-v) + (t - t_{fin}(t,x,-v))v_{fin}(t,x,-v), \\ V_t(x,v) &= -v_{fin}(t,x,-v). \end{aligned}$$

Moreover, we have that  $V_t(x, v) = O_{t,x,v}(v)$  with  $O_{t,x,v}$  an orthogonal transformation, and that for almost every (x, v) in  $\overline{\Omega} \times \mathbb{R}^d$  we have the following

$$\forall t \ge 0, \quad (x,v) = (X_t(X_t(x,-v), -V_t(x,-v)), V_t(X_t(x,-v), -V_t(x,-v))). \tag{2.3.1}$$

**Proof of Proposition 2.3.8** By construction we have that

$$O_{t,x,v} = \mathcal{R}_{x_{fin}(t,x,v)} \circ \cdots \circ \mathcal{R}_{x_1(t,x,v)}$$

It only remains to show the last equation (2.3.1), but it follows directly from the fact that the backward trajectory of (x, v) is the forward trajectory of (x, -v).

We can reach a point on  $\Omega_{stop}$  after a time  $t_1$  and so the forward trajectory of that point during a time  $t > t_1$  does not come back to the original point (since we stayed in  $\Omega_{stop}$  for a period  $t - t_1$ ). However, the set of points that reach  $\Omega_{stop}$  belongs to the set of points that bounce infinitely many times in a finite time and this set is of measure zero (see Proposition 2.3.4).

### **2.3.3** Existence and uniqueness of solution to (2.2.5) - (2.2.7)

### 2.3.3.1 Proof of uniqueness

The uniqueness of a solution with  $u_0$  in  $C_x^1 \cap L_{x,v}^2$  comes directly from the fact that we have a preserved quantity through time, thanks to the specular reflection property. Indeed, let us assume that u is a solution to our free transport equation satisfying specular boundary condition and the initial value problem  $u_0$ . Then, a mere integration by part gives us

$$\forall t \ge 0, \quad \|u(t, \cdot, \cdot)\|_{L^2_{x,v}}^2 = \|u_0\|_{L^2_{x,v}}^2,$$

which directly implies the uniqueness of a solution, since the transport equation (2.2.5) is linear.

### 2.3.3.2 Construction of the solution

It remains to construct a function u that will be constant on the characteristic trajectories and check that we indeed obtain a function that is differentiable in t and x which satisfies the transport equation. The first point of Remark 2.3.7 gives us the answer as we expect the following behaviour

$$u(t, x, v) = u(t - t_1(x, v), x_1(x, v), v_1(x, v)) = \dots = u(t - t_k(x, v), x_k(x, v), v_k(x, v)),$$

up to the point where there are no more rebound in the time interval [0, t]. From there we continue in a straight line.

Thus, we define:  $\forall (t, x, v) \in \mathbb{R}^+ \times \overline{\Omega} \times \mathbb{R}^d$ ,

$$u(t, x, v) = u_0 \left( x_{fin}(t, x, v) - (t - t_{fin}(t, x, v)) v_{fin}(t, x, v), v_{fin}(t, x, v) \right)$$

### 2.3.3.3 Boundary and initial conditions

First of all, u satisfies the initial condition (2.2.6) as n(0, x, v) = 0 (since  $t_{min}(x, v) \ge 0$ ).

u also satifies the specular boundary condition (2.2.7). Indeed, if (x, v) is in  $\Lambda^+$ , then either  $v \cdot n(x) = 0$  and the result is obvious since  $\mathcal{R}_x(v) = v$ , or  $v \cdot n(x) > 0$  and thus  $(x, \mathcal{R}_x(v))$  belongs to  $\Omega_{rebounds}$  so  $t_{min}(x, \mathcal{R}_x(v)) = 0$  (Proposition 2.3.3). An easy induction shows

$$x_k(x,v) = x_{k+1}(x, \mathcal{R}_x(v)), \ v_k(x,v) = v_{k+1}(x, \mathcal{R}_x(v)), \ t_k(x,v) = t_{k+1}(x, \mathcal{R}_x(v)),$$

for all k in  $\mathbb{N}$ .

The last equality gives us that  $n(t, x, v) = n(t, x, \mathcal{R}_x(v)) - 1$  and therefore, combined with the two other equalities,

$$\begin{aligned} x_{fin}(t,x,v) &= x_{fin}(t,x,\mathcal{R}_x(v)), \ v_{fin}(t,x,v) = v_{fin}(t,x,\mathcal{R}_x(v)), \\ t_{fin}(t,x,v) &= t_{fin}(t,x,\mathcal{R}_x(v)), \end{aligned}$$

which leads to the specular reflection boundary condition.

### 2.3.3.4 Time differentiability

Here we prove that u is differentiable in time on  $\mathbb{R}^+$ . Let us fix (x, v) in  $\Omega \times \mathbb{R}^d$ . By construction, we know that n(t, x, v) is piecewise constant. Since  $(t_k(x, v))_{k \in \mathbb{N}}$  is strictly increasing up to the step where it takes the value  $+\infty$ , for  $t_k(x, v) < t < t_{k+1}(x, v)$  we have that for all  $s \in \mathbb{R}$  such that  $t_k(x, v) < t + s < t_{k+1}(x, v)$ ,

$$\begin{aligned} x_{fin}(t, x, v) &= x_{fin}(t + s, x, v), \ v_{fin}(t, x, v) = v_{fin}(t + s, x, v), \\ t_{fin}(t, x, v) &= t_{fin}(t + s, x, v). \end{aligned}$$

Therefore, we have that

$$\frac{u(t+s, x, v) - u(t, x, v)}{s} = \frac{u_0(x_{fin} - (t+s - t_{fin})v_{fin}, v_{fin}) - u_0(x_{fin} - (t - t_{fin})v_{fin}, v_{fin})}{s} \\ \xrightarrow{s}{s \to 0} - v_{fin} \cdot (\nabla_x u_0) (x_{fin} - (t - t_{fin})v_{fin}, v_{fin}),$$

because  $u_0$  is  $C^1$  in x. So u is differentiable at t if t in strictly between two times  $t_k(x, v)$ . We thus find that u is differentiable at t and that its derivative is continuous (since  $x_{fin}$ ,  $v_{fin}$  and  $t_{fin}$  are continuous when x and v are fixed).

In the case  $t = t_k(x, v)$  we can use what we just proved to show that we have the existence of right (except for t = 0) and left limits of  $\partial_t u(t, x, v)$  as t tends to  $t_k(x, v)$ . We use the specular reflection boundary condition of  $u_0$  together with the fact that it is  $C^1$  in x and that  $t_k(x, v) = t_{k+1}(x, \mathcal{R}_x(v))$  to obtain the equality of the two limits.

#### 2.3.3.5 Space differentiability and solvability of the transport equation

Here we prove that u is differentiable in x in  $\Omega$ , which follows directly from the time differentiability. Let us fix t in  $\mathbb{R}^+$  and v in  $\mathbb{R}^d$ , we shall study the differentiability of  $u(t, \cdot, v)$  in the direction of v.

 $\Omega$  is open and so

$$\forall x \in \Omega, \ \exists \delta > 0, \ \forall s \in [-\delta, \delta], \quad x + sv \in \Omega.$$

Thanks to the inductive construction, one find easily that

$$u(t, x + sv, v) = u(t - s, x, v).$$

Therefore, since u is time differentiable, we have that  $u(t, \cdot, v)$  admits a directional derivative in the direction of v and that

$$D_x(v)(u)(t, x, v) = -\partial_t u(t, x, v).$$

## 2.4 The cutoff case: localized "upheaval points"

In this section and the next three we are going to prove a Maxwellian lower bound for a solution to the Boltzmann equation (2.1.1) in the case where the collision kernel satisfies a cutoff property.

The strategy to tackle this result follows the main idea used in [95] and [78] which relies on finding an "upheaval point" (a first minoration uniform in time and space but localised in velocity) and spreading this bound, thanks to the spreading property of the  $Q^+$  operator, in order to include larger and larger velocities.

We gather here two lemmas, proven in [78], that we will frequently use in this section. We remind the reader that we are using Grad's splitting (2.2.8). Let us first give an  $L^{\infty}$  bound on the loss term (Corollary 2.2 in [78]).

**Lemma 2.4.1** Let g be a measurable function on  $\mathbb{R}^d$ . Then

$$\forall v \in \mathbb{R}^d, \quad |L[g](v)| \leqslant C_q^L \langle v \rangle^{\gamma^+},$$

where  $C_q^L$  is defined by:

1. If  $\Phi$  satisfies (2.1.4) with  $\gamma \ge 0$  or if  $\Phi$  satisfies (2.1.5), then

$$C_q^L = \operatorname{cst} n_b C_\Phi e_g.$$

2. If  $\Phi$  satisfies (2.1.4) with  $\gamma \in (-d, 0)$ , then

$$C_g^L = \operatorname{cst} n_b C_\Phi \left[ e_g + l_g^p \right], \quad p > d/(d+\gamma).$$

The spreading property of  $Q^+$  is given by the following lemma (Lemma 2.4 in [78]), where we define

$$l_b = \inf_{\pi/4 \leqslant \theta \leqslant 3\pi/4} b\left(\cos\theta\right). \tag{2.4.1}$$

**Lemma 2.4.2** Let  $B = \Phi b$  be a collision kernel satisfying (2.1.3), with  $\Phi$  satisfying (2.1.4) or (2.1.5) and b satisfying (2.1.6) with  $\nu \leq 0$ . Then for any  $\bar{v} \in \mathbb{R}^d$ ,  $0 < r \leq R$ ,  $\xi \in (0, 1)$ , we have

$$Q^{+}(\mathbf{1}_{B(\bar{v},R)},\mathbf{1}_{B(\bar{v},r)}) \geqslant \operatorname{cst} l_{b}c_{\Phi}r^{d-3}R^{3+\gamma}\xi^{\frac{d}{2}-1}\mathbf{1}_{B(\bar{v},\sqrt{r^{2}+R^{2}}(1-\xi))}$$

As a consequence in the particular quadratic case  $\delta = r = R$ , we obtain

$$Q^{+}(\mathbf{1}_{B(\bar{v},\delta)},\mathbf{1}_{B(\bar{v},\delta)}) \geqslant \operatorname{cst} l_{b}c_{\Phi}\delta^{d+\gamma}\xi^{\frac{a}{2}-1}\mathbf{1}_{B(\bar{v},\delta\sqrt{2}(1-\xi))}$$

for any  $\bar{v} \in \mathbb{R}^d$  and  $\xi \in (0, 1)$ .

The case of the torus, studied in [78], indicates that without rebounding the expected minoration is created after time t = 0 as quickly as one wants. Therefore we expect the same kind of bound to arise on each characteristic trajectory before its first rebound. However, in the case of a bounded domain, rebounds against the boundary can occur very close to the time t = 0 and a rebound preserves only the norm of the velocity. Therefore, we will fail finding a uniformly (in space) small time where a uniform bound arises. Nevertheless, the convexity and the smoothness of the domain implies that grazing collisions against the boundary do not change the velocity very much.

Thus our study will be split in three parts, which are the next three sections. The first step will be to partition the position and velocity spaces so that we have an immediate appearance of an "upheaval point" in each of those partitions. The second one is to obtain a uniform lower bound which will depend only on the norm of the velocity. Then the final part will use the standard spreading method used in [95] and [78] which will allow us to deal with large velocities and derive the exponential lower bound uniformly.

### 2.4.1 Partition of the phase space and first localised lower bounds

In this section we use the continuity of f together with the conservation laws (2.2.4) to obtain a point in the phase space where f is strictly positive. Then, thanks to the continuity of f, its Duhamel representation (2.2.10) and the spreading property of the  $Q^+$  operator (Lemma 2.4.2) we extend this positivity to high velocities at that particular point (Lemma 2.4.3). Finally, the free transport part of the solution f will imply the immediate appearance of the localised lower bounds (Proposition 2.4.4).

Moreover we define constants that we will use in the next two subsections in order to have a uniform lower bound.

We define some shorthand notations. For x in  $\overline{\Omega}$ , v in  $\mathbb{R}^d$  and  $s, t \ge 0$  we denote the point at time s of the forward characteristic passing through (x, v) at time t by

$$X_{s,t}(x,v) = X_s(X_t(x,-v), -V_t(x,-v))$$
$$V_{s,t}(x,v) = V_s(X_t(x,-v), -V_t(x,-v)),$$

which has been derived from (2.3.1).

We start by the strict positivity of our function at one point for all velocities:

**Lemma 2.4.3** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3.

Then there exists  $(x_1, v_1)$  in  $\Omega \times \mathbb{R}^d$  and  $\Delta > 0$  such that for all  $n \in \mathbb{N}$  and all t in  $[0, \Delta]$ , there exists  $r_n > 0$ , depending only on n, and  $\alpha_n(t) > 0$  such that

$$\forall x \in B\left(x_1, \frac{\Delta}{2^n}\right), \ \forall v \in \mathbb{R}^d, \quad f(t, x, v) \ge \alpha_n(t) \mathbf{1}_{B(v_1, r_n)}(v),$$

with  $\alpha_0 > 0$  independent of t and the induction formula

$$\alpha_{n+1}(t) = C_Q \frac{r_n^{d+\gamma}}{4^{d/2-1}} \int_0^{\min(t,\Delta/(2^{n+1}(2r_n+\|v_1\|)))} e^{-sC_L\langle 2r_n+\|v_1\|\rangle^{\gamma^+}} \alpha_n^2(s) \, ds$$

where  $C_Q = cstl_bc_{\Phi}$  is defined in Lemma 2.4.2 and  $C_L = cstn_bC_{\Phi}E_f$  (or  $C_L = cstn_bC_{\Phi}(E_f + L_f^p)$ ) is defined in Lemma 2.4.1, and

$$r_0 = \Delta, \quad r_{n+1} = \frac{3\sqrt{2}}{4}r_n.$$

**Proof of Lemma 2.4.3** The proof is an induction on n.

Step 1: Initialization. We recall the conservation laws satisfied by a solution to the Boltzmann equation, (2.2.4),

$$\forall t \in \mathbb{R}^+, \quad \int_{\Omega} \int_{\mathbb{R}^d} f(t, x, v) \, dx dv = M, \quad \int_{\Omega} \int_{\mathbb{R}^d} |v|^2 \, f(t, x, v) \, dx dv = E,$$

with M > 0 and  $E < \infty$ .

Since  $\Omega$  is bounded, and so is included in, say,  $B(0, R_X)$ , we also have that

$$\forall t \in \mathbb{R}^+, \quad \int_{\Omega} \int_{\mathbb{R}^d} \left( |x|^2 + |v|^2 \right) f(t, x, v) \, dx dv \leqslant \alpha = MR_X^2 + E < +\infty.$$

Therefore if we take t = 0 and  $R_{min} = \sqrt{2\alpha/M}$ , we have the following

$$\int_{B(0,R_{min})} \int_{B(0,R_{min})} f_0(x,v) \, dx \, dv \ge \frac{M}{2} > 0.$$

Therefore we have that there exists  $x_1$  in  $\Omega$  and  $v_1$  in  $B(0, R_{min})$  such that

$$f_0(x_1, v_1) \ge \frac{M}{4\text{Vol}(B(0, R_{min}))^2} > 0.$$

The first step of the induction is then due to the continuity of f at  $(0, x_1, v_1)$ . Indeed, there exists  $\delta_T, \delta_X, \delta_V > 0$  such that

$$\forall t \in [0, \delta_T], \ \forall x \in B(x_1, \delta_X), \ \forall v \in B(v_1, \delta_V), \quad f(t, x, v) \ge \frac{M}{8 \operatorname{Vol}(B(0, R_{min}))^2}.$$

and we define  $\Delta = \min(\delta_T, \delta_X, \delta_V)$ .

Step 2: Proof of the induction. We assume the conjecture is valid for n. Let x be in  $B(x_1, \Delta/2^{n+1})$ , v in  $B(0, ||v_1|| + 2r_n)$  and t in  $[0, \Delta]$ .

We use the fact that f is a mild solution to write f(t, x, v) under its Duhamel form (2.2.10). The control we have on the L operator, Lemma 2.4.1, allows us to bound from above the second integral term (the first term is positive). Moreover, this bound on L is independent on t, x and v since it only depends on an upper bound on the energy  $e_{f(t,x,\cdot)}$ (and its local  $L^p$  norm  $l_{f(t,x,\cdot)}^p$ ) which is uniformly bounded by  $E_f$  (and by  $L_f^p$ ). This yields, for  $\tau_n(t) = \min(t, \Delta/(2^{n+1}(2r_n + ||v_1||)))$ 

$$f(t,x,v) \ge \int_{0}^{\tau_{n}(t)} e^{-sC_{L}\langle ||v_{1}||+2r_{n}\rangle^{\gamma^{+}}} Q^{+} \left[f(s,X_{s,t}(x,v),\cdot),f(s,X_{s,t}(x,v),\cdot)\right] \left(V_{s,t}(x,v)\right) ds,$$
(2.4.2)

where  $C_L = \operatorname{cst} n_b C_{\Phi} E_f$  (or  $C_L = \operatorname{cst} n_b C_{\Phi} (E_f + L_f^p)$ ), see Lemma 2.4.1, and we used  $\|V_{s,t}(x,v)\| = \|v\| \leq 2r_n + \|v_1\|$ .

Besides, we have that  $B(x_1, \Delta) \subset \Omega$  and also

$$\forall s \in \left[0, \frac{\Delta}{2^{n+1}(2r_n + \|v_1\|)}\right], \, \forall v_* \in B(0, \|v_1\| + 2r_n), \quad \|x_1 - (x + sv_*)\| \leq \frac{\Delta}{2^n}$$

which, by definition of the characteristics (see Section 2.3.2), yields

$$\forall s \in [0, \tau_n(t)], \ \forall v_* \in B(0, \|v_1\| + 2r_n), \quad \begin{cases} X_{s,t}(x, v_*) &= x + sv_* \in B\left(x_1, \frac{\Delta}{2^n}\right) \\ V_{s,t}(x, v_*) &= v_*. \end{cases}$$

Therefore, by calling  $v_*$  the integration parameter in the operator  $Q^+$  we can apply the induction property to  $f(s, X_{s,t}(x, v), v_*)$  which implies, in (2.4.2),

$$f(t,x,v) \ge \int_0^{\tau_n(t)} e^{-sC_L \langle ||v_1|| + 2r_n \rangle^{\gamma^+}} \alpha_n^2(s) Q^+ \left[ \mathbf{1}_{B(v_1,r_n)}, \mathbf{1}_{B(v_1,r_n)} \right] \, ds(v).$$

Applying the spreading property of  $Q^+$ , Lemma 2.4.2, with  $\xi = 1/4$  gives us the expected result for the step n + 1 since  $B(v_1, r_{n+1}) \subset B(0, ||v_1|| + 2r_n)$ .

We now have all the tools to prove the next proposition which is the immediate appearance of localised "upheaval points".

**Proposition 2.4.4** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3.

Then there exists  $\Delta > 0$  such that for all  $0 < \tau_0 \leq \Delta$ , there exists  $\delta_T(\tau_0)$ ,  $\delta_X(\tau_0)$ ,  $\delta_V(\tau_0)$ ,  $R_{min}(\tau_0)$ ,  $a_0(\tau_0) > 0$  such that for all N in N there exists  $N_X$  in N<sup>\*</sup> and  $x_1, \ldots, x_{N_X}$  in  $\Omega$  and  $v_1, \ldots, v_{N_X}$  in  $B(0, R_{min}(\tau_0))$  and

- $\bar{\Omega} \subset \bigcup_{1 \leq i \leq N_X} B\left(x_i, \delta_X(\tau_0)/2^N\right);$
- $\forall t \in [\tau_0, \delta_T(\tau_0)], \forall x \in B(x_i, \delta_X(\tau_0)), \forall v \in \mathbb{R}^d,$

 $f(t, x, v) \ge a_0(\tau_0) \mathbf{1}_{B(v_i, \delta_V(\tau_0))}(v),$ 

with  $B(v_i, \delta_V(\tau_0)) \subset B(0, R_{min}(\tau_0)).$ 

**Proof of Proposition** 2.4.4 We are going to use the free transport part of the Duhamel form of f (2.2.10), to create localised lower bounds out of Lemma 2.4.3.

We take  $0 < \tau_0 \leq \Delta$ , where  $\Delta$  is defined in Lemma 2.4.3.

 $\Omega$  is bounded so let us denote its diameter by  $d_{\Omega}$ . Let *n* be big enough such that  $r_n \ge 2d_{\Omega}/\tau_0 + ||v_1||$  and define  $R_{min}(\tau_0) = 2d_{\Omega}/\tau_0$ .

Thanks to Lemma 2.4.3 applied to this particular n we have that

$$\forall t \in \left[\frac{\tau_0}{2}, \Delta\right], \, \forall x \in B(x_1, \Delta/2^n), \quad f(t, x, v) \ge \alpha_n\left(\frac{\tau_0}{2}\right) \mathbf{1}_{B(v_1, r_n)}(v), \tag{2.4.3}$$

where we used the fact that  $\alpha_n(t)$  is an increasing function.

Define

$$a_0(\tau_0) = \frac{1}{2} \alpha_n \left(\frac{\tau_0}{2}\right) e^{-\frac{\tau_0}{2} C_L \left(\frac{2d_\Omega}{\tau_0}\right)^{\gamma^+}}.$$

We remark that f is continuous on the compact  $[\tau_0, \Delta] \times \overline{\Omega} \times B(0, R_{min}(\tau_0))$  and hence uniformly continuous. Therefore it exists  $\delta'_T(\tau_0), \, \delta'_X(\tau_0), \, \delta'_V(\tau_0) > 0$  such that

$$\forall |t - t'| \leq \delta'_T(\tau_0), \ \forall \ \left\| x - x' \right\| \leq \delta'_X(\tau_0), \ \forall \ \left\| v - v' \right\| \leq \delta'_V(\tau_0),$$

$$\left| f(t, x, v) - f(t', x', v') \right| \leq a_0(\tau_0).$$

$$(2.4.4)$$

We conclude our definition by taking

$$\delta_T(\tau_0) = \min(\Delta, \tau_0 + \delta'_T(\tau_0)),$$
  

$$\delta_X(\tau_0) = \min(\delta'_X(\tau_0), \Delta/2^n),$$
  

$$\delta_V(\tau_0) = \min(\delta'_V(\tau_0), r_n).$$

Finally, we take  $N \in \mathbb{N}$  and notice that  $\overline{\Omega}$  is compact so there exists  $x_1, \ldots, x_{N_X}$  in  $\Omega$  such that  $\overline{\Omega} \subset \bigcup_{1 \leq i \leq N_X} B(x_i, \delta_X(\tau_0)/2^N)$ . Moreover, we construct them such that  $x_1$  is the one defined in Lemma 2.4.3.

We then take  $v_1$  to be the one defined in Lemma 2.4.3 and we define

$$\forall i \in \{2, \dots, N_X\}, \quad v_i = \frac{2}{\tau_0}(x_i - x_1).$$

Because  $\Omega$  is convex we have that

$$X_{\tau_0/2,\tau_0}(x_i, v_i) = x_1,$$

$$V_{\tau_0/2,\tau_0}(x_i, v_i) = v_i,$$
(2.4.5)

Using the fact that f is a mild solution of the Boltzmann equation, we write it under its Duhamel form (2.2.10) and we drop the last term which is positive. As in the proof of Lemma 2.4.3 we can control the L operator appearing in the first term in the right-hand side of (2.2.10) (corresponding to the free transport). Thus, we use the Duhamel form (2.2.10) between  $\tau_0$  and  $\tau_0/2$  and we combine it with (2.4.5). This yields

$$\begin{aligned} f(\tau_0, x_i, v_i) & \geqslant \quad f\left(\frac{\tau_0}{2}, x_1, v_i\right) e^{-\frac{\tau_0}{2}C_L \langle \frac{2}{\tau_0}(x_i - x_1) \rangle^{\gamma^+}} \\ & \geqslant \quad \alpha_n \left(\frac{\tau_0}{2}\right) e^{-\frac{\tau_0}{2}C_L \langle \frac{2d_\Omega}{\tau_0} \rangle^{\gamma^+}} \mathbf{1}_{B(v_1, r_n)}(v_i) \\ & \geqslant \quad 2a_0(\tau_0) \mathbf{1}_{B(v_1, r_n)}(v_i), \end{aligned}$$

where we used (2.4.3) for the second inequality. We see here that  $v_i$  belongs to  $B(0, R_{min}(\tau_0))$ and that  $B(0, R_{min}(\tau_0)) \subset B(v_1, r_n)$  and therefore

$$f(\tau_0, x_i, v_i) \ge 2a_0(\tau_0).$$
 (2.4.6)

Finally, combining (2.4.6) with the uniform continuity of f, (2.4.4) we have that for all t in  $[\tau_0, \delta_T(\tau_0)]$ , x in  $B(x_i, \delta_X(\tau_0))$  and v in  $B(v_i, \delta_V(\tau_0))$ ,

$$f(t, x, v) \ge a_0(\tau_0).$$

**Remark 2.4.5** This last proposition tells us that localised lower bounds appear immediately, that is to say after any time  $\tau_0 > 0$ . The exponential lower bound we expect will appear immediately after those initial localised lower bounds, i.e. for all  $\tau_1 > \tau_0$ . Therefore, to shorten notation and lighten our presentation, we are going to study the case of solution to the Boltzmann equation which satisfies Proposition 2.4.4 at  $\tau_0 = 0$ . Then we will immediatly create the exponential lower bound after 0 and apply this result to  $F(t, x, v) = f(t + \tau_0, x, v)$ .

### 2.4.2 A constructive approach to the initial lower bound, Corollary 1

The initial lower bounds we just derived relies on compactness arguments and their construction is therefore not explicit. However, as mentioned in Section 2.2.3, a few more assumptions on  $f_0$  and f suffice to obtain a completely constructive approach for the "upheaval point". This method is based on a property of the iterated  $Q^+$  operator discovered by Pulvirenty and Wennberg [95] and reformulated by Mouhot ([78] Lemma 2.3) as follows.

**Lemma 2.4.6** Let  $B = \Phi b$  be a collision kernel satisfying (2.1.3), with  $\Phi$  satisfying (2.1.4) or (2.1.5) and b satisfying (2.1.6) with  $\nu \leq 0$ . Let g(v) be a nonnegative function on  $\mathbb{R}^d$ with bounded energy  $e_g$  and entropy  $h_g$  and a mass  $\rho_g$  such that  $0 < \rho_g < +\infty$ . Then there exist  $R_0$ ,  $\delta_0$ ,  $\eta_0 > 0$  and  $\bar{v} \in B(0, R_0)$  such that

$$Q^{+}\left(Q^{+}\left(g\mathbf{1}_{B(0,R_{0})},g\mathbf{1}_{B(0,R_{0})}\right),g\mathbf{1}_{B(0,R_{0})}\right) \geqslant \eta_{0}\mathbf{1}_{B(\bar{v},\delta_{0})},$$

with  $R_0$ ,  $\delta_0$ ,  $\eta_0$  being constructive in terms on  $\rho_g$ ,  $e_g$  and  $h_g$ .

We now suppose that  $0 < \rho_{f_0} < +\infty$ ,  $h_{f_0} < +\infty$  and that

$$\forall (x,v) \in \Omega \times \mathbb{R}^d, \quad f_0(x,v) \ge \varphi(v) > 0$$

and we follow the argument used in [78].

By the Duhamel definition (2.2.10) of f being a mild solution and Lemma 2.4.1 we have

$$f(t, X_t(x, v), V_t(x, v)) \ge f_0(x, v)e^{-tC_L\langle v \rangle^{\gamma^+}}$$
(2.4.7)

and

$$f(t,x,v) \ge \int_0^t e^{-(t-s)C_L \langle v \rangle^{\gamma^+}} Q^+ \left[ f(s, X_{s,t}(x,v), \cdot), f(s, X_{s,t}(x,v), \cdot) \right] \left( V_{s,t}(x,v) \right) \, ds.$$

Define t(x, v) > 0 the time of first contact with  $\partial \Omega$  of the trajectory x + sv (see rigorous definition in Proposition 2.3.3). For all t in [0, t(x, v)] we have

$$X_{0,t}(x,v) = x + tv,$$
  

$$V_{0,t}(x,v) = v.$$

Thus, for all  $0 \leq t \leq t(x, v)$ ,

$$f(t,x,v) \ge \int_0^t e^{-(t-s)C_L \langle v \rangle^{\gamma^+}} Q^+ \left[ f(s,x+sv,\cdot), f(s,x+sv,\cdot) \right] (v) \ ds$$

and we can iterate the latter inequality

$$f(t, x, v) \ge \int_{0}^{t} e^{-(t-s)C_{L}\langle v \rangle^{\gamma^{+}}} Q^{+} \left( f(s, x + s'v, \cdot), f(s, x + s'v, \cdot) \right) (\cdot) ds', f(t, x + sv, \cdot) \right] (v) \, ds.$$

$$(2.4.8)$$

(2.4.7) and (2.4.8) are exactly the same bounds than the ones obtained in [78], Step 1 of proof of Proposition 3.2, and we can therefore conclude the same way with Lemma 2.4.6

$$f(t, x, v) \ge a_0(\tau_0) \mathbf{1}_{B(\bar{v}, \delta_0),}$$

as long as v is in  $B(0, R_0)$  and  $0 \leq t \leq \tau_0$ .

The only difference with [78] is the fact that we need  $\tau_0$  to be in [0, t(x, v)], giving local lower bounds instead of a global one.

# 2.4.3 A lower bound depending only on the norm of the velocity: strategy of the proof of Proposition 2.2.4

As stated in the introduction, the spreading property of the bilinear operator  $Q^+$  cannot be used (at least uniformly in time and space) when we are really close to the boundary due to the lack of control over the rebounds. However, if we have a lower bound depending only on the norm of the velocity then the latter bound will not take into account rebounds as they preserve the norm, allowing us to spread this minoration up to an exponential one.

The next two sections are dedicated to the creation of such a uniform lower bound depending solely on the norm of the velocity. In order to do so we restrain the problem without taking into account large velocities and divide the study to two cases: if the trajectory stays close to the boundary or if it does not. In both cases we will start from the localised "upheaval points" constructed in Section 2.4.1 and spread them to the point where one gets a lower bound depending only on the norm of the velocity.

The next sections tackle each of these points. We first study the case when a characteristic reaches a point far from the boundary and finally we focus on the case of grazing characteristics. We fix  $\delta_T$ ,  $\delta_X$ ,  $\delta_V$ ,  $R_{min}$  and  $a_0$  to be the ones described in Proposition 2.4.4 at time  $\tau_0 = 0$ .

The result we will derive out of those studies is Proposition 2.2.4 and from now on, dependencies on physical observables of f ( $E_f$  and  $L_f^{p_{\gamma}}$ ) will be mentioned but will not be explicitly written everytime.

# 2.5 The cutoff case: characteristics passing by a point far from the boundary

In this section we manage to spread the lower bounds created in Proposition 2.4.4 up to a ball in velocity centred at zero as long as the trajectory we look at reaches a point far enough from the boundary.

First, we pick N in  $\mathbb{N}^*$  and cover  $\overline{\Omega}$  with  $\bigcup_{1 \leq i \leq N_X} B(x_i, \delta_X/2^N)$  as in Proposition 2.4.4. Then for  $l \geq 0$  we define

$$\Omega_l = \{ x \in \Omega : d(x, \partial \Omega) \ge l \}, \qquad (2.5.1)$$

where  $d(x, \partial \Omega)$  is the distance from x to the boundary of  $\Omega$ .

For any R > 0 we define two sequences in  $\mathbb{R}^+$  by induction, for all  $\tau \ge 0$  and  $l \ge 0$ ,

$$\begin{aligned}
 r_0 &= \delta_V \\
 r_{n+1} &= \frac{3\sqrt{2}}{4}r_n
 \end{aligned}
 \tag{2.5.2}$$

and

$$\begin{cases} a_0(l,\tau) = a_0 \\ a_{n+1}(l,\tau) = C_Q \frac{r_n^{d+\gamma}}{4^{d/2-1}} \frac{l}{2^{n+3}R} e^{-\tau C_L \langle R \rangle^{\gamma^+}} a_n^2 \left(\frac{l}{8},\tau\right), \end{cases}$$
(2.5.3)

where  $C_Q$  and  $C_L$  were defined in Lemma 2.4.3.

We express the spreading of the lower bound in the following proposition.

**Proposition 2.5.1** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3 and suppose that f satisfies Proposition 2.4.4 with  $\tau_0 = 0$ .

Consider  $0 < \tau \leq \delta_T$  and N in N. Let  $(x_i)_{i \in \{1,...,N_X\}}$  and  $(v_i)_{i \in \{1,...,N_X\}}$  be given as in Proposition 2.4.4 with  $\tau_0 = 0$ .

Then for all n in  $\{0, ..., N\}$  we have that the following holds: for all  $0 < l \leq \delta_X$ , and R > 0 such that  $l/R < \tau$ , for all t in  $[l/(2^n R), \tau]$ , and for all  $x \in \overline{\Omega}$  and  $v \in B(0, R)$ , if there exists  $t_1 \in [0, t - l/(2^n R)]$  such that  $X_{t_1,t}(x, v)$  belongs to  $\Omega_l \cap B(x_i, \delta_X/2^n)$  then

$$f(t, x, v) \ge a_n(l, \tau) \mathbf{1}_{B(v_i, r_n)}(V_{t_1, t}(x, v)),$$

where  $(r_n)$  and  $(a_n)$  are defined by (2.5.2)-(2.5.3).

**Proof of Proposition** 2.5.1 This Proposition will be proved by induction on n.

Step 1: Initialization. The initialisation is simply Proposition 2.4.4 and the first term in the Duhamel formula (2.2.10) starting at  $\tau$ .

Indeed, we use the definition of f being a mild solution to write f(t, x, v) under its Duhamel form (2.2.10) starting at  $t_1$  where both parts are positive. The control we have on the L operator, Lemma 2.4.1, allows us to bound from above the first term. Moreover, this bound on L is independent on x and v (see proof of Lemma 2.4.3). This gives

$$f(t, x, v) \ge e^{-(t-t_1)C_L \langle R \rangle^{\gamma^+}} f(t_1, X_{t_1, t}(x, v), V_{t_1, t}(x, v)).$$
(2.5.4)

Finally, Proposition 2.4.4 applied to  $f(t_1, X_{t_1,t}(x, v), V_{t_1,t}(x, v))$  gives us the property for n = 0.

Step 2: Proof of the induction. We consider the case where the proposition is true for n.

Given  $l \in (0, \delta_X]$ ,  $t \in [l/(2^{n+1}R), \tau]$ ,  $x \in \overline{\Omega}$  and  $v \in B(0, R)$ .

We suppose now that there exists  $t_1 \in [0, t - l/(2^{n+1}R)]$  such that  $X_{t_1,t}(x, v) \in \Omega_l \cap B(x_i, \delta_X/2^{n+1})$ .

Similar to what we did in the first step of the induction, but concentrating on the second part of the Duhamel formula (2.2.10) we conclude that

$$f(t, x, v) \ge$$

$$e^{-C_L \tau \langle R \rangle^{\gamma^+}} \left( \int_{t_1 + \frac{l}{2^{n+2R}}}^{t_1 + \frac{l}{2^{n+2R}}} Q^+ \left[ f(s, X_{s,t}(x, v), \cdot), f(s, X_{s,t}(x, v), \cdot) \right] ds \right) \left( V_{t_1,t}(x, v) \right).$$
(2.5.5)

The goal is now to apply the induction to the triplet  $(s, X_{s,t}(x, v), v_*)$ , where  $v_*$  is the integration parametre inside the  $Q^+$  operator, with  $||v_*|| \leq R$ .

One easily shows that  $X_{s,t}(x,v) = X_{t_1,t}(x,v) + (s-t_1)V_{t_1,t}(x,v)$ , for s in  $[t_1 + \frac{l}{2^{n+3}R}, t_1 + \frac{l}{2^{n+2}R}]$ , and therefore we have that

$$\|X_{t_1,t}(x,v) - X_{s,t}(x,v)\| \leq \frac{l}{2^{n+2}},$$
(2.5.6)

and so that  $X_{s,t}(x,v)$  belongs to  $\Omega_{l-l/2^{n+2}}$ .

Finally, we have to find a point on the characteristic trajectory of  $(s, X_{s,t}(x, v), v_*)$  that is in  $\Omega_{l'}$  for some l'. This is achieved at the time  $t_1$  (see Figure 2.4).



Figure 2.4: Study of  $(s, X_{s,t}(x, v), v_*)$  far from the boundary

Indeed, we have s in  $[t_1 + l/(2^{n+3}R), t_1 + l/(2^{n+2}R)]$  so, for  $||v_*|| \leq R$ 

$$\forall s' \in [t_1, s], \quad \left\| X_{s,t}(x, v) - \left( X_{s,t}(x, v) - (s - s')v_* \right) \right\| \leqslant \frac{l}{2^{n+2}}.$$
 (2.5.7)

This gives us the characteristics trajectory backward starting from s, since  $X_{s,t}(x, v) - (s - s')v_*$  remains in  $\Omega$ , and therefore

$$\forall s' \in [t_1, s], \begin{cases} X_{s', s} \left( X_{s, t}(x, v), v_* \right) = X_{s, t}(x, v) - (s - s') v_* \\ V_{s', s} \left( X_{s, t}(x, v), v_* \right) = v_*. \end{cases}$$

To conclude we just need to gather the upper bounds we found about the trajectories reaching  $(X_{s,t}(x,v), v_*)$  in a time s in  $[t_1 + l/(2^{n+3}R), t_1 + l/(2^{n+2}R)]$ , equations (2.5.6) and (2.5.7)

$$||X_{t_{1},t}(x,v) - X_{t_{1},s}(X_{s,t}(x,v),v_{*})|| \leq \frac{l}{2^{n+1}}.$$

We have that  $X_{t_1,t}(x,v)$  belongs to  $\Omega_l \cap B(x_i, \delta_X/(2^{n+1}))$  and therefore we have that for all s in  $[t_1+l/(2^{n+3}R), t_1+l/(2^{n+2}R)], X_{t_1,s}(X_{s,t}(x,v),v_*)$  belongs to  $\Omega_{l/2} \cap B(x_i, \delta_X/2^n)$ .

Finally, if s belongs to  $[t_1+l/(2^{n+3}R), t_1+l/(2^{n+2}R)]$  we have that  $(l/8)/(2^nR) \leq s \leq \tau$ and  $t_1$  is in  $[0, s - (l/8)/(2^nR)]$ .

We can therefore apply the induction assumption for l' = l/8 inside the  $Q^+$  operator in (2.5.5), recalling that  $V_{t_1,s}(X_{s,t}(x,v), v_*) = v_*$ .

$$f(t,x,v) \ge a_n \left(\frac{l}{8},\tau\right)^2 e^{-C_L \tau \langle R \rangle^{\gamma^+}} \left(\int_{t_1+\frac{l}{2^{n+2}R}}^{t_1+\frac{l}{2^{n+2}R}} Q^+ \left[\mathbf{1}_{B(v_i,r_n)},\mathbf{1}_{B(v_i,r_n)}\right] ds\right) \left(V_{t_1,t}(x,v)\right).$$

Applying the spreading property of  $Q^+$ , Lemma 2.4.2, with  $\xi = 1/4$  gives us the expected result for the step n + 1.

One easily notices that  $(r_n)_{n \in \mathbb{N}}$  is a strictly increasing sequence. Moreover, for all N in  $\mathbb{N}$  we have that for all  $1 \leq i \leq N_X$ ,  $v_i$  belongs to  $B(0, R_{min})$ . Therefore, by taking N large enough (greater than  $N_1$  say) we have that

$$\forall i \in \{1, \dots, N_X\}, \quad B(0, 2R_{min}) \subset B(v_i, r_N).$$

This remark leads directly to the following corollary which stands for Proposition 2.2.4 in the case when a point on the trajectory is far from the boundary of  $\Omega$ .

**Corollary 3** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3 and suppose that f satisfies Proposition 2.4.4 with  $\tau_0 = 0$ .

Let  $\Delta_T$  be in  $(0, \delta_T]$  and take  $\tau_1$  in  $(0, \Delta_T]$ .

Then for all  $0 < l \leq \delta_X$ , there exists  $a(l, \tau_1, \Delta_T) > 0$  and  $0 < \tilde{t}(l, \tau_1, \Delta_T) < \tau_1$  such that for all t in  $[\tau_1, \Delta_T]$ , and every (x, v) in  $\overline{\Omega} \times \mathbb{R}^d$ : if there exists  $t_1 \in [0, t - \tilde{t}(l, \tau_1, \Delta_T)]$  such that  $X_{t_1,t}(x, v)$  belongs to  $\Omega_l$  then

$$f(t, x, v) \ge a(l, \tau_1, \Delta_T) \mathbf{1}_{B(0, 2R_{min})}(v).$$

**Proop of Corollary 3** This is a direct consequence of Proposition 2.5.1.

Indeed, take  $0 < l \leq \delta_X$ ,  $0 < \tau_1 \leq \Delta_T$  and  $R = R(\Delta_T) > 0$  such that  $R \geq 3R_{min}$  and  $l/R \leq \Delta_T$ . Then take  $N_2 \geq N_1$  large enough such that  $l/(2^{N_2}R) < \tau_1$ . We emphasize here that  $N_2$  depends on to  $\tau_1$  so we write  $N_2(\tau_1)$ .

Now apply Proposition 2.5.1 with  $N = N_2(\tau_1)$  and for t in  $[\tau_1, \Delta_T]$ . We obtain exactly Corollary 3 (since  $B(0, 2R_{min}) \subset B(v_i, r_N)$  for all i and  $R \ge 3R_{min}$ ) with

$$a(l, \tau_1, \Delta_T) = a_{N_2(\tau_1)}(l, \Delta_T)$$
 and  $\tilde{t}(l, \tau_1, \Delta_T) = \frac{l}{2^{N_2} R(\Delta_T)}$ ,

and the fact that  $\bigcup_{1 \leq i \leq N_X} B(x_i, \delta_x/2^N)$  covers  $\overline{\Omega}$ .

### 2.6 The cutoff case: geometry and grazing trajectories

We now turn to the case when the characteristic trajectory never escapes a small distance from the boundary of our convex domain  $\Omega$ .

Intuitively, by considering the case where  $\Omega$  is a circle, one can see that such a behaviour is possible only when the angles of collisions with the boundary remain small (which corresponds in high dimension to the scalar product of the velocity with the outside normal being close to zero), or the angle is important but the norm of the velocity or the time of motion is small. Thus, by using the spreading property of the  $Q^+$  operator we may be able to create larger balls in between two rebounds against the boundary because the latters should not change the velocity too much.

The study of grazing collisions will follow this intuition. First of all Section 2.6.1 proves a geometric lemma dealing with the fact that if the velocities are bounded from below and above, then for short times, the possibility for a trajectory to stay very close to the boundary implies that the velocity do not change a lot over time. Then Section 2.6.2 spreads a lower bound, in the same spirit as the last subsection, up to the point when this lower bound covers a centred ball in velocity. Notice that the geometric property forces us to work with velocities whose norm is bounded from below and so we shall have to take into account the speed of the spreading.

### 2.6.1 Geometric study of grazing trajectories

The key point of the study of grazing collisions is the following geometric lemma. We emphasize here that this is the only part of the chapter where we need the fact that  $\Omega$  is  $C^2$ .

**Proposition 2.6.1** Let  $\Omega$  be an open convex bounded  $C^2$  domain in  $\mathbb{R}^d$  and let  $0 < v_m < v_M$ .

Then, for all  $\varepsilon > 0$  there exists  $t_{\varepsilon}(v_M)$  such that for all  $0 < \tau_2 \leq t_{\varepsilon}(v_M)$  there exists  $l_{\varepsilon}(v_m, \tau_2) > 0$  such that for all x in  $\overline{\Omega}$  and all v in  $\mathbb{R}^d$  with  $v_m \leq ||v|| \leq v_M$ ,

$$\left(\forall s \in [0, \tau_2], X_s(x, v) \notin \Omega_{l_{\varepsilon}(v_m, \tau_2)}\right) \Longrightarrow \left(\forall s \in [0, t_{\varepsilon}(v_M)], \|V_s(x, v) - v\| \leqslant \varepsilon\right).$$

Furthermore,  $l_{\varepsilon}(v_m, \cdot)$  is an increasing function.

The following is dedicated to the proof of Proposition 2.6.1.

We recall that for x in  $\overline{\Omega}$  and v in  $\mathbb{R}^d$  we define, see Section 2.3,  $t_{min}(x, v)$  to be the time of the first proper rebound when we start from x with a velocity -v. This means that  $t_{min}(x, v)$  does not take into account the case where a ball rolls on the boundary. This implies that one cannot hope to get continuity of the function  $t_{min}$  because changing the velocity slightly may lead to a proper rebound instead of a rolling movement.

This being said, we define a time of collision against the boundary which will not take into account the possibility of rolling along the boundary of  $\Omega$ . This will not be too restrictive as we are considering a  $C^2$  convex domain and therefore a trajectory that stays on the boundary will only reach a stopping point which happens only on a set of measure zero in the phase space (see Section 2.3). Therefore we define for x in  $\overline{\Omega}$  and v in  $\mathbb{R}^d$ , the first forward contact with the boundary, t(x, v). It exists by the same arguments as for  $t_{min}$ . Notice that if x is on  $\partial\Omega$  then for all  $v \neq 0$  we have that t(x, v) = 0 if and only if  $n(x) \cdot v \ge 0$ , with n(x) being the outward normal to  $\partial\Omega$  at the point x.

We have the following Lemma dealing with the continuity of the outward normal to  $\partial\Omega$  at the first forward contact point which will be of great interest for proving the crucial Proposition 2.6.1.

**Lemma 2.6.2** Let  $\Omega$  be an open convex bounded  $C^1$  domain in  $\mathbb{R}^d$ . Then  $t: (x, v) \longrightarrow t(x, v)$  is continuous from  $\overline{\Omega} \times (\mathbb{R}^d - \{0\})$  to  $\mathbb{R}^+$ . **Proof of Lemma 2.6.2** Let suppose that t is not continuous at  $(x_0, v_0)$  in  $\Omega \times (\mathbb{R}^d - \{0\})$ . Then

$$\exists \varepsilon > 0, \forall N \ge 1, \exists (x_N, v_N), \quad \begin{cases} \|x_0 - x_N\| \le 1/N \\ \|v_0 - v_N\| \le 1/N \end{cases} \text{ and } |t(x_0, v_0) - t(x_N, v_N)| > \varepsilon. \end{cases}$$

If we still denote by  $d_{\Omega}$  the diameter of  $\Omega$ , we obviously have that for all  $N, 0 \leq t(x_N, v_N) \leq d_{\Omega}/||v_N||$ . Thus,  $(t(x_N, v_N))_{N \in \mathbb{N}}$  is a bounded sequence of  $\mathbb{R}$  and we can extract a converging subsequence  $(t(x_{\phi(N)}, v_{\phi(N)}))$  such that  $T = \lim_{N \to +\infty} t(x_{\phi(N)}, v_{\phi(N)})$ .

By construction (see Section 2.3) we have that for all N in  $\mathbb{N}$ ,  $x_{\phi(N)} + t(x_{\phi(N)}, v_{\phi(N)})v_{\phi(N)}$ belongs to  $\partial\Omega$  which is closed. Moreover, this sequence converges to  $x_0 + Tv_0$  which therefore is on  $\partial\Omega$ .

Finally we have that  $|t(x_0, v_0) - T| \ge \varepsilon$ . Since  $\Omega$  is convex, the segment  $[x_0, x_0 + \max(t(x_0, v_0), T)v_0]$  stays in  $\overline{\Omega}$  and intersect the boundary at least at two distinct points. By convexity of the domain, this implies that the extreme points of the latter segment have to be on the boundary which means that  $x_0$  belongs to  $\partial\Omega$  which is a contradiction.

Therefore, t is continuous in  $\Omega \times (\mathbb{R}^d - \{0\})$ . By the definition of t(x, v) we have its continuity at the boundary. Indeed,  $n(x) \cdot v \ge 0$  means we came from inside the domain to reach that point and we have

$$\left\| t(x',v) - t(x,v) \right\| \leqslant \frac{x-x'}{\|v\|}.$$

We are now ready to prove the geometric Proposition 2.6.1.

### **Proof of Proposition 2.6.1** Consider $\varepsilon > 0$ and $0 < v_m < v_M$ .

Step 1: the case of segments. The first step is to understand that if a whole trajectory stays close to the boundary, then the angle made by the velocity with respect to the normal at the point of collision is close to  $\pi/2$  for dimension d = 2. The same behaviour in higher dimensions is described by the scalar product of the direction of the trajectory and the normal being close to zero. One has to remember that controlling  $||V_s(x,v) - v||$  is the same as controlling the scalar products of the trajectory and the normal on the boundary at each collision point (see definition of  $V_s(x, v)$  in Section 2.3).

Let x be on  $\partial \Omega$  and p in  $\mathbb{N}^*$ . We define

$$\Gamma_p(x) = \left\{ |n(x) \cdot v| : v \in \mathbb{S}^{d-1} \text{ s.t. } n(x) \cdot v < 0 \text{ and } \forall s \in [0, t(x, v)], \ x + sv \notin \Omega_{1/p} \right\},$$

with  $\Omega_{1/p}$  being defined by (2.5.1).

 $\Gamma_p(x)$  gives us the values of scalar products between a normal on the boundary and all the directions that create a characteristic trajectory which stays at a distance less than 1/p from the boundary in between two distinct rebounds (see Figure 2.5). This is exactly what we would like to control uniformly on the boundary.

We remark that  $\Gamma_p(x)$  is not empty because  $\Omega$  and, thus,  $\Omega_{1/p}$  are convex and by the geometric theorem of Hahn-Banach we can separate  $\Omega_{1/p}$  and a disjoint convex ball containing x. It is also straightforward, a mere Cauchy-Schwartz inequality, that  $\Gamma_p(x)$  is bounded from above by 1. Therefore we can define, for all p in  $\mathbb{N}^*$ ,

$$\begin{array}{rccc} h_p: & \partial \Omega & \longrightarrow & \mathbb{R}^+ \\ & x & \longmapsto & \sup \Gamma_p(x) \end{array}$$

We are going to prove that  $(h_p)_{p \in \mathbb{N}^*}$  satisfies the following properties: it is a decreasing sequence of functions,  $h_p$  is continuous in x for each  $p \ge 1$  and for all x in  $\partial \Omega$   $(h_p(x))_{p \in \mathbb{N}^*}$ converges to 0.

The fact that  $(h_p)$  is decreasing is obvious.

In order to prove the continuity of  $h_p$  we take an x on the boundary and v in  $\mathbb{S}^{d-1}$ such that  $|n(x) \cdot v|$  is in  $\Gamma_p(x)$ . We have that for all s in [0, t(x, v)]

$$d(x + sv, \partial\Omega) < 1/p$$

The distance to the boundary is a continuous function and [0, t(x, v)] is compact so there exists s(x, v) in the latter interval such that  $d(x+s(x, v)v, \partial\Omega)$  is maximum. Because  $\Omega$  is convex we have that  $\Omega_{1/p}$  is convex and therefore

$$\forall s \in [0, t(x, v)], \quad B\left(x + sv, \frac{d(x + s(x, v)v, \Omega_{1/p})}{2}\right) \cap \Omega_{1/p} = \emptyset.$$

Then for all x' on the boundary such that  $||x - x'|| \leq d(x + s(x, v)v, \Omega_{1/p})/2$  we have that for all s in [0, t(x', v)], x' + sv is not in  $\Omega_{1/p}$ . Lemma 2.6.2 gives us that if x' is close to x then t(x', v) > 0 and thus v is not tangential at x' either. Moreover  $\Omega$  is  $C^2$  so the outward normal to the boundary is continuous and therefore for x' even closer to x we have that v is such that  $|n(x') \cdot v|$  is also in  $\Gamma_p(x')$ . To conclude, we notice that the scalar product is continuous and therefore for all  $\eta > 0$  we obtain

$$-\eta \leq \left| \left| n(x') \cdot v \right| - \left| n(x) \cdot v \right| \right| \leq \eta,$$

when x' is close enough to x.

The same arguments with the same constants (since our continuous functions act on compact sets and therefore are uniformly continuous) if x' is close to x then taking  $|n(x') \cdot v|$ in  $\Gamma_{1/p}(x')$  we have  $|n(x) \cdot v|$  in  $\Gamma_{1/p}(x)$  and the same inequality as above. This gives us the continuity of  $h_p$  at x. Indeed, we showed that for all x' close to x and for all element u in  $\Gamma_{1/p}(x)$  we can find an element u' in  $\Gamma_{1/p}(x')$  that is close to u.

Finally, it remains to show that for x on the boundary we have that  $h_p(x)$  tends to 0 as p tends to  $+\infty$ .

One can notice that the vector -n(x) is the maximum possible in  $\Gamma_p(x)$  and is exactly the direction of the diametre in  $\Omega$  passing by x. Hence, simple convexity arguments lead to the fact that if all the segments of the form [x, x - t(x, -n(x))n(x)] intersect  $\Omega_{1/p}$  then we have that for all x on the boundary, there exists  $v_p(x)$  in  $\mathbb{S}^{d-1}$  such that  $n(x) \cdot v_p(x) = -h_p(x)$ . Moreover, the segment  $[x, x + t(x, v_p(x))v_p(x)]$  is tangent to  $\Omega_{1/p}$  and we denote by  $x_p$  its first contact point (see Figure 2.5). The convexity of  $\Omega$  and  $\Omega_{1/p}$  shows that, as p increases,  $x_p$  gets closer to x and to the boundary ( $\Omega$  is convex). Therefore  $v_p(x)$  tends to a tangent vector of the boundary at x. This shows that

$$\lim_{p \to +\infty} h_p(x) = 0$$

in the case where all the segments of the form [x, x - t(x, -n(x))n(x)] intersect  $\Omega_{1/p}$ .

We now come to the case where the segments of the form [x, x - t(x, -n(x))n(x)] do not all intersect  $\Omega_{1/p}$ . If for all p, this segment does not intersect  $\Omega_{1/p}$  this implies by convexity of  $\Omega$  that [x, x - t(x, -n(x))n(x)] is included in  $\partial\Omega$ . But then -n(x) is not only a normal vector to the boundary at x but also a tangential one at x. Geometrically this means that x is a corner of  $\partial\Omega$  and n(x) is ill-defined. This is impossible for  $\Omega$  being  $C^2$ . Hence, for all x on the boundary, it exists p(x) such that the segment at x intersect  $\Omega_{p(x)}$ . However,  $\Omega$  is  $C^2$  and we also have Lemma 2.6.2. Those two facts implies that p(x) is continuous on  $\partial\Omega$  which is compact and therefore p(x) reaches a maximum. Let us call this maximum P. For all  $p \ge P$ , all the segments of the form [x, x - t(x, -n(x))n(x)], xin  $\partial\Omega$ , intersect  $\Omega_P$  and we conclude thanks to the previous case.

Thanks to these three properties and the fact that  $\partial\Omega$  is compact, we are able to use Dini's theorem. We therefore find that  $(h_p)_{p\in\mathbb{N}^*}$  converges uniformly to 0. By taking  $p_{\varepsilon}$ large enough we have that for a segment of a characteristic trajectory joining two points on the boundary to be outside  $\Omega_{p_{\varepsilon}}$  we must have  $\Gamma_{p_{\varepsilon}} \leq \varepsilon$  for any x on the boundary (see Figure 2.5).

Step 2: more general trajectories. We take x in  $\partial\Omega$  and v such that  $v_m \leq ||v|| \leq v_M$ and we suppose that for a given t > 0

$$\forall s \in [0, t], \quad X_s(x, v) \notin \Omega_{1/p_{(\varepsilon/2N_{max})}},$$

- 96 -

 $N_{max}$  to be define later.

We are about to find a uniformly small time such that trajectories having at least two collisions against the boundary do not undergo an important evolution of velocity. This will be achieved thanks to the facts that  $||v|| \leq v_M$  and that the maximum of the scalar product is attained at a critical vector and which is the only one that needs to be controlled.

Thanks to Proposition 2.3.4,  $(X_s(x,v))_s$  has countably many rebounds against the boundary (almost surely a finite number in fact). We denote by  $(t_i)_{i\in\mathbb{N}}$  the sequence of times between consecutive collisions and by  $(l_i)_{i\in\mathbb{N}}$  the distance travelled during these respective times. We have that

$$\forall i \in \mathbb{N}, \quad l_i = |v| t_i \quad \text{and} \quad v_m t \leqslant \sum_{i \in \mathbb{N}} l_i \leqslant v_M t.$$

Therefore, for all  $\eta > 0$ , there exists  $N_{\eta}(x, v)$  in  $\mathbb{N}$  such that

$$\sum_{i>N_{\eta}(x,v)} t_i \leqslant \eta.$$
(2.6.1)

By continuity of t(x, v), see Lemma 2.6.2, and the fact that t(x, v) = 0 if and only if  $n(x) \cdot v \ge 0$ , we have that for  $\eta$  small enough (2.6.1) yields

$$\sum_{i>N_{\eta}(x,v)} |n(x_i) \cdot v_i| \leqslant \varepsilon/4, \tag{2.6.2}$$

where  $v_i$  is the velocity after the  $i^{th}$  rebound and  $x_i$  is the  $i^{th}$  footprint.

t(x, v) is uniformly continuous on the compact  $\partial \Omega \times \{|v| = v_M\}$  (see Lemma 2.6.2) therefore the footprints of  $(X_s(x, v))_{s \in [0,t]}$  are uniformly continuous and therefore there exists  $\alpha_X^{(1)} > 0$  and  $N_{max}$  in  $\mathbb{N}$  such that

$$\forall x, x' \in \partial \Omega \quad \text{s.t.} \quad \left\| x - x' \right\| \leq \alpha_X, \ \forall v_m \leq |v| \leq v_M, \quad N_\eta(x, v) \leq N_{max} - 1.$$
(2.6.3)

We have now defined  $N_{max}$ .

The first property to notice is that if  $(X_s(x, v))_{s \in [0,t]}$  has at least two rebounds against the boundary, then at each of them the scalar product between the incoming velocity and the outward normal is less than  $\varepsilon/2N_{max}$ .

Secondly,  $\Omega$  is  $C^2$  and therefore n(x) is uniformly continuous on the boundary. Thus, the specular reflection operator  $\mathcal{R}_x$  is uniformly continuous on  $\partial \Omega \times B(0, v_M)$ :

$$\exists \alpha_X^{(2)} > 0, \, \forall x, x' \in \partial \Omega \quad \text{s.t.} \quad \left\| x - x' \right\| \leqslant \alpha_X, \quad \left\| \mathcal{R}_x - \mathcal{R}_{x'} \right\| \leqslant \varepsilon / 4N_{max}. \tag{2.6.4}$$

We want to be sure that straight trajectories stay in our domain of uniformity so we consider

$$t \leq t_{\varepsilon}(v_M) = \max\left(\frac{\alpha_X}{v_M}, \frac{1}{p_{\varepsilon/2N_{max}}v_M}\right),$$

where  $\alpha_X = \min(\alpha_X^{(1)}, \alpha_X^{(2)})$  defined in (2.6.3) and (2.6.4). To conclude, thanks to (2.6.3) and (2.6.2), if  $(X_s(x, v))_{s \in [0,t]}$  collides at least twice with the boundary then

$$\forall s \in [0, t], \quad \|v - V_s(x, v)\| \leq 2 \sum_{i \in \mathbb{N}} |n(x_i) \cdot v_i| \leq 2 \sum_{i \leq N_{max} - 1} \frac{\varepsilon}{4N_{max}} + 2\frac{\varepsilon}{4} = \varepsilon.$$

Roughly speaking we do not allow the velocities near the critical direction to bounce against the wall and for the grazing ones we run them for a short time, preventing them from escaping a small neighbourhood where the collisions behave almost the same everywhere (see Figure 2.5).

To conclude our proof, it only remains to find  $l \leq 1/p_{\varepsilon/2N_{max}}$  that prevents trajectories staying in  $\Omega_l$  but go through only one rebound with a scalar product greater than  $\varepsilon/2$ from happening. This is easily achieved by taking l small enough such that not a single trajectory with a scalar product greater than  $\varepsilon/2N_{max}$  can stay inside  $\Omega_l$  during a time  $\tau$ . Indeed, one part of these trajectories will overcome a straight line of lenght at least  $v_m \tau/2$ and making a scalar product greater than  $\varepsilon/2N_{max}$ . The distance from the boundary of the extremal point of these straight lines is therefore, by convexity, uniformly bounded from below (e.g. in dimension 2 it is bounded by  $v_m \tau \varepsilon/4N_{max}$ . Taking  $l_{\varepsilon}(v_m, \tau)$  being the minimum between this lower bound and  $1/p_{\varepsilon/2N_{max}}$  gives us the required distance from the boundary.



Figure 2.5: Control on grazing trajectories

**Remark 2.6.3** In the case of  $\Omega$  is a strictly convex  $C^3$  domain, the proof of Proposition 2.6.1 can be easily made constructive thanks to the tools developed by Guo [57].

In that case we have the existence of  $\xi : \mathbb{R}^d \longrightarrow \mathbb{R}$  to be  $C^3$  such that

$$\Omega = \{ x \in \mathbb{R}^d, \quad \xi(x) < 0 \}$$

and such that  $\nabla \xi \neq 0$  on  $\partial \Omega$  and there exists  $C_{\xi} > 0$  such that

$$\partial_{ij}\xi(x)v_iv_j \geqslant C_{\xi} \|v\|^2$$

for all x in  $\overline{\Omega}$  and all v in  $\mathbb{R}^d$ . It allows us to define the following bounded functional along a characteristic trajectories  $(X_s, V_s)$ ,

$$\alpha(s) = \xi^2(X_s) + \left[V_s \cdot \nabla \xi(X_s)\right]^2 - 2\left[V_s \cdot \nabla^2 \xi(X_s) \cdot V_s\right] \xi(X_s) \ge 0.$$

The latter functional satisfies that if  $X_{s_0}$  is on  $\partial \Omega$  then

$$\alpha(s_0) = [V_{s_0} \cdot \nabla \xi(X_{s_0})]^2 = [V_{s_0} \cdot n(X_{s_0})]^2 |\nabla \xi(X_{s_0})|^2.$$

 $\alpha$  thus encodes the evolution of the scalar product between the velocity of the trajectory and the normal to  $\Omega$  at the footprints of the characteristic. If the characteristic trajectory starts with a velocity v such that  $v_m \leq ||v|| \leq v_M$ , as in Proposition 2.6.1, Lemma 1 and Lemma 2 of [57] shows that in between two consecutive collision with the boundary at time  $s_1$  and  $s_2$  we have the existence of  $C_{\xi} > 0$  such that

$$|s_1 - s_2| \geq C_{\xi} \frac{\sqrt{\alpha(s_1)}}{v_M^2}, \qquad (2.6.5)$$

$$e^{C_{\xi}(v_m+1)s_1}\alpha(s_1) \leqslant e^{C_{\xi}(v_M+1)s_2}\alpha(s_2),$$
 (2.6.6)

$$e^{-C_{\xi}(v_M+1)s_1}\alpha(s_1) \ge e^{-C_{\xi}(v_M+1)s_2}\alpha(s_2).$$
 (2.6.7)

With (2.6.5) we can control the minimum time between two consecutive collisions with the boundary and therefore the minimum lenght of a segment between two consecutive collisions, uniformly in x and v (since  $\nabla \xi$  is bounded from below on  $\partial \Omega$  and non-vanishing). We therefore obtain a uniform maximum number of collisions during the given time T. Finally, (2.6.6) and (2.6.7) bounds uniformly the evolution of the scalar product between two consecutive collision and therefore the maximum evolution of  $V_s(x, v)$  on the whole trajectory for a given time T. Plugging those constructive constants into the study we just made gives explicit constants in Proposition 2.6.1.

Now that we understand how grazing trajectories behave geometrically we can turn our attention to their effects combined with the spreading property of the Boltzmann  $Q^+$  operator.

### 2.6.2 Spreading effect along grazing trajectories

In order to use the geometrical behaviour of grazing characteristic trajectories, one needs to consider velocities that are bounded from below. However, we would like to spread a lower bound up to ball centred at 0 where a lower bound on the norm of velocities is impossible. We shall overcome this problem using the flexibility of the spreading property of the  $Q^+$  operator, Lemma 2.4.2, which allows us to extend the radius of the ball from 0 up to  $\sqrt{2}$  times the initial radius.

The idea is to spread the initial lower bound by induction as long as the origin is strictly outside, where we are allowed to use the geometrical property of grazing characteristics. Finally, a last iteration of the spreading property, not requiring any *a priori* knowledge on characteristics, will include 0 in the lower bound.

In Corollary 3 we can fix a special time  $\tau_1$  of crossing the frontier of some  $\Omega_l$  allowing us to derive a lower bound for our function in this special case. The second case of grazing trajectories is dealt with Proposition 2.6.1 where we can find an l for  $\Omega_l$  to control the evolution of the velocity. Our goal now will be to find all the constants that are still free and to finally find a time of collision small enough that it will remain the same during all the iteration scheme.

We now fix all the constants that remain to be fixed in Corollary 3 thanks to Proposition 2.6.1.

Let

$$\Delta_T = \min\left(\delta_T, t_{\delta_V/4}(3R_{min})\right). \tag{2.6.8}$$

Next we define, for  $\xi$  in (0, 1),

$$\begin{cases} r_0(\xi) = \delta_V \\ r_{n+1}(\xi) = \sqrt{2}(1-\xi)r_n(\xi) - \frac{\delta_V}{4}. \end{cases}$$
(2.6.9)

We have that  $(r_n(1/2 - 5/(8\sqrt{2})))_{n \in \mathbb{N}}$  is a strictly increasing sequence. Therefore, it exists  $N_{max}$  such that

$$r_{N_{max}}\left(\frac{1}{2}-\frac{5}{8\sqrt{2}}\right) \geqslant 2R_{min}.$$

Now we fix N in  $\mathbb{N}^*$  greater than  $N_{max}$ . With this N and Proposition 2.4.4 at  $\tau_0 = 0$ , we construct  $v_1, \ldots, v_{N_X}$ .

For i in  $\{1, \ldots, N\}$  we take  $\xi^{(i)}$  in  $(0, 1/4 - 5/(8\sqrt{2})]$  and we define  $N_{max}(i)$  to be such

that  $0 \notin B\left(v_i, r_n(\xi^{(i)})\right)$  for all  $n < N_{max}(i)$  and  $0 \in B\left(v_i, r_{N_{max}(i)}(\xi^{(i)})\right)$ . We can in fact take  $\xi^{(i)}$  such that  $0 \in \text{Int}\left(B\left(v_i, r_{N_{max}(i)}(\xi^{(i)})\right)\right)$ .

Therefore we have that for all i in  $\{1, \ldots, N_X\}$ ,

$$\delta_i = \|v_i\| - r_{N_{max}(i)-1}(\xi^{(i)}) \ge 0,$$

which is strictly positive if and only if  $N_{max}(i) > 0$ . We consider

$$v_m = \min_{i \in \{1, \dots, N_X\}} \{\delta_i; \, \delta_i > 0\}.$$
(2.6.10)

We can now define:

$$\forall 0 < \tau \leq \Delta_T, \qquad R(\tau) = \max\left(3R_{min}, \frac{2\delta_X}{\tau} + 1\right),$$
(2.6.11)

$$\tau_1(\tau) = \tau - \frac{2\delta_X}{R(\tau)} > 0, \qquad (2.6.12)$$

$$\tilde{t}(\tau) = \tilde{t}(l(\tau), \tau_1(\tau), \Delta_T).$$
(2.6.13)

Finally, we define  $l(\tau)$ 

$$\forall 0 < \tau \leq \Delta_T, \qquad \tau_2(\tau) = \min\left(\Delta_T, \frac{\delta_X}{R(\tau)}\right),$$
(2.6.14)

$$l(\tau) = \min\left(\delta_X, l_{\delta_V/4}(v_m, \tau_2(\tau))\right).$$
 (2.6.15)

We also build up the following sequence, where R, l and  $\tau_1$  depend on  $\tau$ ,

$$\begin{cases} b_0^{(i)}(\tau, \Delta_T) = a_0 e^{-(\Delta_T - \tau)C_L \langle R \rangle^{\gamma^+}} \\ b_{n+1}^{(i)}(\tau, \Delta_T) = \min \left( C_Q r_n^{d+\gamma}(\xi^{(i)})^{d/2 - 1} \frac{\delta_X}{2^{n+2}R} e^{-\tau C_L \langle R \rangle^{\gamma^+}} b_n^{(i)}(\tau, \Delta_T)^2; a(l, \tau_1, \Delta_T) \right) \\ \end{cases}$$
(2.6.16)

 $\xi^{(i)}$  was defined above and  $a(l, \tau, \Delta_T)$  was defined in Corollary 3.

We are now ready to state the next Proposition which is the complement of Proposition 2.5.1 in the case when the trajectory stays close to the boundary. We remind the reader that  $0 < \tilde{t}(\tau) < \tau_1(\tau)$ .

**Proposition 2.6.4** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3 and suppose that f satisfies Proposition 2.4.4 with  $\tau_0 = 0$ .

Consider  $0 < \tau \leq \Delta_T$  and take *i* in  $\{1, \ldots, N_X\}$  such that  $N_{max}(i) > 1$ . For all *n* in  $\{0, \ldots, N_{max}(i) - 1\}$  we have that for all *t* in  $[\tau - \delta_X/(2^n R(\tau)), \Delta_T]$ , all *x* in  $B(x_i, \delta_X/2^n)$  and all *v* in  $B(0, R(\tau))$ , if

$$\forall s \in [0, t - \tilde{t}(\tau)], \quad X_{s,t}(x, v) \notin \Omega_{l(\tau)}$$

then

$$f(t, x, v) \ge b_n^{(i)}(\tau, \Delta_T) \mathbf{1}_{B(v_i, r_n(\xi^{(i)}))}(v),$$

all the constants being defined in (2.6.8), (2.6.9), (2.6.15), (2.6.11), (2.6.12), (2.6.13) and (2.6.16).

**Proof of Proposition** 2.6.4 We are going to use the same kind of induction we used to prove Proposition 2.5.1. So we start by fixing *i* such that  $N_{max}(i) > 1$ .

Step 1: Initialization. The initialisation is simply Proposition 2.4.4 and the first term in the Duhamel formula (2.2.10) starting at  $\tau$ , with the control from above on L thanks to Lemma 2.4.1.

Stef 2: Proof of the induction. We consider the case where the Proposition is true at  $n \leq N_{max}(i) - 2$ .

We take t in  $[\tau - \delta_X/(2^{n+1}R(\tau)), \Delta_T]$ , x in  $B(x_i, \delta_X/2^{n+1})$  and all v in  $B(0, R(\tau))$ .

We suppose now that for all  $s \in [0, t - \tilde{t}(\tau)]$  we have that  $X_{s,t}(x, v)$  does not belongs to  $\Omega_{l(\tau)}$ .

To shorten notation we will skip the dependence in  $\tau$  of the constant.

We use the definition of f being a mild solution to write f(t, x, v) under its Duhamel form (2.2.10) where both parts are positive. As in the proof of Proposition 2.5.1, we control, uniformly on t, x and v, the L operator from above. This yields

$$f(t,x,v) \ge e^{-C_L \tau \langle R \rangle^{\gamma^+}} \int_{t-\frac{\delta_X}{2^{n+1}R}}^{t-\frac{\delta_X}{2^{n+2}R}} Q^+ \left[ f(s, X_{s,t}(x,v), \cdot), f(s, X_{s,t}(x,v), \cdot) \right] \left( V_{s,t}(x,v) \right) \, ds,$$

$$(2.6.17)$$

where we used  $||V_{s,t}(x,v)|| = ||v|| \leq R$ . We also emphasize here that this inequality holds true thanks to the definition of (2.6.11):

$$t - \frac{\delta_X}{2^{n+1}R} \ge \tau - \frac{\delta_X}{R} > 0.$$

The goal is now to apply the induction to the triplet  $(s, X_{s,t}(x, v), v_*)$ , where  $v^*$  is the integration parameter inside the  $Q^+$  operator, with  $||v_*|| \leq R$ .

We notice first that for all s in  $[t - \delta_X/(2^{n+1}R), t - \delta_X/(2^{n+2}R)]$ 

$$\begin{aligned} \|x_i - X_{s,t}(x,v)\| &\leqslant \quad \frac{\delta_X}{2^{n+1}} + \|x - X_{s,t}(x,v)\| \\ &\leqslant \quad \frac{\delta_X}{2^{n+1}} + (t-s)R \leqslant \frac{\delta_X}{2^n}, \end{aligned}$$

so that for all s in  $[t - \delta_X/(2^{n+1}R), t - \delta_X/(2^{n+2}R)], X_{s,t}(x,v)$  belongs to  $B(x_i, \delta_X/2^n)$ .

We also note that

$$\left[t - \frac{\delta_X}{2^{n+1}R}, t - \frac{\delta_X}{2^{n+2}R}\right] \subset \left[\tau - \frac{\delta_X}{2^n R}, \Delta_T\right].$$

We have two different cases to consider for  $(X_{s',s}(X_{s,t}(x,v),v_*))_{s'\in[0,s-\tilde{t}]}$ .

Either for some s' in  $[0, s - \tilde{t}]$ ,  $X_{s',s}(X_{s,t}(x, v), v_*)$  belongs to  $\Omega_l$  and then we can apply Corollary 3:

$$f(s, X_{s,t}(x, v), v_*) \geq a(l, \tau_1, \Delta_T) \mathbf{1}_{B(0, 2R_{min})}(v_*) \\\geq b_n^{(i)}(\tau, \Delta_T) \mathbf{1}_{B(v_i, r_n(\xi^{(i)}))}(v),$$
(2.6.18)

since  $v_i$  is in  $B(0, R_{min})$ .

Or for all s' in  $[0, s - \tilde{t}] \subset [0, \tau_2]$ ,  $X_{s',s}(X_{s,t}(x, v), v_*)$  does not belong to  $\Omega_l$  and then we can apply our induction property at rank n and we reach the same lower bound (2.6.18).

Plugging (2.6.18) into (2.6.17) implies, thanks to the spreading property of  $Q^+$ , Lemma 2.4.2 with  $\xi = \xi^{(i)}$ ,

$$f(t, x, v) \geq (2.6.19)$$

$$C_Q r_n^{d+\gamma}(\xi^{(i)})^{d/2-1} e^{-\tau C_L \langle R \rangle^{\gamma^+}} (b_n^{(i)})^2 \int_{t-\frac{\delta_X}{2^{n+1}R}}^{t-\frac{\delta_X}{2^{n+2}R}} \mathbf{1}_{B(v_i,\sqrt{2}(1-\xi^{(i)})r_n(\xi^{(i)}))} (V_{s,t}(x, v)) \ ds.$$

To conclude we use the fact that for all s in  $[0, t - \tilde{t}]$  we have that  $X_{s,t}(x, v)$  does not belong to  $\Omega_l$  and that  $t - \tilde{t} > \tau_2$ . Moreover,  $n + 1 \leq N_{max}(i) - 1$  and so if v belongs to  $B(v_i, r_n(\xi^{(i)}))$  we have that  $v_m \leq ||v||$ . We apply Proposition 2.6.1, raising

$$\forall s \in \left[t - \frac{\delta_X}{2^{n+1}R}, t - \frac{\delta_X}{2^{n+2}R}\right], \quad \|v - V_{s,t}(x,v)\| \leq \frac{\delta_V}{4}.$$

Therefore, if v belongs to  $B(v_i, r_{n+1}(\xi^{(i)}))$  we have that  $V_{s,t}(x, v)$  belongs to  $B(v_i, \sqrt{2}(1-\xi^{(i)})r_n(\xi^{(i)}))$  for all s in  $[t - \delta_X/(2^{n+1}R), t - \delta_X/(2^{n+2}R)]$ .

Therefore if v belongs to  $B(v_i, r_{n+1}(\xi^{(i)}))$  we can compute explicitly (2.6.19) and obtain the expected induction.

Thanks to Proposition 2.6.4, we can build, for all x and all v, a lower bound that will contain 0 in its interior after another use of the spreading property of the  $Q^+$  operator. The next Corollary is the complement of Corollary 3.

**Corollary 4** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3 and suppose that f satisfies Proposition 2.4.4 with  $\tau_0 = 0$ .

Let  $\Delta_T$  be defined by (2.6.8).

There exists  $r_V > 0$  such that for all  $\tau \in (0, \Delta_T]$  there exists  $b(\tau) > 0$  such that for all t in  $[\tau, \Delta_T]$ 

If, for  $\tilde{t}(\tau)$  and  $l(\tau)$  being defined by (2.6.13) – (2.6.15),

$$\forall s \in [0, t - \tilde{t}(\tau)], \quad X_{s,t}(x, v) \notin \Omega_{l(\tau)}.$$

Then

$$f(t, x, v) \ge b(\tau) \mathbf{1}_{B(0, r_V)}(v).$$

**Proof of Corollary** (4) We are going to use the spreading property of  $Q^+$  one more time. We recall that we chose  $N \ge N_{max} \ge N_{max}(i)$  for all *i*. By definition of  $N_{max}(i)$ ,

$$\forall i \in \{1, \dots, N_X\}, \quad 0 \in \operatorname{Int}\left(B\left(v_i, r_{N_{max}(i)}(\xi^{(i)})\right)\right).$$

We define

$$r_V = \min\left\{r_{N_{max}(i)}(\xi^{(i)}) - \|v_i\|; \ i \in \{1, \dots, N_X\}\right\},\$$

which only depends on  $\delta_V$  and  $(v_i)_{i \in \{1, \dots, N_X\}}$ . By construction we see that

$$\forall i \in \{1, \dots, N_X\}, B(0, r_V) \subset B\left(v_i, r_{N_{max}(i)}(\xi^{(i)})\right).$$
(2.6.20)

Now we take  $\tau$  in  $(0, \Delta_T]$  and we take t in  $[\tau, \Delta_T]$ , x in  $B(x_i, \delta_X/2^N)$  and v in  $B(0, R(\tau))$  such that

$$\forall s \in [0, t - \tilde{t}(\tau)], \quad X_{s,t}(x, v) \notin \Omega_{l(\tau)},$$

We have that t is in  $[\tau - \delta_X/(2^{N_{max}(i)-1}R(\tau)), \Delta_T]$  and x in  $B(x_i, \delta_X/2^{N_{max}(i)-1})$  $(N \ge N_{max}(i))$ . By the same methods we reached (2.6.19), we obtain for  $n = N_{max}(i)$ 

$$f(t, x, v) \geq (2.6.21)$$

$$C_Q r_n^{d+\gamma}(\xi^{(i)})^{d/2-1} e^{-\tau C_L \langle R \rangle^{\gamma^+}} (b_n^{(i)})^2 \int_{t-\frac{\delta_X}{2^{n+1}R}}^{t-\frac{\delta_X}{2^{n+2}R}} \mathbf{1}_{B(v_i,\sqrt{2}(1-\xi^{(i)})r_n(\xi^{(i)}))} (V_{s,t}(x, v)) \ ds.$$

This time the conclusion is different because we cannot bound the velocity from below since our lower bound contains 0. However, (2.6.20) allows us to bound from below the integrand in (2.6.21) by a function depending only on the norm. Moreover,  $||v|| = ||V_{s,t}(x, v)||$ along characteristic trajectories (see Proposition (2.3.8)). Thus we obtain the expected result by taking

$$b(\tau) = \min\left\{b_{N_{max}(i)}^{(i)}; i \in \{1, \dots, N_X\}\right\}.$$

# 2.7 Maxwellian lower bound in the cutoff case: proof of Theorem 2.2.3

This section gathers all the results we proved above and proves the main Theorem in the case of a cut-off collision kernel.

### **2.7.0.1** Proof of Proposition (2.2.4)

By combining Corollary 3 and Corollary 4 we can deal with any kind of characteristic trajectory. This is expressed by the following lemma.

**Lemma 2.7.1** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3 and suppose that f satisfies Proposition 2.4.4 with  $\tau_0 = 0$ . There exists  $\Delta_T > 0$  and  $r_V > 0$  such that for all  $0 < \tau \leq \Delta_T$  there exists  $a(\tau)$  and

$$\forall t \in [\tau, \Delta_T], \ a.e. \ (x, v) \in \bar{\Omega} \times \mathbb{R}^d, \quad f(t, x, v) \ge a(\tau) \mathbf{1}_{B(0, r_V)}(v).$$

**Proof of Lemma** 2.7.1 In Corollary 4 we constructed  $\Delta_T$  and  $r_V$ .

We now take  $\tau$  in  $(0, \Delta_T]$  and consider t in  $[\tau, \Delta_T]$ , (x, v) in  $\overline{\Omega} \times \mathbb{R}^d$  where f is a mild solution of the Boltzmann equation.

We remind the reader that  $l(\tau)$  and  $\tilde{t}(\tau)$  have been introduced in (2.6.15) and (2.6.13). Either  $(X_{s,t}(x,v))_{s\in[0,t-\tilde{t}(\tau)]}$  meets  $\Omega_{l(\tau)}$  and then we use Corollary 3 to get

$$f(t, x, v) \ge a(l(\tau), \tau_1(\tau), \Delta_T) \mathbf{1}_{B(0, r_V)}(v).$$

Or  $(X_{s,t}(x,v))_{s\in[0,t-\tilde{t}(\tau)]}$  stays out of  $\Omega_{l(\tau)}$  and then we use Corollary 4 to get

$$f(t, x, v) \ge b(\tau) \mathbf{1}_{B(0, r_V)}(v).$$

We obtain Lemma 2.7.1 with  $a(\tau) = \min(a(l(\tau), \tau_1(\tau), \Delta_T), b(\tau))$ .

We now have all the tools to prove Proposition 2.2.4.

### **Proof of Proposition** 2.2.4 Let $\tau$ be strictly positive and consider t in $[\tau/2, \tau]$ .

**First case**. We suppose that f satisfies Proposition 2.4.4 with  $\tau_0 = 0$ .

We can compare t with  $\Delta_T$  constructed in Lemma 2.7.1.

If  $t \leq \Delta_T$  then we can apply the latter lemma and obtain for almost every (x, v) in  $\overline{\Omega} \times \mathbb{R}^d$ 

$$f(t, x, v) \ge a\left(\frac{\tau}{2}\right) \mathbf{1}_{B(0, r_V)}(v).$$

$$(2.7.1)$$

If  $t \ge \Delta_T$  then we can use Duhamel formula (2.2.10) and bound f(t, x, v) by its value at time  $\Delta_T$  (as we did in the first step of the induction in the proof of Proposition 2.5.1) and use Lemma 2.7.1 at  $\Delta_T$ . This gives, for  $||v|| \le r_V$ ,

$$f(t, x, v) \geq f(\Delta_T, X_{\Delta_T, t}(x, v), V_{\Delta_T, t}(x, v))e^{-(t - \Delta_T)C_L \langle r_V \rangle^{\gamma^+}}$$
$$\geq a(\Delta_T)e^{-(\tau - \Delta_T)C_L \langle r_V \rangle^{\gamma^+}} \mathbf{1}_{B(0, r_V)}(V_{\Delta_T, t}(x, v))$$
$$= a(\Delta_T)e^{-(\tau - \Delta_T)C_L \langle r_V \rangle^{\gamma^+}} \mathbf{1}_{B(0, r_V)}(v).$$
(2.7.2)

We just have to take the minimum of the two lower bounds (2.7.1) and (2.7.2) to obtain Proposition 2.2.4.

Second case. We do not assume anymore that f satisfies Proposition 2.4.4 with  $\tau_0 = 0$ .

Thanks to Proposition 2.4.4 with  $\tau_0 = \tau/4$  we have that

$$\forall t \leq 0, \ \forall x \in \overline{\Omega}, \ v \in \mathbb{R}^d, \quad F(t, x, v) = f(t + \tau_0, x, v)$$

is a mild solution of the Boltzmann equation satisfying exactly the same bounds as f in Theorem 2.2.3 and such that F has the property of Proposition 2.4.4 at 0 (note that all the constants depend on  $\tau_0$ ).

Hence, we can apply the first step for t' in  $[\tau/4, 3\tau/4]$  and F(t', x, v). This gives us the expected result for f(t, x, v) for  $t = t' + \tau_0$  in  $[\tau/2, \tau]$ .

### 2.7.1 Proof of Theorem 2.2.3

As was mentioned in Section 2.1.2, the main difficulty in the proof is to create a lower bound depending only on the norm of the velocity. This has been achieved thanks to Proposition 2.2.4. If we consider this proposition as the start of an induction then it leads to exactly the same process developed by Mouhot in [78], Section 3. Therefore we will just explain how to go from Proposition 2.2.4 to Theorem 2.2.3, without writing too many details.

First of all, by using the spreading property of the  $Q^+$  operator once again we can grow the lower bound derived in Proposition 2.2.4.

**Proposition 2.7.2** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.3.

For all  $\tau$  in (0,T), there exists  $R_0 > 0$  such that

$$\forall n \in \mathbb{N}, \forall t \in \left[\tau - \frac{\tau}{2^{n+1}}, \tau\right], \forall (x, v) \in \bar{\Omega} \times \mathbb{R}^d, f(t, x, v) \ge a_n(\tau) \mathbf{1}_{B(0, r_n)}(v),$$

with the induction formulae

$$a_{n+1}(\tau) = \operatorname{cst} C_e \frac{a_n^2(\tau) r_n^{d+\gamma} \xi_n^{d/2+1}}{2^{n+1}}$$
 and  $r_{n+1} = \sqrt{2}(1-\xi_n) r_n$ 

where  $(\xi_n)_{n\in\mathbb{N}}$  is any sequence in (0,1) and  $r_0 = r_V$ ,  $a_0(\tau)$  and  $C_e$  only depend on  $\tau$ ,  $E_f$ (and  $L_f^{p_{\gamma}}$  if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ ).

Indeed, we take the result in Proposition 2.2.4 to be the first step of our induction and then, for n in N and  $0 < \tau < T$ , the Duhamel form of f gives

$$f(t, x, v) \geq \int_{\tau - \frac{\tau}{2^{n+1}}}^{\tau - \frac{\tau}{2^{n+2}}} e^{-C_L(t-s)\langle v \rangle^{\gamma^+}} Q^+ \left( f(s, X_{s,t}(x, v), \cdot), f(s, X_{s,t}(x, v), \cdot) \right) \left( V_{s,t}(x, v) \right) ds,$$

for t in  $[\tau - \tau/2^{n+2}, \tau]$ .

Using the induction hypothesis together with the spreading property of  $Q^+$  (Lemma 2.4.2) leads us, as in the proofs of Propositions 2.5.1 and 2.6.4, to a bigger ball in velocity, centred at 0. The only issue is to avoid the *v*-dependence in  $\exp\left[-C_L(t-s)\langle v \rangle^{\gamma^+}\right]$  which can easily be achieved as shown at the end of the proof of Proposition 3.2 in [78]. This is exactly the same result as Proposition 3.2 in [78], but with the added uniformity in x.

As in Lemma 3.3 in [78], we can take an appropriate sequence  $(\xi_n)_{n \in \mathbb{N}}$  and look at the asymptotic behaviour of  $(a_n(\tau))_{n \in \mathbb{N}}$ . We obtain the following

$$\forall \tau > 0, \ \exists \rho_{\tau}, \theta_{\tau} > 0, \ \forall (x, v) \in \bar{\Omega} \times \mathbb{R}^{d}, \quad f(t, x, v) \ge \frac{\rho_{\tau}}{(2\pi\theta_{\tau})^{d/2}} e^{-\frac{|v|^{2}}{2\theta}}.$$

Notice that, again, the result is uniform in space, since the previous one was, and that the constants  $\rho_{\tau}$  and  $\theta_{\tau}$  only depend on  $\tau$  and the physical quantities associated to f.

To conclude, it remains to make the result uniform in time. As noticed in [78], Lemma 3.5, the results we obtained so far do not depend on an explicit form of  $f_0$  but just on uniform bounds and continuity that are satisfied at all times, positions and velocities. Therefore, we can do the same arguments starting at any time and not t = 0. So if we take  $\tau > 0$  and consider  $\tau \leq t < T$  we just have to make the proof start at  $t - \tau$  to obtain Theorem 2.2.3.

# 2.8 Exponential lower bound in the non cutoff case: proof of Theorem 2.2.6

In this section we prove the immediate appearance of an exponential lower bound for solutions to the Boltzmann equation (2.1.1) in the case of a collision kernel satisfying the non cutoff property.

The definition of being a mild solution in the case of a non cutoff collision kernel, Definition 2.2.5 and equation (2.2.12), shows that we are in fact dealing with an almost cutoff kernel to which we add a non locally integrable remainder. The strategy will mainly follow what we did in the case of a cutoff collision kernel with the addition of controlling the loss due to the added term.

As in the last section, we shall first prove that solutions to the Boltzmann equation can be uniformly bounded from below by a lower bound depending only on the norm of the velocity and then use the proof given for the non cutoff case in [78]. We will do that by proving the immediate appearance of localised "upheaval points" and spreading them up to the point where we reach a uniform lower bound that includes a ball in velocity centred at the origin. The spreading effect will be done both in the case where the trajectories
reach a point far from the boundary and in the case of grazing trajectories. At this point we will spread this lower bound on the norm of the velocity up to the exponential lower bound we expect.

We gather here two lemmas, proved in [78], which we shall use in this section. They control the  $L^{\infty}$ -norm of the linear operator  $S_{\varepsilon}$  and of the bilinear operator  $Q_{\varepsilon}^{1}$ . We first give a property satisfied by the linear operator S, (2.2.12), which is Corollary 2.2 in [78], where we define

$$m_b = \int_{\mathbb{S}^{d-1}} b\left(\cos\theta\right) \left(1 - \cos\theta\right) d\sigma = \left|\mathbb{S}^{d-2}\right| \int_0^\pi b\left(\cos\theta\right) \left(1 - \cos\theta\right) \sin^{d-2}\theta \, d\theta.$$
(2.8.1)

**Lemma 2.8.1** Let g be a measurable function on  $\mathbb{R}^d$ . Then

$$\forall v \in \mathbb{R}^d, \quad |S[g](v)| \leq C_q^S \langle v \rangle^{\gamma^+},$$

where  $C_g^S$  is defined by:

1. If  $\Phi$  satisfies (2.1.4) with  $\gamma \ge 0$  or if  $\Phi$  satisfies (2.1.5), then

$$C_q^S = \operatorname{cst} m_b C_\Phi e_q.$$

2. If  $\Phi$  satisfies (2.1.4) with  $\gamma \in (-d, 0)$ , then

$$C_g^S = \operatorname{cst} m_b C_\Phi \left[ e_g + l_g^p \right], \quad p > d/(d+\gamma).$$

We will compare the lower bound created by the cutoff part of our kernel to the remaining part  $Q_{\varepsilon}^1$ . To do so we need to control its  $L^{\infty}$ -norm. This is achieved thanks to Lemma 2.5 in [78], which we recall here.

**Lemma 2.8.2** Let  $B = \Phi b$  be a collision kernel satisfying (2.1.3), with  $\Phi$  satisfying (2.1.4) or (2.1.5) and b satisfying (2.1.6) with  $\nu \in [0, 2)$ . Let f, g be measurable functions on  $\mathbb{R}^d$ . Then

1. If  $\Phi$  satisfies (2.1.4) with  $2 + \gamma \ge 0$  or if  $\Phi$  satisfies (2.1.5), then

$$\forall v \in \mathbb{R}^d, \quad \left| Q_b^1(g, f)(v) \right| \leq \operatorname{cst} m_b C_\Phi \, \|g\|_{L^1_{\tilde{\gamma}}} \, \|f\|_{W^{2,\infty}} \, \langle v \rangle^{\tilde{\gamma}}.$$

2. If  $\Phi$  satisfies (2.1.4) with  $2 + \gamma < 0$ , then

$$\forall v \in \mathbb{R}^d, \quad \left| Q_b^1(g, f)(v) \right| \leq \operatorname{cst} m_b C_\Phi \left[ \left\| g \right\|_{L^1_{\tilde{\gamma}}} + \left\| g \right\|_{L^p} \right] \left\| f \right\|_{W^{2,\infty}} \langle v \rangle^{\tilde{\gamma}}$$

with  $p > d/(d + \gamma + 2)$ .

#### 2.8.1 A lower bound only depending on the norm of the velocity

In this section we prove the following proposition, which is exactly Proposition 2.2.4 in the non-cutoff framework.

**Proposition 2.8.3** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.6.

For all  $0 < \tau < T$  there exists  $a_0(\tau) > 0$  such that

$$\forall t \in [\tau/2, \tau], \, \forall (x, v) \in \bar{\Omega} \times \mathbb{R}^d, \quad f(t, x, v) \ge a_0(\tau) \mathbf{1}_{B(0, r_V)}(v),$$

 $r_V$  and  $a_0(\tau)$  only depending on  $E_f$ ,  $E'_f$ ,  $W_f$  (and  $L_f^{p_\gamma}$  if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ ).

**Proof of Proposition 2.8.3** As before, we would like to create localised "upheaval points" (as the ones created in Proposition 2.4.4) and then extend them. Both steps are done, as in the cutoff case, by induction along the characteristics.

We have the following inequality

$$Q_{\varepsilon}^{+}(f,f) + Q_{\varepsilon}^{1}(f,f) \ge Q_{\varepsilon}^{+}(f,f) - \left|Q_{\varepsilon}^{1}(f,f)\right|.$$

$$(2.8.2)$$

From the definition of being a mild solution in the non-cutoff case (Definition 2.2.5), for any  $0 < \varepsilon < \varepsilon_0$ ,

$$(2.8.3)$$

$$f(t, X_t(x, v), V_t(x, v)) = f_0(x, v) \exp\left[-\int_0^t (L_{\varepsilon} + S_{\varepsilon}) \left[f(s, X_s(x, v), \cdot)\right](V_s(x, v)) \, ds\right]$$

$$+ \int_0^t \exp\left(-\int_s^t (L_{\varepsilon} + S_{\varepsilon}) \left[f(s', X_{s'}(x, v), \cdot)\right](V_{s'}(x, v)) \, ds'\right)$$

$$(Q_{\varepsilon}^+ + Q_{\varepsilon}^1) \left[f(s, X_s(x, v), \cdot), f(s, X_s(x, v), \cdot)\right](V_s(x, v)) \, ds.$$

Due to Lemmas 2.4.1, 2.8.1 and 2.8.2 we find that

$$L_{\varepsilon}[f] \leqslant C_f n_{b_{\varepsilon}^{CO}} \langle v \rangle^{\gamma^+}, \quad S_{\varepsilon}[f] \leqslant C_f m_{b_{\varepsilon}^{NCO}} \langle v \rangle^{\gamma^+}$$
 (2.8.4)

and

$$\left|Q_{\varepsilon}^{1}(f,f)\right| \leqslant C_{f} m_{b_{\varepsilon}^{NCO}} \langle v \rangle^{(2+\gamma)^{+}}$$

$$(2.8.5)$$

where  $C_f > 0$  is a constant depending on  $E_f$ ,  $E'_f$ ,  $W_f$  (and  $L_f^{p_{\gamma}}$  if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ ).

The proof of Proposition 2.8.3 is divided into three different inductions that are dealt with in the same way as in the proof of Proposition 2.2.4. Each induction represents a step in the proof: one to create localised initial lower bounds (Lemma 2.4.3), another one to deal with non-grazing trajectories (Proposition 2.5.1) and the final one for grazing trajectories (Proposition 2.6.4). Therefore, we will just point out below the only changes we need to make those inductions work in the non-cutoff case.

In all the inductions in the cutoff case, the key point of the induction was to control at each step quantities of the form

$$f(t,x,v) \geq \int_{t_n^{(1)}}^{t_n^{(2)}} \exp\left(-\int_s^t \left(L_{\varepsilon} + S_{\varepsilon}\right) \left[f(s', X_{s'}(x,v), \cdot)\right](V_{s'}(x,v)) \, ds'\right) \\ \left(Q_{\varepsilon}^+ + Q_{\varepsilon}^1\right) \left[f(s, X_s(x,v), \cdot), f(s, X_s(x,v), \cdot)\right](V_s(x,v)) \, ds$$

where  $(t_n^{(1)})_{n \in \mathbb{N}}$ ,  $(t_n^{(2)})_{n \in \mathbb{N}}$  are defined differently for grazing and non-grazing trajectories (see proofs of Propositions 2.5.1 and 2.6.4).

Much like those previous induction, and using (2.8.2), (2.8.3) and (2.8.4) – (2.8.5), if  $f(t, x, v) \ge a_n \mathbf{1}_{B(\bar{v}, r_n)}$  then

$$f(t, x, v) \ge \int_{t_n^{(1)}}^{t_n^{(2)}} e^{-C_f^{\varepsilon}(R)} \left( a_n^2 Q_{\varepsilon}^+ [\mathbf{1}_{B(\bar{v}, r_n)}, \mathbf{1}_{B(\bar{v}, r_n)}] - C_f m_{b_{\varepsilon}^{NCO}} \langle R \rangle^{(2+\gamma)^+} \right) (V_s(x, v)) \, ds,$$

which leads to

$$f(t,x,v) \ge \int_{t_n^{(1)}}^{t_n^{(2)}} e^{-C_f^{\varepsilon}(R)}$$

$$\left(a_n^2 \operatorname{cst} l_{b_{\varepsilon}^{CO}} c_{\Phi} r_n^{d+\gamma} \xi_n^{\frac{d}{2}-1} \mathbf{1}_{B(\bar{v}, r_n \sqrt{2}(1-\xi_n))} - C_f m_{b_{\varepsilon}^{NCO}} \langle R \rangle^{(2+\gamma)^+} \right) (V_s(x,v)) \, ds,$$
(2.8.6)

due to the spreading property of  $Q_{\varepsilon}^+$  (see Lemma 2.4.2) and using the shorthand notation  $C_f^{\varepsilon}(R) = C_f(n_{b_{\varepsilon}^{CO}} + m_{b_{\varepsilon}^{NCO}})\langle R \rangle^{\gamma^+}$ .

To conclude we notice that, thanks to the definitions (2.4.1), (2.2.9) and (2.8.1),

$$l_{b_{\epsilon}^{CO}} \ge l_{b}$$

and

$$n_{b_{\varepsilon}^{CO}} \underset{\varepsilon \to 0}{\sim} \frac{b_0}{\nu} \varepsilon^{-\nu}, \quad m_{b_{\varepsilon}^{NCO}} \underset{\varepsilon \to 0}{\sim} \frac{b_0}{2 - \nu} \varepsilon^{2-\nu}$$
 (2.8.7)

if  $\nu$  belongs to (0, 2) and

$$n_{b_{\varepsilon}^{CO}} \underset{\varepsilon \to 0}{\sim} b_0 \left| \log \varepsilon \right|, \quad m_{b_{\varepsilon}^{NCO}} \underset{\varepsilon \to 0}{\sim} \frac{b_0}{2 - \nu} \varepsilon^2$$
 (2.8.8)

for  $\nu = 0$ .

Thus, at each step of the inductions we just have to redo the proofs done in the cutoff case and choose  $\varepsilon = \varepsilon_n$  small enough such that

$$C_f m_{b_{\varepsilon_n}^{NCO}} \langle R \rangle^{(2+\gamma)^+} \leqslant \frac{1}{2} a_n^2 \operatorname{cst} l_b c_\Phi r_n^{d+\gamma} \xi_n^{\frac{d}{2}-1}.$$
(2.8.9)

Proposition 2.8.3 follows directly from these choices plugged into the study of the cutoff case. ■

#### 2.8.2 Proof of Theorem 2.2.6

Now that we proved the immediate appearance of a lower bound depending only on the norm of the velocity we can spread it up to an exponential lower bound. As in Section 2.7.1, we thoroughly follow the proof of Theorem 2.1 of [78]. The proof in our case is exactly the same induction, starting from Proposition 2.8.3. Therefore we only briefly describe how to construct the expected exponential lower bound. For more details we refer the reader to [78], Section 4.

We start by spreading the initial lower bound (Proposition 2.8.3) by induction where, at each step, we use the spreading property of the  $Q_{\varepsilon_n}^+$  operator and fix  $\varepsilon_n$  small enough to obtain a strictly positive lower bound (see (2.8.9)).

There is, however, a subtlety in the non-cutoff case that we have to deal with. Indeed, at each step of the induction we choose an  $\varepsilon_n$  of decreasing magnitude, but at the same time in each step the action of the operator  $-(Q_{\varepsilon}^- + Q_{\varepsilon}^2)$  behaves like (see (2.8.6))

$$\exp\left[-C_f\left(m_{b_{\varepsilon_n}^{NCO}}+n_{b_{\varepsilon_n}^{CO}}\right)(t_n^{(1)}-t_n^{(2)})\langle v\rangle^{\gamma^+}\right].$$

By (2.8.7) - (2.8.8), as  $\varepsilon_n$  tends to 0 we have that  $n_{b_{\varepsilon_n}^{CO}}$  goes to  $+\infty$  and so the action of  $-(Q_{\varepsilon}^- + Q_{\varepsilon}^2)$  seems to decrease the lower bound to 0 exponentially fast. The idea to overcome this difficulty is to find a time interval  $t_n^{(1)} - t_n^{(2)} = \Delta_n$  at each step to be sufficiently small to counterbalance the effect of  $n_{b_{\varepsilon_n}^{CO}}$ .

More precisely, by starting from Proposition 2.8.3 as the first step of our induction,

taking

$$t_n^{(1)} = \left(\sum_{k=0}^{n+1} \Delta_k\right) \tau, \quad t_n^{(2)} = \left(\sum_{k=0}^n \Delta_k\right) \tau$$

in (2.8.6) and fixing  $\varepsilon_n$  by (2.8.9) we can prove the following induction property

**Proposition 2.8.4** Let f be the mild solution of the Boltzmann equation described in Theorem 2.2.6.

For all  $\tau$  in (0,T) and any sequence  $(\Delta_n)_{n\in\mathbb{N}}$  such that  $\sum_{n\geq 0} \Delta_n = 1$ ,

$$\forall n \in \mathbb{N}, \forall t \in \left[ \left( \sum_{k=0}^{n} \Delta_k \right) \tau, \tau \right], \forall (x,v) \in \bar{\Omega} \times \mathbb{R}^d, f(t,x,v) \ge a_n(\tau) \mathbf{1}_{B(0,r_n)}(v) \le \frac{1}{2} \mathbf{1}_{$$

with the induction formulae

$$a_{n+1} = \operatorname{cst} \Delta_{n+1} \exp\left[-[\tilde{C}_f a_n^2 r_n^{d+\gamma-\tilde{\gamma}} \xi_n^{d/2-1}]^{-\frac{\nu}{2-\nu}} \left(\sum_{k \ge n+1} \Delta_k\right) r_n^{\gamma+1}\right] a_n^2 r_n^{\gamma+d} \xi_n^{d/2+1}$$

if  $\nu$  is in (0,2),

$$a_{n+1} = \operatorname{cst} \Delta_{n+1} \exp\left[-\operatorname{cst} \log[\tilde{C}_f a_n^2 r_n^{d+\gamma-\tilde{\gamma}} \xi_n^{d/2-1}] \left(\sum_{k \ge n+1} \Delta_k\right) r_n^{\gamma+1}\right] a_n^2 r_n^{\gamma+d} \xi_n^{d/2+1}$$

if  $\nu = 0$  and

$$r_{n+1} = \sqrt{2}r_n(1-\xi_n),$$

where  $(\xi_n)_{n\in\mathbb{N}}$  is any sequence in (0,1) and  $r_0 = r_V$ ,  $a_0(\tau)$  and  $\tilde{C}_f$  depend only on  $\tau$ ,  $E_f$ ,  $E'_f$ ,  $W_f$  (and  $L_f^{p_{\gamma}}$  if  $\Phi$  satisfies (2.1.4) with  $\gamma < 0$ ).

We emphasize here that the induction formulae are obtained thanks to the use of equivalences (2.8.7) and (2.8.8) inside the exponential term

$$e^{-C_f \left(m_{b_{\varepsilon_n}^{NCO}} + n_{b_{\varepsilon_n}^{CO}}\right) (t_n^{(1)} - t_n^{(2)}) \langle R \rangle^{\gamma^+}} \ge e^{-C_f \left(m_{b_{\varepsilon_n}^{NCO}} + n_{b_{\varepsilon_n}^{CO}}\right) \left(\sum_{k \ge n+1} \Delta_k\right) \langle R \rangle^{\gamma^+}}$$

(see step 2 of proof of Proposition 4.2, Section 4 in [78]).

As we obtain exactly the same induction formulae as in [78], the asymptotic behaviour of the coefficients  $a_n$  is the same. Thus, by choosing an appropriate sequence  $(\Delta_n)_{n \in \mathbb{N}}$ , as done in [78], we can construct the expected exponential lower bound independently of time.

## Part II

# THE HYDRODYNAMICAL LIMITS OF THE BOLTZMANN EQUATION

## Chapter 3

# From many-body problems to physics of continua

Scientists work on establishing mathematical descriptions of physical phenomena in order to understand and foresee Nature. Different points of view, different scales, can be considered to translate physical dynamics into equations. However, even though the resulting theories look different in terms of equations and behaviour, they should model the same phenomena but at different scales. In this chapter we explore some physical and mathematical links between particles motion, gas dynamics and fluid mechanics in order to prove the mathematical coherence of the various physical modellings of Nature.

This chapter is far from being an exhaustive overview since it restrains itself to the framework of point particles in the framework of classical mechanics. It however motivates and introduces the important concept of hydrodynamical limits of the Boltzmann equation.

#### Contents

3.1 Part	articles, gases and fluids in physics		
3.1.1	Newton, Hamiltonian systems and the motion of particles 121		
3.1.2	Euler, continuous medium and fluid mechanics		
3.1.3	Maxwell, kinetic theory and gas dynamics		
3.2 Hilb	pert's sixth problem: the mathematical coherence of models124		
3.2.1	From micro to macro models: the law of large numbers 124		
3.2.2	The thermodynamical limit: the chaos assumption		
3.2.3	The hydrodynamical limits: the Knudsen number		
	3.2.3.1 A dimensionless reformulation of the Boltzmann equation 130		
	3.2.3.2 The compressible Euler limit		
	3.2.3.3 The asymptotic compressible Navier-Stokes limit 133		
	3.2.3.4 The incompressible hydrodynamical limits 135		
3.3 A b	rief description of the following chapters		

#### 3.1 Particles, gases and fluids in physics

The study of the motion of large amounts of particles is of great importance in physics and several laws and models had been derived. Whether one wants to describe the movement of nanoscopic particles under the influence of an electromagnetic field or to foresee the evolution of clouds of galaxies, it all comes down to finding a physical theory that agrees with observation.

In this section we briefly present the historical evolutions of the description of natural phenomena in order to understand their distinctive features as well as the links that can be made in between them. The framework presented here is the whole space but bounded domains can be considered with appropriate boundary conditions to be added to the equations.

#### 3.1.1 Newton, Hamiltonian systems and the motion of particles

In 1687, Isaac Newton published his *Principia Mathematica* [87] where he wrote the basis of what we nowadays call the classical mechanics. He described his celebrated second law of motion stating that the net force applied to a body produces an acceleration that is proportional to its mass.

With this tool we are able to describe the motion of N particles of mass m and radius r, evolving in  $\mathbb{R}^d$  and subject to an external potential V and a two-body interaction potential  $\Phi$ . Each of the particle is represented by its position and velocity  $(x_i, v_i)$  and the latter couple satisfies the following system of ordinary differential equations, where we put m in the definition of V and  $\Phi$ .

$$\frac{dx_i}{dt} = v_i,$$
  
$$\frac{dv_i}{dt} = -\nabla_x V(x_i) - \sum_{\substack{j=1\\j\neq i}}^N \nabla \Phi(x_i - x_j).$$

This system of nonlinear equations is known as the N body problem, which is very complicated as soon as N is greater than or equal to 3. This difficulty is due to the lack of a sufficient number of conserved quantities along the motion (which is called integrability of the system and imposes constraints on the motion itself) and was already noticed by Poincaré in [92]. However, in this specific case the force is the gradient of a potential and this dynamical system is thus Hamiltonian because with Hamiltonian operator

$$H = \frac{1}{2} \sum_{i=1}^{N} |v_i|^2 + \sum_{i=1}^{N} V(x_i) + \sum_{\substack{j=1\\ j \neq i}}^{N} \Phi(x_i - x_j).$$

If N is very large then one is only interested in an average behaviour of the group of particles. We therefore turn to the N-particle distribution function  $F_N$  in the phase space

$$\mathcal{D}_N = \left\{ (t, x_1, v_1, \dots, x_N, v_N) \in \mathbb{R}^+ \times \mathbb{R}^{2dN}, \, \forall i \neq j, \quad |x_i - x_j| > r \right\}.$$

In the case of Hamiltonian dynamical systems, the distribution  $F_N$  had been shown to satisfy the Liouville equation,

$$\partial_t F_N + \text{Liou}(F_N) = 0, \quad \text{in } \mathcal{D}_N,$$
(3.1.1)

where the Liouville operator reads

$$\text{Liou} = \sum_{i=1}^{N} \left( \frac{\partial H}{\partial v_i} \cdot \frac{\partial}{\partial x_i} - \frac{\partial H}{\partial x_i} \cdot \frac{\partial}{\partial v_i} \right).$$

In 1884, Josiah Gibbs [45] emphasized the importance of the Liouville equation as the fundamental equation of statistical mechanics.

#### 3.1.2 Euler, continuous medium and fluid mechanics

The study of the motion of fluids awaited a bit longer and a precise mathematical models appeared only with Leonhard Euler in his *Principes généraux du mouvement des fluids*, [43] published in 1755. Considering a fluid like a continuous medium rather than a group of individual molecules, Euler derived the first equations of fluid dynamics which bares his name and concerns the mass, the momentum and the energy of the fluid. Euler equations link the evolution of the density  $\rho$ , the mean velocity u, the inner pression p and the energy E of the fluid. They read

$$\partial_t \rho + \nabla_x \cdot (\rho u) = 0,$$
  

$$\rho \partial_t u + \rho u \cdot \nabla_x u + \nabla_x p = 0,$$
  

$$\partial_t E + \nabla_x \cdot (u (E + p)) = 0.$$
  
(3.1.2)

The first remark we can make is that those equations imply the conservation of total mass, momentum and energy of the fluid which is the minimum that classical mechanics ask for. However, Euler modelled a fluid without friction and thus cannot explain the viscosity phenomena we can observe.

The first step towards modelling of viscid fluids was made by Claude-Louis Navier in 1822 in his *Mémoire sur les lois du mouvement des fluides* [85], where he introduced a shear stress tensor to describe the inner force created by the motion of the fluid. Unfortunately, the way he tackled shear stress did not match real observations and the ultimate

improvements to the theory of viscid fluids were achieved by Adhémar Barré de Saint-Venant ([11], written in 1834 and published in 1843) and George Stokes ([103], 1845) by considering a shear stress tensor that is proportional to the gradient of the velocity. This lead to the so-called Navier-Stokes equations for newtonian fluids, where we denote the mean temperature of the fluid by  $\theta$ ,

$$\partial_t \rho + \nabla_x \cdot (\rho u) = 0,$$
  

$$\rho \partial_t u + \rho u \cdot \nabla_x u + \nabla_x p = \nu \Delta_x u,$$
  

$$\partial_t E + \nabla_x \cdot (u (E + p)) = \kappa \Delta_x \theta,$$
  
(3.1.3)

where  $\nu$  and  $\kappa$  are respectively the dynamic viscosity and the thermal conductivity of the fluid. The interested reader can find a derivation of those equations from the laws of physics as well as other types of fluids in [12].

**Remark 3.1.1** We can think of the Euler equations as the limit of the Navier-Stokes equations when the viscosity of the fluids goes to 0.

Finally, when the shear stress is very important, namely when  $\nu$  tends to infinity, one obtains the Stokes equations for viscid fluids.

$$\partial_t \rho + \nabla_x \cdot (\rho u) = 0,$$
  

$$\rho \partial_t u + \nabla_x p = \nu \Delta_x u,$$
  

$$\partial_t E = \kappa \Delta_x \theta,$$
  
(3.1.4)

#### 3.1.3 Maxwell, kinetic theory and gas dynamics

As previously emphasized, the many particles problem is really intricate and its complexity makes it almost impossible to use when one realises that a mole of gas contains more than  $6,02 \times 10^{23}$  particles or that the Milky Way is constituted of approximately  $10^{11}$  stars.

To overcome this issue, James Clerk Maxwell in 1867 [75] (in a weak formulation based on the physical observables of a system) and Ludwig Boltzmann in 1872 [18][17] developed the founding principles of kinetic theory. This theory proposes to take a statistical approach to model the dynamics of particles when they are so numerous that the individual behaviours are of little interest. Basically, one should try to understand the evolution of the distribution function f(t, x, v) of particles in the phase space as N tends to infinity; the quantity f(t, x, v)dxdv stands for the probability of finding a particles in [x, x + dx]with velocity in [v, v + dv] at time t. These thoughts lead to several models in statistical physics, depending on which interactions, which systems, are taken into account. In this part we will focus on the Boltzmann equation, already described in Chapter 1, which reads

$$\partial_t f + v \cdot \nabla_x f = Q(f, f). \tag{3.1.5}$$

# 3.2 Hilbert's sixth problem: the mathematical coherence of models

The different theories briefly described above have been established rather independently from each other and real observations validate them as being the relevant mathematical description of physics. However, they model, at different scales, the same underlying phenomenon that is the interaction between particles.

If we study a system of N particles then looking at their average dynamics when N is large should bring us to the kinetic theory framework for rarefied gases. Moreover, a strongly compressed gas becomes a fluid at very high pressure and therefore there should exist a link between kinetic equations and fluid equations.

At the International Congress of Mathematicians held in Paris in 1900, Hilbert emphasized the importance of mathematically deriving the coherence of all those physical models. More precisely, Hilbert's sixth problem aims at building up a unified description of mechanics, from microscopic atoms to macroscopic continuum. One would like to understand mathematically how macroscopic properties of fluids and gases, such as viscosity or irreversibility, evolving at an observation timescale  $T_{obs}$ , can arise from reversible microscopic dynamics, where the mean time between two consecutive collisions is of microscopic timescale  $T_{coll}$  (see Figure 3.1).

In this section, we formally study the possible convergences between the different physical models and give some of the existing results in the field that proved these convergences rigorously. We will restrict ourselves to the case when particles moves in a boundary free domain such as  $\mathbb{R}^d$  or the torus  $\mathbb{T}^d$ .

We shall give a more thorough study of the hydrodynamical limits as these will be the purpose of the next two chapters.

#### 3.2.1 From micro to macro models: the law of large numbers

In this section we briefly present some strategies to go directly from Hamiltonian systems to macroscopic dynamics. It deals with a problem that is completely transversal to our work on Boltzmann equation and its hydrodynamical limits and we will thus not go into details but we give a short description for the sake of completeness of our manuscript.



Figure 3.1: Transitions between the different level of description (from [98])

For rigorous proofs and a deeper insight of the theory, we refer the reader to [91], in the case of Euler limit, and to [42][97], for the Navier-Stokes limit. What follows can be found in [98] Chapter 1.

We would like to understand how microscopic properties generate macroscopic dynamics. We start with N particles with positions and velocities  $(x_i, v_i)$  which satisfy the Liouville equation (3.1.1). The physical observables of the system of N particles are the mass and momentum densities

$$M_N(t,x) = \frac{1}{N} \sum_{i=1}^N \delta(x - x_i(t)),$$
  

$$P_N(t,x) = \frac{1}{N} \sum_{i=1}^N v_i(t) \delta(x - x_i(t)).$$

where  $\delta$  is the Dirac measure at the origin.

The microscopic dynamics happen much faster and more localized than the fluid mechanics and one thus has to work at different time and space scales. We denote by (x, t)the microscopic variables and by  $(\tilde{x}, \tilde{t})$  the macroscopic ones. We have a ratio  $\varepsilon$  between space scale that we will make go to zero:

$$\tilde{x} = \varepsilon x.$$

The ratio between time scales will define the fluid dynamics towards which our Liouville equation converges. Indeed, if we define the typical density  $\rho = N/L^3$ , with L being a typical macroscopic lenght, we have the three possible outcomes

- 1.  $\rho \sim \varepsilon$  and  $\tilde{t} = \varepsilon t$ : the number of collision per particle is then finite and this is the Grad limit,
- 2.  $\rho \sim 1$  and  $\tilde{t} = \varepsilon t$ : the number of collision per particle is of order  $\varepsilon^{-1}$  and this is the Euler limit,
- 3.  $\rho \sim 1$  and  $\tilde{t} = \varepsilon^2 t$ : the number of collision per particle is of order  $\varepsilon^{-2}$  and this is the Diffusive limit.

In all the cases, the goal is to compute the equations satisfied, in a weak sense since we are dealing with probability measures, by  $M_N(\tilde{t}/\varepsilon, \tilde{x}/\varepsilon)$  and  $P_N(\tilde{t}/\varepsilon, \tilde{x}/\varepsilon)$  and to compute the limiting equations as  $\varepsilon$  goes to zero. We are looking at probability measures so the convergence has to be understood as a convergence in the sense of the law of large numbers as N tends to infinity, with respect to the density function  $F_N$ .

Some results have been rigorously proven in some special settings where we have ergodicity of the system. We refer to the references given at the begin of this section for more details.

#### 3.2.2 The thermodynamical limit: the chaos assumption

As the number N of particles in a system becomes very large the N-body problem is too intricate to offer an interesting description of how the system behaves. Moreover, one is more interested by the global evolution of the system than the actual motion of one particular particle in the case where they are indistinguishable. A statistical approach is preferable and the Liouville equation is easier to handle, see Section 3.1.1.

The derivation from microscopic dynamics to mesoscopic scales is rather hard even for short range interaction potentials. Indeed, it depends exactly on the positions and velocities of particles when we would like to only care about the probability distribution of the latters. In other words, there is no global description of the interacting forces inside the system. Therefore, some assumptions have to be made in order to ensure the statistical stability of the mesoscopic dynamics in the limit  $N \to \infty$ .

We give here a brief and formal derivation of the Boltzmann equation from the laws on Newton. Most of this section follows closely [30] and [44].

The Liouville equation (3.1.1) in the case of a sole two-body interaction potential  $\Phi_N$  reads, when the diameter of each particle is denoted by r,

$$\partial_t F_N + \sum_{i=1}^N v_i \cdot \nabla_{x_i} F_N - \sum_{i=1}^N \sum_{\substack{j=1\\j\neq i}}^N \nabla_x \Phi_N(x_i - x_j) \cdot \nabla_{v_i} F_N = 0,$$

in the phase space

$$\mathcal{D}_N = \left\{ (t, x_1, v_1, \dots, x_N, v_N) \in \mathbb{R}^+ \times \mathbb{R}^{2dN}, \ \forall i \neq j, \quad |x_i - x_j| > r \right\}.$$

We add on  $F_N$  the assumption that particles are indistinguishable, which translates into

$$\forall \sigma \in \mathfrak{S}_N, \quad f(t, x_1, v_1, \dots, x_N, v_N) = f(t, x_{\sigma(1)}, v_{\sigma(1)}, \dots, x_{\sigma(N)}, v_{\sigma(N)}),$$

where  $\mathfrak{S}_N$  denotes the group of permutations of the set  $\{1, \ldots, N\}$ .

We want to extract the average behaviour of a particle, that is to say the first marginal associated to  $F_N$ , which we denote by  $f_N^{(1)}(t, x_1, v_1)$ .

The thermodynamical limit is the resulting equation satisfied by  $f_N^{(1)}$  when we let N go to infinity. Looking at the Liouville equation, the main difficulty will be to understand the term  $\nabla_x \Phi_N$  in the limit of infinitely many particles. However, as N goes to infinity, the energy of the system has to remain bounded and we thus have to assume that the energy of each interaction  $via \Phi_N$  is small. We therefore need to rescale the potential  $\Phi_N$  and we present the two ways of doing it.

#### The mean-field limit:

In that case we consider that the range of the interaction stays macroscopic but that its amplitude decreases like 1/N. This way we have that  $\Phi_N = \Phi/N$  for a given macroscopic potential  $\Phi$ .

In that case, integrating the Liouville equation against  $(x_2, v_2, \ldots, x_N, v_N)$  to obtain the equation of the first marginal and taking the limit as N goes to infinity yields

$$\partial_t f + v \cdot \nabla_x f + F \cdot \nabla_v f = 0,$$

where  $f = \lim_{N \to \infty} f_N^{(1)}$  and

$$F(t,x) = -\nabla_x \left( \Phi * \int_{\mathbb{R}^d} f(t,x,v) dv \right).$$

This strategy of the mean-field limit generates a lot of interestic mathematical studies. As an example we mention the Vlasov-Poisson equation used to describe plasmas with Coulomb interaction potential (even if the rigorous derivation remains an open problem)

$$\Phi(x) = \frac{q^2}{4\pi\varepsilon_0 |x|},$$

where q is the electric charge of a particle and  $\varepsilon_0$  is the vacuum permittivity.

However, we will not deal with this type of rescaling for  $\Phi_N$  and we refer the interested reader to a review by Golse [47].

#### Collisional dynamics:

The rescaling of  $\Phi_N$  we are interested in is the one where interactions become localized in the space variable, acting like collisions between particles. Basically, we suppose that the strenght of the interaction  $\Phi_N$  stays O(1) but its range is very small. The Liouville equation associated to this problem reads

$$\partial_t F_N + \sum_{i=1}^N v_i \cdot \nabla_{x_i} F_N - \sum_{i=1}^N \sum_{\substack{j=1\\j \neq i}}^N \frac{1}{l} \nabla_x \Phi\left(\frac{x_i - x_j}{l}\right) \cdot \nabla_{v_i} F_N = 0, \qquad (3.2.1)$$

where l is the range of interaction of  $\Phi_N$  and is microscopic. Moreover, this equation has to be satisfied in the following domain

$$\mathcal{D}_N = \left\{ (t, x_1, v_1, \dots, x_N, v_N) \in \mathbb{R}^+ \times \mathbb{R}^{2dN}, \, \forall i \neq j, \quad |x_i - x_j| > l \right\}.$$

If we integrate the Liouville equation against  $(x_2, v_2, \ldots, x_N, v_N)$  we clearly see that, compared to the mean-field limit case where  $\Phi_N$  implies that each particle feels the average force generated by all the other particles,  $f_N^{(1)}$  will depend on  $f_N^{(2)}$  via the term  $l^{-1}\nabla_x \Phi(l^{-1}(x_1 - x_2))$ . We thus need to compute the equation satisfied by the second marginal which depends, by the same considerations, of  $f_N^{(3)}$ . By induction we construct a hierarchy of N equations from the first marginal to the  $N^{th}$  one  $(F_N$  itself). This system of equations is called the BBGKY hierarchy, from Bogoliubov [16], Born and Green [19], Kirkwood [62][63] and Yvon [115] (see also [30][44][96]).

A requirement to derive the BBGKY hierarchy is to define boundary conditions on  $\partial \mathcal{D}_N$ , which is the set where at least two particles are in contact, in order to integrate by parts in the integrated Liouville equation. We suppose that the collision between particles  $(x_i, v_i)$  and  $(x_j, v_j)$  are elastic collisions, that is

- 1. they are localized in time and space so the positions of the particles remain unchanged and the particles collide at a given time,
- 2. they are perfectly elastic which means that the momentum and the energy are preserved: if we denote  $v'_i$  and  $v'_i$  the outcoming velocities we have

$$\begin{cases} & v'_i + v'_j = v_i + v_j \\ & |v'_i|^2 + |v'_j|^2 = |v_i|^2 + |v_j|^2 \end{cases}$$

Of course, for this elastic collisions to define boundary conditions and physical dynamics, we have to put aside the problematic case when three or more particles collide at the same time or when infinitely many collisions happen in a finite time. Fortunately, the set of initial data leading to such outcome is of Lebesgue measure zero (see Proposition 2.1.1 in [44] for the hard sphere case).

We integrate the Liouville equation (3.2.1) against the last N - s coordinates and take into account the boundary conditions. The complete computations can be found in [44] Chapter 4 thanks to truncated marginals. To simplify here we just assume that it holds true for marginals. The BBGKY hierarchy reads, at least in a weak sense,

$$\partial_t f_N^{(s)} + \sum_{i=1}^N v_i \cdot \nabla_{x_i} f_N^{(s)} - \frac{1}{l} \sum_{\substack{i,j=1\\j \neq i}}^N \nabla_x \Phi\left(\frac{x_i - x_j}{l}\right) \cdot \nabla_{v_i} f_N^{(s)} = \sum_{m=1}^{N-s} m\binom{N-s}{m} Q_l(f_N^{(s+m)}),$$

for all  $s \in \{1 \dots N\}$ . Here  $Q_l$  is an operator encoding the boundary conditions. Namely, it involves the integral over the N - s spherical particles potentially colliding with the free s particles. Therefore, the particles having a diameter l, one can expect a uniform convergence as N tends to infinity if our gas satisfies the Boltzmann-Grad scaling:

$$\lim_{N \to +\infty} N l^{d-1} = O(1).$$

The thermodynamical limit consists in understanding the BBGKY hierarchy when N goes to infinity in the Boltzmann-Grad scaling setting.

As mentioned before, the marginals can only be understood thanks to higher order marginals. To obtain the Boltzmann equation, one has to prove that the limit  $F_N$  is of a tensor product form

$$\lim_{N \to +\infty} F_N = \bigotimes_{n \in \mathbb{N}} f,$$

where f is a density function that satisfies the Boltzmann equation. This tensor product has to be understood in the sense of marginal

$$\forall n \in \mathbb{N}, \quad \lim_{N \to +\infty} f_N^{(s)} = \underbrace{f \otimes \cdots \otimes f}_{s \text{ times}}.$$

Such a property is called the chaos assumption and it means that particles asymptotically behave independently of each other, in a weak sense.

An important feature that does not happen in the mean-field case is the fact that the collisional dynamics framework defines a past and a future for the system. Indeed, when two particles bounce against each other, they are no longer independent of each other since the laws of elasticity define their velocities after the collision. The chaos assumption thus implies a choice for the arrow of time.

Deriving Boltzmann equation rigorously from the Liouville equation, and the BBGKY

hierarchy, still requires a lot of studies. The first result was due to Lanford [65] where he gave a proof about existence of solution to the BBGKY hierarchy, their convergence and the propagation of the chaos assumption in time. Recently, [44] and [96] filled in the missing details in Lanford's proof in the case of hard spheres (billiard balls with  $\Phi = 0$ ) and short range potentials ( $\Phi$  with compact support and unbounded near the origin).

Unfortunately, up to now, the proofs hold for very short time, smaller than the mean free time between two consecutive collisions.

Let us briefly mention that another approach has been proposed by Kac [60] to derive the spatially homogeneous Boltzmann equation from a stochastic process underlying the dynamics of particles instead of using Newton's law. In that case, dynamics of velocities are viewed as stochastic processes with jumps standing for collisions. This strategy has been useful to obtain results about the Boltzmann equation, such as insights of Cercignani's conjecture on the entropy decay for entropy-entropy production methods for instance. A recent state of the art about the subject can be found in [76].

#### 3.2.3 The hydrodynamical limits: the Knudsen number

Fluids dynamics are determined by some properties of fluids such as their compressibility and their viscosity. These parameters are expressed in terms of dimensionless coefficients that encode the physical properties of the fluid. For instance, the Mach number  $Ma = u_0/c^*$ , where  $u_0$  is the bulk velocity of the flow and  $c_*$  is the speed of sound in the medium, determines the compressibility of the fluids and the Reynolds number Re informs about the viscosity of the fluids. The smaller these numbers are, the less compressible or viscous the fluid is.

At the mesoscopic scale of the Boltzmann equation, only microscopic features are governing the dynamics but one expects that the macroscopic properties arise from such dynamics but at different time and space scales. What follows is a gathering of results, thoughts and suggestions made essentially in [46][111][98].

We will not discuss the case of bounded domain as this is more intricate and notably fewer results has been proven. We refer the interested reader to Section 4.4 of [98] or [9] for the particular case of incompressible Euler limit. Note that some results presented here cannot hold true with boundaries because of the existence of a boundary layer phenomenon, as noticed in [73] for compressible Euler limit for which the Prandtl layer occurs.

#### 3.2.3.1 A dimensionless reformulation of the Boltzmann equation

The macroscopic dynamics are visible at a much bigger scale than the microscopic interactions between particles. We therefore consider a macroscopic length scale  $l_0$ , which can be the size of the domain where the flow evolves, and an observation time scale  $t_0$ , which can be seen as  $T_{obs}$  in Figure 3.1. By choosing a reference temperature  $\theta_0$  we define a thermal speed  $c^*$  associated to this temperature (see Section 2.2 of [98]).  $c_*$  would be the speed of sound in the case of a monoatomic gas. We set new nondimensional variables

$$t' = \frac{t}{t_0}, \quad x' = \frac{x}{l_0}, \quad v' = \frac{v}{c^*},$$

as well as a nondimensional distribution function

$$f'(t', x', v') = \frac{l_0^3 c_0^3}{N_0} f(t, x, v),$$

where  $N_0$  is the average number of particles in a volume  $l_0^3$ ;  $\rho_0 = N_0/l_0^3$  is therefore the mean macroscopic density of the gas.

As noticed in [46] or [98], the Boltzmann operatore Q is expressed in density per unit of time and therefore defines a new microscopic time scale  $\tau_0$  in the following sense

$$\int_{\mathbb{R}^d \times \mathbb{R}^d \times \mathbb{S}^{d-1}} B(v, v_*, \sigma) M_{(\rho_0, 0, \theta_0)}(v) M_{(\rho_0, 0, \theta_0)}(v_*) \, d\sigma dv_* dv = \frac{\rho_0}{\tau_0}.$$

 $\tau_0$  is the mean free time between two consecutive collisions of a particle at equilibrium  $M_{(\rho_0,0,\theta_0)}$  and thus define the mean free path

$$\lambda_0 = c_* \tau_0.$$

Finally we take a rescaled collisional kernel  $B'(v', v'_*, \sigma) = \rho_0 \tau_0 B(v, v_*, \sigma)$  and we obtain a nondimensional form of the Boltzmann equation, where we dropped the prime notations:

$$\operatorname{Ma}\partial_t f + v \cdot \nabla_x f = \frac{1}{\operatorname{Kn}} Q(f, f),$$
(3.2.2)

where Ma is the Mach number of the flow and Kn is the Knudsen number, which is the inverse of the average number of collisions for each particle per unit of time. One can choose different length and time scales to study fluctuations around a reference flow instead. In that case one obtains the Strouhal number instead of the Mach number and this choice leads to different hydrodynamical models (see [98]).

Considering fluids as gases where particles are in contact suggests that the Knudsen number governs the convergence from Boltzmann equation to fluid equations. The hydrodynamical limits study the evolution of solutions to the rescaled Boltzmann equation (3.2.2) as Kn goes to zero. Moreover, fluid dynamics essentially amount to the laws of conservation of mass, momentum and energy. We thus expect to derive them only using the conservation laws and the entropy decrease fulfilled by the Boltzmann equation.

#### 3.2.3.2 The compressible Euler limit

For now on we denote the Knudsen number by  $\varepsilon$ . We look at the family of distributions  $(f_{\varepsilon})_{\varepsilon>0}$  that satisfy (3.2.2) for all  $\varepsilon$  in (0, 1], that is

$$\varepsilon \mathrm{Ma} \partial_t f_\varepsilon + \varepsilon v \cdot \nabla_x f_\varepsilon = Q(f_\varepsilon, f_\varepsilon).$$

Therefore, at least formally, if  $f_{\varepsilon}$  tends to f as  $\varepsilon$  goes to zero we have that Q(f, f) = 0which leads to (see Section 1.1.2.4)

$$f(t, x, v) = M_{\rho, u, \theta}(t, x, v) = \frac{\rho(t, x)}{(2\pi\theta(t, x))^{d/2}} \exp\left(-\frac{|v - u(t, x)|^2}{2\theta(t, x)}\right).$$

Thanks to Section 1.1.2.4 we have the conservation of mass, momentum and energy for all the  $f_{\varepsilon}$  and so this gives in the limit

$$\operatorname{Ma} \int_{\mathbb{R}^d} M_{\rho,u,\theta} \, dv + \nabla_x \cdot \int_{\mathbb{R}^d} v M_{\rho,u,\theta} \, dv = 0$$
  
$$\operatorname{Ma} \int_{\mathbb{R}^d} v M_{\rho,u,\theta} \, dv + \nabla_x \cdot \int_{\mathbb{R}^d} v \otimes v M_{\rho,u,\theta} \, dv = 0$$
  
$$\operatorname{Ma} \int_{\mathbb{R}^d} \frac{1}{2} |v|^2 M_{\rho,u,\theta} \, dv + \nabla_x \cdot \int_{\mathbb{R}^d} \frac{1}{2} |v|^2 v M_{\rho,u,\theta} \, dv = 0$$

which is easily computed into

$$\begin{aligned} \operatorname{Ma}\partial_{t}\rho + \nabla_{x} \cdot (\rho u) &= 0, \\ \operatorname{Ma}\partial_{t} (\rho u) + \nabla_{x} \cdot (\rho u \otimes u) + \nabla_{x} (\rho \theta) &= 0, \\ \operatorname{Ma}\partial_{t} \left( \rho \left( \frac{1}{2} |u|^{2} + \frac{d}{2} \theta \right) \right) + \nabla_{x} \cdot \left( u\rho \left( \frac{1}{2} |u|^{2} + \frac{d+2}{2} \theta \right) \right) &= 0. \end{aligned}$$

These equations are the compressible Euler equations for a perfect monoatomic gas where the pressure is  $p = \rho \theta$ , the thermal energy is  $\theta/2$  per degree of freedom and the internal energy is thus  $d\theta/2$ . We remark here that the H-theorem (see Section 1.1.2.4) leads to

$$\operatorname{Ma}\partial_t \left( \rho \log \frac{\rho}{\theta^{d/2}} \right) + \nabla_x \cdot \left( \rho u \log \frac{\rho}{\theta^{d/2}} \right) \leqslant 0,$$

which is the characterization of physically relevant solutions to the Euler system, known as the Lax admissibility condition.

There are few rigorous results about the derivations above and we refer the reader to Section 6.2 of [98] for a bibliography on the subject. Most of the existing results are valid

as long as the compressible Euler theory gives smooth solutions, in other words the proofs fail as soon as a singularity appears in Euler equations, which is a common property of the latter equations (see [101]).

Let us mention here the article by Caflisch [25], extended by Lachowicz [64] to more general initial data by dealing with the problem of initial layer. These articles construct solutions to the Boltzmann equation close to a local Maxwellian  $M_{(\rho,u,\theta)}$  the parameters of which satisfy the Euler equations. Their general strategy was to look for solutions to the Boltzmann equation of the form  $f_{\varepsilon} = M_{(\rho,u,\theta)} + \varepsilon g_{\varepsilon}$  which proved itself to be a powerful method that we shall discuss deeper later.

#### 3.2.3.3 The asymptotic compressible Navier-Stokes limit

Hydrodynamical limits study the asymptotic of the observable quantities of solutions  $f_{\varepsilon}$  to (3.2.2) as  $\varepsilon$  goes to zero. As seen in the section above, if we expand  $f_{\varepsilon}$  in terms of  $\varepsilon$ , the zeroth order term has to be an Eulerian maxwellian  $M_{(\rho,u,\theta)}$ . Moreover, Remark 3.1.1 seems to consider Navier-Stokes system as a fluctuation around a global equilibrium of Euler system so the natural question is wether an expansion of  $f_{\varepsilon}$  gives us the compressible Navier-Stokes equation.

The Hilbert's expansion - or its modified version the Chapman-Enskog's expansion where the variables of g are observables of  $f_{\varepsilon}$  and their derivatives - is a formal expansion of  $f_{\varepsilon}$  around the Knudsen number

$$f_{\varepsilon}(t, x, v) = \sum_{i=0}^{+\infty} \varepsilon^{i} g_{i}(t, x, v).$$

The goal is now to plug this expression inside the nondimensional form of the Boltzmann equation, to obtain a hierarchy of partial differential equations and to solve them to obtain solutions  $f_{\varepsilon}$  of that specific form. The Hilbert's expansion is formal but one can look at a finite expansion with a remainder term that is hoped to be small enough. Mathematical properties of these kind of expansions are detailed in the works of Grad [48][49].

We saw before that  $g_0 = M_{(\rho,u,\theta)}$  with  $(\rho, u, \theta)$  satisfying the Euler equations. The equation one gets at zeroth order is

$$\mathrm{Ma}\partial_t g_0 + v \cdot \nabla_x g_0 = L_{g_0}(g_1),$$

where  $L_{g_0}$  is the linearization of the Boltzmann operator around the local Maxwellian  $g_0$ . The properties of the linear operator are therefore required but we will not go into them, we refer the reader to [8] for the ones needed in this derivation and to Chapter 4 for a general study. The important point is that we get a form for  $g_1$  which is

$$g_1(t, x, v) = L_{g_0}^{-1} (\operatorname{Ma}\partial_t g_0 + v \cdot \nabla_x g_0) + \varphi_1(t, x, v), \qquad (3.2.3)$$

where  $\rho_1$ ,  $u_1$  and  $\theta_1$  are functions of t and x and  $\varphi_1$  is in Ker  $(L_{g_0})$ .

If we define

$$A(v) = v \otimes v - \frac{1}{d} |v|^2 I_d \quad \text{and} \quad B(v) = \frac{1}{2} v \left[ |v|^2 - (d+2) \right], \tag{3.2.4}$$

then direct computations with  $g_0 = M_{(\rho,u,\theta)}$  and the Euler equations give us

$$\operatorname{Ma}\partial_t g_0 + v \cdot \nabla_x g_0 = -\left(\frac{1}{2}A(V): D(u) + B(V) \cdot \frac{\nabla_x \theta}{\sqrt{\theta}}\right)g_0, \qquad (3.2.5)$$

where

$$V = \frac{v - u}{\sqrt{\theta}} \quad \text{and} \quad D(u) = \frac{1}{2} \left( \nabla_x u + (\nabla_x u)^T \right) - \frac{1}{d} (\nabla_x \cdot u) I_d.$$

We notice that  $A(V)g_0$  and  $B(V)g_0$  are in Im  $(L_{g_0})$ . Moreover, see [8][46], there exist functions  $\alpha, \beta : \mathbb{R}^+ \longrightarrow \mathbb{R}^+$  such that

$$L_{g_0}^{-1}(A(V)g_0) = \rho^{-1}\theta^{\gamma/2}\alpha(|V|)A(V)g_0 \quad \text{and} \quad L_{g_0}^{-1}(Bg_0) = \rho^{-1}\theta^{\gamma/2}\beta(|V|)B(V)g_0.$$
(3.2.6)

Combining the latter equalities with (3.2.3) and (3.2.5) yields a precise form for  $g_1$ ,

$$g_1 = -\rho^{-1}\theta^{\gamma/2} \left(\frac{1}{2}\alpha\left(|V|\right)A(V): D(u) + \beta\left(,|V|\right)B(V) \cdot \frac{\nabla_x\theta}{\sqrt{\theta}}\right)g_0 + \varphi_1.$$

It remains to compute  $\varphi_1$  - i.e. its observable quantities  $(\rho_1, u_1, \theta_1)$  - so we look at the expansion of the Boltzmann equation at first order in  $\varepsilon$ :

$$\operatorname{Ma}\partial_t g_1 + v \cdot \nabla_x g_1 = L_{g_0}(g_2) + Q(g_1, g_1),$$

for which we still have conservation of mass momentum and energy

$$\int_{\mathbb{R}^d} \psi(v) \left[ \mathrm{Ma} \partial_t g_1 + v \cdot \nabla_x g_1 \right] \, dv = 0,$$

with  $\psi(v) = 1, v, |v|^2$ .

This system of conservation laws together with the equations satisfied by  $(\rho, u, \theta)$  gives that

$$\rho_{\varepsilon} = \rho + \varepsilon \rho_1, \quad u_{\varepsilon} = u + \varepsilon u_1, \quad \theta_{\varepsilon} = \theta + \varepsilon \theta_1$$

satisfy

$$\begin{split} \operatorname{Ma}\partial_{t}\rho_{\varepsilon} + \nabla_{x} \cdot (\rho_{\varepsilon}u_{\varepsilon}) &= 0, \\ \operatorname{Ma}\partial_{t}\left(\rho_{\varepsilon}u_{\varepsilon}\right) + \nabla_{x} \cdot (\rho_{\varepsilon}u_{\varepsilon} \otimes u_{\varepsilon}) + \nabla_{x}\left(\rho_{\varepsilon}\theta_{\varepsilon}\right) &= \varepsilon \nabla_{x}\left[\nu(\rho_{\varepsilon},\theta_{\varepsilon})D(u_{\varepsilon})\right], \\ \operatorname{Ma}\partial_{t}E_{\varepsilon} + \nabla_{x} \cdot \left(u_{\varepsilon}(E_{\varepsilon} + \rho_{\varepsilon}\theta_{\varepsilon})\right) &= \frac{\varepsilon}{2}\nu(\rho_{\varepsilon},\theta_{\varepsilon})D(u_{\varepsilon}) : D(u_{\varepsilon}) + \varepsilon \nabla_{x} \cdot \left[\kappa(\rho_{\varepsilon},\theta_{\varepsilon})\nabla_{x}\theta_{\varepsilon}\right], \end{split}$$

where the internal energy is the one of a monoatomic gas

$$E_{\varepsilon} = \rho_{\varepsilon} \left( \frac{1}{2} \left| u_{\varepsilon} \right|^2 + \frac{d}{2} \theta_{\varepsilon} \right),$$

and the dynamic viscosity and the thermal conductivity are given by

$$\nu(\rho_{\varepsilon}, \theta_{\varepsilon}) = \rho_{\varepsilon}^{-1} \theta_{\varepsilon}^{\gamma/2} \left\langle \alpha\left(|V|\right) A(V), A(V) \right\rangle_{L^{2}_{v}(g_{0})}, \tag{3.2.7}$$

$$\kappa(\rho_{\varepsilon},\theta_{\varepsilon}) = \rho_{\varepsilon}^{-1} \theta_{\varepsilon}^{\gamma/2} \left\langle \alpha\left(|V|\right) A(V), A(V) \right\rangle_{L^{2}_{v}(q_{0})}.$$
(3.2.8)

The observables of the system  $(\rho_{\varepsilon}, u_{\varepsilon}, \theta_{\varepsilon})$  satisfy the Navier-Stokes equation with dissipation terms of order  $\varepsilon$ . Of course, in order to make this rigorous, one has to close the whole system with  $g_2$  and prove that this remainder term is of order  $\varepsilon^2$ . This can be found formally in [8] and a similar result that the one derived by Caffisch in the case of Euler equation has been proven by De Masi, Esposito and Lebowitz [34], for a Navier-Stokes maxwellian with constant mass and temperature. The interesting point is that the macroscopic viscosity and conductivity arose from microscopic phenomenon described by the linear part of the Boltzmann equation.

To conclude with the Hilbert expansion, one can obtain next orders asymptotic hydrodynamical limits. However, the second order yields the Burnett equations which turned out to be irrelevant physically. For further discussions on these schemes see [48] and [111].

#### 3.2.3.4 The incompressible hydrodynamical limits

The compressible Navier-Stokes equation has been recovered from Boltzmann equation only in an asymptotic regime where the dissipativity tends to 0. One can wonder if we can actually obtain the compressible Navier-Stokes equations with dissipative term of order 1.

Unfortunately, this is impossible. The viscosity of a fluid is measured thanks to its Reynolds number Re and one has the von Karman relation

$$\operatorname{Re} = \frac{\operatorname{Ma}}{\operatorname{Kn}}$$

Considering the Knudsen number to go to zero therefore implies that either the Reynolds number explodes or that the Mach number goes to zero as well. Therefore, hydrodynamical limits with finite viscosity must be incompressible, since the Mach number measures compressibility.

Fluids equations has been derived from the physics of continua, considering that in arbitrarily small regions of space the fluid is at equilibrium. Euler or Navier-Stokes equations should thus arise from gas dynamics as a perturbation around a global equilibrium, say  $M_0 = M_{(1,0,1)}$  without loss of generality. We therefore look at solutions to the Boltzmann equation (3.2.2) of the form

$$f_{\varepsilon} = M_0 + \delta_{\varepsilon} h_{\varepsilon},$$

where  $\varepsilon$  still stands for the Knudsen number. We expect that different orders of perturbations  $\delta_{\varepsilon}$  lead to different hydrodynamical regimes. Indeed, properties of the Boltzmann operator, see Chapter 1, yield

$$\operatorname{Ma}\partial_t h_{\varepsilon} + v \cdot \nabla_x h_{\varepsilon} = \frac{1}{\varepsilon} L_{M_0}(h_{\varepsilon}) + \frac{\delta_{\varepsilon}}{\varepsilon} Q(h_{\varepsilon}, h_{\varepsilon}), \qquad (3.2.9)$$

and the term  $\delta_{\varepsilon}$  emphasized the role the linearity of the equation or, on the contrary, the bilinear part depending on the order of magnitude of  $\delta_{\varepsilon}$  compared to  $\varepsilon$ .

In any case, since  $\delta_{\varepsilon}$  tends to zero as  $\varepsilon$  go to zero we have that if  $h_{\varepsilon} \to h$  then formally taking the limit as  $\varepsilon \to 0$  in (3.2.9) gives

$$L_{M_0}(h) = 0$$

which means that h is a fluctuation of a maxwellian, i.e. of the following form:

$$h(t, x, v) = \left[\rho(t, x) + u(t, x) \cdot v + \left(\frac{|v|^2 - d}{2}\right)\theta(t, x)\right] M_0.$$
(3.2.10)

#### The acoustic limit.

The Mach number has to be related to the Knudsen number if one hopes to obtain a viscuous hydrodynamical limit. We briefly mention that in the case where Ma = 1 the Reynolds number tends to infinity whereas the fluid stays compressible. Therefore, one wants to recover the acoustic equations (propagation of waves in the medium) in the limit  $\varepsilon$  to zero. This is indeed the case as long as  $\delta_{\varepsilon} = \varepsilon^{\beta}$  with  $\beta > 1/2$ , see [6][7].

#### The fluid limits.

In the case Ma tends to zero as  $\varepsilon$  goes to zero, taking the limit in the local conservation of mass and momentum associated to (3.2.9) gives

$$abla_x \cdot \langle v, h \rangle_{L^2_v} = 0 \quad \text{and} \quad \nabla_x \cdot \langle v \otimes v, h \rangle_{L^2_v} = 0,$$

which stands for the incompressibility criteria and the Boussinesq approximation

$$\nabla_x \cdot u = 0$$
 and  $\nabla_x (\rho + \theta) = 0.$ 

The Boussinesq relation is an approximation for fluids where density evolutions can be neglected. As a result, in such fluids we can neglect the effects of inner inertia which makes sound waves impossible to develop, which is in adequation with our previous paragraph.

In order to derive the hydrodynamical limit we come back to the conservation laws satisfied by (3.2.9). As an example we consider the conservation of momentum and energy (the conservation of mass being straightforward at least formally) which reads

$$\partial_t \langle v, h_\varepsilon \rangle_{L^2_v} + \frac{1}{\mathrm{Ma}} \nabla_x \cdot \langle v \otimes v, h_\varepsilon \rangle_{L^2_v} = 0,$$

and can be written in terms of A (see (3.2.4)),

$$\partial_t \langle v, h_\varepsilon \rangle_{L^2_v} + \frac{1}{\mathrm{Ma}} \nabla_x \cdot \langle A(v), h_\varepsilon \rangle_{L^2_v} + \nabla_x \left( \frac{1}{\mathrm{Ma}} \langle \frac{1}{d} |v|^2, h_\varepsilon \rangle_{L^2_v} \right) = 0.$$

The last term on the left-hand side seems to explode as  $\varepsilon$  and Ma tend to 0. However, it is a gradient so if we integrate the equality against a divergence free test function this term disappears. For now on, we consider the computations in a weak sense, integrated against divergence free functions. This way, we can only recover solutions to incompressibles fluid equations in the Leray weak sense [66].

The idea is to use the self-adjointness property of  $L_{M_0}$  and express  $L_{M_0}(h_{\varepsilon})$  thanks to (3.2.9). This gives

$$\lim_{\varepsilon \to 0} \frac{1}{\mathrm{Ma}} \langle A(v), h_{\varepsilon} \rangle = \lim_{\varepsilon \to 0} \left[ \frac{\varepsilon}{\mathrm{Ma}} \langle v \otimes L_{M_0}^{-1} \left( A(v) \right), h_{\varepsilon} \rangle - \frac{\delta_{\varepsilon}}{\mathrm{Ma}} \langle L_{M_0}^{-1} \left( A(v) \right), Q(h_{\varepsilon}, h_{\varepsilon}) \rangle \right]$$

The temperature equation is handled the same with the operator B instead of A and to conclude we have to use their properties (3.2.6).

Here we see the importance of the order of magnitude of Ma and  $\delta_{\varepsilon}$  compared to the Knudsen number  $\varepsilon$ . The results are the following:

- Ma =  $\delta_{\varepsilon} = \varepsilon^q$  with 0 < q < 1 leads to incompressible Euler equations since collisions are faster than the dissipation,
- Ma =  $\varepsilon^q$  with 0 < q < 1 and  $\delta_{\varepsilon} = \varepsilon^p$  with p > 1 leads to incompressible Stokes equation since this time the non-linearity vanishes in the limit,
- Ma =  $\delta_{\varepsilon} = \varepsilon$  leads to incompressible Navier-Stokes equations because dissipation and collisions occur at the same scale.

A formal proof of these statements is given in [8][46] and a survey of the existing rigorous results can be found in [111]. As noticed before, the physical quantities associated to  $h_{\varepsilon}$  converges to weak solutions to the fluid models (Leray sense for Navier-Stokes and dissipative sense for Euler). Rigorous proofs of those derivations require compactness arguments. A strategy that proved itself to be useful is to use the framework of renormalized solutions derived by DiPerna and Lions [38], like in [5].

#### **3.3** A brief description of the following chapters

The problem of hydrodynamical limit is also closely related to the Cauchy problem associated to the linearized Boltzmann equation (3.2.9). Indeed, one would like to have a local or global existence in a perturbative setting with some kind of uniformity in the Knudsen number. Such results exist and we refer to Chapter 4 for a state of the art of the matter.

Furthermore, the methods described above do not give explicit rate of convergence or a constructive way of deriving the limit. Such concerns can be dealt with thanks to the study of the explicit form of the linear semigroup in the Fourier space. This point of view was initiated by Ellis and Pinsky [39] and further developed by Bardos and Ukai [10] to obtain uniform convergence in time. The issue of the initial layer at time t = 0 arises from these studies and we refer to Chapter 4 for a deeper understanding of the phenomenon.

The next two chapters deal with the incompressible Navier-Stokes hydrodynamical limit on the torus.

In Chapter 4 we build a Cauchy theory in Sobolev spaces with a maxwellian weight as well as exponential convergence to equilibrium. Then, extending the strategy of [10] to the torus, we prove strong convergence of solutions to the Boltzmann equation towards incompressible Navier-Stokes Leray solutions. Moreover, we give a constructive proof with explicit rates of convergence.

Chapter 5 aims at getting rid of the maxwellian weight as well as derivatives to build a Cauchy theory in Sobolev with polynomial weight. This uses a recent extension theorem for semigroups result by Gualdani, Mischler and Mouhot [51]. Such uniform results should allow to derive solutions to Navier-Stokes equations in large spaces, in particular  $L_x^{\infty}$ .

## Chapter 4

## From Boltzmann to incompressible Navier-Stokes on the torus in Sobolev spaces

We investigate the Boltzmann Equation, depending on the Knudsen number, in the Navier-Stokes perturbative setting on the torus. Using hypocoercivity, we derive a new proof of existence and exponential decay for the solutions of this perturbed equation, with explicit regularity bounds and rates of convergence. These results are uniform in the Knudsen number and thus allow us to obtain a strong derivation of the incompressible Navier-Stokes equations as the Knudsen number tends to 0. Moreover, our method shows that the smaller the Knudsen number, the less control on the v-derivatives of the initial perturbation is needed to have existence and decay. It is also used to deal with other kinetic models. Finally, we show that the study of this hydrodynamical limit is rather different on the torus than the already proven convergences in the whole space as it requires averaging in time, unless the initial layer conditions are satisfied.

#### Contents

4.1	Intr	oduction	1
	4.1.1	The pro	blem and its motivations
	4.1.2	Notation	ns
	4.1.3	Our stra	ategy and results
	4.1.4	Compar	ison with existing results
	4.1.5	Organiz	ation of the chapter $\ldots \ldots 148$
<b>4.2</b>	Mai	n results	3
	4.2.1	Hypocoe	ercivity assumptions
		4.2.1.1	Assumptions on the linear operator $L$
		4.2.1.2	Assumptions on the second order term $\Gamma$ 151
	4.2.2	Stateme	nt of the Theorems $\ldots \ldots 151$
		4.2.2.1	Uniform result for the linear Boltzmann equation 151
		4.2.2.2	Uniform perturbative result for the Boltzmann equation 152
		4.2.2.3	The boundednes of the $v$ -derivatives $\ldots \ldots \ldots \ldots 154$
		4.2.2.4	The hydrodynamical limit on the torus for Maxwellian
			$particles \dots \dots$
4.3	Too	lbox: flu	id projection and a priori energy estimates 157
	4.3.1	Properti	ies concerning the fluid projection $\pi_L$
	4.3.2	A priori	energy estimates
		4.3.2.1	Time evolutions for quantities in $H^1_{x,v}$
		4.3.2.2	Time evolutions for quantities in $H^s_{x,v}$
		4.3.2.3	Time evolutions for orthogonal quantities in $H^s_{x,v}$ 160
4.4	Line	ear case:	proof of Theorem 4.2.1
	4.4.1	The case	$e \ s = 1  \dots  \dots  \dots  162$
	4.4.2	The ind	uction in higher order Sobolev spaces 164
		4.4.2.1	The time evolution of $Q_{l,i}$
		4.4.2.2	The time evolution of $F_s$ and conclusion
4.5	Esti	mate for	the full equation: proof of Proposition 4.2.2166
	4.5.1	The case	$e \ s = 1  \dots  \dots  167$
	4.5.2	The ind	uction in higher order Sobolev spaces 168
		4.5.2.1	The time evolution of $Q_{l,i}$
		4.5.2.2	The time evolution of $F_s$ and conclusion
4.6	Exis	stence ar	nd exponential decay: proof of Theorem 4.2.3 170
	4.6.1	Proof of	the existence of global solutions
		4.6.1.1	Construction of solutions to a linearized problem 170

	4.6.1.2 E	coundedness of the sequence	172
	4.6.1.3 T	The global existence of solutions	173
4.6.2	Proof of th	e exponential decay	174
4.7 Exp	onential de	ecay of $v$ -derivatives: proof of Theorem 4.2.4	175
4.7.1	An a prior	i estimate	176
	4.7.1.1 T	The case $s = 1$	176
	4.7.1.2 T	`he induction in higher order Sobolev spaces	177
	4.7.1.3 T	The time evolution of $\sum Q_{l,i}$	178
	4.7.1.4 T	The time evolution of $F_s$ and conclusion	179
4.7.2	The expon	ential decay: proof of Theorem 4.2.4	180
4.8 Incompressible Navier-Stokes Limit: proof of Theorem			
4.8.1	A converge	ence in finite time	182
	4.8.1.1 S	tudy of the linear part	184
	4.8.1.2 S	tudy of the bilinear part	186
4.8.2	Proof of T	heorem 4.2.5	189
Appendic	es		189
4.A Vali	dation of t	he assumptions	189
4.A.1	The semi-	lassical relaxation	190
4.A.2	Boltzmann	operator with angular cutoff and hard potential $\ . \ .$	191
	4.A.2.1 C	Prthogonality to $\operatorname{Ker}(L)$ : (H5)	191
	4.A.2.2 C	Controlling derivatives: (H4)	192
4.A.3	Landau op	erator with hard and moderately soft potential $\ldots$	193
	4.A.3.1 C	Prthogonality to $\operatorname{Ker}(L)$ : (H5)	194
	4.A.3.2 C	ontrolling derivatives: (H4)	194
4.B Proc	ofs of the r	esults given in the toolbox	194
4.B.1	Proof of P	roposition 4.3.1: $\ldots$	194
4.B.2	A priori en	ergy estimates	196
	4.B.2.1 T	Time evolution of pure $x$ -derivatives	197
	4.B.2.2 T	$\text{ ``ime evolution of } \left\  \nabla_v h \right\ _{L^2_{x,v}}^2 \dots \dots$	197
	4.B.2.3 T	Time evolution of $\langle  abla_x h,  abla_v h \rangle_{L^2_{x,v}}$	199
	4.B.2.4 T	Time evolution of $\left\ \partial_l^j h\right\ _{L^2_{x,v}}^2$ for $ j  \ge 1$ and $ j  +  l  = s$	200
	4.B.2.5 T	$\text{ ``ime evolution of } \langle \partial_{l-\delta_i}^{\delta_i}h, \partial_l^0h\rangle_{L^2_{x,v}} \dots \dots \dots \dots \dots$	201
	4.B.2.6 T	$\text{ ``ime evolution of } \left\  \nabla_v h^{\perp} \right\ _{L^2_{x,v}}^2 \dots \dots$	201
	4.B.2.7 A	new time evolution of $\langle  abla_x^{*,v} h,  abla_v h \rangle_{L^2_{x,v}} \dots \dots$	203
	4.B.2.8 T	$\text{ ``ime evolution of } \left\ \partial_l^j h^\perp\right\ _{L^2_{x,v}}^2,  j \geqslant 1   \text{and}    j + l =s  .$	204

	4.B.2.9	A new time evolution of $\langle \partial_{l-\delta_i}^{\delta_i} h, \partial_l^0 h \rangle_{L^2_{x,v}}$
4.C Proc	of of the	hydrodynamical limit lemmas
4.C.1	Study of	f the linear part 208
	4.C.1.1	Proof of Lemma 4.8.6
	4.C.1.2	Proof of Lemma 4.8.7
	4.C.1.3	Proof of Lemma 4.8.8
$4.\mathrm{C.2}$	Study of	f the bilinear part 213
	4.C.2.1	A simplification without loss of generality 213
	4.C.2.2	Proof of Lemma 4.8.9
	4.C.2.3	Proof of Lemma 4.8.11
	4.C.2.4	Proof of Lemma 4.8.12
	4.C.2.5	Proof of Lemma 4.8.14

#### 4.1 Introduction

This chapter deals with the Boltzmann equation in a perturbative setting as the Knudsen number tends to 0. We consider the latter equation to describe the behaviour of rarefied gas particles moving on  $\mathbb{T}^d$  (flat torus of dimension  $d \ge 2$ ) with velocities in  $\mathbb{R}^d$  when the only interactions taken into account are binary collisions. More precisely, the Boltzmann equation describes the time evolution of the distribution of particles in position and velocity. A formal derivation of the Boltzmann equation from Newton's laws under the rarefied gas assumption can be found in [28], while [30] present Lanford's Theorem (see [65] and [44] for detailed proofs) which rigorously proves the derivation in short times.

Consider the following more general form of it, where we denote the Knudsen number by  $\varepsilon$  .

$$\partial_t f + v \cdot \nabla_x f = \frac{1}{\varepsilon} Q(f, f), \text{ on } \mathbb{T}^d \times \mathbb{R}^d$$
  
= 
$$\int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} \Phi\left(|v - v_*|\right) b\left(\cos \theta\right) \left[f' f'_* - f f_*\right] dv_* d\sigma, \qquad (4.1.1)$$

where f',  $f_*$ ,  $f'_*$  and f are the values taken by f at v',  $v_*$ ,  $v'_*$  and v respectively. Define:

$$\begin{cases} v' = \frac{v + v_*}{2} + \frac{|v - v_*|}{2}\sigma \\ v'_* = \frac{v + v_*}{2} - \frac{|v - v_*|}{2}\sigma \end{cases}, \text{ and } \cos \theta = \left\langle \frac{v - v_*}{|v - v_*|}, \sigma \right\rangle \end{cases}$$

One can find in [28], [30] or [46] that the global equilibria for the Boltzmann equation are the *Maxwellians*  $\mu(v)$ , which are gaussian density functions. Without loss of generality we consider only the case of normalized Maxwellians:

$$\mu(v) = \frac{1}{(2\pi)^{\frac{d}{2}}} e^{-\frac{|v|^2}{2}}$$

The bilinear operator Q(g, h) is given by

$$Q(g,h) = \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} \Phi\left(|v - v_*|\right) b\left(\cos\theta\right) \left[h'g'_* - hg_*\right] dv_* d\sigma.$$

#### 4.1.1 The problem and its motivations

The Knudsen number is the inverse of the average number of collisions for each particle per unit of time. Therefore, as reviewed in [111], one can expect a convergence from the Boltzmann model towards the acoustics and the fluid dynamics as the Knudsen number tends to 0. This latter convergence will be specified. However, these different models describe physical phenomenon that do not evolve at the same timescale. As suggested in previous studies (see [46][111][98]) we can rescale our original equation in time by a factor  $\varepsilon$ , to get rid of these time scale differences. Moreover, they also suggested that a perturbation of order  $\varepsilon$  around the global equilibrium  $\mu(v)$  should approximate, as the Knudsen number tends to 0, the incompressible Navier-Stokes equations.

Therefore we study the following equation

$$\partial_t f_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x f_{\varepsilon} = \frac{1}{\varepsilon^2} Q(f_{\varepsilon}, f_{\varepsilon}) , \text{ on } \mathbb{T}^d \times \mathbb{R}^d,$$
(4.1.2)

under the linearization  $f_{\varepsilon}(t, x, v) = \mu(v) + \varepsilon \mu^{1/2}(v) h_{\varepsilon}(t, x, v)$ . This leads us to the linearized Boltzmann equation

$$\partial_t h_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x h_{\varepsilon} = \frac{1}{\varepsilon^2} L(h_{\varepsilon}) + \frac{1}{\varepsilon} \Gamma(h_{\varepsilon}, h_{\varepsilon}).$$
(4.1.3)

that we will study thoroughly, and where

$$\begin{cases} L(h) &= \left[ Q(\mu, \mu^{\frac{1}{2}}h) + Q(\mu^{\frac{1}{2}}h, \mu) \right] \mu^{-\frac{1}{2}} \\ \Gamma(g, h) &= \frac{1}{2} \left[ Q(\mu^{\frac{1}{2}}g, \mu^{\frac{1}{2}}h) + Q(\mu^{\frac{1}{2}}h, \mu^{\frac{1}{2}}g) \right] \mu^{-\frac{1}{2}}. \end{cases}$$

All along this chapter we consider the Boltzmann equation with hard potential or Maxwellian potential ( $\gamma = 0$ ), that is to say there is a constant  $C_{\Phi} > 0$  such that

$$\Phi(z) = C_{\Phi} z^{\gamma} , \ \gamma \in [0,1],$$

and a strong form of Grad's *angular cutoff* (see [48]), expressed here by the fact that we assume b to be  $C^1$  with the controls from above

$$\forall z \in [-1,1], \ b(z), b(z') \leq C_b,$$

b and  $\Phi$  being defined in equation (4.1.1).

The aim of the present chapter is to use a constructive method to obtain existence and exponential decay for solutions to the linearized Boltzmann equation (4.1.3), uniformly in the Knudsen number.

Such a uniform result is then used to derive explicit rates of convergence for  $(h_{\varepsilon})_{\varepsilon>0}$  towards its limit as  $\varepsilon$  tends to 0. This allows us to prove, and quantify, the convergence from Boltzmann equation to the incompressible Navier-Stokes equations (4.1.4).

#### 4.1.2 Notations

Throughout this chapter, we will use the following notations. For two multi-indexes j and l in  $\mathbb{N}^d$  we define:

- $\partial_l^j = \partial_{v_i} \partial_{x_l},$
- for i in  $\{1, \ldots, d\}$  we denote by  $c_i(j)$  the  $i^{th}$  coordinate of j,
- the length of j will be written  $|j| = \sum_i c_i(j)$ ,
- the multi-index  $\delta_{i_0}$  by :  $c_i(\delta_{i_0}) = 1$  if  $i = i_0$  and 0 elsewhere.

We will write the spaces we are working on  $L_{x,v}^p = L^p (\mathbb{T}^d \times \mathbb{R}^d)$ ,  $L_x^p = L^p (\mathbb{T}^d)$  and  $L_v^p = L^p (\mathbb{R}^d)$ . The Sobolev spaces  $H_{x,v}^s$ ,  $H_x^s$  and  $H_v^s$  are defined in the same way and we denote the standard Sobolev norms by  $\|\cdot\|_{H_{x,v}^s}^2 = \sum_{|j|+|l|\leqslant s} \left\|\partial_l^j \cdot \right\|_{L_{x,v}^2}^2$ .

#### 4.1.3 Our strategy and results

The first step of this chapter is to investigate the equation (4.1.3) in order to obtain existence and exponential decay of solutions in Sobolev spaces  $H_{x,v}^s$ , independently of the Knudsen number  $\varepsilon$ . Moreover, we want all the required smallness assumptions and rates of convergence to be explicit. Such a result has been proved in [56] by studying independently the behaviour of both microscopic and fluid parts of solutions to (4.1.3), we proposed here another method based on hypocoercivity estimates.

Our strategy is to build a norm on Sobolev spaces which is equivalent to the standard norm and which satisfies a Gronwall type inequality.

First, we construct a functional on  $H^s_{x,v}$  by considering a linear combination of  $\left\|\partial_l^j\cdot\right\|_{L^2_{x,v}}^2$ , for all  $|j|+|l| \leq s$ , together with product terms of the form  $\langle\partial_{l-\delta_i}^{\delta_i},\partial_l^0\cdot\rangle_{L^2_{x,v}}$ . The distortion of the standard norm by the addition of mixed terms is necessary to have a relaxation, due to the hypocoercivity property of the linearized Boltzmann equation (4.1.3) (see [82]).

We then study the flow of this functional along time for solutions to the linearized Boltzmann equation (4.1.3). This flow is controlled by energy estimates and, finally, a nontrivial choice of coefficients in the functional yields an equivalence between the functional and the standard Sobolev norm, as well as a Gronwall type inequality, both of them being independent of  $\varepsilon$ .

We first apply this strategy to the linear case (i.e. without considering the bilinear remainder term) and prove that it generates a strong semigroup with a spectral gap and, therefore, an exponential relaxation (Theorem 4.2.1). We then extend our method to the full nonlinear model and obtain an a priori estimate on our functional (Proposition 4.2.2).

This estimate enables us to prove the existence of solutions to the Cauchy problem and their exponential decay as long as the initial data is small enough with a smallness not depending on  $\varepsilon$  (Theorem 4.2.3). We emphasize here that, thanks to the functional we used, the smaller  $\varepsilon$  the less control is needed on the *v*-derivatives of the initial data.

However, these results seem to tell us that the v-derivatives of solutions to equation (4.1.3) can blow-up as  $\varepsilon$  tends to 0. Thus, the last step is to create a new functional, based on the microscopic part of solutions, satisfying the same properties but controlling the v-derivatives as well. The fact that we ask for a control on the microscopic part of solutions to equation (4.1.3) is due to the deep structure of the linear operator L. This leads to the expected exponential decay independently of  $\varepsilon$  even for those v-derivatives (Theorem 4.2.4).

Finally, the chief aim of the present chapter is to derive explicit rates of convergence from solutions to the linearized Boltzmann equation to the incompressible Navier-Stokes equations.

Theorem 4.2.3 tells us that for all  $\varepsilon$  we can build a solution  $h_{\varepsilon}$  to the linearized Boltzmann equation (4.1.3), as long as the initial perturbation is sufficiently small, independently of  $\varepsilon$ . We can then consider the sequence  $(h_{\varepsilon})_{0<\varepsilon\leq 1}$  and study its limit. It appears that it converges weakly in  $L_t^{\infty} H_x^s L_v^2$ , for  $s \ge s_0 > d/2$ , towards a function h. Furthermore, we have the following form for h (see [10])

$$h(t, x, v) = \left[\rho(t, x) + v \cdot u(t, x) + \frac{1}{2}(|v|^2 - N)\theta(t, x)\right]\mu(v)^{1/2},$$

of which physical observables are weak (in the Leray sense [66]) solutions of the linearized incompressible Navier-Stokes equations (p being the pressure function,  $\nu$  and  $\kappa$  being constants determined by L, see Theorem 5 in [46])

$$\partial_t u - \nu \Delta u + u \cdot \nabla u + \nabla p = 0,$$
  

$$\nabla \cdot u = 0,$$
  

$$\partial_t \theta - \kappa \Delta \theta + u \cdot \nabla \theta = 0,$$
  
(4.1.4)

together with the Boussineq relation

$$\nabla(\rho + \theta) = 0. \tag{4.1.5}$$

However, in order to know the initial data of these quantities, we study the Fourier transform on the torus of our linear operator and use Duhamel formula. This gives us a strong convergence result on the time average of  $h_{\varepsilon}$  with an explicit rate of convergence
in finite time. An interpolation between this finite time convergence and the exponential stability of the global equilibria in Boltzmann and Navier-Stokes equations concludes with a strong convergence for all times (Theorem 4.2.5). The way we tackle this convergence allows us to obtain an explicit form for the limit of  $(h_{\varepsilon})_{\varepsilon>0}$ .

#### 4.1.4 Comparison with existing results

For physical purposes, one may assume that  $\varepsilon = 1$  which is a mere normalization and that is why many articles about the linearized Boltzmann equation only deal with this case. The associated Cauchy problem has been worked on over the past fifty years, starting with Grad [50], and it has been studied in different spaces, such as weighted  $L_v^2(H_x^l)$  spaces [107] or weighted Sobolev spaces [53][55][114]. Other results have also been proved in  $\mathbb{R}^d$  instead of the torus, see for instance [88][1][31], but it will not be the purpose of this chapter.

Our chapter explicitly deals with the general case for  $\varepsilon$  and we prove results that are uniform in  $\varepsilon$ , allowing us to consider the hydrodynamical limit as the Knudsen number tends to 0. To solve the Cauchy problem we used an iterative scheme, like in the papers mentioned above, but our strategy yields a condition for the existence of solutions in  $H_{x,v}^s$ (without any weight) which is uniform in  $\varepsilon$  (Theorem 4.6.3). In order to obtain such a result, we had to consider more precise estimates on the bilinear operator  $\Gamma$ , depending on the existence of v-derivatives or not. Bardos and Ukai [10] obtained a similar result in  $\mathbb{R}^d$  but in weighted Sobolev spaces and did not prove any decay.

The behaviour of such global in time solutions has also been studied. Guo worked in weighted Sobolev spaces and proved the boundedness of solutions to equation (4.1.3) in [55], as well as an exponential decay (uniform in  $\varepsilon$ ) in [56]. The norm involved in [55][56] is quite intricate and requires a lot of technical computations. To avoid specific and technical calculations, the theory of hypocoercivity (see [81]) focuses on the properties of the Boltzmann operator and which are quite similar to hypoellipticity. This theory has been used in [82] to obtain exponential decay in standard Sobolev spaces in the case  $\varepsilon = 1$ .

We use the idea of Mouhot and Neumann developed in [82] consisting of considering a functional on  $H^s_{x,v}$  involving mixed scalar products. In this chapter we thus construct such a quadratic form, but with coefficient depending on  $\varepsilon$ . Working in the general case for  $\varepsilon$  yields new calculations and requires the use of certain orthogonal properties of the bilinear operator  $\Gamma$  to overcome these issues. Moreover, we must construct a new norm out of this functional, which controls the v-derivatives by a factor  $\varepsilon$ .

The fact that the study yields a norm containing some  $\varepsilon$  factors prevents us from having a uniform exponential decay for the v-derivatives. We use the idea of Guo, in [56], of looking at the microscopic part of the solution  $h_{\varepsilon}$  everytime we look at a differentiation in v. This idea catches the interesting structure of L on its orthogonal part. Combining this idea with our previous strategy fills the gap for the v-derivatives. Finally, our uniform results enable us to derive a weak convergence in  $H_x^s L_v^2$  towards solutions to the incompressible Navier-Stokes equations, together with the Boussineq relation. We then find a way to obtain strong convergence using the ideas of the Fourier study of the linear operator  $L - v . \nabla_x$ , developed in [39] and [10], combined with Duhamel formula. However, the study done in [10] relies strongly on an argument of stationary phase developed in [109] which is no longer applicable in the torus. Indeed, the Fourier space of  $\mathbb{R}^d$  is continuous and so integration by parts can be used in that frequency space. This tool is no longer available in the frequency space of the torus which is discrete.

Theorem 4.2.5 shows that the behaviour of the hydrodynamical limit is quite different on the torus, where an averaging in time is necessary for general initial data. However, we obtain the same relation between the limit at t = 0 and the initial perturbation  $h_{in}$ and also the existence of an initial layer. That is to say that we have a convergence in  $L^2_{[0,T]} = L^2([0,T])$  if and only if the initial perturbation satisfies some physical properties, which appear to be the same as in  $\mathbb{R}^d$  studied in [10].

This convergence gives a perturbative result for incompressible Navier-Stokes equations in Sobolev spaces around the steady solution. The regularity of the weak solutions we constructed implies that they are in fact strong solutions (see Serrin [99][100] and Lions [67] Section 2.5). Moreover, our uniform exponential decay for solutions to the linearized Boltzmann equation yields an exponential decay for the perturbative solutions of the incompressible Navier-Stokes equations in higher Sobolev spaces. Such an exponential convergence to equilibrium has been derived in  $H_0^1$  for d = 2 or d = 3 in [106], or can be deduced from Proposition 3.7 in [72] in higher Sobolev spaces for small initial data. The general convergence to equilibrium can be found in [74] (small initial data) and in [90] but they focus on the general compressible case and no rate of decay is built.

Furthermore, results that do not involve hydrodynamical limits (existence and exponential decay results) are applicable to a larger class of operators. In Appendix 4.A we prove that those theorems also hold for other kinetic collisional models such as the linear relaxation, the semi-classical relaxation, the linear Fokker-Planck equation and the Landau equation with hard and moderately soft potential.

#### 4.1.5 Organization of the chapter

Section 4.2 is divided in two different subsections.

As mentionned above, we shall use the hypocoercivity of the Boltzmann equation (4.1.1). This hypocoercivity can be described in terms of technical properties on L and  $\Gamma$  and, in order to obtain more general results, we consider them as a basis of our chapter. Thus, subsection 4.2.1 describes them in detail and a proof of the fact that L and  $\Gamma$  indeed satisfy those properties is given in Appendix 4.A. Most of them have been proved in [82] but we require more precise ones to deal with the general case.

The second subsection 4.2.2 is dedicated to a mathematical formulation of the results described in subsection 4.1.3.

As said when we described our strategy (subsection 4.1.3), we are going to study the flow of a functional involving  $L^2_{x,v}$ -norm of x and v derivatives and mixed scalar products. To control this flow in time we compute energy estimates for each of these terms in a toolbox (section 4.3) which will be used and referred to all along the rest of the chapter. Proofs of those energy estimates are given in Appendix 4.B.

Finally, sections 4.4, 4.5, 4.6, 4.7 and 4.8 are the proofs respectively of Theorem 4.2.1 (about the strong semigroup property of the linear part of equation (4.1.3)), Proposition 4.2.2 (an a priori estimates on the constructed functional for the full model), Theorem 4.2.3 (existence and exponential decay of solutions to equation (4.1.3)), Theorem 4.2.4 (showing the uniform boundedness of the *v*-derivatives) and of Theorem 4.2.5 (dealing with the hydrodynamical limit).

We notice here that section 4.6 is divided in two subsection. Subsection 4.6.1 deals with the existence of solutions for all  $\varepsilon > 0$  and subsection 4.6.2 proved the exponential decay of those solutions.

# 4.2 Main results

This section is divided in two parts. The first one translate the hypocoercivity aspects of the Boltzmann operator in terms of mathematical properties for L and  $\Gamma$ . Then, the second one states our results in terms of those assumptions.

#### 4.2.1 Hypocoercivity assumptions

This section is dedicated to the framework and assumptions of the hypocoercivity theory. A state of the art of this theory can be found in [81].

#### **4.2.1.1** Assumptions on the linear operator L

Assumptions in  $H^1_{x,v}$  :

(H1): Coercivity and general controls  $L: L_v^2 \longrightarrow L_v^2$  is a closed and self-adjoint operator with  $L = K - \Lambda$  such that:

- $\Lambda$  is coercive:
  - there exists  $\|.\|_{\Lambda_v}$  norm on  $L_v^2$  such that

$$\forall h \in L^2_v, \ \nu^{\Lambda}_0 \|h\|^2_{L^2_v} \leqslant \nu^{\Lambda}_1 \|h\|^2_{\Lambda_v} \leqslant \langle \Lambda(h), h \rangle_{L^2_v} \leqslant \nu^{\Lambda}_2 \|h\|^2_{\Lambda_v},$$

 $-\Lambda$  has a defect of coercivity regarding its v derivatives:

$$\forall h \in H_v^1, \ \langle \nabla_v \Lambda(h), \nabla_v h \rangle_{L_v^2} \ge \nu_3^{\Lambda} \| \nabla_v h \|_{\Lambda_v}^2 - \nu_4^{\Lambda} \| h \|_{\Lambda_v}^2.$$

• There exists  $C^L > 0$  such that

$$\forall h \in L_v^2, \ \forall g \in L_v^2, \ \langle L(h), g \rangle_{L_v^2} \leqslant C^L \, \|h\|_{\Lambda_v} \, \|g\|_{\Lambda_v},$$

where  $(\nu_s^{\Lambda})_{1 \leq s \leq 4}$  are strictly positive constants depending on the operator and the dimension of the velocities space d.

As in [82], we define a new norm on  $L^2_{x,v}$ :

$$\|.\|_{\Lambda} = \|\|.\|_{\Lambda_v}\|_{L^2_x}$$

#### (H2): Mixing property in velocity

$$\forall \delta > 0 , \exists C(\delta) > 0 , \forall h \in H_v^1 , \quad \langle \nabla_v K(h), \nabla_v h \rangle_{L_v^2} \leqslant C(\delta) \|h\|_{L_v^2}^2 + \delta \|\nabla_v h\|_{L_v^2}^2$$

## (H3): Relaxation to equilibrium

We suppose that the kernel of L is generated by N functions which form an orthonormal basis for Ker(L):

$$\operatorname{Ker}(L) = \operatorname{Span}\{\phi_1(v), \dots, \phi_N(v)\}.$$

Moreover, we assume that the  $\phi_i$  are of the form  $P_i(v)e^{-|v|^2/4}$ , where  $P_i$  is a polynomial.

Furthermore, denoting by  $\pi_L$  the orthogonal projector in  $L_v^2$  on Ker(L) we assume that we have the following local coercivity property:

$$\exists \lambda > 0 , \, \forall h \in L^2_v , \quad \langle L(h), h \rangle_{L^2_v} \leqslant -\lambda \left\| h^{\perp} \right\|_{\Lambda_v}^2,$$

where  $h^{\perp} = h - \pi_L(h)$  denotes the microscopic part of h (the orthogonal to Ker(L) in  $L_v^2$ ).

We are using the same hypothesis as in [82], except that we require the  $\phi_i$  to be of a specific form. This additional requirement allows us to derive properties on the *v*derivatives of  $\pi_L$  that we will state in the toolbox section 4.3.

Then we have two more properties on L in order to deal with higher order Sobolev spaces.

# Assumptions in $H_{x,v}^s$ , s > 1 :

# (H1'): Defect of coercivity for higher derivatives

We assume that L satisfies (H1) along with the following property: for all  $s \ge 1$ , for all |j| + |l| = s such that  $|j| \ge 1$ ,

$$\forall h \in H^s_{x,v} , \quad \langle \partial_l^j \Lambda(h), \partial_l^j h \rangle_{L^2_{x,v}} \geqslant \nu_5^{\Lambda} \left\| \partial_l^j h \right\|_{\Lambda}^2 - \nu_6^{\Lambda} \left\| h \right\|_{H^{s-1}_{x,v}} ,$$

where  $\nu_5^{\Lambda}$  and  $\nu_6^{\Lambda}$  are strictly positive constants depending on L and d. We also define a new norm on  $H_{x,v}^s$ :

$$\|.\|_{H^s_{\Lambda}} = \left(\sum_{|j|+|l| \leqslant s} \left\|\partial_l^j.\right\|_{\Lambda}^2\right)^{1/2}$$

# (H2'): Mixing properties

As above, Mouhot and Neumann extended the hypothesis (H2) to higher Sobolev's spaces: for all  $s \ge 1$ , for all |j| + |l| = s such that  $|j| \ge 1$ ,

$$\forall \delta > 0 , \exists C(\delta) > 0 , \forall h \in H^s_{x,v} , \quad \langle \partial_l^j K(h), \partial_l^j h \rangle_{L^2_{x,v}} \leqslant C(\delta) \left\| h \right\|_{H^{s-1}_{x,v}}^2 + \delta \left\| \partial_l^j h \right\|_{L^2_{x,v}}^2.$$

# 4.2.1.2 Assumptions on the second order term $\Gamma$

To solve our problem uniformly in  $\varepsilon$  we had to precise the hypothesis made in [82] in order to have a deeper understanding of the operator  $\Gamma$ . This lead us to two different assumptions.

#### (H4): Control on the second order operator

 $\Gamma: L_v^2 \times L_v^2 \longrightarrow L_v^2$  is a bilinear symmetric operator such that for all multi-indexes j and l such that  $|j| + |l| \leq s, s \geq 0$ ,

$$\left| \langle \partial_l^j \Gamma(g,h), f \rangle_{L^2_{x,v}} \right| \leqslant \begin{cases} \mathcal{G}^s_{x,v}(g,h) \, \|f\|_{\Lambda} &, \text{ if } j \neq 0 \\ \mathcal{G}^s_x(g,h) \, \|f\|_{\Lambda} &, \text{ if } j = 0 \end{cases}$$

 $\mathcal{G}_{x,v}^s$  and  $\mathcal{G}_x^s$  being such that  $\mathcal{G}_{x,v}^s \leqslant \mathcal{G}_{x,v}^{s+1}$ ,  $\mathcal{G}_x^s \leqslant \mathcal{G}_x^{s+1}$  and satisfying the following property:

$$\exists s_0 \in \mathbb{N} , \forall s \ge s_0 , \exists C_{\Gamma} > 0 , \quad \begin{cases} \mathcal{G}_{x,v}^s(g,h) & \leq C_{\Gamma} \left( \|g\|_{H_{x,v}^s} \|h\|_{H_{\Lambda}^s} + \|h\|_{H_{x,v}^s} \|g\|_{H_{\Lambda}^s} \right) \\ \mathcal{G}_{x}^s(g,h) & \leq C_{\Gamma} \left( \|h\|_{H_{xL_v^s}^sL_v^2} \|g\|_{H_{\Lambda}^s} + \|g\|_{H_{xL_v^s}^sL_v^2} \|h\|_{H_{\Lambda}^s} \right). \end{cases}$$

#### (H5): Orthogonality to the Kernel of the linear operator

$$\forall h, g \in \text{Dom}(\Gamma) \cap L^2_v, \quad \Gamma(g,h) \in \text{Ker}(L)^{\perp}.$$

#### 4.2.2 Statement of the Theorems

# 4.2.2.1 Uniform result for the linear Boltzmann equation

For s in  $\mathbb{N}^*$  and some constants  $(b_{j,l}^{(s)})_{j,l}$ ,  $(\alpha_l^{(s)})_l$  and  $(a_{i,l}^{(s)})_{i,l}$  strictly positive and  $0 < \varepsilon \leq 1$ we define the following functional on  $H^s_{x,v}$ , where we emphasize that there is a dependance on  $\varepsilon$ , which is the key point of our study:

$$\|\cdot\|_{\mathcal{H}^s_{\varepsilon}} = \left[\sum_{\substack{|j|+|l|\leqslant s\\|j|\geqslant 1}} b_{j,l}^{(s)} \varepsilon^2 \left\|\partial_l^j \cdot\right\|_{L^2_{x,v}}^2 + \sum_{|l|\leqslant s} \alpha_l^{(s)} \left\|\partial_l^0 \cdot\right\|_{L^2_{x,v}}^2 + \sum_{\substack{|l|\leqslant s\\i,c_i(l)>0}} a_{i,l}^{(s)} \varepsilon \langle\partial_{l-\delta_i}^{\delta_i} \cdot, \partial_l^0 \cdot\rangle_{L^2_{x,v}}\right]^{\frac{1}{2}}.$$

We first study the linearized equation (4.1.3), without taking into account the bilinear remainder operator. By letting  $\pi_w$  be the projector in  $L^2_{x,v}$  onto Ker(w) we obtained the following semigroup property for L.

**Theorem 4.2.1** If *L* is a linear operator satisfying the conditions (H1'), (H2') and (H3) then there exists  $0 < \varepsilon_d \leq 1$  such that for all *s* in  $\mathbb{N}^*$ ,

- 1. for all  $0 < \varepsilon \leq \varepsilon_d$ ,  $G_{\varepsilon} = \varepsilon^{-2}L \varepsilon^{-1}v \cdot \nabla_x$  generates a  $C^0$ -semigroup on  $H^s_{x,v}$ .
- 2. there exist  $C_G^{(s)}$ ,  $(b_{j,l}^{(s)})$ ,  $(\alpha_l^{(s)})$ ,  $(a_{i,l}^{(s)}) > 0$  such that for all  $0 < \varepsilon \leq \varepsilon_d$ :

$$\left\|\cdot\right\|_{\mathcal{H}^s_{\varepsilon}}^2 \sim \left(\left\|\cdot\right\|_{L^2_{x,v}}^2 + \sum_{|l|\leqslant s} \left\|\partial_l^0\cdot\right\|_{L^2_{x,v}}^2 + \varepsilon^2 \sum_{\substack{|l|+|j|\leqslant s\\|j|\geqslant 1}} \left\|\partial_l^j\cdot\right\|_{L^2_{x,v}}^2\right),$$

and for all h in  $H^s_{x,v}$ ,

$$\langle G_{\varepsilon}(h), h \rangle_{\mathcal{H}^{s}_{\varepsilon}} \leqslant -C_{G}^{(s)} \|h - \pi_{G_{\varepsilon}}(h))\|_{H^{s}_{\Lambda}}^{2}.$$

This theorem gives us an exponential decay for the semigroup generated by  $G_{\varepsilon}$ .

#### 4.2.2.2 Uniform perturbative result for the Boltzmann equation

The next result states that if we add the bilinear remainder operator then it is enough, if  $\varepsilon$  is small enough, to slightly change our new norm to have a control on the solution.

**Proposition 4.2.2** If L is a linear operator satisfying the conditions (H1'), (H2') and (H3) and  $\Gamma$  a bilinear operator satisfying (H4) and (H5) then there exists  $0 < \varepsilon_d \leq 1$  such that for all s in  $\mathbb{N}^*$ , 1. there exist  $K_0^{(s)}$ ,  $K_1^{(s)}$ ,  $K_2^{(s)}$   $(b_{j,l}^{(s)})$ ,  $(\alpha_l^{(s)})$ ,  $(a_{i,l}^{(s)}) > 0$ , independent of  $\Gamma$  and  $\varepsilon$ , such that for all  $0 < \varepsilon \leq \varepsilon_d$ :

$$\|\cdot\|_{\mathcal{H}^s_{\varepsilon}}^2 \sim \left(\|\cdot\|_{L^2_{x,v}}^2 + \sum_{\substack{|l|\leqslant s}} \left\|\partial_l^0\cdot\right\|_{L^2_{x,v}}^2 + \varepsilon^2 \sum_{\substack{|l|+|j|\leqslant s\\|j|\geqslant 1}} \left\|\partial_l^j\cdot\right\|_{L^2_{x,v}}^2\right),$$

2. and for all  $h_{in}$  in  $H^s_{x,v} \cap \operatorname{Ker}(G_{\varepsilon})^{\perp}$  and all g in  $\operatorname{Dom}(\Gamma) \cap H^s_{x,v}$ , if we have a solution h in  $H^s_{x,v}$  to the following equation

$$\partial_t h + \frac{1}{\varepsilon} v \cdot \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(g, h),$$

then

$$\frac{d}{dt} \|h\|_{\mathcal{H}^{s}_{\varepsilon}}^{2} \leqslant -K_{0}^{(s)} \|h\|_{H^{s}_{\Lambda}}^{2} + K_{1}^{(s)} \left(\mathcal{G}^{s}_{x}(g,h)\right)^{2} + \varepsilon^{2} K_{2}^{(s)} \left(\mathcal{G}^{s}_{x,v}(g,h)\right)^{2}.$$

One can remark that the norm constructed above leaves free the x-derivatives while it controls the v ones by a factor  $\varepsilon$ .

We want to emphasize here that this result shows that the derivative of the norm is control by the x-derivatives of  $\Gamma$  and the Sobolev norm of  $\Gamma$ , but weakened by a factor  $\varepsilon^2$ . This is important as our norm  $\|.\|_{\mathcal{H}^s_{\varepsilon}}^2$  controls the  $L^2_v(H^s_x)$ -norm by a factor of order 1 whereas it controls the whole  $H^s_{x,v}$ -norm by a multiplicative factor of order  $1/\varepsilon$ .

**Theorem 4.2.3** Let Q be a bilinear operator such that:

- the equation (4.1.2) admits an equilibrium  $0 \leq \mu \in L^1(\mathbb{T}^d \times \mathbb{R}^d)$ ,
- the linearized operator L = L(h) around  $\mu$  with the scaling  $f = \mu + \varepsilon \mu^{1/2} h$  satisfies (H1'), (H2') and (H3),
- the bilinear remaining term  $\Gamma = \Gamma(h, h)$  in the linearization satisfies (H4) and (H5).

Then there exists  $0 < \varepsilon_d \leq 1$  such that for any  $s \geq s_0$  (defined in (H4)),

1. there exist  $(b_{j,l}^{(s)})$ ,  $(\alpha_l^{(s)})$ ,  $(a_{i,l}^{(s)}) > 0$ , independent of  $\Gamma$  and  $\varepsilon$ , such that for all  $0 < \varepsilon \leq \varepsilon_d$ :

$$\|\cdot\|_{\mathcal{H}^s_{\varepsilon}}^2 \sim \left(\|\cdot\|_{L^2_{x,v}}^2 + \sum_{|l| \leq s} \left\|\partial_l^0 \cdot\right\|_{L^2_{x,v}}^2 + \varepsilon^2 \sum_{\substack{|l|+|j| \leq s \\ |j| \geq 1}} \left\|\partial_l^j \cdot\right\|_{L^2_{x,v}}^2\right),$$

2. there exist  $\delta_s > 0$ ,  $C_s > 0$  and  $\tau_s > 0$  such that for all  $0 < \varepsilon \leq \varepsilon_d$ :

For any distribution  $0 \leq f_{in} \in L^1(\mathbb{T}^d \times \mathbb{R}^d)$  with  $f_{in} = \mu + \varepsilon \mu^{1/2} h_{in} \geq 0$ ,  $h_{in}$  in  $\text{Ker}(G_{\varepsilon})^{\perp}$ and

$$\|h_{in}\|_{\mathcal{H}^s} \leqslant \delta_s,$$

there exists a unique global smooth (in  $H^s_{x,v}$ , continuous in time) solution  $f_{\varepsilon} = f_{\varepsilon}(t, x, v)$ to (4.1.2) which, moreover, satisfies  $f_{\varepsilon} = \mu + \varepsilon \mu^{1/2} h_{\varepsilon} \ge 0$  with:

$$\|h_{\varepsilon}\|_{\mathcal{H}^{s}_{\varepsilon}} \leqslant \|h_{in}\|_{\mathcal{H}^{s}_{\varepsilon}} e^{-\tau_{s}t}.$$

The fact that we are asking  $h_{in}$  to be in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  just states that we want  $f_{in}$  to have the same physical quantities as the global equilibrium  $\mu$ . This is a compulsory requirement as one can easily check that the physical quantities

$$\int_{\mathbb{T}^d \times \mathbb{R}^d} f_{\varepsilon}(x, v) dx dv, \quad \int_{\mathbb{T}^d \times \mathbb{R}^d} v f_{\varepsilon}(x, v) dx dv, \quad \int_{\mathbb{T}^d \times \mathbb{R}^d} |v|^2 f_{\varepsilon}(x, v) dx dv$$

are preserved with time (see [30] for instance).

Notice that the  $\mathcal{H}^s_{\varepsilon}$ -norm is this theorem is the same than the one we constructed in Proposition 4.2.2.

#### 4.2.2.3 The boundednes of the *v*-derivatives

As a corollary we have that the  $H^s_x(L^2_v)$ -norm decays exponentially independently of  $\varepsilon$  but that the only control we have on the  $H^s_{x,v}$  is

$$\|h_{\varepsilon}\|_{H^s_{x,v}} \leqslant \frac{\delta_s}{\varepsilon} e^{-\tau_s t}.$$

This seems to tell us that the v-derivatives can blow-up at a rate  $1/\varepsilon$ . However, Guo, in [56], showed that one can prove that there is no explosion if one controls independently the fluid part and the microscopic part of the solution. This idea, combined with our original one, leads to the construction of a new norm which will only control the microscopic part of the solution whenever we face a derivative in the v variable.

We define the following positive quadratic form

$$\|\cdot\|_{\mathcal{H}^{s}_{\varepsilon\perp}}^{2} = \sum_{\substack{|j|+|l|\leqslant s\\|j|\geqslant 1}} b_{j,l}^{(s)} \left\|\partial_{l}^{j}(\mathrm{Id}-\pi_{L})\right\|_{L^{2}_{x,v}}^{2} + \sum_{|l|\leqslant s} \alpha_{l}^{(s)} \left\|\partial_{l}^{0}\cdot\right\|_{L^{2}_{x,v}}^{2} + \sum_{\substack{|l|\leqslant s\\i,c_{i}(l)>0}} a_{i,l}^{(s)}\varepsilon\langle\partial_{l-\delta_{i}}^{\delta_{i}}\cdot,\partial_{l}^{0}\cdot\rangle_{L^{2}_{x,v}}.$$

**Theorem 4.2.4** Under the same conditions as in Theorem 4.2.3, for all  $s \ge s_0$ , there exist  $(b_{j,l}^{(s)})$ ,  $(\alpha_l^{(s)})$ ,  $(a_{i,l}^{(s)}) > 0$  and  $0 < \varepsilon_d \le 1$  such that for all  $0 < \varepsilon \le \varepsilon_d$ :

1.  $\|\cdot\|_{\mathcal{H}^{s}_{\varepsilon\perp}} \sim \|\cdot\|_{H^{s}_{x,v}}$ , independently of  $\varepsilon$ ,

2. if  $h_{\varepsilon}$  is a solution of 4.1.3 in  $H^s_{x,v}$  with  $\|h_{in}\|_{\mathcal{H}^s_{\varepsilon^+}} \leq \delta'_s$  then

$$\|h_{\varepsilon}\|_{\mathcal{H}^{s}_{\varepsilon\perp}} \leqslant \delta'_{s} e^{-\tau'_{s}t},$$

where  $\delta'_s$  and  $\tau'_s$  are strictly positive constants independent of  $\varepsilon$ .

This theorem builds up a functional that is equivalent to the standard Sobolev norm, independently of  $\varepsilon$ . Thus, it gives us the exponential decay of the *v*-derivatives as well as the decay of the *x*-derivatives. However, the distorted norm used in Theorem 4.2.3 asked less control on the *v*-derivatives for the initial data, suggesting that, in the limit as  $\varepsilon$  goes to zero, almost only the *x*-variable has to be controlled to obtain existence and exponential decay.

#### 4.2.2.4 The hydrodynamical limit on the torus for Maxwellian particles

Our theorem states that one can really expect a convergence of solutions of collisional kinetic models near equilibrium towards a solution of fluid dynamics equations. Indeed, the smallness assumption on the initial perturbation does not depend on the parameter  $\varepsilon$  as long as  $\varepsilon$  is small enough.

We then define the following macroscopic quantities

- the particles density  $\rho_{\varepsilon}(t,x) = \langle \mu(v)^{1/2}, h_{\varepsilon}(t,x,v) \rangle_{L^2_v}$ ,
- the mean velocity  $u_{\varepsilon}(t,x) = \langle v\mu(v)^{1/2}, h_{\varepsilon}(t,x,v) \rangle_{L^2_r}$ ,
- the temperature  $\theta_{\varepsilon}(t,x) = \frac{1}{d} \langle (|v|^2 d)\mu(v)^{1/2}, h_{\varepsilon}(t,x,v) \rangle_{L^2_v}.$

The theorem 4.2.3 tells us that, for  $s \ge s_0$ , the sequence  $(h_{\varepsilon})_{\varepsilon>0}$  converges (up to an extraction) weakly-\* in  $L_t^{\infty}(H_l^s L_v^2)$  towards a function h. Such a weak convergence enables us to use the theorem 1.1 of [10], which is a slight modification of the result in [8] to get that

1. h is in Ker(L), so of the form

$$h(t, x, v) = \left[\rho(t, x) + v \cdot u(t, x) + \frac{1}{2}(|v|^2 - d)\theta(t, x)\right]\mu(v)^{1/2},$$

- 2.  $(\rho_{\varepsilon}, u_{\varepsilon}, \theta_{\varepsilon})$  converges weakly\* in  $L_t^{\infty}(H_x^s)$  towards  $(\rho, u, \theta)$ ,
- 3.  $(\rho, u, \theta)$  satisfies the incompressible Navier-Stokes equations (4.1.4) as well as the Boussineq equation (4.1.5).

If such a result confirms the fact that one can derive the incompressible Navier-Stokes equations from the Boltzmann equation, it does unfortunately neither give us the continuity of h nor the initial condition verified by  $(\rho, u, \theta)$ , depending on  $(\rho_{in}, u_{in}, \theta_{in})$ , macroscopic quantities associated to  $h_{in}$ . Our next, and final step, is therefore to link the last two triplets and so to understand the convergence  $h_{\varepsilon} \to h$  more deeply. This is the purpose of the next, and last, theorem.

**Theorem 4.2.5** Consider  $s \ge s_0$  and  $h_{in}$  in  $H^s_{x,v}$  such that  $||h_{in}||_{\mathcal{H}^s_{\varepsilon}} \le \delta_s$ .

Then,  $(h_{\varepsilon})_{\varepsilon>0}$  exists for all  $0 < \varepsilon \leq \varepsilon_d$  and converges weakly\* in  $L^{\infty}_t(H^s_x L^2_v)$  towards h such that  $h \in Ker(L)$ , with  $\nabla_x \cdot u = 0$  and  $\rho + \theta = 0$ .

Furthermore,  $\int_0^T h dt$  belongs to  $H_x^s L_v^2$  and there exists C > 0 such that,

$$\left\|\int_0^{+\infty} h dt - \int_0^{+\infty} h_{\varepsilon} dt\right\|_{H^s_x L^2_v} \leqslant C \sqrt{\varepsilon \left| \ln(\varepsilon) \right|}.$$

One can have a strong convergence in  $L^2_{[0,T]}H^s_x L^2_v$  only if  $h_{in}$  is in Ker(L) with  $\nabla_x \cdot u_{in} = 0$  and  $\rho_{in} + \theta_{in} = 0$  (initial layer conditions).

Moreover, in that case we have

$$\|h - h_{\varepsilon}\|_{L^{2}_{[0,+\infty)}H^{s}_{x}L^{2}_{v}} \leq C\sqrt{\varepsilon} \, |ln(\varepsilon)|,$$

and for all  $\delta$  in [0,1], if  $h_{in}$  belongs to  $H_x^{s+\delta}L_v^2$ ,

$$\sup_{t \in [0,+\infty)} \|h - h_{\varepsilon}\|_{H^s_x L^2_v}(t) \leqslant C \varepsilon^{\min(\delta, 1/2)}$$

This theorem gives us strong convergences for  $(\rho_{\varepsilon}, u_{\varepsilon}, \theta_{\varepsilon})$  towards  $(\rho, u, \theta)$  but above all it gives us that  $(\rho, u, \theta)$  is the solution to the incompressible Navier-Stokes equations together with the Boussineq equation satisfying the initial conditions:

- $u(0,x) = Pu_{in}(x)$ , where  $Pu_{in}(x)$  is the divergence-free part of  $u_{in}(x)$ ,
- $\rho(0,x) = -\theta(0,x) = \frac{1}{2}(\rho_{in}(x) \theta_{in}(x)).$

Finally, we emphasize that in the case of initial data satisfying the initial layer conditions, the strong convergence in time requires a little bit more regularity from the initial data. This fact was already noticed in  $\mathbb{R}^d$  (see [10] Lemma 6.1) but overcome by considering weighted norms in velocity.

# 4.3 Toolbox: fluid projection and a priori energy estimates

In this section we are going to give some inequalities we are going to use and to refer to throughout the sequel. First we start with some properties concerning the projection in  $L_v^2$  onto Ker(L):  $\pi_L$ . Then, because we want to estimate all the terms appearing in the  $H_{x,v}^s$ -norm to estimate the functionals  $\mathcal{H}_{\varepsilon}^s$  and  $\mathcal{H}_{\varepsilon\perp}^s$ , we will give upper bound on their time derivatives. The proofs are only technical and the interested reader will find them in Appendix 4.B.

We are assuming there that L is having properties (H1'), (H2') and (H3), that  $\Gamma$  satisfies (H4) and (H5) and that  $0 < \varepsilon \leq 1$ .

# 4.3.1 Properties concerning the fluid projection $\pi_L$

We already know that L is acting on  $L_v^2$ , with  $\operatorname{Ker}(L) = \operatorname{Span}(\phi_1, \ldots, \phi_N)$ , with  $(\phi_i)_{1 \leq i \leq N}$ an orthonormal family, we obtain directly a useful formula for the orthogonal projection on  $\operatorname{Ker}(L)$  in  $L_v^2$ ,  $\pi_L$ :

$$\forall h \in L_v^2, \quad \pi_L(h) = \sum_{i=1}^N \left( \int_{\mathbb{R}^d} h \phi_i dv \right) \phi_i. \tag{4.3.1}$$

Plus, (H3) states that  $\phi_i = P_i(v)e^{-|v|^2/4}$ , where  $P_i$  is a polynomial. Therefore, direct computations and Cauchy-Schwarz inequality give that  $\pi_L$  is continuous on  $H^s_{x,v}$  with

$$\forall s \in \mathbb{N}, \exists C_{\pi s} > 0, \forall h \in H^s_{x,v}, \quad \|\pi_L(h)\|^2_{H^s_{x,v}} \leqslant C_{\pi s} \|h\|^2_{H^s_{x,v}}.$$
(4.3.2)

More precisely one can find that for all s in  $\mathbb{N}$ 

$$\forall |j| + |l| = s, \forall h \in H^s_{x,v}, \quad \left\| \partial^j_l \pi_L(h) \right\|_{L^2_{x,v}}^2 \leq C_{\pi s} \left\| \partial^0_l \pi_L(h) \right\|_{L^2_{x,v}}^2.$$
(4.3.3)

Finally, building the  $\Lambda$ -norm one can find that in all the collisional kinetic equations concerned here we have that

$$\exists C_{\pi} > 0, \forall h \in L^2_{x,v}, \ \|\pi_L(h)\|^2_{\Lambda} \leqslant C_{\pi} \ \|h\|^2_{L^2_{x,v}}.$$
(4.3.4)

Then we can also use the properties of the torus to obtain Poincare type inequalities. This can be very useful thanks to the next proposition, which is proved in Appendix 4.B.

**Proposition 4.3.1** Let a and b be in  $\mathbb{R}^*$  and consider the operator  $G = aL - bv \cdot \nabla_x$  acting on  $H^1_{x,v}$ .

If L satisfies (H1) and (H3) then

$$\operatorname{Ker}(G) = \operatorname{Ker}(L).$$

**Remark 4.3.2** In this proposition, Ker(G) has to be understood as linear combinations with constant coefficients of the functions  $\Phi_i$ . This subtlety has to be emphasized since in  $L^2_{x,v}$ , Ker(L) includes all linear combinations of  $\Phi_i$  but with coefficients being functions of x.

Therefore, if we define, for  $0 < \varepsilon \leq 1$ :

$$G_{\varepsilon} = \frac{1}{\varepsilon^2} L - \frac{1}{\varepsilon} v . \nabla_x,$$

then we have a nice description of  $\pi_{G_{\varepsilon}}$ :

$$\forall h \in L^2_{x,v}, \ \pi_{G_{\varepsilon}}(h) = \sum_{i=1}^{N} \left( \int_{\mathbb{T}^d} \int_{\mathbb{R}^d} h\phi_i \, dx dv \right) \phi_i.$$

That means that  $\pi_{G_{\varepsilon}}(h)$  is, up to a multiplicative constant, the mean of  $\pi_L(h)$  over the torus. We deduce that if h belongs to  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$ ,  $\pi_L(h)$  has zero mean on the torus and is an operator not depending on the x variable. Thus we can apply Poincaré inequality on the torus:

$$\forall h \in \operatorname{Ker}(G_{\varepsilon})^{\perp}, \quad \|\pi_{L}(h)\|_{L^{2}_{x,v}}^{2} \leq C_{p} \|\nabla_{x}\pi_{L}(h)\|_{L^{2}_{x,v}}^{2} \leq C_{p} \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2}.$$
(4.3.5)

### 4.3.2 A priori energy estimates

Our work in this chapter is to study the evolution of the norms involved in the definition of the operators  $\mathcal{H}^s_{\varepsilon}$  and  $\mathcal{H}^s_{\varepsilon\perp}$  and to combine them to obtain the results stated above. The Appendix 4.*B* contains the proofs, which are technical computations together with some choices of decomposition, of the following a priori estimates. Note that all the constants  $K_1$ ,  $K_{dx}$  and  $K_{s-1}$  used in the inequalities below are independent of  $\varepsilon$ ,  $\Gamma$  and g, and only depend constructively on the constants defined in the hypocoercivity assumptions or in the subsection above. The number e can be any positive real number and will be chosen later.

We would like to study both linear and non-linear models but they appeared to be very similar. In order to avoid long and similar inequalities we will write in parenthesis terms we need to add for the full model. Let g be a function in  $H^s_{x,v}$ . We now consider a function h in  $\operatorname{Ker}(G_{\varepsilon})^{\perp} \cap H^s_{x,v}$ , for some s in  $\mathbb{N}^*$ , which is solution of the linear (linearized) Boltzmann equation:

$$\partial_t h + \frac{1}{\varepsilon} v \cdot \nabla_x h = \frac{1}{\varepsilon^2} L(h) \left( + \frac{1}{\varepsilon} \Gamma(g, h) \right).$$

We remind the reader that the following notation is used:  $h^{\perp} = h - \pi_L(h)$ .

# **4.3.2.1** Time evolutions for quantities in $H^1_{x,v}$

We write the  $L^2_{x,v}$ -norm estimate

$$\frac{d}{dt} \|h\|_{L^{2}_{x,v}}^{2} \leqslant -\frac{\lambda}{\varepsilon^{2}} \left\|h^{\perp}\right\|_{\Lambda}^{2} \left(+\frac{1}{\lambda} \left(\mathcal{G}_{x}^{0}(g,h)\right)^{2}\right).$$

$$(4.3.6)$$

Then the time evolution of the x-derivatives

$$\frac{d}{dt} \left\| \nabla_x h \right\|_{L^2_{x,v}}^2 \leqslant -\frac{\lambda}{\varepsilon^2} \left\| \nabla_x h^{\perp} \right\|_{\Lambda}^2 \left( +\frac{1}{\lambda} \left( \mathcal{G}^1_x(g,h) \right)^2 \right), \tag{4.3.7}$$

and of the v-derivatives

$$\frac{d}{dt} \left\| \nabla_{v} h \right\|_{L^{2}_{x,v}}^{2} \leqslant \frac{K_{1}}{\varepsilon^{2}} \left\| h^{\perp} \right\|_{\Lambda}^{2} + \frac{K_{dx}}{\varepsilon^{2}} \left\| \nabla_{x} h \right\|_{L^{2}_{x,v}}^{2} - \frac{\nu_{3}^{\Lambda}}{\varepsilon^{2}} \left\| \nabla_{v} h \right\|_{\Lambda}^{2} \qquad (4.3.8)$$

$$\left( + \frac{3}{\nu_{3}^{\Lambda}} \left( \mathcal{G}_{x,v}^{1}(g,h) \right)^{2} \right).$$

Finally, we will need a control on the scalar product as well, as explained in the strategy subsection 4.1.3. Notice that we have some freedom as e can be any positive number.

$$\frac{d}{dt} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} \leqslant \frac{C^L e}{\varepsilon^3} \left\| \nabla_x h^\perp \right\|_{\Lambda}^2 - \frac{1}{\varepsilon} \left\| \nabla_x h \right\|_{L^2_{x,v}}^2 + \frac{2C^L}{e\varepsilon} \left\| \nabla_v h \right\|_{\Lambda}^2 \\
\left( + \frac{e}{C^L \varepsilon} \left( \mathcal{G}_x^1(g,h) \right)^2 \right).$$
(4.3.9)

# **4.3.2.2** Time evolutions for quantities in $H_{x,v}^s$

We consider multi-indexes j and l such that |j| + |l| = s. As in the previous case, we have a control on the time evolution of the pure x-derivatives,

$$\frac{d}{dt} \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 \leqslant -\frac{\lambda}{\varepsilon^2} \left\| \partial_l^0 h^\perp \right\|_{\Lambda}^2 \left( +\frac{1}{\lambda} \left( \mathcal{G}_x^s(g,h) \right)^2 \right).$$
(4.3.10)

In the case where  $|j| \ge 1$ , that is to say when we have at least one derivative in v, we obtained the following upper bound

$$\frac{d}{dt} \left\| \partial_l^j h \right\|_{L^2_{x,v}}^2 \leqslant -\frac{\nu_5^{\Lambda}}{\varepsilon^2} \left\| \partial_l^j h \right\|_{\Lambda}^2 + \frac{3(\nu_1^{\Lambda})^2 d}{\nu_5^{\Lambda}(\nu_0^{\Lambda})^2} \sum_{i,c_i(j)>0} \left\| \partial_{l+\delta_i}^{j-\delta_i} h \right\|_{\Lambda}^2 + \frac{K_{s-1}}{\varepsilon^2} \left\| h \right\|_{H^{s-1}_{x,v}}^2 \\
\left( + \frac{3}{\nu_5^{\Lambda}} \left( \mathcal{G}_{x,v}^s(g,h) \right)^2 \right).$$
(4.3.11)

We may find useful to consider the particular case where |j| = 1,

$$\frac{d}{dt} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h \right\|_{L^{2}_{x,v}}^{2} \leq -\frac{\nu_{5}^{\Lambda}}{\varepsilon^{2}} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h \right\|_{\Lambda}^{2} + \frac{3\nu_{1}^{\Lambda}}{\nu_{5}^{\Lambda}\nu_{0}^{\Lambda}} \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} + \frac{K_{s-1}}{\varepsilon^{2}} \left\| h \right\|_{H^{s-1}_{x,v}}^{2} \left( +\frac{3}{\nu_{5}^{\Lambda}} \left( \mathcal{G}_{x,v}^{s}(g,h) \right)^{2} \right).$$
(4.3.12)

Finally we will need the time evolution of the following scalar product:

$$\frac{d}{dt} \langle \partial_{l-\delta_{i}}^{\delta_{i}} h, \partial_{l}^{0} h \rangle_{L^{2}_{x,v}} \leqslant \frac{C^{L}e}{\varepsilon^{3}} \left\| \partial_{l}^{0} h^{\perp} \right\|_{\Lambda}^{2} - \frac{1}{\varepsilon} \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} + \frac{2C^{L}}{e\varepsilon} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h \right\|_{\Lambda}^{2} \\
\left( + \frac{e}{C^{L}\varepsilon} \left( \mathcal{G}_{x}^{s}(g,h) \right)^{2} \right),$$
(4.3.13)

where we still have some freedom as e is any positive number.

We just emphasize here that one can see that we were careful about which derivatives are involved in the terms that contain  $\Gamma$ . This is because our operator  $\|.\|_{\mathcal{H}^s_{\varepsilon}}$  controls the  $H^s_x(L^2_v)$ -norm by a mere constant whereas it controls the entire  $H^s_{x,v}$ -norm by a factor  $1/\varepsilon$ .

# **4.3.2.3** Time evolutions for orthogonal quantities in $H_{x,v}^s$

For the theorem 4.2.4 we are going to need four others inequalities which are a little bit more intricate as they need to know the shape of  $\pi_L$  as described in the subsection above. The proofs are written in Appendix 4.*B* and we are just looking at the whole equation in the setting g = h.

We want the time evolution of the v-derivatives of the orthogonal (microscopic) part of h, as suggested in [56] this allows us to really take advantage of the structure of the linear operator L on its orthogonal:

$$\frac{d}{dt} \left\| \nabla_{v} h^{\perp} \right\|_{L^{2}_{x,v}}^{2} \leqslant \frac{K_{1}^{\perp}}{\varepsilon^{2}} \left\| h^{\perp} \right\|_{\Lambda}^{2} + K_{dx}^{\perp} \left\| \nabla_{x} h \right\|_{L^{2}_{x,v}}^{2} - \frac{\nu_{3}^{\Lambda}}{2\varepsilon^{2}} \left\| \nabla_{v} h^{\perp} \right\|_{\Lambda}^{2} + \frac{3}{\nu_{3}^{\Lambda}} \left( \mathcal{G}_{x,v}^{1}(h,h) \right)^{2}.$$
(4.3.14)

Then we can have a new bound for the scalar product used before

$$\frac{d}{dt} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} \leqslant \frac{K^{\perp} e}{\varepsilon^3} \left\| \nabla_x h^{\perp} \right\|_{\Lambda}^2 + \frac{1}{4C_{\pi 1}C_{\pi}C_p e\varepsilon} \left\| \nabla_v h^{\perp} \right\|_{\Lambda}^2 
- \frac{1}{2\varepsilon} \left\| \nabla_x h \right\|_{L^2_{x,v}}^2 + \frac{4C_{\pi}}{\varepsilon} \left( \mathcal{G}^1_{x,v}(h,h) \right)^2,$$
(4.3.15)

where e is any number greater than 1.

As usual, we may need the same kind of bounds in higher degree Sobolev spaces. The reader may notice that the bounds we are about to write are more intricate than the ones in the previous section because they involve more terms with less derivatives. We consider multi-indexes j and l such that |j| + |l| = s. This time we really have to divide in two different cases.

Firstly when  $|j| \ge 2$ ,

$$\frac{d}{dt} \left\| \partial_{l}^{j} h^{\perp} \right\|_{L^{2}_{x,v}}^{2} \leqslant -\frac{\nu_{5}^{\Lambda}}{\varepsilon^{2}} \left\| \partial_{l}^{j} h^{\perp} \right\|_{\Lambda}^{2} + \frac{9(\nu_{1}^{\Lambda})^{2} d}{2(\nu_{0}^{\Lambda})^{2} \nu_{5}^{\Lambda}} \sum_{i,c_{i}(j)>0} \left\| \partial_{l+\delta_{i}}^{j-\delta_{i}} h^{\perp} \right\|_{\Lambda}^{2} + K_{dl}^{\perp} \sum_{|l'|\leqslant s-1} \left\| \partial_{l'}^{0} h \right\|_{L^{2}_{x,v}}^{2} + \frac{K_{s-1}^{\perp}}{\varepsilon^{2}} \left\| h^{\perp} \right\|_{H^{s-1}_{x,v}}^{2} + \frac{3}{\nu_{5}^{\Lambda}} \left( \mathcal{G}_{x,v}^{s}(h,h) \right)^{2}.$$
(4.3.16)

Then the case when |j| = 1

$$\frac{d}{dt} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h^{\perp} \right\|_{L^{2}_{x,v}}^{2} \leqslant -\frac{\nu_{5}^{\Lambda}}{\varepsilon^{2}} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h^{\perp} \right\|_{\Lambda}^{2} + K^{\perp}_{dl} \sum_{|l'|=s} \left\| \partial_{l'}^{0} h \right\|_{L^{2}_{x,v}}^{2} + \frac{K^{\perp}_{s-1}}{\varepsilon^{2}} \left\| h^{\perp} \right\|_{H^{s-1}_{x,v}}^{2} + \frac{3}{\nu_{5}^{\Lambda}} \left( \mathcal{G}^{s}_{x,v}(h,h) \right)^{2}.$$
(4.3.17)

Finally we give a new version of the control over the scalar product in higher Sobolev's spaces.

$$\frac{d}{dt} \langle \partial_{l-\delta_{i}}^{\delta_{i}}h, \partial_{l}^{0}h \rangle_{L^{2}_{x,v}} \leqslant \frac{\tilde{K}^{\perp}}{\varepsilon^{3}} e \left\| \partial_{l}^{0}h^{\perp} \right\|_{\Lambda}^{2} + \frac{1}{4C_{\pi s}C_{\pi}de\varepsilon} \left\| \partial_{l-\delta_{i}}^{\delta_{i}}h^{\perp} \right\|_{\Lambda}^{2} - \frac{1}{2\varepsilon} \left\| \partial_{l}^{0}h \right\|_{L^{2}_{x,v}}^{2} + \frac{1}{4d\varepsilon} \sum_{|l'|\leqslant s-1} \left\| \partial_{l}^{0}h \right\|_{L^{2}_{x,v}}^{2} + \frac{2C_{\pi}}{\varepsilon} \left( \mathcal{G}_{x,v}^{s}(h,h) \right)^{2},$$
(4.3.18)

for any  $e \ge 1$ .

# 4.4 Linear case: proof of Theorem 4.2.1

In this section we are looking at the linear equation

$$\partial_t h = G_{\varepsilon}(h), \text{ on } \mathbb{T}^d \times \mathbb{R}^d.$$

Theorem 4.2.1 will be proved by induction on s. We remind here the operator we will work with on  $H^s_{x,v}$ 

• in the case s = 1:

$$\|h\|_{\mathcal{H}^{1}_{\varepsilon}}^{2} = A \|h\|_{L^{2}_{x,v}}^{2} + \alpha \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2} + b\varepsilon^{2} \|\nabla_{v}h\|_{L^{2}_{x,v}}^{2} + a\varepsilon \langle \nabla_{x}h, \nabla_{v}h \rangle_{L^{2}_{x,v}},$$

• in the case s > 1:

$$\|h\|_{\mathcal{H}^{s}_{\varepsilon}}^{2} = \sum_{\substack{|j|+|l| \leq s \\ |j| \geq 1}} b_{j,l}^{(s)} \varepsilon^{2} \left\|\partial_{l}^{j}h\right\|_{L^{2}_{x,v}}^{2} + \sum_{|l| \leq s} \alpha_{l}^{(s)} \left\|\partial_{l}^{0}h\right\|_{L^{2}_{x,v}}^{2} + \sum_{\substack{|l| \leq s \\ i,c_{i}(l) > 0}} a_{i,l}^{(s)} \varepsilon \langle \partial_{l-\delta_{i}}^{\delta_{i}}h, \partial_{l}^{0}h \rangle_{L^{2}_{x,v}}.$$

The Theorem 4.2.1 only requires us to choose suitable coefficients that gives us the expected inequality and equivalence.

Consider  $h_{in}$  in  $H^s_{x,v} \cap \text{Dom}(G_{\varepsilon})$ . Let h be a solution of  $\partial_t h = G_{\varepsilon}(h)$  on  $\mathbb{T}^d \times \mathbb{R}^d$  such that  $h(0, \cdot, \cdot) = h_{in}(\cdot, \cdot)$ .

Notice that if  $h_{in}$  is in  $H^s_{x,v} \cap \text{Dom}(G_{\varepsilon}) \cap \text{Ker}(G_{\varepsilon})$  then we have that the associated solution remains the same in time:  $\partial_t h = 0$ . Therefore the fluid part of a solution does not evolve in time and so the semigroup is identity on  $\text{Ker}(G_{\varepsilon})$ . Besides, we can see directly from the definition and the adjointness property of L that  $h \in \text{Ker}(G_{\varepsilon})^{\perp}$  for all t if  $h_{in}$ belongs in  $\text{Ker}(G_{\varepsilon})^{\perp}$ .

Therefore, to prove the theorem it is enough to consider  $h_{in}$  in  $H^s_{x,v} \cap \text{Dom}(G_{\varepsilon}) \cap \text{Ker}(G_{\varepsilon})^{\perp}$ .

#### **4.4.1** The case s = 1

For now on we assume that our operator L satisfies the conditions (H1), (H2) and (H3) and that  $0 < \varepsilon \leq 1$ .

If (H3) holds for L then we have that  $\varepsilon^{-2}L$  is a non-positive self-adjoint operator on  $L^2_{x,v}$ . Moreover,  $\varepsilon^{-1}v \cdot \nabla_x$  is skew-symmetric on  $L^2_{x,v}$ . Therefore the  $L^2_{x,v}$ -norm decreases along the flow and it can be deduced that  $G_{\varepsilon}$  yields a  $C_0$ -semigroup on  $L^2_{x,v}$  for all positive  $\varepsilon$  (see [61] for general theory and [107] for its use in our case).

Using the toolbox, which is possible since h is in  $\text{Ker}(G_{\varepsilon})^{\perp}$  for all t, we just have to consider the linear combination  $A(4.3.6) + \alpha(4.3.7) + b\varepsilon^2(4.3.8) + a\varepsilon(4.3.9)$  to obtain

$$\frac{d}{dt} \|h\|_{\mathcal{H}_{\varepsilon}^{1}}^{2} \leqslant \frac{1}{\varepsilon^{2}} [bK_{1} - \lambda A] \left\|h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{\varepsilon^{2}} \left[C^{L}ea - \lambda\alpha\right] \left\|\nabla_{x}h^{\perp}\right\|_{\Lambda}^{2} \\
+ \left[\frac{2C^{L}a}{e} - b\nu_{3}^{\Lambda}\right] \left\|\nabla_{v}h\right\|_{\Lambda}^{2} + \left[bK_{dx} - a\right] \left\|\nabla_{x}h\right\|_{L_{x,v}^{2}}^{2}.$$
(4.4.1)

Then we make the following choices:

- 1. We fix b such that  $-\nu_3^{\Lambda}b < -1$ .
- 2. We fix A big enough such that  $[bK_1 \lambda A] \leq -1$ .
- 3. We fix a big enough such that  $[bK_{dx} a] \leq -1$ .
- 4. We fix e big enough such that  $\left[\frac{2C^{L_{a}}}{e} b\nu_{3}^{\Lambda}\right] \leqslant -1.$

5. We fix  $\alpha$  big enough such that  $\begin{bmatrix} C^L ea - \lambda \alpha \end{bmatrix} \leq -1$  and such that  $\begin{cases} a^2 \leq \alpha b \\ b \leq \alpha \end{cases}$ .

This leads to, because  $0 < \varepsilon \leq 1$ :

$$\frac{d}{dt} \|h\|_{\mathcal{H}^1_{\varepsilon}}^2 \leqslant -\left(\left\|h^{\perp}\right\|_{\Lambda}^2 + \left\|\nabla_x h^{\perp}\right\|_{\Lambda}^2 + \left\|\nabla_v h\right\|_{\Lambda}^2 + \left\|\nabla_x h\right\|_{L^2_{x,v}}^2\right).$$

Finally we can apply the Poincaré inequality (4.3.5) together with the equivalence of the  $L^2_{x,v}$ -norm and the  $\Lambda$ -norm on the fluid part  $\pi_L$ , equation (4.3.4), to get

$$\exists C, C' > 0, \quad \begin{cases} \|h\|_{\Lambda}^{2} & \leq C\left(\left\|h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{2} \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2}\right), \\ \|\nabla_{x}h\|_{\Lambda}^{2} & \leq C'\left(\left\|\nabla_{x}h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{2} \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2}\right). \end{cases}$$

Therefore we proved the following result:

$$\exists K > 0, \forall 0 < \varepsilon \leq 1, \quad \frac{d}{dt} \|h\|_{\mathcal{H}^{1}_{\varepsilon}}^{2} \leq -C_{G}^{(1)} \left( \|h\|_{\Lambda}^{2} + \|\nabla_{x,v}h\|_{\Lambda}^{2} \right).$$

With these constants,  $\|.\|_{\mathcal{H}^1_{\varepsilon}}$  is equivalent to

$$\left(\|h\|_{L^2_{x,v}}^2 + \|\nabla_x h\|_{L^2_{x,v}}^2 + \varepsilon^2 \|\nabla_v h\|_{L^2_{x,v}}^2\right)^{1/2}$$

since  $a^2 \leqslant \alpha b$  and  $b \leqslant \alpha$  and hence:

$$A \|h\|_{L^{2}_{x,v}}^{2} + \frac{b}{2} \left( \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2} + \varepsilon^{2} \|\nabla_{v}h\|_{L^{2}_{x,v}}^{2} \right) \leq \|h\|_{\mathcal{H}^{1}_{\varepsilon}}^{2}$$

and

$$\|h\|_{\mathcal{H}_{\varepsilon}^{1}}^{2} \leq A \|h\|_{L^{2}_{x,v}}^{2} + \frac{3\alpha}{2} \left( \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2} + \varepsilon^{2} \|\nabla_{v}h\|_{L^{2}_{x,v}}^{2} \right).$$

The results above gives us the expected theorem for s = 1.

# 4.4.2 The induction in higher order Sobolev spaces

Then we assume that the theorem is true up to the integer s - 1, s > 1. Then we suppose that L satisfies (H1'), (H2') and (H3) and we consider  $\varepsilon$  in (0, 1].

Let  $h_{in}$  be in  $H^s_{x,v} \cap \text{Dom}(G_{\varepsilon}) \cap \text{Ker}(G_{\varepsilon})^{\perp}$  and h be the solution of  $\partial_t h = G_{\varepsilon}(h)$  such that  $h(0, \cdot, \cdot) = h_{in}(\cdot, \cdot)$ .

As before, h belongs to  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  for all t and thus we can use the results given by the toolbox.

Thanks to the proof in the case s = 1 we know that we are able to handle the case where there is only a difference of one derivative between the number of derivatives in xand in v. Therefore, instead of working with the entire norm of  $H_{x,v}^s$ , we will look at an equivalent of the Sobolev semi-norm. We define:

$$F_{s}(t) = \sum_{\substack{|j|+|l|=s\\|j|\geq 2}} \varepsilon^{2} B \left\| \partial_{l}^{j} h \right\|_{L^{2}_{x,v}}^{2} + B' \sum_{\substack{|l|=s\\i,c_{i}(l)>0}} Q_{l,i}(t),$$
  
$$Q_{l,i}(t) = \alpha \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} + b\varepsilon^{2} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h \right\|_{L^{2}_{x,v}}^{2} + a\varepsilon \langle \partial_{l-\delta_{i}}^{\delta_{i}} h, \partial_{l}^{0} h \rangle_{L^{2}_{x,v}},$$

where the constants, strictly positive, will be chosen later.

Like in the section above, we shall study the time evolution of every term involved in  $F_s$  in order to bound above  $dF_s/dt(t)$  with negative coefficients.

#### 4.4.2.1 The time evolution of $Q_{l,i}$

We will first study the time evolution of  $Q_{l,i}$  for given |j| + |l| = s. The toolbox already gave us all the bounds we need and we just have to gather them in the following way:  $\alpha(4.3.10) + b\varepsilon^2(4.3.12) + a\varepsilon(4.3.13)$ . This leads to, because  $0 < \varepsilon \leq 1$ ,

$$\frac{d}{dt}Q_{l,i}(t) \leqslant \frac{1}{\varepsilon^2} \left[ C^L e a - \lambda \alpha \right] \left\| \partial_l^0 h^\perp \right\|_{\Lambda}^2 + \left[ \frac{2C^L a}{e} - \nu_5^{\Lambda} b \right] \left\| \partial_{l-\delta_i}^{\delta_i} h \right\|_{\Lambda}^2 \\
+ \left[ \frac{3\nu_1^{\Lambda}}{\nu_5^{\Lambda} \nu_0^{\Lambda}} b - a \right] \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 + K_{s-1} b \left\| h \right\|_{H^{s-1}_{x,v}}.$$

One can notice that, except for the last term, we have exactly the same kind of bound as in (4.4.1), in the proof of the case s = 1. Therefore we can choose  $\alpha$ , b, a, e, independently of  $\varepsilon$  such that it exists  $s_Q > 0$  and  $C_{s-1} > 0$  such that for all  $0 < \varepsilon \leq 1$ :

•  $Q_{l,i}(t) \sim \left\|\partial_l^0 h\right\|_{L^2_{x,v}}^2 + \varepsilon^2 \left\|\partial_{l-\delta_i}^{\delta_i} h\right\|_{L^2_{x,v}}^2,$ 

• 
$$\frac{d}{dt}Q_{l,i}(t) \leqslant -K_Q \left( \left\| \partial_l^0 h \right\|_{\Lambda}^2 + \left\| \partial_{l-\delta_i}^{\delta_i} h \right\|_{\Lambda}^2 \right) + C_{s-1} \left\| h \right\|_{H^{s-1}_{x,v}},$$

where we used (4.3.4) (equivalence of norms  $L^2_{x,v}$  and  $\Lambda$  on the fluid part) to get

$$\left\|\partial_{l}^{0}h\right\|_{\Lambda}^{2} \leq C'\left(\left\|\partial_{l}^{0}h^{\perp}\right\|_{\Lambda}^{2} + \left\|\partial_{l}^{0}h\right\|_{L^{2}_{x,v}}^{2}\right).$$

# **4.4.2.2** The time evolution of $F_s$ and conclusion

The last result about  $Q_{l,i}$  gives us that

$$F_{s}(t) \sim \sum_{|l|=s} \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} + \varepsilon^{2} \sum_{\substack{|l|+|j|=s\\|j|\ge 1}} \left\| \partial_{l}^{j} h \right\|_{L^{2}_{x,v}}^{2}$$

To study the time evolution of  $F_s$  we just need to combine the evolution of  $Q_{l,i}$  and the one of  $\left\|\partial_l^j h\right\|_{L^2_{x,v}}^2$  which is given in the toolbox by (4.3.11).

$$\frac{d}{dt}F_{s}(t) \leq \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} -\nu_{5}^{\Lambda}B \left\|\partial_{l}^{j}h\right\|_{\Lambda}^{2} + \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} \frac{3(\nu_{1}^{\Lambda})^{2}d}{\nu_{5}^{\Lambda}(\nu_{0}^{\Lambda})^{2}}B\varepsilon^{2}\sum_{i,c_{i}(j)>0} \left\|\partial_{l+\delta_{i}}^{j-\delta_{i}}h\right\|_{\Lambda}^{2} 
-K_{Q}B'\sum_{\substack{|l|=s\\i,c_{i}(l)>0}} \left(\left\|\partial_{l}^{0}h\right\|_{\Lambda}^{2} + \left\|\partial_{l-\delta_{i}}^{\delta_{i}}h\right\|_{\Lambda}^{2}\right) 
+ \left[\sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} K_{s-1}B + \sum_{\substack{|l|=s\\i,c_{i}(l)>0}} B'C_{s-1}\right] \left\|h\right\|_{H_{x,v}^{s-1}}^{2}.$$
(4.4.2)

Then we choose the following coefficients  $B=2/\nu_5^{\Lambda}$  and we can rearrange the sums to obtain

$$\frac{d}{dt}F_{s}(t) \leq \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} \left( \frac{6d(\nu_{1}^{\Lambda})^{2}}{(\nu_{5}^{\Lambda}\nu_{0}^{\Lambda})^{2}}\varepsilon^{2} - 2 \right) \left\| \partial_{l}^{j}h \right\|_{\Lambda}^{2} + \sum_{\substack{|j|+|l|=s\\|j|=1}} \left( \frac{6d(\nu_{1}^{\Lambda})^{2}}{(\nu_{5}^{\Lambda}\nu_{0}^{\Lambda})^{2}}\varepsilon^{2} - K_{Q}B' \right) \left\| \partial_{l}^{j}h \right\|_{\Lambda}^{2} + C_{+}^{(s-1)}(B') \left\| h \right\|_{H^{s-1}_{x,v}}.$$

Therefore we can choose the remaining coefficients:

1.  $\varepsilon_d = \min\left\{1, \sqrt{\frac{(\nu_5^{\Lambda}\nu_0^{\Lambda})^2}{6d(\nu_1^{\Lambda})^2}}\right\},\$ 

2. we fix B' big enough such that 
$$s_Q B' \ge 1$$
 and  $\left(\frac{6d(\nu_1^{\Lambda})^2}{(\nu_5^{\Lambda}\nu_0^{\Lambda})^2}\varepsilon_d^2 - K_Q B'\right) \leqslant -1$ .

Everything is now fixed in  $C_{+}^{(s-1)}(B')$  and therefore it is just a constant  $C_{+}^{(s-1)}$  that does not depend on  $\varepsilon$ . Therefore we then have the final result.

$$\forall 0 < \varepsilon \leqslant \varepsilon_d, \ \frac{d}{dt} F_s(t) \leqslant C_+^{(s-1)} \|h\|_{H^{s-1}_{x,v}}^2 - \left(\sum_{|j|+|l|=s} \left\|\partial_l^j h\right\|_{\Lambda}^2\right).$$

Then, we know that  $\left\|.\right\|_{\Lambda}$  controls the  $L^2\text{-norm.}$  And therefore:

$$\forall \, 0 < \varepsilon \leqslant \varepsilon_d \,, \, \frac{d}{dt} F_s(t) \leqslant C_+^{(s)} \left( \sum_{|j|+|l| \leqslant s-1} \left\| \partial_l^j h \right\|_{\Lambda}^2 \right) - \left( \sum_{|j|+|l|=s} \left\| \partial_l^j h \right\|_{\Lambda}^2 \right).$$

This inequality is true for all s and therefore we can take a linear combination of the  $F_s$  to obtain the following, where  $C_s$  is a constant that does not depend on  $\varepsilon$  since  $C_+^{(s)}$  does not depend on it.

$$\forall \, 0 < \varepsilon \leqslant \varepsilon_d \,, \, \frac{d}{dt} \left( \sum_{p=1}^n C_p F_p(t) \right) \leqslant -C_G^{(s)} \left( \sum_{|j|+|l||\leqslant s} \left\| \partial_l^j h \right\|_{\Lambda}^2 \right).$$

We can use the induction assumption from rank 1 up to rank s - 1 to find that this linear combination is equivalent to

$$\|.\|_{L^2_{x,v}}^2 + \sum_{|l| \leqslant s} \left\|\partial_l^0.\|_{L^2_{x,v}}^2 + \varepsilon^2 \sum_{\substack{|l|+|j| \leqslant s \\ |j| \geqslant 1}} \left\|\partial_l^j.\|_{L^2_{x,v}}^2\right\|$$

and so fits the expected requirements.

# 4.5 Estimate for the full equation: proof of Proposition 4.2.2

We will prove that proposition by induction on s. For now on we assume that L satisfies hypothesis (H1'), (H2') and (H3), that  $\Gamma$  satisfies properties (H4) and (H5) and we take g in  $H^s_{x,v}$ .

So we take  $h_{in}$  in  $H^s_{x,v} \cap \text{Ker}(G_{\varepsilon})^{\perp}$  and we consider the associated solution, denoted by h, of

$$\partial_t h + \frac{1}{\varepsilon} v \cdot \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(g, h).$$

One can notice that thanks to (H5) and the self-adjointness of L, h remains in Ker $(G_{\varepsilon})^{\perp}$  for all times.

Besides, while considering the time evolution we find a term due to  $G_{\varepsilon}$  and another due to  $\Gamma$ . Therefore, we will use the results found in the toobox but including the terms in parenthesis.

#### **4.5.1** The case s = 1

We want to study the following operator on  $H_{x,v}^s$ 

$$\|h\|_{\mathcal{H}^{1}_{\varepsilon}}^{2} = A \|h\|_{L^{2}_{x,v}}^{2} + \alpha \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2} + b\varepsilon^{2} \|\nabla_{v}h\|_{L^{2}_{x,v}}^{2} + a\varepsilon \langle \nabla_{x}h, \nabla_{v}h \rangle_{L^{2}_{x,v}}.$$

Therefore, using the toolbox we just have to consider the linear combination  $A(4.3.6) + \alpha(4.3.7) + b\varepsilon^2(4.3.8) + a\varepsilon(4.3.9)$  to yield

$$\frac{d}{dt} \|h\|_{\mathcal{H}^{1}_{\varepsilon}}^{2} \leqslant \frac{1}{\varepsilon^{2}} \left[ bK_{1} - \lambda A \right] \left\|h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{\varepsilon^{2}} \left[ C^{L}ea - \lambda\alpha \right] \left\|\nabla_{x}h^{\perp}\right\|_{\Lambda}^{2} \\
+ \left[ \frac{2C^{L}a}{e} - b\nu_{3}^{\Lambda} \right] \left\|\nabla_{v}h\right\|_{\Lambda}^{2} + \left[ bK_{dx} - a \right] \left\|\nabla_{x}h\right\|_{L^{2}_{x,v}}^{2} \\
+ \frac{A\nu_{1}^{\Lambda}}{\nu_{0}^{\Lambda}\lambda} \left(\mathcal{G}_{x}^{0}(g,h)\right)^{2} + \left[ \frac{\alpha\nu_{1}^{\Lambda}}{\nu_{0}^{\Lambda}\lambda} + \frac{\nu_{1}^{\Lambda}ea}{C^{L}\nu_{0}^{\Lambda}} \right] \left(\mathcal{G}_{x}^{1}(g,h)\right)^{2} \\
+ \frac{3\nu_{1}^{\Lambda}b}{\nu_{0}^{\Lambda}\nu_{3}^{\Lambda}} \varepsilon^{2} \left(\mathcal{G}_{x,v}^{1}(g,h)\right)^{2}.$$
(4.5.1)

One can see that we obtained exactly the same upper bound as in the proof of the previous theorem, equation (4.4.1), adding the terms involving  $\Gamma$  (remember that  $\mathcal{G}_x^s$  is increasing in s). Therefore we can make the same choices for A,  $\alpha$ , b, a and e, independently of  $\Gamma$  and g, to get that

$$\|h\|_{\mathcal{H}^{1}_{\varepsilon}}^{2} \sim \|h\|_{L^{2}_{x,v}}^{2} + \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2} + \varepsilon^{2} \|\nabla_{v}h\|_{L^{2}_{x,v}}^{2},$$

and that, once those parameters are fixed, there exist  $K_0^{(1)}$ ,  $K_1^{(1)}$ ,  $K_2^{(1)} > 0$  such that for all  $0 < \varepsilon \leq 1$ ,

$$\frac{d}{dt} \|h\|_{\mathcal{H}_{\varepsilon}^{1}}^{2} \leq -K_{0}^{(1)} \left( \|h\|_{\Lambda}^{2} + \|\nabla_{x,v}h\|_{\Lambda}^{2} \right) + K_{1}^{(1)} \left( \mathcal{G}_{x}^{1}(g,h) \right)^{2} + \varepsilon^{2} K_{2}^{(1)} \left( \mathcal{G}_{x,v}^{1}(g,h) \right)^{2},$$

which is the expected result in the case s = 1.

#### 4.5.2 The induction in higher order Sobolev spaces

Then we assume that the theorem is true up to the integer s - 1, s > 1. Then we suppose that L satisfies (H1'), (H2') and (H3) and we consider  $\varepsilon$  in (0, 1].

Since  $h_{in}$  is in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$ , h belongs to  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  for all t and so we can use the results given in the toolbox.

As in the proof in the linear case we define:

$$F_{s}(t) = \sum_{\substack{|j|+|l|=s\\|j|\geq 2}} \varepsilon^{2} B \left\| \partial_{l}^{j} h \right\|_{L^{2}_{x,v}}^{2} + B' \sum_{\substack{|l|=s\\i,c_{i}(l)>0}} Q_{l,i}(t),$$
$$Q_{l,i}(t) = \alpha \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} + b\varepsilon^{2} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h \right\|_{L^{2}_{x,v}}^{2} + a\varepsilon \langle \partial_{l-\delta_{i}}^{\delta_{i}} h, \partial_{l}^{0} h \rangle_{L^{2}_{x,v}},$$

where the constants, strictly positive, will be chosen later.

Like in the section above, we shall study the time evolution of every term involved in  $F_s$  in order to bound above  $dF_s/dt(t)$  with expected coefficients.

# **4.5.2.1** The time evolution of $Q_{l,i}$

We will first study the time evolution of  $Q_{l,i}$  for given |j| + |l| = s. The toolbox already gave us all the bounds we need and we just have to gather them in the following way:  $\alpha(4.3.10) + b\varepsilon^2(4.3.12) + a\varepsilon(4.3.13)$ . This leads to, because  $0 < \varepsilon \leq 1$ ,

$$\frac{d}{dt}Q_{l,i}(t) \leqslant \frac{1}{\varepsilon^2} \left[ C^L ea - \lambda \alpha \right] \left\| \partial_l^0 h^\perp \right\|_{\Lambda}^2 + \left[ \frac{2C^L a}{e} - \nu_5^\Lambda b \right] \left\| \partial_{l-\delta_i}^{\delta_i} h \right\|_{\Lambda}^2 \\
+ \left[ \frac{3\nu_1^\Lambda}{\nu_5^\Lambda \nu_0^\Lambda} b - a \right] \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 + K_{s-1} b \left\| h \right\|_{H^{s-1}_{x,v}} \\
+ \left[ \frac{\alpha\nu_1^\Lambda}{\nu_0^\Lambda \lambda} + \frac{\nu_1^\Lambda ea}{C^L \nu_0^\Lambda} \right] \left( \mathcal{G}_x^s(g,h) \right)^2 + \frac{3\nu_1^\Lambda b}{\nu_0^\Lambda \nu_5^\Lambda} \varepsilon^2 \left( \mathcal{G}_{x,v}^s(g,h) \right)^2.$$

One can notice that, except for the term in  $\|h\|_{H^{s-1}_{x,v}}$ , we have exactly the same kind of bound as in the case s = 1, given by (4.5.1). Therefore we can choose  $\alpha$ , b, a, e, independently of  $\varepsilon$ ,  $\Gamma$  and g such that it exists  $K_Q$ ,  $K_{\Gamma 1}$ ,  $K_{\Gamma 2} > 0$  and  $C_{s-1} > 0$  such that for all  $0 < \varepsilon \leq 1$ :

•  $Q_{l,i}(t) \sim \left\|\partial_l^0 h\right\|_{L^2_{x,v}}^2 + \varepsilon^2 \left\|\partial_{l-\delta_i}^{\delta_i} h\right\|_{L^2_{x,v}}^2,$ 

$$\frac{d}{dt}Q_{l,i}(t) \leqslant -K_Q \left( \left\| \partial_l^0 h \right\|_{\Lambda}^2 + \left\| \partial_{l-\delta_i}^{\delta_i} h \right\|_{\Lambda}^2 \right) + K_{\Gamma 1} \left( \mathcal{G}_x^s(g,h) \right)^2 
+ \varepsilon^2 K_{\Gamma 2} \left( \mathcal{G}_{x,v}^s(g,h) \right)^2 + C_{s-1} \left\| h \right\|_{H^{s-1}_{x,v}},$$

where we used (4.3.4) (equivalence of norms  $L^2_{x,v}$  and  $\Lambda$  on the fluid part) to get

$$\left\|\partial_{l}^{0}h\right\|_{\Lambda}^{2} \leq C'\left(\left\|\partial_{l}^{0}h^{\perp}\right\|_{\Lambda}^{2} + \left\|\partial_{l}^{0}h\right\|_{L^{2}_{x,v}}^{2}\right).$$

# **4.5.2.2** The time evolution of $F_s$ and conclusion

The last result about  $Q_{l,i}$  gives us that

$$F_{s}(t) \sim \sum_{|l|=s} \left\|\partial_{l}^{0}h\right\|_{L^{2}_{x,v}}^{2} + \varepsilon^{2} \sum_{\substack{|l|+|j|=s\\|j|\geqslant 1}} \left\|\partial_{l}^{j}h\right\|_{L^{2}_{x,v}}^{2}$$

so it remains to show that  $F_s$  satisfies the property describe by the theorem for some B and B'.

To study the time evolution of  $F_s$  we just need to combine the evolution of  $Q_{l,i}$  and the one of  $\left\|\partial_l^j h\right\|_{L^2_{x,v}}^2$  which is given in the toolbox by (4.3.11).

$$\frac{d}{dt}F_{s}(t) \leq \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} -\nu_{5}^{\Lambda}B \left\|\partial_{l}^{j}h\right\|_{\Lambda}^{2} + \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} \frac{3(\nu_{1}^{\Lambda})^{2}d}{\nu_{5}^{\Lambda}(\nu_{0}^{\Lambda})^{2}}B\varepsilon^{2}\sum_{i,c_{i}(j)>0} \left\|\partial_{l+\delta_{i}}^{j-\delta_{i}}h\right\|_{\Lambda}^{2} 
-K_{Q}B'\sum_{\substack{|l|=s\\i,c_{i}(l)>0}} \left(\left\|\partial_{l}^{0}h\right\|_{\Lambda}^{2} + \left\|\partial_{l-\delta_{i}}^{\delta_{i}}h\right\|_{\Lambda}^{2}\right) 
+ \left[\sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} K_{s-1}B + \sum_{\substack{|l|=s\\i,c_{i}(l)>0}} B'C_{s-1}\right] \left\|h\right\|_{H_{x,v}^{s-1}}^{2} 
+ \sum_{\substack{|l|=s\\i,c_{i}(l)>0}} B'K_{\Gamma1}\left(\mathcal{G}_{x}^{s}(g,h)\right)^{2} 
+ \varepsilon^{2} \left[\sum_{\substack{|l|=s\\i,c_{i}(l)>0}} B'K_{\Gamma2} + \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} \frac{3\nu_{1}^{\Lambda}}{\nu_{0}^{\Lambda}\nu_{5}^{\Lambda}}B\right] \left(\mathcal{G}_{x,v}^{s}(g,h)\right)^{2}.$$
(4.5.2)

One can easily see that, apart from the terms including  $\Gamma$ , we have exactly the same

bound as in the proof in the linear case, equation (4.4.2). Therefore we can choose B, B'and  $\varepsilon_d$  like we did, thus independent of  $\Gamma$  and g, to have for all  $0 < \varepsilon \leq \varepsilon_d$ 

$$\frac{d}{dt}F_{s}(t) \leqslant C_{+}^{(s-1)} \|h\|_{H^{s-1}_{x,v}}^{2} - \left(\sum_{|j|+|l|=s} \left\|\partial_{l}^{j}h\right\|_{\Lambda}^{2}\right) \\
+ \tilde{K}_{\Gamma 1} \left(\mathcal{G}_{x}^{s}(g,h)\right)^{2} + \varepsilon^{2} \tilde{K}_{\Gamma 2} \left(\mathcal{G}_{x,v}^{s}(g,h)\right)^{2},$$

with  $C^{(s-1)}_+$ ,  $\tilde{K}_{\Gamma 1}$  and  $\tilde{K}_{\Gamma 2}$  positive constants independent of  $\varepsilon$ ,  $\Gamma$  and g.

To conclude we just have to, as in the linear case, take a linear combination of the  $(F_p)_{p \leq s}$  and use the induction hypothesis (remember that both  $\mathcal{G}_{x,v}^p$  and  $\mathcal{G}_x^p$  are increasing functions of p) to obtain the expected result:  $\forall 0 < \varepsilon \leq \varepsilon_d$ ,

$$\frac{d}{dt} \left( \sum_{p=1}^{n} C_p F_p(t) \right) \leqslant - K_0^{(s)} \left( \sum_{|j|+|l||\leqslant s} \left\| \partial_l^j h \right\|_{\Lambda}^2 \right) + K_1^{(s)} \left( \mathcal{G}_x^s(g,h) \right)^2 \\ + \varepsilon^2 K_1^{(s)} \left( \mathcal{G}_{x,v}^s(g,h) \right)^2,$$

with this linear combination being equivalent to

$$\|\cdot\|_{L^2_{x,v}}^2 + \sum_{|l|\leqslant s} \left\|\partial_l^0\cdot\right\|_{L^2_{x,v}}^2 + \varepsilon^2 \sum_{\substack{|l|+|j|\leqslant s\\|j|\geqslant 1}} \left\|\partial_l^j\cdot\right\|_{L^2_{x,v}}^2$$

and so fits the expected requirements.

# 4.6 Existence and exponential decay: proof of Theorem 4.2.3

One can clearly see that solving the kinetic equation (4.1.2) in the setting  $f = \mu + \varepsilon \mu^{1/2} h$ is equivalent to solving the linearized kinetic equation (4.1.3) directly. Therefore we are going to focus only on this linearized equation.

The proof relies on the *a priori* estimate derived in the previous section. We shall use this inequality as a bootstrap to obtain first the existence of solutions thanks to an iteration scheme and then the exponential decay of those solutions, as long as the initial data is small enough.

# 4.6.1 Proof of the existence of global solutions

#### 4.6.1.1 Construction of solutions to a linearized problem

Here we will follow the classical method that is approximating our solution by a sequence of solutions of a linearization of our initial problem. Then we have to construct a functional on Sobolev spaces for which this sequence can be uniformly bounded in order to be able to extract a convergent subsequence.

Starting from  $h_0$  in  $H^s_{x,v} \cap \operatorname{Ker}(G_{\varepsilon})^{\perp}$ , to be define later, we define the function  $h_{n+1}$  in  $H^s_{x,v}$  by induction on  $n \ge 0$ :

$$\partial_t h_{n+1} + \frac{1}{\varepsilon} v \cdot \nabla_x h_{n+1} = \frac{1}{\varepsilon^2} L(h_{n+1}) + \frac{1}{\varepsilon} \Gamma(h_n, h_{n+1})$$

$$h_{n+1}(0, x, v) = h_{in}(x, v),$$
(4.6.1)

First we need to check that our sequence is well-defined.

**Lemma 4.6.1** Let L be satisfying assumptions (H1'), (H2') and (H3), and let  $\Gamma$  be satisfying assumptions (H4) and (H5).

Then, there exists  $0 < \varepsilon_d \leq 1$  such that for all  $s \geq s_0$  (defined in (H4)), there exists  $\delta_s > 0$  such that for all  $0 < \varepsilon \leq \varepsilon_d$ , if  $\|h_{in}\|_{\mathcal{H}^s_{\varepsilon}} \leq \delta_s$  then the sequence  $(h_n)_{n \in \mathbb{N}}$  is well-defined, continuous in time, in  $H^s_{x,v}$  and belongs to  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$ .

**Proof of Lemma** 4.6.1 By induction, let us suppose that for a fixed  $n \ge 0$  we have constructed  $h_n$  in  $H^s_{x,v}$ , which is true for  $h_{in}$ .

Using the previous notation one can see that we are in fact trying to solve the linear equation on the torus:

$$\partial_t h_{n+1} = G_{\varepsilon}(h_{n+1}) + \frac{1}{\varepsilon}\Gamma(h_n, h_{n+1})$$

with  $h_{in}$  as an initial data.

The existence of a solution  $h_{n+1}$  has already been shown for each equation covered by the hypocoercivity theory in the case  $\varepsilon = 1$  (see papers described in the introduction). It was proved by fixed point arguments applied to the Duhamel's formula. In order not to write several times the same estimates one may use our next lemma 4.6.2 together with the Duhamel's formula (instead of considering directly the time derivative of  $h_{n+1}$ ) to get a fixed point argument as long as  $h_{in}$  is small enough, the smallness not depending on  $\varepsilon$ .

As shown in the study of the linear part of the linearized model, under assumptions (H1'), (H2') and (H3)  $G_{\varepsilon}$  generates a  $C^0$ -semigroup on  $H^s_{x,v}$ , for all  $0 < \varepsilon \leq \varepsilon_d$ . Moreover, hypothesis (H4) shows us that  $\Gamma(h_n, \cdot)$  is a bounded linear operator from  $(H^s_{x,v}, E(\cdot))$  to  $(H^s_{x,v}, \|\cdot\|_{H^s_{x,v}})$ . Thus  $h_{n+1}$  is in  $H^s_{x,v}$ .

The belonging to  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  is direct since  $\Gamma(h_n, \cdot)$  is in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  (hypothesis (H5)).

Then we have to strongly bound the sequence, at least in short time, to have a chance to obtain a convergent subsequence, up to an extraction.

#### 4.6.1.2 Boundedness of the sequence

We are about to prove the global existence in time of solutions in  $C(\mathbb{R}^+, \|.\|_{\mathcal{H}^s_{\varepsilon}})$ . That will give us existence of solutions in standard Sobolev's spaces as long as the initial data is small enough in the sense of the  $\mathcal{H}^s_{\varepsilon}$ -norm, which is smaller than the standard  $H^s_{x,v}$ -norm. To achieve that we define a new functional on  $H^s_{x,v}$ 

$$E(h) = \sup_{t \in \mathbb{R}^+} \left( \|h(t)\|_{\mathcal{H}^s_{\varepsilon}}^2 + \int_0^t \|h(s)\|_{H^s_{\Lambda}}^2 \, ds \right).$$
(4.6.2)

**Lemma 4.6.2** Let L be satisfying assumptions (H1'), (H2') and (H3), and let  $\Gamma$  be satisfying assumptions (H4) and (H5).

Then there exists  $0 < \varepsilon_d \leq 1$  such that for all  $s \geq s_0$  (defined in (H4)) there exists  $\delta_s > 0$  independent of  $\varepsilon$ , such that for all  $0 < \varepsilon \leq \varepsilon_d$ , if  $\|h_{in}\|_{\mathcal{H}^s_{\varepsilon}} \leq \delta_s$  then

$$(E(h_n) \leqslant \delta_s) \Rightarrow (E(h_{n+1}) \leqslant \delta_s).$$

#### **Proof of Lemma 4.6.2** We let t > 0.

We know that  $h_{in}$  belongs to  $H^s_{x,v} \cap \operatorname{Ker}(G_{\varepsilon})^{\perp}$ . Moreover we have, thanks to Lemma 4.6.1, that  $(h_n)$  is well-defined, in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  and in  $H^s_{x,v}$ , since  $s \ge s_0$ . Moreover,  $\Gamma$  satisfies (H5). Therefore we can use the Proposition 4.2.2 to write, for  $\varepsilon \le \varepsilon_d$  ( $\varepsilon_d$  being the minimum between the one in Lemma 4.6.1 and the one in Proposition 4.2.2),

$$\frac{d}{dt} \|h_{n+1}\|_{\mathcal{H}^s_{\varepsilon}}^2 \leqslant -K_0^{(s)} \|h_{n+1}\|_{H^s_{\Lambda}}^2 + K_1^{(s)} \left(\mathcal{G}^s_x(h_n, h_{n+1})\right)^2 + \varepsilon^2 K_2^{(s)} \left(\mathcal{G}^s_{x,v}(h_n, h_{n+1})\right)^2.$$

We can use the hypothesis (H4) and the fact that

$$C_{m}\left(\left\|.\right\|_{L^{2}_{x,v}}^{2}+\sum_{|l|\leqslant s}\left\|\partial_{l}^{0}.\right\|_{L^{2}_{x,v}}^{2}+\varepsilon^{2}\sum_{\substack{|l|+|j|\leqslant s\\|j|\geqslant 1}}\left\|\partial_{l}^{j}.\right\|_{L^{2}_{x,v}}^{2}\right)\leqslant\left\|.\right\|_{\mathcal{H}^{s}_{\varepsilon}}^{2}\leqslant C_{M}\left\|.\right\|_{H^{s}_{x,v}},\qquad(4.6.3)$$

to get the following upper bounds:

$$(\mathcal{G}_{x}^{s}(h_{n},h_{n+1}))^{2} \leqslant \frac{C_{\Gamma}^{2}}{C_{m}} \left( \|h_{n}\|_{\mathcal{H}_{\varepsilon}^{s}}^{2} \|h_{n+1}\|_{H_{\Lambda}^{s}}^{2} + \|h_{n+1}\|_{\mathcal{H}_{\varepsilon}^{s}}^{2} \|h_{n}\|_{H_{\Lambda}^{s}}^{2} \right) \left( \mathcal{G}_{x,v}^{s}(h_{n},h_{n+1}) \right)^{2} \leqslant \frac{C_{\Gamma}^{2}}{C_{m}\varepsilon^{2}} \left( \|h_{n}\|_{\mathcal{H}_{\varepsilon}^{s}}^{2} \|h_{n+1}\|_{H_{\Lambda}^{s}}^{2} + \|h_{n+1}\|_{\mathcal{H}_{\varepsilon}^{s}}^{2} \|h_{n}\|_{H_{\Lambda}^{s}}^{2} \right).$$

Therefore we have the following upper bound, where  $K_1$  and  $K_2$  are constants independent of  $\varepsilon$ :

$$\frac{d}{dt} \|h_{n+1}\|_{\mathcal{H}_{\varepsilon}^{s}}^{2} \leqslant -K_{0}^{(s)} \|h_{n+1}\|_{H_{\Lambda}^{s}}^{2} + K_{1} \|h_{n}\|_{\mathcal{H}_{\varepsilon}^{s}}^{2} \|h_{n+1}\|_{H_{\Lambda}^{s}}^{2} + K_{2} \|h_{n+1}\|_{\mathcal{H}_{\varepsilon}^{s}}^{2} \|h_{n}\|_{H_{\Lambda}^{s}}^{2} \\
\leqslant \left[K_{1}E(h_{n}) - K_{0}^{(s)}\right] \|h_{n+1}\|_{H_{\Lambda}^{s}}^{2} + K_{2}E(h_{n+1}) \|h_{n}\|_{H_{\Lambda}^{s}}^{2}.$$

We consider now that  $E(h_n) \leq K_0^{(s)}/2K_1$ . We can integrate the equation above between 0 and t and one obtains

$$\|h_{n+1}\|_{\mathcal{H}^{s}_{\varepsilon}}^{2} + \frac{K_{0}^{(s)}}{2} \int_{0}^{t} \|h_{n+1}\|_{H^{s}_{\Lambda}}^{2} ds \leq \|h_{0}\|_{\mathcal{H}^{s}_{\varepsilon}}^{2} + KE(h_{n+1})E(h_{n}).$$

This is true for all t > 0, then we define  $C = \min\{1, K_0^{(s)}/2\}$ , if  $E(h_n) \leq C/2K$  we have

$$E(h_{n+1}) \leqslant \frac{2}{C} \|h_0\|_{\mathcal{H}^s_{\varepsilon}}^2.$$

Therefore choosing  $M^{(s)} = \min\{C/2K, K_0^{(s)}/2K_1\}$  and  $\delta_s \leq \min\{M^{(s)}C/2, M^{(s)}\}$  gives us the expected result.

# 

# 4.6.1.3 The global existence of solutions

Now we are able to prove the global existence result:

**Theorem 4.6.3** Let L be satisfying assumptions (H1'), (H2') and (H3), and let  $\Gamma$  be satisfying assumptions (H4) and (H5).

Then there exists  $0 < \varepsilon_d \leq 1$  such that for all  $s \geq s_0$  (defined in (H4)), there exists  $\delta_s > 0$ and for all  $0 < \varepsilon \leq \varepsilon_d$ :

If  $\|h_{in}\|_{\mathcal{H}^s_{\varepsilon}} \leq \delta_s$  then there exist a solution of (4.1.3) in  $C(\mathbb{R}^+, E(\cdot))$  and it satisfies, for some constant C > 0,

$$E(h) \leqslant C \|h_{in}\|_{\mathcal{H}^s_{\varepsilon}}^2.$$

**Proof of Theorem 4.6.3** Regarding Lemma 4.6.2, by induction we can strongly bound the sequence  $(h_n)_{n \in \mathbb{N}}$ , as long as  $E(h_0) \leq \delta_s$ , the constant being defined in Lemma 4.6.2. Therefore, defining  $h_0$  to be  $h_{in}$  at t = 0 and 0 elsewhere gives us  $E(h_0) = ||h_{in}||_{\mathcal{H}^s_{\varepsilon}} \leq \delta_s$ . Thus, we have the boundedness of the sequence  $(h_n)_{n \in \mathbb{N}}$  in  $L^{\infty}_t H^s_{x,v} \cap L^1_t H^s_{\Lambda}$ . By compact embeddings into smaller Sobolev's spaces (Rellich theorem) we can take the limit in (4.6.1) as n tends to  $+\infty$ , since  $G_{\varepsilon}$  and  $\Gamma$  are continuous. We obtain h a solution, in  $C(\mathbb{R}^+, E(\cdot))$ , to

$$\begin{cases} \partial_t h + \frac{1}{\varepsilon} v \cdot \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \\ h(0, x, v) = h_{in}(x, v). \end{cases}$$

# 4.6.2 Proof of the exponential decay

The function constructed above, h, is in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  for all  $0 < \varepsilon \leq 1$ . Moreover, this function is clearly a solution of the following equation:

$$\partial_t h = G_{\varepsilon}(h) + \frac{1}{\varepsilon} \Gamma(h, h),$$

with  $\Gamma$  satisfying (H5). Therefore, we can use the a priori estimate on solutions of the full perturbative model concerning the time evolution of the  $\mathcal{H}_{\varepsilon}^{s}$ -norm (where we will omit to write the dependence on s for clearness purpose), Proposition 4.2.2.

$$\frac{d}{dt} \left\|h\right\|_{\mathcal{H}^s_{\varepsilon}}^2 \leqslant -K_0 \left\|h\right\|_{H^s_{\Lambda}}^2 + K_1 \left(\mathcal{G}^s_x(h,h)\right)^2 + \varepsilon^2 K_2 \left(\mathcal{G}^s_{x,v}(h,h)\right)^2.$$

Moreover, using (4.6.3) and hypothesis (H4) to find:

$$\begin{aligned} \left(\mathcal{G}_x^s(h,h)\right)^2 &\leqslant \quad \frac{2C_{\Gamma}^2}{C_m} \left\|h\right\|_{\mathcal{H}_{\varepsilon}^s}^2 \left\|h\right\|_{H_{\Lambda}^s}^2 \\ \left(\mathcal{G}_{x,v}^s(h,h)\right)^2 &\leqslant \quad \frac{2C_{\Gamma}^2}{C_m \varepsilon^2} \left\|h\right\|_{\mathcal{H}_{\varepsilon}^s}^2 \left\|h\right\|_{H_{\Lambda}^s}^2. \end{aligned}$$

Hence, K being a constant independent of  $\varepsilon$ :

$$\frac{d}{dt} \left\|h\right\|_{\mathcal{H}^s_{\varepsilon}}^2 \leqslant \left(K \left\|h\right\|_{\mathcal{H}^s_{\varepsilon}}^2 - K_0\right) \left\|h\right\|_{H^s_{\Lambda}}^2.$$

Therefore, one can notice that if  $\|h_{in}\|_{\mathcal{H}^s_{\varepsilon}}^2 \leq K_0/2K$  then we have that  $\|h\|_{\mathcal{H}^s_{\varepsilon}}^2$  is decreasing in time. Hence, because the  $\Lambda$ -norm controls the  $L^2$ -norm which controls the  $\mathcal{H}$ -norm:

$$\begin{aligned} \frac{d}{dt} \left\|h\right\|_{\mathcal{H}^s_{\varepsilon}}^2 &\leqslant -\frac{K_0}{2} \left\|h\right\|_{H^s_{\Lambda}}^2 \\ &\leqslant -\frac{K_0}{2} \frac{\nu_0^{\Lambda}}{\nu_1^{\Lambda} C_M} \left\|h\right\|_{\mathcal{H}^s_{\varepsilon}}^2. \end{aligned}$$

Then we have directly, by Gronwall's lemma and setting  $\tau_s = K_0 \nu_0^{\Lambda} / 4 \nu_1^{\Lambda} C_M$ ,

$$\|h\|_{\mathcal{H}^s_{\varepsilon}}^2 \leqslant \|h_{in}\|_{\mathcal{H}^s_{\varepsilon}}^2 e^{-2\tau_s}$$

as long as  $\|h_{in}\|_{\mathcal{H}^s_{\varepsilon}}^2 \leq K_0/2K$ , which is the expected result with  $\delta_s \leq \sqrt{K_0/2K}$ .

# 4.7 Exponential decay of *v*-derivatives: proof of Theorem 4.2.4

In order to prove this theorem we are going to state a proposition giving an a priori estimate on a solution to the equation (4.1.3)

$$\partial_t h + \frac{1}{\varepsilon} v \cdot \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon} L(h) + \frac{1}{\varepsilon} \Gamma(h, h) \cdot \frac{1}{\varepsilon} \nabla_x h = \frac{1}{\varepsilon} L(h) \cdot \frac{1}{\varepsilon} \nabla$$

We remind the reader that we work in  $H^s_{x,v}$  with the following positive functional

$$\|\cdot\|_{\mathcal{H}^{s}_{\varepsilon\perp}}^{2} = \sum_{\substack{|j|+|l|\leqslant s\\|j|\geqslant 1}} b_{j,l}^{(s)} \left\|\partial_{l}^{j}(\mathrm{Id}-\pi_{L})\cdot\right\|_{L^{2}_{x,v}}^{2} + \sum_{|l|\leqslant s} \alpha_{l}^{(s)} \left\|\partial_{l}^{0}\cdot\right\|_{L^{2}_{x,v}}^{2} + \sum_{\substack{|l|\leqslant s\\i,c_{i}(l)>0}} a_{i,l}^{(s)}\varepsilon\langle\partial_{l-\delta_{i}}^{\delta_{i}}\cdot,\partial_{l}^{0}\cdot\rangle_{L^{2}_{x,v}}.$$

One can notice that if we choose coefficients  $(b_{j,l}^{(s)})$ ,  $(\alpha_l^{(s)})$ ,  $(a_{i,l}^{(s)}) > 0$  such that  $\|\cdot\|_{\mathcal{H}^s_{1\perp}}^2$  is equivalent to

$$\sum_{\substack{|j|+|l|\leqslant s\\|j|\geqslant 1}} \left\|\partial_l^j (\mathrm{Id} - \pi_L) \cdot \right\|_{L^2_{x,v}}^2 + \sum_{|l|\leqslant s} \left\|\partial_l^0 \cdot \right\|_{L^2_{x,v}}^2$$

then for all  $\varepsilon$  less than some  $\varepsilon_0$ ,  $\|\cdot\|^2_{\mathcal{H}^s_{\varepsilon\perp}}$  is also equivalent to the latter norm with equivalence coefficients not depending on  $\varepsilon$ .

Moreover, using equation (4.3.3), we have that

$$\left\|\partial_{l}^{j}h\right\|_{L^{2}_{x,v}}^{2} \leq C_{\pi s} \left\|\partial_{l}^{0}h\right\|_{L^{2}_{x,v}}^{2} + \left\|\partial_{l}^{j}h^{\perp}\right\|_{L^{2}_{x,v}}^{2} \leq 2C_{\pi s} \left\|\partial_{l}^{0}h\right\|_{L^{2}_{x,v}}^{2} + \left\|\partial_{l}^{j}h\right\|_{L^{2}_{x,v}}^{2},$$

and therefore

$$\sum_{\substack{|j|+|l|\leqslant s\\|j|\geqslant 1}} \left\|\partial_l^j (Id - \pi_L)\right\|_{L^2_{x,v}}^2 + \sum_{|l|\leqslant s} \left\|\partial_l^0 .\right\|_{L^2_{x,v}}^2$$

is equivalent to the standard Sobolev norm. Thus, we will just construct coefficients  $(b_{j,l}^{(s)})$ ,  $(\alpha_l^{(s)})$  and  $(a_{i,l}^{(s)})$  so that  $\|.\|_{\mathcal{H}^s_{1\perp}}^2$  is equivalent to the latter norm and then for  $\varepsilon$  small enough we will have the equivalence, not depending on  $\varepsilon$ , between  $\|\cdot\|_{\mathcal{H}^s_{\varepsilon\perp}}^2$  and the  $H^s_{x,v}$ -norm.

# 4.7.1 An a priori estimate

In this subsection we will prove the following proposition:

**Proposition 4.7.1** If L is a linear operator satisfying the conditions (H1'), (H2') and (H3) and  $\Gamma$  a bilinear operator satisfying (H5) then there exists  $0 < \varepsilon_d \leq 1$  such that for all s in  $\mathbb{N}^*$ ,

1. for  $h_{in}$  in  $Ker(G_{\varepsilon})^{\perp}$  if we have h an associated solution of

$$\partial_t h + \frac{1}{\varepsilon} v \cdot \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(h, h),$$

- 2. there exist  $K_0^{(s)}, K_1^{(s)}, (b_{j,l}^{(s)}), (\alpha_l^{(s)}), (a_{i,l}^{(s)}) > 0$  such that for all  $0 < \varepsilon \leq \varepsilon_d$ :
  - $\|\cdot\|_{\mathcal{H}^s_{\varepsilon^\perp}} \sim \|\cdot\|_{H^s_{x,v}},$
  - $\forall h_{in} \in H^s_{x,v} \cap \operatorname{Ker}(G_{\varepsilon})^{\perp}$ ,

$$\frac{d}{dt} \left\|h\right\|_{\mathcal{H}^{s}_{\varepsilon\perp}}^{2} \leqslant -K_{0}^{(s)} \left(\frac{1}{\varepsilon^{2}} \left\|h^{\perp}\right\|_{H^{s}_{\Lambda}}^{2} + \sum_{1 \leqslant |l| \leqslant s} \left\|\partial_{l}^{0}h\right\|_{L^{2}_{x,v}}^{2}\right) + K_{1}^{(s)} \left(\mathcal{G}^{s}_{x,v}(h,h)\right)^{2}.$$

**Remark 4.7.2** We notice here that in front of the microscopic part of h is a negative constant order  $-1/\varepsilon^2$  which is the same order than the control derived by Guo in [56] for his dissipation rate.

We will prove that proposition by induction on s.

So we take  $h_{in}$  in  $H^s_{x,v} \cap \operatorname{Ker}(G_{\varepsilon})^{\perp}$  and we consider the associated solution of (4.1.3), denoted by h. One can notice that thanks to (H5), h remains in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  for all times and thus we are allowed to use the inequalities given in the toolbox

#### **4.7.1.1** The case s = 1

In that case we have

$$\|h\|_{\mathcal{H}^{1}_{\varepsilon\perp}}^{2} = A \|h\|_{L^{2}_{x,v}}^{2} + \alpha \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2} + b \left\|\nabla_{v}h^{\perp}\right\|_{L^{2}_{x,v}}^{2} + a\varepsilon \langle \nabla_{x}h, \nabla_{v}h \rangle_{L^{2}_{x,v}},$$

with A,  $\alpha$ , b and a strictly positive.

Therefore we can study the time evolution of that operator acting on h by gathering results given in the toolbox. We simply take  $A(4.3.6) + \alpha(4.3.7) + b(4.3.14) + a\varepsilon(4.3.15)$ 

$$\frac{d}{dt} \|h\|_{\mathcal{H}^{1}_{\varepsilon\perp}}^{2} \leqslant \frac{1}{\varepsilon^{2}} \left[ K_{1}^{\perp}b - \lambda A \right] \left\|h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{\varepsilon^{2}} \left[ K^{\perp}ea - \lambda\alpha \right] \left\|\nabla_{x}h^{\perp}\right\|_{\Lambda}^{2} 
+ \frac{1}{\varepsilon^{2}} \left[ \frac{1}{4C_{\pi1}C_{\pi}C_{p}} \frac{a}{e} - b\frac{\nu_{3}^{\Lambda}}{2} \right] \left\|\nabla_{v}h^{\perp}\right\|_{\Lambda}^{2} + \left[ K_{dx}^{\perp}b - \frac{a}{2} \right] \left\|\nabla_{x}h\right\|_{L^{2}_{x,v}}^{2} 
+ K(A, \alpha, b, a) \left(\mathcal{G}_{x,v}^{1}(h, h)\right)^{2},$$
(4.7.1)

with s a fonction only depending on the coefficients appearing in hypocoercivity hypothesis and independent of  $\varepsilon$ .

We directly see that we have exactly the same kind of bound as the one we obtain while working on the a priori estimates for the operator  $||h||_{\mathcal{H}^{1}_{\varepsilon}}$ , equation (4.5.1). Therefore we can choose of coefficients A,  $\alpha$ , b, e and a in the same way (in the right order) and use the same inequalities to finally obtain the expected result:  $\exists s_0, K_1 > 0, \forall 0 < \varepsilon \leq 1$ ,

$$\frac{d}{dt} \|h\|_{\mathcal{H}^{1}_{\varepsilon\perp}}^{2} \leq -s_{0}^{(1)} \left(\frac{1}{\varepsilon^{2}} \left\|h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{\varepsilon^{2}} \left\|\nabla_{x}h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{\varepsilon^{2}} \left\|\nabla_{v}h^{\perp}\right\|_{\Lambda}^{2} + \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2}\right) \\
+ K_{1}^{(1)} \left(\mathcal{G}_{x,v}^{1}(h,h)\right)^{2},$$

with the constants  $s_0^{(1)}$  and  $K_1^{(1)}$  independent of  $\varepsilon$ , and  $\|h\|_{\mathcal{H}_{1\perp}}^2$  equivalent to  $\|h\|_{L^2_{x,v}}^2 + \|\nabla_v h^{\perp}\|_{L^2_{x,v}}^2$ . Therefore, for all  $\varepsilon$  small enough we have the expected result in the case s = 1.

#### 4.7.1.2 The induction in higher order Sobolev spaces

Then we assume that the theorem is true up to the integer s - 1, s > 1. Then we suppose that L satisfies (H1'), (H2') and (H3) and we consider  $\varepsilon$  in (0, 1].

Since  $h_{in}$  is in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$ , h belongs to  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  for all t and so we can use the results given in the toolbox.

As in the proofs of previous sections, we define on  $H^s_{x,v}$ :

$$F_{s}(t) = \sum_{\substack{|j|+|l|=s\\|j|\geq 2}} B \left\| \partial_{l}^{j} h^{\perp} \right\|_{L^{2}_{x,v}}^{2} + B' \sum_{\substack{|l|=s\\i,c_{i}(l)>0}} Q_{l,i}(t),$$
$$Q_{l,i}(t) = \alpha \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} + b \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h^{\perp} \right\|_{L^{2}_{x,v}}^{2} + a\varepsilon \langle \partial_{l-\delta_{i}}^{\delta_{i}} h, \partial_{l}^{0} h \rangle_{L^{2}_{x,v}}$$

where the constants, strictly positive, will be chosen later.

Like in the section above, we shall study the time evolution of every term involved in  $F_s$  in order to bound above  $\frac{dF_s}{dt}(t)$  with expected coefficients. However, in this subsection we will need to control all the  $Q_{l,i}$ 's in the same time rather than treating them separately as we did in the proof of Proposition (4.2.2), because the toolbox tells us that each  $Q_{l,i}$  is controlled by quantities appearing in the others.

# **4.7.1.3** The time evolution of $\sum Q_{l,i}$

Gathering the toolbox inequalities in the following way:  $\alpha(4.3.10) + b(4.3.17) + a\varepsilon(4.3.18)$ . This yields, because  $0 < \varepsilon \leq 1$  and  $\operatorname{Card}\{i, c_i(l) > 0\} \leq d$ ,

$$\begin{split} \frac{d}{dt} \left( \sum_{\substack{|l|=s\\i,c_i(l)>0}} Q_{l,i}(t) \right) &\leqslant \quad \frac{1}{\varepsilon^2} \left[ \tilde{K}^{\perp} ea - \lambda \alpha \right] \sum_{|l|=s} \left\| \partial_l^0 h^{\perp} \right\|_{\Lambda}^2 \\ &\quad + \frac{1}{\varepsilon^2} \left[ \frac{1}{4C_{\pi s} C_{\pi} d} \frac{a}{e} - \nu_5^{\Lambda} b \right] \sum_{\substack{|l|=s\\i,c_i(l)>0}} \left\| \partial_{l-\delta_i}^{\delta_i} h^{\perp} \right\|_{\Lambda}^2 \\ &\quad + \left[ K_{dl}^{\perp} db - \frac{a}{2} \right] \sum_{|l|=s} \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 + \frac{a}{4} \sum_{|l|\leqslant s-1} \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 \\ &\quad + \frac{bK_{s-1}^{\perp}}{\varepsilon^2} \left( \sum_{\substack{|l|+|j|=s\\i,c_i(l)>0}} 1 \right) \left\| h^{\perp} \right\|_{H^{s-1}}^2 + K(\alpha, b, a, e) \left( \mathcal{G}_{x,v}^s(h, h) \right)^2, \end{split}$$

with s a fonction only depending on the coefficients appearing in hypocoercivity hypothesis and independent of  $\varepsilon$ .

One can notice that except for the terms in  $\|h\|_{H^{s-1}_{x,v}}$  and  $\sum_{|l|\leqslant s-1} \|\partial_l^0 h\|_{L^2_{x,v}}^2$ , we have exactly the same bound as in the case s = 1, equation (4.7.1). Therefore we can choose  $\alpha$ , b, a, e, independently of  $\varepsilon$  and  $\Gamma$  such that there exist  $K'_0 > 0$ ,  $K'_1 > 0$  and  $C_0, C_1 > 0$  such that for all  $0 < \varepsilon \leqslant 1$ :

• 
$$\sum_{\substack{|l|=s\\i,c_i(l)>0}} Q_{l,i}(t) \sim \sum_{\substack{|l|=s\\i,c_i(l)>0}} \left( \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 + \left\| \partial_{l-\delta_i}^{\delta_i} h^\perp \right\|_{L^2_{x,v}}^2 \right),$$

$$\frac{d}{dt} \sum_{\substack{|l|=s\\i,c_i(l)>0}} Q_{l,i}(t) \leqslant - K'_0 \left( \frac{1}{\varepsilon^2} \sum_{|l|=s} \left\| \partial_l^0 h^\perp \right\|_{\Lambda}^2 + \frac{1}{\varepsilon^2} \sum_{\substack{|l|=s\\i,c_i(l)>0}} \left\| \partial_{l-\delta_i}^{\delta_i} h^\perp \right\|_{\Lambda}^2 + \sum_{\substack{|l|=s\\i,c_i(l)>0}} \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 \right) + \frac{C_0}{\varepsilon^2} \left\| h^\perp \right\|_{H^{s-1}_{x,v}}^2 + C_1 \sum_{\substack{|l|\leqslant s-1}} \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 + K'_1 \left( \mathcal{G}^s_{x,v}(h,h) \right)^2$$

# 4.7.1.4 The time evolution of $F_s$ and conclusion

We can finally obtain the time evolution of  $F_s$ , using  $\frac{d}{dt} \left\| \partial_l^j h^{\perp} \right\|_{L^2_{x,v}}^2$ , equation (4.3.16), so that there is no more  $\varepsilon$  in front of the  $\Gamma$  term:

$$\begin{split} \frac{d}{dt} F_{s}(t) &\leqslant -B \frac{\nu_{5}^{\lambda}}{\varepsilon^{2}} \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} \left\| \partial_{l}^{j} h^{\perp} \right\|_{\Lambda}^{2} + B \frac{9(\nu_{1}^{\Lambda})^{2} d}{2(\nu_{0}^{\Lambda})^{2} \nu_{5}^{\Lambda}} \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} \sum_{i,c_{i}(j)>0} \left\| \partial_{l+\delta_{i}}^{j} h^{\perp} \right\|_{\Lambda}^{2} \\ &-K_{0}^{\prime} B^{\prime} \left( \frac{1}{\varepsilon^{2}} \sum_{|l|=s} \left\| \partial_{l}^{0} h^{\perp} \right\|_{\Lambda}^{2} + \frac{1}{\varepsilon^{2}} \sum_{\substack{|l|=s\\i,c_{i}(l)>0}} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h^{\perp} \right\|_{\Lambda}^{2} + \sum_{|l|=s} \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} \right) \\ &+ \left( \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} BK_{dl}^{\perp} + B^{\prime} C_{1} \right) \sum_{|l|\leqslant s-1} \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} \\ &+ \frac{1}{\varepsilon^{2}} \left[ \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} BK_{s-1}^{\perp} + B^{\prime} C_{0} \right] \left\| h^{\perp} \right\|_{H^{s-1}_{x,v}}^{2} \\ &+ \left[ \sum_{\substack{|j|+|l|=s\\|j|\geqslant 2}} \frac{3B\nu_{1}^{\Lambda}}{\nu_{0}^{\Lambda}\nu_{5}^{\Lambda}} + B^{\prime} K_{1}^{\prime} \right] \left( \mathcal{G}_{x,v}^{s}(h,h) \right)^{2}, \end{split}$$

Therefore we obtain the same bound (except  $\sum_{|l| \leq s-1} \|\partial_l^0 h\|_{L^2_{x,v}}^2$ ) as in the proof of Proposition 4.2.2, equation (4.5.2), and so by choosing coefficients in the same way we have that there exist  $C^{(s)}_+ > 0$ ,  $0 < \varepsilon_d \leq 1$  and  $K^{(s*)}_1 > 0$ , none of them depending on  $\varepsilon$ , such that for all  $0 < \varepsilon \leq \varepsilon_d$ :

$$\frac{d}{dt}F_{s}(t) \leqslant C_{+}^{(s)}\left(\frac{1}{\varepsilon^{2}}\sum_{|j|+|l|\leqslant s-1}\left\|\partial_{l}^{j}h^{\perp}\right\|_{\Lambda}^{2}+\sum_{|l|\leqslant s-1}\left\|\partial_{l}^{0}h\right\|_{L_{x,v}^{2}}^{2}\right) -\left(\frac{1}{\varepsilon^{2}}\sum_{|j|+|l|=s}\left\|\partial_{l}^{j}h^{\perp}\right\|_{\Lambda}^{2}+\sum_{|l|=s}\left\|\partial_{l}^{0}h\right\|_{L_{x,v}^{2}}^{2}\right) +K_{1}^{(s*)}\left(\mathcal{G}_{x,v}^{s}(h,h)\right)^{2}.$$

This inequality is true for all s and therefore we can take a linear combination of the  $F_s$  to obtain the required result. Using the induction hypothesis on  $F_1$  up to  $F_{s-1}$  we also have the equivalence of norms.

# 4.7.2 The exponential decay: proof of Theorem 4.2.4

Thanks to Theorem 4.2.3, we know that we have a solution to the equation (4.1.3) for any given  $h_{in}$  small enough in the standard Sobolev norm. Call h the associated solution of  $h_{in} \in H^s_{x,v}$  to (4.1.3). Since the existence has been proved we can use the a priori estimate above and the Proposition 4.7.1.

Thus we have

$$\frac{d}{dt} \|h\|_{\mathcal{H}^s_{\varepsilon\perp}}^2 \leqslant -K_0^{(s)} \left( \frac{1}{\varepsilon^2} \left\|h^{\perp}\right\|_{H^s_{\Lambda}}^2 + \sum_{1 \leqslant |l| \leqslant s} \left\|\partial_l^0 h\right\|_{L^2_{x,v}}^2 \right) + K_1^{(s)} \left(\mathcal{G}^s_{x,v}(h,h)\right)^2.$$

As before we can use (4.3.4) (equivalence of norms  $L^2_{x,v}$  and  $\Lambda$  on the fluid part) to get, for |l| > 1,

$$\left\|\partial_l^0 h\right\|_{\Lambda}^2 \leqslant C'\left(\left\|\partial_l^0 h^{\perp}\right\|_{\Lambda}^2 + \left\|\partial_l^0 h\right\|_{L^2_{x,v}}^2\right),$$

and for the case  $|l| \leq 1$  we can apply the Poincare inequality (4.3.5) together with the equivalence of the  $L^2_{x,v}$ -norm and the  $\Lambda$ -norm on the fluid part  $\pi_L$ , (4.3.4) to get

$$\exists C, C' > 0, \begin{cases} \|h\|_{\Lambda}^{2} & \leq C\left(\left\|h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{2} \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2}\right), \\ \|\nabla_{x}h\|_{\Lambda}^{2} & \leq C'\left(\left\|\nabla_{x}h^{\perp}\right\|_{\Lambda}^{2} + \frac{1}{2} \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2}\right). \end{cases}$$

Then we get that

$$\frac{d}{dt} \|h\|_{\mathcal{H}^{s}_{\varepsilon\perp}}^{2} \leqslant -K_{0}^{(s)} \left( \sum_{\substack{|j|+|l|\leqslant s\\|j|\geqslant 1}} \left\|\partial_{l}^{j}h^{\perp}\right\|_{\Lambda}^{2} + \sum_{|l|\leqslant s} \left\|\partial_{l}^{0}h\right\|_{\Lambda}^{2} \right) + K_{1}^{(s)} \left(\mathcal{G}_{x,v}^{s}(h,h)\right)^{2} \\
\leqslant -K_{0}^{(s*)} \|h\|_{H^{s}_{\Lambda}}^{2} + K_{1}^{(s)} \left(\mathcal{G}_{x,v}^{s}(h,h)\right)^{2}.$$

Then for  $s \ge s_0$ , defined in (H4), and because  $\Gamma$  satisfies (H4) we can write

$$\frac{d}{dt} \|h\|_{\mathcal{H}^{s}_{\varepsilon\perp}}^{2} \leqslant \left(K_{1}^{(s)}C_{\Gamma}^{2} \|h\|_{H^{s}_{x,v}}^{2} - K_{0}^{(s*)}\right) \|h\|_{H^{s}_{\Lambda}}^{2}$$

Because  $\|h\|_{\mathcal{H}^s_{\varepsilon^+}}$  and  $\|h\|^2_{H^s_{x,v}}$  are equivalent, independently of  $\varepsilon$ , we finally have

$$\frac{d}{dt} \left\|h\right\|_{\mathcal{H}^s_{\varepsilon\perp}}^2 \leqslant \left(K_1^{(s)} C_{\Gamma}^2 C \left\|h\right\|_{\mathcal{H}^s_{\varepsilon\perp}}^2 - K_0^{(s*)}\right) \left\|h\right\|_{H^s_{\Lambda}}^2$$

Therefore if

$$\|h_{in}\|_{\mathcal{H}^s_{\varepsilon\perp}}^2 \leqslant \frac{K_0^{(s*)}}{2K_1^{(s)}C_{\Gamma}^2C}$$

we have that  $\|h\|_{\mathcal{H}^s_{\varepsilon\perp}}^2$  is always decreasing on  $\mathbb{R}^+$  and so for all t > 0

$$\frac{d}{dt} \left\|h\right\|_{\mathcal{H}^{s}_{\varepsilon\perp}}^{2} \leqslant -\frac{K_{0}^{(s*)}}{2K_{1}^{(s)}C_{\Gamma}^{2}C} \left\|h\right\|_{H^{s}_{\Lambda}}^{2}$$

And the  $H^s_{\Lambda}$ -norm controls the  $H^s_{x,v}$ -norm which is equivalent to the  $\mathcal{H}^s_{\varepsilon\perp}$ -norm. Thus applying Gronwall's lemma gives us the expected exponential decay.

# 4.8 Incompressible Navier-Stokes Limit: proof of Theorem 4.2.5

In this section we consider  $s \ge s_0$ ,  $0 < \varepsilon \le \varepsilon_d$  and we take  $h_{in}$  in  $H^s_{x,v}$  such that  $||h_{in}||_{\mathcal{H}^s_{\varepsilon}} \le \delta_s$ .

Therefore we know, thanks to theorem 4.2.3, that we have a solution  $h_{\varepsilon}$  to the linearized Boltzmann equation

$$\partial_t h_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x h_{\varepsilon} = \frac{1}{\varepsilon^2} L(h_{\varepsilon}) + \frac{1}{\varepsilon} \Gamma(h_{\varepsilon}, h_{\varepsilon}),$$

with  $h_{\varepsilon}(0, x, v) = h_{in}(x, v)$ . Moreover, we also know that  $(h_{\varepsilon})$  tends weakly-\* to h in  $L_t^{\infty}(H_x^s L_v^2)$ .

The first step towards the proof of Theorem 4.2.5 is to derived a convergence rate in finite time. Then, as described in Section 4.1.3, we shall interpolate this result with the exponential decay behaviour of our solutions in order to obtain a global in time convergence.

#### 4.8.1 A convergence in finite time

In Remark 4.8.13, we define  $V_T(\varepsilon)$  and prove the following result

$$\forall T > 0, \quad V_T(\varepsilon) = \sup_{t \in [0,T]} \|h_{\varepsilon} - h\|_{L^{\infty}_x L^2_v} \to 0, \text{ as } \varepsilon \to 0.$$

Thanks to this remark we can give an explicit convergence in finite time.

**Theorem 4.8.1** Consider  $s \ge s_0$  and  $h_{in}$  in  $H^s_{x,v}$  such that  $||h_{in}||_{\mathcal{H}^s_{\sigma}} \le \delta_s$ .

Then,  $(h_{\varepsilon})_{\varepsilon>0}$  exists for all  $0 < \varepsilon \leq \varepsilon_d$  and converges weakly\* in  $L^{\infty}_t(H^s_x L^2_v)$  towards h such that  $h \in Ker(L)$ , with  $\nabla_x \cdot u = 0$  and  $\rho + \theta = 0$ .

Furthermore,  $\int_0^T h dt$  belongs to  $H_x^s L_v^2$  and it exists C > 0 such that for all T > 0,

$$\left\|\int_0^T h dt - \int_0^T h_{\varepsilon} dt\right\|_{H^s_x L^2_v} \leq C \max\{\sqrt{\varepsilon}, \sqrt{T\varepsilon}, TV_T(\varepsilon)\}.$$

One can have a strong convergence in  $L^2_{[0,T]}H^s_x L^2_v$  only if  $h_{in}$  is in Ker(L) with  $\nabla_x \cdot u_{in} = 0$ and  $\rho_{in} + \theta_{in} = 0$  (initial layer conditions). Moreover, in that case we have, for all T > 0,

$$\|h - h_{\varepsilon}\|_{L^{2}_{[0,T]}H^{s}_{x}L^{2}_{v}} \leq C \max\{\sqrt{\varepsilon}, \sqrt{T}V_{T}(\varepsilon)\},\$$

and for all  $\delta$  in [0,1], if  $h_{in}$  belongs to  $H_x^{s+\delta}L_v^2$ ,

$$\sup_{t \in [0,T]} \|h - h_{\varepsilon}\|_{H^s_x L^2_v}(t) \leqslant C \max\{\varepsilon^{\min(\delta, 1/2)}, V_T(\varepsilon)\}$$

**Remark 4.8.2** We mention here that the obligation of an integration in time for non special initial condition is only due to the linear part  $\varepsilon^{-2}L - \varepsilon^{-1}v \cdot \nabla_x$ , whereas the case  $T = +\infty$  is prevented by the second order term  $\Gamma$ .

We proved in the linear case, theorem 4.2.1, that the linear operator  $G_{\varepsilon} = \varepsilon^{-2}L - \varepsilon^{-1}v \cdot \nabla_x$  generates a semigroup  $e^{tG_{\varepsilon}}$  on  $H^s_{x,v}$ . Therefore we can use Duhamel's principle to rewrite our equation under the following form, defining  $u_{\varepsilon} = \Gamma(h_{\varepsilon}, h_{\varepsilon})$ ,
$$h_{\varepsilon} = e^{tG_{\varepsilon}}h_{in} + \int_{0}^{t} \frac{1}{\varepsilon} e^{(t-s)G_{\varepsilon}}u_{\varepsilon}(s)ds$$
  
:=  $U^{\varepsilon}h_{in} + \Psi^{\varepsilon}(u_{\varepsilon}).$  (4.8.1)

The article by Ellis and Pinsky [39] gives us a Fourier theory in x of the semigroup  $e^{tG_{\varepsilon}}$  and therefore we are going to use it to study the strong limit of  $U^{\varepsilon}h_{in}$  and  $\Psi^{\varepsilon}(u_{\varepsilon})$  as  $\varepsilon$  tends to 0. We will denote by  $\mathcal{F}_x$  the Fourier transform in x on the torus (which is discrete) and n the discrete variable associated in  $\mathbb{Z}^d$ .

From [39], we are using Theorem 3.1, rewriten thanks with the Proposition 2.6 and the Appendix II with  $\delta = \lambda/4$  in Proposition 2.3, to get the following theorem

**Theorem 4.8.3** There exists  $n_0 \in \mathbb{R}^{*+}$ , there exists functions

- $\lambda_j : [-n_0, n_0] \longrightarrow \mathbb{C}, \ -1 \leq j \leq 2, \ C^{\infty}$
- $e_j: [-n_0, n_0] \times \mathbb{S}^{d-1} \longrightarrow L^2_v$ ,  $-1 \leq j \leq d, C^{\infty}$  in  $\zeta$  and  $C^0$  in  $\omega$ ,  $(\zeta, \omega) \longmapsto e_j(\zeta, \omega)$

such that

1. for all  $-1 \leq j \leq 2$ ,  $\lambda_j(\zeta) = i\alpha_j\zeta - \beta_j\zeta^2 + \gamma_j(\zeta)$ , where  $\alpha_j \in \mathbb{R}$ , with  $\alpha_0 = \alpha_2 = 0$ ,  $\beta_j < 0 \text{ and } |\gamma_j(\zeta)| \leq C_\gamma |\zeta|^3 \text{ with } |\gamma_j(\zeta)| \leq \frac{\beta_j}{2} |\zeta|^2$ ,

2. for all  $-1 \leq j \leq d$ 

• 
$$e_j(\zeta, \omega) = e_{0j}(\omega) + \zeta e_{1j}(\omega) + \zeta^2 e_{2j}(\zeta, \omega),$$
  
•  $e_{0-1}(\omega)(v) = e_{01}(-\omega)(v) = A\left(1 - \omega \cdot v + \frac{|v|^2 - d}{2}\right) \mu(v)^{1/2},$ 

3. we have  $e^{tG_{\varepsilon}} = \mathcal{F}_x^{-1} \hat{U}(t/\varepsilon^2, \varepsilon n, v) \mathcal{F}_x$  where

$$\hat{U}(t,n,v) = \sum_{j=-1}^{2} \hat{U}_{j}(t,n,v) + \hat{U}_{R}(t,n,v)$$

with the following properties

- for  $-1 \leq j \leq 2$ ,  $\hat{U}_j(t, n, v) = \chi_{|n| \leq n_0} e^{t\lambda_j(|n|)} P_j\left(|n|, \frac{n}{|n|}\right)(v)$ ,
- for  $-1 \leq j \leq 1, P_j\left(\left|n\right|, \frac{n}{\left|n\right|}\right) = e_j\left(\left|n\right|, \frac{n}{\left|n\right|}\right) \otimes e_j\left(\left|n\right|, \frac{-n}{\left|n\right|}\right)$ ,

• 
$$P_2\left(|n|, \frac{n}{|n|}\right) = \sum_{j=2}^{n} e_j\left(|n|, \frac{n}{|n|}\right) \otimes e_j\left(|n|, \frac{-n}{|n|}\right),$$
  
•  $for -1 \leq j \leq 2, P_j\left(|n|, \frac{n}{|n|}\right) = P_{0j}\left(\frac{n}{|n|}\right) + |n|P_{1j}\left(\frac{n}{|n|}\right) + |n|^2 P_{2j}\left(|n|, \frac{n}{|n|}\right),$ 

$$|||\hat{U}_R(t,n,v)|||_{L^2_w} \leqslant C_R e^{-\sigma t}$$

**Remark 4.8.4** This decomposition of the spectrum of the linear operator is based on a low and high frequencies decomposition. It shows that the spectrum of the whole operator can be viewed as a perturbation of the spectrum of the homogeneous linear operator. It can be divided into large eigenvalues, which are negative and therefore create a strong semigroup property for the remainder term, and small eigenvalues around the origin that are smooth perturbations of the homogeneous ones.

This theorem gives us all the tools we need to study the convergence as  $\varepsilon$  tends to 0 since we have an explicit form for the Fourier transform of the semigroup. We also know that this semigroup commutes with the pure x-derivatives. Therefore, studying the convergence in the  $L_x^2 L_v^2$ -norm will be enough to obtain the desired result in the  $H_x^s L_v^2$ -norm. We are going to prove the following convergences in the different settings stated by Theorem 4.2.5

- 1.  $U^{\varepsilon}h_{in}$  tends to  $V(t, x, v)h_{in}$  with  $V(0, x, v)h_{in} = V(0)(h_{in})(x, v)$  where V(0) the projection on the subset of Ker(L) consisting in functions g such that  $\nabla_x \cdot u_g = 0$  and  $\rho_g + \theta_g = 0$ ,
- 2.  $\Psi^{\varepsilon}(u_{\varepsilon})$  converges to  $\Psi(h,h)$  with  $\Psi(h,h)(t=0) = 0$ .

## 4.8.1.1 Study of the linear part

We remind here that we have

$$U^{\varepsilon}h_{in} = \mathcal{F}_x^{-1}\hat{U}^{\varepsilon}(t,n,v)\hat{h}_{in}(n,v)$$

with

$$\hat{U}^{\varepsilon}(t,n,v) = \sum_{j=-1}^{2} \hat{U}_{j}^{\varepsilon}(t,n,v) + \hat{U}_{R}^{\varepsilon}(t,n,v),$$

$$\hat{U}_{j}^{\varepsilon}(t,n,v) = \chi_{|\varepsilon n| \leqslant n_{0}} e^{\frac{i\alpha_{j}t|n|}{\varepsilon} -\beta_{j}t|n|^{2} + \frac{t}{\varepsilon^{2}}\gamma_{j}(|\varepsilon n|)} \left[ P_{0j}\left(\frac{n}{|n|}\right) + \varepsilon |n| \tilde{P}_{1j}\left(|\varepsilon n|, \frac{n}{|n|}\right) \right].$$

We can decompose  $\hat{U}_{i}^{\varepsilon}$  into four different terms

$$\hat{U}_{j}^{\varepsilon}(t,n,v) = e^{\frac{i\alpha_{j}t|n|}{\varepsilon} - \beta_{j}t|n|^{2}} P_{0j}\left(\frac{n}{|n|}\right) \\
+\chi_{|\varepsilon n|\leqslant n_{0}} e^{\frac{i\alpha_{j}t|n|}{\varepsilon} - \beta_{j}t|n|^{2}} \left(e^{\frac{t}{\varepsilon^{2}}\gamma_{j}(|\varepsilon n|)} - 1\right) P_{0j}\left(\frac{n}{|n|}\right) \\
+\chi_{|\varepsilon n|\leqslant n_{0}} e^{\frac{i\alpha_{j}t|n|}{\varepsilon} - \beta_{j}t|n|^{2} + \frac{t}{\varepsilon^{2}}\gamma_{j}(|\varepsilon n|)}\varepsilon |n| \tilde{P}_{1j}\left(|\varepsilon n|, \frac{n}{|n|}\right) \\
+\left(\chi_{|\varepsilon n|\leqslant n_{0}} - 1\right) e^{\frac{i\alpha_{j}t|n|}{\varepsilon} - \beta_{j}t|n|^{2}} P_{0j}\left(\frac{n}{|n|}\right). \\
= U_{0j}^{\varepsilon} + U_{1j}^{\varepsilon} + U_{2j}^{\varepsilon} + U_{3j}^{\varepsilon}.$$
(4.8.2)

**Remark 4.8.5** One can notice that  $U_{00}^{\varepsilon}$  and  $U_{02}^{\varepsilon}$  do not depend on  $\varepsilon$ , since  $\alpha_0 = \alpha_2 = 0$ .

We are going to study each of these four terms in two different lemmas and then add a last lemma to deal with the remainder term  $U_R h_{in}$ . The lemmas will be proven in Appendix 4.C.

**Lemma 4.8.6** For  $\alpha_j \neq 0$   $(j = \pm 1)$  we have that it exists  $C_0 > 0$  such that for all  $T \in [0, +\infty]$ 

$$\left\|\int_0^T U_{0j}^{\varepsilon} h_{in} dt\right\|_{L^2_x L^2_v}^2 \leqslant C_0 \varepsilon^2 \left\|h_{in}\right\|_{L^2_x L^2_v}^2.$$

Moreover we have a strong convergence in the  $L^2_{[0,+\infty)}L^2_xL^2_v$ -norm if and only if  $h_{in}$  satisfies  $\nabla_x \cdot u_{in} = 0$  and  $\rho_{in} + \theta_{in} = 0$ . In that case we have  $U^{\varepsilon}_{0j}h_{in} = 0$ .

**Lemma 4.8.7** For  $-1 \leq j \leq 2$  and for  $1 \leq l \leq 3$  we have that the three following inequalities hold for  $U_{lj}^{\varepsilon}$ 

•  $\exists C_l > 0, \forall T > 0, \quad \left\| \int_0^T U_{lj}^{\varepsilon} h_{in} dt \right\|_{L^2_x L^2_v}^2 \leqslant C_l \varepsilon^2 \left\| h_{in} \right\|_{L^2_x L^2_v}^2,$ 

• 
$$\exists C'_l > 0$$
,  $\left\| U_{lj}^{\varepsilon} h_{in} \right\|_{L^2_{[0,+\infty)} L^2_x L^2_v}^2 \leq C'_l \varepsilon^2 \left\| h_{in} \right\|_{L^2_x L^2_v}^2$ ,

• 
$$\forall \delta \in [0,1], \exists C_{\delta}^{(l)} > 0, \forall t > 0, \quad \left\| U_{lj}^{\varepsilon} h_{in}(t) \right\|_{L^{2}_{x}L^{2}_{v}}^{2} \leqslant C_{\delta}^{(l)} \varepsilon^{2\delta} \left\| h_{in} \right\|_{H^{\delta}_{x}L^{2}_{v}}^{2}.$$

Lemma 4.8.8 For the remainder term we have the two following inequalities

• 
$$\exists C_4 > 0, \forall T > 0, \quad \left\| \int_0^T U_R^{\varepsilon} h_{in} dt \right\|_{L^2_x L^2_v}^2 \leqslant C_4 T \varepsilon^2 \|h_{in}\|_{L^2_x L^2_v}^2,$$
  
•  $\exists C'_4 > 0, \quad \left\| U_R^{\varepsilon} h_{in} \right\|_{L^2_{[0,+\infty)} L^2_x L^2_v}^2 \leqslant C'_4 \varepsilon^2 \|h_{in}\|_{L^2_x L^2_v}^2,$ 

• 
$$\forall t_0 > 0, \exists C_r > 0, \forall t > t_0, \quad \|U_R h_{in}(t)\|_{L^2_x L^2_v}^2 \leq \frac{C_r}{\sqrt{t_0}} \varepsilon \|h_{in}\|_{L^2_x L^2_v}^2.$$

Moreover, the strong convergence up to  $t_0 = 0$  is possible if and only if  $h_{in}$  is in Ker(L). In that case we have

$$\forall \delta \in [0,1], \exists C_{\delta}^{(R)} > 0, \forall t > 0, \| U_{R}^{\varepsilon} h_{in} \|_{L^{2}_{x}L^{2}_{v}}^{2} \leqslant C_{\delta}^{(R)} \varepsilon^{2\delta} \| h_{in} \|_{H^{\delta}_{x}L^{2}_{v}}^{2}.$$

Therefore, gathering lemmas 4.8.6, 4.8.7 and 4.8.8 and reminding Remark 4.8.5, we proved that, as  $\varepsilon$  tends to 0,  $(e^{tG_{\varepsilon}}h_{in})$  converges to

$$V(t,x,v)h_{in}(x,v) = \mathcal{F}_x^{-1} \left[ e^{-\beta_0 t|n|^2} P_{00}\left(\frac{n}{|n|}\right) + e^{-\beta_2 t|n|^2} P_{02}\left(\frac{n}{|n|}\right) \right] \mathcal{F}_x h_{in}.$$
 (4.8.3)

The convergence is strong when we consider the average in time and is strong in  $L_t^2 H_x^s L_v^2$  (and in  $C([0, +\infty), H_x^s L_v^2)$  if  $h_{in}$  is in  $H_x^{s+0} L_v^2$ ) if an only if both conditions found in Lemma 4.8.6 and Lemma 4.8.8 are satisfied. That is to say  $h_{in}$  belongs to Ker(L) with  $\nabla_x \cdot u_{in} = 0$  and  $\rho_{in} + \theta_{in} = 0$ .

Moreover this also allows us to see that  $V(0, x, v)h_{in} = V(0)(h_{in})(x, v)$  where V(0) is the projection on the subset of Ker(L) consisting in functions g such that  $\nabla_x \cdot u_g = 0$  and  $\rho_g + \theta_g = 0$ .

## 4.8.1.2 Study of the bilinear part

We recall here that  $u_{\varepsilon} = \Gamma(h_{\varepsilon}, h_{\varepsilon})$ . Therefore, by hypothesis (H5),  $u_{\varepsilon}$  belongs to  $\operatorname{Ker}(L)^{\perp}$ . Then we know that for all  $-1 \leq j \leq 2$ ,  $P_{0j}\left(\frac{n}{|n|}\right)$  is a projection onto a subspace of  $\operatorname{Ker}(L)$ . Therefore we have that, in the Fourier space,

$$P_j\left(\left|\varepsilon n\right|, \frac{n}{|n|}\right)\hat{u}_{\varepsilon} = \left|\varepsilon n\right| P_{1j}\left(\frac{n}{|n|}\right)\hat{u}_{\varepsilon} + \left|\varepsilon n\right|^2 P_{2j}\left(\left|\varepsilon n\right|, \frac{n}{|n|}\right)\hat{u}_{\varepsilon}.$$

Thus, recalling that

$$\Psi^{\varepsilon}(u_{\varepsilon}) = \int_0^t \frac{1}{\varepsilon} e^{(t-s)G_{\varepsilon}} u_{\varepsilon}(s) ds,$$

we can decompose it

$$\Psi^{\varepsilon}(u_{\varepsilon}) = \sum_{j=-1}^{2} \psi_{j}^{\varepsilon}(u_{\varepsilon}) + \psi_{R}^{\varepsilon}(u_{\varepsilon}),$$

with

$$\begin{split} \psi_{j}^{\varepsilon}(u_{\varepsilon}) &= \mathcal{F}_{x}^{-1}\chi_{|\varepsilon n| \leq n_{0}} \int_{0}^{t} e^{\frac{i\alpha_{j}(t-s)|n|}{\varepsilon} -\beta_{j}(t-s)|n|^{2} + \frac{t-s}{\varepsilon^{2}}\gamma_{j}(|\varepsilon n|)} |n| \left(P_{1j} + \varepsilon |n| P_{2j}\right) \hat{u}_{\varepsilon}(s) ds. \\ &:= \psi_{0j}^{\varepsilon}(u_{\varepsilon}) + \psi_{1j}^{\varepsilon}(u_{\varepsilon}) + \psi_{2j}^{\varepsilon}(u_{\varepsilon}) + \psi_{3j}^{\varepsilon}(u_{\varepsilon}), \end{split}$$

where we have used the same decomposition as in the linear case, equation (4.8.2), substituting t by t - s,  $P_{0j}$  by  $|n| P_{1j}$  and  $\tilde{P}_{1j}$  by  $|n| P_{2j}$ . And

$$\psi_R^{\varepsilon}(u_{\varepsilon}) = \int_0^t \frac{1}{\varepsilon} U_R^{\varepsilon}(t-s) u_{\varepsilon}(s) ds.$$

Like the linear case, Remark 4.8.5,  $\psi_{00}^{\varepsilon}$  and  $\psi_{02}^{\varepsilon}$  do not depend on  $\varepsilon$  and we are going to prove the convergence towards  $\Psi(u) = \mathcal{F}_x^{-1} \left[ \psi_{00}^{\varepsilon}(u) + \psi_{02}^{\varepsilon}(u) \right] \mathcal{F}_x$ , where  $u = \Gamma(h, h)$ . To establish such a result we are going to study each term in three different lemmas and then a fourth one will deal with the remainder term. The lemmas will be proven in Appendix 4.*C*.

**Lemma 4.8.9** For  $\alpha_j \neq 0$   $(j = \pm 1)$  we have the following inequality for  $\psi_{0j}^{\varepsilon}$ :

$$\exists \tilde{C}_0 > 0, \forall T > 0, \quad \left\| \int_0^T \psi_{0j}^{\varepsilon}(u_{\varepsilon}) dt \right\|_{L^2_x L^2_v}^2 \leqslant \tilde{C}_0 T^2 \varepsilon^2 E(h_{\varepsilon})^2.$$

**Remark 4.8.10** We know that  $(h_{\varepsilon})_{\varepsilon>0}$  is bounded in  $L_t^{\infty} H_x^s L_v^2$  (see theorems 4.2.3 and 4.2.4).

This remark gives us the strong convergence to 0 of the average in time and the strong convergence to 0 without averaging in time as long as  $h_{in}$  belongs to Ker(L) in Lemma 4.8.9.

**Lemma 4.8.11** For  $-1 \leq j \leq 2$  and for  $1 \leq l \leq 3$  we have that the three following inequalities hold for  $\psi_{lj}^{\varepsilon}$ 

- $\exists \tilde{C}_l > 0, \forall T > 0, \quad \left\| \int_0^T \psi_{lj}^{\varepsilon}(u_{\varepsilon}) dt \right\|_{L^2_x L^2_v}^2 \leq \tilde{C}_l T \varepsilon^2 E(h_{\varepsilon})^2,$
- $\exists \tilde{C}'_l > 0, \forall T > 0, \quad \left\| \psi_{lj}^{\varepsilon}(u_{\varepsilon}) \right\|_{L^2_{[0,T]}L^2_x L^2_v}^2 \leqslant \tilde{C}'_l \varepsilon^2 E(h_{\varepsilon})^2,$
- $\forall |\delta| \in [0,1], \exists C^{(l)}_{\delta} > 0, \forall T > 0, \quad \left\| \psi^{\varepsilon}_{lj}(u_{\varepsilon})(T) \right\|^{2}_{L^{2}_{x}L^{2}_{v}} \leqslant C^{(l)}_{\delta} \varepsilon^{2\delta} E(\partial^{0}_{\delta}h_{\varepsilon})^{2}.$

Lemma 4.8.12 For the remainder term we have the three following inequalities

- $\exists \tilde{C}_4 > 0, \forall T > 0, \quad \left\| \int_0^T \psi_R^{\varepsilon}(u_{\varepsilon}) dt \right\|_{L^2_x L^2_v}^2 \leqslant \tilde{C}_4 T \varepsilon E(h_{\varepsilon})^2,$
- $\exists \tilde{C}'_4 > 0, \forall T > 0, \quad \|\psi^{\varepsilon}_R(u_{\varepsilon})\|^2_{L^2_{[0,T]}L^2_xL^2_v} \leqslant \tilde{C}'_4 \varepsilon E(h_{\varepsilon})^2,$
- $\exists \tilde{C}_4'' > 0, \forall T > 0, \quad \|\psi_R^{\varepsilon}(u_{\varepsilon})(T)\|_{L^2_x L^2_x}^2 \leq \tilde{C}_4'' \varepsilon E(h_{\varepsilon})^2.$

Gathering all Lemmas 4.8.9, 4.8.11 and 4.8.12 gives us the strong convergence of  $\Psi^{\varepsilon}(u_{\varepsilon}) - \Psi(u_{\varepsilon})$  towards 0, thanks to Remark 4.8.10. It remains to prove that we have indeed the expected convergences of  $\Psi(u_{\varepsilon})$  towards  $\Psi(u)$  as  $\varepsilon$  tends to 0.

We start this last step by a quick remark relying on Sobolev embeddings and giving us a strong convergence of  $h_{\varepsilon}$  towards h in  $L^{\infty}_{[0,T]}L^{\infty}_{x}L^{2}_{v}$ , for T > 0.

**Remark 4.8.13** We know that  $h_{\varepsilon} \to h$  weakly-\* in  $L_t^{\infty} H_x^s L_v^2$ , for  $s \ge s_0 > d/2$ . But we also proved that for all t > 0 that  $(h_{\varepsilon})_{\varepsilon}$  is bounded in  $H_x^s L_v^2$ . Therefore the sequence  $(\|h_{\varepsilon}\|_{L_v^2}, \varepsilon > 0)$  is bounded in  $H_x^s$  and therefore converges strongly in  $H_x^{s'}$  for all s' < s. But, by triangular inequality it comes that

$$\left| \|h_{\varepsilon}\|_{H^{s'}_{x}L^{2}_{v}} - \|h\|_{H^{s'}_{x}L^{2}_{v}} \right| \leq \left\| \|h_{\varepsilon}\|_{L^{2}_{v}} - \|h\|_{L^{2}_{v}} \right\|_{H^{s'}_{x}}.$$

This means that we also have that  $\lim_{\varepsilon \to 0} \|h_{\varepsilon}\|_{H^{s'}_x L^2_v} = \|h\|_{H^{s'}_x L^2_v}$ . The space  $H^{s'}_x L^2_v$  is a Hilbert space and  $h_{\varepsilon}$  tends weakly to h in it, therefore the last result gives us that in fact  $h_{\varepsilon}$  tends strongly to h in  $H^{s'}_x L^2_v$ .

This result is for all t > 0 and all  $s' \leq s$ . Furthermore, s > d/2 and so we can choose s' > d/2. By Sobolev's embedding we obtain that  $h_{\varepsilon}$  tends strongly to h in  $L_x^{\infty} L_v^2$ , for all t > 0. Reminding that  $h_{\varepsilon} \to h$  weakly-\* in  $L_t^{\infty} H_x^s L_v^2$  and we obtain that we have

$$\forall T > 0, \quad V_T(\varepsilon) = \sup_{t \in [0,T]} \|h_{\varepsilon} - h\|_{L^{\infty}_x L^2_v} \to 0, \text{ as } \varepsilon \to 0.$$

**Lemma 4.8.14** We have the following rate of convergence:

- $\exists \tilde{C}_5 > 0, \forall T > 0, \left\| \int_0^T \Psi(u_\varepsilon) dt \int_0^T \Psi(u) dt \right\|_{L^2_x L^2_v}^2 \leq \tilde{C}_5 T^2 V_T(\varepsilon)^2,$
- $\exists \tilde{C}'_5 > 0, \forall T > 0, \|\Psi(u_{\varepsilon}) \Psi(u_{\varepsilon})\|^2_{L^2_{[0,T]}L^2_x L^2_v} \leq \tilde{C}'_5 T V_T(\varepsilon)^2,$
- $\exists \tilde{C}_5'' > 0, \forall T > 0, \|\Psi(u_{\varepsilon}) \Psi(u_{\varepsilon})\|_{L^2_x L^2_v}^2(T) \leqslant \tilde{C}_5'' V_T(\varepsilon)^2.$

Thus, those Lemmas, combined with the study of the linear case (Lemmas 4.8.6, 4.8.7 and 4.8.8) prove the Theorem 4.2.5 with the rate of convergence being the maximum of each rate of convergence. Moreover we have proved

$$h(t, x, v) = V(t, x, v)h_{in}(x, v) + \Psi(t, x, v)(\Gamma(h, h)).$$

## 4.8.2 Proof of Theorem 4.2.5

Thanks to Theorem 4.8.1 we can control the convergence of  $h_{\varepsilon}$  towards h for any finite time T. Then, thanks to the uniqueness property of Theorem 2.1 and the control on the remainder of Theorem 2.3 in [56], in the case of a hard potential collision kernel, one has

$$\forall T > 0, V_T(\varepsilon) \leq C_V \varepsilon.$$

Finally, thanks to Theorem 4.2.3, we have the exponential decay for both  $h_{\varepsilon}$  and h, leading to

$$\|h_{\varepsilon} - h\|_{H^s_x L^2_v} \leqslant 2 \,\|h_{in}\|_{\mathcal{H}^s_{\varepsilon}} \, e^{-\tau_s T}.$$

We define

$$T_M = -\frac{1}{\tau_s} \ln\left(\frac{\varepsilon}{2 \|h_{in}\|_{\mathcal{H}^s_{\varepsilon}}}\right)$$

to get that

$$\forall T \ge T_M, \quad \|h_{\varepsilon} - h\|_{H^s_r L^2_v} \le \varepsilon.$$

This conclude the proof Theorem 4.2.5, by applying Theorem 4.8.1 to  $T_M$ .

# Appendices

# 4.A Validation of the assumptions

As said in the introduction, all the hypocoercivity theory assumptions hold for several different kinetic models. One can find the proof of the assumptions (H1), (H2), (H3), (H1') and (H2') in [82] directly for the linear relaxation (see also [24]), the semi-classical relaxation (see also [86]), the linear Fokker-Planck equation, the Boltzmann equation with hard potential and angular cutoff and the Landau equation with hard and moderately soft potential (both studied in a constructive way in [4] and [79], for the spectral gaps, see also [53] and [54] for the Cauchy problems):

• The Linear Relaxation

$$\partial_t f + v \cdot \nabla_x f = \frac{1}{\varepsilon} \left[ \left( \int_{\mathbb{R}}^d f(t, x, v_*) dv_* \right) \mu(v) - f \right],$$

• The Semi-classical Relaxation

$$\partial_t f + v \cdot \nabla_x f = \frac{1}{\varepsilon} \int_{\mathbb{R}^d} \left[ \mu (1 - \delta f) f_* - \mu_* (1 - \delta f_*) f \right] dv_*,$$

• The Linear Fokker-Planck Equation

$$\partial_t f + v \cdot \nabla_x f = \frac{1}{\varepsilon} \nabla_v \cdot \left( \nabla_v f + f v \right),$$

• The Boltzmann Equation with hard potential and angular cutoff

$$\partial_t f + v \cdot \nabla_x f = \frac{1}{\varepsilon} \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} b(\cos\theta) |v - v_*|^{\gamma} \left[ f' f'_* - f f_* \right] dv_* d\sigma,$$

• The Landau Equation with hard and moderately soft potential

$$\partial_t f + v \cdot \nabla_x f = \frac{1}{\varepsilon} \nabla_v \cdot \left( \int_{\mathbb{R}^d} \Phi(v - v_*) |v - v_*|^{\gamma + 2} \left[ f_*(\nabla f) - f(\nabla f)_* \right] \right).$$

Assumption (H4) is clearly satisfied by the first three as in that case we have either  $\|.\|_{\Lambda_v} = \|.\|_{L_v^2}$  or  $\Gamma = 0$  (see [82]). Moreover, (H5) is obvious in the case of a linear equation. It thus remains to prove properties (H5) for the semi-classical relaxation and (H4) and (H5) for the Boltzmann equation and the Landau equation (since our property (H4) is slightly different from (H4) in [82]).

## 4.A.1 The semi-classical relaxation

In the case of the semi-classical relaxation, the linearization is slightly different. Indeed, the unique global equilibrium associated to an initial data  $f_0$  is (assuming some initial bounds, see [82])

$$f_{\infty} = \frac{\kappa_{\infty}\mu}{1 + \delta\kappa_{\infty}\mu},$$

where  $\kappa_{\infty}$  depends on  $f_0$ .

Thus, we are no longer in the case of a global equilibrium being a Maxwellian. However, a good way of linearizing this equation is (see [82]) considering

$$f = f_{\infty} + \varepsilon \, \frac{\sqrt{\kappa_{\infty}\mu}}{1 + \delta \kappa_{\infty}\mu} \, h.$$

Using such a linearization instead of the one used all along this chapter yields the same general equation (4.1.3) with L and  $\Gamma$  satisfying all the requirements (see [82]). Indeed, one may find that  $\operatorname{Ker}(L) = \operatorname{Span}(f_{\infty}/\sqrt{\mu})$  and then notice that this is not of the form

needed in assumption (H3). However, this is bounded by  $e^{-|v|^2/4}$  and therefore we are still able to use the toolbox (section 4.3, thus all the theorems.

Let us look at the bilinear operator to show that it fulfils hypothesis (H5). A straightforward computation gives us the definition of  $\Gamma$ ,

$$\Gamma(g,h) = \frac{\delta\sqrt{\kappa_{\infty}}}{2} \int_{\mathbb{R}^d} \sqrt{\mu_*} \frac{\mu_* - \mu}{1 + \varepsilon \kappa_{\infty} \mu_*} [hg_* + h_*g] dv_*.$$

Then, multiplying by a function f, integrating over  $\mathbb{R}^d$  and looking at the change of variable  $(v, v_*) \to (v_*, v)$  yields

$$\langle \Gamma(g,h), f \rangle_{L^2_v} = \frac{\delta \sqrt{\kappa_\infty}}{4} \int_{\mathbb{R}^d \times \mathbb{R}^d} (\mu_* - \mu) (gh_* + g_*h) \left[ f \frac{\sqrt{\mu_*}}{1 + \delta \kappa_\infty \mu_*} - f_* \frac{\sqrt{\mu}}{1 + \delta \kappa_\infty \mu} \right] dv dv_*.$$

Therefore, taking f in Ker(L) gives us the expected property.

## 4.A.2 Boltzmann operator with angular cutoff and hard potential

Notice that, compared to [82], we defined  $\Gamma$  in a way that it is symmetric which gives us, using the fact that  $\mu_*\mu = \mu'_*\mu'$ ,

$$\Gamma(g,h) = \frac{1}{2} \int_{\mathbb{R}^d \times (S)^{d-1}} B(\mu^{1/2})_* [g'_*h' + g'h'_* - g_*h - gh_*] dv_* d\sigma_*$$

## 4.A.2.1 Orthogonality to Ker(L): (H5)

A well-known property (see [46] for instance) tells us that for all  $\phi$  in  $L_v^2$  decreasing fast enough at infinity and for all  $\psi$  in  $L_v^2$  one has

$$\int_{\mathbb{R}^d} \Gamma(g,h)(v)\psi(v)dv = \frac{1}{8} \qquad \int_{(\mathbb{R}^d)^2 \times \mathbb{S}^{d-1}} B[g'_*h' + g'h'_* - g_*h - gh_*] \\ ((\mu^{1/2})_*\psi + (\mu^{1/2})\psi_* - (\mu^{1/2})'_*\psi' - (\mu^{1/2})'\psi'_*)dvdv_*d\sigma$$

As shown in [30] or [82] we have that  $\operatorname{Ker}(L) = \operatorname{Span}(1, v_1, \dots, v_d, |v|^2) \mu^{1/2}$  and therefore taking  $\psi$  to be each of these kernel functions gives us (H5).

## 4.A.2.2 Controlling derivatives: (H4)

To prove (H4) we can define

$$\Gamma^{+}(g,h) = \int_{\mathbb{R}^{d} \times (S)^{d-1}} B(\mu^{1/2})_{*} g'_{*} h' \, dv_{*} d\sigma,$$
  
 
$$\Gamma^{-}(g,h) = -\int_{\mathbb{R}^{d} \times (S)^{d-1}} B(\mu^{1/2})_{*} g_{*} h \, dv_{*} d\sigma.$$

By using the change of variable  $u = v - v_*$  we end up with  $\theta$  being a function of u and  $\sigma$  and  $v' = v + f_1(u, \sigma)$  and  $v'_* = v + f_2(u, \sigma)$ ,  $f_1$  and  $f_2$  being functions. Therefore we can make this change of variable, take j and l such that  $|j| + |l| \leq s$  and differentiate our operator  $\Gamma^-$ .

$$\partial_{l}^{j}\Gamma^{-}(g,h) = -\frac{1}{2} \sum_{\substack{j_{0}+j_{1}+j_{2}=j\\l_{1}+l_{2}=l}} \int_{\mathbb{R}^{d}\times\mathbb{S}^{d-1}} b(\cos\theta) |u|^{\gamma} \partial_{0}^{j_{0}} \left(\mu(v-u)^{1/2}\right) \partial_{l_{1}}^{j_{1}}g_{*} \ \partial_{l_{2}}^{j_{2}}h \ dud\sigma.$$

Then we can easily compute that, C being a generic constant,

$$\left|\partial_0^{j_0}\left(\mu(v-u)^{1/2}\right)\right| \leqslant C\mu(v-u)^{1/4}$$

Moreover, we are in the case where  $\gamma > 0$  and therefore we have

$$|u|^{\gamma}\mu(v-u)^{1/4} \leqslant C(1+|v|)^{\gamma}\mu(v-u)^{1/8}.$$

Combining this and the fact that  $|b| \leq C_b$  (angular cutoff considered here), multiplying by a function f and integrating over  $\mathbb{T}^d \times \mathbb{R}^d$  yields, using Cauchy-Schwarz two times,

$$\begin{split} \left| \langle \partial_l^j \Gamma^-(g,h), f \rangle_{L^2_{x,v}} \right| &\leqslant C \sum_{\substack{j_0+j_1+j_2=j\\l_1+l_2=l}} \int_{\mathbb{T}^d \times \mathbb{R}^d} (1+|v|)^\gamma \left| \partial_{l_2}^{j_2} h \right| |f| \left( \int_{\mathbb{R}^d} \mu_*^{1/8} \left| \partial_{l_1}^{j_1} g_* \right| dv_* \right) dv dx \\ &\leqslant \mathcal{G}^s(g,h) \left\| f \right\|_{\Lambda}, \end{split}$$

with

$$\mathcal{G}^{s}(g,h) = C \sum_{|j_{1}|+|l_{1}|+|j_{2}|+|l_{2}|\leqslant s} \left[ \int_{\mathbb{T}^{d}} \left\| \partial_{l_{2}}^{j_{2}} h \right\|_{\Lambda_{v}}^{2} \left\| \partial_{l_{1}}^{j_{1}} g \right\|_{L_{v}^{2}}^{2} dx \right]^{1/2}.$$

At that point we can use Sobolev embeddings (see [22], corollary IX.13) stating that if  $E(s_0/2) > d/2$  then we have  $H_x^{s/2} \hookrightarrow L_x^{\infty}$ .

So, if  $|j_1| + |l_1| \leq s/2$  we have

$$\begin{split} \left\| \partial_{l_{1}}^{j_{1}} g \right\|_{L_{v}^{2}}^{2} &\leq \sup_{x \in \mathbb{T}^{d}} \left\| \partial_{l_{1}}^{j_{1}} g \right\|_{L_{v}^{2}}^{2} \leqslant C_{s} \left\| \left\| \partial_{l_{1}}^{j_{1}} g \right\|_{L_{v}^{2}}^{2} \right\|_{H_{x}^{s/2}} \\ &\leq C_{s} \sum_{|p| \leqslant s/2} \sum_{p_{1} + p_{2} = p} \int_{\mathbb{T}^{d} \times \mathbb{R}^{d}} \partial_{l_{1} + p_{1}}^{j_{1}} g \, \partial_{l_{1} + p_{2}}^{j_{1}} g \, dv dx \qquad (4.A.1) \\ &\leqslant C_{s} \left\| g \right\|_{H_{x,v}^{s}}^{2}, \end{split}$$

by a mere Cauchy-Schwarz inequality.

In the other case,  $|j_2| + |l_2| \leq s/2$  and by same calculations we show

$$\left\|\partial_{l_2}^{j_2}h\right\|_{\Lambda_v}^2 \leqslant C_s \left\|h\right\|_{H^s_\Lambda}^2.$$

Therefore, by just dividing the sum into this two subcases we obtain the result (H4) for  $\Gamma^-$ , noticing that in the case j = 0 equation (4.A.1) has no v derivatives and the Cauchy-Schwarz inequality does not create such derivatives so the control is only made by x-derivatives.

The second term  $\Gamma^+$  is dealt exactly the same way with, at the end (the study of  $\mathcal{G}^s$ ), another change of variable  $(v, v_*) \to (v', v'_*)$  which gives the result since  $(1 + |v'|)^{\gamma} \leq (1 + |v|)^{\gamma} + (1 + |v_*|)^{\gamma}$  if  $\gamma > 0$ .

## 4.A.3 Landau operator with hard and moderately soft potential

The Landau operator is used to describe plasmas and for instance in the case of particles interacting via a Coulomb interaction (see [112] for more details). The particular case of Coulomb interaction alone ( $\gamma = -3$ ) will not be studied here as the Landau linear operator has a spectral gap if and only if  $\gamma \ge -2$  (see [53], for not constructive arguments, [83] for general constructive case and [4] for explicit construction in the case of hard potential  $\gamma > 0$ ) and so only the case  $\gamma \ge -2$  may be applicable in this study.

We can compute straightforwardly the bilinear symmetric operator associated with the Landau equation:

$$\Gamma(g,h) = \frac{1}{2\sqrt{\mu}} \nabla_v \cdot \int_{\mathbb{R}^d} \sqrt{\mu \mu_*} \Phi(v - v_*) \left[ g_* \nabla_v h + h_* \nabla_v g - g(\nabla_v h)_* - h(\nabla_v g)_* \right] dv_*,$$

where  $\Phi : \mathbb{R}^d \longrightarrow \mathbb{R}^d$  is such that  $\Phi(z)$  is the orthogonal projection onto  $\operatorname{Span}(z)^{\perp}$  so

$$\Phi(z)_{ij} = \delta_{ij} - \frac{z_i z_j}{|z|^2}$$

and  $\gamma$  belongs to [-2, 1].

### 4.A.3.1 Orthogonality to Ker(L): (H5)

Let consider a function  $\psi$  in  $C_{x,v}^{\infty}$ . A mere integration by part gives us

$$\langle \Gamma(g,h),\psi\rangle_{L^2_v} = -\frac{1}{2} \int_{\mathbb{R}^d \times \mathbb{R}^d} \nabla_v \left(\frac{\psi}{\sqrt{\mu}}\right) \cdot \left(\sqrt{\mu\mu_*} \Phi(v-v_*)[G]\right) dv_* dv,$$

where

$$G = g_* \nabla_v h + h_* \nabla_v g - g(\nabla_v h)_* - h(\nabla_v g)_*.$$

Then the change of variable  $(v, v_*) \to (v_*, v)$  only changes  $\nabla_v(\psi/\sqrt{\mu})$  to  $[\nabla_v(\psi/\sqrt{\mu})]_*$ and G becomes -G. Therefore we finally obtain

$$\langle \Gamma(g,h),\psi\rangle_{L^2_v} = \frac{1}{4} \int_{\mathbb{R}^d \times \mathbb{R}^d} \sqrt{\mu \mu_*} \Phi(v-v_*)[G] \cdot \left[ \left( \nabla_v \left(\frac{\psi}{\sqrt{\mu}}\right) \right)_* - \nabla_v \left(\frac{\psi}{\sqrt{\mu}}\right) \right] dv_* dv.$$

As shown in [30] or [82] we have that  $\operatorname{Ker}(L) = \operatorname{Span}(1, v_1, \dots, v_d, |v|^2) \mu^{1/2}$ . Computing the term inside brackets for each of these functions gives us 0 or, in the case  $|v|^2 \sqrt{\mu}$ ,  $2(v_* - v)$ .

However, by definition,  $\Phi(v-v_*)[G]$  belongs to  $\operatorname{Span}(v-v_*)^{\perp}$  and therefore  $\Phi(v-v_*)[G] \cdot (v_*-v) = 0$ . So  $\Gamma$  indeed satisfies (H5).

#### 4.A.3.2 Controlling derivatives: (H4)

The article [53] gives us directly the expected result in its Theorem 3, equation (35) with  $\theta = 0$ . The case where there are only x-derivatives is also included if one takes  $\beta = 0$ .

# 4.B Proofs of the results given in the toolbox

We used the estimates given by the toolbox throughout this chapter. This appendix is to prove all of them. It is divided in two parts. The first one is dedicated to the proof of the equality between null spaces whereas the second part deals with the time derivatives inequalities.

## 4.B.1 Proof of Proposition 4.3.1:

We are about to prove the following proposition.

**Proposition 4.B.1** Let a and b be in  $\mathbb{R}^*$  and consider the operator  $G = aL - bv \cdot \nabla_x$ acting on  $H^1_{x,v}$ . If L satisfies (H1) and (H3) then

$$\operatorname{Ker}(G) = \operatorname{Ker}(L).$$

To prove this result we will need a lemma.

**Lemma 4.B.2** Let  $f : \mathbb{T}^d \times \mathbb{R}^d \longrightarrow \mathbb{R}$  be continuous on  $\mathbb{T}^d \times \mathbb{R}^d$  and differentiable in x. If  $v \cdot \nabla_x f(x, v) = 0$  for all (x, v) in  $\mathbb{T}^d \times \mathbb{R}^d$  then f does not depend on x.

**Proof of Lemma** 4.*B*.2 Fix x in  $\mathbb{T}^d$  and v  $\mathbb{Q}$ -free in  $\mathbb{R}^d$ . For y in  $\mathbb{R}^d$  we will denote by  $\overline{y}$  its equivalent class in  $\mathbb{T}^d$ .

We find easily that g is differentiable on  $\mathbb{R}$  and that  $g'(t) = v \cdot \nabla_x f(x, v) = 0$  on  $\mathbb{R}$ . Therefore:

$$\forall t \in \mathbb{R}, f(x + tv, v) = f(x, v).$$

However, a well-known property about the torus is that the set  $\{x + nv, n \in \mathbb{Z}\}$  is dense in  $\mathbb{T}^d$  for all x in  $\mathbb{T}^d$  and  $v \mathbb{Q}$ -free in  $\mathbb{R}^d$ . This combined with the last result and the continuity of f leads to:

$$\forall y \in \mathbb{T}^d, \quad f(y,v) = f(x,v).$$

To conclude it is enough to see that the set of  $\mathbb{Q}$ -free vector in  $\mathbb{R}^d$  is dense in  $\mathbb{R}^d$  and then, by continuity of f in v:

$$\forall y \in \mathbb{T}^d, \forall v \in \mathbb{R}^d, \quad f(y,v) = f(x,v).$$

Now we have all the tools to prove the proposition about the kernel of operators.

**Proof of Proposition** 4.B.1 Since L satisfies (H1) we know that L acts on  $L_v^2$  and that its Kernel functions  $\phi_i$  only depend on v. Thus, we have directly the first inclusion

$$\operatorname{Ker}(L) \subset \operatorname{Ker}(G).$$

Then, let us consider h in  $H^1_{x,v}$  such that G(h) = 0. Because the transport operator  $v \cdot \nabla_x$  is skew-symmetric in  $L^2_{x,v}$  we have

$$0 = \langle G(h), h \rangle_{L^2_{x,v}} = a \int_{\mathbb{T}^d} \langle L(h), h \rangle_{L^2_v} dx.$$

However, because L satisfies (H3) we obtain:

$$0 \ge \lambda \int_{\mathbb{T}^d} \|h(x,.) - \pi_L(h(x,.))\|_{\Lambda_v}^2 \, dx.$$

But  $\lambda$  is strictly positive and thus:

$$\forall x \in \mathbb{T}^d$$
,  $h(x, \cdot) = \pi_L(h(x, \cdot)) = \sum_{i=1}^d c_i(x)\phi_i$ .

Finally we have, by assumption, G(h) = 0 and because  $h(x, \cdot)$  belongs to Ker(L) for all x in  $\mathbb{T}^d$  we end up with

$$\forall (x,v) \in \mathbb{T}^d \times \mathbb{R}^d, \quad v \cdot \nabla_x h(x,v) = 0.$$

By applying the lemma above we then obtain that h does not depend on x. But  $(\phi_i)_{1 \leq i \leq d}$  is an orthonormal family, basis of Ker(L), and therefore we find that for all i,  $c_i$  does not depend on x.

So, we have proved that:

$$\forall (x,v) \in \mathbb{T}^d \times \mathbb{R}^d, \ h(x,v) = \sum_{i=1}^d c_i \phi_i(v).$$

Therefore, h belongs to Ker(L) and only depends on x.

## 4.B.2 A priori energy estimates

In this subsection we derive all the inequalities we used. Therefore, we assume that L satisfies (H1'), (H2') and (H3) while  $\Gamma$  has the properties (H4) and (H5), and we pick g in  $H^s_{x,v}$ . We consider h in  $H^s_{x,v} \cap \operatorname{Ker}(G_{\varepsilon})^{\perp}$  and we assume that h is a solution to (4.1.3):

$$\partial_t h + \frac{1}{\varepsilon} v \cdot \nabla_x h = \frac{1}{\varepsilon^2} L(h) + \frac{1}{\varepsilon} \Gamma(g, h).$$

In the toolbox, we wrote inequalities on function which were solutions of the linear equation. As the reader may notice, we will deal with the second order operator just by applying the first part of (H4) and Young's inequality. Such an inequality only provides two positive terms, and thus by just setting  $\Gamma$  equal to 0 in the next inequalities we get the expected bounds in the linear case (not the sharpest ones though). Therefore we will just describe the more general case and the linear one is included in it.

## 4.B.2.1 Time evolution of pure *x*-derivatives

The operators L and  $\Gamma$  only act on the v variable. Thus, for  $0 \leq |l| \leq s$ ,  $\partial_l^0$  commutes with L and  $v \cdot \nabla_x$ . Remind that  $v \cdot \nabla_x$  is skew-symmetric in  $L^2_{x,v}(\mathbb{T}^d \times \mathbb{R}^d)$  and therefore we can compute

$$\frac{d}{dt}\left\|\partial_l^0 h\right\|_{L^2_{x,v}}^2 = \frac{2}{\varepsilon^2} \langle L(\partial_l^0 h), \partial_l^0 h \rangle_{L^2_{x,v}} + \frac{2}{\varepsilon} \langle \partial_l^0 \Gamma(g,h), \partial_l^0 h \rangle_{L^2_{x,v}}.$$

We can then use hypothesis (H3) to obtain

$$\frac{2}{\varepsilon^2} \langle L(\partial_l^0 h), \partial_l^0 h \rangle_{L^2_{x,v}} \leqslant -\frac{2\lambda}{\varepsilon^2} \left\| (\partial_l^0 h)^\perp \right\|_{\Lambda}^2$$

We also use (H3) to get  $(\partial_l^0 h)^{\perp} = \partial_l^0 h^{\perp}$ .

To deal with the second scalar product, we will use hypothesis (H4) and (H5), which is still valid for  $\partial_l^0 \Gamma$  since  $\pi_L$  only acts on the v variable, followed by a Young inequality with some  $D_1 > 0$ . This yields

$$\begin{split} \frac{2}{\varepsilon} \langle \partial_l^0 \Gamma(g,h), \partial_l^0 h \rangle_{L^2_{x,v}} &= \frac{2}{\varepsilon} \langle \partial_l^0 \Gamma(g,h), \partial_l^0 h^\perp \rangle_{L^2_{x,v}} \\ &\leqslant \frac{2}{\varepsilon} \mathcal{G}_x^s(g,h) \left\| \partial_l^0 h^\perp \right\|_{\Lambda} \\ &\leqslant \frac{D_1}{\varepsilon} \left( \mathcal{G}_x^s(g,h) \right)^2 + \frac{1}{D_1 \varepsilon} \left\| \partial_l^0 h^\perp \right\|_{\Lambda}^2 \end{split}$$

Gathering the last two upper bounds we obtain

$$\frac{d}{dt} \left\| \partial_l^0 h \right\|_{L^2_{x,v}}^2 \leqslant \left[ \frac{1}{D_1 \varepsilon} - \frac{2\lambda}{\varepsilon^2} \right] \left\| \partial_l^0 h^\perp \right\|_{\Lambda}^2 + \frac{D_1}{\varepsilon} \left( \mathcal{G}^s_x(g,h) \right)^2.$$

Finally, taking  $D_1 = \varepsilon / \lambda$  gives us inequalities (4.3.6), (4.3.7) and (4.3.10).

# **4.B.2.2** Time evolution of $\|\nabla_v h\|_{L^2_{x,v}}^2$

For that term we get, by applying the equation satisfied by h, the following:

$$\frac{d}{dt} \|\nabla_v h\|_{L^2_{x,v}}^2 = \frac{2}{\varepsilon^2} \langle \nabla_v L(h), \nabla_v h \rangle_{L^2_{x,v}} - \frac{2}{\varepsilon} \langle \nabla_v (v \cdot \nabla_x h), \nabla_v h \rangle_{L^2_{x,v}} + \frac{2}{\varepsilon} \langle \nabla_v \Gamma(g,h), \nabla_v h \rangle_{L^2_{x,v}}.$$

And by writing the second term on the right-hand side of the equality and integrating by part in x, we have

$$\langle \nabla_v (v \cdot \nabla_x h), \nabla_v h \rangle_{L^2_{x,v}} = \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}}.$$

Therefore the following holds:

$$\frac{d}{dt} \|\nabla_v h\|_{L^2_{x,v}}^2 = \frac{2}{\varepsilon^2} \langle \nabla_v L(h), \nabla_v h \rangle_{L^2_{x,v}} - \frac{2}{\varepsilon} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} + \frac{2}{\varepsilon} \langle \nabla_v \Gamma(g,h), \nabla_v h \rangle_{L^2_{x,v}}.$$

Then we have by (H1) that  $L = K - \Lambda$  and we can estimate each component thanks to (H1) and (H2):

$$\begin{aligned} -\langle \nabla_v \Lambda(h), \nabla_v h \rangle_{L^2_{x,v}} &\leqslant \nu_4^{\Lambda} \|h\|_{L^2_{x,v}}^2 - \nu_3^{\Lambda} \|\nabla_v h\|_{\Lambda}^2, \\ \langle \nabla_v K(h), \nabla_v h \rangle_{L^2_{x,v}} &\leqslant C(\delta) \|h\|_{L^2_{x,v}}^2 + \delta \|\nabla_v h\|_{L^2_{x,v}}^2, \end{aligned}$$

where  $\delta$  is a strictly positive real that we will choose later.

Finally, for a D > 0 that we will choose later, we have the following upper bound, by Cauchy-Schwarz inequality:

$$-\frac{2}{\varepsilon} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} \leqslant \frac{D}{\varepsilon} \| \nabla_x h \|_{L^2_{x,v}}^2 + \frac{\nu_1^{\Lambda}}{D\nu_0^{\Lambda} \varepsilon} \| \nabla_v h \|_{\Lambda}^2,$$

using the fact that  $\|.\|_{L^2_{x,v}}^2 \leq \frac{\nu_1^{\Lambda}}{\nu_0^{\Lambda}} \|.\|_{\Lambda}^2$ . Finally, another Young inequality gives us a control on the last scalar product, for a  $D_2 > 0$  to be chosen later

$$\frac{2}{\varepsilon} \langle \nabla_v \Gamma(g,h), \nabla_v h \rangle_{L^2_{x,v}} \leqslant \frac{D_2}{\varepsilon} \left( \mathcal{G}^1_{x,v}(g,h) \right)^2 + \frac{1}{D_2 \varepsilon} \| \nabla_v h \|_{\Lambda}^2.$$

We gather here the last three inequalities to obtain our global upper bound:

$$\frac{d}{dt} \|\nabla_{v}h\|_{L^{2}_{x,v}}^{2} \leqslant \frac{1}{\varepsilon^{2}} \left(2\nu_{4}^{\Lambda} + 2C(\delta)\right) \|h\|_{L^{2}_{x,v}}^{2} + \frac{D}{\varepsilon} \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2} \\
+ \left(\frac{2\nu_{1}^{\Lambda}\delta}{\nu_{0}^{\Lambda}\varepsilon^{2}} - \frac{2\nu_{3}^{\Lambda}}{\varepsilon^{2}} + \frac{\nu_{1}^{\Lambda}}{D\varepsilon\nu_{0}^{\Lambda}} + \frac{1}{D_{2}\varepsilon}\right) \|\nabla_{v}h\|_{\Lambda}^{2} + \frac{D_{2}}{\varepsilon} \left(\mathcal{G}_{x,v}^{1}(g,h)\right)^{2}.$$

We can go even further since we have  $\|h\|_{L^2_{x,v}}^2 = \|h^{\perp}\|_{L^2_{x,v}}^2 + \|\pi_L(h)\|_{L^2_{x,v}}^2$ . But because h is in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$  we can use the toolbox and the equation (4.3.5) about the Poincaré inequality:

$$\|\pi_L(h)\|_{L^2_{x,v}}^2 \leqslant C_p \|\nabla_x h\|_{L^2_{x,v}}^2.$$

This last inequality yields:

$$\frac{d}{dt} \|\nabla_{v}h\|_{L^{2}_{x,v}}^{2} \leqslant \frac{\nu_{1}^{\Lambda}}{\nu_{0}^{\Lambda}\varepsilon^{2}} \left(2\nu_{4}^{\Lambda} + 2C(\delta)\right) \left\|h^{\perp}\right\|_{\Lambda}^{2} + \left[\frac{C_{p}}{\varepsilon^{2}} \left(2\nu_{4}^{\Lambda} + 2C(\delta)\right) + \frac{D}{\varepsilon}\right] \|\nabla_{x}h\|_{L^{2}_{x,v}}^{2} \\
+ \left[\frac{2\nu_{1}^{\Lambda}\delta}{\nu_{0}^{\Lambda}\varepsilon^{2}} - \frac{2\nu_{3}^{\Lambda}}{\varepsilon^{2}} + \frac{\nu_{1}^{\Lambda}}{D\varepsilon\nu_{0}^{\Lambda}} + \frac{1}{D_{2}\varepsilon}\right] \|\nabla_{v}h\|_{\Lambda}^{2} + \frac{D_{2}}{\varepsilon} \left(\mathcal{G}_{x,v}^{1}(g,h)\right)^{2}.$$

Therefore, we can choose  $\delta = \nu_0^{\Lambda} \nu_3^{\Lambda} / 6\nu_1^{\Lambda}$ ,  $D = 3\nu_1^{\Lambda} \varepsilon / \nu_0^{\Lambda} \nu_3^{\Lambda}$  and  $D_2 = 3\varepsilon / \nu_3^{\Lambda}$  to get the equation (4.3.8).

# **4.B.2.3** Time evolution of $\langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}}$

In the same way, and integrating by part in x then in v we obtain the following equality:

$$\begin{split} \frac{d}{dt} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} \\ &= \frac{2}{\varepsilon^2} \langle L(\nabla_x h), \nabla_v h \rangle_{L^2_{x,v}} - \frac{2}{\varepsilon} \langle \nabla_v (v \cdot \nabla_x h), \nabla_x h \rangle_{L^2_{x,v}} + \frac{2}{\varepsilon} \langle \nabla_x \Gamma(g,h), \nabla_v h \rangle_{L^2_{x,v}}. \end{split}$$

By writing explicitly  $\langle \nabla_v (v \cdot \nabla_x h), \nabla_x h \rangle_{L^2_{x,v}}$  and by integrating by part one can show that the following holds:

$$\langle \nabla_v (v.\nabla_x h), \nabla_x h \rangle_{L^2_{x,v}} = \frac{1}{2} \| \nabla_x h \|_{L^2_{x,v}}^2$$

Therefore we have an explicit formula for that term and we can find the time derivative of the scalar product being:

$$\frac{d}{dt}\langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} = \frac{2}{\varepsilon^2} \langle L(\nabla_x h), \nabla_v h \rangle_{L^2_{x,v}} - \frac{1}{\varepsilon} \| \nabla_x h \|_{L^2_{x,v}}^2 + \frac{2}{\varepsilon} \langle \nabla_x \Gamma(g,h), \nabla_v h \rangle_{L^2_{x,v}}.$$

We can bound above the first term in the right-hand side of the equality thanks to (H1) and then Cauchy-Schwarz in x, with a constant  $\eta > 0$  to be define later.

$$\begin{aligned} \frac{2}{\varepsilon^2} \langle L(\nabla_x h), \nabla_v h \rangle_{L^2_{x,v}} &= \frac{2}{\varepsilon^2} \langle L(\nabla_x h^\perp), \nabla_v h \rangle_{L^2_{x,v}} \\ &\leqslant \frac{C^L}{\varepsilon^2} \int_{\mathbb{T}^d} 2 \left\| \nabla_x h^\perp \right\|_{\Lambda_v} \left\| \nabla_v h \right\|_{\Lambda_v} dx \\ &\leqslant \frac{C^L \eta}{\varepsilon^2} \left\| \nabla_x h^\perp \right\|_{\Lambda}^2 + \frac{C^L}{\eta \varepsilon^2} \left\| \nabla_v h \right\|_{\Lambda}^2. \end{aligned}$$

Then applying hypothesis (H4) and Young's inequality one more time with a constant  $D_3 > 0$  one may find

$$\frac{2}{\varepsilon} \langle \nabla_x \Gamma(g,h), \nabla_v h \rangle_{L^2_{x,v}} \leqslant \frac{D_3}{\varepsilon} \left( \mathcal{G}^1_x(g,h) \right)^2 + \frac{1}{D_3 \varepsilon} \left\| \nabla_v h \right\|_{\Lambda}^2.$$

Hence we end up with the following inequality:

$$\frac{d}{dt} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} \leqslant \frac{C^L \eta}{\varepsilon^2} \left\| \nabla_x h^\perp \right\|_{\Lambda}^2 + \left( \frac{C^L}{\eta \varepsilon^2} + \frac{1}{D_3} \right) \left\| \nabla_v h \right\|_{\Lambda}^2 - \frac{1}{\varepsilon} \left\| \nabla_x h \right\|_{L^2_{x,v}}^2 \\
+ \frac{D_3}{\varepsilon} \left( \mathcal{G}^1_x(g,h) \right)^2.$$

Now define  $\eta = e/\varepsilon$ , e > 0, and  $D_3 = e/C^L$  to obtain equation (4.3.9).

4.B.2.4 Time evolution of 
$$\left\|\partial_l^j h\right\|_{L^2_{x,v}}^2$$
 for  $|j| \ge 1$  and  $|j| + |l| = s$ 

This term is the only term far from what we already did since we are mixing more than one derivative in x and one derivative in v in general. By simply differentiating in time and integrating by part we find the following equality.

$$\begin{split} \frac{d}{dt} \left\| \partial_l^j h \right\|_{L^2_{x,v}}^2 &= \frac{2}{\varepsilon^2} \langle \partial_l^j L(h), \partial_l^j h \rangle_{L^2_{x,v}} - \frac{2}{\varepsilon} \langle \partial_l^j (v.\nabla_x h), \partial_l^j h \rangle_{L^2_{x,v}} \\ &+ \frac{2}{\varepsilon} \langle \partial_l^j \Gamma(g,h), \partial_l^j h \rangle_{L^2_{x,v}} \\ &= \frac{2}{\varepsilon^2} \langle \partial_l^j L(h), \partial_l^j h \rangle_{L^2_{x,v}} - \frac{2}{\varepsilon} \sum_{i,c_i(j)>0} \langle \partial_l^j h, \partial_{l+\delta_i}^{j-\delta_i} h \rangle_{L^2_{x,v}} \\ &+ \frac{2}{\varepsilon} \langle \partial_l^j \Gamma(g,h), \partial_l^j h \rangle_{L^2_{x,v}}. \end{split}$$

We can then apply Cauchy-Schwarz for the terms inside the sum symbol. For each we can use a  $D_{i,l,s} > 0$  but because they play an equivalent role we will take the same D > 0, that we will choose later:

$$-\frac{2}{\varepsilon}\langle \partial_l^j h, \partial_{l+\delta_i}^{j-\delta_i} h \rangle_{L^2_{x,v}} \leqslant \frac{\nu_1^{\Lambda}}{D\nu_0^{\Lambda}\varepsilon} \left\| \partial_l^j h \right\|_{\Lambda}^2 + \frac{D}{\varepsilon} \left\| \partial_{l+\delta_i}^{j-\delta_i} h \right\|_{L^2_{x,v}}^2.$$

Then we can use (H1') and (H2'), with a  $\delta > 0$  we will choose later, to obtain

$$\frac{2}{\varepsilon^2} \langle \partial_l^j L(h), \partial_l^j h \rangle_{L^2_{x,v}} \leqslant \frac{2}{\varepsilon^2} (C(\delta) + \nu_6^{\Lambda}) \|h\|_{H^{s-1}_{x,v}}^2 + \frac{2}{\varepsilon^2} \left( \frac{\delta \nu_1^{\Lambda}}{\nu_0^{\Lambda}} - \nu_5^{\Lambda} \right) \left\| \partial_l^j h \right\|_{\Lambda}^2.$$

Finally, applying (H4) and Young's inequality with a constant  $D_2 > 0$  we obtain

$$\frac{2}{\varepsilon} \langle \partial_l^j \Gamma(g,h), \partial_l^j h \rangle_{L^2_{x,v}} \leqslant \frac{D_2}{\varepsilon} \left( \mathcal{G}^s_{x,v}(g,h) \right)^2 + \frac{1}{D_2 \varepsilon} \left\| \partial_l^j h \right\|_{\Lambda}^2.$$

Combining these three inequality we find an upper bound for the time evolution. Here we also use the fact that the number of i such that  $c_i(j) > 0$  is less or equal to d.

$$\begin{split} \frac{d}{dt} \left\| \partial_l^j h \right\|_{L^2_{x,v}}^2 &\leqslant \quad \left[ \frac{\nu_1^{\Lambda} d}{D \varepsilon \nu_0^{\Lambda}} + \frac{2}{\varepsilon^2} \left( \frac{\delta \nu_1^{\Lambda}}{\nu_0^{\Lambda}} - \nu_5^{\Lambda} \right) + \frac{1}{D_2 \varepsilon} \right] \left\| \partial_l^j h \right\|_{\Lambda}^2 \\ &+ \frac{D}{\varepsilon} \sum_{i,c_i(j) > 0} \left\| \partial_{l+\delta_i}^{j-\delta_i} h \right\|_{L^2_{x,v}}^2 + \frac{2}{\varepsilon^2} (C(\delta) + \nu_6^{\Lambda}) \left\| h \right\|_{H^{s-1}_{x,v}}^2 \\ &+ \frac{D_2}{\varepsilon} \left( \mathcal{G}_{x,v}^s(g,h) \right)^2. \end{split}$$

Hence, we obtain equations (4.3.11) and (4.3.12) by taking  $D = 3\nu_1^{\Lambda} \varepsilon / \nu_0^{\Lambda} \nu_5^{\Lambda}$ ,  $D_2 = 3\varepsilon / \nu_5^{\Lambda}$ and  $\delta = \nu_0^{\Lambda} \nu_5^{\Lambda} / 6\nu_1^{\Lambda}$ . Also note that in (4.3.11) we used  $\left\| \partial_{l+\delta_i}^{j-\delta_i} h \right\|_{L^2_{x,v}}^2 \leqslant \frac{\nu_1^{\Lambda}}{\nu_0^{\Lambda}} \left\| \partial_{l+\delta_i}^{j-\delta_i} h \right\|_{\Lambda}^2$ .

# **4.B.2.5** Time evolution of $\langle \partial_{l-\delta_i}^{\delta_i} h, \partial_l^0 h \rangle_{L^2_{x,v}}$

With no more calculations, we can bound this term in the same way we did for  $\langle \nabla_x h, \nabla_v h \rangle$ . Here we get

$$\frac{d}{dt} \langle \partial_{l-\delta_{i}}^{\delta_{i}} h, \partial_{l}^{0} h \rangle_{L^{2}_{x,v}} \leqslant \frac{C^{L} \eta}{\varepsilon^{2}} \left\| \partial_{l}^{0} h^{\perp} \right\|_{\Lambda}^{2} + \left[ \frac{C^{L}}{\eta \varepsilon^{2}} + \frac{1}{\varepsilon D_{3}} \right] \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h \right\|_{\Lambda}^{2} - \frac{1}{\varepsilon} \left\| \partial_{l}^{0} h \right\|_{L^{2}_{x,v}}^{2} \\ + \frac{D_{3}}{\varepsilon} \left( \mathcal{G}_{x}^{s}(g,h) \right)^{2}.$$

Now define  $\eta = e/\varepsilon$ , e > 0, and  $D_3 = e/C^L$  to obtain equation (4.3.13).

In the next paragraphs, we are setting g = h.

4.B.2.6 Time evolution of  $\left\| 
abla_v h^\perp \right\|_{L^2_{x,v}}^2$ 

By simply differentiating norm and using (H5) to get  $\Gamma(h,h)^{\perp} = \Gamma(h,h)$ , we compute

$$\frac{d}{dt} \left\| \nabla_v h^{\perp} \right\|_{L^2_{x,v}}^2 = 2 \langle \nabla_v (G_{\varepsilon}(h))^{\perp}, \nabla_v h^{\perp} \rangle_{L^2_{x,v}} + \frac{2}{\varepsilon} \langle \nabla_v \Gamma(h,h), \nabla_v h^{\perp} \rangle_{L^2_{x,v}}.$$

By applying (H4) and Young's inequality to the second term on the right-hand side, with a constant  $D_2 > 0$ , and controlling the  $L^2_{x,v}$ -norm by the  $\Lambda$ -norm we obtain:

$$\frac{2}{\varepsilon} \langle \nabla_v \Gamma(h,h), \nabla_v h^\perp \rangle_{L^2_{x,v}} \leqslant \frac{D_2}{\varepsilon} \left( \mathcal{G}^1_{x,v}(h,h) \right)^2 + \frac{1}{\varepsilon D_2} \left\| \nabla_v h^\perp \right\|_{\Lambda}^2.$$

Then we have to control the first term. Just by writing it and decomposing terms in projection onto Ker(L) and onto its orthogonal we yield:

$$\begin{aligned} 2\langle \nabla_{v}(G_{\varepsilon}(h))^{\perp}, \nabla_{v}h^{\perp}\rangle_{L^{2}_{x,v}} &= \frac{2}{\varepsilon^{2}}\langle \nabla_{v}L(h), \nabla_{v}h^{\perp}\rangle_{L^{2}_{x,v}} - \frac{2}{\varepsilon}\langle \nabla_{v}(v\cdot\nabla_{x}h)^{\perp}, \nabla_{v}h^{\perp}\rangle_{L^{2}_{x,v}} \\ &= \frac{2}{\varepsilon^{2}}\langle \nabla_{v}L(h^{\perp}), \nabla_{v}h^{\perp}\rangle_{L^{2}_{x,v}} - \frac{2}{\varepsilon}\langle \nabla_{x}h, \nabla_{v}h^{\perp}\rangle_{L^{2}_{x,v}} \\ &- \frac{2}{\varepsilon}\langle v\cdot\nabla_{v}\nabla_{x}\pi_{L}(h), \nabla_{v}h^{\perp}\rangle_{L^{2}_{x,v}} \\ &+ \frac{2}{\varepsilon}\langle \nabla_{v}\pi_{L}(v\cdot\nabla_{x}h), \nabla_{v}h^{\perp}\rangle_{L^{2}_{x,v}}. \end{aligned}$$

Then we can control the first term on the right-hand side thanks to (H1) and (H2),  $\delta > 0$  to be chosen later:

$$\frac{2}{\varepsilon^2} \langle \nabla_v L(h^{\perp}), \nabla_v h^{\perp} \rangle_{L^2_{x,v}} \leqslant \frac{2(C(\delta) + \nu_4^{\Lambda})\nu_{\Lambda}^1}{\nu_0^{\Lambda} \varepsilon^2} \left\| h^{\perp} \right\|_{\Lambda}^2 + \frac{2}{\varepsilon^2} \left( \frac{\nu_1^{\Lambda} \delta}{\nu_0^{\Lambda}} - \nu_3^{\Lambda} \right) \left\| \nabla_v h^{\perp} \right\|_{\Lambda}^2$$

We apply Cauchy-Schwarz inequality to the next term, with D to be chosen later:

$$-\frac{2}{\varepsilon} \langle \nabla_x h, \nabla_v h^{\perp} \rangle_{L^2_{x,v}} \leqslant \frac{D}{\varepsilon} \| \nabla_x h \|_{L^2_{x,v}}^2 + \frac{\nu_1^{\Lambda}}{\nu_0^{\Lambda} D\varepsilon} \left\| \nabla_v h^{\perp} \right\|_{\Lambda}^2.$$

For the third term we are going to apply Cauchy-Schwarz inequality and then use the property (H3). The latter property tells us that the functions in Ker(L) are of the form a polynomial in v times  $e^{-|v|^2/4}$ . This fact combined with the shape of  $\pi_L$ , equation (4.3.1), shows us that we can control, by a mere Cauchy-Schwarz inequality, the third term. Then the property (4.3.3) yields the following upper bound:

$$\begin{aligned} -\frac{2}{\varepsilon} \langle v \cdot \nabla_v \nabla_x \pi_L(h), \nabla_v h^{\perp} \rangle_{L^2_{x,v}} &\leqslant \quad \frac{\tilde{D}}{\varepsilon} \| v \cdot \nabla_v \pi_L(\nabla_x h) \|_{L^2_{x,v}}^2 + \frac{1}{\tilde{D}\varepsilon} \left\| \nabla_v h^{\perp} \right\|_{L^2_{x,v}}^2 \\ &\leqslant \quad \frac{\tilde{D}C_{\pi 1}}{\varepsilon} \| \nabla_x h \|_{L^2_{x,v}}^2 + \frac{\nu_1^{\Lambda}}{\nu_0^{\Lambda} \tilde{D}\varepsilon} \left\| \nabla_v h^{\perp} \right\|_{\Lambda}^2. \end{aligned}$$

Finally, we first use equation (4.3.3) controling the v-derivatives of  $\pi_L$  and then see that the norm of  $\pi_L(v.f)$  is easily controlled by the norm of f (just use (H3) and the definition of  $\pi_L$  (4.3.1) and apply Cauchy-Schwarz inequality) by a factor  $C_{\pi 1}$  (increase this constant if necessary in (4.3.3)):

$$\begin{aligned} \frac{2}{\varepsilon} \langle \nabla_v \pi_L(v.\nabla_x h), \nabla_v h^\perp \rangle_{L^2_{x,v}} &\leqslant \quad \frac{D'}{\varepsilon} \left\| \nabla_v \pi_L(v.\nabla_x h) \right\|_{L^2_{x,v}}^2 + \frac{1}{\varepsilon D'} \left\| \nabla_v h^\perp \right\|_{L^2_{x,v}}^2 \\ &\leqslant \quad \frac{D'C_{\pi 1}}{\varepsilon} \left\| \pi_L(v.\nabla_x h) \right\|_{L^2_{x,v}}^2 + \frac{1}{\varepsilon D'} \left\| \nabla_v h^\perp \right\|_{L^2_{x,v}}^2 \\ &\leqslant \quad \frac{D'C_{\pi 1}^2}{\varepsilon} \left\| \nabla_x h \right\|_{L^2_{x,v}}^2 + \frac{\nu_1^\Lambda}{\nu_0^\Lambda \varepsilon D'} \left\| \nabla_v h^\perp \right\|_{L^2_{x,v}}^2. \end{aligned}$$

We then gather all those bounds to get the last upper bound for the time derivative of the *v*-derivative.

$$\begin{split} \frac{d}{dt} \left\| \nabla_{v} h^{\perp} \right\|_{L^{2}_{x,v}}^{2} &\leqslant \quad \frac{\nu_{1}^{\Lambda}}{\nu_{0}^{\Lambda} \varepsilon^{2}} \left( 2\nu_{4}^{\Lambda} + 2C(\delta) \right) \left\| h^{\perp} \right\|_{\Lambda}^{2} + \left[ \frac{D}{\varepsilon} + \frac{D'C_{\pi 1}^{2}}{\varepsilon} + \frac{\tilde{D}C_{\pi 1}}{\varepsilon} \right] \left\| \nabla_{x} h \right\|_{L^{2}_{x,v}}^{2} \\ &+ \left[ \frac{2\nu_{1}^{\Lambda} \delta}{\nu_{0}^{\Lambda} \varepsilon^{2}} - \frac{2\nu_{3}^{\Lambda}}{\varepsilon^{2}} + \frac{\nu_{1}^{\Lambda}}{\varepsilon \nu_{0}^{\Lambda}} \left( \frac{1}{D} + \frac{1}{D'} + \frac{1}{\tilde{D}} \right) + \frac{1}{\varepsilon D_{2}} \right] \left\| \nabla_{v} h^{\perp} \right\|_{\Lambda}^{2} \\ &+ \frac{D_{2}}{\varepsilon} \left( \mathcal{G}_{x,v}^{1}(h,h) \right)^{2}. \end{split}$$

Therefore we obtain (4.3.14) by taking  $D = D' = \tilde{D} = 9\nu_1^{\Lambda}\varepsilon/\nu_0^{\Lambda}\nu_3^{\Lambda}$ ,  $\delta = \nu_0^{\Lambda}\nu_3^{\Lambda}/6\nu_1^{\Lambda}$  and  $D_2 = 3\varepsilon/\nu_3^{\Lambda}$ .

# **4.B.2.7** A new time evolution of $\langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}}$

By integrating by part in x then in v we obtain the following equality on the evolution of the scalar product:

$$\frac{d}{dt} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} = 2 \langle \nabla_x G_{\varepsilon}(h), \nabla_v h \rangle_{L^2_{x,v}} + \frac{2}{\varepsilon} \langle \nabla_v \Gamma(h,h), \nabla_x h \rangle_{L^2_{x,v}}.$$

We will bound above the first term as in the previous case and for the second term involving  $\Gamma$  we use (H4) and Young's inequality with a constant  $D_3 > 0$ :

$$2\langle \nabla_v \Gamma(h,h), \nabla_x h \rangle_{L^2_{x,v}} \leq D_3 \left( \mathcal{G}^1_{x,v}(h,h) \right)^2 + \frac{1}{D_3} \left\| \nabla_x h \right\|_{\Lambda}^2$$

We decompose  $\nabla_x h$  thanks to  $\pi_L$  and we use (4.3.4) to control the fluid part of it,

$$2\langle \nabla_{v}\Gamma(h,h), \nabla_{x}h \rangle_{L^{2}_{x,v}} \leq D_{3} \left( \mathcal{G}^{1}_{x,v}(h,h) \right)^{2} + \frac{1}{D_{3}} \left\| \nabla_{x}h^{\perp} \right\|_{\Lambda}^{2} + \frac{C_{\pi}}{D_{3}} \left\| \nabla_{x}h \right\|_{L^{2}_{x,v}}^{2}.$$

Finally we obtain an upper bound for the time-derivative:

$$\frac{d}{dt} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} \leqslant \left[ \frac{C^L \eta}{\varepsilon^2} + \frac{1}{\varepsilon D_3} \right] \left\| \nabla_x h^\perp \right\|_{\Lambda}^2 + \frac{C^L}{\eta \varepsilon^2} \left\| \nabla_v h \right\|_{\Lambda}^2 + \left[ \frac{C_\pi}{\varepsilon D_3} - \frac{1}{\varepsilon} \right] \left\| \nabla_x h \right\|_{L^2_{x,v}}^2 + \frac{D_3}{\varepsilon} \left( \mathcal{G}^1_{x,v}(h,h) \right)^2.$$

But now, we can use the properties (4.3.3) and (4.3.4) of the projection  $\pi_L$  to go further.

$$\begin{aligned} \|\nabla_{v}h\|_{\Lambda}^{2} &\leq 2 \left\|\nabla_{v}h^{\perp}\right\|_{\Lambda}^{2} + 2 \left\|\nabla_{v}\pi_{L}(h)\right\|_{\Lambda}^{2} \\ &\leq 2 \left\|\nabla_{v}h^{\perp}\right\|_{\Lambda}^{2} + 2C_{\pi 1}C_{\pi} \left\|\pi_{L}(h)\right\|_{L^{2}_{x,v}}^{2} \\ &\leq 2 \left\|\nabla_{v}h^{\perp}\right\|_{\Lambda}^{2} + 2C_{\pi 1}C_{\pi}C_{p} \left\|\nabla_{x}h\right\|_{L^{2}_{x,v}}^{2}, \end{aligned}$$

where we used Poincare inequality (4.3.5) because h is in  $\operatorname{Ker}(G_{\varepsilon})^{\perp}$ .

Hence we have a final upper bound for the time derivative:

$$\begin{aligned} \frac{d}{dt} \langle \nabla_x h, \nabla_v h \rangle_{L^2_{x,v}} &\leqslant \left[ \frac{C^L \eta}{\varepsilon^2} + \frac{1}{\varepsilon D_3} \right] \left\| \nabla_x h^\perp \right\|_{\Lambda}^2 \\ &+ \frac{2C^L}{\eta \varepsilon^2} \left\| \nabla_v h^\perp \right\|_{\Lambda}^2 + \left[ \frac{2C^L C_{\pi 1} C_\pi C_p}{\varepsilon^2 \eta} + \frac{C_\pi}{\varepsilon D_3} - \frac{1}{\varepsilon} \right] \left\| \nabla_x h \right\|_{L^2_{x,v}}^2 \\ &+ \frac{D_3}{\varepsilon} \left( \mathcal{G}^1_{x,v}(h,h) \right)^2. \end{aligned}$$

Thus, setting  $\eta = 8eC^L C_{\pi 1} C_{\pi} C_p / \varepsilon$  with  $e \ge 1$  and  $D_3 = 4C_{\pi}$  we obtain equation (4.3.15).

4.B.2.8 Time evolution of  $\left\|\partial_l^j h^{\perp}\right\|_{L^2_{x,v}}^2$ ,  $j \ge 1$  and |j| + |l| = s

We have the following time evolution:

$$\frac{d}{dt} \left\| \partial_l^j h^\perp \right\|_{L^2_{x,v}}^2 = 2 \langle \partial_l^j (G_\varepsilon(h))^\perp, \partial_l^j h^\perp \rangle_{L^2_{x,v}} + \frac{2}{\varepsilon} \langle \partial_l^j \Gamma(h,h), \partial_l^j h^\perp \rangle_{L^2_{x,v}}$$

As above, we apply (H4) for the last term on the right hand side, with a constant  $D_2 > 0$ ,

$$2\langle \partial_l^j \Gamma(h,h), \partial_l^j h^\perp \rangle_{L^2_{x,v}} \leqslant D_2 \left( \mathcal{G}^s_{x,v}(h,h) \right)^2 + \frac{1}{D_2} \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2.$$

Then we evaluate the first term on the right-hand side.

$$\begin{split} 2\langle\partial_{l}^{j}(G_{\varepsilon}(h))^{\perp},\partial_{l}^{j}h^{\perp}\rangle_{L^{2}_{x,v}} &= \frac{2}{\varepsilon^{2}}\langle\partial_{l}^{j}L(h),\partial_{l}^{j}h^{\perp}\rangle_{L^{2}_{x,v}} - \frac{2}{\varepsilon}\langle\partial_{l}^{j}(v.\nabla_{x}h)^{\perp},\partial_{l}^{j}h^{\perp}\rangle_{L^{2}_{x,v}} \\ &= \frac{2}{\varepsilon^{2}}\langle\partial_{l}^{j}L(h^{\perp}),\partial_{l}^{j}h^{\perp}\rangle_{L^{2}_{x,v}} - \frac{2}{\varepsilon}\langle v\cdot\partial_{l}^{j}\pi_{L}(\nabla_{x}h),\partial_{l}^{j}h^{\perp}\rangle_{L^{2}_{x,v}} \\ &- \frac{2}{\varepsilon}\sum_{i,c_{i}(j)>0}\langle\partial_{l+\delta_{i}}^{j-\delta_{i}}h,\partial_{l}^{j}h^{\perp}\rangle_{L^{2}_{x,v}} \\ &+ \frac{2}{\varepsilon}\langle\partial_{l}^{j}\pi_{L}(v\cdot\nabla_{x}h),\partial_{l}^{j}h^{\perp}\rangle_{L^{2}_{x,v}}. \end{split}$$

Then we shall bound each of these four terms on the right-hand side.

We can first use the properties (H1') and (H2') of L to get, for some  $\delta$  to be chosen later,

$$\frac{2}{\varepsilon^2} \langle \partial_l^j L(h^\perp), \partial_l^j h^\perp \rangle_{L^2_{x,v}} \leqslant \frac{2}{\varepsilon^2} \left( C(\delta) + \nu_6^\Lambda \right) \left\| h^\perp \right\|_{H^{s-1}_{x,v}}^2 + \frac{2}{\varepsilon^2} \left( \frac{\nu_1^\Lambda \delta}{\nu_0^\Lambda} - \nu_5^\Lambda \right) \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2.$$

For the three remaining terms we will apply Cauchy-Schwarz inequality and use the properties of  $\pi_L$  concerning v-derivatives and multiplications by a polynomial in v. First

$$\begin{aligned} -\frac{2}{\varepsilon} \langle v \cdot \partial_l^j \pi_L(\nabla_x h), \partial_l^j h^\perp \rangle_{L^2_{x,v}} &\leqslant \quad \frac{D}{\varepsilon} \left\| v \cdot \partial_l^j \pi_L(\nabla_x h) \right\|_{L^2_{x,v}}^2 + \frac{1}{D\varepsilon} \left\| \partial_l^j h^\perp \right\|_{L^2_{x,v}}^2 \\ &\leqslant \quad \frac{DC_{\pi s}}{\varepsilon} \left\| \partial_l^0 (\nabla_x h) \right\|_{L^2_{x,v}}^2 + \frac{\nu_1^\Lambda}{\nu_0^\Lambda D\varepsilon} \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2 \\ &\leqslant \quad \begin{cases} \quad \frac{DC_{\pi s}}{\varepsilon} \sum_{|l'| \leqslant s-1} \left\| \partial_{l'}^0 h \right\|_{L^2_{x,v}}^2 + \frac{\nu_1^\Lambda}{\nu_0^\Lambda D\varepsilon} \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2, \ |j| = 1 \\ \\ \quad \frac{DC_{\pi s}}{\varepsilon} \sum_{|l'| \leqslant s-1} \left\| \partial_{l'}^0 h \right\|_{L^2_{x,v}}^2 + \frac{\nu_1^\Lambda}{\nu_0^\Lambda D\varepsilon} \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2, \ |j| > 1, \end{aligned}$$

where we used that |l| = |s| - |j|. Then

$$-\frac{2}{\varepsilon}\langle\partial_{l+\delta_{i}}^{j-\delta_{i}}h,\partial_{l}^{j}h^{\perp}\rangle_{L^{2}_{x,v}} \leqslant \frac{D'}{\varepsilon}\left\|\partial_{l+\delta_{i}}^{j-\delta_{i}}h\right\|^{2}_{L^{2}_{x,v}} + \frac{\nu_{1}^{\Lambda}}{\nu_{0}^{\Lambda}D'\varepsilon}\left\|\partial_{l}^{j}h^{\perp}\right\|^{2}_{\Lambda}$$

In the case where |j| > 1 we can also use that  $\left\| \partial_{l+\delta_i}^{j-\delta_i} h \right\|_{L^2_{x,v}}^2$  can be decomposed thanks to  $\pi_L$  and its orthogonal projector. Then the fluid part is controlled by the *x*-derivatives only.

And finally

$$\begin{aligned} \frac{2}{\varepsilon} \langle \partial_l^j \pi_L (v \cdot \nabla_x h), \partial_l^j h^\perp \rangle_{L^2_{x,v}} &\leqslant \quad \frac{\tilde{D}}{\varepsilon} \left\| \partial_l^j \pi_L (v \cdot \nabla_x h) \right\|_{L^2_{x,v}}^2 + \frac{1}{\tilde{D}\varepsilon} \left\| \partial_l^j h^\perp \right\|_{L^2_{x,v}}^2 \\ &\leqslant \quad \frac{\tilde{D}C_{\pi s}}{\varepsilon} \left\| \partial_l^0 \nabla_x h \right\|_{L^2_{x,v}}^2 + \frac{\nu_1^\Lambda}{\tilde{D}\nu_0^\Lambda \varepsilon} \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2 \\ &\leqslant \quad \begin{cases} \frac{\tilde{D}C_{\pi s}}{\varepsilon} \sum_{|l'|=s} \left\| \partial_{l'}^0 h \right\|_{L^2_{x,v}}^2 + \frac{\nu_1^\Lambda}{\nu_0^\Lambda \tilde{D}\varepsilon} \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2, \text{ if } |j| = 1 \\ \\ &\frac{\tilde{D}C_{\pi s}}{\varepsilon} \sum_{|l'|\leqslant s-1} \left\| \partial_{l'}^0 h \right\|_{L^2_{x,v}}^2 + \frac{\nu_1^\Lambda}{\nu_0^\Lambda \tilde{D}\varepsilon} \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2, \text{ if } |j| > 1, \end{aligned}$$

We are now able to combine all those estimates to get an upper bound of the timederivative we are looking at. We can also give to different bounds, depending on the size |j|. We also used that the number of *i* such that  $c_i(j) > 0$  is less than *d*.

In the case |j| > 1,

$$\begin{split} \frac{d}{dt} \left\| \partial_l^j h^\perp \right\|_{L^2_{x,v}}^2 &\leqslant \quad \left[ \frac{2}{\varepsilon^2} \left( \frac{\nu_1^\Lambda \delta}{\nu_0^\Lambda} - \nu_5^\Lambda \right) + \frac{\nu_1^\Lambda}{\nu_0^\Lambda \varepsilon} \left( \frac{1}{D} + \frac{d}{D'} + \frac{1}{\tilde{D}} \right) + \frac{1}{D_2} \right] \left\| \partial_l^j h^\perp \right\|_{\Lambda}^2 \\ &+ \frac{D' \nu_1^\Lambda}{2\nu_0^\Lambda \varepsilon} \sum_{i,c_i(j)>0} \left\| \partial_{l+\delta_i}^{j-\delta_i} h^\perp \right\|_{\Lambda}^2 \\ &+ \left[ \frac{DC_{\pi s}}{2\varepsilon} + \frac{D'C_{\pi s}}{\varepsilon} + \frac{\tilde{D}C_{\pi s}}{\varepsilon} \right] \sum_{|l'|\leqslant s-1} \left\| \partial_{l'}^0 h \right\|_{L^2_{x,v}}^2 \\ &+ \frac{2(C(\delta) + \nu_6^\Lambda)}{\varepsilon^2} \left\| h^\perp \right\|_{H^{s-1}_{x,v}}^2 \\ &+ \frac{D_2}{\varepsilon} \left( \mathcal{G}_{x,v}^s(h,h) \right)^2. \end{split}$$

And in the case |j| = 1,

$$\begin{split} \frac{d}{dt} \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h^{\perp} \right\|_{L^{2}_{x,v}}^{2} &\leqslant \quad \left[ \frac{2}{\varepsilon^{2}} \left( \frac{\nu_{1}^{\Lambda} \delta}{\nu_{0}^{\Lambda}} - \nu_{5}^{\Lambda} \right) + \frac{\nu_{1}^{\Lambda}}{\nu_{0}^{\Lambda} \varepsilon} \left( \frac{1}{D} + \frac{1}{D'} + \frac{1}{\tilde{D}} \right) + \frac{1}{D_{2}} \right] \left\| \partial_{l-\delta_{i}}^{\delta_{i}} h^{\perp} \right\|_{\Lambda}^{2} \\ &+ \left[ \frac{DC_{\pi s}}{\varepsilon} + \frac{D'}{\varepsilon} + \frac{\tilde{D}C_{\pi s}}{\varepsilon} \right] \sum_{|l'|=s} \left\| \partial_{l'}^{0} h \right\|_{L^{2}_{x,v}}^{2} \\ &+ \frac{2(C(\delta) + \nu_{6}^{\Lambda})}{\varepsilon^{2}} \left\| h^{\perp} \right\|_{H^{s-1}_{x,v}}^{2} \\ &+ \frac{D_{2}}{\varepsilon} \left( \mathcal{G}_{x,v}^{s}(h,h) \right)^{2}. \end{split}$$

By taking  $D = \tilde{D} = 9\nu_1^{\Lambda} \varepsilon / \nu_0^{\Lambda} \nu_5^{\Lambda}$ ,  $D_2 = 3\varepsilon / \nu_5^{\Lambda}$ ,  $\delta = \nu_0^{\Lambda} \nu_5^{\Lambda} / 6\nu_1^{\Lambda}$  and  $D' = 9\nu_1^{\Lambda} \varepsilon / \nu_0^{\Lambda} \nu_5^{\Lambda}$ , if

|j| = 1, or  $D' = 9\nu_1^{\Lambda} d\varepsilon / \nu_0^{\Lambda} \nu_5^{\Lambda}$ , if |j| > 1, we obtain (4.3.16) and (4.3.17).

# **4.B.2.9** A new time evolution of $\langle \partial_{l-\delta_i}^{\delta_i} h, \partial_l^0 h \rangle_{L^2_{x,v}}$

By integrating by part in x then in v we obtain the following equality on the evolution of the scalar product.

$$\frac{d}{dt}\langle\partial_{l-\delta_{i}}^{\delta_{i}}h,\partial_{l}^{0}h\rangle_{L^{2}_{x,v}} = 2\langle\partial_{l-\delta_{i}}^{\delta_{i}}G_{\varepsilon}(h),\partial_{l}^{0}h\rangle_{L^{2}_{x,v}} + \frac{2}{\varepsilon}\langle\partial_{l-\delta_{i}}^{\delta_{i}}\Gamma(h,h),\partial_{l}^{0}h\rangle_{L^{2}_{x,v}}.$$

We will bound above the first term as in the previous case and for the second term involving  $\Gamma$  we use (H4) and Young's inequality with a constant  $D_3 > 0$ . Moreover, we decompose  $\partial_l^0 h$  into its fluid part and its microscopic part and we apply (4.3.4) on the fluid part. This yields

$$2\langle \partial_{l-\delta_{i}}^{\delta_{i}}\Gamma(h,h), \partial_{l}^{0}h \rangle_{L^{2}_{x,v}} \leqslant D_{3} \left( \mathcal{G}^{s}_{x,v}(h,h) \right)^{2} + \frac{1}{D_{3}} \left\| \partial_{l}^{0}h^{\perp} \right\|_{\Lambda}^{2} + \frac{C_{\pi}}{D_{3}} \left\| \partial_{l}^{0}h \right\|_{L^{2}_{x,v}}^{2}.$$

Finally we obtain an upper bound for the time-derivative:

$$\begin{split} \frac{d}{dt} \langle \partial_{l-\delta_{i}}^{\delta_{i}}h, \partial_{l}^{0}h \rangle_{L^{2}_{x,v}} &\leqslant \quad \left[ \frac{C^{L}\eta}{\varepsilon^{2}} + \frac{1}{D_{3}} \right] \left\| \partial_{l}^{0}h^{\perp} \right\|_{\Lambda}^{2} + \frac{C^{L}}{\eta\varepsilon^{2}} \left\| \partial_{l-\delta_{i}}^{\delta_{i}}h \right\|_{\Lambda}^{2} + \left( \frac{C_{\pi}}{\varepsilon D_{3}} - \frac{1}{\varepsilon} \right) \left\| \partial_{l}^{0}h \right\|_{L^{2}_{x,v}}^{2} \\ &+ \frac{D_{3}}{\varepsilon} \left( \mathcal{G}^{s}_{x,v}(h,h) \right)^{2}. \end{split}$$

Now we can use the properties of  $\pi_L$  concerning the *v*-derivatives, equation (4.3.3), the equivalence of norm under the projection  $\pi_L$ , equation (4.3.4), and Poincare inequality get the following upper bound:

$$\begin{split} \left\| \partial_{l-\delta_{i}}^{\delta_{i}}h \right\|_{\Lambda}^{2} &\leq 2 \left\| \partial_{l-\delta_{i}}^{\delta_{i}}h^{\perp} \right\|_{\Lambda}^{2} + 2 \left\| \partial_{l-\delta_{i}}^{\delta_{i}}\pi_{L}(h) \right\|_{\Lambda}^{2} \\ &\leq 2 \left\| \partial_{l-\delta_{i}}^{\delta_{i}}h^{\perp} \right\|_{\Lambda}^{2} + 2C_{\pi s}C_{\pi} \left\| \partial_{l-\delta_{i}}^{0}(h) \right\|_{L^{2}_{x,v}}^{2} \\ &\leq 2 \left\| \partial_{l-\delta_{i}}^{\delta_{i}}h^{\perp} \right\|_{\Lambda}^{2} + 2C_{\pi s}C_{\pi} \sum_{|l'| \leq s-1} \left\| \partial_{l'}^{0}h \right\|_{L^{2}_{x,v}}^{2}. \end{split}$$

Therefore,

$$\begin{split} \frac{d}{dt} \langle \partial_{l-\delta_{i}}^{\delta_{i}}h, \partial_{l}^{0}h \rangle_{L^{2}_{x,v}} &\leqslant \quad \left[\frac{C^{L}\eta}{\varepsilon^{2}} + \frac{1}{D_{3}}\right] \left\|\partial_{l}^{0}h^{\perp}\right\|_{\Lambda}^{2} + \frac{2C^{L}}{\eta\varepsilon^{2}} \left\|\partial_{l-\delta_{i}}^{\delta_{i}}h^{\perp}\right\|_{\Lambda}^{2} \\ &+ \left(\frac{C_{\pi}}{\varepsilon D_{3}} - \frac{1}{\varepsilon}\right) \left\|\partial_{l}^{0}h\right\|_{L^{2}_{x,v}}^{2} + \frac{2C^{L}C_{\pi s}C_{\pi}}{\eta\varepsilon^{2}} \sum_{|l'|\leqslant s-1} \left\|\partial_{l'}^{0}h\right\|_{L^{2}_{x,v}}^{2} \\ &+ \frac{D_{3}}{\varepsilon} \left(\mathcal{G}^{s}_{x,v}(h,h)\right)^{2}. \end{split}$$

We finally define  $\eta = 8eC^L C_{\pi s} C_{\pi} d/\varepsilon$ , with e > 1, and  $D_3 = 2C_{\pi}$  to yield equation (4.3.18).

# 4.C Proof of the hydrodynamical limit lemmas

In this section we are going to prove all the different lemmas used in section 9.

All along the demonstration we will use this inequality:

$$\forall t > 0, \ k \in \mathbb{N}^*, \ q \ge 0, \ p > 0, \ t^q k^{2p} e^{-atk^2} \leqslant C_p(a) t^{q-p}.$$
 (4.C.1)

## 4.C.1 Study of the linear part

## 4.C.1.1 Proof of Lemma 4.8.6

Fix T in  $[0, +\infty]$ . By integrating we compute

$$\int_{0}^{T} U_{0j}^{\varepsilon} h_{in} dt = \sum_{n \in \mathbb{Z}^{d} - \{0\}} e^{in.x} \left[ \int_{0}^{T} e^{\frac{i\alpha_{j}t|n|}{\varepsilon} - \beta_{j}t|n|^{2}} dt \right] P_{0j} \left( \frac{n}{|n|} \right) \hat{h}_{in}(n,v)$$
$$= \sum_{n \in \mathbb{Z}^{d} - \{0\}} e^{in.x} \frac{\varepsilon}{i\alpha_{j} |n| - \varepsilon \beta_{j} |n|^{2}} \left[ e^{\frac{i\alpha_{j}T|n|}{\varepsilon} - \beta_{j}T|n|^{2}} - 1 \right] P_{0j} \hat{h}_{in}(n,v).$$

The Fourier transform is an isometry in  ${\cal L}^2_x$  and therefore

$$\left\|\int_{0}^{T} U_{0j}^{\varepsilon} h_{in} dt\right\|_{L^{2}_{x} L^{2}_{v}}^{2} \leqslant \varepsilon^{2} \sum_{n \in \mathbb{Z}^{d} - \{0\}} \frac{2}{\alpha_{j}^{2} |n|^{2} + \varepsilon^{2} \beta_{j}^{2} |n|^{4}} \left\|P_{0j}\left(\frac{n}{|n|}\right) \hat{h}_{in}(n, \cdot)\right\|_{L^{2}_{v}}^{2}.$$

Finally, we know that, like  $e_{0j}$ ,  $P_{0j}$  is continuous on the compact  $\mathbb{S}^{d-1}$  and so is bounded. But the latter is a linear operator acting on  $L_v^2$  and therefore it is bounded by  $M_{0j}$  in the operator norm on  $L_v^2$ . Thus

$$\begin{split} \left\| \int_{0}^{T} U_{0j}^{\varepsilon} h_{in} dt \right\|_{L_{x}^{2} L_{v}^{2}}^{2} &\leqslant \varepsilon^{2} \frac{M_{0j}^{2}}{\alpha_{j}^{2}} \sum_{n \in \mathbb{Z}^{d} - \{0\}} \left\| \hat{h}_{in}(n, \cdot) \right\|_{L_{v}^{2}}^{2} \\ &\leqslant \varepsilon^{2} \frac{M_{0j}^{2}}{\alpha_{j}^{2}} \left\| h_{in}(\cdot, \cdot) \right\|_{L_{x}^{2} L_{v}^{2}}^{2}, \end{split}$$

which is the expected result.

Now, let us look at the  $L_x^2$ -norm of this operator, to see how the torus case is different from the case  $\mathbb{R}^d$  studied in [39] and [10].

Consider a direction  $n_1$  in the Fourier transform space of the torus and define  $\phi_{n_1} = \mathcal{F}_x^{-1}(e^{in_1})$ . We have the following equality

$$\langle U_{0j}^{\varepsilon}h_{in},\phi_{n_1}\rangle_{L^2_x} = \langle \hat{U}_{0j}^{\varepsilon}\hat{h}_{in},\hat{\phi}_{n_1}\rangle_{L^2_n} = e^{\frac{i\alpha_jt|n_1|}{\varepsilon} -\beta_jt|n_1|^2} P_{0j}\left(\frac{n_1}{|n_1|}\right)\hat{h}_{in}(n_1,v)$$

If we do not integrate in time, one can easily see that this expression cannot have a limit as  $\varepsilon$  tends to 0 if  $P_{0j}\left(\frac{n_1}{|n_1|}\right)\hat{h}_{in}(n_1,v) \neq 0$ , and so we cannot even have a weak convergence. The difference with the whole space case is this possibility to single out one mode in the frequency space in the case of the torus. This leads to the possible existence of periodic function at a given frequency, the norm of which will never decrease. This is impossible in the case of a continuous Fourier space, as in  $\mathbb{R}^d$ , and well described by the Riemann-Lebesgue lemma.

Therefore we have a convergence without averaging in time if and only if

$$P_{0j}\left(\frac{n_1}{|n_1|}\right)\hat{h}_{in}(n_1,v)=0,$$

for all  $j = \pm 1$  and all direction  $n_1$ . This means that for all  $j = \pm 1$  and all  $n_1$ ,  $\langle e_{0j} \left( \frac{n_1}{|n_1|} \right), \hat{h}_{in} \rangle_{L_v^2} = 0$ . By the expression known (see theorem 4.8.3) of  $e_{0\pm 1}$ , this is true if and only if  $\nabla_x \cdot u_{in} = 0$  and  $\rho_{in} + \theta_{in} = 0$ .

#### 4.C.1.2 Proof of Lemma 4.8.7

This lemma deals with three different terms and we study them one by one because they behaviour are quite different.

**The term**  $U_{1i}^{\varepsilon}$ : We remind that we have

$$\hat{U}_{1j}^{\varepsilon}\hat{h}_{in} = \chi_{|\varepsilon n| \leqslant n_0} e^{\frac{i\alpha_j t|n|}{\varepsilon} - \beta_j t|n|^2} \left( e^{\frac{t}{\varepsilon^2}\gamma_j(|\varepsilon n|)} - 1 \right) P_{0j}\left(\frac{n}{|n|}\right) \hat{h}_{in}(n,v).$$

If we take T > 0, by Parseval identity we get

$$\left\| \int_{0}^{T} U_{1j}^{\varepsilon} h_{in} dt \right\|_{L^{2}_{x}L^{2}_{v}}^{2} = \sum_{n \in \mathbb{Z}^{d} - \{0\}} \chi_{|\varepsilon n \leqslant n_{0}|} \left| \int_{0}^{T} e^{\frac{i\alpha_{j}t|n|}{\varepsilon} - \beta_{j}t|n|^{2}} \left( e^{\frac{t}{\varepsilon^{2}}\gamma_{j}(|\varepsilon n|)} - 1 \right) dt \right|^{2} \left\| P_{0j}\hat{h}_{in} \right\|_{L^{2}_{v}}^{2}.$$

But then we can use the fact that  $|e^a - 1| \leq |a| e^{|a|}$ , the inequalities satisfied by  $\gamma_j$  and the computational inequality (4.C.1) to obtain

$$\begin{aligned} \left| \int_{0}^{T} e^{\frac{i\alpha_{j}t|n|}{\varepsilon} - \beta_{j}t|n|^{2}} \left( e^{\frac{t}{\varepsilon^{2}}\gamma_{j}(|\varepsilon n|)} - 1 \right) dt \right| &\leqslant C_{\gamma}\varepsilon \int_{0}^{T} t |n|^{3} e^{-\frac{t\beta_{j}}{2}|n|^{2}} dt \\ &\leqslant C_{\gamma}\varepsilon C_{3/2} \left( \frac{\beta_{j}}{4} \right) \int_{0}^{T} \frac{1}{\sqrt{t}} e^{-\frac{t\beta_{j}}{4}|n|^{2}} dt \\ &\leqslant C_{\gamma}\varepsilon C_{3/2} \left( \frac{\beta_{j}}{4} \right) \int_{0}^{+\infty} \frac{1}{\sqrt{t}} e^{-\frac{t\beta_{j}}{4}} dt, \end{aligned}$$

which is independent of n and is written  $I\varepsilon$ . Therefore we have the expected inequality, by using the continuity of  $P_{0j}$ ,

$$\left\|\int_{0}^{T} U_{1j}^{\varepsilon} h_{in} dt\right\|_{L^{2}_{x}L^{2}_{v}}^{2} \leqslant \varepsilon^{2} I^{2} M_{0j}^{2} \|h_{in}\|_{L^{2}_{x}L^{2}_{v}}^{2}.$$

The last two inequalities we want to show comes from Parseval's identity, the properties of  $\gamma_j$  and the computational inequality (4.C.1):

$$\begin{aligned} \left\| U_{1j}^{\varepsilon} h_{in} \right\|_{L_{x}^{2} L_{v}^{2}}^{2} &= \sum_{n \in \mathbb{Z}^{d} - \{0\}} \chi_{|\varepsilon n| \leq n_{0}} e^{-2\beta_{j} t |n|^{2}} \left| e^{\frac{t}{\varepsilon^{2}} \gamma_{j} |\varepsilon n|} - 1 \right|^{2} \left\| P_{0j} \left( \frac{n}{|n|} \right) \hat{h}_{in} \right\|_{L_{v}^{2}}^{2} \\ &\leqslant M_{0j}^{2} C_{\gamma}^{2} \varepsilon^{2} \sum_{n \in \mathbb{Z}^{d} - \{0\}} \chi_{|\varepsilon n| \leq n_{0}} t^{2} |n|^{6} e^{-\beta_{j} t |n|^{2}} \left\| \hat{h}_{in} \right\|_{L_{v}^{2}}^{2} \\ &\leqslant M_{0j}^{2} C_{\gamma}^{2} \varepsilon^{2} C_{2} \left( \frac{\beta_{j}}{2} \right) \sum_{n \in \mathbb{Z}^{d} - \{0\}} \chi_{|\varepsilon n| \leq n_{0}} |n|^{2} e^{-\frac{\beta_{j} t}{2} |n|^{2}} \left\| \hat{h}_{in} \right\|_{L_{v}^{2}}^{2}. (4.C.2) \end{aligned}$$

Finally, if we integrate in t between 0 and  $+\infty$  we obtain the expected second inequality of the lemma. If we merely bound  $e^{-\frac{\beta_j t}{2}|n|^2}$  by one and use the fact that  $\chi_{|\varepsilon n| \leq n_0} \leq 1$  and  $\chi_{|\varepsilon n| \leq n_0} \varepsilon^2 |n|^2 \leq n_0^2$  we obtain the third inequality of the lemma for  $\delta = 1$  and  $\delta = 0$ . Then by interpolation we obtain the general case for  $0 \leq \delta \leq 1$ .

The term  $U_{2i}^{\varepsilon}$ : Fix T > 0. By Parseval's identity we have

$$\begin{split} \left\| \int_{0}^{T} U_{2j}^{\varepsilon} h_{in} dt \right\|_{L_{x}^{2} L_{v}^{2}}^{2} &= \sum_{n \in \mathbb{Z}^{d}} \chi_{|\varepsilon n| \leq n_{0}} \left\| \int_{0}^{T} e^{\frac{i\alpha_{j} t |n|}{\varepsilon} - \beta_{j} t |n|^{2} + \frac{t}{\varepsilon^{2}} \gamma_{j}(|\varepsilon n|)} dt \right|^{2} |\varepsilon n|^{2} \left\| \tilde{P}_{1j} \hat{h}_{in} \right\|_{L_{v}^{2}}^{2} \\ &\leqslant \sum_{n \in \mathbb{Z}^{d} - \{0\}} \frac{4}{\beta_{j}^{2} |n|^{4}} |\varepsilon n|^{2} \left\| \tilde{P}_{1j} \left( |\varepsilon n|, \frac{n}{|n|} \right) \hat{h}_{in} \right\|_{L_{v}^{2}}^{2}, \end{split}$$

where we used the inequalities satisfied by  $\gamma$  and integration in time. Then,  $\tilde{P}_{1j}$  is continuous on the compact  $[-n_0, n_0] \times \mathbb{S}^{d-1}$  and so is bounded, as an operator acting on  $L_v^2$ , by  $M_{1j} > 0$ . Hence, Parseval's identity offers us the first inequality of the

The last two inequalities are just using Parseval's identity and the continuity of  $P_{1j}$ . Indeed,

$$\begin{aligned} \left\| U_{2j}^{\varepsilon} h_{in} \right\|_{L_{x}^{2} L_{v}^{2}}^{2} &= \sum_{n \in \mathbb{Z}^{d} - \{0\}} \chi_{|\varepsilon n| \leqslant n_{0}} \left| e^{\frac{i \alpha_{j} t |n|}{\varepsilon} - \beta_{j} t |n|^{2} + \frac{t}{\varepsilon^{2}} \gamma_{j}(|\varepsilon n|)} \right|^{2} |\varepsilon n|^{2} \left\| \tilde{P}_{1j}(n) \hat{h}_{in} \right\|_{L_{v}^{2}}^{2} \\ &\leqslant M_{1j}^{2} \varepsilon^{2} \sum_{n \in \mathbb{Z}^{d} - \{0\}} \chi_{|\varepsilon n| \leqslant n_{0}} |n|^{2} e^{-t\beta_{j} |n|^{2}} \left\| \hat{h}_{in} \right\|_{L_{v}^{2}}^{2}. \end{aligned}$$

We recognize here the same form of inequality (4.C.2). Thus, we obtain the last two inequalities of the statement in the same way.

The term  $U_{3j}^{\varepsilon}$ : We remind the reader that

$$\hat{U}_{3j}^{\varepsilon} = \left(\chi_{|\varepsilon n| \leqslant n_0} - 1\right) e^{\frac{i\alpha_j t|n|}{\varepsilon} - \beta_j t|n|^2} P_{0j}\left(\frac{n}{|n|}\right).$$

We have the following inequality

lemma.

$$\left|\chi_{|\varepsilon n|\leqslant n_0}-1\right|\leqslant \frac{\varepsilon n}{n_0}.$$

Therefore, replacing  $\tilde{P}_{1j}$  by  $\frac{1}{n_0}P_{0j}$  and  $\beta_j$  by  $2\beta_j$  (since  $\frac{t}{\varepsilon^2}\gamma_j(|\varepsilon n|) \leq \frac{t\beta_j}{2}|n|^2$ ) in the proof made for  $U_{2j}^{\varepsilon}$  we obtain the expected three inequalities for  $U_{3j}^{\varepsilon}h_{in}$ , the last one only with  $\delta = 1$ .

To have the last inequality in  $\delta$ , it is enough to bound  $|\chi_{|\varepsilon n| \leq n_0} - 1|$  by 1 and then using the continuity of  $P_{0j}$  to have the result for  $\delta = 0$ . Finally, we interpolate to get the general result for all  $0 \leq \delta \leq 1$ .

## 4.C.1.3 Proof of Lemma 4.8.8

Thanks to Theorem 4.8.3 we have that

$$\|U_R^{\varepsilon}h_{in}\|_{L^2_x L^2_v}^2 = \left\|\hat{U}_R(t/\varepsilon^2, \varepsilon n, v)\hat{h}_{in}\right\|_{L^2_n L^2_v}^2 \leqslant C_R^2 e^{-2\frac{\sigma t}{\varepsilon^2}} \|h_{in}\|_{L^2_x L^2_v}^2.$$

But then we have, thanks to the technical lemma 4.C.1, that  $e^{-2\frac{\sigma t}{\varepsilon^2}} \leq C_{1/2}(2\sigma)\frac{\varepsilon}{\sqrt{t}}$ , which gives us the last two inequalities we wanted. For the first inequality, a mere Cauchy-Schwartz inequality yields

$$\left\|\int_0^T U_R^{\varepsilon} h_{in} dt\right\|_{L^2_x L^2_v}^2 \leqslant T \int_0^T \|U_R^{\varepsilon} h_{in}\|_{L^2_x L^2_v}^2 dt,$$

which gives us the first inequality by integrating in t.

Now, let us suppose that we have the strong convergence down to t = 0. At t = 0 we can write that  $e^{tG_{\varepsilon}} = \text{Id}$  and therefore that:

$$\mathrm{Id} = \chi_{|\varepsilon n| \leq n_0} \sum_{j=-1}^{2} P_j\left(|\varepsilon n|, \frac{n}{|n|}\right) + \hat{U}_R(0, \varepsilon n, v).$$

We have the strong convergence down to 0 as  $\varepsilon$  tends to 0. Therefore, taking the latter equality at  $\varepsilon = 0$  we have, because  $\sum_{j=-1}^{2} P_{0j} = \pi_L$ ,

$$\hat{U}_R(0,0,v) = \mathrm{Id} - \pi_L.$$

Then  $\hat{U}_R \hat{h}_{in}$  tends to 0 as  $\varepsilon$  tends to 0 in  $C([0, +\infty), L_x^2 L_v^2)$  if and only if  $h_{in}$  belongs to  $\operatorname{Ker}(L)$ .

In that case, we can use the proof of Lemma 6.2 of [10] in which they noticed that

$$U_R^{\varepsilon}(t,x,v) = e^{tG_{\varepsilon}} U_R^{\varepsilon}(0,x,v) = e^{tG_{\varepsilon}} \left[ \mathcal{F}_x^{-1} \left( Id - \chi_{|\varepsilon n| \leqslant n_0} \sum_{j=-1}^2 P_j(\varepsilon n) \right) \mathcal{F}_x \right].$$

Thanks to that new form we have that, if  $h_{in} = \pi_L(h_{in})$ ,

$$U_R^{\varepsilon}(t,x,v)h_{in} = e^{tG_{\varepsilon}} \left[ \mathcal{F}_x^{-1} \left( (1 - \chi_{|\varepsilon n| \leqslant n_0}) - |\varepsilon n| \,\chi_{|\varepsilon n| \leqslant n_0} \sum_{j=-1}^2 \tilde{P}_{1j}(\varepsilon n) \right) \hat{h}_{in} \right],$$

because  $\pi_L = \sum_{j=-1}^2 P_{0j}$ .

Therefore we can redo the same estimates we worked out in the previous lemmas and use

the same interpolation method to get the result stated in Lemma 4.8.8.

## 4.C.2 Study of the bilinear part

### 4.C.2.1 A simplification without loss of generality

All the terms we are about to study, apart from the remainder term, are of the following form

$$\psi_{ij}^{\varepsilon}(u_{\varepsilon}) = \int_{0}^{t} \sum_{n \in \mathbb{Z}^{d} - \{0\}} g(t, s, k, x) P(n) \hat{u}_{\varepsilon}(s, k, v) ds,$$

with P(n) being a projector in  $L_v^2$ , bounded uniformly in n.

Looking at the dual definition of the norm of a function in  $L^2_{x,v}$ , we can consider f in  $C^{\infty}_{c}(\mathbb{T}^d \times \mathbb{R}^d)$  such that  $\|f\|_{L^2_{x,v}} = 1$  and take the scalar product with  $\psi^{\varepsilon}_{ij}(u_{\varepsilon})$ . This yields, since P is a projector and thus symmetric,

$$\begin{aligned} \langle \psi_{ij}^{\varepsilon}(u_{\varepsilon}), f \rangle_{L^{2}_{x,v}} &= \int_{\mathbb{T}^{d}} \int_{0}^{t} \sum_{n \in \mathbb{Z}^{d} - \{0\}} g(t, s, k, x) \langle P(n) \hat{u}_{\varepsilon}, f \rangle_{L^{2}_{v}} ds \\ &= \int_{\mathbb{T}^{d}} \int_{0}^{t} \sum_{n \in \mathbb{Z}^{d} - \{0\}} g(t, s, k, x) \langle \hat{u}_{\varepsilon}, P(n) f \rangle_{L^{2}_{v}} ds. \end{aligned}$$
(4.C.3)

We are working in  $L_x^2 L_v^2$  in order to simplify computations as they are exactly the same in higher Sobolev spaces. Therefore, we can assume that hypothesis (H4) is still valid in  $L_v^2$  without loss of generality. This means

$$\langle \hat{u}_{\varepsilon}, P(n)f \rangle_{L^2_v} \leq \|h\|_{L^2_x L^2_v} \|h\|_{\Lambda_v} \|P(n)f\|_{\Lambda_v}.$$
 (4.C.4)

Finally, in terms of Fourier coefficients in x, P(n) is a projector in  $L_v^2$  and uniformely bounded in n as an operator in  $L_v^2$ .

Thus, combining (4.C.4) and the definition of the functional E, (4.6.2), we see that

$$\int_0^T \left\| \hat{f}_{\varepsilon} \right\|_{L^2_x L^2_v}^2 dt$$

is a continuous operator from  $C(\mathbb{R}^+, L^2_x L^2_v, E(\cdot))$  to  $C(\mathbb{R}^+, L^2_x L^2_v, \|\cdot\|_{L^2_x L^2_v})$ . Looking at (4.C.3), we can consider without loss of generality that the following holds (even for the remainder term) for all T > 0:

$$\psi_{ij}^{\varepsilon}(u_{\varepsilon}) = \int_{0}^{t} \sum_{n \in \mathbb{Z}^{d} - \{0\}} g(t, s, k, x) \hat{f}_{\varepsilon}(s, k, v) ds$$

with

$$\int_0^T \left\| \hat{f}_{\varepsilon} \right\|_{L^2_x L^2_v}^2 dt \leqslant M_{ij} E(h_{\varepsilon})^2$$

## 4.C.2.2 Proof of Lemma 4.8.9

For the first inequality, fix T > 0 and integrate by part in t to obtain

$$\begin{split} \int_0^T \psi_{0j}^{\varepsilon}(u_{\varepsilon}) dt &= \sum_{n \in \mathbb{Z}^d - \{0\}} e^{in.x} \int_0^T \left( \int_0^t e^{i\frac{\alpha_j(t-s)}{\varepsilon} |n| - (t-s)\beta_j |n|^2} |n| \, \widehat{f}_{\varepsilon}(s) ds \right) dt \\ &= \sum_{n \in \mathbb{Z}^d - \{0\}} e^{in.x} \frac{\varepsilon}{i\alpha_j |n| - \varepsilon\beta_j |n|} \left[ \int_0^T \left( e^{i\frac{\alpha_j(T-s)}{\varepsilon} |n| - (T-s)\beta_j |n|^2} - 1 \right) \widehat{f}_{\varepsilon}(s) ds \right]. \end{split}$$

Finally we can use Parseval's identity

$$\begin{split} \left\| \int_0^T \psi_{0j}^{\varepsilon}(u_{\varepsilon}) dt \right\|_{L^2_x L^2_v}^2 &\leqslant \quad \sum_{n \in \mathbb{Z}^d - \{0\}} \frac{\varepsilon^2}{\varepsilon^2 \beta_j^2 \left|n\right|^2 + \alpha_j^2} T \int_0^T 2 \left\| \hat{f}_{\varepsilon}(s, n, v) \right\|_{L^2_v}^2 ds \\ &\leqslant \quad \frac{2M_{1j}^2}{\alpha_j^2} T \varepsilon^2 E(h_{\varepsilon})^2, \end{split}$$

where we used the subsection above and Parseval's identity again. This is exactly the expected result.

## 4.C.2.3 Proof of Lemma 4.8.11

We divide this proof in three paragraphes, each of them studying a different term.

**The term**  $\psi_{1j}^{\varepsilon}$ : We will just prove the last two inequalities and then merely applying Cauchy-Schwarz inequality will lead to the first one.

Fix t > 0. By a change of variable we can write

$$\psi_{1j}^{\varepsilon}(u_{\varepsilon}) = \sum_{n \in \mathbb{Z}^d - \{0\}} e^{ik.x} \chi_{|\varepsilon n| \leqslant n_0} \int_0^t e^{\frac{i\alpha_j s}{\varepsilon} |n| - \beta_j s |n|^2} \left( e^{\frac{s}{\varepsilon^2} \gamma_j(|\varepsilon n|)} - 1 \right) |n| \, \hat{f}_{\varepsilon}(t-s) ds.$$

By the study made in the proof of Lemma 4.8.7 we have that

$$\begin{split} \left| \int_0^t e^{\frac{i\alpha_j s}{\varepsilon} |n| - \beta_j s |n|^2} \left( e^{\frac{s}{\varepsilon^2} \gamma_j (|\varepsilon n|)} - 1 \right) |n| \, \hat{f}_{\varepsilon}(t-s) ds \right| \\ \leqslant C_\gamma \, |n|^4 \, \varepsilon \int_0^t s e^{-\frac{\beta_j s}{2} |n|^2} \left| \hat{f}_{\varepsilon}(t-s) \right| ds. \end{split}$$

Then we use the computational inequality (4.C.1) and a Cauchy-Schwarz to obtain

$$\begin{split} \left| \int_{0}^{t} e^{\frac{i\alpha_{j}s}{\varepsilon} |n| - \beta_{j}s|n|^{2}} \left( e^{\frac{s}{\varepsilon^{2}}\gamma_{j}(|\varepsilon n|)} - 1 \right) |n| \, \hat{u}_{\varepsilon} ds \right| \\ &\leq \varepsilon C_{\gamma} C_{1} \left( \frac{\beta_{j}}{4} \right) |n|^{2} \int_{0}^{t} e^{-\frac{\beta_{j}s}{4}|n|^{2}} \left| \hat{f}_{\varepsilon} \right| ds \\ &\leq \varepsilon C_{\gamma} C_{1} \left( \frac{\beta_{j}}{4} \right) |n|^{2} \sqrt{\frac{4}{\beta_{j} |n|^{2}}} \left[ \int_{0}^{t} e^{-\frac{\beta_{j}(t-s)}{4}|n|^{2}} \left| \hat{f}_{\varepsilon} \right|^{2} ds \right]^{1/2}.$$
(4.C.5)

We can obtain the result by using Parseval's identity, denoting C a constant independent of  $\varepsilon$  and T, the continuity of  $P_{1j}$  and the computational inequality (4.C.1).

$$\left\|\psi_{1j}^{\varepsilon}(u_{\varepsilon})(t)\right\|_{L^{2}_{x}L^{2}_{v}}^{2} \leqslant C \sum_{n \in \mathbb{Z}^{d} - \{0\}} \chi_{|\varepsilon n| \leqslant n_{0}} \varepsilon^{2} |n|^{2} \int_{0}^{t} e^{-\frac{\beta_{j}(t-s)s}{4}|n|^{2}} \left\|\hat{f}_{\varepsilon}(s)\right\|_{L^{2}_{v}}^{2} ds$$

If we merely bound  $e^{-\frac{\beta_j(t-s)}{2}|n|^2}$  by one and use the fact that  $\chi_{|\varepsilon n| \leq n_0} \leq 1$  and  $\chi_{|\varepsilon n| \leq n_0} \varepsilon^2 |n|^2 \leq n_0^2$  we obtain the third inequality of the lemma for  $\delta = 1$  and  $\delta = 0$ . Then by interpolation we obtain the general case for  $0 \leq \delta \leq 1$ .

If we integrate in t between 0 and a fixed T > 0, a mere integration by part yields the expected control on the  $L^2_{t,x,v}$ -norm. Finally, from the latter control and a Cauchy-Schwarz inequality we deduce the first inequality.

**The term**  $\psi_{2j}^{\varepsilon}$ : As in the case  $\psi_{1j}^{\varepsilon}$ , we are going to prove the third inequality only. Fix T > 0, a change of variable gives us

$$\psi_{2j}^{\varepsilon}(u_{\varepsilon}) = \sum_{n \in \mathbb{Z}^d - \{0\}} e^{ik \cdot x} \chi_{|\varepsilon n| \leq n_0} \int_0^T e^{\frac{i\alpha_j s}{\varepsilon} |n| - \beta_j s |n|^2 + \frac{s}{\varepsilon^2} \gamma_j(|\varepsilon n|)} \varepsilon |n|^2 \hat{f}_{\varepsilon}(T - s) ds.$$

We can see that

$$\left|\int_0^T e^{\frac{i\alpha_j s}{\varepsilon}|n|-\beta_j s|n|^2 + \frac{s}{\varepsilon^2}\gamma_j(|\varepsilon n|)} \varepsilon |n|^2 \hat{f}_{\varepsilon}(T-s)ds\right| \leqslant \varepsilon |n|^2 \int_0^T e^{-\frac{\beta_j s}{2}|n|^2} \left|\hat{f}_{\varepsilon}(T-s)\right| ds.$$

This bound is of the same form as equation (4.C.5). Therefore we have the same result.

The term  $\psi_{3j}^{\varepsilon}$ : As above, we will show the third inequality only. Fix T > 0, we can write

$$\psi_{3j}^{\varepsilon}(u_{\varepsilon}) = \sum_{n \in \mathbb{Z}^{d} - \{0\}} e^{ik.x} (\chi_{|\varepsilon n| \leqslant n_{0}} - 1) \int_{0}^{T} e^{\frac{i\alpha_{j}s}{\varepsilon} |n| - \beta_{j}s|n|^{2}} |n| \hat{f}_{\varepsilon}(T - s, n, v) ds.$$

Looking at the fact that  $|\chi_{|\varepsilon n| \leq n_0} - 1| \leq \frac{\varepsilon |n|}{n_0}$ , we find the same kind of inequality as equation (4.C.5). Thus, we reach the same result.

## 4.C.2.4 Proof of Lemma 4.8.12

We remind the reader that

$$\Psi_R^{\varepsilon}(u_{\varepsilon}) = \int_0^t \frac{1}{\varepsilon} U_R^{\varepsilon}(t-s) f_{\varepsilon}(s) ds,$$

and that, by Theorem 4.8.3,

$$\|U_R^{\varepsilon}f_{\varepsilon}\|_{L^2_xL^2_v}^2 \leqslant C_R^2 e^{-2\frac{\sigma t}{\varepsilon^2}} \|f_{\varepsilon}\|_{L^2_xL^2_v}^2.$$

Hence, a Cauchy-Schwarz inequality gives us the third inequality for  $\|\psi_R^{\varepsilon}(u_{\varepsilon})(T)\|_{L^2_x L^2_v}^2$ , and then the two others inequality stated above.

### 4.C.2.5 Proof of Lemma 4.8.14

We remind the reader that

$$\Psi(u) = \mathcal{F}_x^{-1} \left[ \psi_{00}^{\varepsilon}(u) + \psi_{02}^{\varepsilon}(u) \right] \mathcal{F}_x.$$

As above, and because in that case  $\alpha_j = 0$ , we can write  $\psi_{0j}^{\varepsilon}(u_{\varepsilon} - u)(T)$ , for some T > 0, and apply a Cauchy-Schwarz inequality:

$$\begin{split} \left\|\psi_{0j}^{\varepsilon}(u_{\varepsilon}-u)\right\|_{L^{2}_{x}L^{2}_{v}}^{2}(T) &= \sum_{n\in\mathbb{Z}^{d}-\{0\}}|n|^{2}\int_{\mathbb{R}^{d}}\left|\int_{0}^{T}e^{-s\beta_{j}|n|^{2}}P_{1j}\widehat{\Gamma}(h_{\varepsilon}-h,h_{\varepsilon}+h)ds\right|^{2}dv\\ &\leqslant \frac{M^{2}_{1j}}{\beta_{j}^{2}}\sup_{t\in[0,T]}\left\|\Gamma(h_{\varepsilon}-h,h_{\varepsilon}+h)\right\|_{L^{2}_{x}L^{2}_{v}}^{2}.\end{split}$$

But because  $\mathbb{T}^d$  is bounded in  $\mathbb{R}^d$  and thanks to (H4) and the boundedness of  $(h_{\varepsilon})_{\varepsilon}$ and h (both bounded by M) in  $H^s_x L^2_v$  (Theorem 4.2.3), we can have the following control:

$$\|\Gamma(h_{\varepsilon} - h, h_{\varepsilon} + h)\|_{L^{2}_{x}L^{2}_{v}}^{2} \leqslant 4M^{2}C^{2}_{\Gamma} \text{Volume}(\mathbb{T}^{d}) \|h_{\varepsilon} - h\|_{L^{\infty}_{x}L^{2}_{v}}.$$

Therefore we obtain the last inequality and the first two just come from Cauchy-Schwarz inequality.
## Chapter 5

## The Incompressible Navier-Stokes limit in polynomial weighted spaces

We study the Boltzmann equation on the d-dimensional torus in a perturbative setting around a global equilibrium under the Navier-Stokes linearisation. We use a recent functional analysis breakthrough to prove that the linear part of the equation generates a  $C^{0}$ semigroup with exponential decay in Sobolev spaces with polynomial weight, independently on the Knudsen number. Finally we show a Cauchy theory and an exponential decay for the perturbed Boltzmann equation, uniformly in the Knudsen number, in Sobolev spaces with polynomial weight. The polynomial weight is almost optimal and furthermore, this result only requires derivatives in the space variable and allows to connect to solutions to the incompressible Navier-Stokes equations in these spaces.

This is a joint work with Sara Merino-Aceituno and Clément Mouhot, both from the University of Cambridge.

## Contents

5.1	Intr	oduction $\ldots \ldots 221$
	5.1.1	The problem and its motivations
	5.1.2	Existing results
	5.1.3	Our contributions and strategy
	5.1.4	Organization of the chapter
<b>5.2</b> Main results		
	5.2.1	Notations
	5.2.2	Results about the full linear part
	5.2.3	Existence, uniqueness and trend to equilibrium
5.3	The	linear part: a $C^0$ -semigroup in spaces with polynomial
	weig	th, proof of Theorem 5.2.1
	5.3.1	Strategy of the proof
	5.3.2	Decomposition of the operator and assumption $(A2)(ii)$ 232
	5.3.3	Dissipativity estimates for $\mathcal{B}_{\varepsilon}^{(\delta)}$ , assumption (A2)(i) 234
	5.3.4	Estimates on the iterated convolution product, assumption $(A2)(iii)238$
	5.3.5	Proof of Theorem 5.2.1
5.4	An	a priori estimate for the full perturbed equation: proof
of Theorem 5.2.3		
	5.4.1	Description of the problem and notations
	5.4.2	Study of equation (5.4.3) in $\mathcal{E}$
	5.4.3	Study of equations (5.4.4) in $E$
	5.4.4	Proof of Theorem 5.4.1

## 5.1 Introduction

This chapter deals with the Boltzmann equation in a perturbative setting as the Knudsen number tends to zero. This equation rules the dynamics of rarefied gas particles moving on the flat torus in dimension d,  $\mathbb{T}^d$ , when the only interactions taken into account are binary collisions. More precisely, the Boltzmann equation describes the time evolution of the distribution f = f(t, x, v) of particles in position x and velocity v. A formal derivation of the Boltzmann equation from Newton's laws under the rarefied gas assumption can be found in [28], while [30] presents Lanford's Theorem (see [65] and [44] for detailed proofs) which rigorously proves the derivation in short times.

We denote the Knudsen number by  $\varepsilon$  and the Boltzmann equation reads

$$\partial_t f + v \cdot \nabla_x f = \frac{1}{\varepsilon} Q(f, f) , \text{ on } \mathbb{T}^d \times \mathbb{R}^d,$$

where Q is the Boltzmann collision operator given by

$$Q(f,f) = \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} B\left(|v - v_*|, \cos \theta\right) \left[f'f'_* - ff_*\right] dv_* d\sigma.$$

The Boltzmann kernel operator B encodes the physics of the collision process and f',  $f_*$ ,  $f'_*$  and f are the values taken by f at v',  $v_*$ ,  $v'_*$  and v respectively, where

$$v' = \frac{v + v_*}{2} + \frac{|v - v_*|}{2}\sigma$$
  

$$v'_* = \frac{v + v_*}{2} - \frac{|v - v_*|}{2}\sigma$$
, and  $\cos \theta = \left\langle \frac{v - v_*}{|v - v_*|}, \sigma \right\rangle.$ 

The Boltzmann collision operator comes from a symmetric bilinear operator Q(g, h)defined by

$$Q(g,h) = \frac{1}{2} \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} B\left( |v - v_*|, \cos \theta \right) \left[ h'g'_* + h'_*g' - hg_* - h_*g \right] dv_* d\sigma.$$

It is well-known (see [28], [30] or [46] for example) that the global equilibria for the Boltzmann equation are the *Maxwellians*, which are gaussian density functions depending only on the v variable. Without loss of generality we consider only the case of normalized Maxwellians:

$$\mu(v) = \frac{1}{(2\pi)^{\frac{d}{2}}} e^{-\frac{|v|^2}{2}}.$$

In this chapter we will assume that the Boltzmann collision kernel is of the following

form

$$B\left(|v - v_*|, \cos\theta\right) = \Phi\left(|v - v_*|\right) b\left(\cos\theta\right),\tag{5.1.1}$$

with  $\Phi$  and b positive functions. This hypothesis is satisfied for all physical model and is more convenient to work with but do not impede the generality of our results.

We also restrict ourselves to the case of hard potential or Maxwellian potential ( $\gamma = 0$ ), that is to say there is a constant  $C_{\Phi} > 0$  such that

$$\Phi(z) = C_{\Phi} z^{\gamma}, \quad \gamma \in [0, 1], \tag{5.1.2}$$

with a strong form of Grad's *angular cutoff* (see [48]), expressed here by the fact that we assume b to be  $C^1$  with the controls from above

$$\forall z \in [-1, 1], \quad b(z), \ b(z') \leqslant C_b.$$
 (5.1.3)

## 5.1.1 The problem and its motivations

The Knudsen number is the inverse of the average number of collisions for each particle per unit of time. Therefore, as reviewed in [111], one can expect a convergence, in some sense, from the Boltzmann model towards the acoustics and the fluids dynamics as the Knudsen number tends to 0. However, these different models describe physical phenomena that do not evolve at the same timescale and the right rescaling to approximate the incompressible Navier-Stokes equation (see [8][46][111][98]) is the following equation

$$\partial_t f_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x f_{\varepsilon} = \frac{1}{\varepsilon^2} Q(f_{\varepsilon}, f_{\varepsilon}) , \text{ on } \mathbb{T}^d \times \mathbb{R}^d,$$
 (5.1.4)

under the linearization  $f_{\varepsilon}(t, x, v) = \mu(v) + \varepsilon h_{\varepsilon}(t, x, v)$ . This leads to the perturbed Boltzmann equation

$$\partial_t h_{\varepsilon} + \frac{1}{\varepsilon} v \cdot \nabla_x h_{\varepsilon} = \frac{1}{\varepsilon^2} \mathcal{L}(h_{\varepsilon}) + \frac{1}{\varepsilon} Q(h_{\varepsilon}, h_{\varepsilon}), \qquad (5.1.5)$$

where we defined

$$\mathcal{L}(h) = 2Q(\mu, h).$$

The hydrodynamical limit of the perturbed equation is the system of equations satisfied by the limit, as  $\varepsilon$  tends to 0, of the hydrodynamical fluctuations that are the following physical observables of  $h_{\varepsilon}$ :

$$\begin{split} \rho_{\varepsilon}(t,x) &= \int_{\mathbb{R}^d} h_{\varepsilon}(t,x,v) \, dv, \\ u_{\varepsilon}(t,x) &= \int_{\mathbb{R}^d} v h_{\varepsilon}(t,x,v) \, dv, \\ \theta_{\varepsilon}(t,x) &= \frac{1}{d} \int_{\mathbb{R}^d} (|v|^2 - d) h_{\varepsilon}(t,x,v) \, dv \end{split}$$

Note that  $(\rho_{\varepsilon}, u_{\varepsilon}, \theta_{\varepsilon})$  are the linearised fluctuations of the mass, momentum and the thermal energy around the global equilibrium  $\mu$ .

In our perturbative framework, previous studies [8][10][23] (and also Chapter 4) show that the hydrodynamical limits  $\rho$ , u and  $\theta$  are the weak (in the Leray sense [66]) solutions of the linearized incompressible Navier-Stokes equations:

$$\partial_t u - \nu \Delta u + u \cdot \nabla u + \nabla p = 0,$$
  

$$\nabla \cdot u = 0,$$
  

$$\partial_t \theta - \kappa \Delta \theta + u \cdot \nabla \theta = 0,$$
  
(5.1.6)

where p is the pressure function and  $\nu$  and  $\kappa$  are constants determined by L (see [8] or [46] Theorem 5). They also satisfy the Boussineq relation

$$\nabla(\rho + \theta) = 0. \tag{5.1.7}$$

The aim of the present chapter is to use a constructive method to obtain existence and exponential decay for solutions to the perturbed Boltzmann equation (5.1.4), uniformly in the Knudsen number. One will thus be allowed to extract a converging (at least weakly) subsequence of  $h_{\varepsilon}$  converging to the incompressible Navier-Stokes equations [10][8][23] (see also Chapter 4). Such uniform results have been obtained on the torus in Sobolev spaces with exponential weight  $H_{x,v}^s(\mu^{-1/2})$  in [56][23] and the present work improves this strong weight to a polynomial weight without the need of derivatives in the velocity variable (see also Chapter 4).

#### 5.1.2 Existing results

The first part of our work is to prove that the linear part of the Boltzmann equation

$$\mathcal{G}_{\varepsilon} = \frac{1}{\varepsilon^2} \mathcal{L} - \frac{1}{\varepsilon} v \cdot \nabla_x$$

generates a strongly continuous semigroup with an exponential decay in Lebesgue and Sobolev spaces with polynomial weight, namely  $1 + |v|^k$  for some k large enough.

It has been known for long that the linear Boltzmann operator  $\mathcal{L}$  is a self-adjoint non positive linear operator in the space  $L_v^2(\mu^{-1/2})$ . Moreover it has a spectral gap  $\lambda_0$ . This has been proved in [27][48][49] with non constructive methods for hard potential with cutoff and in [14][15] in the Maxwellian case. These results were made constructive in [4][79] for more general collision operators. One can easily extend this spectral gap to Sobolev spaces  $H_v^s(\mu^{-1/2})$  (see for instance [51] Section 4.1).

The next step is to see if the latter properties about  $\mathcal{L}$  in the velocity space can be

transposed when one adds the skew-symmetric transport operator  $-v \cdot \nabla_x$ . The first results were obtained in [107] where  $\mathcal{G}_1$  was proven to generate a strong continuous semigroup in  $L_v^2 H_x^s (\mu^{-1/2})$  and in  $L_v^{\infty} H_x^s (\mu^{-1/2}(1+|v|)^k)$ , for *s* and *k* large enough. Then [82] obtained constructively this result in  $H_{x,v}^s (\mu^{-1/2})$  using hypocoercivity properties of the Boltzmann linear operator. Finally, a recent breakthrough proving abstract extension of semigroups [51] showed that  $\mathcal{G}_1$  generates a  $C^0$ -semigroup in all the Sobolev spaces of the form  $W_v^{\alpha,q} W_x^{\beta,p}(m)$ , for *m* being an exponential weight (including maxwellian density if q = p = 2) or a polynomial weight  $(1 + |v|)^k$ , as long as  $\alpha \leq \beta$  and *k* is large enough depending on *q* (with k > 2 in the case q = 1).

The full Boltzmann equation perturbed around a global equilibrium  $\mu(v)$  (5.1.5) has also been studied in the case  $\varepsilon = 1$ . The associated Cauchy problem has been worked on over the past fifty years, starting with Grad [50], and it has been studied in different spaces, such as  $L_v^2 H_x^s (\mu^{-1/2})$  spaces [107] or  $H_{x,v}^s (\mu^{-1/2}(1+|v|)^k)$  [55][114]. The Cauchy theory was then extended to  $H_{x,v}^s (\mu^{-1/2})$  where an exponential trend to equilibrium has also been obtained. This was obtained using hypocoercivity properties of the linear operator [82] or nonlinear estimates on fluid and microscopic parts of the equation [56]. Recently, [51] proved existence and uniqueness for (5.1.5) in more the general spaces  $(W_v^{\alpha,1} \cap W_v^{\alpha,q}) W_x^{\beta,p} (1+|v|)^k)$  for  $\alpha \leq \beta$  and  $\beta$  and k large enough with explicit thresholds. This result therefore gets rid of the exponential weight needed in the previous studies.

All the results presented above hold in the case of the torus. We refer the reader interested in the Cauchy problem, both for the torus and the whole space, to the review [110].

For physical purposes, these studies for  $\varepsilon = 1$  are relevant since mere rescalings or changes of physical units changes (5.1.4) to the case where the Knudsen number equals 1. However, if one wants to study the hydrodynamical limits of the Boltzmann equation, one needs to obtain explicit dependencies on the Knudsen number. Using hypocoercivity methods [23] (see also Chapter 4) gave a constructive uniform approach on the semigroup generated by  $\mathcal{G}_{\varepsilon}$  in  $H^s_{x,v}$  ( $\mu^{-1/2}$ ) and its exponential decay. The study of the full perturbed Boltzmann equation (5.1.5) taking into account the dependencies on the Knudsen number has been obtained [56][23] in the same spaces  $H^s_{x,v}$  ( $\mu^{-1/2}$ ), for *s* large enough (see also Chapter 4). More precisely, for initial data sufficiently close to  $\mu$  there exists a unique non-negative solution to (5.1.4) and it decays exponentially fast towards its equilibrium. The smallness assumption was proven to be independent of the Knudsen number as well as the rate of decay and the methods used in [23] (see also Chapter 4) are constructive.

## 5.1.3 Our contributions and strategy

The present work brings two major improvements.

In the spirit of [51], we first prove that  $\mathcal{G}_{\varepsilon}$  generates a strong continuous semigroup in Sobolev spaces  $W_v^{\alpha,1}W_x^{\beta,p}\left(1+|v|\right)^k\right)$  for  $\alpha \leq \beta$  and  $\beta$  and k large enough with explicit thresholds. It is done by starting from existing results in  $H_{x,v}^s\left(\mu^{-1/2}\right)$  and then decomposing the linear operator  $\mathcal{G}_{\varepsilon}$  into a dissipative part and a regularising part that is then treated in more and more regular spaces up to the space where the semigroup properties have been derived in previous articles. We thus improve the existing result [23], also given in Chapter 4. Our main contribution is an adapted version of the abstract extension theorem developed in [51] that takes into account the dependencies on the Knudsen number as well as a careful study of the dissipative and the regularising parts of the operator  $\mathcal{G}_{\varepsilon}$ .

The second contribution of this chapter is the solution to the Cauchy problem with exponential trend to equilibrium, independently on  $\varepsilon$ , in spaces

$$W_v^{\alpha,1} W_x^{\beta,1} \left( 1 + |v|^{2+0} \right)$$
 and  $W_v^{\alpha,1} H_x^{\beta} \left( 1 + |v|^{2+0} \right)$ 

for  $\beta$  large enough and all  $\alpha \leq \beta$ . First, this result makes the recent study [51] uniform in the Knudsen number. Second, it improves the Cauchy theory developed uniformly in  $\varepsilon$ in [56][23] by dropping the exponential weight and the *v*-derivatives. Moreover, one can notice that the polynomial weight is almost the optimal one for the Boltzmann equation (conservation of mass and energy).

The main issue to obtain uniform results is that the bilinear operator  $\varepsilon^{-1}Q$  cannot be treated as a mere perturbation that evolves under the flow of  $S_{\mathcal{G}_{\varepsilon}}$ , the semigroup generated by  $\mathcal{G}_{\varepsilon}$ , since the latter has an exponential decay of order O(1) that is negligeable compared to  $O(\varepsilon^{-1})$  as  $\varepsilon$  tends to zero. We develop an analytic point of view about the extension theorem in [51] and include the bilinear term. More precisely, we decompose the perturbed equation (5.1.5) into a hierarchy of equations taking place in spaces that have more and more regularity up to  $H^s_{x,v}(\mu^{-1/2})$  where estimates had been derived in [23] (see also Chapter 4). At each step we use the dissipative part of the linear operator to control the remainder term  $\varepsilon^{-1}Q$  whereas the regularising part is controlled in spaces with higher regularity.

## 5.1.4 Organization of the chapter

Section 5.2 first introduces the different notations and definitions we will use throughout the chapter and then states the precise theorems we prove in this work. Section 5.2.2 deals with the semigroup generated by the full linear operator  $\varepsilon^{-2}\mathcal{L} - \varepsilon^{-1}v \cdot \nabla_x$  whereas Section 5.2.3 is dedicated to the full Boltzmann equation.

The full linear part of the Boltzmann operator is proven to generate a strongly continuous semigroup in Lebesgue and Sobolev spaces with polynomial weight in Section 5.3.

We start with Section 5.3.1, a thorough description of our strategy and a version of the extension theorem of [51] that takes into account the dependencies in  $\varepsilon$ .

We show in this section that  $\varepsilon^{-2}\mathcal{L} - \varepsilon^{-1}v \cdot \nabla_x$  can be decompose into a regularising operator in the velocity variable (Section 5.3.2) and a dissipative one (Section 5.3.3).

We then combine the last two properties to gain regularity both in space and velocity (Section 5.3.4) to finally prove the existence and exponential decay of the associated semigroup (Section 5.3.5).

The last section, Section 5.4, proves existence, uniqueness and exponential decay of solutions to the perturbed Boltzmann equation (5.1.5).

Section 5.4.1 gives a new point of view on the extension we used to generate the semigroup associated to  $\varepsilon^{-2}\mathcal{L} - \varepsilon^{-1}v \cdot \nabla_x$  and how it can be used with the bilinear operator. This strategy is developed through Sections 5.4.2 and 5.4.3 and it leads to the proof of the exponential decay towards equilibrium in Section 5.4.4.

## 5.2 Main results

#### 5.2.1 Notations

We gather here the notations we will use throughout this chapter.

Function spaces. We first define the following shorthand notation,

$$\langle \cdot \rangle = \sqrt{1 + |\cdot|^2}.$$

The convention we choose is to index the space by the name of the concerned variable so we have, for p in  $[1, +\infty]$ ,

$$L^{p}_{[0,T]} = L^{p}([0,T]), \quad L^{p}_{x} = L^{p}(\mathbb{T}^{d}), \quad L^{p}_{v} = L^{p}(\mathbb{R}^{d}).$$

Let p and q be in  $[1, +\infty)$ ,  $\alpha$  and  $\beta$  in  $\mathbb{N}$  and  $m : \mathbb{R}^d \longrightarrow \mathbb{R}^+$  a strictly positive measurable function. For any multi-indexes  $j = (j_1, \ldots, j_d)$  and  $l = (l_1, \ldots, l_d)$  in  $\mathbb{N}^d$  we denote the  $(j, l)^{th}$  partial derivative by

$$\partial_l^j = \partial_x^l \partial_v^j.$$

We define the space  $W_v^{\alpha,q} W_x^{\beta,p}(m)$  by the norm

$$\|f\|_{W^{\alpha,q}_v W^{\beta,p}_x(m)} = \sum_{\substack{|j| \leqslant \alpha, |l| \leqslant \beta \\ |l|+|j| \leqslant \max(\alpha,\beta)}} \left\| \left(\partial_l^j f\right) m \right\|_{L^q_v L^p_x},$$

where we used the Lebesgue norm

$$||g||_{L^{q}_{v}L^{p}_{x}} = \left[ \int_{\mathbb{R}^{d}} \left( \int_{\mathbb{T}^{d}} |f(x,v)|^{p} dx \right)^{q/p} dv \right]^{1/q}.$$

**Linear Boltzmann operator.** First we use a writing convention. This chapter aims at extending results known in a small space E, namely  $H_{x,v}^s(\mu^{-1/2})$  with s sufficiently large, into a larger space  $\mathcal{E}$ , namely Lebesgue and Sobolev spaces with polynomial weight. We will use curly letters for operators in  $\mathcal{E}$  and their non-curly equivalent to denote their restriction to E. For instance, we will denote

$$\mathcal{L}|_E = L$$

The linear Boltzmann operator L has several properties we will use throughout this chapter (see [28][30][112][51] for instance).

L is a closed self-adjoint operator in  $L^2_v\left(\mu^{-1/2}\right)$  with kernel

$$\operatorname{Ker}(L) = \operatorname{Span}\left\{\phi_0(v), \dots, \phi_{d+1}(v)\right\}\mu_{d+1}(v)$$

where  $\phi_0(v) = 1$ , for i = 1, ..., d we defined  $\phi_i(v) = v_i$  and  $\phi_{d+1} = \left(|v|^2 - d\right) / \sqrt{2d}$ . The family  $(\phi_i)_{0 \leq i \leq d+1}$  is an orthonormal basis of Ker (L) in  $L_v^2(\mu^{-1/2})$  and we denote  $\pi_L$  the orthogonal projection onto Ker (L) in  $L_v^2(\mu^{-1/2})$ :

$$\pi_L(h) = \sum_{i=0}^{d+1} \left( \int_{\mathbb{R}^d} h(u)\phi_i(u) \, du \right) \phi_i(v)\mu(v), \tag{5.2.1}$$

and we define  $\pi_L^{\perp} = \mathrm{Id} - \pi_L$ . We will also denote the full linear Boltzmann operator by

$$\mathcal{G}_{\varepsilon} = \frac{1}{\varepsilon^2} L - \frac{1}{\varepsilon} v \cdot \nabla_x \cdot$$

For s in  $\mathbb{N}$  we will use the convention

$$(\mathcal{G}_{\varepsilon})|_{H^s_{x,v}(\mu^{-1/2})} = G_{\varepsilon}$$

It has been proven ([23] Proposition 3.1 or see Proposition 4.3.1 in Chapter 4) that the kernel of  $G_{\varepsilon}$  does not depend on  $\varepsilon$  and that its generators in  $L^2_{x,v}(\mu^{-1/2})$  are the same

than the ones of Ker (L). We therefore have that the orthogonal projection onto Ker  $(G_{\varepsilon})$ in  $L^2_{x,v}(\mu^{-1/2})$  is given by

$$\Pi_G(h) = \Pi_{G_{\varepsilon}}(h) = \sum_{i=0}^{d+1} \left( \int_{\mathbb{T}^d \times \mathbb{R}^d} h(x, u) \phi_i(u) \, dx \, du \right) \phi_i(v) \mu(v), \tag{5.2.2}$$

and we define  $\Pi_G^{\perp} = \mathrm{Id} - \Pi_G$ .

Note that for a function h in  $L^2_{x,v}(\mu^{-1/2})$  we have that

$$\forall (x,v) \in \mathbb{T}^d \times \mathbb{R}^d, \quad \Pi_G(h)(x,v) = \int_{\mathbb{T}^d} \pi_L(h(x_*,\cdot))(v) \, dx_*.$$

## 5.2.2 Results about the full linear part

We first deal with  $\mathcal{G}_{\varepsilon}$ , the linear part of the perturbed Boltzmann operator. We prove that it generates a strongly continuous semigroup with an exponential decay in Lebesgue and Sobolev spaces with a weight  $\langle v \rangle^k$  as long as k is large enough. The precise statement is the following.

**Theorem 5.2.1** Let B be a Boltzmann collision kernel satisfying (5.1.1)-(5.1.2)-(5.1.3). There exists  $0 < \varepsilon_d \leq 1$  such that for all p, q in  $[1, +\infty]$ , all  $\alpha$ ,  $\beta$  in  $\mathbb{N}$  with  $\alpha \leq \beta$  and all  $k > k_a^*$ , where

$$k_q^* = \frac{3 + \sqrt{49 - 48/q}}{2} + \gamma \left(1 - \frac{1}{q}\right), \tag{5.2.3}$$

with  $\gamma$  defined in (5.1.2),

- 1. for all  $0 < \varepsilon \leq \varepsilon_d$ ,  $\mathcal{G}_{\varepsilon} = \varepsilon^{-2} \mathcal{L} \varepsilon^{-1} v \cdot \nabla_x$  generates a  $C^0$ -semigroup,  $S_{\mathcal{G}_{\varepsilon}}(t)$ , on  $W_v^{\alpha,q} W_x^{\beta,p} (\langle v \rangle^k)$ ,
- 2. for all  $\tau > 0$ , there exist  $C_{\mathcal{G}}(\tau), \lambda_0 > 0$ , such that for all  $0 < \varepsilon \leq \varepsilon_d$  and for all  $h_{in}$ in  $W_v^{\alpha,q} W_x^{\beta,p}(\langle v \rangle^k)$ , for all  $t \ge \tau$

$$\|S_{\mathcal{G}_{\varepsilon}}(t)(h_{in}) - \Pi_{\mathcal{G}}(h_{in})\|_{W_{v}^{\alpha,q}W_{x}^{\beta,p}(\langle v \rangle^{k})} \leq C_{\mathcal{G}}(\tau)e^{-\lambda_{0}t}\|h_{in} - \Pi_{\mathcal{G}}(h_{in})\|_{W_{v}^{\alpha,q}W_{x}^{\beta,p}(\langle v \rangle^{k})},$$

where  $\Pi_{\mathcal{G}}$  is the spectral projector onto  $\operatorname{Ker}(\mathcal{G}_{\varepsilon})$  which is given, for all  $\varepsilon$ , by

$$\Pi_{\mathcal{G}}(g) = \sum_{i=0}^{d+1} \left( \int_{\mathbb{T}^d \times \mathbb{R}^d} g\phi_i \, dx dv \right) \phi_i \mu.$$
(5.2.4)

The constants  $\varepsilon_d$ ,  $C_{\mathcal{G}}(\tau)$  and  $\lambda_0$  are constructive and only depends on d, p, q, k,  $\alpha$ ,  $\beta$  and the kernel of the Boltzmann operator.

We refer to [61] and [51] Section 2 for definitions and properties of spectral projectors.

Remark 5.2.2 We can make a couple of remarks about this theorem.

- 1. It has been proven in [23] Section 3 (or Chapter 4 Section 4.3), that in  $H^1_{x,v}(\mu^{-1/2})$ , Ker  $(G_{\varepsilon})$  does not depend on  $\varepsilon$  if  $\varepsilon$  is positive and we therefore can define  $\Pi_G = \Pi_{G_{\varepsilon}}$ . During the proof of Theorem 5.2.1 we will show that  $(\Pi_{\mathcal{G}_{\varepsilon}})|_{H^s_{x,v}(\mu^{-1/2})} = \Pi_{G_{\varepsilon}}$  and thus  $\Pi_{\mathcal{G}}$  is well-defined and is independent of  $\varepsilon$  and given by (5.2.2).
- 2. As noticed in [51], the rate of decay  $\lambda_0$  can be taken equal to the spectral gap of  $\mathcal{L}|_{H^s_{x,v}(\mu^{-1/2})}$  (see [23] or Chapter 4), for s as large as wanted, when k is big enough (and we obtained a constructive threshold).
- 3. Finally, we emphasize that in the case q = 1, the result holds for all k > 2. This is almost the minimal regularity  $L_v^2(1+|v|^2)$  for the Boltzmann equation, that is solutions with bounded mass and energy.

## 5.2.3 Existence, uniqueness and trend to equilibrium

A physically relevant requirement for solutions to the Boltzmann equation are that their mass, momentum and energy are preserved with time. This is also an *a priori* property of the Boltzmann equation on the torus (see [112] Chapter 1 Section 2 for instance) which reads

$$\forall t \ge 0, \quad \int_{\mathbb{T}^d \times \mathbb{R}^d} \begin{pmatrix} 1 \\ v \\ |v|^2 \end{pmatrix} f_{\varepsilon}(t, x, v) \, dx dv = \int_{\mathbb{T}^d \times \mathbb{R}^d} \begin{pmatrix} 1 \\ v \\ |v|^2 \end{pmatrix} f_{\varepsilon}(0, x, v) \, dx dv.$$

If one expects trend to the equilibrium  $\mu(v)$  for the solutions  $f_{\varepsilon} = \mu + \varepsilon h_{\varepsilon}$  of the Boltzmann equation (5.1.4) then it must be that

$$\forall t \ge 0, \quad \int_{\mathbb{T}^d \times \mathbb{R}^d} \begin{pmatrix} 1 \\ v \\ |v|^2 \end{pmatrix} h_{\varepsilon}(t, x, v) \, dx dv = 0,$$

that is  $\Pi_{\mathcal{G}_{\varepsilon}}(h_{\varepsilon}(t,\cdot,\cdot)) = 0$  for all t, which is a property that is indeed preserved along time for solution to the perturbed Boltzmann equation (5.1.5), see [23] or Chapter 4 for instance. We hence state the following theorem answering the Cauchy problem and the exponential convergence towards the equilibrium  $\mu$ .

**Theorem 5.2.3** Let B be a Boltzmann collision kernel satisfying (5.1.1)-(5.1.2)-(5.1.3)and let p = 1 or p = 2. There exists  $0 < \varepsilon_d \leq 1$  and  $\beta_0$  in  $\mathbb{N}$  such that

• for all  $\alpha$ ,  $\beta$  in  $\mathbb{N}$  such that  $\beta \ge \beta_0$  and  $\alpha \le \beta$  and for all k > 2 define

$$\mathcal{E}^p = W_v^{\alpha,1} W_x^{\beta,p} \left( \langle v \rangle^k \right),$$

• for any  $\lambda'_0$  in  $(0, \lambda_0)$  ( $\lambda_0$  defined in Theorem 5.2.1) there exist  $C_{\alpha,\beta}$ ,  $\eta_{\alpha,\beta} > 0$  such that for any  $0 < \varepsilon \leq \varepsilon_d$ , for any distribution  $0 \leq f_{in} = \mu + \varepsilon h_{in}$ :

If

- (i)  $h_{in}$  is in  $\operatorname{Ker}(\mathcal{G}_{\varepsilon})^{\perp}$  in  $\mathcal{E}^p$ ,
- (*ii*)  $||h_{in}||_{\mathcal{E}^p} \leq \eta_{\alpha,\beta}$ ,

then there exists a unique global solution  $f_{\varepsilon} = f_{\varepsilon}(t, x, v)$  to (5.1.4) in  $\mathcal{E}^p$  which, moreover, satisfies  $f_{\varepsilon} = M + \varepsilon h_{\varepsilon} \ge 0$  with:

- $h_{\varepsilon}$  belongs to  $\operatorname{Ker}(\mathcal{G}_{\varepsilon})^{\perp}$  for all times,
- •

$$\|h_{\varepsilon}\|_{\mathcal{E}^p} \leqslant C_{\alpha,\beta} \,\|h_{in}\|_{\mathcal{E}^p} \, e^{-\lambda'_0 t}.$$

The constants  $C_{\alpha,\beta}$  and  $\eta_{\alpha,\beta}$  are constructive and depends only on  $\alpha$ ,  $\beta$ , k, d,  $\lambda'_0$  and the kernel of the Boltzmann operator.

# 5.3 The linear part: a $C^0$ -semigroup in spaces with polynomial weight, proof of Theorem 5.2.1

In this section we focus on the linear part of the perturbed Boltzmann equation in  $W_v^{\alpha,q} W_x^{\beta,p} (\langle v \rangle^k)$ . We thus consider the following equation:

$$\partial_t h = \mathcal{G}_{\varepsilon}(h). \tag{5.3.1}$$

## 5.3.1 Strategy of the proof

If we denote  $\mathcal{E} = W_v^{\alpha,q} W_x^{\beta,p} (\langle v \rangle^k)$  and  $E = H_{x,v}^s (\mu^{-1/2})$  we have that  $E \subset \mathcal{E}$ , dense with continuous embedding for *s* large enough. [23] Theorem 2.1 (with the norm of Theorem 2.4) states that  $G_{\varepsilon} = (\mathcal{G}_{\varepsilon})|_E$  generates a strongly continuous semigroup in *E* with exponential decay (these results are given in Chapter 4 Section 4.2). Theorem 5.2.1 can therefore be understood as the possibility to extend properties of  $G_{\varepsilon}$  in a small space *E* to  $\mathcal{G}_{\varepsilon}$  in a larger space  $\mathcal{E}$ .

This issue of extending spectral gap properties as well as semigroup properties has been first tackled by Mouhot to obtain constructive rates of convergence to equilibrium for the homogeneous Boltzmann equation [80]. Recently, Gualdani, Mischler and Mouhot [51] proposed a more abstract approach that allows to deal with the full linear operator. In their work, they proved that if some conditions on  $\mathcal{G}_{\varepsilon}$  and  $\mathcal{G}_{\varepsilon}$  are satisfied then we can pass on some semigroup properties from E to  $\mathcal{E}$ . The main argument of the proof of Theorem 5.2.1 is to show that we can use their result in our setting, independently of  $\varepsilon$ .

To be more precise, we give below a modified version of their main functional analysis theorem which is combination of Theorem 2.13 and Lemma 2.17 where we added dependencies on  $\varepsilon$ .

We refer to [51] Section 2 for the definition of hypodissipativity (roughly speaking it is a dissipative property in a different norm on a Banach space) and the definition of the convolution of two semigroups of operators (denoted by the symbol (\*)). In the sequel we will use  $\mathscr{C}(E)$  for the set of closed operators on E and  $\mathscr{B}(E)$  for the set of bounded operators on E. For any operator G in  $\mathscr{C}(E)$  we denote  $\mathbb{R}(G)$  its range and  $\Sigma(G)$  its spectrum.

**Theorem 5.3.1 (Modified extension theorem from [51])** Let  $\varepsilon$  be a parameter such that  $0 < \varepsilon \leq 1$ .

Let  $E, \mathcal{E}$  be two Banach spaces with  $E \subset \mathcal{E}$  dense with continuous embedding, and consider  $G_{\varepsilon}$  in  $\mathscr{C}(E), \mathcal{G}_{\varepsilon}$  in  $\mathscr{C}(\mathcal{E})$  with  $(\mathcal{G}_{\varepsilon})|_{E} = G_{\varepsilon}$  and a > 0.

We assume the following

(A1)  $G_{\varepsilon}$  generates a semigroup  $S_{G_{\varepsilon}}$  on E,  $G_{\varepsilon} + a$  is hypodissipative on  $R(\mathrm{Id} - \Pi_{G_{\varepsilon},a})$  and

$$\Sigma(G_{\varepsilon}) \cap \{z \in \mathbb{C}, \operatorname{Re}(z) > -a\} = \{0\}$$
 is a discrete eigenvalue.

(A2) There exists  $\mathcal{A}_{\varepsilon}, \mathcal{B}_{\varepsilon}$  in  $\mathscr{C}(\mathcal{E})$  such that  $\mathcal{G}_{\varepsilon} = \mathcal{A}_{\varepsilon} + \mathcal{B}_{\varepsilon}$  (with corresponding restrictions  $A_{\varepsilon}, B_{\varepsilon}$  on E) and there exist some "intermediate spaces" (not necessarily ordered)

$$E = \mathcal{E}_J, \, \mathcal{E}_{J-1}, \ldots, \, \mathcal{E}_2, \, \mathcal{E}_1 = \mathcal{E}_J$$

such that, still denoting  $\mathcal{B}_{\varepsilon} := (\mathcal{B}_{\varepsilon})|_{\mathcal{E}_i}$  and  $\mathcal{A}_{\varepsilon} := (\mathcal{A}_{\varepsilon})|_{\mathcal{E}_i}$ 

- (i)  $(\mathcal{B}_{\varepsilon} + a/\varepsilon^2)$  is hypodissipative on  $\mathcal{E}_j$ ;
- (ii)  $\mathcal{A}_{\varepsilon} \in \mathscr{B}(\mathcal{E}_{j})$  with  $\|\mathcal{A}_{\varepsilon}\|_{\mathscr{B}(\mathcal{E}_{i})} \leq C_{A}/\varepsilon^{2};$

(iii) there are some constants  $l_0, l_1 \in \mathbb{N}^*$ ,  $C \ge 1$ ,  $K \in \mathbb{R}$ ,  $\alpha \in [0, 1)$  such that

$$\forall t \ge 0, \quad \|T_{l_0}(t)\|_{\mathscr{B}(\mathcal{E}_j, \mathcal{E}_{j+1})} \le C \frac{e^{Kt/\varepsilon^2}}{\varepsilon^{l_1} t^{\alpha}},$$

for  $1 \leq j \leq J-1$ , with the notation  $T_l := (\mathcal{A}_{\varepsilon}S_{\mathcal{B}_{\varepsilon}})^{(*l)}$ .

Then  $\mathcal{G}_{\varepsilon}$  is hypodissipative in  $\mathcal{E}$  and for all a' < a there exists  $n = n(a') \ge 1$  and some positive constants  $C_{a'}$  and  $C'_{a'}$  such that

$$||T_n(t)||_{\mathscr{B}(\mathcal{E})} \leqslant \frac{C_{a'}}{\varepsilon^{nl_1/l_0}} e^{-a't/\varepsilon^2};$$
(5.3.2)

$$S_{\mathcal{G}_{\varepsilon}}(t) = S_{G_{\varepsilon}}(t)\Pi_{\mathcal{G}} + \sum_{l=0}^{n-1} (-1)^{l} \left( \mathrm{Id} - \Pi_{\mathcal{G}} \right) S_{\mathcal{B}_{\varepsilon}} * T_{l}(t) + (-1)^{n} \left[ (\mathrm{Id} - \Pi_{\mathcal{G}}) S_{G_{\varepsilon}} \right] * T_{n}(t); \quad (5.3.3)$$

$$\left\| \left| S_{\mathcal{G}_{\varepsilon}}(t) - S_{G_{\varepsilon}}(t) \Pi_{\mathcal{G}} - (-1)^{n} \left[ \left( \operatorname{Id} - \Pi_{\mathcal{G}} \right) S_{G_{\varepsilon}} \right] * T_{n}(t) \right\|_{\mathscr{B}(\mathcal{E})} \leqslant C_{a'}^{\prime} \frac{t^{n}}{\varepsilon^{n(2+l_{1}/l_{0})}} e^{-a't/\varepsilon^{2}}, \quad (5.3.4)$$

where  $\Pi_{\mathcal{G}}$  has been defined in (5.2.4).

We will use Theorem 5.3.1 to directly prove Theorem 5.2.1. Indeed, [23] Theorem 2.1 states that  $G_{\varepsilon}$  generates a strongly continuous semigroup with exponential decay in  $E = H_{x,v}^s (\mu^{-1/2})$ , which is the required assumption (A1) (properties about the spectral gap of the spectrum can be found in [4]). Therefore if  $\mathcal{G}_{\varepsilon}$  fulfils hypothesis (A2) then it generates a strongly continuous semigroup, with an exponential decay of order a' for all a' < a, since for all  $\alpha$ ,  $\beta$ ,  $\eta > 0$ , all  $t > t_0 > 0$  and all  $0 < 2\eta' < \eta$ ,

$$\frac{t^{\alpha}}{\varepsilon^{2\beta}}e^{-\eta\frac{t}{\varepsilon^{2}}} \leqslant C_{\beta,\eta'}t^{\alpha-\beta}e^{-(\eta-\eta')\frac{t}{2\varepsilon^{2}}} \leqslant C_{\beta,\eta'}t^{\alpha-\beta}e^{-(\eta-\eta')t} \leqslant C_{t_{0},\alpha,\beta,\eta'}e^{-(\eta-2\eta')t}, \quad (5.3.5)$$

for  $0 < \varepsilon \leq 1$ .

#### **5.3.2** Decomposition of the operator and assumption (A2)(ii)

In this section we find a decomposition  $\mathcal{G}_{\varepsilon} = \mathcal{A}_{\varepsilon} + \mathcal{B}_{\varepsilon}$  that will fit the requirements (A1) - (A2) of Theorem 5.3.1. This decomposition has been found in [51] in the case  $\varepsilon = 1$ . We will use exactly the same operators but including the dependencies in  $\varepsilon$ . All the results presented in the rest of this section are true for  $\varepsilon = 1$  (see [51] Section 4) so we will try to relate as much as possible our computations with the ones for  $\varepsilon = 1$ .

For  $\delta$  in (0, 1), to be chosen later, we consider  $\Theta_{\delta} = \Theta_{\delta}(v, v_*, \sigma)$  in  $C^{\infty}$  that is bounded

by one on the set

$$\{|v| \leq \delta^{-1} \text{ and } 2\delta \leq |v - v_*| \leq \delta^{-1} \text{ and } |\cos \theta| \leq 1 - 2\delta\}$$

and whose support is included in

$$\{|v| \leq 2\delta^{-1} \text{ and } \delta \leq |v - v_*| \leq 2\delta^{-1} \text{ and } |\cos \theta| \leq 1 - \delta\}.$$

We define the splitting

$$\mathcal{G}_{\varepsilon} = \mathcal{A}_{\varepsilon}^{(\delta)} + \mathcal{B}_{\varepsilon}^{(\delta)},$$

with

$$\mathcal{A}_{\varepsilon}^{(\delta)}h(v) = \frac{1}{\varepsilon^2} \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} \Theta_{\delta} \left[ \mu'_* h' + \mu' h'_* - \mu h_* \right] b\left(\cos\theta\right) |v - v_*|^{\gamma} \, d\sigma dv_*$$

and

$$\mathcal{B}_{\varepsilon}^{(\delta)}h(v) = \mathcal{B}_{2,\varepsilon}^{(\delta)}h(v) - \frac{1}{\varepsilon^2}\nu(v)h(v) - \frac{1}{\varepsilon}v\cdot\nabla_x h(v),$$

where

$$\mathcal{B}_{2,\varepsilon}^{(\delta)}h(v) = \frac{1}{\varepsilon^2} \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} \left(1 - \Theta_\delta\right) \left[\mu'_* h' + \mu' h'_* - \mu h_*\right] b\left(\cos\theta\right) |v - v_*|^{\gamma} \, d\sigma dv_*$$

and  $\nu(v)$  is the standard collision frequency

$$\nu(v) = \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} b\left(\cos\theta\right) |v - v_*|^{\gamma} \mu_* \, d\sigma dv_*.$$

Note that there exists  $\nu_0$ ,  $\nu_1 > 0$  such that

$$\forall v \in \mathbb{R}^d, \quad \nu_0(1+|v|^{\gamma}) \leq \nu(v) \leq \nu_1(1+|v|^{\gamma}).$$
 (5.3.6)

We have that

$$\mathcal{A}_{\varepsilon}^{(\delta)} = \frac{1}{\varepsilon^2} \mathcal{A}_1^{(\delta)} \quad \text{and} \quad \mathcal{B}_{2,\varepsilon}^{(\delta)} = \frac{1}{\varepsilon^2} \mathcal{B}_{2,1}^{(\delta)}.$$

We therefore obtain the following controls on  $\mathcal{A}_{\varepsilon}^{(\delta)}$ .

**Proposition 5.3.2** For all  $0 < \varepsilon < \varepsilon^d$ , for any q in  $[1, +\infty]$  and  $\alpha \ge 0$ , the operator  $\mathcal{A}_{\varepsilon}^{(\delta)}$ maps  $L_v^q$  into  $W_v^{\alpha,q}$  with compact support. There exists  $C_{\delta,\alpha,q}, R_{\delta} > 0$  independent of  $\varepsilon$  such that

$$\forall h \in L_v^q, \, \operatorname{supp}\left(\mathcal{A}_{\varepsilon}^{(\delta)}h\right) \subset B(0, R_{\delta}), \quad \left\|\mathcal{A}_{\varepsilon}^{(\delta)}h\right\|_{W_v^{\alpha, q}} \leqslant \frac{C_{\delta, \alpha, q}}{\varepsilon^2} \|h\|_{L_v^q}.$$

Moreover, for any p in  $[1, +\infty]$  and for all h in  $L_v^q L_x^p$ ,

$$\left\|\mathcal{A}_{\varepsilon}^{(\delta)}h\right\|_{L_{v}^{q}L_{x}^{p}} \leqslant \left\|\mathcal{A}_{\varepsilon}^{(\delta)}\left(\|h\|_{L_{x}^{p}}\right)\right\|_{L_{v}^{q}}$$

**Remark 5.3.3** We notice here that this Proposition gives the point (A2)(ii) of Theorem 5.3.1 if the  $\mathcal{E}_j$  are Sobolev spaces.

**Proof of Proposition 5.3.2** The kernel of the operator  $\mathcal{A}_{\varepsilon}^{(\delta)}$  is of compact support so its Carleman representation (see [27]) gives the existence of  $k^{(\delta)}$  in  $C_c^{\infty} (\mathbb{R}^d \times \mathbb{R}^d)$  such that

$$\mathcal{A}_{\varepsilon}^{(\delta)}h(v) = \frac{1}{\varepsilon^2} \int_{\mathbb{R}^d} k^{(\delta)}(v, v_*)h(v_*) \, dv_*, \qquad (5.3.7)$$

and therefore the control on  $\left\|\mathcal{A}_{\varepsilon}^{(\delta)}h\right\|_{W_{v}^{\alpha,q}}$  is straightforward.

The control of  $\left\|\mathcal{A}_{\varepsilon}^{(\delta)}h\right\|_{L^{q}_{v}L^{p}_{x}}$  comes directly from Minkowski's integral inequality which states

$$\left[\int_{\mathbb{T}^d} \left(\int_{\mathbb{R}^d} k^{(\delta)}(v,v_*)h(x,v_*)dv_*\right)^p dx\right]^{1/p} \leqslant \int_{\mathbb{R}^d} \left(\int_{\mathbb{T}^d} k^{(\delta)}(v,v_*)^p h(x,v_*)^p dx\right)^{1/p} dv_*.$$

## 5.3.3 Dissipativity estimates for $\mathcal{B}_{\varepsilon}^{(\delta)}$ , assumption (A2)(i)

One can find in [51] proof of Lemma 4.14 case (W2) and (W3) the following estimate on the operator  $\mathcal{B}_{\varepsilon}^{(\delta)}$  in the case  $\varepsilon = 1$ .

**Lemma 5.3.4** For all p, q in  $[1, +\infty]$ , for all k > 2 and for any  $\delta$  in (0, 1) and all h in  $L^q_v L^p_x(\langle v \rangle^k)$ ,

$$\int_{\mathbb{R}^d} \langle v \rangle^{kq} \|h\|_{L^p_x}^{q-p} \left( \int_{\mathbb{T}^d} \operatorname{sgn}(h) |h|^{p-1} \mathcal{B}_1^{(\delta)} h \, dx \right) \, dv \leq \left[ \Lambda_{k-\gamma/q',q}(\delta) - 1 \right] \|h\|_{L^q_v L^p_x(\langle v \rangle^k \nu^{1/q})}^q,$$

where q' is the conjugate exponent of 1/q and  $\Lambda_{k,q}(\delta)$  is a constructive constant such that

$$\lim_{\delta \to 0} \Lambda_{k,q}(\delta) = \phi_q(k) = \left(\frac{4}{k+2}\right)^{1/q} \left(\frac{4}{k-1}\right)^{1-1/q}.$$

**Remark 5.3.5** As noticed in [51] Remark 4.3, the quantity  $\phi_q(k)$  is strictly less than one for k bigger than a constant  $k_q^{**}$ . The constant  $k_q^*$  we are considering is not optimal and is such that  $\phi_q(k - \gamma/q') < 1$ , where q' is the conjugate exponent of q. This appearance of  $k - \gamma/q'$  is due to a loss of weight of order  $\nu^{-1/q'}$  in the estimate of the spectral gap, see proof of Proposition 5.3.6.

In the case of the Boltzmann operator with hard potential and angular cutoff, point (A2)(i) is fulfilled by  $\mathcal{B}_{\varepsilon}^{(\delta)}$  for  $\delta$  small enough. This is the purpose of the following lemma. We recall here that  $\nu_0 = \inf_{v \in \mathbb{R}^d} (\nu(v)) > 0$  and that we define

$$\left\|\cdot\right\|_{W_{v}^{\alpha,q}W_{x}^{\beta,p}\left(\langle v\rangle^{k}\right)}=\sum_{\substack{|l|+|j|\leqslant \max\left(\alpha,\beta\right)\\|j|\leqslant\alpha,|l|\leqslant\beta}}\left\|\partial_{l}^{j}\cdot\right\|_{L_{v}^{q}L_{x}^{p}\left(\langle v\rangle^{k}\right)}.$$

**Proposition 5.3.6** Consider p, q in  $[1, +\infty]$ ,  $k > k_q^*$ , defined by (5.2.3), and  $\alpha$ ,  $\beta$  in  $\mathbb{N}$  such that  $\alpha \leq \beta$ .

Then there exists  $\delta_{k,q}$  in (0,1) such that for all  $0 < \delta \leq \delta_{k,q}$  there exists  $\lambda_0 = \lambda_0(k,q,\delta)$  in  $(0,\nu_0)$  such that for all  $0 < \varepsilon \leq 1$ ,

- $\lambda_0(k,q,\delta)$  tends to  $\lambda_0^*(k,q)$  as  $\delta$  goes to 0,
- $\lambda_0^*(k,q)$  tends to  $\nu_0$  when k goes to  $+\infty$ ,
- $\left(\mathcal{B}_{\varepsilon}^{(\delta)} + \lambda_0/\varepsilon^2\right)$  is dissipative in  $W_v^{\alpha,q} W_x^{\beta,p}\left(\langle v \rangle^k\right)$ .

**Proof of Proposition** 5.3.6 Let  $h_0$  be in  $W_v^{\alpha,q} W_x^{\beta,p}(\langle v \rangle^k)$  and considert h to be a solution to the linear equation

$$\partial_t h = \mathcal{B}_{\varepsilon}^{(\delta)} h = \mathcal{B}_{2,\varepsilon}^{(\delta)} h - \frac{1}{\varepsilon^2} \nu h - \frac{1}{\varepsilon} v \cdot \nabla_x h, \qquad (5.3.8)$$

with initial value  $h_0$ .

Since the x-derivative commutes with the equation we can consider only the case when  $\beta = \alpha$ . The proof is split into two parts. First we prove Proposition 5.3.6 in the case  $\alpha = 0$  and then we study the case with v-derivatives.

Step 1: the case  $\alpha = 0$ . Take p, q in  $[1, +\infty)$ . We recall that

$$\|h\|_{L^q_v L^p_x(\langle v \rangle^k)} = \left[ \int_{\mathbb{R}^d} \left( 1 + |v|^k \right)^q \left( \int_{\mathbb{T}^d} |h|^p \ dx \right)^{q/p} \ dv \right]^{1/q}$$

Therefore we can compute

$$\frac{d}{dt} \|h\|_{L^q_v L^p_x(\langle v \rangle^k)} = \|h\|^{1-q}_{L^q_v L^p_x(\langle v \rangle^k)} \times \int_{\mathbb{R}^d} \left(1 + |v|^k\right)^q \|h\|^{q-p}_{L^p_x} \left(\int_{\mathbb{T}^d} \operatorname{sgn}(h) |h|^{p-1} \mathcal{B}^{(\delta)}_{\varepsilon} h \, dx\right) \, dv.$$
(5.3.9)

Observing that

$$\int_{\mathbb{T}^d} \operatorname{sgn}(h) |h|^{p-1} v \cdot \nabla_x h \, dx = \frac{1}{p} v \cdot \int_{\mathbb{T}^d} \nabla_x \left( |h|^p \right) \, dx = 0,$$

we deduce

$$\frac{d}{dt} \|h\|_{L^q_v L^p_x(\langle v \rangle^k)} = \|h\|_{L^q_v L^p_x(\langle v \rangle^k)}^{1-q} \times \frac{1}{\varepsilon^2} \int_{\mathbb{R}^d} \left(1 + |v|^k\right)^q \|h\|_{L^p_x}^{q-p} \left(\int_{\mathbb{T}^d} \operatorname{sgn}(h) |h|^{p-1} \mathcal{B}_1^{(\delta)} h \, dx\right) \, dv.$$

We can therefore use Lemma 5.3.4 which leads to

$$\frac{d}{dt} \left\|h\right\|_{L^q_v L^p_x(\langle v \rangle^k)} \leqslant -\frac{1}{\varepsilon^2} \left[1 - \Lambda_{k-\gamma/q',q}(\delta)\right] \left\|h\right\|^q_{L^q_v L^p_x(\langle v \rangle^k \nu^{1/q})} \left\|h\right\|^{1-q}_{L^q_v L^p_x(\langle v \rangle^k)}, \quad (5.3.10)$$

We already noticed that  $\Lambda_{k-1/q',q}(\delta)$  is strictly less than 1 for  $\delta$  smaller than some  $\delta_{k,q}$  (see Remark 5.3.5). Therefore, because  $\nu(v) \ge \nu_0$  for all v we have that for all  $\delta$  smaller than  $\delta_{k,q}$  the following holds,

$$\frac{d}{dt} \|h\|_{L^q_v L^p_x(\langle v \rangle^k)} \leqslant -\frac{\nu_0}{\varepsilon^2} \left[1 - \Lambda_{k-\gamma/q',q}(\delta)\right] \|h\|_{L^q_v L^p_x(\langle v \rangle^k)},$$

This concludes the proof of Proposition 5.3.6 for  $\alpha = 0$  and  $1 \leq p, q < +\infty$ . The cases  $p = \infty$  and  $q = \infty$  are respectively dealt with by taking the limit  $p \to \infty$  and  $q \to \infty$  which is possible since  $\delta_{k,q}$  is independent of p and can be chosen to converge to a strictly positive constant when q goes to  $\infty$ , thanks to the definition of  $\Lambda_{k,q}(\delta)$ .

## Step 2: the case with v-derivatives. Take p, q in $[1, +\infty]$ and $\alpha = \beta = 1$ .

Since the x-derivative commutes with (5.3.8) the equation satisfied by h, we have that (5.3.10) holds for x-derivatives. Notice that  $1 - q \leq 0$  gives

$$\frac{d}{dt} \left( \|h\|_{L_v^q L_x^p(\langle v \rangle^k)} + \|\nabla_x h\|_{L_v^q L_x^p(\langle v \rangle^k)} \right) 
\leq -\frac{\nu_0^{1-1/q}}{\varepsilon^2} \left[ 1 - \Lambda_{k-\gamma/q',q}(\delta) \right] \left( \|h\|_{L_v^q L_x^p(\langle v \rangle^k \nu^{1/q})} + \|\nabla_x h\|_{L_v^q L_x^p(\langle v \rangle^k \nu^{1/q})} \right).$$
(5.3.11)

Applying a v-derivatives to (5.3.8) yields

$$\partial_t \nabla_v h = \mathcal{B}^{(\delta)}_{\varepsilon}(\nabla_v h) + \left(\nabla_v B^{(\delta)}_{\varepsilon}\right)(h)$$
  
=  $\mathcal{B}^{(\delta)}_{\varepsilon}(\nabla_v h) - \frac{1}{\varepsilon}\nabla_x h + \mathcal{R}^{(\delta)}_{\varepsilon}(h),$ 

where  $\mathcal{R}_{\varepsilon}^{(\delta)}(h) = \left(\nabla_{v}\mathcal{B}_{2,\varepsilon}^{(\delta)}\right)(h) - \frac{1}{\varepsilon^{2}}\nabla_{v}(\nu)h = \frac{1}{\varepsilon^{2}}\mathcal{R}_{1}^{(\delta)}(h).$ From (5.3.11), our computations in Step 1 with  $\delta \leq \delta_{k,q}$  and the following norm

$$\|h\|_{W_{v}^{1,q}W_{x}^{1,p}(\langle v \rangle^{k})_{\eta}} = \|h\|_{L_{v}^{q}L_{x}^{p}(\langle v \rangle^{k})} + \|\nabla_{x}h\|_{L_{v}^{q}L_{x}^{p}(\langle v \rangle^{k})} + \eta \|\nabla_{v}h\|_{L_{v}^{q}L_{x}^{p}(\langle v \rangle^{k})},$$

with  $\eta > 0$  to be fixed later, we obtain

$$\begin{aligned} \frac{d}{dt} \|h\|_{W_{v}^{1,q}W_{x}^{1,p}\left(\langle v\rangle^{k}\right)_{\eta}} \\ &\leqslant -\frac{\nu_{0}^{1-1/q}}{\varepsilon^{2}} \left[1 - \Lambda_{k-\gamma/q',q}(\delta)\right] \left(\|h\|_{L_{v}^{q}L_{x}^{p}\left(\langle v\rangle^{k}\nu^{1/q}\right)} + \|\nabla_{x}h\|_{L_{v}^{q}L_{x}^{p}\left(\langle v\rangle^{k}\nu^{1/q}\right)}\right) \\ &- \eta \frac{\nu_{0}^{1-1/q}}{\varepsilon^{2}} \left[1 - \Lambda_{k-\gamma/q',q}(\delta)\right] \|\nabla_{v}h\|_{L_{v}^{q}L_{x}^{p}\left(\langle v\rangle^{k}\nu^{1/q}\right)} \\ &- \frac{\eta}{\varepsilon} \|\nabla_{v}h\|_{L_{v}^{q}L_{x}^{p}\left(\langle v\rangle^{k}\right)}^{1-q} \int_{\mathbb{R}^{d}} \left(\langle v\rangle^{k}\right)^{q} \|\nabla_{v}h\|_{L_{x}^{p}}^{q-p} \left(\int_{\mathbb{T}^{d}} \operatorname{sgn}(h) |\nabla_{v}h|^{p-1} \nabla_{x}h \, dx\right) \, dv \\ &+ \frac{\eta}{\varepsilon^{2}} \|\nabla_{v}h\|_{L_{v}^{q}L_{x}^{p}\left(\langle v\rangle^{k}\right)}^{1-q} \int_{\mathbb{R}^{d}} \left(\langle v\rangle^{k}\right)^{q} \|\nabla_{v}h\|_{L_{x}^{p}}^{q-p} \left(\int_{\mathbb{T}^{d}} \operatorname{sgn}(h) |\nabla_{v}h|^{p-1} \mathcal{R}_{1}^{(\delta)}(h) \, dx\right) \, dv. \end{aligned}$$

We take the absolute value and use Hölder inequality twice on the last two terms which makes the terms in  $\nabla_v h$  disappear, and this gives

$$\begin{split} \frac{d}{dt} \|h\|_{W_v^{1,q}W_x^{1,p}(\langle v \rangle^k)_\eta} \\ &\leqslant -\frac{\nu_0^{1-1/q}}{\varepsilon^2} \left[1 - \Lambda_{k-\gamma/q',q}(\delta)\right] \left(\|h\|_{L_v^q L_x^p(\langle v \rangle^k \nu^{1/q})} + \eta \|\nabla_v h\|_{L_v^q L_x^p(\langle v \rangle^k \nu^{1/q})}\right) \\ &\quad + \frac{1}{\varepsilon^2} \left(\varepsilon \eta \nu_0^{-1/q} - \nu_0^{1-1/q} \left[1 - \Lambda_{k-\gamma/q',q}(\delta)\right]\right) \|\nabla_x h\|_{L_v^q L_x^p(\langle v \rangle^k \nu^{1/q})} \\ &\quad + \frac{\eta}{\varepsilon^2} \left\|\mathcal{R}_1^{(\delta)}(h)\right\|_{L_v^q L_x^p(\langle v \rangle^k)}. \end{split}$$

One can find in [51] proof of Lemma 4.14 case (W2) and (W3) the following estimate

$$\left\| \mathcal{R}_{1}^{(\delta)}(h) \right\|_{L_{v}^{q} L_{x}^{p}\left(\langle v \rangle^{k}\right)} \leq C_{\delta} \left\| h \right\|_{L_{v}^{q} L_{x}^{p}\left(\langle v \rangle^{k} \nu^{1/q}\right)},$$

where  $C_{\delta} > 0$  is a constant only depending on  $\delta$ .

Because  $\varepsilon \leq 1$ , this latter estimates yields

$$\frac{d}{dt} \|h\|_{W_{v}^{1,q}W_{x}^{1,p}(\langle v \rangle^{k})_{\eta}} \leqslant \frac{1}{\varepsilon^{2}} \left( C_{\delta}\eta - \nu_{0}^{1-1/q} \left[ 1 - \Lambda_{k-\gamma/q',q}(\delta) \right] \right) \|h\|_{L_{v}^{q}L_{x}^{p}(\langle v \rangle^{k}\nu^{1/q})} 
+ \frac{1}{\varepsilon^{2}} \left( \eta\nu_{0}^{-1/q} - \nu_{0}^{1-1/q} \left[ 1 - \Lambda_{k-\gamma/q',q}(\delta) \right] \right) \|\nabla_{x}h\|_{L_{v}^{q}L_{x}^{p}(\langle v \rangle^{k}\nu^{1/q})} 
- \eta \frac{\nu_{0}^{1-1/q}}{\varepsilon^{2}} \left[ 1 - \Lambda_{k-\gamma/q',q}(\delta) \right] \|\nabla_{v}h\|_{L_{v}^{q}L_{x}^{p}(\langle v \rangle^{k}\nu^{1/q})},$$
(5.3.12)

which concludes the proof if we take  $\eta$  small enough in terms of  $\delta$ , for  $\delta \leq \delta_{k,q}$ .

The case where  $1 < \alpha = \beta$  is dealt with in the same way with the norm

$$\|h\|_{W^{\alpha,q}_{v}W^{\alpha,p}_{x}\left(\langle v\rangle^{k}\right)_{\eta}} = \sum_{0 \leqslant |j|+|l| \leqslant \alpha} \eta^{|j|} \left\|\partial_{l}^{j}h\right\|_{L^{q}_{v}L^{p}_{x}\left(\langle v\rangle^{k}\right)}.$$

with  $\eta$  small enough in terms of  $\delta$ .

#### **5.3.4** Estimates on the iterated convolution product, assumption (A2)(*iii*)

In order to use Theorem 5.3.1, it remains to show that our equation (5.3.1) satisfies hypothesis (A2)(iii), that is we need to control the iterated quantities  $T_l := \left(\mathcal{A}_{\varepsilon}^{(\delta)} S_{\mathcal{B}_{\varepsilon}^{(\delta)}}\right)^{(*l)}$ for some l in  $\mathbb{N}$ . The following proposition describes such controls when p = 1.

**Proposition 5.3.7** Consider  $k > k_q^*$ , defined by (5.2.3), and s in  $\mathbb{N}$ . For any  $\delta$  in  $(0, \delta_{k,q}]$  and any  $\lambda'_0$  in  $(0, \lambda_0)$  ( $\delta_{k,q}$  and  $\lambda_0$  defined in Proposition 5.3.6), there exists  $C_1 = C_1(\lambda'_0, \delta) > 0$  and  $R = R(\delta) > 0$  such that for any  $t \ge 0$ ,

$$\forall n \in \mathbb{N}, \quad \text{supp } T_n(t)h \subset K := B(0, R)$$

and

$$\forall s \ge 1, \quad \|T_1(t)h\|_{W^{s+1,1}_{x,v}(K)} \le C_1 \frac{e^{-\frac{\lambda'_0}{\varepsilon^2}t}}{\varepsilon^2 t} \|h\|_{W^{s+1,1}_v W^s_x(\langle v \rangle^k)}, \qquad (5.3.13)$$

$$\forall s \ge 0, \quad \|T_2(t)h\|_{W^{s+1/2,1}_{x,v}(K)} \le C_1 \frac{e^{-\frac{1}{\varepsilon^2}t}}{\varepsilon^4} \|h\|_{W^{s,1}_{x,v}(\langle v \rangle^k)}.$$
(5.3.14)

## **Proof of Proposition 5.3.7**

Most of the proof is an adaptation of [51] proof of Lemma 4.19 to keep track of the dependencies on  $\varepsilon$ . We will refer to it when we are using some of its computations.

**Control of**  $T_1(t)h$ : The x-derivatives commutes with  $T_1(t)$  and therefore it is enough to consider h in  $W_v^{s,1}W_x^{1,1}(\langle v \rangle^k)$ , with  $s \ge 1$ , and to control  $||T_1(t)h||_{W_v^{s+1,1}W_x^{1,1}(K)}$ . This

5.3 The linear part: a  $C^0$ -semigroup in spaces with polynomial weight, proof of Theorem 5.2.1

gives

$$\|T_1(t)h\|_{W_v^{s+1,1}W_x^{1,1}(K)} \leq \|T_1(t)h\|_{W_v^{s+1,1}L_x^1(K)} + \|\nabla_x T_1(t)h\|_{W_v^{s+1,1}L_x^1(K)}.$$
 (5.3.15)

The first term is easily dealt with thanks to the estimate on  $\mathcal{A}_{\varepsilon}^{(\delta)}$ , Proposition 5.3.2, and the dissipativity property of  $\mathcal{B}_{\varepsilon}^{(\delta)}$ , Proposition 5.3.6,

$$\|T_{1}(t)h\|_{W_{v}^{s+1,1}L_{x}^{1}(K)} = \left\|\mathcal{A}_{\varepsilon}^{(\delta)}S_{\mathcal{B}_{\varepsilon}^{(\delta)}}h\right\|_{W_{v}^{s+1,1}L_{x}^{1}(K)} \leqslant \frac{C}{\varepsilon^{2}}e^{-\frac{\lambda_{0}}{\varepsilon^{2}}t}\|h\|_{L_{v}^{1}L_{x}^{1}(\langle v \rangle^{k})}.$$
 (5.3.16)

For the second term, define  $f(t) = S_{\mathcal{B}_{c}^{(\delta)}}h$  and

$$D_t = \varepsilon^{-1} t \nabla_x + \nabla_v. \tag{5.3.17}$$

By direct computations we have that

$$\varepsilon^{-1}t\nabla_x T_1(t)h = \mathcal{A}_{\varepsilon}^{(\delta)}(D_t f) - \left(\nabla_v \mathcal{A}_{\varepsilon}^{(\delta)}\right)f,$$

which leads to, by Proposition 5.3.2,

$$\varepsilon^{-1}t \|\nabla_x T_1(t)h\|_{W^{s+1,1}_v L^1_x(K)} \leqslant \frac{C}{\varepsilon^2} \left[ \|D_t f\|_{L^1_{x,v}(\langle v \rangle^k)} + \|f\|_{L^1_{x,v}(\langle v \rangle^k)} \right].$$
(5.3.18)

The dissipativity property of  $\mathcal{B}_{\varepsilon}^{(\delta)}$ , in particular (5.3.10) with q = 1, yields

$$\frac{d}{dt} \|f\|_{L^{1}_{x,v}\left(\langle v \rangle^{k}\right)} \leqslant -\frac{\lambda_{0}}{\varepsilon^{2}} \|f\|_{L^{1}_{x,v}\left(\langle v \rangle^{k}\nu\right)}.$$
(5.3.19)

Direct computations yields

$$\partial_t (D_t f) = \mathcal{B}_{\varepsilon}^{(\delta)} (D_t f) + \frac{1}{\varepsilon^2} \mathcal{J}^{(\delta)} f,$$

where

$$\mathcal{J}^{(\delta)} = \nabla_v \left( \mathcal{B}_1^{(\delta)}(\cdot) \right) - \mathcal{B}_1^{(\delta)} \left( \nabla_v(\cdot) \right)$$
(5.3.20)

is independent of  $\varepsilon$  and satisfies (see [51] proof of Lemma 4.19) for all g in  $L_v^1(\langle v \rangle^k \nu)$ 

$$\left\| \mathcal{J}^{(\delta)}g \right\|_{L^1_v(\langle v \rangle^k)} \leq C_{\delta} \|g\|_{L^1_v(\langle v \rangle^k \nu)}.$$

In the same way as proof of Proposition 5.3.6 we obtain

$$\frac{d}{dt} \left\| D_t f \right\|_{L^1_{x,v}\left(\langle v \rangle^k\right)} \leqslant -\frac{\lambda_0}{\varepsilon^2} \left\| D_t f \right\|_{L^1_{x,v}\left(\langle v \rangle^k \nu\right)} + \frac{C_\delta}{\varepsilon^2} \left\| f \right\|_{L^1_v\left(\langle v \rangle^k \nu\right)}.$$
(5.3.21)

We then consider  $\lambda'_0$  in  $(0, \lambda_0)$  and define  $\eta = (\lambda_0 - \lambda'_0)/C_{\delta}$ . We compute, with (5.3.19),

$$\frac{d}{dt}\left[e^{\frac{\lambda'_0}{\varepsilon^2}t}\left(\eta \left\|D_t f\right\|_{L^1_{x,v}\left(\langle v\rangle^k\right)} + \left\|f\right\|_{L^1_{x,v}\left(\langle v\rangle^k\right)}\right)\right] \leqslant 0,$$

and thus

$$\|D_t f\|_{L^1_{x,v}(\langle v \rangle^k)} + \|f\|_{L^1_{x,v}(\langle v \rangle^k)} \leqslant \eta^{-1} e^{-\frac{\lambda_0}{\varepsilon^2} t} \|h\|_{W^{1,1}_v L^1_x(\langle v \rangle^k)}.$$
(5.3.22)

To conclude we plug (5.3.22) into (5.3.18) and we combine it with (5.3.16) into (5.3.15). This yields, because  $s \ge 1$ ,

$$\|T_1(t)h\|_{W_v^{s+1,1}W_x^{1,1}(K)} \leqslant C \frac{e^{-\frac{\lambda'_0}{\varepsilon^2}t}}{\varepsilon^2 t} \|h\|_{W_v^{s,1}L_x^1(\langle v \rangle^k)},$$

which implies the expected result (5.3.13) because  $T_1(t)$  commutes with x-derivatives.

**Control of**  $T_2(t)h$ : For  $s \ge 0$  we can interpolate (for interpolation theory in Sobolev spaces see [13] Chapters 6) between (5.3.16) and (5.3.13) to get

$$\|T_1(t)h\|_{W^{s+1/2,1}_{x,v}(K)} \leq C \frac{e^{-\frac{\lambda'_0}{\varepsilon^2}t}}{\varepsilon^2 \sqrt{t}} \|h\|_{W^{s,1}_v L^1_x(\langle v \rangle^k)}.$$

Then, we firstly use the inequality above and secondly (5.3.16) to obtain

$$\begin{aligned} \|T_{2}(t)h\|_{W^{s+1/2,1}_{x,v}(K)} &\leqslant \int_{0}^{t} \|T_{1}(t-s)T_{1}(s)h\|_{W^{s+1/2,1}_{x,v}(K)} ds \\ &\leqslant \frac{C}{\varepsilon^{4}} e^{-\frac{\lambda'_{0}t}{\varepsilon^{2}}} \left( \int_{0}^{t} \frac{e^{-\frac{\lambda_{0}-\lambda'_{0}}{\varepsilon^{2}}s}}{\sqrt{t-s}} ds \right) \|h\|_{W^{s,1}_{v}L^{1}_{x}(\langle v \rangle^{k})} \,, \end{aligned}$$

which is the expected result (5.3.14).

The aim is to link our space  $L_v^q L_x^p (\langle v \rangle^k)$  to the space  $H_{x,v}^s (\mu^{-1/2})$ . We thus state the following control on the iterated convolution in the case p = 2.

**Proposition 5.3.8** Consider  $k > k_q^*$ , defined by (5.2.3), and s in  $\mathbb{N}$ .

For any  $\delta$  in  $(0, \delta_{k,q}]$  there exists  $C_2 = C_2(\delta) > 0$  and  $R = R(\delta) > 0$  such that for any  $t \ge 0$ ,

$$\forall n \in \mathbb{N}, \quad \text{supp } T_n(t)h \subset K := B(0, R)$$

and

$$\forall s \ge 0, \quad \|T_2(t)h\|_{H^{s+1/2}_{x,v}(K)} \le \frac{C_T}{\varepsilon^{5/2}} \|h\|_{H^s_{x,v}(\langle v \rangle^k)}.$$
(5.3.23)

## **Proof of Proposition 5.3.8** Consider h in $W^{s,2}_{x,v}(\langle v \rangle^k)$ , s in $\mathbb{N}$ .

This Proposition is easier than when p = 1 because there exists velocity averaging lemmas in this framework, as discussed in [51] Remark 4.21. The *x*-derivative commutes with  $T_1$  and therefore we suppose there is no derivative in space.

Define  $f(t) = S_{\mathcal{B}^{(\delta)}}(t)(h)$  so that f is solution to the kinetic equation

$$\partial_t f + \frac{1}{\varepsilon} v \cdot \nabla_x f = s_{\varepsilon}(t, x, v),$$

with  $s_{\varepsilon}(t, x, v) = -\varepsilon^{-2}\nu f + \varepsilon^{-2}\mathcal{B}_{2,\varepsilon}^{(\delta)}f.$ 

Let j be a multi-index such that  $|j| \leq s$ . We apply  $\partial_0^j$  to the latter equation, which gives

$$\partial_t \left( \partial_0^j f \right) + \frac{1}{\varepsilon} v \cdot \nabla_x \left( \partial_0^j f \right) = \partial_0^j s_\varepsilon(t, x, v) + \frac{1}{\varepsilon} \sum_{|i|+|l|=|j|} a_{i,l} \partial_l^i f, \qquad (5.3.24)$$

where  $a_{i,j}$  are non-negative numbers.

A classical averaging lemma (see [20] Lemma 1 and [21] in which we emphasize the dependencies in  $\varepsilon$ ) reads, for (5.3.24) with  $\partial_0^j f(0, x, v) = \partial_0^j h(x, v)$ , for all  $\psi$  in  $\mathcal{D}(\mathbb{R}^d)$ 

$$\left\| \int_{\mathbb{R}^d} \partial_0^j f(t, x, v) \psi(v) \, dv \right\|_{L^2_t \left( H^{1/2}_x \right)}$$

$$\leq \frac{C}{\sqrt{\varepsilon}} \left( \left\| \partial_0^j h(x, v) \right\|_{L^2_{x,v}} + \left\| \partial_0^j s_\varepsilon \right\|_{L^2_{t,x,v}} + \frac{1}{\varepsilon} \left\| \sum_{|i|+|l|=|j|} a_{i,l} \partial_l^i f \right\|_{L^2_{t,x,v}} \right).$$

$$(5.3.25)$$

We use [51], Lemmas 4.4 and 4.7, in order to bound the terms involving  $\mathcal{B}_{2,\varepsilon}^{(\delta)} = \varepsilon^{-2} \mathcal{B}_{2,1}^{(\delta)}$ we have that

$$\|s_{\varepsilon}\|_{H^{s}_{x,v}\left(\langle v\rangle^{k}\right)} \leqslant \frac{1}{\varepsilon^{2}} \|s_{1}\|_{H^{s}_{x,v}\left(\langle v\rangle^{k}\right)} \leqslant \frac{C}{\varepsilon^{2}} \|f\|_{H^{s}_{x,v}\left(\langle v\rangle^{k}\nu\right)} \leqslant \frac{C}{\varepsilon^{2}} e^{-\frac{\lambda_{0}}{\varepsilon^{2}}t} \|h\|_{H^{s}_{x,v}\left(\langle v\rangle^{k}\nu\right)},$$

where the last inequality comes from the hypodissipativity properties of  $S_{\mathcal{B}_{\varepsilon}}(t)$ , see Proposition 5.3.6.

Using the dissipativity properties of  $S_{\mathcal{B}_{\varepsilon}}(t)$  one more time we deduce that

$$\|T_{1}(t)h\|_{L^{2}_{t}\left(H^{s+1/2}_{x,v}(\langle v\rangle^{k})\right)} \leqslant \frac{C}{\varepsilon^{5/2}} \|h\|_{H^{s}_{x,v}\left(\langle v\rangle^{k}\nu\right)}.$$
(5.3.26)

To conclude we notice that  $\int_0^t T_1(t-s)T_1(s) ds$  is a continuous linear operator on the Hilbert space  $H_{x,v}^{s+1/2}(K)$  and thus we can see it as an element of  $H_{x,v}^{s+1/2}(K)$  by Riesz's representation theorem. Hence, thanks to Cauchy-Schwartz,

where we used Proposition 5.3.2 and the fact that  $S_{\mathcal{B}_{\varepsilon}^{(\delta)}}$  is a contraction semigroup on  $H_{x,v}^s$  with spectral gap  $\lambda'_0/\varepsilon^2$ .

## 5.3.5 Proof of Theorem 5.2.1

As we explained it in Section 5.3.1, the proof of Theorem 5.2.1 is direct from the application of Theorem 5.3.1. This theorem is clearly applicable in our case and we emphasize it through the extreme case of no derivative in space or velocity variables.

Indeed, we consider s in  $\mathbb{N}$  to be chosen big enough later. We define  $\mathcal{E} = L_v^q L_x^p (\langle v \rangle^k)$ and  $E = H_{x,v}^s (\mu^{-1/2})$  and we have  $E \subset \mathcal{E}$  for s big enough (dense with continuous embedding). Indeed, in the case  $q \ge 2$  and  $p \ge 2$ , standard Sobolev embeddings (see [22] Section IX.3.) imply  $E \subset L_v^q L_x^p (\mu^{-1/2})$ . In the case p < 2 we have, on the torus,  $L_x^2 \subset L_x^p$ and  $H_x^s \subset L_x^2$  by the same Sobolev embeddings. Finally, in the case q < 2 we have that  $L_v^2 (\mu^{-1/2}) \subset L_v^q (\langle v \rangle^k)$  (it can be done by a mere Cauchy-Schwarz inequality) and the same Sobolev embeddings give  $H_v^s (\mu^{-1/2}) \subset L_v^2 (\mu^{-1/2})$ .

On the torus we have the following embedding:  $L_x^p \subset L_x^1$ . Thanks to Proposition 5.3.2 and Proposition 5.3.6 we obtain (same arguments as (5.3.16))

$$\|T_1(t)h\|_{\mathcal{E}} \leq C \left\| A_{\varepsilon}^{(\delta)} S_{\mathcal{B}_{\varepsilon}^{(\delta)}} h \right\|_{L^q_v L^1_x(K)} \leq \frac{C}{\varepsilon^2} e^{-\frac{\lambda_0}{\varepsilon^2} t} \|h\|_{L^1_v L^1_x(\langle v \rangle^k)}.$$
(5.3.27)

We therefore define  $\mathcal{E}_2 = L_v^1 L_x^1 (\langle v \rangle^k)$ .

Then we define by  $\mathcal{E}_j = W_{x,v}^{(j-2)/2,1}(\langle v \rangle^k)$  for j from 2 to m with m big enough such that  $W_{x,v}^{(m-1)/2,1}(\langle v \rangle^k) \subset L^2_{x,v}(\langle v \rangle^k)$ . Then we denote  $\mathcal{E}_j = H_{x,v}^{(j-m-1)/2}(\langle v \rangle^k)$  up J-1 where  $H_{x,v}^{(J-m-2)/2}(\langle v \rangle^k) \subset E$ .

Point (A1) of Theorem 5.3.1 is satisfied thanks to [4] and [23] Theorem 2.1 (with the norm of Theorem 2.4), point (A2)(i) by Proposition 5.3.6 and point (A2)(ii) by Proposition 5.3.2. Finally, point (A2)(iii) is given by (5.3.27) for  $\mathcal{E}$  and  $\mathcal{E}_1$ , then by Proposition 5.3.7 (5.3.14) up to  $\mathcal{E}_m$  and by Proposition 5.3.8 from  $\mathcal{E}_m$  to  $\mathcal{E}_J$  and E.

## 5.4 An *a priori* estimate for the full perturbed equation: proof of Theorem 5.2.3

In this section we work in  $W_v^{\alpha,1} H_x^{\beta} (\langle v \rangle^k)$  or in  $W_v^{\alpha,1} W_x^{\beta,1} (\langle v \rangle^k)$ , with  $\alpha \leq \beta$  on the full perturbed Boltzmann equation

$$\partial_t h = \mathcal{G}_{\varepsilon}(h) + \frac{1}{\varepsilon}Q(h,h).$$

## 5.4.1 Description of the problem and notations

When  $\varepsilon = 1$ , the linear part  $\mathcal{G}_{\varepsilon}$  has the same order of magnitude than the bilinear term Q in the linearized Boltzmann equation (5.1.5). In this case, Theorem 5.2.1 suffices to obtain existence and exponential decay since the contraction property of the semigroup  $S_{\mathcal{G}_1}$  controls the bilinear part for small initial data (see [51]).

In the general case,  $S_{\mathcal{G}_{\varepsilon}}$  only generates a semigroup with a spectral gap of order 1, insufficient to control  $\varepsilon^{-1}Q$ . However, [56][23] (and Chapter 4) show that a careful study of  $\varepsilon^{-1}Q$  compared to  $G_{\varepsilon}$  yields existence and exponential decay of solutions to (5.1.5) in  $H_{x,v}^s\left(\mu^{-1/2}\right)$  for *s* large enough (see Theorem 5.4.7 for an adapted version of this result). Our strategy is to use the same kind of ideas as when we extended the semigroup properties from  $H_{x,v}^{\beta}\left(\mu^{-1/2}\right)$  to  $W_v^{\alpha,1}H_x^{\beta}\left(\langle v \rangle^k\right)$  and  $W_v^{\alpha,1}W_x^{\beta,1}\left(\langle v \rangle^k\right)$  but including the bilinear term. Namely, we shall decompose the partial differential equation (5.1.5) into a system of partial differential equations from  $W_v^{\alpha,1}H_x^{\beta}\left(\langle v \rangle^k\right)$  or  $W_v^{\alpha,1}W_x^{\beta,1}\left(\langle v \rangle^k\right)$  to  $H^{\beta}\left(\mu^{-1/2}\right)$  and use the perturbative estimates of [23] (that are given in Chapter 4).

As noticed in Remark 2.16 of [51], Theorem 5.3.1 extending the semigroup generated by  $\mathcal{G}_{\varepsilon}$  in  $H^s(\mu^{-1/2})$  to  $L_v^1 L_x^{\infty}(\langle v \rangle^k)$  can be interpreted as a decomposition of

$$\partial_t f = \mathcal{G}_{\varepsilon} f,$$

into a system of partial differential equations, involving operators  $\mathcal{G}_{\varepsilon} = \mathcal{A}_{\varepsilon} + \mathcal{B}_{\varepsilon}$  (defined in Section 5.3.2), with  $f = f^1 + \cdots + f^J$  satisfying

- $f^1$  is in  $L^1_v L^\infty_x(\langle v \rangle^k)$  and  $f^1_{in} = f_{in}$  in  $Ker(\mathcal{G}_\varepsilon)^{\perp}$ ,
- for all  $2 \leq j \leq J 1$ ,  $f^j$  is in  $\mathcal{E}_j$  and  $f^j_{in} = 0$ ,
- $f^J$  is in  $H^s(\mu^{-1/2})$ ,  $f^J_{in} = 0$  and in that space we can use the contraction property of  $S_{\mathcal{G}_{\varepsilon}}$ .

We will decompose the linearized Boltzmann equation in a similar way than the one explained above. We shall define a sequence of spaces  $(\mathcal{E}_j)_{1 \leq j \leq J}$ . In each space  $\mathcal{E}_j$ ,  $1 \leq j \leq J-1$ , a piece of the bilinear term, of order  $\varepsilon^{-1}$ , will be added and controlled by the dissipativity property of  $\mathcal{B}_{\varepsilon}^{(\delta)}$ , of order  $\varepsilon^{-2}$ . Contrary to the study in the linear case, the bilinear operator generates terms involving functions in all the spaces  $\mathcal{E}_j$  which have to be compared and controlled. This imposes to construct  $(\mathcal{E}_j)_{1 \leq j \leq J}$  as a nested sequence.

The difficult part of the linear operator, namely  $\mathcal{A}_{\varepsilon}^{(\delta)}$ , enjoys a regularising effect and could therefore be treated in more regular spaces. Of course, our decomposition will be much easier since we solely want to go from an exponential weight into a polynomial weight Sobolev spaces, without losing any derivatives in x or v.

In order to shorten notations we define, for p = 1, 2 and k to be defined later,

$$\mathcal{E}^{p} = W_{v}^{\alpha,1} W_{x}^{\beta,p} \left( \langle v \rangle^{k} \right) \quad \text{and} \quad E = H_{x,v}^{\beta} \left( \mu^{-1/2} \right).$$
(5.4.1)

We take  $h_{in}$  in  $\mathcal{E}^p$  and we decompose the partial differential equation,

$$\partial_t h = \mathcal{G}_{\varepsilon}(h) + \frac{1}{\varepsilon}Q(h,h) = \mathcal{A}_{\varepsilon}^{(\delta)}(h) + \mathcal{B}_{\varepsilon}^{(\delta)}(h) + \frac{1}{\varepsilon}Q(h,h)$$

into an equivalent system of partial differential equations for the following decomposition

$$h(t, x, v) = h^{0}(t, x, v) + h^{1}(t, x, v),$$
(5.4.2)

with

1. In  $\mathcal{E}^p$ ,  $h_{t=0}^0 = h_{in}$  and

$$\partial_t h^0 = \mathcal{B}^{(\delta)}_{\varepsilon}(h^0) + \frac{1}{\varepsilon}Q(h^0, h^0) + \frac{2}{\varepsilon}Q(h^0, h^1), \qquad (5.4.3)$$

2. In  $E, h_{t=0}^1 = 0$  and

$$\partial_t h^1 = \mathcal{G}_{\varepsilon}(h^1) + \frac{1}{\varepsilon} Q(h^1, h^1) + \mathcal{A}_{\varepsilon}^{(\delta)}(h^0).$$
(5.4.4)

The aim of this Section is to establish the following estimate of solutions to the system (5.4.3) - (5.4.4).

**Theorem 5.4.1** Let p = 1 or p = 2. There exist  $\beta_0$  in  $\mathbb{N}$  and  $\varepsilon_d$  in (0, 1] depending on d and the kernel of the Boltzmann operator such that:

For all  $\beta \ge \beta_0$ , for any  $\delta$  in  $(0, \delta_{k,1}]$  and any  $\lambda'_0$  in  $(0, \lambda_0)$  ( $\delta_{k,1}$  and  $\lambda_0$  defined in Proposition 5.3.6) there exist  $C_\beta$ ,  $\eta_\beta > 0$  such that for any  $0 < \varepsilon \le \varepsilon_d$  and  $h_{in}$  in  $\mathcal{E}^p$ , if

(i)  $\|h_{in}\|_{\mathcal{E}^p} \leq \eta_{\beta}$ ,

(ii)  $(h^0, h^1)$  is solution to the system (5.4.3) - (5.4.4),

then

$$\left\|h^{0}+h^{1}\right\|_{\mathcal{E}^{p}} \leqslant C_{\beta} \left\|h_{in}\right\|_{\mathcal{E}^{p}} e^{-\lambda_{0}^{\prime} t}.$$

The constants  $C_{\beta}$  and  $\eta_{\beta}$  are constructive and depends only on  $\beta$ , d,  $\delta$ ,  $\lambda'_0$  and the kernel of the Boltzmann operator.

**Remark 5.4.2 (Link with Theorem 5.2.3)** The existence and uniqueness for the perturbed Boltzmann equation (5.1.5) in  $\mathcal{E}^p$  has been proved for  $\varepsilon = 1$ , that is equivalent of  $\varepsilon$ fixed with constant depending on it, in [51] Theorems 5.3 and 5.5 respectively. The constants, as well as the smallness assumption on the initial data, in the theorem above are independent of  $\varepsilon$  and therefore this a priori result combined with existence and uniqueness developed in [51] and in [23] (for existence and uniqueness in E, see statements in Chapter 4) implies the existence and uniqueness independently of  $\varepsilon$  which is Theorem 5.2.3.

The next subsections deal with the estimates one can get for solutions to the system (5.4.3) - (5.4.4). We study each of them independently and the *a priori* exponential decay will be a straightforward application of these results together with a maximum principle argument.

Section 5.4.2 focuses on the *a priori* study of the equation in  $\mathcal{E}^p$ . Section 5.4.3 deals with (5.4.4) in *E*. Finally, Section 5.4.4 gathers the previous results to prove Theorem 5.4.1.

## 5.4.2 Study of equation (5.4.3) in $\mathcal{E}$

In this section we prove the following general proposition about the equation taking place in  $\mathcal{E}^p = W_v^{\alpha,1} W_x^{\beta,p} (\langle v \rangle^k)$ , for p = 1 or p = 2. We define the shorthand notation

$$\mathcal{E}^p_{\nu} = W^{\alpha,1}_v W^{\beta,p}_x \left( \langle v \rangle^k \nu \right).$$

**Proposition 5.4.3** Let p = 1 or p = 2 and  $0 < \varepsilon \leq 1$ . Let  $k > k_1^* = 2$ ,  $\beta > 2d/p$ . Let  $h_{in}$  be in  $\mathcal{E}^p$  and  $h^1$  in  $\mathcal{E}^p_{\nu}$ .

For any  $\delta$  in  $(0, \delta_{k,1}]$  and any  $\lambda'_0$  in  $(0, \lambda_0)$  ( $\delta_{k,1}$  and  $\lambda_0$  defined in Proposition 5.3.6) there exist  $\eta_0 > 0$  such that

(i)  $\|h_{in}\|_{\mathcal{E}^p} \leq \eta_0$ ,  $\|h^1\|_{\mathcal{E}^p_u} \leq \eta_0$ ,

(ii)  $h^0$  satisfies  $h^0_{t=0} = h_{in}$  and is solution to

$$\partial_t h^0 = \mathcal{B}_{\varepsilon}^{(\delta)}(h^0) + \frac{1}{\varepsilon}Q(h^0, h^0) + \frac{2}{\varepsilon}Q(h^0, h^1) + \frac{2}{\varepsilon}Q(h^0, h^1) + \frac{2}{\varepsilon}Q(h^0, h^1) + \frac{2}{\varepsilon}Q(h^0, h^1) + \frac{2}{\varepsilon}Q(h^0, h^0) + \frac{2}{\varepsilon}Q(h^0, h^0$$

then

$$\left\|h^{0}\right\|_{\mathcal{E}^{p}} \leqslant e^{-\frac{\lambda'_{0}}{\varepsilon^{2}}t} \left\|h_{in}\right\|_{\mathcal{E}^{p}}.$$

The constant  $\eta_0$  is constructive and depends only on  $\delta$ ,  $\lambda'_0$  and the kernel of the Boltzmann operator.

We need to control the bilinear term Q, which is given by the following lemma.

**Lemma 5.4.4** For all p = 1, 2 and  $\alpha$ ,  $\beta$  in  $\mathbb{N}$  such that  $\beta > 2d/p$ , there exists  $C_{\beta,p} > 0$  such that all f and g

$$\|Q(f,g)\|_{\mathcal{E}^p} \leqslant C_{\beta,p} \left( \|g\|_{\mathcal{E}^p_{\mathcal{U}}} \|f\|_{\mathcal{E}^p} + \|g\|_{\mathcal{E}^p} \|f\|_{\mathcal{E}^p_{\mathcal{U}}} \right).$$

This lemma has been proved in Lemma 5.16 in [51], which is adapted from interpolation results in [3] or duality arguments as in [84] Theorem 2.1.

#### **Proof of Proposition 5.4.3**

Consider  $\delta$  in  $(0, \delta_{k,1}]$  and  $\lambda'_0$  in  $(0, \lambda_0)$ . Take p = 1 or p = 2 and  $\beta > 2d/p$ . We have that

$$\partial_t h^0 = \mathcal{B}_{\varepsilon}^{(\delta)}(h^0) + \frac{1}{\varepsilon}Q(h^0, h^0) + \frac{2}{\varepsilon}Q(h^0, h^1).$$

Thanks to the dissipativity of property of  $\mathcal{B}_{\varepsilon}^{(\delta)}$ , more precisely the proof of Lemma 5.3.6, we have

$$\begin{split} & \frac{d}{dt} \left\| h^0 \right\|_{\mathcal{E}^p} \leqslant -\frac{\lambda_0}{\varepsilon^2 \nu_0} \left\| h^0 \right\|_{\mathcal{E}^p_{\nu}} + \frac{1}{\varepsilon} \left| \langle Q(h^0, h^0) + 2Q(h^0, h^1), h^0 \rangle_{\mathcal{E}^p} \right| \\ & \leqslant -\frac{\lambda_0}{\varepsilon^2 \nu_0} \left\| h^0 \right\|_{\mathcal{E}^p_{\nu}} + \frac{1}{\varepsilon} \left\| Q(h^0, h^0) + 2Q(h^0, h^1) \right\|_{\mathcal{E}^p}, \end{split}$$

where we used the scalar product notation to refer to the product operator appearing in  $W_v^{\alpha,1}W_x^{\beta,p}$  when one differentiates  $\|h\|_{W_v^{\alpha,1}W_x^{\beta,p}(\langle v \rangle^k)}$  (of the same form as (5.3.9)). For the second inequality we used Hölder inequality between  $L_x^p$  and  $L_x^{p/(p-1)}$  inside the product operator:

$$\int_{\mathbb{T}^d} \operatorname{sgn}(h^0) \left| h^0 \right|^{p-1} F(h^0) \, dx \leq \left\| h^0 \right\|_{L^p_x}^{p-1} \left\| F(h^0) \right\|_{L^p_x}$$

Then estimating Q using Lemma 5.4.4 yields

$$\frac{d}{dt} \left\| h^0 \right\|_{\mathcal{E}^p} \leqslant -\frac{1}{\varepsilon^2} \left[ \frac{\lambda_0}{\nu_0} - 2\varepsilon C_{\beta,p} \left( \left\| h^0 \right\|_{\mathcal{E}^p} + \frac{2}{\nu_0} \left\| h^1 \right\|_{\mathcal{E}^p_\nu} \right) \right] \left\| h^0 \right\|_{\mathcal{E}^p_\nu}, \tag{5.4.5}$$

$$l \ \nu_0 = \inf \left( \nu(v) \right) > 0.$$

we recall  $\nu_0 = \inf_{v \in \mathbb{R}^d} (\nu(v)) > 0$ 

Therefore, if

$$\|h^1\|_{\mathcal{E}^p_{\nu}} \leqslant \varepsilon^{-1} \frac{(\lambda_0 - \lambda'_0)}{8C_{\beta,p}} \quad \text{and} \quad \|h_{t=0}\|_{\mathcal{E}^p} \leqslant \varepsilon^{-1} \frac{(\lambda_0 - \lambda'_0)}{4\nu_0 C_{\beta,p}}$$

then  $\|h^0\|_{\mathcal{E}^p}$  is always decreasing in time with

$$\frac{d}{dt} \left\| h^0 \right\|_{\mathcal{E}^p} \leqslant -\frac{\lambda'_0}{\varepsilon^2 \nu_0} \left\| h^0 \right\|_{\mathcal{E}^p_{\nu}},$$

which hence yields the expected exponential decay by Grönwall Lemma.

#### 

## **5.4.3** Study of equations (5.4.4) in E

In the space  $E = H_{x,v}^{\beta} (\mu^{-1/2})$ , solutions to the perturbed Boltzmann equation enjoy an exponential decay. More precisely, [23] derived a precise Grönwall that we will now use to obtain estimates on the solution  $h^1$ . We will use the following shorthand notation

$$E_{\nu} = H_{x,v}^{\beta} \left( \mu^{-1/2} \nu^{1/2} \right)$$

In this section we use the previous theorem to obtain exponential decay of  $h^1$  in E. This result is stated in the following proposition, where  $C_t^0$  denotes the space of time-continuous functions.

**Proposition 5.4.5** Let p = 1 or p = 2,  $0 < \varepsilon \leq \varepsilon^d \leq 1$ ,  $\beta \geq s_0$  and  $\alpha \leq \beta$  ( $\varepsilon_d$  and  $s_0$  being constructive constants that will be defined in Theorem 5.4.7). Let  $h_{in}$  be in  $\mathcal{E}^p$  and  $h^0$  in  $C_t^0 \mathcal{E}^p$ .

For any  $\delta$  in  $(0, \delta_{k,1}]$  and any  $\lambda'_0$  in  $(0, \lambda_0)$  ( $\delta_{k,1}$  and  $\lambda_0$  defined in Proposition 5.3.6) there

exist  $\eta_1, C_1 > 0$  such that if

(i)  $\|h_{in}\|_{\mathcal{E}^p} \leq \eta_1$ ,

(*ii*) there exists  $C_0 > 0$  such that  $\|h^0\|_{\mathcal{E}^p} \leq C_0 e^{-\frac{\lambda_0 + \lambda'_0}{2\varepsilon^2}t} \|h_{in}\|_{\mathcal{E}^p}$ ,

(iii)  $h^1$  satisfies  $h^1_{t=0} = 0$  and is solution to

$$\partial_t h^1 = \mathcal{G}_{\varepsilon}(h^1) + \frac{1}{\varepsilon}Q(h^1, h^1) + \mathcal{A}_{\varepsilon}^{(\delta)}(h^0)$$

then

$$\left\|h^{1}\right\|_{E} \leqslant C_{1} e^{-\lambda_{0}^{\prime} t} \left\|h_{in}\right\|_{\mathcal{E}^{p}}$$

The constants  $C_1$  and  $\eta_1$  are constructive and depends only on  $\delta$ ,  $\lambda'_0$  and the kernel of the Boltzmann operator.

In order to prove Proposition 5.4.5 we need a new control on the bilinear term.

For any operator  $F: E \times E \longrightarrow E$ , we will say that F satisfies the property (H) if the following holds.

## **Property** (H):

- 1. for all  $g^1$ ,  $g^2$  in E we have  $\pi_L \left( F(g^1, g^2) \right) = 0$ , where  $\pi_L$  is the orthogonal projection on Ker (L) in  $L^2_v \left( \mu^{-1/2} \right)$  (see (5.2.1)),
- 2. for all s' > 0 there exists  $\mathcal{F}_F^{s'} : E \times E \longrightarrow \mathbb{R}^+$  such that for all multi-indexes j and l such that  $|j| + |l| \leq s'$ ,

$$\left| \langle \partial_l^j F(g^1, g^2), g^3 \rangle_{L^2_{x,v}(\mu^{-1/2})} \right| \leqslant \mathcal{F}_F^{s'}(g^1, g^2) \left\| g^3 \right\|_{L^2_{x,v}(\mu^{-1/2}\nu^{1/2})},$$

with  $\mathcal{F}_F^{s'} \leqslant \mathcal{F}_F^{s'+1}$ .

**Lemma 5.4.6** The Boltzmann linear operator Q satisfies the property (H) with

$$\forall s > d, \ \exists C_s > 0, \quad \mathcal{F}_Q^s(g,h) \leq C_s \left[ \|f\|_E \|g\|_{E_{\mathcal{U}}} + \|f\|_{E_{\mathcal{U}}} \|g\|_E \right].$$

The latter control on the bilinear part is from [23] Appendix A.2 (see Chapter 4 Appendix 4.A).

**Proof of Proposition** 5.4.5 We state below the estimate derived in [23] (note that this is a version of [23] Theorem 2.4 extended by estimates proved in [23] Propositions 2.2 and 7.1).

**Theorem 5.4.7** There exist  $0 < \varepsilon_d \leq 1$  and  $s_0$  in  $\mathbb{N}$  such that for any  $s \geq s_0$  and any  $\lambda_0''$  in  $(0, \lambda_0)$  there exists  $\delta_s$ ,  $C_s > 0$  such that,

• for any  $h_{in}$  in  $H^s_{x,v}\left(\mu^{-1/2}\right)$  with

$$\|h_{in}\|_{H^s_{x,v}\left(\mu^{-1/2}\right)} \leqslant \delta_s,$$

• for any operator  $F : H^s_{x,v}(\mu^{-1/2}) \times H^s_{x,v}(\mu^{-1/2}) \longrightarrow H^s_{x,v}(\mu^{-1/2})$  satisfying the property (H);

Then for all  $0 < \varepsilon \leq \varepsilon_d$  and for all  $g^1$ ,  $g^2$  in  $H^s_{x,v}(\mu^{-1/2})$ , if h is a solution to

$$\begin{cases} & \partial_t h = G_{\varepsilon}(h) + \frac{1}{\varepsilon} F(g^1, g^2) \\ & h_{t=0} = h_{in}, \end{cases}$$

and h is in Ker  $(G_{\varepsilon})$  for all time, then

$$\forall t \in \mathbb{R}^+, \quad \frac{d}{dt} \|h\|_{H^s_{x,v}(\mu^{-1/2})}^2 \leqslant -\frac{2\lambda_0''}{\nu_0^2} \|h\|_{H^s_{x,v}(\mu^{-1/2}\nu)}^2 + C_s \left(\mathcal{F}_F^s(g^1, g^2)\right)^2.$$

Now, let  $\lambda''$  be in  $(0, \lambda_0)$ ,  $s \ge s_0$  and  $0 < \varepsilon \le \varepsilon_d$ .

The proof of Proposition 5.4.5 will be done in two steps. First we study the projection of  $h^1$  onto Ker  $(G_{\varepsilon})$  and then its orthogonal part.

Estimate on the projection part. We have that, see the decomposition (5.4.2), that  $h^1 = h - h^0$  with h solution to the perturbed Boltzmann equation and thus satisfying  $\Pi_{\mathcal{G}}(h) = 0$ . We therefore have that

$$\Pi_{\mathcal{G}}(h^1) = -\Pi_{\mathcal{G}}(h^0).$$

Moreover, Theorem 5.2.1 tells us that  $\Pi_{\mathcal{G}}$  and  $\Pi_{G}$  coincide on E and thus

$$\Pi_G(h^1) = -\Pi_\mathcal{G}(h^0),$$

and assumption (*ii*) together with the shape of  $\Pi_{\mathcal{G}}$  (see (5.2.4)), there exists a constant  $C_{\Pi} > 0$ , depending only on the dimension d and s and the constant  $C_0$ , such that

$$\left\|\Pi_G(h^1)\right\|_{E_{\nu}} \leqslant C_{\Pi} e^{-\frac{\lambda_0 + \lambda'_0}{2\varepsilon^2}t} \left\|h_{in}\right\|_{\mathcal{E}^p}.$$
(5.4.6)

Estimate on the orthogonal part. Applying  $\Pi_G^{\perp} = \text{Id} - \Pi_G$ , the orthogonal projection onto  $(\text{Ker}(G_{\varepsilon}))^{\perp}$  in  $L^2_{x,v}(\mu^{-1/2})$ , to the differential equation satisfied by  $h^1$  yields

$$\partial_t \left( \Pi_G^{\perp}(h^1) \right) = G_{\varepsilon}(h^1) + \Pi_G^{\perp} \left( \frac{1}{\varepsilon} Q(h^1, h^1) + \mathcal{A}_{\varepsilon}^{(\delta)}(h^0) \right)$$
  
$$= G_{\varepsilon} \left( \Pi_G^{\perp}(h^1) \right) + \Pi_G^{\perp} \left( \frac{1}{\varepsilon} Q(h^1, h^1) + \mathcal{A}_{\varepsilon}^{(\delta)}(h^0) \right).$$
(5.4.7)

Moreover, we have by definition of  $\Pi_G$  and  $\pi_L$  (see (5.2.4) and (5.2.1)) that

$$(\pi_L(h) = 0) \implies (\Pi_G(h) = 0)$$

and therefore

$$\Pi_G^{\perp}\left(Q(h^1,h^1)\right) = Q(h^1,h^1),$$

since Q satisfies property (H).1. by Lemma 5.4.6. Plugging the latter equality into (5.4.7) gives

$$\partial_t \left( \Pi_G^{\perp}(h^1) \right) = G_{\varepsilon} \left( \Pi_G^{\perp}(h^1) \right) + \frac{1}{\varepsilon} Q(h^1, h^1) + \Pi_G^{\perp} \left( \mathcal{A}_{\varepsilon}^{(\delta)}(h^0) \right).$$

By definition,  $\Pi_G^{\perp}(h^1)$  is in  $(\text{Ker}(G_{\varepsilon}))^{\perp}$  for all time and thanks to the control on the Boltzmann operator Q in E (Lemma 5.4.6), we are able to use Theorem 5.4.7 with  $\lambda_0 > \lambda'_0$  to which we have to add the source term  $\Pi_G^{\perp}\left(\mathcal{A}_{\varepsilon}^{(\delta)}(h^0)\right)$ . This yields the following differential inequality, where we denote by C any positive constant independent of  $\varepsilon$ ,

$$\frac{d}{dt} \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E}^{2} \tag{5.4.8}$$

$$\leq -\frac{2\lambda_{0}''}{\nu_{0}^{2}} \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E_{\nu}}^{2} + C \left( \mathcal{F}_{Q}^{s}(h^{1},h^{1}) \right)^{2} + \left| \langle \Pi_{G}^{\perp} \left( \mathcal{A}_{\varepsilon}^{(\delta)}(h^{0}) \right), \Pi_{G}^{\perp}(h^{1}) \rangle_{E} \right|$$

$$\leq -\frac{2\lambda_{0}''}{\nu_{0}^{2}} \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E_{\nu}}^{2} + C \left\| h^{1} \right\|_{E}^{2} \left\| h^{1} \right\|_{E_{\nu}}^{2} + \left\| \Pi_{G}^{\perp} \left( \mathcal{A}_{\varepsilon}^{(\delta)}(h^{0}) \right) \right\|_{E} \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E},$$

where we used a Cauchy-Schwarz inequality on the last term on the right-hand side.

Then we can decompose  $h^1 = \prod_G (h^1) + \prod_G^{\perp} (h^1)$  to get first

$$\begin{split} \left\|h^{1}\right\|_{E}^{2} \left\|h^{1}\right\|_{E_{\nu}}^{2} \leqslant 4 \left\|\Pi_{G}^{\perp}(h^{1})\right\|_{E}^{2} \left\|\Pi_{G}^{\perp}(h^{1})\right\|_{E_{\nu}}^{2} + \frac{8}{\nu_{0}^{2}} \left\|\Pi_{G}(h^{1})\right\|_{E_{\nu}}^{2} \left\|\Pi_{G}^{\perp}(h^{1})\right\|_{E_{\nu}}^{2} \\ + \frac{4}{\nu_{0}^{2}} \left\|\Pi_{G}(h^{1})\right\|_{E_{\nu}}^{4}, \end{split}$$

into which we can plug the control on  $\left\|\Pi_G(h^1)\right\|_{E_{\nu}}^2$  we derived in (5.4.6) to obtain, with

 $\|h_{in}\| \leqslant \eta_1,$ 

$$\|h^1\|_E^2 \|h^1\|_{E_{\nu}}^2 \leqslant 4 \|\Pi_G^{\perp}(h^1)\|_E^2 \|\Pi_G^{\perp}(h^1)\|_{E_{\nu}}^2 + C\eta_1^2 \|\Pi_G^{\perp}(h^1)\|_{E_{\nu}}^2 + Ce^{-\frac{2(\lambda_0 + \lambda_0')}{\varepsilon^2}t} \|h_{in}\|_{\mathcal{E}^p}^4.$$

$$(5.4.9)$$

And finally, this inequality together with assumption (ii) gives the existence of a constant  $C_A > 0$  such that

$$\left\|\Pi_{G}^{\perp}\left(\mathcal{A}_{\varepsilon}^{(\delta)}(h^{0})\right)\right\|_{E}\left\|\Pi_{G}^{\perp}(h^{1})\right\|_{E} \leqslant \frac{C_{A}}{\varepsilon^{2}} \left\|h_{in}\right\|_{\mathcal{E}^{p}} e^{-\frac{\lambda_{0}+\lambda_{0}'}{2\varepsilon^{2}}t} \left\|\Pi_{G}^{\perp}(h^{1})\right\|_{E}.$$
(5.4.10)

We plug (5.4.9) and (5.4.10) into (5.4.8) and obtain, with C and C' being positive constants independent of  $\varepsilon$ ,

$$\frac{d}{dt} \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E}^{2} \leqslant - \left[ \frac{2\lambda_{0}''}{\nu_{0}^{2}} - \left( 4 \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E}^{2} + C\eta_{1}^{2} \right) \right] \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E_{\nu}}^{2} \\
+ C' \left( \left\| h_{in} \right\|_{\mathcal{E}^{p}}^{4} + \frac{1}{\varepsilon^{2}} \left\| h_{in} \right\|_{\mathcal{E}^{p}} \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E} \right) e^{-\frac{\lambda_{0} + \lambda_{0}'}{2\varepsilon^{2}}t}.$$

We now choose  $\eta_1$  sufficiently small so that

$$C\eta_1^2 \leqslant \frac{\lambda_0'' - \lambda_0'}{\nu_0^2},$$

which in turns implies

$$\frac{d}{dt} \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E}^{2} \leqslant - \left[ \frac{\lambda_{0}^{\prime\prime} + \lambda_{0}^{\prime}}{\nu_{0}^{2}} - 4 \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E}^{2} \right] \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E_{\nu}}^{2} + C^{\prime} \left( \left\| h_{in} \right\|_{\mathcal{E}^{p}}^{4} + \frac{1}{\varepsilon^{2}} \left\| h_{in} \right\|_{\mathcal{E}^{p}} \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E} \right) e^{-\frac{\lambda_{0} + \lambda_{0}^{\prime}}{2\varepsilon^{2}} t}.$$
(5.4.11)

We define

$$\eta_* = \frac{\lambda_0'' - \lambda_0'}{4\nu_0^2}$$

We have that  $h_{t=0}^1 = 0$  so we can define

$$t_0 = \sup\{t > 0, \quad \left\|\Pi_G^{\perp}(h^1)\right\|_E^2 < \eta_*\}.$$

Suppose that  $t_0 < +\infty$ , we therefore have for all t in  $[0, t_0]$ 

$$\frac{d}{dt} \left\| \Pi_G^{\perp}(h^1) \right\|_E^2 \leqslant -\frac{2\lambda_0'}{\nu_0^2} \left\| \Pi_G^{\perp}(h^1) \right\|_{E_{\nu}}^2 + C' \left( \|h_{in}\|_{\mathcal{E}^p}^4 + \frac{\sqrt{\eta_*}}{\varepsilon^2} \|h_{in}\|_{\mathcal{E}^p} \right) e^{-\frac{\lambda_0 + \lambda_0'}{2\varepsilon^2} t},$$

which gives

$$\forall t \in [0, t_0], \quad \frac{d}{dt} \left\| \Pi_G^{\perp}(h^1) \right\|_E^2 \leqslant -2\lambda_0' \left\| \Pi_G^{\perp}(h^1) \right\|_E^2 + C' \left( \|h_{in}\|_{\mathcal{E}^p}^4 + \frac{\sqrt{\eta_*}}{\varepsilon^2} \|h_{in}\|_{\mathcal{E}^p} \right) e^{-\frac{\lambda_0 + \lambda_0'}{2\varepsilon^2}t},$$

and by Gronwall lemma with  $\Pi_G^{\perp}(h^1)_{(t=0)} = 0$ ,

$$\begin{aligned} \forall t \in [0, t_0], \quad \left\| \Pi_G^{\perp}(h^1) \right\|_E^2 &\leqslant C' \left( \left\| h_{in} \right\|_{\mathcal{E}^p}^4 + \frac{\sqrt{\eta_*}}{\varepsilon^2} \left\| h_{in} \right\|_{\mathcal{E}^p} \right) \left( \int_0^t e^{-\frac{\lambda_0 + \lambda'_0}{2\varepsilon^2} s} e^{2\lambda'_0 s} \, ds \right) e^{-2\lambda'_0 t} \\ &\leqslant C' \left( \varepsilon^2 \left\| h_{in} \right\|_{\mathcal{E}^p}^4 + \sqrt{\eta_*} \left\| h_{in} \right\|_{\mathcal{E}^p} \right) \left( \int_0^{+\infty} e^{-\frac{\lambda_0 - \lambda'_0}{2} u} \, du \right) e^{-2\lambda'_0 t}, \end{aligned}$$

where we used the change of variable  $u = \varepsilon^{-2}s$  and we considered  $\varepsilon \leq 1/4$  (which only amounts to decreasing  $\varepsilon_d$ ).

Hence, there exists K > 0 independent of  $\varepsilon$  such that

$$\forall t \in [0, t_0], \quad \left\| \Pi_G^{\perp}(h^1) \right\|_E^2 \leqslant K(\eta_1^4 + \eta_1 \sqrt{\eta_*}).$$

If we thus chose  $\eta_1$  sufficiently small such that  $(\eta_1^4 + \eta_1 \sqrt{\eta_*})K < \eta_*/2$  we reach a contradiction when t goes to  $t_0$  since  $\|\Pi_G^{\perp}(h^1)\|_E^2(t_0) \ge \eta_*$ . Therefore, choosing  $\eta_1$  small enough independently on  $\varepsilon$  implies first that  $t_0 = +\infty$  and second that

$$\forall t \in [0, +\infty), \quad \left\| \Pi_{G}^{\perp}(h^{1}) \right\|_{E}^{2} \leq C \left\| h_{in} \right\|_{\mathcal{E}^{p}}^{2} e^{-2\lambda_{0}' t}.$$
 (5.4.12)

**End of the proof.** By just decomposing  $h^1$  into its projection and orthogonal part and using the estimates (5.4.6) and (5.4.12) gives the expected exponential decay for  $h^1$  in E.

## 5.4.4 Proof of Theorem 5.4.1

Let p = 1 or p = 2,  $\lambda''$  be in  $(0, \lambda_0)$ ,  $\beta \ge \beta_0 = s_0$  and  $0 < \varepsilon \le \varepsilon_d$ . All the constants used in this section are the ones constructed in Proposition 5.4.3 with  $(\lambda_0 + \lambda'_0)/2$  and Proposition 5.4.5 with  $\lambda'_0$ .

E is continuously embedded in  $\mathcal{E}^p_{\nu}$  because  $L^2_v(\mu^{-1/2}) \subset L^2_v(\langle v \rangle^k)$  (mere Cauchy-Schwarz inequality) and  $L^2_x \subset L^1_x$  because  $\mathbb{T}^d$  is bounded. Hence, there exists  $C_{E,\mathcal{E}} > 0$  such that

$$\frac{1}{\nu_0} \|\cdot\|_{\mathcal{E}^p} \leqslant \|\cdot\|_{\mathcal{E}^p_\nu} \leqslant C_{E,\mathcal{E}} \|\cdot\|_E.$$
(5.4.13)

We define

$$\eta = \min\left(\eta_0, \eta_1, \frac{\eta_0}{2C_{E,\mathcal{E}}C_1}\right),\,$$

and we assume  $||h_{in}||_{\mathcal{E}^p} \leq \eta$ . Since  $h_{t=0}^1 = 0$  we also define

$$t_0 = \sup\{t > 0, \|h^1\|_{\mathcal{E}^p_{\nu}} < \eta_0\}.$$

Suppose that  $t_0 < +\infty$ . Then, thanks to Proposition 5.4.3 we have that

$$\forall t \in [0, t_0], \quad \left\| h^0 \right\|_{\mathcal{E}^p} \leqslant \left\| h_{in} \right\|_{\mathcal{E}^p} e^{-\frac{\lambda_0 + \lambda'_0}{2\varepsilon^2} t}$$

We can thus apply Proposition 5.4.5 and get

$$\forall t \in [0, t_0], \quad \left\|h^1\right\|_E \leqslant C_1 \left\|h_{in}\right\|_{\mathcal{E}^p} e^{-\lambda'_0 t} \leqslant C_1 \eta \leqslant \frac{\eta_0}{2C_{E,\mathcal{E}}},$$

which is in contradiction with the definition of  $t_0$  thanks to (5.4.13). Therefore  $t_0 = +\infty$ and we have the expected exponential decay stated in Theorem 5.4.1 for all time.
# Part III

# A QUANTIC VERSION OF BOLTZMANN EQUATION FOR GASES OF BOSONS AND FERMIONS

# Chapter 6

# The homogeneous Boltzmann-Nordheim equation for bosons: local existence and uniqueness

The Boltzmann-Nordheim equation is a modification, based on physical considerations, of the Boltzmann equation that describes the dynamics of the distribution of particles in a quantum gas composed by bosons or fermions. We investigate the homogeneous Boltzmann-Nordheim equation for the particular case of bosons. We solve existence and uniqueness locally in time for any initial data that are bounded and with finite mass and energy, without any assumption of isotropy. We also show that moments of all order appear immediately for such solutions. Finally, we discuss the phenomenon of Bose-Einstein condensate in a gas of bosons at low temperature and the recent results associated to it.

## Contents

259 sate261 262 264 264 264 265 266 266 271 272
sate 261 262 264 264 264 265 265 266 271 272
<ul> <li>262</li> <li>264</li> <li>264</li> <li>265</li> <li>266</li> <li>271</li> <li>272</li> </ul>
<ul> <li>264</li> <li>264</li> <li>265</li> <li>266</li> <li>271</li> <li>272</li> </ul>
264 265 266 271 272
. 265 . 266 . 271 . 272
. 266 . 271 . 272
. 271 . 272
. 272
. 274
. 276
. 276
. 279
n 280
. 281
. 282
. 285
. 288
. 290
. 291
. 293
n 297
. 201
•

### 6.1 Introduction

This chapter deals with the dynamics of the distribution of particles in time and velocity,  $f(t, v) \ge 0$  in  $\mathbb{R}^+ \times \mathbb{R}^d$   $(d \ge 2)$ , for a dilute homogeneous quantum gas of bosons. In greater generality, the dynamics of particles undergoing binary collisions in quantum statistics is given by the Boltzmann-Nordheim equation

$$\partial_t f = Q(f), \text{ on } \mathbb{R}^+ \times \mathbb{R}^d$$
  
= 
$$\int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} B(v, v_*, \theta) \left[ f'(1 + \alpha f) f'_*(1 + \alpha f_*) - f(1 + \alpha f') f_*(1 + \alpha f'_*) \right] dv_* d\sigma,$$

where f',  $f_*$ ,  $f'_*$  and f are the values taken by f at v',  $v_*$ ,  $v'_*$  and v respectively and B is the collision kernel which encodes the physical properties of the collision process. Define:

$$\begin{cases} v' = \frac{v + v_*}{2} + \frac{|v - v_*|}{2}\sigma \\ v'_* = \frac{v + v_*}{2} - \frac{|v - v_*|}{2}\sigma \end{cases}, \text{ and } \cos \theta = \left\langle \frac{v - v_*}{|v - v_*|}, \sigma \right\rangle.$$

This equation has been derived by Nordheim (see [89]) using quantum statistics considerations. Basically, when  $\alpha = 0$  one recovers exactly the Boltzmann equation which rules the dynamics of particles in a dilute gas when only elastic binary collisions are taken into account. The main difference with the Boltzmann-Nordheim equation is that in quantum statistics the probability of two particles colliding not only depends on the number of particles undergoing the collision but also the number of particles already in the final state the latter collision yields. In the case of fermions ( $\alpha = -1$ ), this probability decreases and in the case of bosons it increases ( $\alpha = 1$ ).

The collision kernel  $B \ge 0$  contains all the information about the interaction between two particles with velocities v and  $v_*$ , and is determined by physics. We can mention here that one can derive this type of equations from Newton mechanics (coupled with quantum effects in the case of the Boltzmann-Nordheim equation) at least formally, see [28] or [30] for the classical mechanics case and [89] or [32] in the quantum case. However, if mathematically rigorous derivations are known for small times for the classical Boltzmann equation (Landford's theorem, see [65] or more recently [44][96]), we do not have, at the moment, the same kind of proof for the Boltzmann-Nordheim equation.

### 6.1.1 The problem and its motivations

All along this chapter we will assume that the collision kernel B can be decomposed as

$$B(v, v_*, \theta) = \Phi(|v - v_*|) b(\cos \theta),$$

which is a common assumption as it is more convenient and also covers a wide range of physical applications.

Moreover, we will consider only kernels with hard potentials or Maxwellian potentials ( $\gamma = 0$  hereinbelow), that is to say there is a constant  $C_{\Phi} > 0$  such that

$$\Phi(z) = C_{\Phi} z^{\gamma}, \ \gamma \in [0, 1], \tag{6.1.1}$$

and satisfying Grad's angular cutoff (see [48]), expressed here by the fact that we assume  $b \circ \cos$  to be continuous on  $(0, \pi)$  and to be integrable on the sphere:

$$l_b = \int_{\mathbb{S}^{d-1}} b\left(\cos\theta\right) d\sigma = \left|\mathbb{S}^{d-2}\right| \int_0^\pi b\left(\cos\theta\right) \sin^{d-2}\theta \, d\theta < \infty.$$
(6.1.2)

All those assumptions allow us to rewrite the Boltzmann-Nordheim equation with  $\alpha = 1$ , into the equation we are going to study

$$\partial_t f = C_{\Phi} \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} |v - v_*|^{\gamma} b(\cos \theta) \left[ f' f'_* (1 + f + f_*) - f f_* (1 + f' + f'_*) \right] dv_* d\sigma.$$
(6.1.3)

with the following decomposition

$$\partial_t f = Q^+(f) - fQ^-(f)$$

where we defined

$$Q^{+}(f) = C_{\Phi} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v - v_{*}|^{\gamma} b(\cos \theta) f' f'_{*}(1 + f + f_{*}) dv_{*} d\sigma, \qquad (6.1.4)$$

$$Q^{-}(f) = C_{\Phi} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v - v_{*}|^{\gamma} b(\cos \theta) f_{*}(1 + f' + f'_{*}) dv_{*} d\sigma.$$
(6.1.5)

In this chapter, we are first interested in the existence and uniqueness properties of (6.1.3). Then, we shall understand and quantify the possible appearance, in finite time, of a Bose-Einstein condensate in a gas of bosons. This condensate is a concentration of mass in velocity at the mean velocity. In mathematical terms, this can be seen as the appearance of a dirac function in the solution of the equation (6.1.3), noticeable by a blow-up in finite time.

Such a concentration is physically expected, based on various experiments and numerical simulations (see [40] for an overview of these results), as long as the temperature Tof the gas is below a critical temperature  $T_c(M_0)$  which depends on the mass  $M_0$  of the bosonic gas.

### 6.1.2 *A priori* expectations for the creation of a Bose-Einstein condensate

In this section, we use some properties of the Boltzmann-Nordheim equation for bosons to understand why a concentration phenomenon is expected. We emphasize here that everything is done *a priori* and should not be considered as a rigorous proof.

The first thing to notice is the symmetry property of the Boltzmann-Nordheim operator.

**Lemma 6.1.1** Let f be such that Q(f) is well-defined. Then for all  $\Psi(v)$  we have

$$\int_{\mathbb{R}^d} Q(f)\Psi \, dv = \frac{C_\Phi}{2} \int_{\mathbb{R}^d \times \mathbb{R}^d \times \mathbb{S}^{d-1}} q(f)(v, v_*) \left[\Psi'_* + \Psi' - \Psi_* - \Psi\right] \, d\sigma dv dv_*,$$

with

$$q(f)(v, v_*) = |v - v_*|^{\gamma} b(\cos \theta) ff_* (1 + f' + f'_*)$$

This result is well-known for the Boltzmann equation and is simply a play with the changes of variables  $(v, v_*) \rightarrow (v_*, v)$  and  $(v, v_*) \rightarrow (v', v'_*)$  and the symmetries of the operator q(f). A straightforward consequence is the *a priori* conservation of mass, momentum and energy for a solution of (6.1.3), f, associated to an initial data  $f_0$ , that is

$$\int_{\mathbb{R}^d} \begin{pmatrix} 1\\ v\\ |v|^2 \end{pmatrix} f(v) \, dv = \int_{\mathbb{R}^d} \begin{pmatrix} 1\\ v\\ |v|^2 \end{pmatrix} f_0(v) \, dv \tag{6.1.6}$$

The entropy associated to (6.1.3) is the following operator

$$S(f) = \int_{\mathbb{R}^d} \left[ (1+f) \log(1+f) - f \log(f) \right] \, dv$$

which is, a priori, always increasing in time. It has been proven (see [58]) that for given mass  $M_0$ , momentum  $v_0$  and energy  $E_0$ , there exists a unique maximizer of S of mass  $M_0$ , momentum  $v_0$  and energy  $E_0$  and this maximizer is of the form

$$F_{BE}(v) = m_0 \delta(v - v_0) + \frac{1}{e^{\frac{\beta}{2}(|v - v_0|^2 - \mu)} - 1},$$
(6.1.7)

with  $m_0 \ge 0$ ,  $\beta$  in  $(0, +\infty]$  is the inverse of the equilibrium temperature and  $-\infty < \mu \le 0$ is the chemical potential. Moreover, the following equality is satisfied:  $\mu . m_0 = 0$ . Besides, functions of the form  $F_{BE}$  fulfilling the same constraints are the only maximizers of the entropy. Therefore, for a given initial data  $f_0$ , the associated solution of the Boltzmann-Nordheim equation (6.1.3) should converge, in some sense, to the function  $F_{BE}$  associated to the physical quantities of  $f_0$ . Hence the appearance of a dirac function at  $v_0$  if  $m_0 \neq 0$ .

One can find in [69] or [41] that for a given  $(M_0, v_0, E_0)$  we have that  $m_0 = 0$  if and only if

$$M_0 \leqslant \frac{\zeta(3/2)}{(\zeta(5/2))^{3/5}} \left(\frac{4\pi}{3}\right)^{3/5} E_0^{3/5}.$$
(6.1.8)

According to [32], Chap. 2, the kinetic temperature of a bosonic gas is given by

$$T = \frac{m}{3k_B} \frac{E_0}{M_0}$$

which implies that, by plugging it into (6.1.8),  $m_0 = 0$  if and only if  $T \ge T_c(M_0)$  where we compute

$$T_c(M_0) = \frac{m\zeta(5/2)}{2\pi k_B \zeta(3/2)} \left(\frac{M_0}{\zeta(3/2)}\right)^{2/3}.$$

In the equations above,  $k_B$  is the constant of Boltzmann.

Initial data satisfying (6.1.8) are called subcritical (or critical in case of equality).

Therefore, for low temperature  $T < T_c(M_0)$  we expect our solution to split into a regular part and a dirac mass at  $v_0$  as it converges towards its equilibrium  $F_{BE}$  with  $m_0 \neq 0$ . Spohn, in [102], used this idea of a splitting into a regular and a singular part to derive a physical quantitative study of the Bose-Einstein condensate and its interactions with the normal fluid, in the case of radially symmetric (isotropic) solutions.

#### 6.1.3 Comparison with previous results

The first theorem of the present chapter deals with local-in-time existence and uniqueness of solutions to the bosonic Boltzmann-Nordheim equation for bounded initial datum  $f_0$  with bounded mass and energy (second moment).

The issue of existence and uniqueness for the homogeneous bosonic Boltzmann-Nordheim equation has been studied recently, especially by X. Lu [69][70][71] and M. Escobedo and J. J. L. Velázquez [40][41]. However, all those studies focused on the case of radially symmetric solutions  $f(t, v) = f(t, |v|^2)$  and in the case of hard potential with angular cut-off.

In his papers [69] and [70], X. Lu developped a global-in-time Cauchy theory for isotropic initial data with bounded mass and energy and extended the concept of solutions for isotropic distributions. In these cases he proved existence and uniqueness of radially symmetric solutions that preserve mass and energy. Moreover, he showed the boundedness of moments of order s > 2 as long as the initial data has a moment of order s.

Very recently, M. Escobedo and J. J. L. Velázquez in [40] used an idea developped by Carleman for the Boltzmann equation ([26]) in order to obtain a result of uniqueness and existence locally in time for radially symmetric solutions in the spaces  $L^{\infty}(1+|v|^{6+0})$ . We discussed above that the creation of the Bose-Einstein condensate leads to a blow-up in finite time. Therefore one cannot expect more than local-in-time results in  $L^{\infty}$ -spaces.

The *a priori* conservation of mass, momentum and energy seems to imply that the most natural space to tackle the Cauchy problem is  $L_2^1$ , the space of positive functions with bounded mass and energy. This was indeed the case for the homogeneous Boltzmann equation (see [68] and [77]). However, our quick look at the Bose-Einstein condensate told us that one may physically expect that a solution to (6.1.3) is bounded up to the appearance of a blow-up. Moreover, the  $L^{\infty}$ -norm is of great importance in the study of the Boltzmann-Nordheim operator in order to be able to deal with the trilinear part of the operator Q. Therefore it seems that the natural framework of the homogeneous Boltzmann-Nordheim equation for bosons is  $L_2^1 \cap L^{\infty}$ .

The present work shows a local-in-time existence and uniqueness result for initial data in  $L_2^1 \cap L^\infty$  without any isotropic requirement. Along the way, it also proves the immediate appearance of moments of all orders for these solutions.

The issue of the creation of a condensate of Bose-Einstein has been extensively studied experimentally and numerically in physics (see [40] for references on these results). Mathematically, a formal derivation of some properties of this condensate as well as its interactions with the regular part of the bosonic part has been studied in [102] in an isotropic framework.

In the series of papers [69][70][71], X. Lu proved, with not entirely constructive methods, a condensate phenomenon in the limit t goes to infinity. Indeed, he proved that the isotropic solutions he constructed tend to the regular part of their associated equilibrium  $F_{BE}$  (see (6.1.7)). But for low temperatures, the regular part of  $F_{BE}$  does not have the same mass than the initial solutions. This loss of mass proves the creation of a singular part in the limit. As mentionned in [71], this argument does not require the solution to be isotropic and the condensate Lu catches is to be understood as a concentration phenomenon in the limit t goes to infinity. This limiting behaviour neither prove nor prevent the creation of a Bose-Einstein condensate in finite time.

The appearance of Bose-Einstein in finite time has been mathematically shown in a recent breakthrough [40][41]. In the article [40] the authors showed that if the initial data is isotropic in  $L^{\infty}(1 + |v|^{6+0})$  and satisfies some properties about its distribution of mass near  $|v|^2 = 0$  then the associated isotropic solution is only define in finite time and its  $L^{\infty}$ -norm blows up. They achieve this work thanks to a thorough study of the concentration phenomenon occuring in a bosonic gas. The article [41] proves that supercritical initial

data indeed satisfy the blow-up assumptions in the case of radially symmetric solutions.

#### 6.1.4 Our strategy

We tackle the issue of the existence of solutions with an approximative scheme (see Section 6.7). More precisely, we truncate the Boltzmann-Nordheim operator Q and solve the associated differential equation using a Euler scheme. The sequence of functions we obtain is then proved to be weakly compact and goes to a solution of (6.1.3). The key ingredients are a new control on the operator  $Q^+$  for high and small relative velocities  $v - v_*$  as well as an extended version of Povzner inequality (see Section 6.3).

The proof of the uniqueness follows very closely the proof of uniqueness developed by S. Mischler and B. Wennberg in [77] for the homogeneous Boltzmann equation. Our extended version of Povzner inequality matches the main features of their proof. The main issue is the control of terms of the form  $|v - v_*|^{2+\gamma}$  that appear when one studies the evolution of the energy of solutions. This is achieve by the fact that bounded solutions of (6.1.3) happen to have more regularity (see Proposition 6.4.1) and thanks to an explicit control on the explosion at t = 0 of the moment of order  $2 + \gamma$  of solutions to (6.1.3) (see Proposition 6.5.5). The speed of the blow-up is exactly the one required to use a Nagumo's type uniqueness criterion in small times. The uniqueness for later time uses a Gronwall-type lemma which is available thanks to the boundedness of the moment of order  $2 + \gamma$  whenever t > 0 (see Section 6.5).

#### 6.1.5 Organisation of the chapter

Section 6.2 is dedicated to the statement and the description of the main results proved in this chapter.

The first problem we shall deal with is the uniqueness result. As said when we described our strategy (Section 6.1.4), this part requires the control of a little bit more than the  $L^{\infty}$ norm as well as the control of moments of order greater than 2.

A very important tool is an extended version of the Povzner inequality (first derived in [94]) and we shall use it throughout this chapter. The statement of this lemma and its proof are given in Section 6.3.

Section 6.4 focuses on an *a priori* boundedness property of solutions to the bosonic Boltzmann-Nordheim equation. Proposition 6.4.1 will allow us to control terms of the form  $|v|^{\gamma} f(t, v)$  in  $L^{\infty}$ .

The next section, Section 6.5, deals with the moments of solutions to (6.1.3). It is divided in two subsections. The first one is dedicated to the immediate appearance of bounded moments of all order, see Proposition 6.5.1. Then, Section 6.5.2 quantifies the explosion near t = 0 of the moment of order  $2 + \gamma$ .

Finally, Section 6.6 proves the uniqueness of bounded solutions preserving mass and energy.

Then we turn to the proof of existence of such bounded, mass and energy preserving solutions in Section 6.7. We construct our sequence of approximations in Section 6.7.2 and derive some of their properties. Section 6.7.3 shows that this sequence converges toward a mass-preserving solution of (6.1.3) and finally Section 6.7.4 proves that this limit is also energy-preserving.

### 6.2 Main results

We begin with the notations we shall use all along the chapter.

We are going to use spaces in the v and the t variables. Therefore, to shorten notations, we will index by v or t the spaces we are working on. The subscript v will always refer as  $\mathbb{R}^d$ , for instance  $L_v^1 = L^1(\mathbb{R}^d)$ ,  $L_{[0,T],v}^\infty = L^\infty([0,T] \times \mathbb{R}^d)$ . Moreover, we define

$$L_{2,v}^{1} = \left\{ f \in L_{v}^{1}, \quad \left\| (1+|v|^{2})f \right\|_{L_{v}^{1}} < +\infty \right\}.$$

Finally, we denote, for all s and t in  $\mathbb{R}^+$ ,

$$M_s(t) = \int_{\mathbb{R}^d} |v|^s f(t, v) \, dv.$$
 (6.2.1)

The first main theorem is the Cauchy problem for the Boltzmann-Nordheim equation for bosons.

### **Theorem 6.2.1** Let $f_0(v)$ be in $L^1_{2,v} \cap L^{\infty}_v$ .

Then there exists  $T_0 > 0$ , depending only on  $C_{\Phi}$ ,  $l_b$ ,  $\gamma$ ,  $\|f_0\|_{L^1_{2,v}}$  and  $\|f_0\|_{L^{\infty}_v}$ , such that there exists a unique f in  $L^{\infty}_{loc}([0, T_0), L^1_{2,v} \cap L^{\infty}_v)$  solution on (6.1.3) on  $[0, T_0) \times \mathbb{R}^d$  that preserves mass and energy.

Moreover, this solution satisfies

- $\bullet \ T_0=+\infty \quad or \quad \lim_{T\to T_0^-}\|f\|_{L^\infty_{[0,T]\times \mathbb{R}^d}}=+\infty,$
- f preserves the momentum of  $f_0$ ,
- for all s > 0 and for all  $0 < T < T_0$ ,

$$M_s(t) \in L^{\infty}_{loc}\left([T, T_0]\right).$$

• for all  $T < T_0$ ,

$$\sup_{[0,T]\times\mathbb{R}^d} \left( f(t,v) + \int_0^t \left(1+|v|^{\gamma}\right) f(s,v) \, ds \right) < \infty.$$

**Remark 6.2.2** We empasize here that moments appear as soon as t is strictly positive. However, we only get that  $M_s(t)$  is in  $L_{loc}^{\infty}([T, T_0))$ . This is slightly weaker than the result derived in [77] for the Boltzmann equation but it is explained by the fact that at  $T_0$  we can obtain a blow-up of the  $L^{\infty}$ -norm. The latter norm is not required for the control of the bilinear Boltzmann operator but is of great importance for the trilinear part of the Boltzmann-Nordheim operator.

Let us mention here that Theorem 6.2.1 implies a Bose-Einstein concentration phenomenon as time goes to infinity for subcritical initial data if they are globally defined. Indeed, Lu ([71] Theorem 2) proved in the case  $T < T_c(M_0)$  that distributional solutions (not necessarily isotropic) with finite mass and energy present a concentration phenomenon in the limit t goes to infinity.

The latter argument is however non explicit and does not prove any blow-up in finite time whereas [40] gives the appearance of a Bose-Einstein condensate in finite time in the isotropic setting. A work in progress is the proof of the creation of a condensate in finite time in our more general framework.

### 6.3 An extended version of a Povzner-type inequality

This section is dedicated to proving a refinement of a result in [77], which extends a Povzner-type inequality (see [94]) which captures the geometry of the collisions inside the Boltzmann kernel. The statement of the lemma is very close to Lemma 2.2 in [77].

**Lemma 6.3.1** Assume that  $b(\theta)$  is a locally bounded function and consider  $F \ge 1$  a function in  $L^{\infty}(\mathbb{R}^d \times \mathbb{R}^d \times \mathbb{S}^{d-1})$ .

For a given function 
$$\psi$$
 let

$$K_{\psi}(v,v_{*}) = \int_{\mathbb{S}^{d-1}} F(v,v_{*},\sigma)b(\theta) \left(\psi(|v_{*}'|^{2}) + \psi(|v'|^{2}) - \psi(|v_{*}|^{2}) - \psi(|v|^{2})\right) d\sigma.$$

Then one can write  $K_{\psi}(v, v_*) = G_{\psi}(v, v_*) - H_{\psi}(v, v_*)$ , where  $G_{\psi}$  and  $H_{\psi}$  satisfies the following inequalities (where we omit the subscript  $\psi$ ).

Let  $\chi(v, v_*) = 1 - \mathbf{1}_{\{|v| < |v_*| < 2|v|\}}$  then

i) If  $\psi(x) = x^{1+\alpha}$  with  $\alpha > 0$  then

$$|G(v, v_*)| \leq C_G \alpha (|v| |v_*|)^{1+\alpha}$$

and

$$H(v, v_*) \ge C_H \alpha \left( |v|^{2+2\alpha} + |v_*|^{2+2\alpha} \right) \chi(v, v_*).$$

ii) If  $\psi(x) = x^{1+\alpha}$  with  $-1 < \alpha < 0$  then

$$|G(v, v_*)| \leq C_G |\alpha| (|v| |v_*|)^{1+\alpha}$$

and

$$-H(v, v_*) \ge C_H |\alpha| \left( |v|^{2+2\alpha} + |v_*|^{2+2\alpha} \right) \chi(v, v_*).$$

iii) If  $\psi$  is a positive convex function that can be written  $\psi(x) = x\phi(x)$ , where  $\phi$  is concave, increasing to infinity, and such that for any  $\varepsilon > 0$  and any  $\alpha$  in (0,1), it satisfies  $(\phi(x) - \phi(\alpha x)) x^{\varepsilon} \to \infty$  as  $x \to \infty$ . Then, for all  $\varepsilon > 0$ ,

$$|G(v, v_*)| \leq C_G |v| \phi (|v|^2) |v_*| \phi (|v_*|^2)$$

and

$$H(v, v_*) \ge C_H \left( |v|^{2-\varepsilon} + |v_*|^{2-\varepsilon} \right) \chi(v, v_*).$$

In addition, there is a constant C > 0 such that  $\phi'(x) \leq C/(1+x)$  implies  $G(v, v_*) \leq C_G |v| |v_*|$ .

The constants in the Lemma depends on  $\alpha$ ,  $\psi$ ,  $\varepsilon$ , b and  $\|F\|_{L^{\infty}_{n,n,\sigma}}$ .

**Remark 6.3.2** As noticed in [77], the operator  $H_{\psi}$  can be taken monotonous in  $\psi$  in the following sense. If  $\psi_1 - \psi_2 \ge 0$  is convex then  $H_{\psi_1} - H_{\psi_2} \ge 0$ . This property will prove itself really useful to apply Lemma 6.3.1 to truncated sequences converging to convex functions.

**Proof of Lemma** 6.3.1 The proof of this result has been done in Lemma 2.2 of [77] in the case where F = 1. Therefore, our goal will be to compare our new operators H and G with  $H_1$  and  $G_1$  (obtained when F = 1) in each of the three cases.

We start with the term H which is quite straightforward. We define H to be the following operator, which coincides with  $H_1$  in [77] when F = 1,

$$H(v, v_*) = 4 \int_0^{\pi/2} \left[ \bar{b}(\theta) \bar{F}(v, v_*, \theta) + \bar{b}(\pi/2 - \theta) \bar{F}(v, v_*, \pi/2 - \theta) \right] \\ \times \left[ \cos^2\theta \,\psi \left( |v|^2 \right) + \sin^2\theta \,\psi \left( |v_*|^2 \right) - \psi \left( |v|^2 \cos^2\theta + |v_*|^2 \sin^2\theta \right) \right] \, d\theta,$$

$$(6.3.1)$$

where  $\bar{b}(\theta) = b(\theta) |d\sigma|(\theta)$  and  $\bar{F}(v, v_*, \theta) = \int_{-\pi}^{\pi} F(v, v_*, \sigma) d\omega$  where  $(\theta, \omega)$  are spherical coordinates parametrising  $\mathbb{S}^{d-1}$  ( $\omega$  is then a (d-2)-uple of angles).

The core of the proof is the fact that the term

$$\cos^2\theta \ \psi\left(|v|^2\right) + \sin^2\theta \ \psi\left(|v_*|^2\right) - \psi\left(|v|^2\cos^2\theta + |v_*|^2\sin^2\theta\right)$$

keeps the same sign if  $\psi$  is convex (positive sign) and if  $\psi$  is concave (negative sign). We have that

$$1 \leqslant F(v, v_*, \theta) \leqslant \|F\|_{L^{\infty}_{v, v_*, \sigma}}$$

and therefore when  $\psi$  is convex we have

$$H(v, v_*) \geqslant H_1(v, v_*)$$

and if  $\psi$  is concave

$$H(v, v_*) \leq ||F||_{L^{\infty}_{v, v_*, \sigma}} H_1(v, v_*).$$

This yields the expected inequalities i), ii) and iii) for the operator H since they hold true for  $H_1$ .

The proof for the operator G is more intricate and we shall write it in dimension d = 3 for sake of simplicity.

We follow the proof in [77] and we parametrise the sphere  $\mathbb{S}^2$  by

$$\mathbb{S}^2 = \{ (\theta, \omega), \quad -\pi \leqslant \omega \leqslant \pi, \ 0 \leqslant \theta \leqslant \pi/2 \}$$

with the measure

$$d\sigma = 4\sin\theta\cos\theta\,d\theta d\omega.$$

To shorten notation we define, for a given v and a given  $v_*$ 

$$Y(\theta) = |v|^2 \cos^2 \theta + |v_*|^2 \sin^2 \theta,$$
  
$$Z(\theta) = 2 |v| |v_*| \sin \theta \cos \theta.$$

In these coordinates, with the notations above, we have geometrically that

$$|v'|^2 = Y(\theta) + \tau Z(\theta) \cos \omega$$
$$|v'_*|^2 = Y(\pi/2 - \theta) - \tau Z(\theta) \cos \omega$$

where  $\tau$  denotes the sine of the angle between the vector v and v\*.

With these notations we obtain

$$K(v, v_*) = G(v, v_*) - H(v, v_*)$$

where H is given by (6.3.1) (after the change of variable  $\theta \to \pi/2 - \theta$ ) and

$$G(v, v_*) = 4 \int_0^{\pi/2} \bar{b}(\theta) \int_{-\pi}^{\pi} F(v, v_*, \sigma) \left[ \psi \left( Y(\theta) + \tau Z(\theta) \cos \omega \right) - \psi \left( Y(\theta) \right) \right] d\omega d\theta + 4 \int_0^{\pi/2} \bar{b}(\theta) \int_{-\pi}^{\pi} F(v, v_*, \sigma) \left[ \psi \left( Y(\pi/2 - \theta) - \tau Z(\theta) \cos \omega \right) - \psi \left( Y(\pi/2 - \theta) \right) \right] d\omega d\theta.$$

The two terms on the right-hand side will be treated the same way and therefore we focus only on

$$I = 4 \int_0^{\pi/2} \bar{b}(\theta) \int_{-\pi}^{\pi} F(v, v_*, \sigma) \left[ \psi \left( Y(\theta) + \tau Z(\theta) \cos \omega \right) - \psi \left( Y(\theta) \right) \right] \, d\omega d\theta.$$

Since  $\psi$  is increasing in all the cases we have that  $\psi(Y(\theta) + \tau Z(\theta) \cos \omega) - \psi(Y(\theta))$  is positive when  $-\pi/2 \leq \omega \leq \pi/2$  and negative elsewhere on  $[-\pi, \pi]$ . Thus,

$$\begin{aligned} |I| &\leqslant 4 \|F\|_{L^{\infty}_{v,v_{*},\sigma}} \int_{0}^{\pi/2} \bar{b}(\theta) \int_{-\pi}^{\pi} \left| \psi \left( Y(\theta) + \tau Z(\theta) \cos \omega \right) - \psi \left( Y(\theta) \right) \right| d\omega d\theta \\ &= 8 \|F\|_{L^{\infty}_{v,v_{*},\sigma}} \int_{0}^{\pi/2} \bar{b}(\theta) \int_{0}^{\pi} \left| \psi \left( Y(\theta) + \tau Z(\theta) \cos \omega \right) - \psi \left( Y(\theta) \right) \right| d\omega d\theta \\ &= 8 \|F\|_{L^{\infty}_{v,v_{*},\sigma}} \int_{0}^{\pi/2} \bar{b}(\theta) \qquad (6.3.2) \\ &\times \int_{0}^{\pi/2} \left[ \psi \left( Y(\theta) + \tau Z(\theta) \cos \omega \right) - \psi \left( Y(\theta) - \tau Z(\theta) \cos \omega \right) \right] d\omega d\theta, \end{aligned}$$

where we just made the change of variable  $\omega \to \pi - \omega$  on  $[\pi/2, \pi]$ .

Upper bound in cases i) and ii). In these cases, we have that  $\psi$  is twice differentiable and therefore we can integrate by part twice in the integral with respect to  $\omega$ . The first time we consider 1 to be the derivative of  $\omega$  to get

$$\begin{split} |I| \leqslant 8 \, \|F\|_{L^{\infty}_{v,v*,\sigma}} \int_{0}^{\pi/2} \bar{b}(\theta) \tau Z \\ \times \int_{0}^{\pi/2} \left( \omega \sin \omega \left[ \psi'(Y + \tau Z \cos \omega) + \psi'(Y - \tau Z \cos \omega) \right] \right) \, d\omega d\theta, \end{split}$$

and in the second integration by part considers  $\omega \sin \omega$  as a derivative to get

$$\begin{split} |I| \leqslant 8 \, \|F\|_{L^{\infty}_{v,v_{*},\sigma}} \int_{0}^{\pi/2} \bar{b}(\theta) \tau^{2} Z^{2} \int_{0}^{\pi/2} \left( \sin \omega - \omega \cos \omega \right) \\ & \times \left[ \psi''(Y + \tau Z \cos \omega) - \psi''(Y - \tau Z \cos \omega) \right] \, d\omega d\theta \qquad (6.3.3) \\ & + 16 \, \|F\|_{L^{\infty}_{v,v_{*},\sigma}} \int_{0}^{\pi/2} \bar{b}(\theta) \tau Z(\theta) \psi'(Y(\theta)) \, d\theta. \end{split}$$

On  $[0, \pi/2]$ , sin  $\omega - \omega \cos \omega$  is positive and thus

$$(\sin \omega - \omega \cos \omega) \left[ \psi''(Y + \tau Z \cos \omega) - \psi''(Y - \tau Z \cos \omega) \right] \\ \leqslant (\sin \omega - \omega \cos \omega) \left[ \left| \psi''(Y + \tau Z \cos \omega) \right| + \left| \psi''(Y - \tau Z \cos \omega) \right| \right],$$

which is the integrand dealt with in the case F = 1 in Lemma 2.2 in [77]. Hence, (6.3.3) becomes

$$|I| \leq \|F\|_{L^{\infty}_{v,v_{*},\sigma}} |I_{1}| + 16(1+\alpha) \|F\|_{L^{\infty}_{v,v_{*},\sigma}} \int_{0}^{\pi/2} \bar{b}(\theta)\tau Z(\theta)Y(\theta)^{\alpha} d\theta.$$
(6.3.4)

It only remains to control the last integral which can be achieve thanks to the fact that for  $\theta$  in  $[0, \pi/2]$ ,

$$Z(\theta)Y(\theta)^{\alpha} \le |v| |v_*| \left( |v|^2 + |v_*|^2 \right)^{\alpha}.$$
(6.3.5)

In the case  $-1 < \alpha < 0$  we have easily that (6.3.5) yields

$$Z(\theta)Y(\theta)^{\alpha} \leq 2^{\alpha} \left( |v|^2 + |v_*|^2 \right)^{1+\alpha},$$

which, combined with (6.3.4) gives us the expected inequality in point ii).

In the case  $\alpha > 0$  we use (6.3.5) in two different ways. First of all we notice the following

$$\forall \frac{|v|}{2} \le |v_*| \le 2 |v|, \quad Z(\theta) Y(\theta)^{\alpha} \le 2^{2+\alpha} 5^{\alpha} \left(|v| |v_*|\right)^{\alpha+1}.$$
(6.3.6)

Then basic computations yields

$$\forall \varepsilon > 0, \ \forall v, v_*, \quad |v| \, |v_*| \left( |v|^2 + |v_*|^2 \right)^{\alpha} \leq \frac{1}{\epsilon^{\alpha}} \left( |v| \, |v_*| \right)^{1+\alpha} + \varepsilon \left( |v|^2 + |v_*|^2 \right)^{1+\alpha}. \tag{6.3.7}$$

To conclude in that case we gather (6.3.6) and (6.3.7) to obtain that

$$\forall \varepsilon > 0, \quad Z(\theta)Y(\theta)^{\alpha} \leq C_{\varepsilon} \left( |v| |v_*| \right)^{1+\alpha} + \varepsilon \left( |v|^2 + |v_*|^2 \right)^{1+\alpha} \chi(v, v_*).$$

This last bound combined with (6.3.4) gives the inequality of point i), up to the fact that we choose  $\varepsilon$  small enough so that the second term in the right-hand side of the inequality above can be included in the inequality satisfied by H, which only leads to a slight change of definition for H in that case.

Upper bound in cases iii). We start from (6.3.2)

$$\begin{split} |I| \leqslant 8 \, \|F\|_{L^{\infty}_{v,v_{*},\sigma}} \\ \times \int_{0}^{\pi/2} \bar{b}(\theta) \int_{0}^{\pi/2} \left[ \psi \left( Y(\theta) + \tau Z(\theta) \cos \omega \right) - \psi \left( Y(\theta) - \tau Z(\theta) \cos \omega \right) \right] \, d\omega d\theta. \end{split}$$

In case *iii*) we consider  $\psi(x) = x\phi(x)$  with  $\psi$  being convex and  $\phi$  being concave. Therefore, the latters are almost everywhere differentiable with

$$\psi \left( Y - \tau Z \cos \omega \right) \geq \psi \left( Y \right) - \tau Z \cos \omega \psi'(Y),$$
  
$$\phi \left( Y + \tau Z \cos \omega \right) \leq \phi \left( Y \right) + \tau Z \cos \omega \phi'(Y).$$

Hence, developing every term in (6.3.2) yields

$$\begin{split} |I| \leqslant & \|F\|_{L^{\infty}_{v,v_{*},\sigma}} \\ & \times \int_{0}^{\pi/2} \bar{b}(\theta) \int_{0}^{\pi/2} \left[ 2\tau Z \cos \omega \phi(Y) + 2\tau Z \cos \omega Y \phi'(Y) + \tau^{2} Z^{2} \cos^{2} \omega \phi'(Y) \right] \, d\omega d\theta. \end{split}$$

We recognize here the term I for F = 1, see proof of Lemma 2.2 in [77]. Therefore

$$\left|I\right| \leqslant 2 \left\|F\right\|_{L^{\infty}_{v,v_{*},\sigma}} \left|I_{1}\right|,$$

and hence *iii*) follows directly from the case where F = 1.

This concludes the proof of Lemma 6.3.1.  $\blacksquare$ 

# 6.4 A priori control on the $L_v^{\infty} \left( (1+|v|^{\gamma})L_t^1 \right)$

This section is dedicated to proving an *a priori* estimate in the  $L_v^{\infty}$  space for solutions to (6.1.3), in small times. We cannot expect more than small times as we know from [40] that, even for radially symmetric solutions, there exists solutions with a blow-up in finite time.

We will prove the following result

**Proposition 6.4.1** Let  $f_0(v)$  be in  $L^1_{2,v} \cap L^{\infty}_v$ .

Let f be a non-negative solution of (6.1.3) in  $L^{\infty}_{[0,T_0)}(L^1_{2,v} \cap L^{\infty}_v)$ , with initial value  $f_0$ , satisfying the conservation of mass and energy.

Then for all  $0 \leq T < T_0$  there exists  $C_T > 0$  such that following controls holds

$$\sup_{[0,T]\times\mathbb{R}^d} \left( f(t,v) + \int_0^t \left(1 + |v|^\gamma\right) f(s,v) \, ds \right) \leqslant C_T.$$

### 6.4.1 Some properties of the Boltzmann-Nordheim operator

Here we gather and prove some useful properties about the positive operator  $Q^-$  and  $Q^+$ . First, we have the following control on the negative part

**Lemma 6.4.2** Let  $f \ge 0$  be in  $L^1_{2,v}$ . Then there exists  $C_{\gamma} > 0$  (given by (6.4.2)) such that

$$\forall v \in \mathbb{R}^d, \quad Q^-(f)(v) \ge C_{\Phi} l_b \left(1 + |v|^{\gamma}\right) \|f\|_{L^1_v} - C_{\Phi} C_{\gamma} l_b \|f\|_{L^1_{2,v}}.$$

**Proof of Lemma 6.4.2** We have that

$$Q^{-}(f)(v) = C_{\Phi} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v - v_{*}|^{\gamma} b(\cos \theta) f_{*} \left[ 1 + f_{*}' + f' \right] dv_{*} d\sigma.$$

We supposed that f is positive, thus

$$Q^{-}(f)(v) \ge C_{\Phi} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v - v_*|^{\gamma} b(\cos \theta) f_* \, dv_* d\sigma$$

Since  $\gamma$  is in [0, 1], we know that the following triangular inequality holds

$$(|v|^{\gamma} - |v_*|^{\gamma}) \le |v - v_*|^{\gamma} \le (|v|^{\gamma} + |v_*|^{\gamma}).$$
(6.4.1)

This yields

$$Q^{-}(f)(v) \geq C_{\Phi} l_b \int_{\mathbb{R}^d} \left( (1+|v|^{\gamma}) - (1+|v_*|^{\gamma}) \right) f_* \, dv_*$$
  
$$\geq C_{\Phi} l_b \left[ (1+|v|^{\gamma}) \, \|f\|_{L^1_v} - C_{\gamma} \int_{\mathbb{R}^d} (1+|v_*|^2) f_* \, dv_* \right].$$

because  $\gamma \leq 1$  and so there exists  $C_{\gamma} > 0$  such that for all  $x \ge 0$ ,

$$(1+x^{\gamma}) \leqslant C_{\gamma}(1+x^2).$$
 (6.4.2)

**Remark 6.4.3** If  $\int_{\mathbb{R}^d} f \ln f \, dv$  was finite then it would be possible to lower bound  $Q^-(f)$  by a quantity that is strictly positive. Unfortunately, this quantity decreases in the case of the classical Boltzmann equation whereas the decrease of entropy for Boltzmann-Nordheim is given by the decrease of

$$\int_{\mathbb{R}^d} \left[ (1+f) \ln(1+f) - f \ln f \right] \, dv$$

which does not bring any knowledge about a non-concentration property for f.

Moreover we also have the following general bound on the positive part

**Lemma 6.4.4** Let f and h be in  $L_{2,v}^1 \cap L_v^\infty (1+|v|^\gamma)$ . Then we have that for all  $\lambda > 0$  there exists  $C(\lambda) > 0$  such that

$$\lim_{\lambda \to 0} C(\lambda) = 0$$

and such that

$$\left\| C_{\Phi} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v - v_{*}|^{\gamma} b\left(\cos\theta\right) f' h'_{*} dv_{*} d\sigma \right\|_{L^{\infty}_{v}}$$

$$\leq C_{\gamma} C_{\Phi} l_{b} \left\| h \right\|_{L^{1}_{2,v}} \left[ C(\lambda) \left\| (1 + |v|^{\gamma}) f \right\|_{L^{\infty}_{v}} + \frac{2^{d-2}}{\lambda^{d-1}} C_{\gamma} \left\| f \right\|_{L^{1}_{2,v}} \right],$$

$$(6.4.3)$$

and

$$\left\| C_{\Phi} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v - v_{*}|^{\gamma} b\left(\cos\theta\right) f' h'_{*} dv_{*} d\sigma \right\|_{L^{\infty}_{v}}$$

$$\leq \left(1 + |v|^{\gamma}\right) C_{\gamma} C_{\Phi} l_{b} \left\|h\right\|_{L^{1}_{2,v}} \left[ C(\lambda) \left\|f\right\|_{L^{\infty}_{v}} + \frac{2^{d-2}}{\lambda^{d-1}} \left\|f\right\|_{L^{1}_{v}} \right],$$

$$(6.4.4)$$

where  $C_{\gamma}$  has been defined in (6.4.2) and  $C(\lambda)$  is given by (6.4.7).

**Proof of Lemma** 6.4.4 The  $L_v^{\infty}$ -norm is intricate and for this purpose we write the operator under another form. We use the Carleman representation of the operator (see [27]), which uses the final velocities after a collision, v' and  $v'_*$ , as the parameters we integrate against:

$$\int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} |v - v_*|^{\gamma} b\left(\cos\theta\right) f' h'_* dv_* d\sigma$$
  
=  $C_{\Phi} \int_{\mathbb{R}^d} dv' \int_{E_{vv'}} dv'_* \frac{1}{|v - v'|^{d-1}} \tilde{B}\left(2v - v' - v'_*, \frac{v' - v'_*}{|v' - v'_*|}\right) f' h'_*.$ 

In this form we have that  $E_{vv'}$  is the hyperplane orthogonal to v - v' going through v and the new operator  $\tilde{B}$  is such that

$$\tilde{B}(z,\omega) = 2^{d-2} \left\langle \frac{z}{|z|}, \omega \right\rangle^{d-2} B(z,\sigma).$$

With this new representation we have that

$$\begin{aligned} \left| \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v - v_{*}|^{\gamma} b\left(\cos\theta\right) f' h_{*}' dv_{*} d\sigma \right| \\ &\leq 2^{d-2} C_{\Phi} l_{b} \int_{\mathbb{R}^{d} \times \mathbb{R}^{d}} \frac{\left(1 + |v'|^{\gamma}\right) \left(1 + |v_{*}'|^{\gamma}\right)}{|v - v'|^{d-1}} f' h_{*}' dv' dv_{*}' \\ &\leq 2^{d-2} C_{\Phi} l_{b} C_{\gamma} \left\|h\right\|_{L^{1}_{2,v}} \int_{\mathbb{R}^{d}} \frac{\left(1 + |v'|^{\gamma}\right) f'}{|v - v'|^{d-1}} dv'. \end{aligned}$$
(6.4.5)

Now we are going to split our integral into velocities far from v and velocities close to v. Let us consider  $\lambda > 0$ ,

$$\int_{\mathbb{R}^d} \frac{(1+|v'|^{\gamma})f'}{|v-v'|^{d-1}} \, dv' \leqslant \|(1+|v|^{\gamma})f\|_{L^{\infty}_v} \int_{|v-v'|\leqslant\lambda} \frac{dv'}{|v-v'|^{d-1}} + \frac{C_{\gamma}}{\lambda^{d-1}} \, \|f\|_{L^1_{2,v}} \,. \tag{6.4.6}$$

The function  $1/|x|^{d-1}$  is integrable near 0 and therefore we can define

$$C(\lambda) = 2^{d-2}(1+\lambda) \int_{|x| \le \lambda} \frac{dv'}{|x|^{d-1}},$$
(6.4.7)

which fulfils the requirements of (6.4.3) in Lemma 6.4.4.

In (6.4.6), instead of taking  $\|(1+|v|^{\gamma})f\|_{L^{\infty}_{u}}$  we could also use

$$(1 + |v'|^{\gamma}) \leq (1 + |v - v'|^{\gamma})(1 + |v|^{\gamma})$$

and the fact that  $0 \leq \gamma \leq 1$ . Then taking the  $L_v^{\infty}$ -norm of f leads to the expected (6.4.4) with  $C(\lambda)$  described by (6.4.7).

### 6.4.2 A priori estimate: proof of Proposition 6.4.1

Let f be a solution of the Boltzmann-Nordheim equation (6.1.3), satisfying the assumptions of Proposition 6.4.1. This means that

$$\forall (t,v) \in [0,T_0) \times \mathbb{R}^d, \quad f(t,v) = f_0(v) + \int_0^t Q(f(s,\cdot))(v) \, ds. \tag{6.4.8}$$

We consider  $0 \leq T < T_0$  and we define the following quantities, for  $0 \leq t < T$  and v in  $\mathbb{R}^d$ :

$$e(f)(t,v) = \int_0^t (1+|v|^{\gamma}) f(s,v) \, ds,$$
  
$$E(f)(t,v) = f(t,v) + \int_0^t (1+|v|^{\gamma}) f(s,v) \, ds$$

In (6.4.8), we apply to Q(f) the Lemmas 6.4.2 and 6.4.4 (with  $\lambda_1$  and  $\lambda_2$  to be defined later) together with the conservations laws satisfied by f (6.1.6) to get

$$\begin{split} f(t,v) &\leqslant \|f_0\|_{L^{\infty}_{v}} - C_{\Phi} l_b \,\|f_0\|_{L^{1}_{v}} \int_{0}^{t} (1+|v|^{\gamma}) \,f(s,v) \,ds \\ &+ C_0 \int_{0}^{t} f(s,v) \,ds \\ &+ C_0 \left[ C(\lambda_1) \sup_{|u-v| \leqslant \lambda_1} \left( e(f)(t,u) \right) + \frac{2^{d-2}C_{\gamma}}{\lambda_1^{d-1}} T \,\|f_0\|_{L^{1}_{2,v}} \right] \\ &+ 2C_0 \,\|f\|_{L^{\infty}_{[0,T] \times \mathbb{R}^d}} \left[ C(\lambda_2) \sup_{|u-v| \leqslant \lambda_2} \left( e(f)(t,u) \right) + \frac{2^{d-2}C_{\gamma}}{\lambda_2^{d-1}} T \,\|f_0\|_{L^{1}_{2,v}} \right], \end{split}$$
(6.4.9)

where we set

$$M_0 = \min\{1, C_{\Phi} l_b \| f_0 \|_{L^1_v}\},$$
  

$$C_0 = C_{\gamma} C_{\Phi} l_b \| f_0 \|_{L^1_{2,v}}.$$

We emphasize here that we slightly changed Lemma 6.4.4 since we put the integrale in time before taking the supremum in v. Which is obtain by exactly the same proof but integrating first in time.

By bounding all quantities in (6.4.9) in time and velocities, one gets

$$\begin{split} M_{0}E(f)(t,v) &\leqslant \left[ C_{0}C(\lambda_{1}) + 2C_{0}C(\lambda_{2}) \left\| f \right\|_{L^{\infty}_{[0,T]\times\mathbb{R}^{d}}} \right] \sup_{[0,T]\times B(v,\lambda_{1}+\lambda_{2})} E(f) \\ &+ \left[ \left\| f_{0} \right\|_{L^{\infty}_{v}} + C_{0}T \left\| f \right\|_{L^{\infty}_{[0,T]\times\mathbb{R}^{d}}} + C_{0}C_{\gamma}2^{d-2} \left\| f_{0} \right\|_{L^{1}_{2,v}} T \left( \frac{1}{\lambda_{1}^{d-1}} + \frac{\left\| f \right\|_{L^{\infty}_{t,v}}}{\lambda_{2}^{d-1}} \right) \right] \end{split}$$

To conclude, we notice that by assumption  $||f||_{L^{\infty}_{[0,T]\times\mathbb{R}^d}}$  is finite and therefore we fix  $\lambda_1$ and  $\lambda_2$  small enough such that  $C_0C(\lambda_1) + 2C_0C(\lambda_2) ||f||_{L^{\infty}_{[0,T]\times\mathbb{R}^d}} \leq M_0/2$ . Then we take the supremum over t in [0,T] and v in  $\mathbb{R}^d$  to obtain the expected result.

### 6.5 Creation of moments of all order

In this section we prove that moments of all order appear immediately for solutions of the Boltzmann-Nordheim equation, as long as they are in  $L^{\infty}_{\text{loc}}([0, T_0), L^1_{2,v} \cap L^{\infty}_v)$ .

The first part of this section is dedicated to the proof of this *a priori* result. It thoroughly follows the proof established in [77] for the Boltzmann equation which was relying on a subtle Povzner inequality. Our extension of their Povzner-type inequality, see Section 6.3, allows us to apply their methods directly to the Boltzmann-Nordheim equation.

Then, in a second part we quantify the explosion of the  $(2 + \gamma)^{\text{th}}$  moment as time goes to 0. This estimate will be of great importance in the proof of the uniqueness, see Section 6.6. Here again we copy the arguments of [77] thanks to the extension of Povzner inequality, Lemma 6.3.1.

All the details of the proofs are exactly the same as for the Boltzmann equation given by S. Mischler and B. Wennberg in [77]. However, we still write them down roughly in order to show that they are indeed a straight combination of their proofs and our Povznertype inequality. Basically we show that we can apply our inequality each time they applied theirs and that the outcome is the same.

#### 6.5.1 A priori estimate on the moments of a solution

The immediate appearance of moments is characterized by the following proposition.

**Proposition 6.5.1** Let  $f_0(v)$  be in  $L^1_{2,v} \cap L^{\infty}_v$ .

Let f be a non-negative solution of (6.1.3) in  $L_{loc}^{\infty}([0,T_0), L_{2,v}^1 \cap L_v^{\infty})$ , with initial value  $f_0$ , satisfying the conservation of mass and energy. Then for all for all s > 0 and for all  $0 < T < T_0$ ,

$$\int_{\mathbb{R}^d} |v|^s f(t,v) \, dv \in L^{\infty}_{loc}\left([T,T_0]\right)$$

The proof of that proposition is done by induction and requires two lemmas, which gives the same estimates as the ones for the Boltzmann equation in [77]. The first one is the initialisation of the induction, it controls the  $L^1_{2+\gamma/2,v}$ -norm, and the second lemma gives an inductive bound on moments.

We start by taking  $f_0$ , f,  $T_0$  as in Proposition 6.5.1. We have that  $f_0$  is positive and such that  $(1 + |v|^2) f_0(v)$  is in  $L_v^1$ . Proposition A1 in the appendix of [77] gives the existence of  $\psi$  a positive convex function on  $\mathbb{R}^+$  such that there exists C > 0 such that

$$\int_{\mathbb{R}^d} \psi\left(|v|^2\right) f_0(v) \, dv \leqslant C.$$

Moreover,  $\psi$  can be written  $\psi(x) = x\phi(x)$ , where  $\phi$  is concave, increasing to infinity, and such that for any  $\varepsilon > 0$  and any  $\alpha$  in (0, 1), it satisfies  $(\phi(x) - \phi(\alpha x)) x^{\varepsilon} \to \infty$  as  $x \to \infty$ .

**Lemma 6.5.2** We have that for all T in  $[0, T_0)$  there exists  $c_T, C_T > 0$  such that for all  $0 \leq t \leq T$ ,

$$\int_{\mathbb{R}^d} f(t,v)\psi\left(|v|^2\right) dv + c_T \int_0^t \int_{\mathbb{R}^d} f(\tau,v) \left[M_{2+\gamma/2} + \psi\left(|v|^2\right)\right] dv d\tau$$
  
$$\leqslant \int_{\mathbb{R}^d} \psi\left(|v|^2\right) f_0(v) dv + C_T t.$$
(6.5.1)

**Proof of Lemma** 6.5.2 We fix T in  $[0, T_0)$  and we consider  $0 \le t \le T$ .

As proved in the proof of uniqueness in [77], we can construct an increasing sequence  $(\psi_n)_{n\in\mathbb{N}}$  of convex function converging pointwise to  $\psi$  and such that  $\psi_{n+1} - \psi_n$  is convex. The  $\psi_n$  are such that there exists a sequence  $(p_n)_{n\in\mathbb{N}}$  of polynomial of order 1 such that  $\psi_n - p_n$  is of compact support.

Moreover, for a given F satisfying the assumptions of Lemma 6.3.1, we have that  $H_{\psi_n}$  is non-negative and converges pointwise to  $H_{\psi}$  (see Remark 6.3.2) and  $|G_{\psi_n}(v, v_*)| \leq C_G |v| |v_*|$  for all n.

We know that f preserves mass and energy and therefore

$$\int_{\mathbb{R}^d} \left[ f(t,v) - f_0(v) \right] \psi_n\left( |v|^2 \right) \, dv = \int_{\mathbb{R}^d} \left[ f(t,v) - f_0(v) \right] \left( \psi_n\left( |v|^2 \right) - p_n\left( |v|^2 \right) \right) \, dv$$

Now,  $\psi_n - p_n$  is of compact support so we can use the fact that f is solution to the Boltzmann-Nordheim equation and the integral property of the operator Q, Lemma 6.1.1. This yields

$$\int_{\mathbb{R}^d} \left[ f(t,v) - f_0(v) \right] \psi_n \left( |v|^2 \right) dv$$
$$= \frac{C_\Phi}{2} \int_0^t \int_{\mathbb{R}^d \times \mathbb{R}^d \times \mathbb{S}^{d-1}} q(f)(\tau, v, v_*) \left[ \psi'_{n*} + \psi'_n - \psi_{n*} - \psi_n \right] dv dv_* d\tau,$$

with

$$q(f)(\tau, v, v_*) = |v - v_*|^{\gamma} b(\cos \theta) f(\tau) f_*(\tau) \left(1 + f'(\tau) + f'_*(\tau)\right).$$

We can decompose the left handside as in point *iii*) of Lemma 6.3.1 with  $F(v, v_*, \sigma) = 1 + f'_* + f'$  which fulfils the assumptions needed since f is in  $L^{\infty}_{[0,T]}$ . We obtain

$$\int_{\mathbb{R}^d} f(t,v)\psi_n\left(|v|^2\right) dv + \frac{C_{\phi}}{2} \int_0^t \int_{\mathbb{R}^d \times \mathbb{R}^d} f(\tau)f(\tau)_* |v - v_*|^{\gamma} H_{\psi_n} dv_* dv d\tau$$
$$= \int_{\mathbb{R}^d} f_0(v)\psi_n\left(|v|^2\right) dv + \frac{C_{\phi}}{2} \int_0^t \int_{\mathbb{R}^d \times \mathbb{R}^d} f(\tau)f(\tau)_* |v - v_*|^{\gamma} G_{\psi_n} dv_* dv d\tau (6.5.2)$$

Thanks to Lemma 6.3.1, our operators  $H_{\psi_n}$ ,  $G_{\psi_n}$ ,  $H_{\psi}$  and  $G_{\psi}$  satisfies exactly the same properties than the operators  $H_{\psi_n}$ ,  $G_{\psi_n}$ ,  $H_{\psi}$  and  $G_{\psi}$  in step 1 of proof of Theorem 1.1' in [77], with  $C_H$  and  $C_G$  depending on T which has been fixed. Equality (6.5.2) is exactly equality (3.4) in step 1 of proof of Theorem 1.1' in [77]. Thus we can compute this equality in exactly the same way as Mischler and Wennberg did.

This yields the inequality of Lemma 6.5.2.  $\blacksquare$ 

We turn to the induction property.

#### **Lemma 6.5.3** Let T be in $(0, T_0)$ .

For all n in  $\mathbb{N}$  there exists  $T_n > 0$  as small as we want such that

$$M_{2+(2n+1)\gamma/2}(T_n) < \infty$$

and such that for all t in  $[T_n, T]$  there exists  $C_T > 0$  and  $c_{T_n,T} > 0$  such that

$$M_{2+(2n+1)\gamma/2}(t) + c_T \int_{T_n}^t \left[ M_{2+(2n+1)\gamma/2}(\tau) + M_{2+(2n+3)\gamma/2}(\tau) \right] d\tau \leqslant C_{T_n,T}(1+t), \quad (6.5.3)$$

where  $M_s(t)$  is the moment of order s at time t, see (6.2.1).

**Proof of Lemma** 6.5.3 The fact that there exists  $T_{n+1}$  as small as we want such that  $M_{2+(2n+3)\gamma/2}(T_n) < \infty$  is given by the second term on the left-hand side of inequality (6.5.3) at rank n, and from the second term on the left-hand side of inequality (6.5.1) in Lemma 6.5.2 for n = 0.

Then the proof amounts to using Povzner inequality of Lemma 6.3.1 in exactly the same way as in [77], this time considering the function  $\psi$  to be  $\psi(x) = x^{1+(2n+3)\gamma/4}$  (so we use point *i*) of Lemma 6.3.1). We can therefore follow the proof of [77] (step 2 of proof of Theorem 1.1') and apply our Lemma 6.3.1 with  $F(v, v_*, \sigma) = 1 + f' + f'_*$  and the constants  $C_H$  and  $C_G$  depending on T via  $||f||_{[0,T],v}$ .

We are now able to finish the proof of the main proposition of this section. **Proof of Proposition** 6.5.1 First of all we notice that f is in  $L_{2,v}^1$  and therefore the Proposition is true for all s in [0, 2]. For s > 2 we just have to notice that the first term on the left-hand side of inequality (6.5.3) in Lemma 6.5.3 gives that  $M_{2+(2n+1)\gamma/2}(t) \leq C_{T_n,T}(1+T)$  for all t in  $[T_n,T]$ ,  $T_n$  given but being as small as we want. This being true for all n gives that  $M_s$  is bounded on  $[T_1, T_2]$  for all  $0 < T_1 \leq T_2 < T_0$ .

**Remark 6.5.4** We can emphasize here that this result is slightly different from the one in the Boltzmann equation. Indeed, in that case  $T_0 = +\infty$  and the bounds on the moments on  $[T, T_0)$  only depend on T. For the Boltzmann-Nordheim equation in our settings we can only reach locally bounded moments since we require the boundedness of the solution f in order to apply Povzner inequality. This boundedness property is only local (as shown by the explosion at  $T_0^-$  of the  $L^\infty$ -norm) and so we cannot expect the moments to be in  $L_{[0,T_0]}^\infty$  even if  $T_0 = +\infty$ .

### 6.5.2 Control of the explosion of the $L^1_{2+\gamma,v}$ -norm at time 0

In this section we show that  $M_{2+\gamma}$  explodes at most like 1/t when t goes to zero. This is the purpose of the next proposition.

**Proposition 6.5.5** Let  $f_0(v)$  be in  $L^1_{2,v} \cap L^{\infty}_v$ . Let f be a non-negative solution of (6.1.3) in  $L^{\infty}_{loc}([0,T_0), L^1_{2,v} \cap L^{\infty}_v)$ , with initial value  $f_0$ , satisfying the conservation of mass and energy. Then there exists  $0 < \tau < T_0$  and there exists  $C_{\tau} > 0$  such that

$$\forall t \in (0, \tau], \quad M_{2+\gamma}(t) \leqslant \frac{C_{\tau}}{t}.$$

**Proof of Proposition** 6.5.5 We take  $0 < t \leq T < T_0$ . Thanks to Proposition 6.5.1 we know that  $M_s(t)$  is bounded by a constant  $C_T > 0$  depending on T and t.

The technical Lemma 6.1.1 yields

$$\frac{d}{dt}M_{2+\gamma}(t) = \frac{C_{\phi}}{2} \int_{\mathbb{R}^d \times \mathbb{R}^d} |v - v_*|^{\gamma} ff_* K_{1+\gamma/2}(v, v_*) \, dv_* dv, \tag{6.5.4}$$

where  $K_{1+\gamma/2}(v, v_*)$  is given in Lemma 6.3.1 for  $\psi(x) = x^{1+\gamma/2}$ . We can use point *i*) of Lemma 6.3.1 since *f* is bounded on [0, T]. Hence, (6.5.4) becomes

$$\frac{d}{dt}M_{2+\gamma}(t) \leqslant \frac{C_{\phi}}{2} \int_{\mathbb{R}^d \times \mathbb{R}^d} |v - v_*|^{\gamma} ff_* \left[ C'_G |v|^{1+\gamma/2} |v_*|^{1+\gamma/2} - C'_H |v|^{2+\gamma} \right] dv_* dv,$$

where  $C_G$  and  $C_H$  are given in Lemma 6.3.1 (up to a multiplicative constant only depending on  $\gamma$ ).

Since f preserves the mass and the energy and since  $0 \leq \gamma \leq 1$ , a mere triangular inequality in the first term in the integral yields

$$\frac{d}{dt}M_{2+\gamma}(t) \leqslant C_{\phi}C_{\gamma} \|f_0\|_{L^1_{2,v}} M_{1+3\gamma/2} - \frac{C_{\phi}C'_H}{2} \|f_0\|_{L^1_v} M_{2+\gamma},$$
(6.5.5)

where  $C_{\gamma}$  has been defined in (6.4.2). Besides,

$$\forall \varepsilon > 0, \quad |v|^{1+3\gamma/2} \leqslant \varepsilon \, |v|^{2+\gamma} + C_{\varepsilon} \, |v|^{2\gamma}.$$

Then, because  $2\gamma \leq 2$  and taking  $\varepsilon$  small enough, (6.5.5) shows that there exists  $c_T$  and  $C_T$  positive constants depending on T and independent of t such that

$$\frac{d}{dt}M_{2+\gamma}(t) \leqslant c_T - C_T M_{2+2\gamma}(t).$$

We have the following Holder's inequality

$$M_{2+\gamma} \leqslant M_2^{1/2} M_{2+2\gamma}^{1/2}$$

and therefore

$$\frac{d}{dt}M_{2+\gamma}(t) \leqslant c_T - C_T M_{2+\gamma}^2(t).$$

So we have two cases to consider. Either  $M_{2+\gamma}(t)$  is bounded when t goes to 0 and then Proposition 6.5.5 is proven. Or there exists  $\tau$  such that  $M_{2+\gamma}(\tau) \ge \sqrt{2c_T/C_T}$  and then for all  $t \le \tau$ ,  $M_{2+\gamma}(t)$  is decreasing and

$$\forall t \in (0,\tau], \quad \frac{d}{dt} M_{2+\gamma}(t) \leqslant -\frac{C_T}{2} M_{2+\gamma}^2(t),$$

which gives the expected bound on  $M_{2+\gamma}(t)$ .

# 6.6 Uniqueness of solution for the Boltzmann-Nordheim equation

In this section we prove that there exists at most one solution to the Boltzmann-Nordheim equation for bosons (6.1.3) in the space  $L^{\infty}_{\text{loc}}([0, T_0), L^1_{2,v} \cap L^{\infty}_v)$  for  $T_0 > 0$ .

The proof relies on precise estimates on the  $L_v^1$ , the  $L_{2,v}^1$  and the  $L_v^\infty$ -norms of the difference of two solutions. The main problem arises when, in order to control the  $L_{2,v}^1$ -norm, one has to deal with terms of the form  $|v|^{2+\gamma}$ . A careful study allows us to control this weight thanks to the  $2 + \gamma$  moment of the solution (which has been studied in Section 6.5.2) and the fact that if f in  $L_{loc}^\infty([0, T_0), L_{2,v}^1 \cap L_v^\infty)$  is a solution to the Boltzmann-

Nordheim equation then, by Proposition 6.4.1,

$$\forall T \in [0, T_0), \ \exists N_T > 0, \quad \sup_{[0, T] \times \mathbb{R}^d} E[f](t, v) \leqslant N_T,$$
(6.6.1)

where

$$E[f](t,v) = f(t,v) + \int_0^t (1+|v|^{\gamma}) f(s,v) \, ds.$$

These estimates lead to a system of three differential and non-differential inequalities that we solve thanks to an extended Nagumo's uniqueness criterion for small times and an extended Gronwall lemma for larger times.

In the end, we prove the following theorem.

**Theorem 6.6.1** Let  $h_0(v)$  be in  $L_{2,v}^1 \cap L_v^\infty$ . Let f and g be two non-negative solutions of (6.1.3) in  $L_{loc}^\infty([0,T_0), L_{2,v}^1 \cap L_v^\infty)$  satisfying the conservation of mass and energy. If f and g have the same initial data  $h_0$  then f = g on  $[0,T_0)$ .

For now on we take f and g satisfying the assumptions of Theorem 6.6.1. In order to shorten notations we still denote by  $N_T$  the maximum of  $N_T$  for f and for g, defined in (6.6.1).

## **6.6.1** Evolution of $||f - g||_{L^{1}_{+}}$

First of all we can write the following algebraic identity which we are going to use throughout this section.

$$abc - def = \frac{1}{2}(a - d)(bc + ef) + \frac{a + d}{4}\left[(b - e)(c + f) + (c - f)(b + e)\right].$$
 (6.6.2)

We have the following differential inequality,

**Lemma 6.6.2** For all T in  $[0, T_0)$ , there exists  $C_T > 0$  such that for all t in [0, T],

$$\frac{d}{dt} \|f - g\|_{L^1_v} \leq C_T \left[ \|f - g\|_{L^1_v} + \|f - g\|_{L^1_{2,v}} \right].$$

 $C_T$  only depends on  $C_{\Phi}$ ,  $l_b$ ,  $\gamma$ ,  $\|h_0\|_{L^1_{2,v}}$  and  $N_T$  (see (6.6.1)).

**Proof of Lemma** 6.6.2 We fix T in  $[0, T_0)$  and we consider t in [0, T]. Thanks to the

integral property of Q, Lemma 6.1.1,

$$\frac{d}{dt} \|f - g\|_{L^{1}_{v}} = \int_{\mathbb{R}^{d}} \operatorname{sgn}(f - g) \partial_{t} (f - g) \, dv = \int_{\mathbb{R}^{d}} \operatorname{sgn}(f - g) \left(Q(f) - Q(g)\right) \, dv 
= \frac{C_{\Phi}}{2} \int_{\mathbb{R}^{d} \times \mathbb{R}^{d} \times \mathbb{S}^{d-1}} b\left(\cos \theta\right) |v - v_{*}|^{\gamma} P(f, g) \left[\Psi'_{*} + \Psi' - \Psi_{*} - \Psi\right] \, d\sigma dv_{*} dv,$$
(6.6.3)

where we wrote  $\Psi(t, v) = \operatorname{sgn}(f - g)(t, v)$  and

$$P(f,g) = ff_*(1+f'+f'_*) - gg_*(1+g'+g'_*)$$
(6.6.4)

We easily have that  $|[\Psi'_* + \Psi' - \Psi_* - \Psi] \leq 4.$ 

Furthermore, using the arithmetic identity (6.6.2) we compute

$$\begin{aligned} |P(f,g)| \leqslant & \frac{1+2N_T}{2} \left| f - g \right| (f_* + g_*) + \frac{1+N_T}{2} (f+g) \left| f_* - g_* \right| \\ & + \frac{1}{4} (f+g) (f_* + g_*) \Big[ \left| f' - g' \right| + \left| f'_* - g'_* \right| \Big]. \end{aligned}$$

We plug these two inequalities inside (6.6.3) which we cut into for integrals. The change of variable  $(v, v_*) \rightarrow (v_*, v)$  shows that the first two terms are equal as well as the last two. Thus,

$$\frac{d}{dt} \|f - g\|_{L^{1}_{v}} \leq 2C_{\Phi} l_{b} \left(1 + 2N_{T}\right) \int_{\mathbb{R}^{d} \times \mathbb{R}^{d}} |v - v_{*}|^{\gamma} |f - g| \left(f_{*} + g_{*}\right) dv_{*} dv + C_{\Phi} \int_{\mathbb{R}^{d} \times \mathbb{R}^{d} \times \mathbb{S}^{d-1}} b\left(\cos \theta\right) |v - v_{*}|^{\gamma} (f + g) (f_{*} + g_{*}) \left|f' - g'\right| \, d\sigma dv_{*} dv$$

The first integral is easily dealt with by a mere triangular inequality together with (6.4.2). In the second integral we use the change of variable  $(v, v_*) \rightarrow (v', v'_*)$  and the terms in v' and  $v'_*$  are dealt with Lemma 6.4.4, inequality (6.4.4) with  $\lambda = 1$ . Therefore,

$$\begin{aligned} \frac{d}{dt} \|f - g\|_{L^{1}_{v}} &\leq 2C_{\Phi} l_{b} \left(1 + 2N_{T}\right) C_{\gamma} \left(\|h_{0}\|_{L^{1}_{2,v}} \|f - g\|_{L^{1}_{v}} + \|h_{0}\|_{L^{1}_{v}} \|f - g\|_{L^{1}_{2,v}}\right) \\ &+ C_{\gamma} C_{\Phi} l_{b} \|h_{0}\|_{L^{1}_{2,v}} \left[2C(1)N_{T} + 2^{d-1}C_{\gamma} \|h_{0}\|_{L^{1}_{2,v}}\right] \int_{\mathbb{R}^{d}} \left(1 + |v|^{\gamma}\right) |f - g| \ dv. \end{aligned}$$

By setting  $C_T$  the maximum among the multiplicative constants above, we reach the inequality of Lemma 6.6.2.

# 6.6.2 Evolution of $||f - g||_{L^{1}_{2,v}}$

The differential inequality satisfies by  $||f - g||_{L^{1}_{2,v}}$  is more intricate and requires to control the  $(2 + \gamma)^{\text{th}}$  moments of g and f by the  $L^{1}_{v}$ -norm of the difference. This is achieve thanks

to the next lemma.

**Lemma 6.6.3** For all T in  $[0, T_0)$ , there exists  $C_T > 0$  such that for all t in [0, T],

$$\frac{d}{dt} \|f - g\|_{L^{1}_{2,v}} \leq C_{T} \left[ M_{2+\gamma}(t) \|f - g\|_{L^{1}_{v}} + \|f - g\|_{L^{1}_{2,v}} + \|f - g\|_{L^{\infty}_{[0,T],v}} \right].$$

 $M_{2+\gamma}$  is the  $(2+\gamma)^{th}$  moment of f+g (see (6.2.1)) and  $C_T$  only depends on  $C_{\Phi}$ ,  $l_b$ ,  $\gamma$ ,  $\|h_0\|_{L^1_{2,\eta}}$  and  $N_T$  (see (6.6.1)).

**Proof of Lemma** 6.6.3 We fix T in  $[0, T_0)$  and we consider t in [0, T]. As in the proof of Lemma 6.6.2 we have

$$\frac{d}{dt} \|f - g\|_{L^1_v} = \frac{C_\Phi}{2} \int_{\mathbb{R}^d \times \mathbb{R}^d \times \mathbb{S}^{d-1}} b |v - v_*|^{\gamma} P(f, g) \left[ \Psi'_* + \Psi' - \Psi_* - \Psi \right] \, d\sigma dv_* dv, \quad (6.6.5)$$

where this time we wrote  $\Psi(t, v) = \operatorname{sgn}(f - g)(t, v) \left(1 + |v|^2\right)$  and P(f, g) is still given by (6.6.4).

Using the algebraic inequality (6.6.2) and using the change of variable  $(v, v_*) \rightarrow (v_*, v)$ we obtain

$$\frac{d}{dt} \|f - g\|_{L^1_v} = C_{\Phi} \left( \frac{1}{2} I_1 + \frac{1}{4} I_2 + \frac{1}{8} I_3 + \frac{1}{4} I_4 \right)$$
(6.6.6)

with

$$\begin{split} I_{1} &= \int_{\mathbb{R}^{d} \times \mathbb{R}^{d} \times \mathbb{S}^{d-1}} b|v - v_{*}|^{\gamma} \left[ G(\Psi) - \Psi \right] (f - g)(f_{*} + g_{*}) \, d\sigma dv_{*} dv, \\ I_{2} &= \int_{\mathbb{R}^{d} \times \mathbb{R}^{d} \times \mathbb{S}^{d-1}} b|v - v_{*}|^{\gamma} \left[ G(\Psi) - \Psi \right] (f - g)(f_{*}(f' + f'_{*}) + g_{*}(g' + g'_{*})) \, d\sigma dv_{*} dv, \\ I_{3} &= \int_{\mathbb{R}^{d} \times \mathbb{R}^{d} \times \mathbb{S}^{d-1}} b|v - v_{*}|^{\gamma} \left[ G(\Psi) - \Psi \right] (f + g)(f_{*} - g_{*})(f' + f'_{*} + g' + g'_{*}) \, d\sigma dv_{*} dv, \\ I_{4} &= \int_{\mathbb{R}^{d} \times \mathbb{R}^{d} \times \mathbb{S}^{d-1}} b|v - v_{*}|^{\gamma} \left[ G(\Psi) - \Psi \right] (f + g)(f_{*} + g_{*})(f'_{*} - g'_{*}) \, d\sigma dv_{*} dv, \end{split}$$

where we defined  $G(\Psi) = \Psi'_* + \Psi' - \Psi_*$  and we have straightforwardly

$$|G(\Psi)| \leq 3 + |v'|^2 + |v'_*|^2 + |v_*|^2 = 2\left(1 + |v_*|^2\right) + \left(1 + |v|^2\right).$$
(6.6.7)

Thanks to the latter bound on  $G(\Psi)$  and the fact  $\Psi(f-g) = (1+|v|^2)|f-g|$  we find

$$|I_{1}| \leq 2l_{b} \int_{\mathbb{R}^{d} \times \mathbb{R}^{d}} \left(1 + |v_{*}|^{2}\right) \left(|v|^{\gamma} + |v_{*}|^{\gamma}\right) |f - g| \left(f_{*} + g_{*}\right) dv dv_{*}$$
  
$$\leq 4l_{b} C_{\gamma}^{2} \|h_{0}\|_{L_{2,v}^{1}} \|f - g\|_{L_{2,v}^{1}} + 2l_{b} C_{\gamma} M_{2+\gamma} \|f - g\|_{L_{v}^{1}}, \qquad (6.6.8)$$

where  $C_{\gamma}$  has been defined in (6.4.2).

The term  $I_2$  is dealt exactly the same way, remembering the f and g are bounded by  $N_T$ .

$$|I_2| \leq 8N_T l_b C_{\gamma}^2 \|h_0\|_{L^1_{2,v}} \|f - g\|_{L^1_{2,v}} + 4N_T l_b C_{\gamma} M_{2+\gamma} \|f - g\|_{L^1_v}.$$
(6.6.9)

In the term  $I_3$  we make the change of variable  $(v, v_*) \to (v_*, v)$  and after bounding the terms in v' and  $v'_*$  by  $N_T$  we recover  $|I_1|$ . Therefore,

$$|I_3| \leq 8N_T l_b C_{\gamma}^2 \|h_0\|_{L^1_{2,v}} \|f - g\|_{L^1_{2,v}} + 4N_T l_b C_{\gamma} M_{2+\gamma} \|f - g\|_{L^1_v}.$$
(6.6.10)

The last term,  $I_4$ , is more intricate and we have to deal with it carefully so that the terms of order  $2 + \gamma$  in v only appear in front of the  $L_{2,v}^1$ -norm of f - g.

First of all, thanks to (6.6.7), we have

$$|G(\Psi) - \Psi| \leq 2\left[\left(1 + |v|^2\right) + \left(1 + |v_*|^2\right)\right],$$

so that

$$|I_4| \leq 2 \int_{\mathbb{R}^{2d} \times \mathbb{S}^{d-1}} b|v - v_*|^{\gamma} \left[ \left( 1 + |v|^2 \right) + \left( 1 + |v_*|^2 \right) \right] (f+g)(f_* + g_*) \left| f'_* - g'_* \right| \, d\sigma dv_* dv.$$

Then, the change of variable  $(v, v_*) \to (v_*, v)$  followed by the change of variable  $\sigma \to -\sigma$ , which brings v' to  $v'_*$  and reciprocally, gives

$$\begin{aligned} \int_{\mathbb{R}^d \times \mathbb{R}^d \times \mathbb{S}^{d-1}} b\left(\cos \theta\right) |v - v_*|^{\gamma} \left(1 + |v_*|^2\right) (f + g)(f_* + g_*) \left|f'_* - g'_*\right| \, d\sigma dv_* dv \\ &= \int_{\mathbb{R}^d \times \mathbb{R}^d \times \mathbb{S}^{d-1}} b\left(-\cos \theta\right) |v - v_*|^{\gamma} \left(1 + |v|^2\right) (f + g)(f_* + g_*) \left|f'_* - g'_*\right| \, d\sigma dv_* dv. \end{aligned}$$

Therefore, if we denote  $\tilde{b}(x) = b(x) + b(-x)$  we obtain

$$|I_{4}| \leq 2 \int_{\mathbb{R}^{d} \times \mathbb{R}^{d} \times \mathbb{S}^{d-1}} \tilde{b}|v - v_{*}|^{\gamma} \left(1 + |v|^{2}\right) (f + g)(f_{*} + g_{*}) \left|f_{*}' - g_{*}'\right| d\sigma dv_{*} dv$$

$$\leq 16 l_{b} C_{\gamma} \|h_{0}\|_{L_{2,v}^{1}}^{2} \|f - g\|_{L_{[0,T],v}^{\infty}}$$

$$+ 2 \int_{\mathbb{R}^{d}} dv \left(1 + |v|^{2}\right) |v|^{\gamma} (f + g) \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \tilde{b}(f_{*} + g_{*}) \left|f_{*}' - g_{*}'\right| d\sigma dv_{*}.$$
(6.6.11)

The last integral is dealt with in the same way as in the proof of Lemma 6.4.4, (6.4.6), by studying the cases v' is close to v and when not. We use the Carleman representation of

this integral, which reads, with  $v_* = v'_* + v' - v$ ,

$$\begin{split} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \tilde{b}(f_{*} + g_{*}) \left| f_{*}' - g_{*}' \right| \, d\sigma dv_{*} &= \int_{\mathbb{R}^{d} \times E_{vv'}} \frac{\tilde{b}}{|v - v'|^{d-1}} (f_{*} + g_{*}) \left| f_{*}' - g_{*}' \right| \, dv' dv_{*}' \\ &\leqslant \int_{\mathbb{R}^{d}} \left| f_{*}' - g_{*}' \right| \left( \int_{\mathbb{R}^{d}} \frac{\tilde{b}}{|v - v'|^{d-1}} (f_{*} + g_{*}) \, dv' \right) \, dv_{*}' \\ &\leqslant l_{b} \int_{\mathbb{R}^{d}} \left| f_{*}' - g_{*}' \right| \left( 2C(1)N_{T} + 2^{d-2} \times 2 \, \|h_{0}\|_{L_{v}^{1}} \right) \, dv_{*}', \end{split}$$

where C(1) is defined in (6.4.7).

We plug the latter inequality into (6.6.11) to obtain the following control from above

$$|I_4| \leq 16l_b C_\gamma \|h_0\|_{L^1_{2,v}}^2 \|f - g\|_{L^\infty_{[0,T],v}} + 4l_b C_\gamma \left(C(1)N_T + 2^{d-2} \|h_0\|_{L^1_v}\right) M_{2+\gamma} \|f - g\|_{L^1_v}.$$
(6.6.12)

To conclude we gather (6.6.8), (6.6.9), (6.6.10) and (6.6.12) into (6.6.6)

# 6.6.3 Control of $||f - g||_{L^{\infty}_{[0,T],v}}$

For the  $L^{\infty}$ -norm of f - g, we derive the following inequality

**Lemma 6.6.4** There exists  $\tau$  in  $[0, T_0)$ , there exists  $C_{\tau} > 0$  such that for all t in  $[0, \tau]$  and for all m in  $\mathbb{N}$ ,

$$\|f - g\|_{L^{\infty}_{[0,t],v}} \leqslant C_{\tau} \sup_{[0,t],v} \|f - g\|_{L^{1}_{2,v}}.$$

 $C_{\tau}$  only depends on  $C_{\Phi}$ ,  $l_b$ ,  $\gamma$ ,  $\|h_0\|_{L^1_{2,v}}$ ,  $\tau$  and  $N_{\tau}$  (see (6.6.1)).

**Proof of Lemma** 6.6.2 We fix T in  $[0, T_0)$  and we consider t in [0, T]. We have the following decomposition

$$f(t,v) - g(t,v) = C_{\Phi} \int_0^t \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} b(\cos\theta) |v - v_*|^{\gamma} P(f',g') \, d\sigma dv_* ds$$
$$-C_{\Phi} \int_0^t \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} b(\cos\theta) |v - v_*|^{\gamma} P(f,g) \, d\sigma dv_* ds$$
$$= \int_0^t J_1(s,v) \, ds + \int_0^t J_2(s,v) \, ds,$$

where P is given by (6.6.4).

We have that at t = 0, f(0, v) - g(0, v) = 0 for all v and therefore

$$|f - g|(t, v) = \int_0^t \partial_t |f - g|(s, v) \, ds = \int_0^t \operatorname{sgn}(f - g)(s, v) \partial_t (f(s, v) - g(s, v)) \, ds$$
  

$$\leqslant \int_0^t |J_1|(s, v) \, ds + \int_0^t \operatorname{sgn}(f - g)(s, v) J_2(s, v) \, ds.$$
(6.6.13)

We start by the first term  $J_1$ . Using the algebraic equality (6.6.2) and the definition of P one can bound P(f', g') by

$$|P(f',g')| \leq (1+N_T) |f'-g'| (f'_*+g'_*) + \frac{(1+2N_T)}{2} |f'_*-g'_*| (f'+g') + \frac{1}{2} ||f-g||_{L^{\infty}_{[0,t],v}} (f'+g') (f'_*+g'_*).$$

The change of variable  $\sigma \to -\sigma$  sends v' to  $v'_*$  and reciprocally. Therefore we have

$$\int_{0}^{t} |J_{1}| \, ds \leq 2C_{\Phi}(1+2N_{T}) \int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \tilde{b}\left(\cos\theta\right) |v-v_{*}|^{\gamma} \left|f_{*}'-g_{*}'\right| \left(f'+g'\right) d\sigma dv_{*} ds \\ + \frac{C_{\Phi}}{2} \left\|f-g\right\|_{L_{[0,t],v}^{\infty}} \int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} b\left(\cos\theta\right) |v-v_{*}|^{\gamma} \left|f_{*}'+g_{*}'\right| \left(f'+g'\right) d\sigma dv_{*} ds,$$

where we defined  $\tilde{b}(x) = b(x) + b(-x)$ .

For both integrals, we use Lemma 6.4.4 with  $\lambda > 0$  to be chosen later. This yields

$$\int_{0}^{t} |J_{1}| \, ds \leqslant 4C_{\gamma}C_{\Phi}l_{b}(1+2N_{T}) \sup_{[0,t]} \left( \|f-g\|_{L^{1}_{2,v}} \right) \left[ 2C(1)N_{T}+2^{d-2}t \, \|h_{0}\|_{L^{1}_{2,v}} \right] \\ + \frac{1}{2}C_{\gamma}C_{\Phi}l_{b} \, \|f-g\|_{L^{\infty}_{[0,t],v}} \, 2 \, \|h_{0}\|_{L^{1}_{2,v}} \left[ 2C(\lambda)N_{T}+\frac{2^{d-2}}{\lambda^{d-1}}2t \, \|h_{0}\|_{L^{1}_{v}} \right].$$

We choose  $\lambda$  small enough such that

$$2C_{\gamma}C_{\Phi}l_b \|h_0\|_{L^1_{2,v}} C(\lambda)N_T \leqslant \frac{1}{4}, \qquad (6.6.14)$$

and we define  $\tau < T$  such that

$$C_{\gamma}C_{\Phi}l_b \|h_0\|_{L^1_{2,v}} \frac{2^{d-1}}{\lambda^{d-1}} \|h_0\|_{L^1_v} \tau \leqslant \frac{1}{4}.$$
(6.6.15)

These choices of constants lead to

$$\forall t \in [0,\tau], \quad \int_0^t |J_1| \ ds \leqslant \frac{1}{2} \|f - g\|_{L^{\infty}_{[0,t],v}} + C_{\tau} \sup_{[0,t]} \|f - g\|_{L^1_{2,v}}, \tag{6.6.16}$$

with  $C_{\tau}$  a constant depending on  $\tau$ .

We now turn to the last term  $J_2$  in (6.6.13). By using (6.6.2) and the change of variable  $\sigma \to -\sigma$  we get

$$\begin{split} \int_{0}^{t} \operatorname{sgn}(f-g)(s,v) J_{2}(s,v) \, ds \\ &\leqslant \frac{C_{\Phi}}{2} \int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} b \, |v-v_{*}|^{\gamma} \operatorname{sgn}(f-g)(g-f) \\ &\qquad \times \left(f_{*}(1+f'+f'_{*})+g_{*}(1+g'+g'_{*})\right) dv_{*} ds \\ &\qquad + \frac{C_{\Phi}}{4} (2+4N_{T}) l_{b} \int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v-v_{*}|^{\gamma} \, |f_{*}-g_{*}| \, (f+g) \, dv_{*} ds \\ &\qquad + \frac{C_{\Phi}}{4} \int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \tilde{b} \, |v-v_{*}|^{\gamma} \, (f+g)(f_{*}+g_{*}) \, |f'_{*}-g'_{*}| \, d\sigma dv_{*} ds, \end{split}$$
(6.6.17)

where we wrote  $\tilde{b}(x) = b(x) + b(-x)$ .

The second integral is easily dealt with and we have

$$\int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} |v - v_{*}|^{\gamma} |f_{*} - g_{*}| (f + g) \, dv_{*} ds \leq 2N_{T} C_{\gamma} \sup_{[0,t],v} ||f - g||_{L^{1}_{2,v}}$$
(6.6.18)

The third and last integral is a bit more intricate and we use the Carleman representation of the integral against  $(\sigma, v_*)$ . We emphasize that in the integral against  $(v', v'_*)$  we denote  $v_* = v'_* + v' - v$ . This yields

$$\int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \tilde{b} |v - v_{*}|^{\gamma} (f + g)(f_{*} + g_{*}) \left| f_{*}' - g_{*}' \right| \, d\sigma dv_{*} ds$$

$$\leq \int_{0}^{t} (g + f) \left( \int_{\mathbb{R}^{d}} \left| f_{*}' - g_{*}' \right| \left( 1 + \left| v_{*}' \right|^{\gamma} \right) \left[ \int_{\mathbb{R}^{d}} \frac{(1 + \left| v' \right|^{\gamma})}{\left| v - v' \right|^{d-1}} (f_{*} + g_{*}) \, dv' \right] \, dv_{*}' \right) \, ds$$

and we follow the idea developed in Lemma 6.4.4 with  $\lambda=1$ 

$$\begin{split} &\int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \tilde{b} \left| v - v_{*} \right|^{\gamma} (f+g) (f_{*} + g_{*}) \left| f_{*}' - g_{*}' \right| \, d\sigma dv_{*} ds \\ &\leqslant \int_{0}^{t} \left( 1 + \left| v \right|^{\gamma} \right) (g+f) \\ & \left( \int_{\mathbb{R}^{d}} \left| f_{*}' - g_{*}' \right| \left( 1 + \left| v_{*}' \right|^{\gamma} \right) \left[ 2N_{T} C(1) + 2^{d-1} \left\| h_{0} \right\|_{L^{1}_{2,v}} \left( 1 + \left| v_{*}' \right|^{\gamma} \right) \right] \, dv_{*}' \right) \, ds \end{split}$$

 $\gamma$  is in [0, 1] and thus we have

$$\int_{0}^{t} \int_{\mathbb{R}^{d} \times \mathbb{S}^{d-1}} \tilde{b} |v - v_{*}|^{\gamma} (f + g) (f_{*} + g_{*}) \left| f_{*}' - g_{*}' \right| \, d\sigma dv_{*} ds 
\leqslant 2C_{\gamma} N_{T} \left[ 2N_{T} C(1) + 2^{d-1} \left\| h_{0} \right\|_{L^{1}_{2,v}} \right] \sup_{[0,t],v} \left\| f - g \right\|_{L^{1}_{2,v}}$$
(6.6.19)

The first integral on the left-hand side of (6.6.17) is negative because f and g are both non-negative solutions. Thus we plug (6.6.18) and (6.6.19) in (6.6.17), which gives

$$\forall t \in [0,T], \quad \int_0^t \operatorname{sgn}(f-g)(s,v) J_2(s,v) \, ds \leqslant C_T \sup_{[0,t],v} \|f-g\|_{L^1_2,v}, \tag{6.6.20}$$

where  $C_T$  is a constant depending on T.

Therefore, if we take  $\tau$  given by (6.6.15), we can use (6.6.16) and (6.6.20) inside (6.6.13). This yields the result given in the statement of Lemma 6.6.4.

#### 6.6.4 Uniqueness result: proof of Theorem 6.6.1

In this section we prove the uniqueness result stated in Theorem 6.6.1.

We set  $\tau$  to be the minimum between  $\tau$  in Proposition 6.5.5 and  $\tau$  in Lemma 6.6.4. Throughout this section, C will stand for a positive generic constant depending only on  $\tau$ ,  $N_{\tau}$ , on the parameters of the operator Q and on  $\|h_0\|_{L^{\frac{1}{2}}_{row}}$ .

We use Lemma 6.6.2, Lemma 6.6.3 and Lemma 6.6.4 together to see that there exists  $\tau$  such that if t belongs to  $[0, \tau]$  then

$$\frac{d}{dt} \|f - g\|_{L^{1}_{v}} \leqslant C \left[ \|f - g\|_{L^{1}_{v}} + \|f - g\|_{L^{1}_{2,v}} \right] 
\frac{d}{dt} \|f - g\|_{L^{1}_{2,v}} \leqslant C \left[ M_{2+\gamma}(t) \|f - g\|_{L^{1}_{v}} + \|f - g\|_{L^{1}_{2,v}} + \|f - g\|_{L^{\infty}_{[0,T],v}} \right]$$

$$\|f - g\|_{L^{\infty}_{[0,t],v}} \leqslant C \sup_{[0,t],v} \|f - g\|_{L^{1}_{2,v}}.$$
(6.6.21)

The  $L_v^1$ ,  $L_{2,v}^1$  and  $L_{[0,t],v}^\infty$ -norms of f and g are bounded by assumption. Therefore, the first inequality in (6.6.21) gives  $||f - g||_{L_v^1} \leq Ct$ .

Moreover, Proposition 6.5.5 says that for t in  $(0, \tau]$ ,  $M_{2+\gamma}(t) \leq \frac{C_{\tau}}{t}$ , where  $C_{\tau}$  has been defined in Proposition 6.5.5 and therefore the second inequality in (6.6.21) gives  $||f - g||_{L^{1}_{2,v}} \leq Ct$ .

We can use these results to get  $||f - g||_{L^{\infty}_{[0,t],v}} \leq Ct$  in the third inequality in (6.6.21).

We can use this argument again to obtain that in fact  $||f - g||_{L_v^1} \leq Ct^2$ ,  $||f - g||_{L_{2,v}^1} \leq Ct^2$  and  $||f - g||_{L_{0,t],v}^{\infty}} \leq Ct^2$ . By induction we obtain that for all n in  $\mathbb{N}$ ,  $||f - g||_{L_v^1} \leq Ct^n$ ,  $||f - g||_{L_{2,v}^1} \leq Ct^n$  and  $||f - g||_{L_{[0,t],v}^{\infty}} \leq Ct^n$ .

**Remark 6.6.5** We emphasize here that one would like to take the limit as n goes to  $+\infty$  to obtain the uniqueness on short times. Unfortunately, C is a generic constant and we do not explicitly mentionned that this constant is increasing with n.

Therefore the three norms are time-differentiable at 0 with their time-derivatives being 0 at t = 0. Therefore we can use (6.6.21) for all t in  $[0, \tau]$  and combining the second with the third one we get

$$\forall T \in [0,\tau], \quad \frac{d}{dt} \|f - g\|_{L^{1}_{2,v}} \leqslant \frac{K_{1}}{t} \|f - g\|_{L^{1}_{2,v}} + K_{2} \sup_{[0,t],v} \|f - g\|_{L^{1}_{2,v}}, \quad (6.6.22)$$

where  $K_1, K_2 > 0$  only depend on  $\tau, h_0$  and the operator Q.

We fix  $n \ge K_1$  and we defined  $X(t) = \|f - g\|_{L^1_{2,v}}/t^n$ . We have that  $X(t) \le C_{\tau}t^2$  which means that X(t) is continuous at 0 and also right-differentiable at 0 with X'(0) = 0.

We differentiate X(t) in the same spirit that Nagumo's fixed point theorem. The main difference relies on the fact that we shall have to deal with terms of the form  $\sup X$  in the differential inequality. Thanks to (6.6.22) we have

$$\frac{d}{dt}X(t) = \frac{1}{t^n} \left( \frac{d}{dt} \|f - g\|_{L^1_{2,v}} - \frac{n}{t} \|f - g\|_{L^1_{2,v}} \right) \\
\leqslant \frac{K_2}{t^n} \sup_{[0,t],v} \|f - g\|_{L^1_{2,v}} \\
\leqslant K_2 \sup_{[0,t],v} X(s).$$

We integrate in time between 0 and t and because X(t) is positive we obtain

$$X(t) \leqslant tK_2 \sup_{[0,t],v} X(s)$$

and by induction we obtain

$$\forall t \in [0,\tau], \ \forall m \in \mathbb{N}, \quad X(t) \leq (tK_2)^m \sup_{[0,t],v} X(s).$$

Hence, we can take the limit as m goes to  $+\infty$  for all  $t < 1/K_2$ . Which means that

$$\forall t \in [0, \min(\tau, 1/K_2)], \quad X(t) = 0$$

and as a result, if we denote  $\tau_1 = \min(\tau, 1/K_2)$ ,

$$\forall t \in [0, \tau_1], \quad \|f - g\|_{L^1_{2,v}} = 0.$$
 (6.6.23)

To conclude for all time in  $[0, \tau]$  we know that for  $t \ge \tau_1$ ,  $M_{2+\gamma}(t)$  is bounded by  $C_{\tau}/\tau_1$ (see Proposition 6.5.5) and therefore (6.6.22) becomes

$$\forall T \in [\tau_1, \tau], \quad \frac{d}{dt} \| f - g \|_{L^1_{2,v}} \leqslant K_{\tau_1} \| f - g \|_{L^1_{2,v}} + K_2 \sup_{[0,t],v} \| f - g \|_{L^1_{2,v}},$$

which we can multiply by  $e^{-K_{\tau_1}t}$ , which is decreasing in t, to use an extended Gronwall lemma:

$$\forall T \in [\tau_1, \tau], \quad \frac{d}{dt} \left( e^{-K_{\tau_1} t} \| f - g \|_{L^1_{2,v}} \right)(t) \leqslant K_2 \sup_{[0,t],v} \left( e^{-K_{\tau_1} s} \| f - g \|_{L^1_{2,v}}(s) \right),$$

which once again gives uniqueness between  $[\tau_1, 2\tau_1]$  and by induction we obtain that

$$||f - g||_{L^1_{2,v}} = 0$$

on  $[0, \tau]$ .

Finally, the uniqueness on [0, T] is obtain by interating the latter proof starting from  $\tau$  to go up to  $2\tau$ . Indeed,  $\tau$  only depends on the operator Q,  $N_T$  and on  $||h_0||_{L^1_{2,v}}$  which is equal to  $||g||_{L^1_{2,v}}$  and  $||f||_{L^1_{2,v}}$  at time  $\tau$  since these two solutions preserves mass and energy. Therefore starting the proof at  $\tau$  will give us the uniqueness between  $\tau$  and  $2\tau$ . By induction we have that f = g on [0,T] for all  $0 \leq T < T_0$ .

### 6.7 Local existence of solutions

This section is dedicated to proving the following theorem

**Theorem 6.7.1** Let  $f_0(v)$  be in  $L^1_{2,v} \cap L^{\infty}_v$ .

Then there exists  $T_0 > 0$ , depending only on  $C_{\Phi}$ ,  $l_b$ ,  $\gamma$ ,  $\|f_0\|_{L^1_{2,v}}$  and  $\|f_0\|_{L^{\infty}_v}$ , such that there exists f in  $L^{\infty}_{loc}([0,T_0), L^1_{2,v} \cap L^{\infty}_v)$  solution on (6.1.3) on  $[0,T_0) \times \mathbb{R}^d$  such that

- $T_0 = +\infty$  or  $\lim_{T \to T_0^-} \|f\|_{L^\infty_{[0,T] \times \mathbb{R}^d}} = +\infty$ ,
- f preserves the mass, momentum and energy of  $f_0$ ,
- for all  $T < T_0$ ,

$$\sup_{[0,T]\times\mathbb{R}^d} \left( f(t,v) + \int_0^t \left(1 + |v|^\gamma\right) f(s,v) \, ds \right) < \infty.$$

For now on, we take  $f_0$ , non identically 0, in  $L^1_{2,v} \cap L^{\infty}_v$ .

The proof of this theorem relies on a time discretisation of equation (6.1.3) together with an approximation of the Boltzmann-Nordheim operator Q. This raises a sequence of functions  $(f_n)_{n \in \mathbb{N}}$  that will be proven to be approximations of a solution of the Boltzmann-Nordheim equation. This step is done by establishing the weak convergence of the sequence  $(f_n)_{n \in \mathbb{N}}$  to the unique solution of (6.1.3) (see Theorem 6.6.1).
We shall first derive some properties for truncated operators approximating the Boltzmann-Nordheim operator. Then we define some constants and construct a sequence of functions and finally show that this sequence convergences to a solution of equation (6.1.3).

### 6.7.1 Some properties of truncated operators

This idea of approximating the collision kernel in the case of hard potentials is a common one in the Boltzmann equation litterature (see for instance [2] or [77]). We consider now the following truncated operators, where n is a positive integer,

$$Q_n(f) = C_{\Phi} \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} \left( |v - v_*| \wedge n \right)^{\gamma} b \left[ f' f'_* (1 + f + f_*) - f f_* (1 + f' + f'_*) \right] dv_* d\sigma.$$

where  $a \wedge b = \min(a, b)$ .

We associate to these operator the natural decomposition (6.1.4) - (6.1.5):

$$Q_n(f) = -fQ_n^-(f) + Q_n^+(f).$$

The truncated operators are much easier to handle because they are easily bounded in  $L_{2,v}^1 \cap L_v^\infty$ , which is not the case for the full Boltzmann-Nordheim operator Q. Indeed, we have the following controls on the negative part

**Lemma 6.7.2** Let f be in  $L^1_{2,v} \cap L^{\infty}_v$ . Then we have the following inequalities

- $\|fQ_n^-(f)\|_{L^1_{2,v}} \leq C_{\Phi} l_b n^{\gamma} \left(1 + 2 \|f\|_{L^{\infty}_v}\right) \|f\|_{L^1_{2,v}}^2$ ,
- $\|Q_n^-(f)\|_{L^{\infty}_v} \leq C_{\Phi} l_b n^{\gamma} \left(1 + 2 \|f\|_{L^{\infty}_v}\right) \|f\|_{L^1_v},$
- if, moreover,  $f \ge 0$ , then there exists  $C_{\gamma} > 0$  (defined by (6.4.2)) such that

$$\forall v \in \mathbb{R}^d, \quad Q_n^-(f)(v) \ge C_\Phi l_b \left( n^\gamma \wedge (1+|v|^\gamma) \right) \|f\|_{L_v^1} - C_\Phi C_\gamma l_b \|f\|_{L_{2,v}^1}.$$

**Proof of Lemma 6.7.2** We have that

$$Q_n^{-}(f)(v) = C_{\Phi} \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} (n \wedge |v - v_*|)^{\gamma} b(\cos \theta) f_* \left[ 1 + f'_* + f' \right] dv_* d\sigma.$$

Therefore the first two inequalities are trivially obtained by bounding  $f'_* + f'$  by  $2 ||f||_{L^{\infty}_v}$ and the kernel by  $n^{\gamma}b(\cos\theta)$  and then taking the supremum in v or integrating against  $(1+|v|^2) dv$ . Now we suppose that f is positive and we copy the proof of Lemma 6.4.2, thus

$$\begin{aligned} Q_n^-(f)(v) &\geqslant C_{\Phi} \int_{\mathbb{R}^d \times \mathbb{S}^{d-1}} (n \wedge |v - v_*|)^{\gamma} b(\cos \theta) f_* \, dv_* d\sigma \\ &\geqslant C_{\Phi} l_b \left[ \int_{|v - v_*| \leqslant n} |v - v_*|^{\gamma} f_* \, dv_* + \int_{|v - v_*| \geqslant n} n^{\gamma} f_* \, dv_* \right]. \end{aligned}$$

Since  $\gamma$  is in [0, 1], we know that the following triangular inequality holds

$$(|v|^{\gamma} - |v_*|^{\gamma}) \leq |v - v_*|^{\gamma} \leq (|v|^{\gamma} + |v_*|^{\gamma}).$$

This yields

$$\begin{aligned} Q_n^-(f)(v) & \geqslant \quad C_{\Phi} l_b \left[ \int_{|v-v_*| \leqslant n} \left( (1+|v|^{\gamma}) - (1+|v_*|^{\gamma}) \right) f_* \, dv_* + \int_{|v-v_*| \geqslant n} n^{\gamma} f_* \, dv_* \right] \\ & \geqslant \quad C_{\Phi} l_b \left[ \left( n^{\gamma} \wedge (1+|v|^{\gamma}) \right) \|f\|_{L_v^1} - C_{\gamma} \int_{|v-v_*| \leqslant n} (1+|v_*|^2) f_* \, dv_* \right], \end{aligned}$$

by definition of  $C_{\gamma}$ , see (6.4.2). We obtained the expected lower bound.

Moreover we also have the following bounds on the positive part

**Lemma 6.7.3** Let f be in  $L_{2,v}^1 \cap L_v^\infty$ . Then we have the following inequalities

- $\|Q_n^+(f)\|_{L^1_{2,v}} \leq C_{\Phi} l_b n^{\gamma} \left(1 + 2 \|f\|_{L^{\infty}_v}\right) \|f\|^2_{L^1_{2,v}},$
- for all  $\lambda > 0$ ,

$$\begin{split} \left\|Q_{n}^{+}(f)\right\|_{L_{v}^{\infty}} \leqslant C_{\Phi} l_{b} C_{\gamma} \left\|f\right\|_{L_{2,v}^{1}} \left(1+2 \left\|f\right\|_{L_{v}^{\infty}}\right) \left[\int_{|u-v|\leqslant\lambda} \frac{\left(n^{\gamma} \wedge (1+|u|^{\gamma})\right) f(u)}{|u-v|^{d-1}} \, du + \frac{C_{\gamma}}{\lambda^{d-1}} \left\|f\right\|_{L_{2,v}^{1}}\right]. \end{split}$$

**Proof of Lemma 6.7.3** For the first inequality, we just have to notice that, after the change of variable  $(v', v'_*) \to (v, v_*)$  one obtains

$$\int_{\mathbb{R}^d} (1+|v|^{\gamma}) Q_n^+(f) \, dv = \int_{\mathbb{R}^d} (1+|v|^{\gamma}) f Q_n^-(f) \, dv,$$

and we deal with the  $L^1_{2,v}$ -norm the same way we did in proof of Lemma 6.7.2.

For the  $L_v^{\infty}$ -norm, we use exactly the same approach than in Lemma 6.4.4, using Carleman representation. This yields

$$\left|Q_{n}^{+}(f)(v)\right| \leq 2^{d-2} C_{\Phi} l_{b} \left[1+2 \left\|f\right\|_{L_{v}^{\infty}}\right] \int_{\mathbb{R}^{d} \times \mathbb{R}^{d}} \frac{\left(\left|v'-v'_{*}\right| \wedge n\right)^{\gamma}}{\left|v-v'\right|^{d-1}} f' f'_{*} \, dv' dv'_{*}.$$
(6.7.1)

We just have to deal with the integral whether  $(1 + |v|^{\gamma})$  is smaller than  $n^{\gamma}$  or not.

In the case where  $(1 + |v|^{\gamma}) \leq n^{\gamma}$  we bound the integral from above in the following way,

$$\int_{\mathbb{R}^{d} \times \mathbb{R}^{d}} \frac{\left(|v'-v'_{*}| \wedge n\right)^{\gamma}}{|v-v'|^{d-1}} f'f'_{*} dv' dv'_{*} \leqslant \int_{\mathbb{R}^{d} \times \mathbb{R}^{d}} \frac{|v'-v'_{*}|^{\gamma}}{|v-v'|^{d-1}} f'f'_{*} dv' dv'_{*} \\
\leqslant C_{\gamma} \|f\|_{L^{1}_{2,v}} \int_{\mathbb{R}^{d}} \frac{1+|v'|^{\gamma}}{|v-v'|^{d-1}} f' dv', \quad (6.7.2)$$

since  $\gamma$  is in [0, 1] and therefore, thanks to (6.4.2),

$$|v' - v'_*|^{\gamma} \leq (|v'|^{\gamma} + |v'_*|^{\gamma}) \leq C_{\gamma} (1 + |v'_*|^2) (1 + |v'|^{\gamma})$$

We obtain the expected result in the case  $(1 + |v|^{\gamma}) \leq n^{\gamma}$  by splitting the integral in (6.7.2) into small and large relative velocities  $v - v_*$ .

The case  $(1+|v|^{\gamma}) \ge n^{\gamma}$  is dealt with in exactly the same way but we bound  $(|v'-v'_*| \land n)$  by n in (6.7.2). This gives us the expected result.

### 6.7.2 Construction of a sequence of approximations

We now fix a positive integer n and we want to discretise in time the Boltzmann-Nordheim equation associated to the truncated operators  $Q_n$ . Thus we need to work on a closed interval. More precisely, we shall solve the truncated Boltzmann-Nordheim equation

$$\partial_t f_n = Q_n(f_n)$$

by a implicit Euler scheme on an interval  $[0, T_0]$ ,  $T_0$  not depending on n.

To this end, we require to fix some constants (like the ones appearing in Lemma 6.7.3) that we are going to define below.

First of all, in order to shorten notations, we define

$$C_L = C_\Phi l_b \, \|f_0\|_{L^1_u} \,, \tag{6.7.3}$$

$$C_0 = \frac{C_{\Phi} l_b C_{\gamma} \|f_0\|_{L^1_{2,v}}}{\min\left(1, C_L\right)}$$
(6.7.4)

where  $C_{\gamma}$  has been defined in (6.4.2). Then, we define

$$K_0 = \frac{2 \|f_0\|_{L_v^{\infty}}}{\min(1, C_L)}.$$
(6.7.5)

We have that  $|u|^{1-d}$  is integrable near 0 in  $\mathbb{R}^d$ . Therefore we can consider  $\lambda$  strictly positive such that

$$C(\lambda) = \int_{|u| \le \lambda} \frac{1}{|u|^{d-1}} \, du \le \frac{1}{2C_0 \left(1 + 2K_0\right)}.$$
(6.7.6)

Now we are able to define the time interval we shall work on,

$$T_0 = \frac{K_0}{4C_0} \left[ 1 + (1 + 2K_0) \frac{C_{\gamma}}{\lambda^{d-1}} \|f_0\|_{L^1_{2,v}} \right]^{-1}.$$
 (6.7.7)

We emphasize here that all the constants are independent of the integer n.

We consider the following explicit Euler scheme on  $[0, T_0]$ ,

$$\begin{cases} f_n^{(0)}(v) = f_0(v) \\ f_n^{(k+1)}(v) = f_n^{(k)}(v) \left(1 - \Delta_n Q_n^-\left(f_n^{(k)}\right)\right) + \Delta_n Q_n^+\left(f_n^{(k)}\right), & \text{for } k \in \left\{0, \dots, \frac{T_0}{\Delta_n}\right\}, \end{cases}$$
(6.7.8)

where  $Q_n^-$  and  $Q_n^+$  have been defined in (6.1.4) – (6.1.5).  $\Delta_n$  is the time step such that

$$\Delta_n = \min\left(1, \frac{1}{2C_{\Phi}l_b n^{\gamma} \|f_0\|_{L_v^1} [1 + 2K_0]}\right).$$
(6.7.9)

We first need to prove that the sequence  $\left(f_n^{(k)}\right)_{k \in \left\{0,\dots,\frac{T_0}{\Delta_n}\right\}}$  is well-defined. This is the purpose of the next proposition.

**Proposition 6.7.4** For all k in  $\{0, \ldots, T_0/\Delta_n\}$ , we have that  $f_n^{(k)}$ , see (6.7.8), is welldefined and

*i*) 
$$f_n^{(k)} \ge 0$$
,  
*ii*)  $\left\| f_n^{(k)} \right\|_{L_v^1} = \| f_0 \|_{L_v^1}, \left\| |v|^2 f_n^{(k)} \right\|_{L_v^1} = \left\| |v|^2 f_0 \right\|_{L_v^1} \text{ and } \int_{\mathbb{R}^d} v f_n^{(k)} \, dv = \int_{\mathbb{R}^d} v f_0 \, dv$ ,

- 292 -

iii)

$$\begin{aligned} f_n^{(k)}(v) &\leqslant f_0(v) - C_L \sum_{i=0}^{k-1} \Delta_n \left( n^{\gamma} \wedge (1+|v|^{\gamma}) \right) f_n^{(j)} \\ &+ k \Delta_n C_{\Phi} C_{\gamma} l_b \, \|f_0\|_{L^1_{2,v}} \left[ 1 + (1+2K_0) \, \frac{C_{\gamma}}{\lambda_1^{d-1}} \, \|f_0\|_{L^1_{2,v}} \right] \\ &+ C_{\Phi} l_b C_{\gamma} \, \|f_0\|_{L^1_{2,v}} \left( 1 + 2K_0 \right) \int_{|u-v| \leqslant \lambda} \frac{\sum_{i=0}^{k-1} \Delta_n \left( n^{\gamma} \wedge (1+|u|^{\gamma}) \right) f_n^{(i)}(u)}{|u-v|^{d-1}} \, du. \end{aligned}$$

iv) if we define

$$E_n(f_n^{(k)}) = \sup_{v \in \mathbb{R}^d} \left[ f_n^{(k)}(v) + \Delta_n \sum_{j=0}^{k-1} \left( n^{\gamma} \wedge (1+|v|^{\gamma}) \right) f_n^{(j)}(v) \right]$$

then

$$E_n(f_n^{(k)}) \leqslant K_0,$$

where  $C_{\gamma}$  has been defined in (6.4.2) and  $C_L$ ,  $\lambda$ ,  $K_0$ ,  $T_0$  and  $\Delta_n$  have been defined in (6.7.3) – (6.7.9).

**Remark 6.7.5** In particular, point iv) gives a uniform control on the  $L_v^{\infty}$ -norm of  $f_n^{(k)}$  which is bounded by  $K_0$ .

**Proof of Proposition** 6.7.4 The proof of the proposition above will be done by induction. The step k = 0 is direct since  $K_0 \ge ||f_0||_{L_v^{\infty}}$  (see (6.7.5)). So let us assume that this is true at rank k with  $k < T_0/\Delta_n$ .

Combining Lemma 6.7.2 and points ii) and iv) of Proposition 6.7.4 at rank k we have that

$$\Delta_n \left\| Q_n^- \left( f_n^{(k)} \right) \right\|_{L^{\infty}_v} \leqslant \Delta_n C_{\Phi} l_b n^{\gamma} \left\| f_0 \right\|_{L^1_v} (1 + 2K_0) \leqslant \frac{1}{2}.$$

Therefore we have that, by definition of  $f_n^{(k+1)}$ , (6.7.8),

$$f_n^{(k+1)}(v) \ge \frac{1}{2} f_n^{(k)}(v) + \Delta_n Q_n^+ \left( f_n^{(k)} \right)$$

and because  $f_n^{(k)} \ge 0$  we obtain that  $f_n^{(k+1)} \ge 0$ , which is *i*).

Furthermore,  $f_n^{(k+1)} \ge 0$  implies that, thanks to Definition (6.7.8),

$$\left\| \begin{pmatrix} 1\\v\\|v|^2 \end{pmatrix} f_n^{(k+1)} \right\|_{L_v^1} = \int_{\mathbb{R}^d} \begin{pmatrix} 1\\v\\|v|^2 \end{pmatrix} f_n^{(k+1)}(v) \, dv$$
$$= \int_{\mathbb{R}^d} \begin{pmatrix} 1\\v\\|v|^2 \end{pmatrix} f_n^{(k)}(v) \, dv + \Delta_n \int_{\mathbb{R}^d} \begin{pmatrix} 1\\v\\|v|^2 \end{pmatrix} Q_n(f_n^{(k)})(v) \, dv.$$

The last term on the right hand side is zero since  $Q_n$  satisfies the same integral property than the non-truncated Boltzmann-Nordheim operator, Lemma 6.1.1. Hence,  $f_n^{(k+1)}$ satisfies point *ii*).

In order to prove *iii*) we use Lemma 6.7.2 ( $f_n^{(k)}$  being positive), Lemma 6.7.3 and the fact that  $\left\|f_n^{(k)}\right\|_{L_n^{\infty}} \leq K_0$  in the definition of  $f_n^{(k+1)}$ , (6.7.8). This yields

$$\begin{aligned}
f_n^{(k+1)}(v) &\leqslant f_n^{(k)}(v) - C_L \Delta_n \left( n^{\gamma} \wedge (1+|v|^{\gamma}) \right) f_n^{(k)} \\
&+ \Delta_n C_{\Phi} C_{\gamma} l_b \left\| f_0 \right\|_{L^{1}_{2,v}} \left[ 1 + (1+2K_0) \frac{C_{\gamma}}{\lambda^{d-1}} \left\| f_0 \right\|_{L^{1}_{2,v}} \right] \\
&+ \Delta_n C_{\Phi} l_b C_{\gamma} \left\| f_0 \right\|_{L^{1}_{2,v}} \left( 1 + 2K_0 \right) \int_{|u-v| \leqslant \lambda} \frac{(n^{\gamma} \wedge (1+|u|^{\gamma})) f_n^{(k)}(u)}{|u-v|^{d-1}} \, du.
\end{aligned}$$

Then, by applying *iii*) for  $f_n^{(k)}$  we obtain *iii*) for  $f_n^{(k+1)}$ .

Thanks to iv) at rank k we have that for all v in  $\mathbb{R}^d$ 

$$\Delta_n \sum_{j=0}^{k-1} \left( n^{\gamma} \wedge (1+|v|^{\gamma}) \right) f_n^{(j)}(v) \leqslant k \Delta_n n^{\gamma} K_0.$$

Thus,  $\sup_{v \in \mathbb{R}^d} \Delta_n \sum_{j=0}^{k-1} \left( n^{\gamma} \wedge (1+|v|^{\gamma}) \right) f_n^{(j)}(v)$  exists and is finite.

Hence, we can consider property *iii*) at rank k + 1 and take the essential supremum over v in  $\mathbb{R}^d$ , noticing that  $k + 1 \leq T_0/\Delta_n$ ,

$$E_n(f_n^{(k+1)}) \leqslant \frac{K_0}{4} + T_0 C_0 \left[ 1 + (1 + 2K_0) \frac{C_{\gamma}}{\lambda^{d-1}} \|f_0\|_{L^{1}_{2,v}} \right]$$
$$C_0 C(\lambda) (1 + 2K_0) E(f_n^{(k+1)}),$$
$$\leqslant \frac{K_0}{2} + \frac{1}{2} E_n(f_n^{(k+1)}),$$

by definition of  $T_0$ , see (6.7.7) and of  $\lambda$ , see (6.7.6). This gives us the expected result iv)

for  $f_n^{(k+1)}$ .

## 6.7.3 Convergence towards a mass and momentum preserving solution of the Boltzmann-Nordheim equation

For each n in  $\mathbb{N}$  we have built a sequence of functions  $\left(f_n^{(k)}\right)_{k \in \{0,\dots,T_0/\Delta_n\}}$  in  $L^1_{2,v} \cap L^{\infty}_v$ . We shall see these functions as piecewise constant functions of time. Therefore we define

$$\forall n \in \mathbb{N}, \, \forall (t,v) \in [k\Delta_n, (k+1)\Delta_n) \times \mathbb{R}^d, \quad f_n(t,v) = f_n^{(k)}(v). \tag{6.7.10}$$

We are about to prove that  $(f_n)_{n\in\mathbb{N}}$  converges weakly in  $L^1([0,T_0]\times\mathbb{R}^d)$  towards f in  $L^1([0,T_0], L^1_{2,v}) \cap L^{\infty}([0,T_0]\times\mathbb{R}^d)$ , the mass and energy preserving solution of the Boltzmann-Nordheim equation (6.1.3) with initial data  $f_0$  (which is unique thanks to Theorem 6.6.1). This is the purpose of the following proposition.

**Proposition 6.7.6** There exists f in  $L^1([0, T_0] \times \mathbb{R}^d) \cap L^\infty([0, T_0] \times \mathbb{R}^d)$  such that a subsequence of  $(f_n)_{n \in \mathbb{N}}$ , see (6.7.10) and (6.7.8), converges towards f weakly in  $L^1([0, T_0] \times \mathbb{R}^d)$  and weakly-\* in  $L^\infty([0, T_0] \times \mathbb{R}^d)$ . Moreover, f satisfies

- f is a solution of the Boltzmann-Nordheim equation (6.1.3) with initial data  $f_0$ ,
- f is positive and for all t in  $[0, T_0]$ ,  $||f(t, \cdot)||_{L^1_v} = ||f_0||_{L^1_v}$  and  $\int_{\mathbb{R}^d} v f(t, v) dv = \int_{\mathbb{R}^d} v f_0(v) dv$ ,
- recalling the definition of  $K_0 > 0$  (see (6.7.5)), f satisfies

$$\sup_{[0,T_0]\times\mathbb{R}^d} \left( f(t,v) + \int_0^t \left(1 + |v|^\gamma\right) f(s,v) \, ds \right) \leqslant 2K_0.$$

**Proof of Proposition** 6.7.6 Thanks to point *ii*) of Proposition 6.7.4 we have that  $(f_n)_{n \in \mathbb{N}}$  is bounded in  $L^1([0, T_0] \times \mathbb{R}^d)$  but also have a uniform bound on its second moment. Therefore it is a tight sequence. Moreover, point *iv*) of Proposition 6.7.4 gives us that  $(f_n)_{n \in \mathbb{N}}$  is bounded in  $L^{\infty}([0, T_0] \times \mathbb{R}^d)$  and thus it is equi-integrable. The Dunford-Pettis theorem concludes that  $(f_n)_{n \in \mathbb{N}}$  is weakly compact in  $L^1([0, T_0] \times \mathbb{R}^d)$ .

Therefore, there exists f in  $L^1([0, T_0] \times \mathbb{R}^d)$  such that there exists a subsequence of  $(f_n)_{N \in \mathbb{N}}$ , that we will keep denoting by  $f_n$ , which converges weakly in  $L^1([0, T_0] \times \mathbb{R}^d)$  towards f.

Point i) of Proposition 6.7.4 tells us that  $f \ge 0$ . The sequence  $(f_n(t, \cdot))_{N \in \mathbb{N}}$  is tight and its tightness property is independent of the time t (see the uniform control of the second moment, point *ii*) of Proposition 6.7.4). Therefore, for all  $\epsilon > 0$ , it exists  $R_{\varepsilon} > 0$  such that for all n and t we have

$$\int_{\mathbb{R}^d} \mathbf{1}_{\{|v| \leqslant R_\varepsilon\}} f_n(t, v) \, dv \ge \|f_0\|_{L^1_v} - \varepsilon.$$

Since  $\mathbf{1}_{\{|v| \leq R_{\varepsilon}\}}$  is in  $L_v^{\infty}$  we can take the weak limit as n tends to  $+\infty$  in the inequality above to obtain

$$\|f(t,\cdot)\|_{L^1_v} \ge \int_{\mathbb{R}^d} \mathbf{1}_{\{|v| \le R_\varepsilon\}} f(t,v) \, dv \ge \|f_0\|_{L^1_v} - \varepsilon,$$

this being true for all  $\epsilon$ . Thus,  $\|f(t, \cdot)\|_{L^1_v} \ge \|f_0\|_{L^1_v}$ . But Fatou's Lemma offers us straightforwardly the opposite inequality.

This indicates that for all t in  $[0, T_0]$ ,  $||f(t, \cdot)||_{L_v^1} = ||f_0||_{L_v^1}$ . A similar argument proves that  $f(t, \cdot)$  has the same momentum as  $f_0$  for all  $t \ge 0$ .

The last point of Proposition 6.7.4, shows that  $(f_n)_{n \in \mathbb{N}}$  is bounded in  $L^{\infty}([0, T_0] \times \mathbb{R}^d)$ and therefore is weakly-\* compact in this space. We can extract a subsequence of  $f_n$ , still denoted by  $f_n$ , which converges weakly-\* in  $L^{\infty}([0, T_0] \times \mathbb{R}^d)$ . But since  $f_n$  converges weakly in  $L^1([0, T_0] \times \mathbb{R}^d)$  to f and therefore the weak-\* limit in  $L^{\infty}([0, T_0] \times \mathbb{R}^d)$  can only be f.

Thus f belongs to  $L^{\infty}([0, T_0] \times \mathbb{R}^d)$ .

Thanks to point iv) of Proposition 6.7.4, we have, for all n in  $\mathbb{N}$  and k in  $\{0, \ldots, T_0/\Delta_n - 1\}$ , for all (t, v) in  $[k\Delta_n, (k+1)\Delta_n) \times \mathbb{R}^d$ ,

$$f_{n}(t,v) + \int_{0}^{t} (n^{\gamma} \wedge (1+|v|^{\gamma})) f_{n}(s,v) \, ds = f_{n}^{(k)}(v) + \sum_{j=0}^{k-1} \Delta_{n} (n^{\gamma} \wedge (1+|v|^{\gamma})) f_{n}^{(j)}(v) + \Delta_{n} (t-k\Delta_{n}) (n^{\gamma} \wedge (1+|v|^{\gamma})) f_{n}^{(k)}(v) \leqslant K_{0} + \Delta_{n}^{2} n^{\gamma} K_{0} \leqslant 2K_{0}.$$
(6.7.11)

Therefore, if we define

$$\forall n \in \mathbb{N}, \,\forall (t,v) \in [0,T_0] \times \mathbb{R}^d, \quad g_n(t,v) = \int_0^t \left(n^\gamma \wedge (1+|v|^\gamma)\right) f_n(s,v) \, ds, \qquad (6.7.12)$$

we have that the sequence  $(g_n)_{n \in \mathbb{N}}$  is bounded in  $L^{\infty}([0, T_0] \times \mathbb{R}^d)$  and therefore is weakly-\* compact. There exists a subsequence, still denoted by  $g_n$ , that converges weakly-\* in  $L^{\infty}([0, T_0] \times \mathbb{R}^d)$  to, say, g.

Besides, since  $0 \leq \gamma \leq 1$ , we have that  $(g_n)_{n \in \mathbb{N}}$  is bounded in  $L^1([0, T_0] \times \mathbb{R}^d)$  and such

that for all t in  $[0, T_0]$ ,

$$\int_{\mathbb{R}^d} |v|^{2-\gamma} g_n(t,v) \, dv \leqslant 2C_{\gamma} T_0 \, \|f_0\|_{L^1_{2,v}} \,,$$

where  $C_{\gamma}$  has been defined in (6.4.2). Therefore  $(g_n)_{n \in \mathbb{N}}$  is tight and equi-integrable and therefore is weakly compact in  $L^1([0, T_0] \times \mathbb{R}^d)$ . As we did for  $f_n$  we obtain that  $g_n$ converges (up to a subsequence) weakly to g in  $L^1([0, T_0] \times \mathbb{R}^d)$ .

We are going to prove that

$$g(t,v) = \int_0^t (1+|v|^{\gamma})f(s,v) \, ds. \tag{6.7.13}$$

As we emphasised before, the compactness properties of  $f_n$  and  $g_n$  are the same for  $f_n(t, \cdot)$ and  $g_n(t, \cdot)$  for a given t, because our bounds are independent of t. Therefore we fix a t in  $[0, T_0]$  and we take  $\phi$  in  $C_c^{\infty}(\mathbb{R}^d)$ . By weak convergence of  $f_n$  in  $L^1([0, T_0] \times \mathbb{R}^d)$  we have

$$\int_{\mathbb{R}^d} \int_0^t \left(1 + |v|^\gamma\right) \phi(v) f(s, v) \, ds dv = \lim_{n \to \infty} \int_{\mathbb{R}^d} \int_0^t \left(1 + |v|^\gamma\right) \phi(v) f_n(t, v) \, ds dv = \lim_{n \to \infty} I_n.$$

But we have the following

$$I_n = \int_{(1+|v|^{\gamma}) \leq n^{\gamma}} \phi(v) g_n(t,v) \, dv + \int_0^t \int_{(1+|v|^{\gamma}) > n^{\gamma}} \left( n^{\gamma} \wedge (1+|v|^{\gamma}) \right) \phi(v) f_n(s,v) \, dv ds.$$

 $\phi$  is of compact support so for n big enough we have that

$$\forall s \in [0,t], \quad \int_{(1+|v|^{\gamma}) > n^{\gamma}} \left( n^{\gamma} \wedge (1+|v|^{\gamma}) \right) \phi(v) f_n(s,v) \, dv = 0$$

Finally, by the weak convergence of  $g_n$  in  $L_v^1$  we obtain

$$\lim_{n \to \infty} I_n = \int_{\mathbb{R}^d} g(t, v) \phi(v) \, dv.$$

Thus,

$$\forall \phi \in C_c^{\infty}(\mathbb{R}^d), \quad \int_{\mathbb{R}^d} \phi(v) \left[ g(t,v) - \int_0^t \left( 1 + |v|^{\gamma} \right) f(s,v) \, ds \right] \, dv = 0.$$

This gives us the expected equality (6.7.13), since both functions are in  $L_v^1$ .

As a result, we have that  $\int_0^t (1+|v|^{\gamma}) f(s,v) ds$  is in  $L^{\infty}([0,T_0] \times \mathbb{R}^d)$ . Thanks to (6.7.11) we also find

$$\sup_{[0,T_0]\times\mathbb{R}^d} \left( f(t,v) + \int_0^t \left(1 + |v|^\gamma\right) f(s,v) \, ds \right) \leqslant 2K_0. \tag{6.7.14}$$

To conclude the proof of Proposition 6.7.6 it remains to show that f is a solution of the Boltzmann-Nordheim equation (6.1.3). However, this is now pretty straightforward since  $g_n(t, \cdot)$  converges weakly-\* in  $L_v^{\infty}$  and  $f_n(t, \cdot)$  converges weakly in  $L_v^1$ , for all t. Therefore, thanks to the definition of  $f_n^{(k)}$ , see (6.7.8), we can take the limit in

$$\int_{\mathbb{R}^d} f_n(t,v)\phi(v) \, dv = \int_{\mathbb{R}^d} \phi(v) \left[ f_0(v) + \int_0^t Q_n(f_n(s,\cdot))(v) \right] \, ds dv,$$

for all test functions  $\phi$  in  $C_c^{\infty}(\mathbb{R}^d)$ . Indeed,  $Q^n$  is basically a convolution operator with a kernel not growing faster than  $(n^{\gamma} \wedge (1+|v|^{\gamma}))$ . Then, since we have the equality (6.7.13), we obtain that for all test function  $\phi$ 

$$\int_{\mathbb{R}^d} \phi(v) \left[ f(t,v) - \left( f_0(v) + \int_0^t Q(f(s,\cdot))(v) \, ds \right) \right] = 0,$$

and thanks to (6.7.14) we have that  $f(t,v) - \left(f_0(v) + \int_0^t Q(f(s,\cdot))(v) \, ds\right)$  belongs to  $L_v^1 \cap L_v^\infty$  and therefore we obtain

$$f(t,v) = f_0(v) + \int_0^t Q(f(s,\cdot))(v) \, ds,$$

which means that f is a solution of the Boltzmann-Nordheim equation (6.1.3).

## 6.7.4 Preservation of the energy

This section is devoted to the proof of the following result, which is the fact that f preserves the energy of the initial data.

**Proposition 6.7.7** Let f be the function obtained in Proposition 6.7.6.

Then for all t in  $[0, T_0]$ ,

$$\left\| |v|^2 f(t,v) \right\|_{L^1_v} = \left\| |v|^2 f_0(v) \right\|_{L^1_v}.$$

**Proof of Proposition** 6.7.7 We have, thanks to Proposition 6.7.4, that for all t in  $[0, T_0]$  the sequence  $(|v|^2 f_n(t, v))_{n \in \mathbb{N}}$  is bounded in  $L_v^1$  with the following preservation of the  $L_v^1$ -norm,

$$\forall n \in \mathbb{N}, \forall t \in [0, T_0], \quad \int_{\mathbb{R}^d} |v|^2 f_n(t, v) \, dv = \left\| |v|^2 f_0 \right\|_{L^1_v}$$

Therefore, we fix t and notice that for all R > 0 we have that, since  $f_n \ge 0$ ,

$$\int_{\mathbb{R}^d} \mathbf{1}_{\{|v| \le R\}} |v|^2 f_n(t, v) \, dv \le \left\| |v|^2 f_0 \right\|_{L^1_v}.$$
(6.7.15)

As we saw in the proof of Proposition 6.7.6,  $(f_n(t, \cdot))_{n \in \mathbb{N}}$  converges weakly (up to a subsequence) in  $L_v^1$  towards f. We can then take the limit as n in (6.7.15),

$$\int_{\mathbb{R}^d} \mathbf{1}_{\{|v| \leq R\}} |v|^2 f(t, v) \, dv \leq \left\| |v|^2 f_0 \right\|_{L^1_v},$$

which is true for all R. Thus, the positivity of f yields

$$\int_{\mathbb{R}^d} |v|^2 f(t,v) \, dv \leqslant \|f_0\|_{L^1_v} \,. \tag{6.7.16}$$

It remains to prove the opposite inequality.

To this end we shall show that  $(|v|^2 f_n(t, v))_{n \in \mathbb{N}}$  is tight in  $L_v^1$ , uniformly in t. Indeed, such a tighness property will yiels

$$\forall \varepsilon > 0, \; \exists R_{\varepsilon} > 0, \; \forall n, t, \quad \int_{\mathbb{R}^d} \mathbf{1}_{\{|v| \leqslant R_{\varepsilon}\}} \left| v \right|^2 f_n(t, v) \; dv \ge \left\| \left| v \right|^2 f_0 \right\|_{L^1_v} - \varepsilon$$

and since  $\mathbf{1}_{\{|v| \leq R_{\varepsilon}\}}$  is in  $L_v^{\infty}$  we can take the weak limit as n tends to  $+\infty$  in the inequality above to obtain (remember that f is positive)

$$\left\| |v|^2 f(t,v) \right\|_{L^1_v} \ge \int_{\mathbb{R}^d} \mathbf{1}_{\{|v| \le R_\varepsilon\}} |v|^2 f(t,v) \, dv \ge \left\| |v|^2 f_0 \right\|_{L^1_v} - \varepsilon,$$

this being true for all  $\epsilon$ .

The tightness of our sequence of approximation is dealt with below.  $\blacksquare$ 

We have that  $f_0$  is positive and such that  $(1 + |v|^2) f_0(v)$  is in  $L_v^1$ . Proposition A1 in the appendix of [77] gives the existence of  $\psi$  a positive convex function on  $\mathbb{R}^+$  such that there exists C > 0 such that

$$\int_{\mathbb{R}^d} \psi\left(\left|v\right|^2\right) f_0(v) \, dv \leqslant C.$$

Moreover,  $\psi$  can be written  $\psi(x) = x\phi(x)$ , where  $\phi$  is concave, increasing to infinity, and such that for any  $\varepsilon > 0$  and any  $\alpha$  in (0, 1), it satisfies  $(\phi(x) - \phi(\alpha x)) x^{\varepsilon} \to \infty$  as  $x \to \infty$ .

To prove that  $(|v|^2 f_n(t, v))_{n \in \mathbb{N}}$  is tight in  $L_v^1$  we are going to need the technical lemma about Povner-type inequality, Lemma 6.3.1.

The tightness of  $(|v|^2 f_n(t, v))_{n \in \mathbb{N}}$  directly follows from the following control of the tail of the distribution  $f_n(t, v)$ .

**Proposition 6.7.8** For all n in  $\mathbb{N}$ , for all k in  $\{0, \ldots, T_0/\Delta_n\}$ ,

$$\int_{\mathbb{R}^d} f_n^{(k)}(v)\psi\left(|v|^2\right) \, dv \leqslant \int_{\mathbb{R}^d} \psi\left(|v|^2\right) f_0(v) + \frac{1}{2}(k+1)\Delta_n C_G C_\gamma^2 \, \|f_0\|_{L^1_{2,v}}^2 \,,$$

where  $C_G$  has been defined in Lemma 6.3.1.

**Remark 6.7.9** We would like to emphasize here that  $C_G$  depends on b,  $\psi$  (which depends only on  $f_0$ ) and an upper bound for  $\left\|f_n^{(k)}\right\|_{L^{\infty}_v}$  which is bounded by  $K_0$  (see point iv) of Proposition 6.7.4). Therefore,  $C_G$  is a constant of our problem, independent of k and n.

**Proof of Proposition** 6.7.8 The proof will be done by induction. The case k = 0 is obvious so let us assume that this is true up to rank  $k < T_0/\Delta_n$ . To shorten computation we set

$$M_n^{(k)} = \int_{\mathbb{R}^d} f_n^{(k)}(v)\psi\left(|v|^2\right) dv$$

By definition of  $f_n^{(k+1)}$ , see (6.7.8), we obtain, after integrating in v and the use of the usual changes of variables  $(v, v_*) \to (v_*, v)$  and  $(v, v_*) \to (v', v'_*)$ ,

$$\begin{split} M_n^{(k+1)} &= M_n^{(k)} + \Delta_n \int_{\mathbb{R}^d} Q_n(f_n^{(k)})(v) \, dv \\ &= M_n^{(k)} + \frac{\Delta_n}{2} \int_{\mathbb{R}^d \times \mathbb{R}^d} (n \wedge |v - v_*|)^{\gamma} \, f_n^{(k)}(v) f_n^{(k)}(v_*) \\ & \times \left[ \int_{\mathbb{S}^{d-1}} \left[ 1 + f_n^{(k)}(v') + f_n^{(k)}(v'_*) \right] b(\cos \theta) \left( \psi'_* + \psi' - \psi_* - \psi \right) d\sigma \right] dv_* dv \\ &= M_n^{(k)} + \frac{\Delta_n}{2} \int_{\mathbb{R}^d \times \mathbb{R}^d} (n \wedge |v - v_*|)^{\gamma} \, f_n^{(k)}(v) f_n^{(k)}(v_*) \left[ G(v, v_*) - H(v, v_*) \right] dv_* dv. \end{split}$$
(6.7.17)

We can use Lemma 6.3.1 with

$$G(v, v_*) \leqslant C_G |v| |v_*|,$$
  

$$H(v, v_*) \geqslant 0.$$

This yields, applied to (6.7.17) because  $f_n^{(k)}$  is positive (see Proposition 6.7.4),

$$\begin{aligned} M_n^{(k+1)} &\leqslant M_n^{(k)} + \frac{\Delta_n}{2} C_G \int_{\mathbb{R}^d \times \mathbb{R}^d} |v - v_*|^{\gamma} |v| \, |v_*| \, f_n^{(k)}(v) f_n^{(k)}(v_*) \, dv_* dv \\ &\leqslant M_n^{(k)} + \frac{\Delta_n}{2} C_G \left[ \int_{\mathbb{R}^d} (1 + |v|^{\gamma}) \, |v| \, f_n^{(k)}(v) \, dv \right]^2.
\end{aligned}$$

Because  $\gamma$  belongs to [0, 1] we also have (up to a change of constant  $C_{\gamma}$  in (6.4.2))

$$(1+|v|^{\gamma})|v| \leq C_{\gamma} (1+|v|^2),$$

which yields, thanks to the preservation of the  $L_{2,v}^1$ -norm (see Proposition 6.7.4),

$$M_n^{(k+1)} \leqslant M_n^{(k)} + \frac{\Delta_n}{2} C_G C_\gamma^2 \|f_0\|_{L^1_{2,v}}^2$$

Applying the induction hypothesis at rank k gives us the expected result.

The tightness of  $(f_n(t, \cdot))_{n \in \mathbb{N}}$  follows straightforwardly from the growing property of  $\psi$ and Proposition 6.7.8 which states the following control, uniform in n and t,

$$\int_{\mathbb{R}^d} f_n(t,v)\psi\left(|v|^2\right) \, dv \leqslant \int_{\mathbb{R}^d} f_0(v)\psi\left(|v|^2\right) \, dv + \frac{T_0}{2} C_G C_{\gamma}^2 \, \|f_0\|_{L^1_{2,v}}^2 < \infty,$$

since  $k \leq T_0/\Delta_n$  and  $C_G$  is a constant (see Remark 6.7.9).

This concludes the proof of Theorem 6.7.1 since f is a positive solution of (6.1.3), mass and energy preserving and in  $L^{\infty}_{\text{loc}}([0, T_0), L^1_{2,v} \cap L^{\infty}_v)$ . Therefore, see Theorem 6.6.1, fis the unique solution satisfying those properties and f preserves the momentum. Thus the sequence  $(f_n)_{n \in \mathbb{N}}$  converges weakly in Proposition 6.7.6, not only just a subsequence, towards f.

If at  $T_0$  we have that  $||f||_{L^{\infty}_{t,v}} \leq M$  then we can apply our proof starting at  $T_0$  and construct a solution up to  $T_1$  (depending only on M, which depends only on  $f_0$ , and  $||f(T_0, \cdot)||_{L^1_{2,v}} = ||f_0||_{L^1_{2,v}}$ ) and by uniqueness we have in fact extended f to  $T_1$ . We can inductively build a solution on [0, T) up to the point where

$$\lim_{T\to T^-}\|f\|_{L^\infty_{[0,T]\times\mathbb{R}^d}}=+\infty.$$

# Bibliography

- ALEXANDRE, R., MORIMOTO, Y., UKAI, S., XU, C.-J., AND YANG, T. The Boltzmann equation without angular cutoff in the whole space: II, Global existence for hard potential. *Anal. Appl. (Singap.)* 9, 2 (2011), 113–134. 147
- [2] ARKERYD, L. On the boltzmann equation. Arch. Rational Mech. Anal (1972), 1–34.
   291
- [3] ARKERYD, L. Stability in L<sup>1</sup> for the spatially homogeneous Boltzmann equation. Arch. Rational Mech. Anal. 103, 2 (1988), 151–167. 246
- [4] BARANGER, C., AND MOUHOT, C. Explicit spectral gap estimates for the linearized Boltzmann and Landau operators with hard potentials. *Rev. Mat. Iberoamericana* 21, 3 (2005), 819–841. 35, 189, 193, 223, 232, 243
- [5] BARDOS, C., GOLSE, F., AND LEVERMORE, C. D. Fluid dynamic limits of kinetic equations. II. Convergence proofs for the Boltzmann equation. *Comm. Pure Appl. Math.* 46, 5 (1993), 667–753. 138
- [6] BARDOS, C., GOLSE, F., AND LEVERMORE, C. D. Acoustic and Stokes limits for the Boltzmann equation. C. R. Acad. Sci. Paris Sér. I Math. 327, 3 (1998), 323–328. 136
- [7] BARDOS, C., GOLSE, F., AND LEVERMORE, C. D. The acoustic limit for the Boltzmann equation. Arch. Ration. Mech. Anal. 153, 3 (2000), 177–204. 136
- [8] BARDOS, C., GOLSE, F., AND LEVERMORE, D. Fluid dynamic limits of kinetic equations. I. Formal derivations. J. Statist. Phys. 63, 1-2 (1991), 323–344. 38, 133, 134, 135, 138, 155, 222, 223
- [9] BARDOS, C., GOLSE, F., AND PAILLARD, L. The incompressible Euler limit of the Boltzmann equation with accommodation boundary condition. *Commun. Math. Sci.* 10, 1 (2012), 159–190. 130
- [10] BARDOS, C., AND UKAI, S. The classical incompressible Navier-Stokes limit of the Boltzmann equation. *Math. Models Methods Appl. Sci.* 1, 2 (1991), 235–257. 39, 138, 146, 147, 148, 155, 156, 209, 212, 223

- [11] BARRÉ DE SAINT-VENANT, A. Mémoire sur la dynamique des fluides. 1834. 123
- [12] BATCHELOR, G. K. An introduction to fluid dynamics, paperback ed. Cambridge Mathematical Library. Cambridge University Press, Cambridge, 1999. 123
- [13] BERGH, J., AND LÖFSTRÖM, J. Interpolation spaces. An introduction. Springer-Verlag, Berlin-New York, 1976. Grundlehren der Mathematischen Wissenschaften, No. 223. 240
- [14] BOBYLËV, A. V. The method of the Fourier transform in the theory of the Boltzmann equation for Maxwell molecules. *Dokl. Akad. Nauk SSSR 225*, 6 (1975), 1041– 1044. 223
- BOBYLËV, A. V. The theory of the nonlinear spatially uniform Boltzmann equation for Maxwell molecules. In *Mathematical physics reviews, Vol. 7*, vol. 7 of *Soviet Sci. Rev. Sect. C Math. Phys. Rev.* Harwood Academic Publ., Chur, 1988, pp. 111–233.
   223
- [16] BOGOLIUBOV, N. Problems of Dynamical Theory in Statistical Physics. Studies in Statistical Mechanics, New York, 1962. 128
- [17] BOLTZMANN, L. Lectures on gas theory. Translated by Stephen G. Brush from the 1896-1898 edition. University of California Press, Berkeley, 1964. 123
- [18] BOLTZMANN, L. Weitere Studien über das Wärme gleichgenicht unfer Gasmoläkuler. Translation "Further Studies on the Thermal Equilibrium of Gas Molecules". Reprinted in The Kinetic Theory of Gases of the 1872 edition. 123
- [19] BORN, M., AND GREEN, H. S. A general kinetic theory of liquids I. The molecular distribution functions. Proc R Soc Med 188, 1012 Ser A (1946), 10–18. 128
- [20] BOUCHUT, F., AND DESVILLETTES, L. Averaging lemmas without time Fourier transform and application to discretized kinetic equations. Proc. Roy. Soc. Edinburgh Sect. A 129, 1 (1999), 19–36. 241
- [21] BOUDIN, L., AND DESVILLETTES, L. On the singularities of the global small solutions of the full Boltzmann equation. *Monatsh. Math.* 131, 2 (2000), 91–108. 241
- [22] BREZIS, H. Analyse fonctionnelle. Collection Mathématiques Appliquées pour la Maîtrise. [Collection of Applied Mathematics for the Master's Degree]. Masson, Paris, 1983. Théorie et applications. [Theory and applications]. 192, 242
- [23] BRIANT, M. A constructive method from Boltzmann to incompressible Navier-Stokes on the torus. Preprint. 41, 223, 224, 225, 227, 229, 231, 232, 243, 245, 247, 248, 249

- [24] CÁCERES, M. J., CARRILLO, J. A., AND GOUDON, T. Equilibration rate for the linear inhomogeneous relaxation-time Boltzmann equation for charged particles. *Comm. Partial Differential Equations 28*, 5-6 (2003), 969–989. 189
- [25] CAFLISCH, R. E. The fluid dynamic limit of the nonlinear Boltzmann equation. Comm. Pure Appl. Math. 33, 5 (1980), 651–666. 133
- [26] CARLEMAN, T. Sur la théorie de l'équation intégrodifférentielle de Boltzmann. Acta Math. 60, 1 (1933), 91–146. 27, 56, 58, 263
- [27] CARLEMAN, T. Problèmes mathématiques dans la théorie cinétique des gaz. Publ.
   Sci. Inst. Mittag-Leffler. 2. Almqvist & Wiksells Boktryckeri Ab, Uppsala, 1957. 21, 223, 234, 273
- [28] CERCIGNANI, C. The Boltzmann equation and its applications, vol. 67 of Applied Mathematical Sciences. Springer-Verlag, New York, 1988. 15, 18, 19, 21, 54, 62, 143, 221, 227, 259
- [29] CERCIGNANI, C. Chapter 1 the boltzmann equation and fluid dynamics. vol. 1 of Handbook of Mathematical Fluid Dynamics. North-Holland, 2002, pp. 1 – 69. 43
- [30] CERCIGNANI, C., ILLNER, R., AND PULVIRENTI, M. The mathematical theory of dilute gases, vol. 106 of Applied Mathematical Sciences. Springer-Verlag, New York, 1994. 15, 18, 24, 54, 62, 126, 128, 143, 154, 191, 194, 221, 227, 259
- [31] CHAI, X. The Boltzmann equation near Maxwellian in the whole space. Commun. Pure Appl. Anal. 10, 2 (2011), 435–458. 147
- [32] CHAPMAN, S., AND COWLING, T. G. The mathematical theory of nonuniform gases, third ed. Cambridge Mathematical Library. Cambridge University Press, Cambridge, 1990. An account of the kinetic theory of viscosity, thermal conduction and diffusion in gases, In co-operation with D. Burnett, With a foreword by Carlo Cercignani. 15, 43, 259, 262
- [33] CHEN, J., AND YANG, M. Z. Linear transport equation with specular reflection boundary condition. *Transport Theory Statist. Phys.* 20, 4 (1991), 281–306. 28, 59, 69, 71
- [34] DE MASI, A., ESPOSITO, R., AND LEBOWITZ, J. L. Incompressible Navier-Stokes and Euler limits of the Boltzmann equation. *Comm. Pure Appl. Math.* 42, 8 (1989), 1189–1214. 135
- [35] DESVILLETTES, L., AND VILLANI, C. On the spatially homogeneous landau equation for hard potentials part ii : h-theorem and applications. *Communications in Partial Differential Equations 25*, 1-2 (2000), 261–298. 27

- [36] DESVILLETTES, L., AND VILLANI, C. On the trend to global equilibrium in spatially inhomogeneous entropy-dissipating systems: the linear Fokker-Planck equation. Comm. Pure Appl. Math. 54, 1 (2001), 1–42. 27, 57
- [37] DESVILLETTES, L., AND VILLANI, C. On the trend to global equilibrium for spatially inhomogeneous kinetic systems: the Boltzmann equation. *Invent. Math. 159*, 2 (2005), 245–316. 27, 57
- [38] DIPERNA, R. J., AND LIONS, P.-L. On the Cauchy problem for Boltzmann equations: global existence and weak stability. Ann. of Math. (2) 130, 2 (1989), 321–366. 138
- [39] ELLIS, R. S., AND PINSKY, M. A. The first and second fluid approximations to the linearized Boltzmann equation. J. Math. Pures Appl. (9) 54 (1975), 125–156.
  39, 138, 148, 183, 209
- [40] ESCOBEDO, M., AND VELÁZQUEZ, J. J. Finite time blow-up for the bosonic Nordheim equation. Preprint. 44, 45, 46, 260, 262, 263, 266, 271
- [41] ESCOBEDO, M., AND VELÁZQUEZ, J. J. On the blow-up of supercritical solutions of the nordheim equation for bosons. Preprint. 45, 262, 263
- [42] ESPOSITO, R., MARRA, R., AND YAU, H. T. Navier-Stokes equations for stochastic particle systems on the lattice. *Comm. Math. Phys.* 182, 2 (1996), 395–456. 125
- [43] EULER, L. General laws of the motion of fluids. Izv. Ross. Akad. Nauk Mekh. Zhidk. Gaza, 6 (1999), 26–54. 122
- [44] GALLAGHER, I., SAINT-RAYMOND, L., AND TEXIER, B. From newton to boltzmann: the case of short-range potentials. Preprint. 19, 126, 128, 129, 130, 143, 221, 259
- [45] GIBBS, J. W. The scientific papers of J. Willard Gibbs. Vol. II: Dynamics, vector analysis and multiple algebra, electromagnetic theory of light, etc. Dover Publications Inc., New York, 1961. 122
- [46] GOLSE, F. From kinetic to macroscopic models, 1998. 25, 26, 31, 32, 130, 131, 134, 138, 143, 144, 146, 191, 221, 222, 223
- [47] GOLSE, F. The mean-field limit for the dynamics of large particle systems. In Journées "Équations aux Dérivées Partielles". Univ. Nantes, Nantes, 2003, pp. Exp. No. IX, 47. 127

- [48] GRAD, H. Principles of the kinetic theory of gases. In Handbuch der Physik (herausgegeben von S. Flügge), Bd. 12, Thermodynamik der Gase. Springer-Verlag, Berlin, 1958, pp. 205–294. 22, 40, 55, 133, 135, 144, 222, 223, 260
- [49] GRAD, H. Asymptotic theory of the Boltzmann equation. II. In Rarefied Gas Dynamics (Proc. 3rd Internat. Sympos., Palais de l'UNESCO, Paris, 1962), Vol. I. Academic Press, New York, 1963, pp. 26–59. 133, 223
- [50] GRAD, H. Asymptotic equivalence of the Navier-Stokes and nonlinear Boltzmann equations. In Proc. Sympos. Appl. Math., Vol. XVII. Amer. Math. Soc., Providence, R.I., 1965, pp. 154–183. 147, 224
- [51] GUALDANI, M. P., MISCHLER, S., AND MOUHOT, C. Factorization for nonsymmetric operators and exponential H-theorem. 40, 41, 42, 138, 223, 224, 225, 226, 227, 229, 231, 232, 234, 235, 237, 238, 239, 241, 243, 245, 246
- [52] GUO, Y. Singular solutions of the Vlasov-Maxwell system on a half line. Arch. Rational Mech. Anal. 131, 3 (1995), 241–304. 28, 59, 69, 71
- [53] GUO, Y. The Landau equation in a periodic box. Comm. Math. Phys. 231, 3 (2002), 391–434. 147, 189, 193, 194
- [54] GUO, Y. The Vlasov-Poisson-Boltzmann system near Maxwellians. Comm. Pure Appl. Math. 55, 9 (2002), 1104–1135. 189
- [55] GUO, Y. Classical solutions to the Boltzmann equation for molecules with an angular cutoff. Arch. Ration. Mech. Anal. 169, 4 (2003), 305–353. 147, 224
- [56] GUO, Y. Boltzmann diffusive limit beyond the Navier-Stokes approximation. Comm. Pure Appl. Math. 59, 5 (2006), 626–687. 37, 145, 147, 154, 160, 176, 189, 223, 224, 225, 243
- [57] GUO, Y. Decay and continuity of the Boltzmann equation in bounded domains. Arch. Ration. Mech. Anal. 197, 3 (2010), 713–809. 59, 65, 101
- [58] HUANG, K. Statistical mechanics. John Wiley & Sons Inc., New York, 1963. 261
- [59] HWANG, H. J. Regularity for the Vlasov-Poisson system in a convex domain. SIAM J. Math. Anal. 36, 1 (2004), 121–171 (electronic). 28, 59, 69, 71
- [60] KAC, M. Foundations of kinetic theory. In Proceedings of the Third Berkeley Symposium on Mathematical Statistics and Probability, 1954–1955, vol. III (1956), University of California Press, Berkeley and Los Angeles, pp. 171–197. 130

- [61] KATO, T. Perturbation theory for linear operators. Classics in Mathematics. Springer-Verlag, Berlin, 1995. Reprint of the 1980 edition. 162, 229
- [62] KIRKWOOD, J. G. The Statistical Mechanical Theory of Transport Processes I. General Theory. The Journal of Chemical Physics 14, 3 (1945). 128
- [63] KIRKWOOD, J. G. Errata: Statistical Mechanics of Transport Processes I. General Theory. The Journal of Chemical Physics 14, 5 (1946). 128
- [64] LACHOWICZ, M. On the initial layer and the existence theorem for the nonlinear Boltzmann equation. Math. Methods Appl. Sci. 9, 3 (1987), 342–366. 133
- [65] LANFORD, III, O. E. Time evolution of large classical systems. In Dynamical systems, theory and applications (Recontres, Battelle Res. Inst., Seattle, Wash., 1974).
  Springer, Berlin, 1975, pp. 1–111. Lecture Notes in Phys., Vol. 38. 19, 130, 143, 221, 259
- [66] LERAY, J. Sur le mouvement d'un liquide visqueux emplissant l'espace. Acta Math. 63, 1 (1934), 193–248. 38, 137, 146, 223
- [67] LIONS, P.-L. Mathematical topics in fluid mechanics. Vol. 1, vol. 3 of Oxford Lecture Series in Mathematics and its Applications. The Clarendon Press Oxford University Press, New York, 1996. Incompressible models, Oxford Science Publications. 148
- [68] LU, X. Conservation of energy, entropy identity, and local stability for the spatially homogeneous Boltzmann equation. J. Statist. Phys. 96, 3-4 (1999), 765–796. 263
- [69] LU, X. A modified Boltzmann equation for Bose-Einstein particles: isotropic solutions and long-time behavior. J. Statist. Phys. 98, 5-6 (2000), 1335–1394. 44, 262, 263
- [70] LU, X. On isotropic distributional solutions to the Boltzmann equation for Bose-Einstein particles. J. Statist. Phys. 116, 5-6 (2004), 1597–1649. 44, 262, 263
- [71] LU, X. The Boltzmann equation for Bose-Einstein particles: velocity concentration and convergence to equilibrium. J. Stat. Phys. 119, 5-6 (2005), 1027–1067. 44, 46, 262, 263, 266
- [72] MAJDA, A. J., AND BERTOZZI, A. L. Vorticity and incompressible flow, vol. 27 of Cambridge Texts in Applied Mathematics. Cambridge University Press, Cambridge, 2002. 148
- [73] MASMOUDI, N., AND SAINT-RAYMOND, L. From the Boltzmann equation to the Stokes-Fourier system in a bounded domain. *Comm. Pure Appl. Math.* 56, 9 (2003), 1263–1293. 130

- [74] MATSUMURA, A., AND NISHIDA, T. Initial-boundary value problems for the equations of motion of compressible viscous and heat-conductive fluids. *Comm. Math. Phys.* 89, 4 (1983), 445–464. 148
- [75] MAXWELL, J. C. On the Dynamical Theory of Gases. Philosophical Transactions of the Royal Society of London 157 (1867), 49–88. 123
- [76] MISCHLER, S., AND MOUHOT, C. Kac's program in kinetic theory. Invent. Math. 193, 1 (2013), 1–147. 130
- [77] MISCHLER, S., AND WENNBERG, B. On the spatially homogeneous Boltzmann equation. Ann. Inst. H. Poincaré Anal. Non Linéaire 16, 4 (1999), 467–501. 45, 263, 264, 266, 267, 268, 270, 271, 276, 277, 278, 291, 301
- [78] MOUHOT, C. Quantitative lower bounds for the full Boltzmann equation. I. Periodic boundary conditions. Comm. Partial Differential Equations 30, 4-6 (2005), 881–917.
  28, 56, 57, 58, 62, 82, 83, 88, 89, 109, 110, 111, 114, 115
- [79] MOUHOT, C. Explicit coercivity estimates for the linearized Boltzmann and Landau operators. Comm. Partial Differential Equations 31, 7-9 (2006), 1321–1348. 35, 189, 223
- [80] MOUHOT, C. Rate of convergence to equilibrium for the spatially homogeneous Boltzmann equation with hard potentials. *Communications in Mathematical Physics* 261 (2006), 629–672. 231
- [81] MOUHOT, C. Quelques résultats d'hypocoercitivité en théorie cinétique collisionnelle. In Séminaire: Équations aux Dérivées Partielles. 2007–2008, Sémin. Équ. Dériv. Partielles. École Polytech., Palaiseau, 2009, pp. Exp. No. XVI, 21. 147, 149
- [82] MOUHOT, C., AND NEUMANN, L. Quantitative perturbative study of convergence to equilibrium for collisional kinetic models in the torus. *Nonlinearity* 19, 4 (2006), 969–998. 36, 37, 145, 147, 148, 150, 151, 189, 190, 191, 194, 224
- [83] MOUHOT, C., AND STRAIN, R. M. Spectral gap and coercivity estimates for linearized Boltzmann collision operators without angular cutoff. J. Math. Pures Appl. (9) 87, 5 (2007), 515–535. 193
- [84] MOUHOT, C., AND VILLANI, C. Regularity theory for the spatially homogeneous Boltzmann equation with cut-off. Arch. Ration. Mech. Anal. 173, 2 (2004), 169–212.
   246
- [85] NAVIER, C.-L. Mémoire sur les lois du mouvement des fluides. 1827. 122

- [86] NEUMANN, L., AND SCHMEISER, C. Convergence to global equilibrium for a kinetic fermion model. SIAM J. Math. Anal. 36, 5 (2005), 1652–1663 (electronic). 189
- [87] NEWTON, I. Philosophiae naturalis principia mathematica. Vol. I. Harvard University Press, Cambridge, Mass., 1972. Reprinting of the third edition (1726) with variant readings, Assembled and edited by Alexandre Koyré and I. Bernard Cohen with the assistance of Anne Whitman. 121
- [88] NISHIDA, T., AND IMAI, K. Global solutions to the initial value problem for the nonlinear Boltzmann equation. Publ. Res. Inst. Math. Sci. 12, 1 (1976/77), 229–239. 147
- [89] NORDHEIM, L. On the kinetic method in the new statistics and its application in the electron theory of conductivity. Proc. Roy. Soc. London Ser. A 119 (1928), 689. 43, 259
- [90] NOVOTNÝ, A., AND STRAŠKRABA, I. Convergence to equilibria for compressible Navier-Stokes equations with large data. Ann. Mat. Pura Appl. (4) 179 (2001), 263–287. 148
- [91] OLLA, S., VARADHAN, S. R. S., AND YAU, H.-T. Hydrodynamical limit for a Hamiltonian system with weak noise. *Comm. Math. Phys.* 155, 3 (1993), 523–560. 125
- [92] POINCARÉ, H. *Œuvres. Tome VII.* Les Grands Classiques Gauthier-Villars. [Gauthier-Villars Great Classics]. Éditions Jacques Gabay, Sceaux, 1996. Masses fluides en rotation. Principes de mécanique analytique. Problème des trois corps. [Rotating fluid masses. Principles of analytic mechanics. Three-body problem], With a preface by Jacques Lévy, Reprint of the 1952 edition. 121
- [93] PORITSKY, H. The billiard ball problem on a table with a convex boundary—an illustrative dynamical problem. Ann. of Math. (2) 51 (1950), 446–470. 28, 59
- [94] POVZNER, A. J. On the Boltzmann equation in the kinetic theory of gases. Mat. Sb. (N.S.) 58 (100) (1962), 65-86. 45, 264, 266
- [95] PULVIRENTI, A., AND WENNBERG, B. A Maxwellian lower bound for solutions to the Boltzmann equation. *Comm. Math. Phys.* 183, 1 (1997), 145–160. 27, 28, 56, 57, 58, 65, 82, 83, 88
- [96] PULVIRENTI, M., SAFFIRIO, C., AND SIMONELLA, S. On the validity of the Boltzmann equation for short range potentials. Preprint. 19, 128, 130, 259
- [97] QUASTEL, J., AND YAU, H.-T. Lattice gases, large deviations, and the incompressible Navier-Stokes equations. Ann. of Math. (2) 148, 1 (1998), 51–108. 125

- [98] SAINT-RAYMOND, L. Hydrodynamic limits of the Boltzmann equation, vol. 1971 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2009. 31, 32, 125, 130, 131, 132, 144, 222
- [99] SERRIN, J. On the interior regularity of weak solutions of the Navier-Stokes equations. Arch. Rational Mech. Anal. 9 (1962), 187–195. 148
- [100] SERRIN, J. The initial value problem for the Navier-Stokes equations. In Nonlinear Problems (Proc. Sympos., Madison, Wis., 1962). Univ. of Wisconsin Press, Madison, Wis., 1963, pp. 69–98. 148
- [101] SIDERIS, T. C. Formation of singularities in three-dimensional compressible fluids. Comm. Math. Phys. 101, 4 (1985), 475–485. 133
- [102] SPOHN, H. Kinetics of the Bose-Einstein condensation. Phys. D 239, 10 (2010), 627–634. 262, 263
- [103] STOKES, G. G. Mathematical and physical papers. Volume 1. Cambridge Library Collection. Cambridge University Press, Cambridge, 2009. Reprint of the 1880 original. 123
- [104] TABACHNIKOV, S. Billiards. Panor. Synth., 1 (1995), vi+142. 28, 29, 59, 70, 71, 75
- [105] TABACHNIKOV, S. Geometry and billiards, vol. 30 of Student Mathematical Library. American Mathematical Society, Providence, RI, 2005. 28, 59, 70
- [106] TEMAM, R. Navier-Stokes equations. AMS Chelsea Publishing, Providence, RI, 2001. Theory and numerical analysis, Reprint of the 1984 edition. 148
- [107] UKAI, S. On the existence of global solutions of mixed problem for non-linear Boltzmann equation. Proc. Japan Acad. 50 (1974), 179–184. 36, 147, 162, 224
- [108] UKAI, S. Les solutions globales de l'équation de Boltzmann dans l'espace tout entier et dans le demi-espace. C. R. Acad. Sci. Paris Sér. A-B 282, 6 (1976), Ai, A317–A320. 36
- [109] UKAI, S. The incompressible limit and the initial layer of the compressible Euler equation. J. Math. Kyoto Univ. 26, 2 (1986), 323–331. 148
- [110] UKAI, S., AND YANG, T. Mathematical theory of the Boltzmann equation. Lecture Notes Series, no. 8, Liu Bie Ju Centre for Mathematical Sciences, City University of Hong Kong, 2006. 224
- [111] VILLANI, C. Limites hydrodynamiques de l'équation de Boltzmann (d'après C. Bardos, F. Golse, C. D. Levermore, P.-L. Lions, N. Masmoudi, L. Saint-Raymond).

*Astérisque*, 282 (2002), Exp. No. 893, ix, 365–405. Séminaire Bourbaki, Vol. 2000/2001. 31, 32, 130, 135, 138, 143, 144, 222

- [112] VILLANI, C. A review of mathematical topics in collisional kinetic theory. In Handbook of mathematical fluid dynamics, Vol. I. North-Holland, Amsterdam, 2002, pp. 71–305. 15, 20, 21, 26, 27, 57, 62, 193, 227, 229
- [113] VILLANI, C. Cercignani's conjecture is sometimes true and always almost true. Comm. Math. Phys. 234, 3 (2003), 455–490. 27, 57
- [114] YU, H. Global classical solutions of the Boltzmann equation near Maxwellians. Acta Math. Sci. Ser. B Engl. Ed. 26, 3 (2006), 491–501. 147, 224
- [115] YVON, J. La théorie statistique des fluides et l'équation d'état. Actualités scientifiques et industrielles. Hermann & cie, 1935. 128

MARC BRIANT

UNIVERSITY OF CAMBRIDGE, CCA-DPMMS, CENTRE FOR MATHEMATICAL SCIENCES WILBERFORCE ROAD, CAMBRIDGE CB3 0WA, UK

E-MAIL: briant.maths@gmail.com